

13 December 2018

Page: 1/154

Original: English

BRAZIL - CERTAIN MEASURES CONCERNING TAXATION AND CHARGES

AB-2017-7 AB-2017-8

Reports of the Appellate Body

(18-7793)

- 2 -

Table of Contents

1	INTRODUCTION	1
2	ARGUMENTS OF THE PARTICIPANTS	8
3	ARGUMENTS OF THE THIRD PARTICIPANTS	8
4	ISSUES RAISED IN THIS APPEAL	8
5	ANALYSIS OF THE APPELLATE BODY	1
5.1	Articles III: 2 and III: 4 of the GATT 19942	1
	1 Whether the Panel erred in finding that imported finished and intermediate ICT products e taxed in excess of like domestic finished and intermediate ICT products inconsistently with cle III: 2, first sentence, of the GATT 1994	2
	2 Whether the Panel erred in finding that the ICT programmes are inconsistent with cle III:4 of the GATT 1994	1
	3 Whether the Panel erred in finding that the ICT programmes are inconsistent with cle 2.1 of the TRIMs Agreement	6
IN(mc	4 Whether the Panel erred in finding that the accreditation requirements under the VAR-AUTO programme are inconsistent with Article III:4 of the GATT 1994 because they are the burdensome for companies seeking accreditation as importers/distributors as opposed to mestic manufacturers	7
	5 Whether the Panel erred in finding that the INOVAR-AUTO programme is inconsistent with cle 2.1 of the TRIMs Agreement	0
5.2	Article III:8(b) of the GATT 19944	1
5.2	1 Introduction	1
5.2	2 The legal standard under Article III:8(b) of the GATT 19944	2
5.2 GA	3 Whether the Panel erred in its interpretation and application of Article III:8(b) of the T 19944	6
5.2	4 Conclusion with respect to Article III:8(b)	5
5.2	5 Separate opinion of one Appellate Body Member on Article III:8(b) of the GATT 19945	6
5.3	Article 3.1(a) of the SCM Agreement	0
5.3	1 Article 1.1(a)(1)(ii) of the SCM Agreement: PEC and RECAP programmes	0
5.4	Article 3.1(b) the SCM Agreement	0
5.4	1 Article 1.1(a)(1)(ii) of the SCM Agreement: ICT programmes	0
5.4	2 Article 3.1(b) the SCM Agreement - import substitution	1
5.4	3 The European Union's and Japan's appeal concerning the in-house scenario	2
5.5	Article I:1 of the GATT 1994 and the Enabling Clause11	1
5.5 uno	1 Whether the Panel erred in finding that the claims raised by the European Union and Japan er Article I:1 of the GATT 1994 were within its terms of reference	1
in f	2 Whether the Panel erred in its interpretation of paragraph 2(b) of the Enabling Clause and nding that the differential tax treatment under the INOVAR-AUTO programme was not justified er that provision	
in f	3 Whether the Panel erred in its interpretation of paragraph 2(c) of the Enabling Clause and nding that the differential tax treatment under the INOVAR-AUTO programme was not justified er that provision	
5.5	4 Conclusion with respect to the Enabling Clause	3
5.5 5.6	4 Conclusion with respect to the Enabling Clause	

- 3 -

5.6.2 Relevant legal standards under Articles 11 and 12.7 of the DSU and Article 4.7 of the SCM Agreement
5.6.3 Whether the Panel acted inconsistently with Articles 11 and 12.7 of the DSU in recommending that Brazil withdraw the prohibited subsidies within 90 days
6 FINDINGS AND CONCLUSIONS
6.1 Articles III: 2 and III: 4 of the GATT 1994 and Article 2.1 of the TRIMs Agreement
6.1.1 Whether the Panel erred in finding that imported finished ICT products were taxed in excess of like domestic finished ICT products inconsistently with Article III:2, first sentence, of the GATT 1994
6.1.2 Whether the Panel erred in finding that imported intermediate ICT products were taxed in excess of like domestic intermediate ICT products inconsistently with Article III: 2, first sentence, of the GATT 1994
6.1.3 Whether the Panel erred in finding that the accreditation requirements under the ICT programmes accord treatment less favourable to imported products than that accorded to like domestic products inconsistently with Article III:4 of the GATT 1994
6.1.4 Whether the Panel erred in finding that the ICT programmes are inconsistent with Article III: 4 of the GATT 1994 by virtue of the lower administrative burden on companies purchasing incentivized domestic intermediate products
6.1.5 Whether the Panel erred in finding that the PPBs and other production-step requirements under the ICT programmes are contingent upon the use of domestic goods, inconsistently with Article III: 4 of the GATT 1994
6.1.6 Whether the Panel erred in finding that the ICT programmes are inconsistent with Article 2.1 of the TRIMs Agreement
6.1.7 Whether the Panel erred in finding that the accreditation requirements under the INOVAR-AUTO programme are inconsistent with Article III:4 of the GATT 1994 because they are more burdensome for companies seeking accreditation as importers/distributors as opposed to domestic manufacturers
6.1.8 Whether the Panel erred in finding that the INOVAR-AUTO programme is inconsistent with Article 2.1 of the TRIMs Agreement
6.2 Article III:8(b) of the GATT 1994
6.3 Article 3.1(a) of the SCM Agreement
6.4 Article 3.1(b) of the SCM Agreement
6.5 Article I: 1 of the GATT 1994 and the Enabling Clause
6.5.1 Whether the Panel erred in finding that the claims raised by the European Union and Japan under Article I:1 of the GATT 1994 were within its terms of reference
6.5.2 Whether the Panel erred in its interpretation of paragraph 2(b) of the Enabling Clause and in finding that the differential tax treatment under the INOVAR-AUTO programme was not justified under that provision
6.5.3 Whether the Panel erred in its interpretation of paragraph 2(c) of the Enabling Clause and in finding that the differential tax treatment under the INOVAR-AUTO programme was not justified under that provision
6.6 Article 4.7 of the SCM Agreement
6.7 Recommendation

- 4 _____

ABBREVIATIONS USED IN THESE REPORTS

Abbrovietion	Description
Abbreviation	Description
Anti-Dumping	Agreement on Implementation of Article VI of the General Agreement on Tariffs
Agreement	and Trade 1994
CIDE	Contribution of Intervention in the Economic Domain
COFINS	Contribution to Social Security Financing
COFINS-Importation	Contribution to Social Security Financing applicable to Imports of Goods or Services
DSB	Dispute Settlement Body
DSU	Understanding on Rules and Procedures Governing the Settlement of Disputes
ECAs	economic complementation agreements
ECA No. 2	Economic Complementation Agreement No. 2 between Brazil and Uruguay
ECA No. 14	Economic Complementation Agreement No. 14 between Argentina and Brazil
ECA No. 55	Economic Complementation Agreement No. 55 between MERCOSUR and Mexico
Enabling Clause	Decision on Differential and More Favourable Treatment, Reciprocity, and Fuller Participation of Developing Countries, Decision of 28 November 1979
European Union's	Request for the Establishment of a Panel by the European Union, WT/DS472/5
panel request	,
GATT 1994	General Agreement on Tariffs and Trade 1994
GSP	Generalized System of Preferences
ICT	Information and Communication Technology
ICT products	Information and Communication Technology, Automation and Related Goods
ICT programme	tax treatment established under the Informatics, PADIS, PATVD, and Digital
ror programme	Inclusion programmes
INMETRO	National Institute of Metrology, Quality and Technology
INOVAR-AUTO	programme of Incentive to the Technological Innovation and Densification of the
programme	Automotive Supply Chain
IPI tax	Tax on Industrialised Products
Japan's panel request	Request for the Establishment of a Panel by Japan, WT/DS497/3
LAIA	Latin American Integration Association
MDIC MERCOSUR	Ministry of Development, Industry and Trade Mercado Común del Sur (Southern Common Market)
MFN	most-favoured nation
NTMS	non-tariff measures
PADIS programme	programme of Incentives for the Semiconductors Sector
PASEP	Civil Service Asset Formation Programme
PATVD	Programme of Support to the Technological Development of the Industry of Digital TV Equipment
PEC programme	regime for Predominantly Exporting Companies
PIS	Social Integration Programme
PIS/PASEP	Social Integration Programme/Civil Service Asset Formation Programme
PIS/PASEP-	Social Integration and Civil Service Asset Formation Programmes contribution
Importation	applicable to Imports of Foreign Goods or Services
PPB	Basic Productive Process
PTA	preferential trade arrangement
R&D	research and development
RECAP programme	Special Regime for the Purchase of Capital Goods for Exporting Enterprises
REIMCOMP	Special Regime to Incentive Computers for Educational Use
RETAERO	Special Regime for the Brazilian Aerospace Industry
RTA	regional trade agreement
S&D	special and differential treatment
SCM Agreement	Agreement on Subsidies and Countervailing Measures
TRIMs Agreement	Agreement on Trade-Related Investment Measures
Vienna Convention	Vienna Convention on the Law of Treaties, Done at Vienna, 23 May 1969,
	1155 UNTS 331; 8 International Legal Materials 679
Working Procedures	Working Procedures for Appellate Review
WTO	World Trade Organization

PANEL EXHIBITS CITED IN THESE REPORTS

Exhibit No.	Short Title (if any)	Full Case Title and Citation
BRA-93	1980 Treaty of Montevideo	Treaty of Montevideo, concluded at Montevideo on 12 August 1980
BRA-100	Law 10,637/2002	Law 10,637 of 30 December 2002
BRA-114		Communication from LAIA to the Chair of the Trade and Development Committee of the WTO, 19 May 2016 (original version)

Exhibit	Short Title	Full Case Title and Citation
No.	(if any)	
BRA-115		Communication from LAIA to the Chair of the Trade and Development Committee of the WTO, 19 May 2016 (translated version)
JE-7	Decree 5,906/2006	Decree 5,906 of 26 September 2006 (regulating Articles 4, 9, 11, and 16-A of Law 8,248 of 23 October 1991) (Informatics Law)
JE-8	Decree 6,759/2009	Decree 6,759 of 5 February 2009 governing the administration of customs activities and the inspection, control and taxation of foreign trade operations
JE-9	Decree 7,212/2010	Decree 7,212 of 15 June 2010 regulating the charging, inspection, collection and administration of the Industrial Goods Tax (IPI)
JE-31	Interministerial Implementing Order 93/2013	Interministerial Implementing Order (<i>Portaria</i>) 93 of 1 April 2013 establishing a Basic Production process for optical splice closures produced in Brazil
JE-32	Interministerial Implementing Order 103/2013	Interministerial Implementing Order (<i>Portaria</i>) 103 of 2 April 2013 on the Basic Production Process for products for speed alarms, tracking and control produced in Brazil
JE-71	Law 11,484/2007	Law 11,484, of 31 May 2007 providing for the incentives to the Digital TV equipment and electronic semiconductor components industries and on the intellectual property protection of integrated circuit topographies, establishing the Semiconductor Technological Development Support Program (PADIS) and Support Program for the Technological Development of the Digital TV Equipment Industry (PATVD)
JE-73	Decree 6,233/2007	Decree 6,233, of 11 October 2007 establishing the criteria for the purpose of accreditation to the Semiconductor Technological Development Support Program (PADIS), which grants income tax exemption and reduces PIS/PASEP, COFINS and IPI contributions to zero
JE-75	Decree 8,247/2014	Decree 8,247 of 23 May 2014, amending Decree 6,233/2007
JE-76	Regulatory Instruction RFB 852/2008	Regulatory Instruction RFB 852 of 13 June 2008 laying down procedures for qualification for the Programme for the Technological Development of the Semiconductor Industry (PADIS) by Secretary of the Federal Revenue Department of Brazil
JE-94	Law 10,637/2002	Law 10,637 of 30 December 2002
JE-95	Law 12,715/2012	Law 12,715 of 17 September 2012
JE-132	Decree 7,819/2012	Decree 7,819 of 3 October 2012 regulating Articles 40 to 44 of Law 12,715 of 17 September 2012, which contain provisions on the Incentive Scheme for Technological Innovation and Consolidation of the Automotive Supply Chain (INOVAR-AUTO) and Articles 5 and 6 of Law 12,546 of 14 December 2011, which contain provisions on a reduction of the Industrial Goods Tax, as the case dictates
JE-163 (rev)	Decree 350/1991	Decree 350 of 21 November 1991 on the Treaty for the Formation of a Common Market between the Republic of Argentina, the Federative Republic of Brazil, the Republic of Paraguay and the Oriental Republic of Uruguay (MERCOSUR)
JE-164 (rev)	Decree 4,458/2002	Decree 4,458 of 5 November 2002 on the Economic Complementation Agreement No. 55 between MERCOSUR and the United Mexican States
JE-165 (rev)	Decree 6,500/2008	Decree 6,500 of 2 July 2008 on the Economic Complementation Agreement No. 14 between the Republic of Argentina and the Federative Republic of Brazil
JE-181	Law 10,865/2004	Law 10,865 of 30 April 2004
JE-182	Law 11,196/2005	Law 11,196 of 21 November 2005 (RECAP)
JE-186		Exporters 2014 study on Brazilian exports – delays in refunds (original version)
JE-191	Normative Instruction RFB 948/2009	Normative Instruction (Instrução Normativa) RFB 948 of 15 June 2009
JE-193	Normative Instruction SRF 595/2005	Normative Instruction SRF 595 of 27 December 2005
JE-203 (rev)	Decree 6,518/2008	Decree 6,518 of 30 July 2008 providing for the implementation of the 68th Additional Protocol to Economic Complementation Agreement No. 2, signed between the Governments of the Federative Republic of Brazil and the Oriental Republic of Uruguay
JE-204 (rev)	Decree 7,658/2011	Decree 7,658 of 23 December 2011 laying down provisions concerning the implementation of the 69th Additional Protocol to Economic Complementation Agreement No. 2 signed by the Governments of the Federal Republic of Brazil and the Oriental Republic of Uruguay

- 5 -

- 6 -

CASES CITED IN THESE REPORTS

Short Title	Full Case Title and Citation
Argentina – Hides and Leather	Panel Report, Argentina – Measures Affecting the Export of Bovine Hides and the Import of Finished Leather, WT/DS155/R and Corr.1, adopted 16 February 2001, DSR 2001:V, p. 1779
Australia – Automotive Leather II	Panel Report, Australia – Subsidies Provided to Producers and Exporters of Automotive Leather, WT/DS126/R, adopted 16 June 1999, DSR 1999:III, p. 951
Australia – Salmon	Appellate Body Report, Australia - Measures Affecting Importation of Salmon, WT/DS18/AB/R, adopted 6 November 1998, DSR 1998:VIII, p. 3327
Brazil – Aircraft	Appellate Body Report, <i>Brazil – Export Financing Programme for Aircraft</i> , WT/DS46/AB/R, adopted 20 August 1999, DSR 1999:III, p. 1161
Brazil – Aircraft	Panel Report, <i>Brazil – Export Financing Programme for Aircraft</i> , WT/DS46/R, adopted 20 August 1999, as modified by Appellate Body Report WT/DS46/AB/R, DSR 1999:III, p. 1221
Brazil – Retreaded Tyres	Appellate Body Report, <i>Brazil – Measures Affecting Imports of Retreaded Tyres</i> , WT/DS332/AB/R, adopted 17 December 2007, DSR 2007:IV, p. 1527
Brazil – Taxation	Panel Reports, <i>Brazil – Certain Measures Concerning Taxation and Charges</i> , WT/DS472/R, Add.1 and Corr.1 / WT/DS497/R, Add.1 and Corr.1, circulated to WTO Members 30 August 2017
Canada – Aircraft	Appellate Body Report, <i>Canada – Measures Affecting the Export of Civilian Aircraft</i> , WT/DS70/AB/R, adopted 20 August 1999, DSR 1999:III, p. 1377
Canada – Aircraft	Panel Report, <i>Canada – Measures Affecting the Export of Civilian Aircraft</i> , WT/DS70/R, adopted 20 August 1999, upheld by Appellate Body Report WT/DS70/AB/R, DSR 1999: IV, p. 1443
Canada – Aircraft (Article 21.5 – Brazil)	Appellate Body Report, <i>Canada – Measures Affecting the Export of Civilian Aircraft – Recourse by Brazil to Article 21.5 of the DSU</i> , WT/DS70/AB/RW, adopted 4 August 2000, DSR 2000:1X, p. 4299
Canada – Aircraft Credits and Guarantees	Panel Report, <i>Canada – Export Credits and Loan Guarantees for Regional Aircraft</i> , WT/DS222/R and Corr.1, adopted 19 February 2002, DSR 2002:III, p. 849
Canada – Autos	Appellate Body Report, <i>Canada – Certain Measures Affecting the Automotive Industry</i> , WT/DS139/AB/R, WT/DS142/AB/R, adopted 19 June 2000, DSR 2000:VI, p. 2985
Canada – Autos	Panel Report, <i>Canada – Certain Measures Affecting the Automotive Industry</i> , WT/DS139/R, WT/DS142/R, adopted 19 June 2000, as modified by Appellate Body Report WT/DS139/AB/R, WT/DS142/AB/R, DSR 2000:VII, p. 3043
Canada – FIRA	GATT Panel Report, Canada – Administration of the Foreign Investment Review Act, L/5504, adopted 7 February 1984, BISD 30S, p. 140
Canada – Periodicals	Appellate Body Report, <i>Canada – Certain Measures Concerning Periodicals</i> , WT/DS31/AB/R, adopted 30 July 1997, DSR 1997:1, p. 449
Canada – Renewable Energy / Canada – Feed-in Tariff Program	Appellate Body Reports, <i>Canada – Certain Measures Affecting the Renewable Energy Generation Sector / Canada – Measures Relating to the Feed-in Tariff Program</i> , WT/DS412/AB/R / WT/DS426/AB/R, adopted 24 May 2013, DSR 2013:1, p. 7
Chile – Alcoholic Beverages	Appellate Body Report, <i>Chile – Taxes on Alcoholic Beverages</i> , WT/DS87/AB/R, WT/DS110/AB/R, adopted 12 January 2000, DSR 2000:1, p. 281
Chile – Price Band System (Article 21.3(c))	Award of the Arbitrator, <i>Chile – Price Band System and Safeguard Measures</i> <i>Relating to Certain Agricultural Products – Arbitration under Article 21.3(c) of the</i> <i>DSU</i> , WT/DS207/13, 17 March 2003, DSR 2003:111, p. 1237
Chile – Price Band System (Article 21.5 – Argentina)	Appellate Body Report, <i>Chile – Price Band System and Safeguard Measures Relating to Certain Agricultural Products – Recourse to Article 21.5 of the DSU by Argentina</i> , WT/DS207/AB/RW, adopted 22 May 2007, DSR 2007:11, p. 513
China – HP-SSST (Japan) / China – HP-SSST (EU)	Appellate Body Reports, <i>China – Measures Imposing Anti-Dumping Duties on</i> <i>High-Performance Stainless Steel Seamless Tubes ("HP-SSST") from Japan /</i> <i>China – Measures Imposing Anti-Dumping Duties on High-Performance Stainless</i> <i>Steel Seamless Tubes ("HP-SSST") from the European Union</i> , WT/DS454/AB/R and Add.1 / WT/DS460/AB/R and Add.1, adopted 28 October 2015, DSR 2015:IX, p. 4573

- 7 -

Short Title	Full Case Title and Citation
China – Publications and Audiovisual Products	Appellate Body Report, <i>China – Measures Affecting Trading Rights and Distribution Services for Certain Publications and Audiovisual Entertainment Products</i> , WT/DS363/AB/R, adopted 19 January 2010, DSR 2010:1, p. 3
China – Rare Earths	Appellate Body Reports, <i>China – Measures Related to the Exportation of Rare Earths, Tungsten, and Molybdenum</i> , WT/DS431/AB/R / WT/DS432/AB/R / WT/DS433/AB/R, adopted 29 August 2014, DSR 2014:III, p. 805
Colombia – Textiles	Appellate Body Report, <i>Colombia – Measures Relating to the Importation of Textiles, Apparel and Footwear</i> , WT/DS461/AB/R and Add.1, adopted 22 June 2016, DSR 2016:III, p. 1131
Dominican Republic – Import and Sale of Cigarettes	Appellate Body Report, <i>Dominican Republic – Measures Affecting the Importation and Internal Sale of Cigarettes</i> , WT/DS302/AB/R, adopted 19 May 2005, DSR 2005:XV, p. 7367
EC – Asbestos	Appellate Body Report, <i>European Communities – Measures Affecting Asbestos and Asbestos-Containing Products</i> , WT/DS135/AB/R, adopted 5 April 2001, DSR 2001:VII, p. 3243
EC – Bananas III (Article 21.5 – Ecuador II) / EC – Bananas III (Article 21.5 – US)	Appellate Body Reports, <i>European Communities – Regime for the Importation,</i> <i>Sale and Distribution of Bananas – Second Recourse to Article 21.5 of the DSU by</i> <i>Ecuador</i> , WT/DS27/AB/RW2/ECU, adopted 11 December 2008, and Corr.1 / <i>European Communities – Regime for the Importation, Sale and Distribution of</i> <i>Bananas – Recourse to Article 21.5 of the DSU by the United States,</i> WT/DS27/AB/RW/USA and Corr.1, adopted 22 December 2008, DSR 2008:XVIII, p. 7165
EC – Bed Linen (Article 21.5 – India)	Appellate Body Report, <i>European Communities – Anti-Dumping Duties on Imports of Cotton-Type Bed Linen from India – Recourse to Article 21.5 of the DSU by India</i> , WT/DS141/AB/RW, adopted 24 April 2003, DSR 2003:III, p. 965
EC – Chicken Cuts (Article 21.3(c))	Award of the Arbitrator, European Communities – Customs Classification of Frozen Boneless Chicken Cuts – Arbitration under Article 21.3(c) of the DSU, WT/DS269/13, WT/DS286/15, 20 February 2006
EC – Export Subsidies on Sugar (Article 21.3(c))	Award of the Arbitrator, <i>European Communities – Export Subsidies on Sugar – Arbitration under Article 21.3(c) of the DSU</i> , WT/DS265/33, WT/DS266/33, WT/DS283/14, 28 October 2005, DSR 2005:XXIII, p. 11581
EC – Fasteners (China)	Appellate Body Report, European Communities – Definitive Anti-Dumping Measures on Certain Iron or Steel Fasteners from China, WT/DS397/AB/R, adopted 28 July 2011, DSR 2011:VII, p. 3995
EC – Hormones	Appellate Body Report, <i>EC Measures Concerning Meat and Meat Products</i> (<i>Hormones</i>), WT/DS26/AB/R, WT/DS48/AB/R, adopted 13 February 1998, DSR 1998:1, p. 135
EC – Poultry	Appellate Body Report, <i>European Communities – Measures Affecting the Importation of Certain Poultry Products</i> , WT/DS69/AB/R, adopted 23 July 1998, DSR 1998: V, p. 2031
EC – Sardines	Appellate Body Report, <i>European Communities – Trade Description of Sardines</i> , WT/DS231/AB/R, adopted 23 October 2002, DSR 2002:VIII, p. 3359
EC – Seal Products	Appellate Body Reports, <i>European Communities – Measures Prohibiting the Importation and Marketing of Seal Products</i> , WT/DS400/AB/R / WT/DS401/AB/R, adopted 18 June 2014, DSR 2014:1, p. 7
EC – Selected Customs Matters	Appellate Body Report, <i>European Communities – Selected Customs Matters</i> , WT/DS315/AB/R, adopted 11 December 2006, DSR 2006:IX, p. 3791
EC – Tariff Preferences	Appellate Body Report, <i>European Communities – Conditions for the Granting of Tariff Preferences to Developing Countries</i> , WT/DS246/AB/R, adopted 20 April 2004, DSR 2004:111, p. 925
EC – Tube or Pipe Fittings	Appellate Body Report, <i>European Communities – Anti-Dumping Duties on Malleable Cast Iron Tube or Pipe Fittings from Brazil</i> , WT/DS219/AB/R, adopted 18 August 2003, DSR 2003:VI, p. 2613
EC and certain member States – Large Civil Aircraft	Appellate Body Report, <i>European Communities and Certain Member States</i> – <i>Measures Affecting Trade in Large Civil Aircraft</i> , WT/DS316/AB/R, adopted 1 June 2011, DSR 2011:1, p. 7
EC and certain member States – Large Civil Aircraft	Panel Report, European Communities and Certain Member States – Measures Affecting Trade in Large Civil Aircraft, WT/DS316/R, adopted 1 June 2011, as modified by Appellate Body Report WT/DS316/AB/R, DSR 2011:11, p. 685

- 8 -

Short Title	Full Case Title and Citation
EC and certain member States – Large Civil Aircraft (Article 21.5 – US)	Appellate Body Report, European Communities and Certain Member States – Measures Affecting Trade in Large Civil Aircraft – Recourse to Article 21.5 of the DSU by the United States, WT/DS316/AB/RW and Add.1, adopted 28 May 2018
EC and certain member States – Large Civil Aircraft (Article 21.5 – US)	Panel Report, European Communities and Certain Member States – Measures Affecting Trade in Large Civil Aircraft – Recourse to Article 21.5 of the DSU by the United States, WT/DS316/RW and Add.1, adopted 28 May 2018, as modified by Appellate Body Report WT/DS316/AB/RW
Guatemala – Cement I	Appellate Body Report, <i>Guatemala – Anti-Dumping Investigation Regarding</i> <i>Portland Cement from Mexico</i> , WT/DS60/AB/R, adopted 25 November 1998, DSR 1998:IX, p. 3767
India – Agricultural Products	Panel Report, India – Measures Concerning the Importation of Certain Agricultural Products, WT/DS430/R and Add.1, adopted 19 June 2015, as modified by Appellate Body Report WT/DS430/AB/R, DSR 2015:V, p. 2663
Indonesia – Autos	Panel Report, Indonesia – Certain Measures Affecting the Automobile Industry, WT/DS54/R, WT/DS55/R, WT/DS59/R, WT/DS64/R, Corr.1 and Corr.2, adopted 23 July 1998, and Corr.3 and Corr.4, DSR 1998:VI, p. 2201
Indonesia – Import Licensing Regimes	Appellate Body Report, Indonesia – Importation of Horticultural Products, Animals and Animal Products, WT/DS477/AB/R, WT/DS478/AB/R, and Add.1, adopted 22 November 2017
Italy – Agricultural Machinery	GATT Panel Report, Italian Discrimination Against Imported Agricultural Machinery, L/833, adopted 23 October 1958, BISD 7S, p. 60
Japan – Agricultural Products II	Appellate Body Report, <i>Japan – Measures Affecting Agricultural Products</i> , WT/DS76/AB/R, adopted 19 March 1999, DSR 1999:1, p. 277
Japan – Alcoholic Beverages II	Appellate Body Report, <i>Japan – Taxes on Alcoholic Beverages</i> , WT/DS8/AB/R, WT/DS10/AB/R, WT/DS11/AB/R, adopted 1 November 1996, DSR 1996:I, p. 97
Japan – Apples	Appellate Body Report, <i>Japan – Measures Affecting the Importation of Apples</i> , WT/DS245/AB/R, adopted 10 December 2003, DSR 2003:IX, p. 4391
Korea – Alcoholic Beverages	Appellate Body Report, <i>Korea – Taxes on Alcoholic Beverages</i> , WT/DS75/AB/R, WT/DS84/AB/R, adopted 17 February 1999, DSR 1999:1, p. 3
Korea – Alcoholic Beverages (Article 21.3(c))	Award of the Arbitrator, <i>Korea – Taxes on Alcoholic Beverages – Arbitration under Article 21.3(c) of the DSU</i> , WT/DS75/16, WT/DS84/14, 4 June 1999, DSR 1999:II, p. 937
Korea – Commercial Vessels	Panel Report, <i>Korea – Measures Affecting Trade in Commercial Vessels,</i> WT/DS273/R, adopted 11 April 2005, DSR 2005:VII, p. 2749
Korea – Dairy	Appellate Body Report, <i>Korea – Definitive Safeguard Measure on Imports of Certain Dairy Products</i> , WT/DS98/AB/R, adopted 12 January 2000, DSR 2000:1, p. 3
Korea – Various Measures on Beef	Appellate Body Report, <i>Korea – Measures Affecting Imports of Fresh, Chilled and Frozen Beef</i> , WT/DS161/AB/R, WT/DS169/AB/R, adopted 10 January 2001, DSR 2001:1, p. 5
Mexico – Corn Syrup (Article 21.5 – US)	Appellate Body Report, <i>Mexico – Anti-Dumping Investigation of High Fructose Corn</i> <i>Syrup (HFCS) from the United States – Recourse to Article 21.5 of the DSU by the</i> <i>United States</i> , WT/DS132/AB/RW, adopted 21 November 2001, DSR 2001:XIII, p. 6675
Mexico – Taxes on Soft Drinks	Panel Report, <i>Mexico – Tax Measures on Soft Drinks and Other Beverages</i> , WT/DS308/R, adopted 24 March 2006, as modified by Appellate Body Report WT/DS308/AB/R, DSR 2006:1, p. 43
Thailand – Cigarettes (Philippines)	Appellate Body Report, <i>Thailand – Customs and Fiscal Measures on Cigarettes from the Philippines</i> , WT/DS371/AB/R, adopted 15 July 2011, DSR 2011:IV, p. 2203
US – Anti-Dumping Methodologies (China)	Appellate Body Report, United States – Certain Methodologies and Their Application to Anti-Dumping Proceedings Involving China, WT/DS471/AB/R and Add.1, adopted 22 May 2017
US – Carbon Steel	Appellate Body Report, United States – Countervailing Duties on Certain Corrosion-Resistant Carbon Steel Flat Products from Germany, WT/DS213/AB/R and Corr.1, adopted 19 December 2002, DSR 2002:IX, p. 3779

- 9 -

Short Title Full Case Title and Citation US - Carbon Steel Appellate Body Report, United States - Countervailing Measures on Certain Hot-Rolled Carbon Steel Flat Products from India, WT/DS436/AB/R, adopted (India) 19 December 2014, DSR 2014: V, p. 1727 US – Certain EC Appellate Body Report, United States - Import Measures on Certain Products from the European Communities, WT/DS165/AB/R, adopted 10 January 2001, Products DSR 2001:1, p. 373 US - Countervailing Appellate Body Report, United States - Countervailing Measures Concerning Certain Products from the European Communities, WT/DS212/AB/R, adopted Measures on Certain EC Products 8 January 2003, DSR 2003: I, p. 5 US - FSC Appellate Body Report, United States - Tax Treatment for "Foreign Sales Corporations", WT/DS108/AB/R, adopted 20 March 2000, DSR 2000; III, p. 1619 Panel Report, United States - Tax Treatment for "Foreign Sales Corporations", US - FSC WT/DS108/R, adopted 20 March 2000, as modified by Appellate Body Report WT/DS108/AB/R, DSR 2000: IV, p. 1675 US - FSC Appellate Body Report, United States - Tax Treatment for "Foreign Sales (Article 21.5 - EC) Corporations" - Recourse to Article 21.5 of the DSU by the European Communities, WT/DS108/AB/RW, adopted 29 January 2002, DSR 2002: I, p. 55 US - FSC Panel Report, United States - Tax Treatment for "Foreign Sales Corporations" -(Article 21.5 - EC) Recourse to Article 21.5 of the DSU by the European Communities, WT/DS108/RW, adopted 29 January 2002, as modified by Appellate Body Report WT/DS108/AB/RW, DSR 2002:1, p. 119 Appellate Body Report, United States - Measures Affecting the Cross-Border US - Gambling Supply of Gambling and Betting Services, WT/DS285/AB/R, adopted 20 April 2005, DSR 2005: XII, p. 5663 (and Corr.1, DSR 2006: XII, p. 5475) Appellate Body Report, United States - Anti-Dumping Measures on Certain US - Hot-Rolled Steel Hot-Rolled Steel Products from Japan, WT/DS184/AB/R, adopted 23 August 2001, DSR 2001: X, p. 4697 US – Large Civil Appellate Body Report, United States – Measures Affecting Trade in Large Civil Aircraft Aircraft (Second Complaint), WT/DS353/AB/R, adopted 23 March 2012, (2nd complaint) DSR 2012:1, p. 7 US - Large Civil Panel Report, United States - Measures Affecting Trade in Large Civil Aircraft (Second Complaint), WT/DS353/R, adopted 23 March 2012, as modified by Aircraft (2nd Appellate Body Report WT/DS353/AB/R, DSR 2012:11, p. 649 complaint) GATT Panel Report, United States - Measures Affecting Alcoholic and Malt US - Malt Beverages Beverages, DS23/R, adopted 19 June 1992, BISD 39S, p. 206 US - Offset Act (Byrd Panel Report, United States - Continued Dumping and Subsidy Offset Act of 2000, Amendment) WT/DS217/R, WT/DS234/R, adopted 27 January 2003, as modified by Appellate Body Report WT/DS217/AB/R, WT/DS234/AB/R, DSR 2003: II, p. 489 US - Offset Act (Byrd Award of the Arbitrator, United States - Continued Dumping and Subsidy Offset Amendment) Act of 2000 - Arbitration under Article 21.3(c) of the DSU, WT/DS217/14, (Article 21.3(c)) WT/DS234/22, 13 June 2003, DSR 2003: III, p. 1163 US – Oil Country Appellate Body Report, United States – Sunset Reviews of Anti-Dumping Measures Tubular Goods Sunset on Oil Country Tubular Goods from Argentina, WT/DS268/AB/R, adopted 17 December 2004, DSR 2004: VII, p. 3257 Reviews US – Section 211 Appellate Body Report, United States - Section 211 Omnibus Appropriations Act of 1998, WT/DS176/AB/R, adopted 1 February 2002, DSR 2002:11, p. 589 Appropriations Act US - Section 337 GATT Panel Report. United States Section 337 of the Tariff Act of 1930. L/6439. adopted 7 November 1989, BISD 36S, p. 345 Tariff Act Appellate Body Report, United States - Final Countervailing Duty Determination US - Softwood with Respect to Certain Softwood Lumber from Canada, WT/DS257/AB/R, adopted Lumber IV 17 February 2004, DSR 2004: II, p. 571 US - Softwood Appellate Body Report, United States - Investigation of the International Trade Commission in Softwood Lumber from Canada - Recourse to Article 21.5 of the Lumber VI (Article 21.5 -DSU by Canada, WT/DS277/AB/RW, adopted 9 May 2006, and Corr.1, DSR 2006: XI, p. 4865 Canada) Appellate Body Report, United States - Conditional Tax Incentives for Large Civil US - Tax Incentives Aircraft, WT/DS487/AB/R and Add.1, adopted 22 September 2017

- 10 -

Short Title	Full Case Title and Citation
US – Tax Incentives	Panel Report, United States - Conditional Tax Incentives for Large Civil Aircraft, WT/DS487/R and Add.1, adopted 22 September 2017, as modified by Appellate Body Report WT/DS487/AB/R
US – Tuna II (Mexico)	Appellate Body Report, <i>United States – Measures Concerning the Importation,</i> <i>Marketing and Sale of Tuna and Tuna Products</i> , WT/DS381/AB/R, adopted 13 June 2012, DSR 2012: IV, p. 1837
US – Upland Cotton	Panel Report, <i>United States – Subsidies on Upland Cotton</i> , WT/DS267/R, Add.1 to Add.3 and Corr.1, adopted 21 March 2005, as modified by Appellate Body Report WT/DS267/AB/R, DSR 2005:11, p. 299
US – Upland Cotton (Article 21.5 – Brazil)	Appellate Body Report, United States – Subsidies on Upland Cotton – Recourse to Article 21.5 of the DSU by Brazil, WT/DS267/AB/RW, adopted 20 June 2008, DSR 2008:111, p. 809
US – Wheat Gluten	Appellate Body Report, United States – Definitive Safeguard Measures on Imports of Wheat Gluten from the European Communities, WT/DS166/AB/R, adopted 19 January 2001, DSR 2001:II, p. 717
US – Wool Shirts and Blouses	Appellate Body Report, <i>United States – Measure Affecting Imports of Woven Wool Shirts and Blouses from India</i> , WT/DS33/AB/R, adopted 23 May 1997, and Corr.1, DSR 1997:1, p. 323

- 11 -

World Trade Organization Appellate Body

Brazil – Certain Measures Concerning Taxation and Charges

Brazil, Appellant/Appellee European Union, Other Appellant¹/Appellee²/Third Participant³ Japan, Other Appellant⁴/Appellee⁵/Third Participant⁶

Argentina, *Third Participant* Australia, *Third Participant* Canada, *Third Participant* China, *Third Participant* Colombia, *Third Participant* India, *Third Participant* Korea, *Third Participant* Russia, *Third Participant* Singapore, *Third Participant*⁷ South Africa, *Third Participant*⁸ Chinese Taipei, *Third Participant*⁹ Turkey, *Third Participant* Ukraine, *Third Participant*¹⁰ United States, *Third Participant* AB-2017-7 AB-2017-8

Appellate Body Division:

Graham, Presiding Member Van den Bossche, Member Zhao, Member

1 INTRODUCTION

1.1. Brazil, the European Union, and Japan each appeal certain issues of law and legal interpretations developed in the Panel Reports, *Brazil – Certain Measures Concerning Taxation and Charges*.¹¹

1.2. On 31 October 2014, the European Union requested the establishment of a panel pursuant to Article 6 of the Understanding on Rules and Procedures Governing the Settlement of Disputes (DSU), Article XXIII of the General Agreement on Tariffs and Trade 1994 (GATT 1994), Articles 4.4 and 30 of the Agreement on Subsidies and Countervailing Measures (SCM Agreement), and Article 8 of the Agreement on Trade-Related Investment Measures (TRIMs Agreement) concerning certain tax measures adopted by Brazil through various programmes aimed at the information and communication technology (ICT) and automotive sectors.¹² On 17 December 2014, pursuant to the request of the European Union, a Panel was established to consider this complaint.¹³

1.3. On 17 September 2015, Japan requested the establishment of a panel pursuant to Articles 4.7 and 6 of the DSU, Article XXIII of the GATT 1994, Articles 4.4 and 30 of the SCM Agreement, and Article 8 of the TRIMs Agreement concerning certain tax measures adopted by Brazil through various

- ⁶ In DS472 only.
- ⁷ In DS497 only.
- ⁸ In DS472 only. ⁹ In DS472 only.
- ¹⁰ In DS472 only.

¹¹ WT/DS472/R, WT/DS497/R, 30 August 2017.

¹² Request for the Establishment of a Panel by the European Union, WT/DS472/5 (European Union's panel request).

¹³ Panel Reports, para. 1.5 (referring to the Minutes of the DSB Meeting held on 17 December 2014, WT/DSB/M/353).

¹ In DS472 only.

² In DS472 only.

³ In DS497 only.

⁴ In DS497 only.

⁵ In DS497 only.

programmes aimed at the ICT and automotive sectors.¹⁴ On 28 September 2015, pursuant to the request by Japan, a Panel was established to consider this complaint.¹⁵

1.4. Following the establishment of the Panel upon request from Japan, and pursuant to Article 9.3 of the DSU, the Panels were composed with the same persons and adopted their Joint Working Procedures and joint timetable on 9 October 2015.¹⁶

1.5. The taxes and contributions relevant for the purposes of the present appeals are: (i) the Tax on Industrialised Products (IPI tax)¹⁷; (ii) the Social Integration Programme/Civil Service Asset Formation Programme (PIS/PASEP) contribution and the Contribution to Social Security Financing (COFINS)¹⁸; (iii) the Social Integration and Civil Service Asset Formation Programmes contribution applicable to Imports of Foreign Goods or Services (PIS/PASEP-Importation) and the Contribution to Social Security Financing applicable to Imports of Goods or Services (COFINS-Importation)¹⁹; and (iv) the Contribution of Intervention in the Economic Domain (CIDE).²⁰

1.6. The measures at issue can be divided into three groups of measures through which Brazil provides exemptions, reductions, or suspensions of the federal taxes and contributions mentioned above. The first group of measures concerns the ICT sector and comprises tax treatment granted under: (i) the Informatics programme; (ii) the programme of Incentives for the Semiconductors Sector (PADIS programme); (iii) the programme of Support for the Technological Development of the Industry of Digital TV Equipment (PATVD programme); and (iv) the programme for Digital Inclusion (Digital Inclusion programme).²¹ The second group comprises tax treatment granted under the programme of Incentive to the Technological Innovation and Densification of the Automotive Supply Chain (INOVAR-AUTO programme), which targets the automotive sector. The third group of measures comprises tax treatment granted under: (i) the regime for Predominantly

¹⁸ The PIS/PASEP contribution and the COFINS contribution are Brazilian federal contributions that apply to gross revenues earned by all types of legal entities. These contributions are subject, as a general rule, to the non-cumulative regime and operate as taxes on value added, where companies generate credits in relation to their purchases of taxed products, which they can offset with debits when paying their PIS/PASEP and COFINS liabilities. The system of tax credits ensures that prior-stage PIS/PASEP and COFINS contributions can be deducted at each stage of the supply chain. (Panel Reports, paras. 2.21-2.29)

¹⁹ The PIS/PASEP-Importation and COFINS-Importation contributions are the import-focused variants of the PIS/PASEP and COFINS contributions. These contributions apply to individual import transactions and are levied upon importation of goods. The taxable base is the customs value of the goods. Importers are subject to the non-cumulative regime, so they can offset the amounts paid upon importation with their domestic PIS/PASEP and COFINS liabilities. Importers, therefore, pay the contributions only with respect to the value added, i.e. the difference between the customs value and the importers' sales price. The tax credit mechanism with respect to the non-cumulative PIS/PASEP and COFINS domestic regime applies equally in the case of importers. (Panel Reports, paras. 2.30-2.34)

²⁰ The CIDE contribution is a Brazilian federal contribution that applies to remittances/royalty payments abroad. The tax rate is 10%, and the taxable persons are legal entities holding a license of use or acquiring technological knowledge, as well as those entering into agreements that imply technological transfer, made with persons residing or domiciled abroad. The basis of calculation is the amount paid, credited, delivered, used, or remitted every month to persons residing or domiciled abroad. (Panel Reports, paras. 2.35-2.36)

²¹ Panel Reports, para. 2.37.

¹⁴ Request for the Establishment of a Panel by Japan, WT/DS497/3 (Japan's panel request).

¹⁵ Panel Reports, para. 1.9 (referring to Minutes of the DSB Meeting held on 28 September 2015,

WT/DSB/M/368).

¹⁶ Panel Reports, paras. 1.14-1.15.

¹⁷ The IPI tax is a Brazilian federal tax that applies to all national or foreign industrialized (i.e. manufactured) products. IPI tax rates are product-specific and linked to the price or value of the industrialized product on which it is imposed. In the case of domestic products, the taxable base is the transaction value, while for imported products, the taxable base is the customs value plus the import duties and charges paid. The IPI tax is not paid directly by the entity bearing the ultimate burden of paying the IPI tax. In the case of domestic products, the IPI tax is charged by the industrial establishment selling the industrialized product to the entity that buys the industrialized product. The taxes retained by the industrial establishment selling the industrialized product must be remitted to the Federal Revenue Service on a monthly basis. In the case of imported products, the IPI tax is charged by the customs authorities to the importer of the industrialized product during the customs clearance process. The IPI tax is non-cumulative, and the tax due in each transaction shall be compensated by the amount charged in previous transactions. This is effected by means of a credit system, by which the tax due on products entering the facilities of the taxpayer is deducted from the amount due on the exit of products in the same taxation period, which, in turn, means that when a taxpayer remits to the government the IPI tax collected for any given transaction, the taxpayer may deduct the IPI tax paid in earlier stages of the supply chain, thereby ensuring that the tax is levied only on the value added. (Panel Reports, paras. 2.3-2.20)

Exporting Companies (PEC programme); and (ii) the Special Regime for the Purchase of Capital Goods for Exporting Enterprises (RECAP programme).²²

1.7. The Informatics programme provides for exemptions and reductions on the IPI tax on the sale of information technology goods. It also provides for suspensions of the IPI tax on the purchase or import of raw materials, intermediate goods, and packaging materials used in the production of information technology, and automation goods incentivized under the programme.²³ In order to benefit from the tax treatment, companies must obtain an accreditation.²⁴ The eligible companies under the Informatics programme are companies that: (i) develop or produce information technology and automation goods and services in compliance with the relevant Basic Productive Processes (PPBs)²⁵; and (ii) invest in information technology research and development (R&D) activities in Brazil.²⁶ Moreover, under this programme, products that have obtained the status of "developed in Brazil" are subject to additional tax reductions.²⁷

1.8. The PADIS programme exempts, through zero rates, accredited companies from paying certain taxes with respect to semiconductors and information displays, as well as inputs, tools, equipment, machinery, and software for the production of semiconductors and displays.²⁸ In order to obtain accreditation, legal persons must: (i) invest in R&D in Brazil; and (ii) engage in certain activities in Brazil with respect to semiconductor electronic devices, information displays, and inputs and equipment intended for the manufacture of electronic semiconductor devices and information displays.²⁹

1.9. The PATVD programme exempts accredited companies from paying certain taxes with respect to radio frequency signal transmitting equipment for digital television (digital television transmission equipment), as well as machinery, apparatus, instruments, equipment, inputs, and software for the production of digital television transmission equipment (production goods).³⁰ In order to obtain accreditation, legal persons must: (i) invest in R&D in Brazil; (ii) engage in developing and manufacturing activities of digital television transmission equipment; and (iii) either comply with the relevant PPB or, alternatively, meet the criteria for a product to be considered "developed in Brazil".³¹

1.10. The Digital Inclusion programme exempts, through zero rates, Brazilian retailers from paying PIS/PASEP and COFINS contributions with respect to the sale of certain digital consumer goods produced in Brazil in accordance with the relevant PPBs.³²

1.11. The INOVAR-AUTO programme provides for reduction of the IPI tax burden on certain motor vehicles either: (i) through presumed IPI tax credits granted to accredited companies; or (ii) through reduced IPI tax rates on the importation of vehicles originating in certain countries, as well as on certain domestic vehicles.³³ All companies using presumed IPI tax credits, and certain companies using reduced IPI tax rates, must obtain one of three forms of accreditation: (i) domestic manufacturer; (ii) importer/distributor; or (iii) investor.³⁴ In order to obtain accreditation, a

²⁶ Panel Reports, para. 2.48.

²⁷ Panel Reports, para. 2.67. For an information technology or automation good to be considered as "developed in Brazil", two requirements must be met: (i) products must comply with the specifications, rules, and standards laid out in Brazilian legislation; and (ii) the specifications, projects, and developments must have been carried out in Brazil by technicians of proven skill in such activities who are residents and domiciled in Brazil. (Panel Reports, para. 2.68)

²⁸ Panel Reports, para. 2.71.

- ³² Panel Reports, para. 2.91.
- ³³ Panel Reports, para. 2.97.

²² Panel Reports, para. 2.37.

²³ Panel Reports, para. 2.39.

²⁴ Panel Reports, para. 2.49.

²⁵ The Panel defined a PPB as "the minimum set of operations performed at a manufacturing facility that characterizes the actual industrialization of a given product". (Panel Reports, para. 2.62 (referring to Decree 5,906 of 26 September 2006 (regulating Articles 4, 9, 11, and 16-A of Law 8,248 of 23 October 1991) (Informatics Law) (Decree 5,906/2006) (Panel Exhibit JE-7), Article 16)) In other words, the Panel explained that PPBs indicate the minimum stages or steps of the manufacturing process of a product that must be performed in Brazil. (Ibid.) On appeal, the participants do not take issue with the definition of a PPB set out by the Panel.

²⁹ Panel Reports, para. 2.76.

³⁰ Panel Reports, para. 2.82.

³¹ Panel Reports, para. 2.87. (emphasis omitted)

³⁴ Panel Reports, paras. 2.112-2.113.

- 14 -

company must comply with certain requirements of both a general and specific nature. All such companies must comply with the same two general requirements and also with certain additional specific requirements that vary by the type of accreditation.³⁵ A company applying for accreditation as a domestic manufacturer shall comply with the two general requirements³⁶ as well as "three out of four specific requirements, one of which must be the performance of a minimum number of defined manufacturing and engineering infrastructure activities in Brazil" ³⁷ A company applying for accreditation as importer/distributor shall comply with the two general requirements and "the following three specific requirements: (i) investments in R&D in Brazil; (ii) expenditure on engineering, basic industrial technology and capacity-building of suppliers in Brazil; and, (iii) participation in the vehicle labelling programme by [the] National Institute of Metrology, Quality and Technology (INMETRO)".³⁸ A company applying for accreditation as an investor shall submit to the Ministry of Development, Industry and Trade (MDIC) an investment project containing a description and the technical features of the vehicles to be imported and manufactured. Accreditation shall be granted once the investment project is approved by that Ministry. An investor shall be required to apply for a specific accreditation for every factory, plant, or industrial project that it plans to establish.39

1.12. Under the PEC programme, the IPI tax and the PIS/PASEP, COFINS, PIS/PASEP-Importation, and COFINS-Importation contributions are suspended with respect to raw materials, intermediate goods, and packaging materials purchased by predominantly exporting companies.⁴⁰ Similarly, under the RECAP programme, the PIS/PASEP, COFINS, PIS/PASEP-Importation, and COFINS-Importation contributions are suspended with respect to purchases of new machinery, tools, apparatus, instruments, and equipment by predominantly exporting companies.⁴¹

1.13. Additional factual aspects of the measures at issue and the relevant taxes and contributions are set forth in greater detail in paragraphs 2.1-2.176 of the Panel Reports.

1.14. Before the Panel, the European Union and Japan raised claims, inter alia, under Articles III:2, III:4, and III:5 of the GATT 1994; Article 2.1 of the TRIMs Agreement; and Article 3.1(b) of the SCM Agreement with respect to tax treatment established under the Informatics, PADIS, PATVD, and Digital Inclusion programmes (the ICT programmes) and the INOVAR-AUTO programme. Brazil invoked Article XX(a) of the GATT 1994 to justify certain inconsistencies with respect to the PATVD programme and Articles XX(b) and XX(g) to justify certain inconsistencies with respect to the INOVAR-AUTO programme. The European Union and Japan also raised claims under Article I:1 of the GATT 1994 with respect to the INOVAR-AUTO programme, in response to which Brazil invoked the Decision on Differential and More Favourable Treatment, Reciprocity, and Fuller Participation of Developing Countries, Decision of 28 November 1979 (Enabling Clause) as a defence. With respect to the PEC and RECAP programmes, the European Union and Japan raised claims under Article 3.1(a) of the SCM Agreement.42

1.15. In the Panel Reports, circulated to Members of the World Trade Organization (WTO) on 30 August 2017, the Panel first addressed two broad defences raised by Brazil concerning the ICT and INOVAR-AUTO programmes. First, the Panel rejected Brazil's argument that Article III of the GATT 1994, Article 2 of the TRIMs Agreement, and Article 3.1(b) of the SCM Agreement are inapplicable to "pre-market" measures.⁴³ The Panel found that these provisions are not per se

³⁵ Panel Reports, para. 2.114.

³⁶ The two general requirements that all companies seeking accreditation under the INOVAR-AUTO programme must meet are: (i) compliance with tax obligations at the federal level; and (ii) commitment to achieve certain minimum levels of energy efficiency for products marketed in Brazil. (Panel Reports, para. 2.121)

³⁷ Panel Reports, para. 2.115. (fn omitted)

³⁸ Panel Reports, para. 2.116. (fn omitted)

³⁹ Panel Reports, para. 2.117.

⁴⁰ Panel Reports, paras. 2.148 and 2.150 (referring to Law 10,637 of 30 December 2002 (PIS/PASEP non-cumulative) (Law 10,637/2002) (Panel Exhibits JE-94 and BRA-100), Article 29, section 1.11 (for IPI tax); Law 10,865 of 30 April 2004 providing for the Contribution to Social Integration and the Civil Servants Investment Programmes and the Contribution to the Financing of Social Security levied on the import of goods and services and sets forth other provisions (Law 10,865/2004) (Panel Exhibit JE-181), Article 40 (for PIS/PASEP and COFINS contributions)).

⁴¹ Panel Reports, paras. 2.163 and 7.1138. We note the definition of a predominantly exporting company in paragraphs 5.144-5.145 of these Reports below. ⁴² Panel Reports, paras. 3.1-3.3.

⁴³ Panel Reports, para. 7.70.

- 15 -

inapplicable to certain measures, in particular "pre-market" measures, directed at producers.⁴⁴ Second, the Panel rejected Brazil's argument that the ICT and INOVAR-AUTO programmes constitute payment of subsidies exclusively to domestic producers within the meaning of Article III:8(b) of the GATT 1994, and therefore are exempted from the disciplines of Article III of the GATT 1994 and Article 2.1 of the TRIMs Agreement. To the Panel, those aspects of a measure resulting in product discrimination are not exempted *per se* from these disciplines, even if the measure takes the form of a subsidy paid exclusively to domestic producers.⁴⁵ The Panel also made the following findings concerning the measures at issue:

- a. with respect to the ICT programmes:
 - the production-step requirements under the Informatics, PADIS, PATVD, and Digital Inclusion programmes; and the requirement for products to obtain the status of "developed in Brazil", under the Informatics, PATVD, and Digital Inclusion programmes, result in imported products being subject to internal taxes in excess of those applied to like domestic products, inconsistently with Article III: 2, first sentence, of the GATT 1994⁴⁶;
 - ii. the production-step requirements under the Informatics, PADIS, PATVD, and Digital Inclusion programmes, and the requirement for products to obtain the status of "developed in Brazil", under the Informatics, PATVD, and Digital Inclusion programmes; the aspect of the mechanism for the calculation of the amount of resources required to be invested in R&D under the Informatics and PADIS programmes relating to the deductible part; and the lower administrative burden on companies purchasing domestic incentivized intermediate products under the Informatics and PADIS programmes accord to imported products treatment less favourable than that accorded to like domestic products, inconsistently with Article III: 4 of the GATT 1994⁴⁷;
 - iii. it is not necessary to make findings with respect to the complaining parties' claims under Article III:5 of the GATT 1994 in order to secure a positive solution to this dispute, and the Panel therefore exercises judicial economy with respect to these claims⁴⁸;
 - iv. the Informatics, PADIS, PATVD, and Digital Inclusion programmes constitute trade-related investment measures, and the aspects of these programmes found to be inconsistent with Articles III:2 and III:4 of the GATT 1994 are also inconsistent with Article 2.1 of the TRIMs Agreement⁴⁹;
 - v. the tax exemptions, reductions, and suspensions granted under the Informatics, PADIS, PATVD, and Digital Inclusion programmes are subsidies within the meaning of Article 1.1 of the SCM Agreement, which are contingent upon the use of domestic over imported goods within the meaning of Article 3.1(b) of the SCM Agreement, and thus are prohibited subsidies, inconsistent with Articles 3.1(b) and 3.2 of the SCM Agreement⁵⁰; and
 - vi. those aspects of the PATVD programme found to be inconsistent with the GATT 1994 and the TRIMs Agreement are not justified under Article XX(a) of the GATT 1994.⁵¹

⁴⁷ Panel Reports, paras. 7.318, 8.5.b, and 8.16.c.

⁴⁴ Panel Reports, paras. 7.70, 8.2, and 8.13.

⁴⁵ Panel Reports, paras. 7.87, 8.3, and 8.14.

⁴⁶ Panel Reports, paras. 7.173-7.174, 8.5.a, and 8.16.a. The Panel exercised judicial economy with respect to Japan's claim under Article III:2, second sentence, of the GATT 1994. (Panel Reports, paras. 7.180 and 8.16.b)

⁴⁸ Panel Reports, paras. 7.347, 8.5.c, and 8.16.d.

⁴⁹ Panel Reports, paras. 7.365, 8.5.d, and 8.16.e.

⁵⁰ Panel Reports, paras. 7.500, 8.5.e, and 8.16.f.

⁵¹ Panel Reports, paras. 7.626, 8.5.f, and 8.16.g. On appeal, Brazil does not challenge this finding by the Panel.

- b. with respect to the INOVAR-AUTO programme:
 - certain aspects of the accreditation process, the system of rules on accrual and calculation of presumed tax credits, and the rules on the use of presumed tax credits resulting from expenditure on strategic inputs and tools in Brazil result in imported products being subject to internal taxes in excess of those applied to like domestic products, inconsistently with Article III:2 of the GATT 1994⁵²;
 - ii. certain aspects of the accreditation process, the system of rules on accrual and calculation of presumed tax credits, and the rules on the use of presumed tax credits resulting from expenditure on strategic inputs and tools in Brazil; the accreditation requirement to perform a minimum number of manufacturing steps in Brazil; that aspect of the rules on accrual of presumed IPI tax credits pertaining to expenditure in strategic inputs and tools; and those aspects of the accreditation requirements to invest in R&D in Brazil and make expenditures on engineering, basic industrial technology, and capacity-building of suppliers in Brazil, pertaining to the purchase of Brazilian laboratory equipment, accord less favourable treatment to imported products than that accorded to like domestic products, inconsistently with Article III:4 of the GATT 1994⁵³;
 - iii. it is not necessary to make findings with respect to the complaining parties' claims under Article III:5 of the GATT 1994 in order to secure a positive solution to this dispute, and the Panel therefore exercises judicial economy with respect to these claims⁵⁴;
 - iv. the INOVAR-AUTO programme constitutes a trade-related investment measure, and those aspects of the programme found to be inconsistent with Articles III: 2 and III: 4 of the GATT 1994 are also inconsistent with Article 2.1 of the TRIMs Agreement⁵⁵;
 - v. tax reductions through presumed tax credits granted under the INOVAR-AUTO programme are subsidies within the meaning of Article 1.1 of the SCM Agreement and contingent upon the use of domestic over imported goods within the meaning of Article 3.1(b) of the SCM Agreement, and thus are prohibited subsidies, inconsistent with Articles 3.1(b) and 3.2 of the SCM Agreement⁵⁶;
 - vi. those aspects of the INOVAR-AUTO programme found to be inconsistent with the GATT 1994 and the TRIMs Agreement are not justified under Article XX(b) or XX(g) of the GATT 1994⁵⁷;
 - vii. the tax reductions accorded to imported products from Mercado Común del Sur (Southern Common Market) (MERCOSUR) members and Mexico under the INOVAR-AUTO programme are advantages granted by Brazil to products originating in those countries, which are not accorded immediately and unconditionally to like products originating in other WTO Members, inconsistently with Article I:1 of the GATT 1994⁵⁸;
 - viii. the complaining parties were not under a burden to invoke the Enabling Clause in their panel requests, and their claims under Article I:1 of the GATT 1994 are therefore within the Panel's terms of reference⁵⁹; and

⁵² Panel Reports, paras. 7.688, 8.6.a, and 8.17.a. The Panel exercised judicial economy with respect to Japan's claim under Article III:2, second sentence, of the GATT 1994. (Panel Reports, paras. 7.691 and 8.17.b) On appeal, Brazil, however, does not challenge the Panel's finding under Article III:2, first sentence, of the GATT 1994.

⁵³ Panel Reports, paras. 7.772-7.773, 8.6.b, and 8.17.c.

⁵⁴ Panel Reports, paras. 7.792, 8.6.c, and 8.17.d.

⁵⁵ Panel Reports, paras. 7.806, 8.6.d, and 8.17.e.

⁵⁶ Panel Reports, paras. 7.847, 8.6.e, and 8.17.f.

⁵⁷ Panel Reports, paras. 7.965, 7.1011, 8.6.f, and 8.17.g. On appeal, Brazil does not challenge these findings by the Panel.

⁵⁸ Panel Reports, paras. 7.1048, 8.6.g, and 8.17.h.

⁵⁹ Panel Reports, paras. 7.1083, 7.1120, 8.6.h, and 8.17.i.

- ix. the tax reductions accorded to imported products from Argentina, Mexico, and Uruguay and found to be inconsistent under Article I:1 of the GATT 1994 are not justified under paragraph 2(b) or 2(c) of the Enabling Clause⁶⁰; and
- c. with respect to the PEC and RECAP programmes:
 - i. the tax suspensions granted thereunder are subsidies within the meaning of Article 1.1 of the SCM Agreement and contingent upon export performance within the meaning of Article 3.1(a) of the SCM Agreement, and thus are prohibited subsidies, inconsistent with Articles 3.1(a) and 3.2 of the SCM Agreement.⁶¹

1.16. In accordance with Article 19.1 of the DSU, and having found that Brazil acted inconsistently with its obligations under Articles I:1, III:2, and III:4 of the GATT 1994; Article 2.1 of the TRIMs Agreement; and Articles 3.1(a), 3.1(b), and 3.2 of the SCM Agreement with respect to the measures at issue, the Panel recommended that the Dispute Settlement Body (DSB) request Brazil to bring its measures into conformity with its obligations under the covered agreements.⁶² Pursuant to Article 4.7 of the SCM Agreement, the Panel further recommended that Brazil withdraw the subsidies found to be WTO-inconsistent within 90 days.⁶³

1.17. On 28 September 2017, Brazil notified the DSB, pursuant to Articles 16.4 and 17 of the DSU, of its intention to appeal certain issues of law covered in the Panel Reports and certain legal interpretations developed by the Panel, and filed a Notice of Appeal⁶⁴ and an appellant's submission pursuant to Rule 20 and Rule 21, respectively, of the Working Procedures for Appellate Review⁶⁵ (Working Procedures).

1.18. On 29 September 2017, the Appellate Body Secretariat transmitted the Working Schedule for Appeal drawn up by the Appellate Body Division hearing these appeals, setting out the deadlines for filing further written submissions.

1.19. On 3 October 2017, the European Union notified the DSB, pursuant to Articles 16.4 and 17 of the DSU, of its intention to appeal certain issues of law covered in the Panel Reports and certain legal interpretations developed by the Panel, and filed a Notice of Other Appeal and an other appellant's submission⁶⁶ pursuant to Rule 23 of the Working Procedures. On the same date, Japan also notified the DSB, pursuant to Articles 16.4 and 17 of the DSU, of its intention to appeal certain issues of law covered in the Panel Reports and certain legal interpretations developed by the Panel, and filed a Notice of Other Appeal and an other appellant's submission⁶⁶ pursuant to Articles 16.4 and 17 of the DSU, of its intention to appeal certain issues of law covered in the Panel Reports and certain legal interpretations developed by the Panel, and filed a Notice of Other Appeal and an other appellant's submission⁶⁷ pursuant to Rule 23 of the Working Procedures.

1.20. On 9 October 2017, the Appellate Body Division hearing these appeals received a joint letter from Argentina, Australia, Canada, and the United States requesting an extension of the deadline for the filing of their third participants' submissions in these proceedings (the joint request). On 9 October 2017, the Appellate Body Division hearing these appeals invited the participants and the other third participants, if they so wished, to comment on the joint request by 11 October 2017. No objections were received. On 12 October 2017, the Appellate Body Division hearing these appeals issued a Procedural Ruling deciding, in accordance with Rule 16(2) of the Working Procedures, to extend the deadline for the filing of third participants' submissions by one week to 26 October 2017.

1.21. On 16 October 2017, Brazil, the European Union, and Japan each filed an appellee's submission.⁶⁸ On 26 October 2017, Argentina, Australia, and the United States each filed a third participant's submission.⁶⁹ On the same day, Canada, China, Colombia, Korea, Russia, Singapore, South Africa, the Separate Customs Territory of Taiwan, Penghu, Kinmen and

⁶⁰ Panel Reports, paras. 7.1097, 7.1121, 8.6.i, and 8.17.j.

⁶¹ Panel Reports, paras. 7.1238, 8.7, and 8.18.

⁶² Panel Reports, paras. 8.9 and 8.20.

⁶³ Panel Reports, paras. 8.11 and 8.22.

⁶⁴ WT/DS472/8; WT/DS497/6.

⁶⁵ WT/AB/WP/6, 16 August 2010.

⁶⁶ WT/DS472/9.

⁶⁷ WT/DS497/7

⁶⁸ Pursuant to Rule 22 of the Working Procedures.

⁶⁹ Pursuant to Rule 24(1) of the Working Procedures.

- 18 -

Matsu (Chinese Taipei), and Turkey each notified its intention to appear at the oral hearing as a third participant.⁷⁰ On 15 June 2018, India and Ukraine each notified its intention to appear at the oral hearing as a third participant.⁷¹

1.22. On 24 November 2017, the participants and third participants were informed that, in accordance with Rule 15 of the Working Procedures, the Chair of the Appellate Body had notified the Chair of the DSB of the Appellate Body's decision to authorize Appellate Body Member Mr Peter Van den Bossche to complete the disposition of these appeals, even though his second term of office was due to expire before the completion of the appellate proceedings.

1.23. On 27 November 2017, the Chair of the Appellate Body notified the Chair of the DSB that the Appellate Body would not be able to circulate its Reports in these appeals within the 60-day period pursuant to Article 17.5 of the DSU, or within the 90-day period pursuant to the same provision.⁷² The Chair of the Appellate Body explained that this was due to a number of factors, including the enhanced workload of the Appellate Body in 2017, scheduling difficulties arising from appellate proceedings running in parallel with an overlap in the composition of the Divisions hearing the appeals, the number and complexity of the issues raised in this and concurrent appellate Body Secretariat. Although the appeals in this dispute were initiated on 28 September 2017, due to the multiple appeals pending before the Appellate Body. The reduced number of Appellate Body Members, and the shortage of staff in the Appellate Body Secretariat, work on these appeals could gather pace only in March 2018. On 19 November 2018, the Chair of the Appellate Body informed the Chair of the DSB that the Reports in these proceedings would be circulated no later than 13 December 2018.⁷³

1.24. The oral hearing was held on 19 and 20 June 2018. The participants and three third participants (Argentina, Australia, and the United States) made opening statements. The participants and one third participant (the United States) responded to questions posed by the Members of the Appellate Body Division hearing these appeals. The participants made closing statements. None of the third participants made closing statements.

2 ARGUMENTS OF THE PARTICIPANTS

2.1. The claims and arguments of the participants are reflected in the executive summaries of their written submissions provided to the Appellate Body.⁷⁴ The Notices of Appeal and Other Appeal, and the executive summaries of the participants' claims and arguments, are contained in Annexes A and B of the Addendum to these Reports, WT/DS472/AB/R/Add.1, WT/DS497/AB/R/Add.1.

3 ARGUMENTS OF THE THI RD PARTICIPANTS

3.1. The arguments of the third participants that filed a written submission (Argentina, Australia, and the United States) are reflected in the executive summaries of their written submissions provided to the Appellate Body⁷⁵, and are contained in Annex C of the Addendum to these Reports, WT/DS472/AB/R/Add.1, WT/DS497/AB/R/Add.1.

4 ISSUES RAISED IN THIS APPEAL

4.1. The following issues are raised in these appeals by Brazil:

⁷⁰ Pursuant to Rule 24(2) of the Working Procedures.

⁷¹ Pursuant to Rule 24(4) of the Working Procedures.

⁷² WT/DS472/10; WT/DS497/8.

⁷³ WT/DS472/11; WT/DS497/9.

⁷⁴ Pursuant to the Appellate Body's communication on "Executive Summaries of Written Submissions in Appellate Proceedings" and "Guidelines in Respect of Executive Summaries of Written Submissions in Appellate Proceedings" (WT/AB/23, 11 March 2015).

⁷⁵ Pursuant to the Appellate Body's communication on "Executive Summaries of Written Submissions in Appellate Proceedings" and "Guidelines in Respect of Executive Summaries of Written Submissions in Appellate Proceedings" (WT/AB/23, 11 March 2015).

- a. with respect to the Panel's findings under Article III: 8(b) of the GATT 1994:
 - whether the Panel erred in its interpretation and application of Article III:8(b) of the GATT 1994 in finding that subsidies that are provided exclusively to domestic producers pursuant to Article III:8(b) are not *per se* exempted from the disciplines of Article III of the GATT 1994⁷⁶;
- b. with respect to the Panel's findings under Articles III:2 and III:4 of the GATT 1994 and Article 2.1 of the TRIMs Agreement:
 - whether the Panel erred in the application of Article III: 2 of the GATT 1994 in finding that imported finished and intermediate Information and Communication Technology, Automation and Related Goods (ICT products) were taxed in excess of like domestic finished and intermediate ICT products;
 - ii. whether the Panel erred in the application of Article III: 4 of the GATT 1994 in finding that the accreditation requirements under the ICT programmes accord treatment less favourable to imported ICT products than that accorded to like domestic ICT products inconsistently with Article III: 4 of the GATT 1994;
 - whether the Panel erred in the application of Article III: 4 of the GATT 1994 in finding that the ICT programmes are inconsistent with Article III: 4 of the GATT 1994 by virtue of the lower administrative burden on companies purchasing incentivized domestic intermediate products;
 - iv. whether the Panel erred in finding that the PPBs and other production-step requirements under the ICT programmes are contingent upon the use of domestic goods, inconsistently with Article III: 4 of the GATT 1994;
 - v. consequently, whether the Panel erred in finding that the above-mentioned aspects of the ICT programmes are inconsistent with Article 2.1 of the TRIMs Agreement;
 - vi. whether the Panel erred in the application of Article III:4 of the GATT 1994 in finding that the accreditation requirements under the INOVAR-AUTO programme are inconsistent with Article III:4 because they are more burdensome for companies seeking accreditation as importers/distributors as opposed to domestic manufacturers; and
 - vii. consequently, whether the Panel erred in finding that the above-mentioned aspect of the INOVAR-AUTO programme is inconsistent with Article 2.1 of the TRIMs Agreement;
- c. with respect to the Panel's findings under Article 3.1(a) of the SCM Agreement:
 - i. whether the Panel erred in the application of Article 1.1(a)(1)(ii) of the SCM Agreement by identifying the benchmark treatment as the treatment applicable to non-accredited companies and rejecting the treatment of structural credit accumulators as the benchmark treatment;
 - ii. conditionally, in the event that the Appellate Body upholds the Panel's findings concerning the benchmark treatment, whether the Panel:
 - erred in the application of Article 1.1(a)(1)(ii) of the SCM Agreement in its comparison of the benchmark treatment with the challenged treatment under the PEC and RECAP programmes;
 - erred in finding that "cash availability" and "implicit interest income" are government revenue "otherwise due" within the meaning of Article 1.1(a)(1)(ii) of the SCM Agreement; and

⁷⁶ Panel Reports, para. 7.87.

- failed to conduct an objective assessment of the facts of the case under Article 11 of the DSU in its assessment of a piece of evidence submitted by Brazil;
- d. with respect to the Panel's findings under Article 3.1(b) of the SCM Agreement:
 - i. whether the Panel erred in the application of Article 1.1(a)(1)(ii) of the SCM Agreement by: (i) failing to compare the tax treatment accorded to the group of taxpayers in the benchmark treatment with the group of taxpayers under the ICT programmes; and (ii) finding that "cash availability" and "implicit interest income" are government revenue otherwise due within the meaning of Article 1.1(a)(1)(ii) of the SCM Agreement; and
 - whether the Panel erred in finding that the PPBs and other production-step requirements under the ICT programmes and the accreditation requirement to perform a number of manufacturing steps in Brazil under the INOVAR-AUTO programme constitute a requirement to use domestic over imported goods under Article 3.1(b) of the SCM Agreement;
- e. with respect to the Panel's findings under Article I:1 of the GATT 1994 and the Enabling Clause:
 - i. whether the Panel erred in its interpretation of paragraph 4(a) of the Enabling Clause in finding that the European Union and Japan did not have the burden to raise the Enabling Clause in their panel requests; and consequently, whether the Panel erred in finding that the claims raised by the European Union and Japan under Article I:1 of the GATT 1994 with regard to the INOVAR-AUTO programme were within its terms of reference;
 - ii. whether the Panel erred in its interpretation of paragraph 2(b) of Enabling Clause and in finding that the differential tax treatment under the INOVAR-AUTO programme was not justified under that provision; and
 - iii. whether the Panel erred in its interpretation of paragraph 2(c) of the Enabling Clause and in finding that the differential tax treatment under the INOVAR-AUTO programme was not justified under that provision; and
- f. with respect to the Panel's findings under Article 4.7 of the SCM Agreement:
 - i. whether the Panel acted inconsistently with Articles 11 and 12.7 of the DSU in recommending that Brazil withdraw the prohibited subsidies found to exist within 90 days.

4.2. The following issues are raised in these appeals by the European Union and Japan with respect to the Panel's findings concerning the in-house scenario under the PPBs and other production-step requirements:

- a. whether the Panel's failure to make findings on the in-house scenario, in its analysis of the claims under Article III:4 of the GATT 1994, Article 2.1 of the TRIMs Agreement, and Article 3.1(b) of the SCM Agreement, in relation to the ICT programmes and the INOVAR-AUTO programme, constitutes false judicial economy and a failure to make an objective assessment of the matter before the Panel, inconsistent with Article 11 of the DSU (raised by the European Union and Japan);
- b. conditionally, in the event that the Appellate Body finds that the Panel did not exercise judicial economy, whether the Panel acted inconsistently with Article 11 of the DSU by failing to provide coherent reasoning for its findings (raised by Japan);
- c. in the event that that the Appellate Body considers that the Panel correctly exercised judicial economy by not making specific findings in the in-house scenario, whether the Appellate Body should review and modify the Panel's legal interpretation and findings to "make it clearer", pursuant to Article 17.3 of the DSU, that the Panel did not need to rule

- 21 -

on both the "in-house" and "outsourcing" scenarios because the measures at issue are *per se* inconsistent with the relevant covered agreements (raised by the European Union and Japan); and

d. in the event that the Appellate Body rejects the above-mentioned claims by the European Union, whether the Panel erred in the application of Article III: 4 of the GATT 1994, Article 2.1 of the TRIMs Agreement, and Article 3.1(b) of the SCM Agreement, by failing to consider the European Union's claims under these provisions in light of all the relevant facts of the case (raised by the European Union).

5 ANALYSIS OF THE APPELLATE BODY

5.1 Articles III:2 and III:4 of the GATT 1994

5.1. We begin our analysis with Brazil's claims on appeal concerning Articles III: 2 and III: 4 of the GATT 1994, namely that the Panel erred in finding that: (i) the ICT programmes are inconsistent with Article III: 2, first sentence, of the GATT 1994 because under these programmes imported ICT products are taxed in excess of like domestic ICT products⁷⁷; (ii) the ICT programmes are inconsistent with Article III: 4 of the GATT 1994 because they accord to imported products treatment less favourable than that accorded to like domestic products⁷⁸; and (iii) the accreditation requirements under the INOVAR-AUTO programme are inconsistent with Article III: 4 of the GATT 1994 because they accord to that accorded to like domestic products treatment with Article III: 4 of the GATT 1994 because they accord to imported products treatment of the GATT 1994 because they accord to imported products treatment less favourable than that accorded to like domestic products treatment with Article III: 4 of the GATT 1994 because they accord to imported products treatment less favourable than that accorded to like domestic products treatment less favourable than that accorded to like domestic products treatment less favourable than that accorded to like domestic products.⁷⁹

5.2. We are, however, mindful of the fact that a substantial part of Brazil's appeal of the Panel's findings under Articles III: 2 and III: 4 of the GATT 1994 is closely connected with Brazil's appeal that the Panel erred in its interpretation and application of Article III:8(b) of the GATT 1994, given that Brazil advanced arguments under Article III:8(b) as a "general defence"⁸⁰ to all of the complaining parties' claims of inconsistency with Article III. For instance, in its appeal of the Panel's findings under Articles III:2 and III:4, Brazil contends that the Panel erred in failing to establish, "at the threshold"⁸¹, that the ICT and INOVAR-AUTO programmes constitute the "payment of subsidies to domestic producers" under Article III:8(b) and therefore are not subject to the disciplines of Articles III:2 and III:4.⁸² While Brazil puts forth stand-alone arguments that the Panel erred in its interpretation of Article III:8(b), it also advances arguments highlighting the Panel's findings under Articles III:2 and III:4. We address all of these connected arguments in section 5.2 of these Reports.

5.3. At this juncture, we also consider it useful to recall certain terminologies employed by the Panel in this dispute regarding the categories of products involved under the ICT and INOVAR-AUTO programmes. We do not understand the participants to take issue or dispute on appeal the Panel's understanding set out below.

5.4. The Panel noted that the European Union's and Japan's claims with respect to the ICT and INOVAR-AUTO programmes pertained to two distinct types of products: "incentivised products" and inputs for the "incentivised products".⁸³ The Panel further noted that Brazil subcategorized the

⁷⁷ Brazil's appellant's submission, para. 15.

⁷⁸ Brazil's appellant's submission, para. 15. We note that Brazil's appeal under Article III:4 of the GATT 1994 also concerns those aspects of the ICT programmes that allegedly contain local content requirements that the Panel found to be inconsistent with Article III:4 and, consequently, with Article 2.1 of the TRIMs Agreement. However, we do not address these aspects of Brazil's claims on appeal in this section. We address them in section 5.4.2.4 of these Reports when addressing Brazil's claims under Article 3.1(b) of the SCM Agreement.

⁷⁹ Brazil's appellant's submission, para. 15. We note that Brazil's appeal under Article III:4 of the GATT 1994 also concerns those aspects of the INOVAR-AUTO programme that allegedly contain local content requirements that the Panel found to be inconsistent with Article III:4 and, consequently, with Article 2.1 of the TRIMs Agreement. However, we do not address these aspects of Brazil's claims on appeal in this section. We address them in section 5.4.2.4.3 of these Reports when addressing Brazil's claims under Article 3.1(b) of the SCM Agreement.

⁸⁰ Panel Reports, para. 7.71.

⁸¹ Brazil's appellant's submission, paras. 15 and 42.

⁸² Brazil's appellant's submission, paras. 15, 42, and 247.

⁸³ Panel Reports, para. 7.23.

- 22 -

"incentivised products" as "intermediate" and "finished" products.⁸⁴ The Panel acknowledged that these terms are not "treaty language", but nonetheless considered that "these distinctions form a useful analytical tool in understanding the arguments and claims of the parties."85

5.5. The Panel noted that a "finished product", as identified by Brazil, is a product that will not undergo any further manufacturing and will be "incentivised" if "the company producing them is accredited under a particular programme."86 The Panel explained that, "[i]f a finished product is 'incentivised', it means that it receives a particular tax benefit on its sale."87

5.6. "Intermediate products", the Panel explained, will be subject to further manufacturing and will also be "incentivised" if the company producing them is accredited under a particular programme.⁸⁸ The Panel noted that, "[i]f an intermediate product is incentivised, it will be subject to a particular tax benefit on its sale."⁸⁹ The Panel also recognized that "such intermediate products will be further used in the production of a 'finished' product."90

5.7. With the aforesaid background in mind, we now turn to address Brazil's claim on appeal that the Panel erred in finding that imported finished and intermediate ICT products were taxed in excess of like domestic ICT products inconsistently with Article III:2, first sentence, of the GATT 1994.

5.1.1 Whether the Panel erred in finding that imported finished and intermediate ICT products were taxed in excess of like domestic finished and intermediate ICT products inconsistently with Article III:2, first sentence, of the GATT 1994

5.8. Before the Panel, Brazil put forth a broad defence contending that Article III of the GATT 1994 does not apply to the ICT programmes because "the disciplines of Article III govern discrimination on *products*, whereas the challenged programmes are not product-related but rather impose process and production-step requirements"⁹¹ and "concern only pre-market obligations" regarding production and investment in R&D by producers.92

5.9. The Panel considered that the "plain text" of Article III "is sufficient to refute Brazil's argument"93 and explained that "there is no reason why a measure directed at a producer rather than a product could not 'affect' the internal sale, offering for sale, purchase, etc. of domestic and imported products."94 The Panel concluded that "Article III ... is not per se inapplicable to certain measures, in particular 'pre-market' measures directed at producers"95 and therefore that the "general defence by Brazil ... fails."96

5.10. Thereafter, the Panel turned to address the European Union's and Japan's claims under Article III:2, first sentence, of the GATT 1994. The Panel found that the complaining parties have made a prima facie case that the relevant tax treatment provided for in the ICT programmes was exclusively based on the origin of the products and that Brazil had failed to successfully rebut the prima facie case of likeness made by the complaining parties.⁹⁷ The Panel then considered whether

⁸⁴ Panel Reports, para. 7.23.

⁸⁵ Panel Reports, para. 7.23.

⁸⁶ Panel Reports, para. 7.24.

⁸⁷ Panel Reports, para. 7.24.

⁸⁸ Panel Reports, para. 7.25.

⁸⁹ Panel Reports, para. 7.25. 90 Panel Reports, para. 7.25.

⁹¹ Panel Reports, para. 7.58. (emphasis original)

⁹² Panel Reports, para. 7.61 (referring to Brazil's first written submission to the Panel, paras. 167, 173, 181, 184, 229, 231-232, 287, 327, 340, 377, 461, 477, 489, 578, 586, and 744 and sections 5.1.2.3.1, 5.2.2.3.1, 5.3.2.3.1, and 5.4.3.2.1 (DS472) and paras. 125, 140, 143, 182, 185-186, 208, 238, 276, 290, 318, 333, 398, 413, 424, 426, 497, 514, and 519 and sections 4.1.2.4.1, 4.2.2.4.1, 4.3.2.4.1, and 4.4.2.4.1 (DS497)).

⁹³ Panel Reports, para. 7.63.

⁹⁴ Panel Reports, para. 7.63.

⁹⁵ Panel Reports, para. 7.70. 96 Panel Reports, para. 7.67.

⁹⁷ Panel Reports, paras. 7.139-7.140.

- 23 -

the relevant tax treatment provided for in the ICT programmes resulted in imported ICT products being taxed in excess of like domestic ICT products.⁹⁸

5.11. With respect to *finished ICT products*, the Panel noted that "neither the complaining parties nor Brazil question[s] the fact that the challenged programmes establish different levels of taxation."99 Recalling the relevant requirements under each of the ICT programmes¹⁰⁰, the Panel noted that the tax reductions and exemptions apply to the relevant domestic ICT products, provided that "the companies that manufacture these products, which must be located (and operate) in Brazil, fulfil certain requirements."¹⁰¹ The Panel considered that only ICT products manufactured in Brazil can satisfy the requirements to benefit from the tax reductions or exemptions.¹⁰² As for imported finished ICT products, the Panel noted that these products will never be able to qualify for the tax reductions and exemptions established in the ICT programmes because "such products are never manufactured in Brazil by companies located or operating in Brazil; such finished products are never produced in accordance with the relevant PPBs or similar production requirements."¹⁰³ The Panel further noted that imported ICT products developed outside of Brazil will never be able to obtain "the status of being 'developed in Brazil' and thus will never be able to gualify for the additional reductions".¹⁰⁴ For these reasons, the Panel found that, contrary to domestic finished ICT products manufactured in Brazil by accredited companies, "like imported ICT finished products cannot benefit from the tax reductions and exemptions (including through zero rates) established in the ICT programmes and are, therefore, subject to a higher tax burden than like domestic ICT products."105

5.12. With respect to *intermediate ICT products*, the Panel recalled that Brazil contended that "there is no difference in the tax treatment between intermediate ICT products manufactured by accredited and non-accredited companies."¹⁰⁶ The Panel proceeded to examine "how the Brazilian tax system applies to both incentivized domestic intermediate ICT products and like imported intermediate ICT products in order to determine whether the latter bear a higher tax burden".¹⁰⁷ The Panel observed that:

In the case of sales of *incentivized domestic intermediate ICT products,* which are manufactured *by accredited companies,* companies purchasing these products do not pay the IPI tax and PIS/PASEP and COFINS contributions (under the PADIS programme) or do not pay or pay a reduced IPI tax (under the Informatics programme). Consequently, they do not obtain any credit (in the case of tax exemptions) or they obtain a reduced credit (in the case of tax reductions) to offset against debits when paying their monthly liabilities.

In the case of sales of *imported intermediate ICT products*, which are never incentivized because they are manufactured by companies that cannot get accredited, the companies purchasing the imported (and, therefore, non-incentivized) intermediate ICT products must pay the IPI tax and PIS/PASEP and COFINS contributions. As a result of this payment, they obtain a tax credit that can later be used to offset debits from the same taxes and contributions, or ask for compensation with other taxes or reimbursement.¹⁰⁸

5.13. The Panel considered that, to the extent the transaction involves the payment of a tax as well as the granting of a tax credit, "these two elements must be taken into account in order to make an overall assessment of the actual tax burden imposed on imported intermediate ICT products, on the one hand, and like domestic incentivized intermediate ICT products, on the other."¹⁰⁹ The Panel

⁹⁸ Panel Reports, para. 7.142.

⁹⁹ Panel Reports, para. 7.149 (referring to Brazil's first written submissions to the Panel, paras. 336, 382, and 485 (DS472) and paras. 286, 324, and 422 (DS497)).

¹⁰⁰ Panel Reports, para. 7.150.

¹⁰¹ Panel Reports, para. 7.150. (fn omitted)

¹⁰² Panel Reports, para. 7.151 (referring to Panel Reports, para. 7.135).

¹⁰³ Panel Reports, para. 7.151.

¹⁰⁴ Panel Reports, para. 7.151 (referring to Panel Reports, para. 7.136).

¹⁰⁵ Panel Reports, para. 7.154.

¹⁰⁶ Panel Reports, para. 7.157.

¹⁰⁷ Panel Reports, para. 7.160.

¹⁰⁸ Panel Reports, paras. 7.161-7.162. (fn omitted; emphasis original)

¹⁰⁹ Panel Reports, para. 7.164 (referring to Panel Report, Argentina – Hides and Leather, para. 11.184).

- 24 -

found that "there is a different effective tax burden" on imported ICT products vis-à-vis like domestic ICT products for two reasons: (i) the availability of cash flow for those companies that benefit from the tax exemption or reduction, and (ii) the "time-value" of money.¹¹⁰ For these reasons, the Panel concluded that "imported intermediate ICT products, which are never incentivized, are subject to a higher tax burden than like domestic incentivized intermediate ICT products."111

5.14. As a preliminary matter, we note that, on appeal, Brazil takes issue with the Panel's finding that "Article III of the GATT 1994 is not per se inapplicable to certain measures, in particular 'pre-market' measures directed at producers."¹¹² Brazil contends that Article III deals with non-discrimination between domestic and imported products, "thus being applicable only to measures which affect a product once it has been produced and enters the marketplace"¹¹³ and "not to production (pre-market)" measures.¹¹⁴ In response to questioning at the oral hearing in these appellate proceedings, Brazil, however, confirmed that it does not wish to pursue any further this line of argument regarding the inapplicability of Article III to producer-related measures.

5.15. Article III:2, first sentence, of the GATT 1994 stipulates that "[t]he products of the territory of any Member imported into the territory of any other Member shall not be subject, directly or indirectly," to internal taxes in excess of those applied to like domestic products. The Appellate Body has explained that "[t]his language suggests that the provision applies to a broad range of measures."¹¹⁵ Article III:2, first sentence, thus has a broad scope of application by not only disciplining internal taxes that directly affect products but also internal taxes that indirectly affect products.¹¹⁶ Moreover, we observe that Article III:1 of the GATT 1994 is couched in broad and inclusive language referring to among other things internal taxes, other internal charges, laws, regulations and requirements affecting the internal sale, and the offering for sale of products. The Appellate Body has explained that "Article III:1 constitutes part of the context of Article III:2, in the same way that it constitutes part of the context of each of the other paragraphs in Article III."¹¹⁷ These considerations suggest that, while the focus of Article III: 2, first sentence is, in particular, "on the treatment accorded to 'products'"¹¹⁸, it does not exclude from its scope measures that are on their face directed at producers, which nevertheless subject the product concerned to taxation in excess, and thereby have an impact on the conditions of competition.¹¹⁹

5.16. Article III: 4 of the GATT 1994 prohibits treatment less favourable to products of foreign origin imported into the territory of a Member than that accorded to like domestic products. Article III: 4 specifies that this obligation is applicable with respect to all laws, regulations, and requirements "affecting their internal sale, offering for sale, purchase, transportation, distribution or use". The Appellate Body has interpreted the word "affecting" as "having a 'broad scope of application'". 120 The Appellate Body has found that "measures that restrict the rights of traders may violate GATT obligations with respect to trade in goods", recalling, in particular, that "restrictions imposed on investors, wholesalers, and manufacturers, as well as on points of sale and ports of entry, have been found to be inconsistent with Article III: 4."¹²¹ Thus, measures directed at producers that may affect the "specific transactions, activities and uses"¹²² mentioned in Article III:4, such that they modify

¹¹⁹ However, whether measures directed at producers subject the product concerned to taxation in excess, and thereby have an impact on the conditions of competition, must be determined on the basis of the facts and circumstances at issue.

¹¹⁰ Panel Reports, para. 7.169.

¹¹¹ Panel Reports, para. 7.172.

¹¹² Brazil's appellant's submission, para. 263 (quoting Panel Reports, para. 7.70).

¹¹³ Brazil's appellant's submission, para. 264. (emphasis original)

¹¹⁴ Brazil's appellant's submission, para. 265.

¹¹⁵ Appellate Body Report, *Thailand – Cigarettes (Philippines)*, para. 112. (fn omitted) ¹¹⁶ The Appellate Body has found that "[a]ny measure that indirectly affects the conditions of competition between imported and like domestic products would come within the provisions of Article III:2,

first sentence." (Appellate Body Report, Canada - Periodicals, p. 19)

¹¹⁷ Appellate Body Report, Japan - Alcoholic Beverages II, p. 18.

¹¹⁸ Appellate Body Reports, Canada – Renewable Energy / Canada – Feed-in Tariff Program, para. 5.62.

¹²⁰ Appellate Body Report, US – FSC (Article 21.5 – EC), para. 210. The Appellate Body explained that "[i]t is ... not any 'laws, regulations and requirements' which are covered by Article III: 4, but only those which affect' the specific transactions, activities and uses mentioned in that provision." (Appellate Body Report, US - FSC (Article 21.5 - EC), para. 208 (emphasis original))

¹²¹ Appellate Body Report, China - Publications and Audiovisual Products, para. 227. (fn omitted)

¹²² Appellate Body Report, US - FSC (Article 21.5 - EC), para. 208.

the conditions of competition to the detriment of imported products vis-à-vis like domestic products, fall within the scope of Article III: 4.¹²³

5.17. With this background in mind, we now turn to address Brazil's claim on appeal that the Panel erred in finding that imported finished ICT products were taxed in excess of like domestic finished ICT products inconsistently with Article III:2, first sentence, of the GATT 1994.

5.1.1.1 Whether the Panel erred in finding that imported finished ICT products were taxed in excess of like domestic finished ICT products

5.18. On appeal, Brazil contends that the Panel failed "to undertake a thorough analysis of the case presented by complainants and to carefully scrutinize the design, structure and operation of the ICT programmes in applying Article **III:2** ... **to the facts of this dispute**".¹²⁴ Instead, according to Brazil, the Panel completely disregarded the fact that the complaining parties had not submitted any evidence that "the tax treatment under the ICT programmes resulted in *de jure* discrimination based exclusively on the origin of products in a manner contrary to the principles set forth in Article III."¹²⁵ Brazil submits that "there is nothing in the design, structure and operation of the challenged measures that by 'necessary implication' would amount to *de jure* tax discrimination within the meaning of Article III: 2."¹²⁶

5.19. In response, the European Union submits that "the Panel focused on the impact of the tax incentives on domestic versus imported *products*" in order to conclude that imported ICT products, which are manufactured outside Brazil, will never be able to obtain the tax reductions and exemptions available under the ICT programmes to like domestic ICT products manufactured by accredited companies.¹²⁷ According to the European Union, the Panel found that imported ICT products will be subject to taxes in excess of those applied to finished domestic ICT products by "looking into the design, structure and operation of the ICT programmes".¹²⁸ The European Union submits that "[n]o more is required to find discrimination under Article III:2, first sentence, of the GATT 1994."¹²⁹

5.20. Japan submits that the Panel did not "simply assume" that the ICT programmes were inconsistent with Article III: 2.¹³⁰ According to Japan, the Panel engaged in a thorough analysis of Article III: 2 and its application to the ICT programmes, ultimately finding that these programmes led to imported products being taxed in excess of domestic products in violation of Article III: 2, first sentence.¹³¹

5.21. We recall that, **as part of its analysis concerning "whether the complaining parties** ... made a *prima facie* case that the relevant ICT programmes draw distinctions between the relevant products based exclusively on origin"¹³², the Panel noted that, in order for an ICT product to be subject to the tax treatment provided for in the ICT programmes, companies that manufacture these products "must be located (and operate) in Brazil"¹³³ and "must comply with one or more"¹³⁴ of the following requirements:

a. invest in R&D in Brazil (in the case of the Informatics, PADIS, and PATVD programmes);

¹²³ However, the determination of whether such measures modify the conditions of competition to the detriment of imported products so as to accord treatment less favourable within the meaning of Article III:4 of the GATT 1994 must be determined on the basis of the facts and circumstances at issue.

¹²⁴ Brazil's appellant's submission, para. 87.

¹²⁵ Brazil's appellant's submission, para. 88.

¹²⁶ Brazil's appellant's submission, para. 93.

¹²⁷ European Union's appellee's submission, para. 107 (referring to Panel Reports, para. 7.151). (emphasis original)

¹²⁸ European Union's appellee's submission, para. 107.

¹²⁹ European Union's appellee's submission, para. 107.

¹³⁰ Japan's appellee's submission, para. 53.

¹³¹ Japan's appellee's submission, para. 53 (referring to Panel Reports, paras. 7.154 and 7.172).

¹³² Panel Reports, para. 7.127.

¹³³ Panel Reports, para. 7.150.

¹³⁴ Panel Reports, para. 7.134.

- 26 -

b. manufacture in Brazil in accordance with the relevant PPBs (in the case of the Informatics, PADIS, PATVD, and Digital Inclusion programmes) or carry out certain manufacturing steps in Brazil (in the case of the PADIS and PATVD programmes); and/or

c. develop the products in Brazil (in order to obtain additional tax reductions under the Informatics programme or to obtain tax exemptions (through zero rates) under the PATVD programme).¹³⁵

5.22. With respect to the requirement that ICT products be manufactured in accordance with the relevant PPBs¹³⁶, the Panel considered that "only ICT products manufactured in Brazil can meet this requirement given that PPBs require that a certain number of manufacturing operations that characterize the effective 'production' of a certain product be performed in Brazil."¹³⁷ The Panel further considered that the same applies for the requirement that certain manufacturing steps be performed in Brazil.¹³⁸ The Panel found that, "by necessary implication, only a product manufactured in Brazil can benefit from the tax treatment under the ICT programmes."¹³⁹ As for the requirement that products be "developed in Brazil", the Panel recalled that "the relevant Implementing Order explains that in order to meet this requirement, the 'specifications, projects and developments [of the products benefiting from the relevant tax treatment] must be carried out in Brazil'."¹⁴⁰ The Panel considered that imported products that have been "developed" outside of Brazil, but are like domestic products developed in Brazil, "can never meet this requirement, and therefore cannot gualify for the relevant tax treatment".¹⁴¹ The Panel concluded that, for the purposes of Article III: 2, "all incentivized products that receive the tax treatment under the challenged programmes[] can be considered to be Brazilian domestic products."¹⁴² On this basis, the Panel found that the complaining parties had made a prima facie case that "the differential tax treatment provided for in the challenged ICT programmes is exclusively based on the origin of the products."143

5.23. Brazil accepts that "[i]t is undisputed that foreign producers cannot be accredited under the ICT programmes."¹⁴⁴ We note that "'finished' products will be 'incentivised' if the company producing them is accredited under a particular programme" and "[i]f a finished product is 'incentivised', it means that it receives a particular tax benefit on its sale."¹⁴⁵ Imported finished ICT products are therefore not eligible for either tax reductions or exemptions under the ICT programmes and consequently, bear the full tax burden, as opposed to like domestic finished ICT products. The Panel rightly considered that imported finished ICT products cannot qualify for the relevant tax treatment¹⁴⁶ because they: (i) "are never manufactured in Brazil by companies located or operating **in Brazil [and]** ... are never produced in accordance with the relevant PPBs or similar production requirements"; and (ii) "will never be able to obtain the status of being 'developed in Brazil".¹⁴⁷ This, in our view, demonstrates that the Panel analysed the design, structure, and operation of the ICT programmes and came to the conclusion that they are designed in such a manner that only "Brazilian domestic products"¹⁴⁸ can "benefit from the tax reductions or exemptions"¹⁴⁹ to the exclusion of imported finished ICT products. We, therefore, reject Brazil's argument that there is

¹³⁵ Panel Reports, para. 7.134. (fns omitted)

¹³⁶ The Panel defined a PPB as "the minimum set of operations performed at a manufacturing facility that characterizes the actual industrialization of a given product". (Panel Reports, para. 2.62 (referring to Decree 5,906/2006 (Panel Exhibit JE-7), Article 16)). In other words, the Panel explained that PPBs indicate the minimum stages or steps of the manufacturing process of a product that must be performed in Brazil. (Ibid.) On appeal, the participants do not take issue with the definition of a PPB set out by the Panel.

¹³⁷ Panel Reports, para. 7.135 (referring to Brazil's first written submissions to the Panel, para. 128 (DS472) and para. 85 (DS497)).

¹³⁸ Panel Reports, para. 7.135.

¹³⁹ Panel Reports, para. 7.135.

¹⁴⁰ Panel Reports, para. 7.136 (referring to Panel Reports, fn 228 to para. 2.87).

¹⁴¹ Panel Reports, para. 7.136. (fn omitted)

¹⁴² Panel Reports, para. 7.137.

¹⁴³ Panel Reports, para. 7.139.

¹⁴⁴ Brazil's appellant's submission, para. 97.

¹⁴⁵ Panel Reports, para. 7.24.

¹⁴⁶ Panel Reports, para. 7.136.

¹⁴⁷ Panel Reports, para. 7.151. ¹⁴⁸ Panel Reports, para. 7.137.

¹⁴⁹ Panel Reports, para. 7.137.

¹⁴⁹ Panel Reports, para. 7.151 (referring to Panel Reports, para. 7.135).

- 27 -

nothing in the design, structure, and operation of the ICT programmes that would amount to *de jure* discrimination based exclusively on the origin of products.¹⁵⁰

5.24. Brazil further contends that, "[g]iven the non-cumulative nature of Brazil's indirect taxes", the application of the credit-debit system to the products manufactured by non-accredited companies "results in principle in the same tax burden" as for products manufactured by accredited companies, to which the credit-debit system does not apply.¹⁵¹ **Thus, Brazil asserts that "[i]t was** ... incumbent upon the Panel to examine evidence of the *effects* of the ICT programmes on the *effective* tax burden borne by the imported products in order to establish a violation of Article III: 2.¹⁵²

5.25. In response, the European Union submits that Brazil "clearly indicated" before the Panel that its rebuttal based on the functioning of the credit-debit system was "relevant only with regard to incentivised intermediate products".¹⁵³ In any event, the European Union submits that Brazil's contention regarding imported finished ICT products is "wrong on substance".¹⁵⁴ The European Union explains that the sale of a finished product to a distributor or a consumer represents the last possible stage of the application of the product taxes in question. The European Union reasons that, "[i]f at that stage the tax rate is higher for imported products than for like domestic incentivized products, then the tax burden imposed on the product is higher for the former products."¹⁵⁵

5.26. Japan recalls that the Panel noted that, with respect to finished ICT products, none of the parties disputed that the ICT programmes establish different levels of taxation, and that only ICT products manufactured in Brazil could satisfy the requirements to qualify for tax reductions.¹⁵⁶ Japan contends that the Panel conducted a thorough and detailed analysis of the actual tax burden under the ICT programmes.¹⁵⁷

5.27. We note that, with respect to finished ICT products, Brazil argued before the Panel that "any possible difference in taxation aims at compensating accredited companies for the costs they must incur in order to meet the requirements provided for in the challenged programmes."¹⁵⁸ As noted above, imported finished ICT products are not eligible for either tax reductions or exemptions because foreign producers cannot be accredited under the ICT programmes and, consequently, bear the full tax burden, as opposed to like domestic finished ICT products. For intermediate ICT products, Brazil, however, argued that "there is no difference in the tax burden on imported products compared to domestic products, because its tax system is neutral in terms of tax collection throughout the production chain."¹⁵⁹ Therefore, it appears that Brazil does not dispute that finished ICT products are subject to different levels of taxation. Indeed, as noted by the Panel, with respect to finished products, "neither the complaining parties nor Brazil question[s] the fact that the challenged programmes establish different levels of taxation"¹⁶⁰ and further that the parties "acknowledge[]" that "the level of taxation on like imported and domestic *finished* [ICT] products is different because of the tax reductions and exemptions provided for in the relevant ICT programmes."¹⁶¹ These considerations suggest that Brazil is conflating its defence with respect to finished ICT products with that of intermediate ICT products by drawing arguments that it made in the specific context of intermediate ICT products. In any event, Brazil's justification that "any possible difference in taxation" under the ICT programmes with respect to domestic finished ICT products is aimed at "compensating accredited companies for the costs they must incur in order to meet the requirements" seems to flow largely from policy reasons and considerations.¹⁶² As also noted by the

¹⁵⁶ Japan's appellee's submission, para. 56 (referring to Panel Reports, paras. 7.149 and 7.151).

¹⁵⁰ Brazil's appellant's submission, paras. 88 and 93.

¹⁵¹ Brazil's appellant's submission, para. 97.

¹⁵² Brazil's appellant's submission, para. 98. (emphasis original)

¹⁵³ European Union's appellee's submission, para. 96. (emphasis omitted)

¹⁵⁴ European Union's appellee's submission, para. 104.

¹⁵⁵ European Union's appellee's submission, para. 104.

¹⁵⁷ Japan's appellee's submission, para. 58.

¹⁵⁸ Panel Reports, para. 7.95 (referring to Brazil's first written submissions to the Panel, paras. 156 and 205 (DS472) and paras. 105 and 161 (DS497)).

¹⁵⁹ Panel Reports, para. 7.95 (referring to Brazil's first written submissions to the Panel, para. 185 (DS472) and para. 144 (DS497)).

¹⁶⁰ Panel Reports, para. 7.149 (referring to Brazil's first written submissions to the Panel, paras. 336, 382, and 485 (DS472) and paras. 286, 324, and 422 (DS497)). (emphasis added)

¹⁶¹ Panel Reports, para. 7.149. (emphasis added)

¹⁶² Panel Reports, para. 7.95.

Panel¹⁶³, the justification for a measure that is found to be inconsistent with Article III:2, first sentence, can be assessed, for example, in the context of the general exceptions under Article XX of the GATT 1994.

5.28. That said, we note that a finished product "is a product that will not undergo any further manufacturing"¹⁶⁴ and therefore the sale of a finished product represents the last stage of a transaction. In the case of an imported finished ICT product, when an importer sells the imported finished ICT product to a wholesaler, retailer, or distributor, the importer will charge the IPI tax to the wholesaler, retailer, or distributor and remit the tax to the Brazilian Government.¹⁶⁵ In contrast, in the case of a like domestic finished ICT product that is subject to IPI tax exemption or reduction under the ICT programmes, the seller does not charge any tax or charges a reduced tax, as the case may be, to the wholesaler, retailer, or distributor.¹⁶⁶ At this last stage the tax rate is thus higher for imported finished ICT products than for like domestic finished ICT products, and the tax burden on the former is necessarily in excess of that on the latter.¹⁶⁷ In other words, imported finished ICT products. In these circumstances, we fail to see how the credit-debit system, as Brazil contends, applies and offsets any tax burden in the case of imported finished ICT products.¹⁶⁸

5.29. Thus, we agree with the Panel that, because imported finished ICT products are "subject to a higher tax burden than like domestic ICT products", they are consequently "taxed in excess of like domestic finished ICT products, contrary to Article III:2, first sentence, of the GATT 1994".¹⁶⁹

5.1.1.2 Whether the Panel erred in finding that imported intermediate ICT products were taxed in excess of like domestic intermediate ICT products

5.30. We now turn to intermediate ICT products regarding which Brazil asserts on appeal that the Panel erred in its assessment of "the impact of the tax suspension and exemptions granted under the ICT programmes in the beginning or in the middle of the production chain".¹⁷⁰ According to Brazil, the Panel "opted to ignore the fact that the credit-**debit system in a value added tax ... ensures** that the amount collected at each step of production is equivalent to the value added at that step".¹⁷¹ Brazil thus submits that "[i]n the end, the tax burden of a product subject to the payment of a tax, which generates a credit, and a product that is subject to a suspension, but receives no credit, will be the same."¹⁷²

5.31. In response, the European Union submits that the Panel did not agree with Brazil that "the suspension or exemption of indirect value-added taxes in the beginning or in the middle of the production chain does not affect the final amount of tax collected by the tax authorities."¹⁷³ Instead, the European Union contends that the Panel clearly indicated that it was concerned with determining the effective tax burden. The European Union further contends that the Panel clearly stated that it was required to take account of the functioning of the tax credit-debit system in assessing the effective tax burden imposed on imported products compared to the burden imposed on incentivized domestic products.¹⁷⁴ The European Union asserts that "Brazil does not take issue with the Panel's description of the Brazilian tax debit-**credit mechanism ... which p**rovide[s] the factual basis for the Panel analysis."¹⁷⁵

- ¹⁶⁸ Brazil's appellant's submission, para. 97.
- ¹⁶⁹ Panel Reports, para. 7.154. (fn omitted)
- ¹⁷⁰ Brazil's appellant's submission, para. 100.
- ¹⁷¹ Brazil's appellant's submission, para. 101.
- ¹⁷² Brazil's appellant's submission, para. 101.
- ¹⁷³ European Union's appellee's submission, para. 127 (referring to Brazil's appellant's submission, para. 101).

¹⁷⁴ European Union's appellee's submission, para. 126 (referring to Panel Reports, paras. 7.160 and 7.164).

¹⁶³ Panel Reports, para. 7.153.

¹⁶⁴ Panel Reports, para. 7.24.

¹⁶⁵ Panel Reports, para. 2.14.

¹⁶⁶ Panel Reports, para. 2.15.

¹⁶⁷ European Union's appellee's submission, para. 104.

¹⁷⁵ European Union's appellee's submission, para. 132.

- 29 -

5.32. Japan submits that "the ICT programmes' structure and design impose origin-based discrimination with respect to tax burdens imposed on products covered by the programmes."¹⁷⁶ According to Japan, the Panel came to this conclusion after "thoroughly analysing the actual tax burden under the programmes as well as by fully addressing the arguments put forward".¹⁷⁷

5.33. The Panel described the functioning of the credit-debit system, in particular, in paragraphs 2.11-2.13 of the Panel Reports. On appeal, Brazil neither takes issue with the understanding set out by the Panel in the above-mentioned paragraphs of the Panel Reports, nor has raised a claim under Article 11 of the DSU. That said, we recall that when comparing the situations concerning the sale of "incentivized domestic intermediate ICT products", on the one hand, and "imported ... intermediate ICT products", on the other hand, the Panel explained that:

First, the tax exemptions (including through zero rates) ... **do** not involve any payment by the purchaser of the incentivized domestic intermediate ICT product but, at the same time, do not generate any tax credit that the purchaser can later use to offset future tax liabilities (i.e. tax debits). Second, the tax reduc**tions** ... **involve a smaller tax** payment by the purchaser of the incentivized domestic intermediate ICT product but, at the same time, generate a lower tax credit to be used later to offset future tax liabilities. Third, the situation absent the tax exemptions and/or reductions involves the full payment of the tax by the purchaser and, at the same time, the granting to the purchaser of tax credits to be used later to offset tax liabilities.

5.34. Therefore, when tax exemptions or reductions are applied, no credit or lower tax credit is accrued because the tax is not due, or is due at a lower rate, whereas under the credit-debit non-cumulative system that applies to imported (and, therefore, non-incentivized) intermediate ICT products, a tax credit (of the same value as the tax paid) is granted to the purchaser. However, while on the face of it, the tax system may appear to be neutral in terms of tax collection with respect to intermediate ICT products (whether domestic or imported), the Panel, in our view, rightly considered that "a thorough look into the operation of the tax holistically is necessary in order to determine the effective tax burden on the products at issue."¹⁷⁹ This exercise, as the Panel also noted, would require taking into consideration "the granting of tax credits to purchasers of imported (and, therefore, non-incentivized) intermediate ICT products following the payment of the tax".¹⁸⁰

5.35. Article III:2, first sentence, of the GATT 1994 is concerned with the protection of "the equal competitive relationship between imported and domestic products".¹⁸¹ The Appellate Body explained that "the words of the first sentence require an examination of the conformity of an internal tax measure with Article III" by determining, first, "whether the taxed imported and domestic products are 'like'" and, second, "whether the taxes applied to the imported products are 'in excess of' those applied to the like domestic products".¹⁸² With respect to the second element, the Appellate Body has found that "[e]ven the smallest amount of 'excess' is too much."¹⁸³ A determination of whether an infringement of Article III: 2, first sentence, exists must be made on the basis of an overall assessment of the actual tax burdens imposed on imported products, on the one hand, and like domestic products, on the other hand.¹⁸⁴ The Panel was mindful that, under the credit-debit system, "the same tax rate may be applied"¹⁸⁵ but nonetheless considered, and rightly so, that "the fact that the nominal value of the tax collected may be identical is not determinant of the tax measure's consistency under Article III: 2."¹⁸⁶ The Panel took into account Brazil's argument that the effect of the tax treatment under the ICT programmes in relation to intermediate products is "neutral"

- ¹⁸² Appellate Body Report, Japan Alcoholic Beverages II, pp. 18-19.
- ¹⁸³ Appellate Body Report, Japan Alcoholic Beverages II, p. 23.
- ¹⁸⁴ Panel Report, Argentina Hides and Leather, para. 11.184.
- ¹⁸⁵ Panel Reports, para. 7.167.
- ¹⁸⁶ Panel Reports, para. 7.165.

¹⁷⁶ Japan's appellee's submission, para. 64.

¹⁷⁷ Japan's appellee's submission, para. 64.

¹⁷⁸ Panel Reports, para. 7.163.

¹⁷⁹ Panel Reports, para. 7.164.

¹⁸⁰ Panel Reports, para. 7.164.

¹⁸¹ Appellate Body Report, *Japan – Alcoholic Beverages II*, p. 16. (fn omitted) The Appellate Body explained that "the first sentence of Article III:2 is, in effect, an application of [the] general principle" of Article III:1 of the GATT 1994. (Ibid., p. 18) We also recall that the panel in *Argentina – Hides and Leather* considered that "the purpose of Article III:2, first sentence, **... is to ensure '**equality of competitive conditions between imported and like domestic products." (Panel Report, *Argentina – Hides and Leather*, para. 11.182 (quoting Appellate Body Report, *Canada – Periodicals*, p. 18))

- 30 -

because "both products covered by the relevant ICT programmes and products outside these programmes are subject to the same tax burden."¹⁸⁷ However, the Panel ultimately found that "there is a different effective tax burden on imported ICT products vis-à-vis like domestic ICT products for two reasons: the availability of cash flow for those companies that benefit from the tax exemption or reduction, and the 'time-value' of money."¹⁸⁸

5.36. In this regard, Brazil contends on appeal that the relevant Brazilian taxes are "not paid up front but are levied monthly on the whole of the previous month's activities" and, therefore, "it is simply not correct to conclude that the payment of taxes necessarily reduces the cash flow availability of purchasers of non-incentivized products, or that credits lose their value over time, to the detriment of imported products."¹⁸⁹ Brazil submits that the Panel "improperly concluded" that the ICT programmes result in a higher tax burden on imported intermediate ICT products because "the payment of tax up front would limit the cash availability of the purchaser and generate a tax credit, the value of which diminishes over time."¹⁹⁰

5.37. In response, the European Union explains that "the payment of a tax at the moment of the purchase of an intermediate product deprives the purchaser of the sum paid and therefore affects its cash-flow."¹⁹¹ By the same token, the European Union notes that "money depreciates over time through the effect of inflation."¹⁹² Thus, the European Union submits that, if the purchase of an imported intermediate product involves the payment of a tax at the moment of the purchase and the generation of a tax credit that can be compensated at some later point in time, "it follows necessarily that the imported intermediate product carries a tax burden that is effectively heavier [than] that imposed on the purchase of an incentivised domestic intermediate product, which is tax exempted or subject to a lower tax rate."¹⁹³

5.38. Japan contends that, during the time lag between purchase of inputs and sale of outputs, funds related to the payment of taxes are temporarily unavailable. The ICT programmes, according to Japan, "mitigate the negative financial consequences of this lag, by reducing or eliminating the required IPI, PIS/PASEP, and COFINS taxes otherwise due" when domestic manufacturers of intermediate goods purchase inputs covered by the ICT programmes.¹⁹⁴ Thus, Japan submits that "the Panel properly concluded that 'this has the effect of limiting the availability of cash flow by companies purchasing imported intermediate ICT products and results in a higher effective tax burden on these products.¹⁹⁵

5.39. Under the credit-debit system, purchases of non-incentivized imported intermediate ICT products involve the payment of a tax upfront that is not faced by companies that purchase incentivized like domestic intermediate ICT products, which are exempted from the relevant taxes.¹⁹⁶ Even in the case of tax reductions, companies purchasing incentivized like domestic intermediate ICT products have to pay a lower tax as compared to companies purchasing non-incentivized imported intermediate ICT products. Indeed, as the Panel also noted, "the tax to be paid would be lower than the tax for like imported intermediate ICT products, which are not incentivized."¹⁹⁷ We fail to see how these situations do not have the effect of limiting the availability of cash flow for companies purchasing non-incentivized imported intermediate ICT products.¹⁹⁸ The fact that purchasers of imported intermediate ICT products have to pay the relevant taxes under the

- ¹⁹³ European Union's appellee's submission, para. 134.
- ¹⁹⁴ Japan's appellee's submission, para. 62.

¹⁹⁶ Panel Reports, para. 7.170.

¹⁸⁷ Brazil's appellant's submission, para. 102.

¹⁸⁸ Panel Reports, para. 7.169 (referring to European Union's second written submission to the Panel, paras. 57-59 and 331; response to Panel question No. 43, paras. 188-191; Japan's second written submission to the Panel, para. 143).

¹⁸⁹ Brazil's appellant's submission, para. 107. (fn omitted)

¹⁹⁰ Brazil's appellant's submission, para. 103.

¹⁹¹ European Union's appellee's submission, para. 134.

¹⁹² European Union's appellee's submission, para. 134.

¹⁹⁵ Japan's appellee's submission, para. 62 (quoting Panel Reports, para. 7.170).

¹⁹⁷ Panel Reports, para. 7.170.

¹⁹⁸ We observe that, in the context of its assessment of whether the tax treatment under the

ICT programmes constitutes financial contributions, the Panel noted that "the Brazilian Government ... will hold the advantage of cash availability or cash flow ... that could be generated on the full amount of that tax that it has collected from the seller", noting further the European Union's contention that "the benchmark rate of the ... Central Bank [of Brazil] ... was, at the time of writing of its first written submission, 13.25%." (Panel Reports, para. 7.433 and fn 790 thereto (additional fns to para. 7.433 omitted))

- 31 -

ICT programmes, irrespective of the point in time, compared to purchasers of incentivized like domestic intermediate ICT products, who do not have to pay the relevant tax or pay a reduced amount, "limit[s] the availability of cash flow"¹⁹⁹, resulting in a higher effective tax burden on imported intermediate ICT products.

5.40. Brazil also contends that the "tax credits accrued from the purchase of imported inputs do not necessarily have to be compensated with debits related to the same tax, and they can be used before the sale of the final product."²⁰⁰ Brazil, therefore, submits that no heavier tax burden is imposed on imported intermediate ICT products.²⁰¹

5.41. We observe that the Panel noted that "Brazil has indicated that the time period that it takes a company to offset its tax credits can be very short."202 The Panel also noted that in cases where the IPI tax debits are lower than the IPI tax credits, and the company buying a product cannot offset the credits with debits after a three-month period, "it can request compensation of the credits with other taxes, or reimbursement from the Brazilian Government."203 In this regard, we are mindful of the fact that Brazil also indicated to the Panel that the process for compensation with other taxes or reimbursement can take a long time.²⁰⁴ The Panel, in particular, noted that "[t]he reimbursement process may take from several months to years."²⁰⁵ The value of the tax credit that is generated upon the payment of the relevant tax on the sale of a non-incentivized imported intermediate ICT product will depreciate over time until it is used or adjusted. To that extent, in as much as there is a time lag between the accrual of the tax credit and the adjustment or use thereof, it necessarily results in the value of money (in the form of accrued tax credits) depreciating over time. Therefore, imported intermediate ICT products, the purchase of which is subject to a payment of tax upfront, bear a higher tax burden than that faced by the incentivized like domestic intermediate ICT products, which benefit from tax exemption or reduction. These considerations support the Panel's finding that, in the case of imported intermediate ICT products, even if credits are generated and can be offset later in time, they are subject to a higher tax burden than like incentivized domestic intermediate ICT products purchased from accredited companies, "due to depreciation in the value of money over time". 206

5.42. Accordingly, we agree with the Panel that "imported intermediate ICT products are taxed in excess of like domestic incentivized intermediate ICT products contrary to Article III:2, first sentence, of the GATT 1994."207

5.1.2 Whether the Panel erred in finding that the ICT programmes are inconsistent with Article III:4 of the GATT 1994

5.43. We now turn to consider Brazil's claim on appeal that the Panel erred in finding that the ICT programmes accord to imported products treatment less favourable than that accorded to like domestic products inconsistently with Article III: 4 of the GATT 1994, and, consequently, also erred in its finding under Article 2.1 of the TRIMs Agreement.

5.44. Before the Panel, the European Union and Japan argued that the conditions for accreditation necessary for products to obtain the tax advantages under the ICT programmes and the lower administrative burden on companies purchasing domestic incentivized intermediate ICT products

¹⁹⁹ Panel Reports, para. 7.170.

²⁰⁰ Brazil's appellant's submission, para. 108.

²⁰¹ Brazil's appellant's submission, para. 108.

²⁰² Panel Reports, para. 7.171.

²⁰³ Panel Reports, para. 7.171 (referring to Brazil's second written submission to the Panel, paras. 11-12

and 17). ²⁰⁴ Panel Reports, para. 7.171 (referring to Brazil's closing statement at the first meeting of the Panel, ²⁰⁵ and 234; comments on para 2.28 of the draft para. 15; second written submission to the Panel, paras. 185 and 234; comments on para. 2.28 of the draft descriptive part of the Panel Reports). We observe that Brazil made these statements in the context of describing the phenomenon of tax credit accumulation. (Brazil's second written submission to the Panel, paras. 182-185 and 234)

²⁰⁵ Panel Reports, para. 2.20. (fn omitted)

²⁰⁶ Panel Reports, para. 7.171. (fn omitted)

²⁰⁷ Panel Reports, para. 7.172.

accord less favourable treatment to imported ICT products than that accorded to like domestic ICT products inconsistently with Article III: 4.208

5.45. The Panel first considered the claim with respect to the conditions of accreditation, i.e. "the conditions that companies must fulfil in respect of particular incentivized products to become accredited producers of, and thus [become] eligible for the tax incentives on, those products." 209 The Panel noted that "the purpose of complying with the requirements for accreditation" is to obtain a tax exemption, reduction, or suspension on the sales or purchases of products.²¹⁰ The Panel recalled that "only products produced in Brazil can satisfy the conditions ... for accreditation ... and only products developed in Brazil can satisfy the requirement of being 'developed in Brazil'."211 Imported like products, the Panel recalled, "cannot satisfy those requirements, and thus can never qualify for the tax exemptions, reductions or suspensions granted under the relevant ICT programmes".²¹² The Panel further recalled that, as a consequence of these accreditation requirements, "imported *finished* and *intermediate* products bear a higher tax burden than like domestic *finished* and *intermediate* products."²¹³ The Panel considered that "the conditions for accreditation, which when fulfilled create a lower internal tax burden on domestic products than on like imported products, modify the conditions of competition to the detriment of the imported products."²¹⁴ The Panel, therefore, concluded that "the accreditation requirements of the ICT programmes, by restricting access to the tax incentives only to domestic products, result in less favourable treatment being accorded to imported products than to like domestic products."215

5.46. With respect to the claim concerning the lower administrative burden placed on companies purchasing domestic incentivized intermediate ICT products, the Panel noted that a purchaser of an incentivized intermediate product subject to a tax exemption or reduction will not need to anticipate any tax, or will only need to anticipate a reduced amount of tax.²¹⁶ However, the Panel considered that the purchaser of a non-incentivized intermediate ICT product will need to anticipate the full amount of those taxes when purchasing the products, and will have to undergo the administrative procedure to claim a tax credit in relation to those taxes, to request compensation with the same or other taxes, or to ask for reimbursement.²¹⁷ The Panel reasoned that, when faced with a decision to choose from either "a product whose purchase will entail no payment of taxes" or "a product whose purchase will entail the payment of the tax and the administrative burden that comes with the procedure of offsetting the tax with other debits (or requesting compensation or reimbursement)", a purchaser, under normal circumstances, "will prefer to avoid the administrative burden that comes with the payment of the tax".²¹⁸ Recalling that an imported intermediate ICT product will never be eligible for the tax reductions or exemptions, the Panel, therefore, found that "the incentive to avoid the administrative burden, by purchasing incentivized intermediate products, modifies the conditions of competition between domestic and like imported products, to the detriment of the imported products."219

²⁰⁸ Panel Reports, para. 7.181. The European Union and Japan also raised a claim regarding the aspect of the mechanism for the calculation of the amount of resources required to be invested in R&D under the Informatics and PADIS programmes, where the amounts paid when purchasing incentivized products are deducted from the calculation. (Ibid.) The Panel found that "the mechanisms under the Informatics and PADIS programmes for the calculation of the amount of resources required to be invested in R&D accord to imported products treatment less favourable than that accorded to like domestic products, and thus are inconsistent with Article III: 4 of the GATT 1994." (Panel Reports, para. 7.243) On appeal, Brazil does not challenge this finding by the Panel. (Brazil's response to questioning at the oral hearing)

²⁰⁹ Panel Reports, para. 7.214.

²¹⁰ Panel Reports, para. 7.197.

²¹¹ Panel Reports, para. 7.223 (referring to Panel Reports, paras. 7.135-7.140).

²¹² Panel Reports, para. 7.223 (referring to Panel Reports, paras. 7.135-7.140).

²¹³ Panel Reports, para. 7.224 (referring to Panel Reports, paras. 7.154 and 7.172). (emphasis original)

²¹⁴ Panel Reports, para. 7.225. (fn omitted)

²¹⁵ Panel Reports, para. 7.225.

²¹⁶ Panel Reports, para. 7.252.

²¹⁷ Panel Reports, para. 7.252.

²¹⁸ Panel Reports, para. 7.253.

²¹⁹ Panel Reports, para. 7.254. (emphasis original)

5.1.2.1 Whether the Panel erred in finding that the accreditation requirements under the ICT programmes accord treatment less favourable to imported products than that accorded to like domestic products inconsistently with Article III:4 of the GATT 1994

5.47. On appeal, Brazil asserts that the Panel's analysis with respect to the accreditation requirements under the ICT programmes "conflates tax discrimination, which is under the purview of Article III:2 ... with regulatory discrimination, to which Article III:4 applies".²²⁰ Brazil contends that "the Panel ... seems to have found that [the] accreditation requirements under the ICT programmes are inconsistent with Article III:4 because they allegedly restrict access to the tax incentives only to domestic products."²²¹ Brazil submits that in order to find a violation of Article III:4 of the GATT 1994, the Panel was required to identify a regulatory discrimination other than the differences in tax treatment that may result in a violation of Article III:2 of the GATT 1994. Therefore, Brazil contends that "[t]o the extent that the Panel's findings that the ICT programmes violate Article III:4 were based on this flawed reasoning, they should be reversed."²²²

5.48. In response, the European Union submits that "[t]he conditions for accreditation amount to a regulatory discrimination since only domestic products are incentivised; that the incentive consists in lower taxation when compared to imported like products is the consequence of the fact that the incentives are only granted to domestic products in detriment of imported like products."²²³ The European Union further submits that the Panel, having recalled the relevant Appellate Body jurisprudence, found that the same measure, or certain aspects of the same measures, can fall both within Articles III: 2 and III: 4 of the GATT 1994, and that Brazil has not appealed this finding.²²⁴

5.49. Japan disagrees with Brazil's claim that the Panel conflated Articles III:2 and III:4 of the GATT 1994.²²⁵ Japan submits that the Panel determined that, as a result of the accreditation requirements, imported finished and intermediate ICT products bear a higher tax burden than domestic finished and intermediate ICT products.²²⁶ Japan submits that, "[f]rom this factual premise, the Panel correctly concluded that the accreditation requirements modify the conditions of competition to the detriment of imported products."²²⁷

5.50. Article III:4 of the GATT 1994 prohibits Members from according treatment less favourable to products of foreign origin than that accorded to like domestic products. It specifies that this obligation is applicable with respect to all laws, regulations, and requirements "affecting their internal sale, offering for sale, purchase, transportation, distribution or use". The Appellate Body found that "the mere fact that a Member draws regulatory distinctions between imported and like domestic products is, in itself, not determinative of whether imported products are treated less favourably within the meaning of Article III:4."²²⁸ Rather, the Appellate Body explained that "what is relevant is whether such regulatory differences distort the conditions of competition to the detriment of imported products."²²⁹ Thus, according to the Appellate Body, whether a measure involves treatment less favourable "must be founded on a careful analysis of the contested measure and of its implications in the marketplace".²³⁰

5.51. We note that the aspect of the ICT programmes challenged by the complaining parties as being inconsistent with Article III: 4 concerned the accreditation requirements, the fulfilment of which enabled the obtaining of the relevant tax exemption, reduction, or suspension on the sales or purchases of ICT products. Indeed, as the Panel also recognized, "[i]n respect of Article III: 4, the complaining parties challenge the laws, regulations and requirements, namely the accreditation requirements for gaining access to the tax incentives."²³¹ The Panel further considered that

²²⁴ European Union's appellee's submission, para. 174.

- ²²⁶ Japan's appellee's submission, para. 68 (referring to Panel Reports, paras. 7.154-7.172).
- ²²⁷ Japan's appellee's submission, para. 68 (referring to Panel Reports, para. 7.225).
- ²²⁸ Appellate Body Report, *Thailand Cigarettes (Philippines)*, para. 128. (fn omitted)
- ²²⁹ Appellate Body Report, *Thailand Cigarettes (Philippines)*, para. 128.

²³⁰ Appellate Body Report, *US* – *FSC (Article 21.5 – EC)*, para. 215. The Appellate Body, however, cautioned that "the examination need not be based on the *actual effects* of the contested measure in the marketplace." (Ibid. (emphasis original; fn omitted))

²²⁰ Brazil's appellant's submission, para. 138.

²²¹ Brazil's appellant's submission, para. 139.

²²² Brazil's appellant's submission, para. 140.

²²³ European Union's appellee's submission, para. 176.

²²⁵ Japan's appellee's submission, para. 69.

²³¹ Panel Reports, para. 7.220.

- 34 -

"the aspects challenged by the complaining parties under Article III: 4 are different from, albeit related to, the differential tax treatment of like domestic and imported products that they challenge under Article III: 2."²³² We recall that the Panel also noted that "a single measure can be inconsistent with two or more provisions of Article III at the same time" since "multiple features of a single measure may operate simultaneously" and therefore, in such a situation, "different aspects of the same measure could be considered to be covered by the disciplines of either or both Article III:2 and III: 4."233

5.52. As noted, Brazil contends that the Panel's analysis with respect to the accreditation requirements conflates tax discrimination under Article III:2 with regulatory discrimination under Article III: 4.²³⁴ However, the Panel was mindful of the fact that the ICT programmes included both fiscal and regulatory aspects that were applicable to the products at issue.²³⁵ While the accreditation requirements relate to regulatory aspects²³⁶, the tax exemptions or reductions under the ICT programmes relate to fiscal aspects. This is evident from the Panel's conclusion that the conditions for accreditation "modify the conditions of competition to the detriment of the imported products" by creating "a lower internal tax burden on domestic products than on like imported products".²³⁷ It is undisputed that, in order to be eligible for the tax exemption, reduction, or suspension under the ICT programmes, companies must fulfil the accreditation requirements. The accreditation requirements under the ICT programmes therefore result in less favourable treatment for imported ICT products in the form of the differential tax burden that imported ICT products are subjected to by virtue of the fact that "foreign producers cannot be accredited under the ICT programmes."238 The consequence being, as the Panel also noted, imported ICT products "can never qualify for the tax exemptions, reductions or suspensions". 239

5.53. We note that the aspects of the ICT programmes found to be inconsistent with Article III:2, first sentence, and Article III: 4 are distinct. In the case of Article III: 2, first sentence, the aspect of the ICT programme found to be inconsistent is the differential tax treatment that results in a higher tax burden on imported ICT products, i.e. imported ICT products are taxed in excess of like domestic ICT products. Whereas, for the purposes of Article III: 4, the aspect of the ICT programmes found to be inconsistent is the accreditation requirements that result in less favourable treatment in the form of the differential burden that imported ICT products are subjected to. We do not see why that cannot be the case since different aspects of the same measure may be found to be inconsistent with one or more paragraphs of Article III of the GATT 1994. As the Appellate Body explained in Thailand - Cigarettes (Philippines), "[a]Ithough Thailand may be correct in stating that prior WTO reports have examined measures consisting of 'administrative requirements relating to the sale of imported products' under Article III: 4", it does not exclude the possibility that "if such requirements subject imported and like domestic products to internal taxes or other internal charges, the same

²³² Panel Reports, para. 7.220. We observe that the Panel also noted that:

The complaining parties allege that certain tax aspects of the programmes, insofar as they apply to incentivized finished and intermediate products, are discriminatory and inconsistent with Article III:2. The complaining parties also allege that the specific conditions and criteria to be satisfied by incentivized *finished* and *intermediate* products in order to receive the tax advantages[] are inconsistent with Article III:4 of the GATT 1994, by consisting of conditions for obtaining the advantage in respect of the incentivized products.

(Ibid., para. 7.35 (emphasis original))

²³³ Panel Reports, para. 7.34. (fn omitted)

²³⁴ Brazil's appellant's submission, para. 138.

²³⁵ Panel Reports, para. 7.35.

²³⁶ The Panel, as we recall, found that "the purpose of complying with the requirements for accreditation is to obtain a tax exemption, reduction or suspension on the sales or purchases of products, so it is clear that the requirements at issue affect the sale, offering for sale or purchase of products." (Panel Reports, para. 7.197) ²³⁷ Panel Reports, para. 7.225. (fn omitted)

²³⁸ Brazil's appellant's submission, para. 97.

²³⁹ Panel Reports, para. 7.223.

measures, or certain aspects of the same measures, could not also be scrutinized under Article III:2."240

5.54. In light of the foregoing considerations, we agree with the Panel that the accreditation requirements of the ICT programmes, by restricting access to the tax incentives only to domestic products, modify the conditions of competition to the detriment of imported products, and result in less favourable treatment being accorded to imported ICT products than to like domestic ICT products inconsistently with Article III:4 of the GATT 1994.²⁴¹

5.1.2.2 Whether the Panel erred in finding that the ICT programmes are inconsistent with Article III:4 of the GATT 1994 by virtue of the lower administrative burden on companies purchasing incentivized domestic intermediate products

5.55. We now turn to Brazil's claim on appeal that the Panel erred in finding that imported intermediate ICT products are subject to more onerous administrative requirements than like domestic intermediate ICT products.²⁴²

5.56. On appeal, Brazil asserts that the Panel's reasoning concerning the alleged lower administrative burden on companies purchasing domestic incentivized intermediate products "has no grounds, either in law or in the facts of the present dispute".²⁴³ Brazil submits that "[t]he Panel misconstrued the functioning of the Brazilian tax system and found an administrative burden in the operation of the debit and credit tax system where there is none."²⁴⁴ Brazil explains that a "careful review" of the functioning of the credit-debit system applicable to "IPI, PIS, PASEP and COFINS taxes under the Brazilian Tax Code shows unequivocally that incentivised domestic intermediate products are not subject to a lower administrative burden when compared to like imported intermediate products".²⁴⁵

5.57. In response, the European Union submits that the Panel's reasoning is firmly grounded on the functioning of the credit-debit system as described in the Panel Reports.²⁴⁶ The European Union adds that, while Brazil makes the claim that the Panel's findings have no factual or legal ground, "it is unable to argue that the Panel violated Article 11 of the DSU by failing to make an objective assessment of the facts of the case when it described the functioning of the Brazilian tax system and the credit-debit mechanism."²⁴⁷ The European Union recalls that the panel²⁴⁸ and "it was Brazil itself who explained that offsetting tax credits in certain situations can be burdensome, and can take years."²⁴⁹

5.58. Japan submits that "Brazil cannot seek a de novo review of the facts at this stage of the proceeding."²⁵⁰ Japan contends that Brazil's argument also fails to account for the "undisputed fact" that purchasers of non-incentivized imported intermediate ICT products will have to "claim compensation from the Brazilian Government, which could take years, whereas the purchasers of incentivized domestic intermediate products do not".²⁵¹

²⁴⁰ Appellate Body Report, *Thailand – Cigarettes (Philippines)*, fn 144 to para. 114 (quoting Thailand's appellant's submission, para. 69). (additional text thereto omitted) We also observe that a similar proposition was considered by the panel in *Mexico – Taxes on Soft Drinks* where the panel found that the measure at issue in that dispute (i.e. the Law on the Special Tax on Production and Services (LIEPS) that provided for the soft drink tax, the distribution tax, and the bookkeeping requirements) was inconsistent with both Articles III:2 and III:4 of the GATT 1994. (Panel Report, *Mexico – Taxes on Soft Drinks*, paras. 8.59, 8.96, and 8.123) Particularly, with respect to the claim under Article III:4, the panel found that "through the soft drink tax, the distribution tax and the bookkeeping requirements, Mexico accords less favourable treatment to imported non-cane sugar sweeteners ... than that accorded to like products of national origin ... inconsistent[ly] with Article III:4." (Panel Report, *Mexico – Taxes on Soft Drinks*, para. 8.123)

²⁴¹ Panel Reports, para. 7.225.

²⁴² Brazil's appellant's submission, para. 121.

²⁴³ Brazil's appellant's submission, para. 152.

²⁴⁴ Brazil's appellant's submission, para. 152.

²⁴⁵ Brazil's appellant's submission, para. 153.

²⁴⁶ European Union's appellee's submission, para. 204.

²⁴⁷ European Union's appellee's submission, para. 205.

²⁴⁸ European Union's appellee's submission, para. 206 (referring to Panel Reports, para. 2.1).

²⁴⁹ European Union's appellee's submission, para. 206.

²⁵⁰ Japan's appellee's submission, para. 73.

²⁵¹ Japan's appellee's submission, para. 73. (fn omitted)

5.59. We have considered above that the Panel took into account the functioning of the credit-debit system. We are mindful of Brazil's contention that this "system ... operates as a ledger in which both purchases of inputs and intermediate goods and sales of final goods are recorded, and debits and credits are offset"252 and that there is "no[] change in the event of suspensions, reductions or exemptions of indirect taxes".²⁵³ Brazil's contention would not, however, change the fact that purchasers of imported intermediate ICT products that are not incentivized under the ICT programmes will have to anticipate and pay the full amount of tax due on such imported intermediate ICT products. Although any such tax paid on the purchase of imported intermediate ICT products will generate a corresponding tax credit in favour of the purchaser, nonetheless, offsetting this tax credit entails an administrative burden that is not faced and/or faced to a lesser extent by a purchaser of domestic intermediate ICT products that are incentivized.²⁵⁴ This is the case because, under the credit-debit system, "if the tax credit cannot be offset by debits after three taxation periods", the process of compensating the tax credit with other federal taxes, or reimbursement thereof can "be burdensome for companies, and can take years".²⁵⁵ We therefore reject Brazil's argument that the Panel erroneously "found an administrative burden in the operation of the debit and credit tax system" in the context of its analysis under Article III: 4.256

5.60. The ICT programmes are designed in a manner that creates incentives for the market participants, that is, purchasers of intermediate ICT products, to behave in a manner that has the "direct practical effect"²⁵⁷ of treating imported intermediate ICT products less favourably than like domestic intermediate ICT products. In this case, by creating an incentive to purchase incentivized domestic intermediate ICT products in order to be relieved from and/or to face reduced administrative burdens. Accordingly, we agree with the Panel that, "when faced with a decision to choose", a purchaser, "under normal circumstances, will prefer to avoid the administrative burden that comes with the payment of the tax"²⁵⁸ and thus prefer to purchase incentivized domestic intermediate ICT products. 259

5.61. In light of the foregoing considerations, we agree with the Panel that "the ICT programmes are inconsistent with Article III: 4 of the GATT 1994, because they accord to imported intermediate products treatment less favourable than that accorded to like domestic intermediate products, due to the lower administrative burden imposed on firms purchasing incentivised intermediate products."²⁶⁰

5.1.3 Whether the Panel erred in finding that the ICT programmes are inconsistent with Article 2.1 of the TRIMs Agreement

5.62. Before the Panel, the European Union and Japan argued that the ICT programmes are inconsistent with Article 2.1 of the TRIMs Agreement, both independently and in conjunction with Article 2.2 and paragraph 1(a) of that Agreement's Illustrative List, because they are inconsistent with Article III of the GATT 1994.²⁶¹ Brazil agreed that the ICT programmes are investment measures, submitting, however, that they do not relate to trade in goods.²⁶²

5.63. The Panel found that "the ICT programmes affect, and ... are aimed at promoting, investment" and "also have an impact on trade, by affecting the sale and purchase of imported products, including

paras. 690, 841, 998, and 1118; Japan's first written submission to the Panel, paras. 371, 437, 479-496, and 533-548). ²⁶² Panel Reports, para. 7.351.

²⁵² Brazil's appellant's submission, para. 155.

²⁵³ Brazil's appellant's submission, para. 155.

²⁵⁴ In this regard, we note that purchasers of incentivized intermediate domestic ICT products, in most cases, will not need to anticipate the tax due on the purchase of such intermediate ICT products since the Informatics, PADIS, PATVD, and Digital Inclusion programmes provide for tax exemptions, through zero rates, to accredited companies selling domestic intermediate ICT products. It is only in the context of the IPI tax reduction provided under the Informatics programme that purchasers of domestic intermediate ICT products may have to anticipate and pay the reduced amount of IPI tax due.

²⁵⁵ Panel Reports, para. 7.251 (referring to Brazil's first written submission to the Panel, para. 702 (DS497)).

²⁵⁶ Brazil's appellant's submission, para. 152.

²⁵⁷ Appellate Body Report, Korea - Various Measures on Beef, para. 145.

²⁵⁸ Panel Reports, para. 7.253.

²⁵⁹ Panel Reports, para. 7.254.

²⁶⁰ Panel Reports, para. 7.255. (emphasis original)

²⁶¹ Panel Reports, para. 7.348 (referring to European Union's first written submission to the Panel,

- 37 -

the inputs used in the production of incentivized *finished* and *intermediate* products".²⁶³ The Panel concluded that the ICT programmes are trade-related investment measures within the meaning of the TRIMs Agreement.²⁶⁴ Having so found, the Panel recalled its findings that certain aspects of the ICT programmes are inconsistent with Articles III: 2 and III: 4 of the GATT 1994.²⁶⁵ Therefore, the Panel found that "those aspects of the ICT programmes found to be inconsistent with Article III: 2 and III: 4 ... are also inconsistent with Article 2.1 of the TRIMs Agreement."²⁶⁶

5.64. On appeal, Brazil does not make any specific arguments in connection with the Panel's finding under Article 2.1 of the TRIMs Agreement. Rather, Brazil's request for reversal of the Panel's finding under that provision is premised on us reversing the Panel's findings under Article III: 4 of the GATT 1994.²⁶⁷ We have, however, for the reasons stated above, agreed with the Panel's findings that certain aspects of the ICT programmes are inconsistent with Article III: 4 of the GATT 1994. Consequently, we agree with the Panel's finding that those aspects of the ICT programmes found to be inconsistent with Article III: 4 are also inconsistent with Article 2.1 of the TRIMs Agreement.²⁶⁸

5.1.4 Whether the Panel erred in finding that the accreditation requirements under the INOVAR-AUTO programme are inconsistent with Article III:4 of the GATT 1994 because they are more burdensome for companies seeking accreditation as importers/distributors as opposed to domestic manufacturers

5.65. We now turn to Brazil's claim on appeal with respect to Article III:4 of the GATT 1994 concerning the INOVAR-AUTO programme. Brazil contends that the Panel erred in finding that "the accreditation requirements under the INOVAR-AUTO programme were ... discriminatory because **they would be** ... more burdensome" for companies seeking accreditation as importers/distributors than for domestic manufacturers.²⁶⁹ We note that, unlike in the context of the ICT programme foreign manufacturers cannot be accredited²⁷⁰, under the INOVAR-AUTO programme foreign manufacturers can be accredited as importers/distributors and thus be entitled to the relevant tax benefits.²⁷¹ Therefore, the claim of error under Article III:4 with respect to the INOVAR-AUTO programme is primarily directed towards, as Brazil contends, the Panel's alleged erroneous conclusion that "foreign companies seeking accreditation under the INOVAR-AUTO programme must go through a more burdensome accreditation process than domestic manufacturers."²⁷²

5.66. We recall that with respect to the accreditation requirements under the INOVAR-AUTO programme, the Panel noted that "the purpose of complying with the requirements for accreditation is to obtain presumed IPI tax credits on the sale of products."²⁷³ The Panel noted that there are three types of accreditation that entitle companies to accrue and use presumed IPI tax credits: (i) accreditation for domestic manufacturers; (ii) accreditation for importers/distributors; and (iii) accreditation for investors.²⁷⁴ The Panel recalled that foreign manufacturers, as such, cannot obtain accreditation under the INOVAR-AUTO programme, because they are not located or do not operate in Brazil.²⁷⁵ The Panel observed that such foreign manufacturers would have to become accredited as importers/distributors in order for their products to be eligible for the tax benefits.²⁷⁶ The Panel recalled its earlier conclusion that "this results in a higher tax burden on imported vehicles manufacturers" because "foreign manufacturers are required to be legally

²⁶⁶ Panel Reports, para. 7.363.

²⁶⁸ Panel Reports, para. 7.363.

²⁶⁹ Brazil's appellant's submission, para. 278.

- ²⁷² Brazil's appellant's submission, para. 286 (quoting Panel Reports, para. 7.658).
- ²⁷³ Panel Reports, para. 7.706.

²⁷⁴ Panel Reports, para. 7.656. The Panel noted that "[a]II accredited companies must comply with two general requirements and with a series of specific requirements that differ depending on the type of accreditation." (Ibid. (fn omitted))

²⁶³ Panel Reports, para. 7.360. (emphasis original)

²⁶⁴ Panel Reports, para. 7.361.

²⁶⁵ Panel Reports, para. 7.363 (referring to Panel Reports, paras. 7.174 and 7.318-7.319).

²⁶⁷ Having sought reversal of the Panel's findings under Article III:4 of the GATT 1994, specifically with respect to paragraphs 7.318-7.319, 8.5.b, and 8.16.c of the Panel Reports, Brazil submits that "[a]s a result, Brazil also requests that the Appellate Body *reverse* the Panel's finding in paragraphs 7.365, 8.5(d) and 8.16(e) of the Panel Report[s] that in respect of those elements the ICT programmes are inconsistent with Article 2.1 of the TRIMs Agreement." (Brazil's appellant's submission, para. 172 (emphasis original))

²⁷⁰ Brazil's appellant's submission, para. 97.

²⁷¹ Panel Reports, para. 7.731 (referring to Panel Reports, paras. 7.656-7.661).

²⁷⁵ Panel Reports, para. 7.731.

²⁷⁶ Panel Reports, para. 7.731 (referring to Panel Reports, paras. 7.656-7.661).

- 38 -

established in Brazil" in order to be accredited as importers/distributors, and "foreign manufacturers seeking accreditation as importers/distributors are required to comply with more accreditation requirements than domestic manufacturers in order to obtain the tax benefits."²⁷⁷ Consequently, the **Panel considered that "[f]oreign manufacturers ... bear a higher burden than domestic manufacturers** in becoming accredited."²⁷⁸

5.67. For these reasons, the Panel found that, "under the INOVAR-AUTO programme, the conditions for accreditation in order to receive presumed tax credits ... accord less favourable treatment to imported products than that accorded to like domestic products, inconsistently with Article III: 4 of the GATT 1994."²⁷⁹

5.68. On appeal, Brazil takes issue with the Panel's finding that the accreditation requirements under the INOVAR-AUTO programme are discriminatory because "they would be allegedly more burdensome to 'importers/distributors' than to 'domestic manufacturers'" and therefore result in less favourable treatment being accorded to imported finished motor vehicles.²⁸⁰ Brazil submits that, in reaching its conclusions on the discriminatory impact of the types of accreditation provided under the INOVAR-AUTO programme, "the Panel limited itself to conducting a *quantitative* analysis of the requirements provided under INOVAR-AUTO [programme] (number of requirements)."²⁸¹ According to Brazil, the "mere difference in the number of requirements" to be fulfilled by domestic manufacturers and importers/distributors was "enough" for the Panel to find that the accreditation requirements were inconsistent with Article III: 4.²⁸² Brazil contends that, in order to find a "potential discriminatory impact", the Panel was required to have conducted a "*qualitative* analysis of the requirements taking into account the actual and relative weight of such requirements in terms of the overall burden imposed to companies – foreign and domestic alike – that request accreditation to the programme".²⁸³

5.69. In response, the European Union recalls that, in order to be eligible for the presumed IPI tax credits, "companies must obtain accreditation, and in order to obtain accreditation companies must comply with certain requirements."²⁸⁴ The European Union notes that the Panel distinguished two different consequences flowing from this design of the INOVAR-AUTO programme. First, according to the European Union, the Panel "rightly noted that the requirement with respect to domestic manufacturing activities is automatically complied with by domestic manufacture[r]s, whereas foreign manufacturers should comply with three (instead of two) different requirements".²⁸⁵ Second, in the European Union's view, "the Panel considered that foreign companies seeking to establish themselves in Brazil face a supplementary burden in comparison to domestic producers already established in Brazil" in as much as "compliance with the accreditation criteria necessarily entails establishing in Brazil, with a corresponding administrative and economic burden."²⁸⁶ Thus, the European Union submits that "no more was required" for the Panel to determine that the accreditation requirements modify the conditions of competition to the detriment of imported motor vehicles and in favour of like domestic motor vehicles.²⁸⁷

5.70. Japan notes that Brazil asserts that "the Panel should have conducted a '*qualitative* analysis of the requirements taking into account the actual and relative weight of such requirements in terms of the overall burden imposed to companies – foreign and domestic alike – that request accreditation to the programme."²⁸⁸ Japan submits that Brazil, however, does not provide any legal authority for this proposition, and, "beyond generalized assertions, fails to explain why the Panel's thorough analysis of the burden imposed on foreign products was inadequate".²⁸⁹ According to Japan, the

²⁷⁷ Panel Reports, para. 7.731 (referring to Panel Reports, paras. 7.656-7.661).

²⁷⁸ Panel Reports, para. 7.731 (referring to Panel Reports, paras. 7.656-7.661).

²⁷⁹ Panel Reports, para. 7.772.

²⁸⁰ Brazil's appellant's submission, para. 278 (referring to Panel Reports, para. 7.732).

²⁸¹ Brazil's appellant's submission, para. 287. (emphasis original)

²⁸² Brazil's appellant's submission, para. 287.

²⁸³ Brazil's appellant's submission, para. 288. (emphasis original)

²⁸⁴ European Union's appellee's submission, para. 359 (referring to Panel Reports, paras. 2.113-2.122).

²⁸⁵ European Union's appellee's submission, para. 362.

²⁸⁶ European Union's appellee's submission, para. 363 (quoting Panel Reports, para. 7.660 (fns omitted)).

²⁸⁷ European Union's appellee's submission, para. 364.

²⁸⁸ Japan's appellee's submission, para. 114 (quoting Brazil's appellant's submission, para. 288 (emphasis original)).

²⁸⁹ Japan's appellee's submission, para. 114.

- 39 -

Panel properly reached the conclusion that the INOVAR-AUTO programme, through the accreditation requirements, favours domestic over imported products and it performed an analysis that included all relevant elements.²⁹⁰

5.71. We begin by recalling that, under the INOVAR-AUTO programme, "all companies using presumed IPI tax credits, and certain companies using reduced IPI tax rates, must obtain one of three forms of accreditation"²⁹¹: (i) domestic manufacturers; (ii) importers/distributors; or (iii) investors.²⁹² The "purpose" of complying with the requirements for accreditation is "to obtain presumed IPI tax credits on the sale of products".²⁹³ Thus, in order to enjoy a reduction on IPI tax liability on a product, in this case motor vehicles, companies "must comply with a set of requirements"²⁹⁴, for example, the performance of a minimum number of defined manufacturing and engineering infrastructure activities in Brazil, or investments in R&D in Brazil.

5.72. It is undisputed that, in order for companies to obtain any sort of accreditation that entitles them to accrue and use presumed IPI tax credits, they must either be located and operate in Brazil, in the case of domestic manufacturers and importers/distributors, or be in the process of establishing in the country as domestic manufacturers, in the case of investors.²⁹⁵ This in turn implies, as the Panel also acknowledged, that "foreign companies exclusively located outside Brazil that manufacture products imported into Brazil cannot, per se, get accredited and, consequently, cannot accrue and use presumed IPI tax credits."²⁹⁶ The only viable way for foreign manufacturers to be able to enjoy the benefit of the presumed IPI tax credits in reducing their IPI tax liability under the INOVAR-AUTO programme is to become accredited as importers/distributors.²⁹⁷ However, in order to do so, foreign manufacturers must, first and foremost, be located and operate in Brazil.²⁹⁸ This indicates that foreign manufacturers seeking accreditation as importers/distributors face a corresponding burden that necessarily comes with having to operate, or establish themselves, in Brazil, unlike domestic manufacturers, who already operate or are established in Brazil. We, therefore, agree with the Panel that "unlike for domestic manufacturers, for foreign manufacturers compliance with the accreditation criteria necessarily entails establishing in Brazil, with a corresponding administrative and economic burden." 299

5.73. Moreover, we note that, in order to become accredited as importers/distributors, a company must comply with the following three specific requirements: (i) investments in R&D in Brazil; (ii) expenditure on engineering, basic industrial technology, and capacity-building of suppliers in Brazil; and (iii) participation in the vehicle-labelling programme by INMETRO.³⁰⁰ A fourth requirement also exists, which calls for the performance in Brazil of certain manufacturing steps.³⁰¹ These activities cannot be considered to be typical for foreign manufacturers seeking to import motor vehicles into Brazil. The fact that foreign manufacturers have to undertake these activities to get accredited as importers/distributors implies that foreign manufacturers face a burden that domestic manufacturers do not face. We also see merit in the Panel's reasoning that foreign manufacturers that are accredited as importers/distributors do not carry out manufacturing activities in Brazil, and there is no reason "why they would purchase strategic inputs and tools in Brazil for the manufacture of motor vehicles"³⁰² or, for that matter, invest in R&D or make expenditures on engineering, basic industrial technology, and capacity-building of suppliers in Brazil.

5.74. On the other hand, we observe that, in order to be accredited, domestic manufacturers need to comply with three out of four specific requirements. One of these must be the performance of a minimum number of defined manufacturing and engineering infrastructure activities in Brazil.³⁰³

²⁹⁰ Japan's appellee's submission, para. 114.

²⁹¹ Panel Reports, para. 2.112.

²⁹² Panel Reports, para. 2.113.

²⁹³ Panel Reports, para. 7.706.

²⁹⁴ Brazil's appellant's submission, para. 266.

²⁹⁵ Panel Reports, para. 7.657 (referring to Brazil's responses to Panel questions Nos. 28 and 57).

²⁹⁶ Panel Reports, para. 7.657.

²⁹⁷ The Panel noted that Brazil acknowledged that accreditation as investors is "a temporary accreditation which eventually becomes an accreditation as a manufacturer in Brazil". (Panel Reports, fn 1053

to para. 7.657 (referring to Brazil's response to Panel question No. 28))

²⁹⁸ Panel Reports, para. 7.660 (referring to Brazil's response to Panel question No. 57).

²⁹⁹ Panel Reports, para. 7.660.

³⁰⁰ Panel Reports, para. 7.658.

³⁰¹ Panel Reports, fn 1056 to para. 7.658.

³⁰² Panel Reports, para. 7.733 (referring to Panel Reports, paras. 7.662-7.672).

³⁰³ Panel Reports, para. 7.658.

- 40 -

The other two requirements must be among the following three: (i) investments in R&D in Brazil; (ii) expenditure on engineering, basic industrial technology, and capacity-building of suppliers in Brazil; or (iii) participation in the vehicle-labelling programme by INMETRO.³⁰⁴ Almost all of these requirements can be considered to be typical of the nature of activity carried out by a domestic manufacturer. Indeed, any domestic manufacturer will carry out and perform a minimum number of manufacturing activities in Brazil, and in that process is also likely to make investments in R&D in Brazil and make expenditures in the categories indicated in the INOVAR-AUTO programme.³⁰⁵ To that extent, we agree with the Panel that the performance of a minimum number of manufacturing activities in Brazil "is inherent to any domestic manufacturer".³⁰⁶ This in turn, as the Panel rightly noted, results in domestic manufacturers being subject *de facto* to only two other specific requirements.³⁰⁷

5.75. Therefore, the INOVAR-AUTO programme is designed in such a manner that the accreditation requirements thereunder adversely modify the "equality of competitive conditions for imported products"³⁰⁸ compared to like domestic products. This is so because, first, foreign manufacturers seeking accreditation as importers/distributors in order to enjoy a reduction on IPI tax liability have to operate or establish themselves in Brazil with the corresponding burden unlike domestic manufacturers, who already operate and are established in Brazil. Second, foreign manufacturers seeking accreditation as importers/distributors are required to comply with more accreditation requirements and undertake certain activities prescribed under the INOVAR-AUTO programme, which are, in any event, as we have considered above, not typical for foreign manufacturers seeking to import motor vehicles into Brazil. Brazil contends that "the Panel limited itself to conducting a quantitative analysis"³⁰⁹ and instead should have "conducted a qualitative analysis of the [accreditation] requirements".³¹⁰ Brazil does not, however, elaborate on the type or kind of qualitative analysis that the Panel should have undertaken. In any event, the Panel, in our view, conducted a qualitative analysis in as much as it recognized that "even if it were to consider that the number of requirements imposed on domestic manufacturers and importers/distributors is the same", foreign manufacturers bear a higher burden since "compliance with the accreditation criteria necessarily entails establishing in Brazil, with a corresponding administrative and economic burden", which is not faced by domestic manufacturers who are already established in Brazil.³¹¹ These considerations are sufficient to support the conclusion that "[f]oreign manufacturers ... bear a higher burden than domestic manufacturers in becoming accredited" in order for "their products to be eligible for the tax benefits".³¹² Thus, the accreditation requirements "modify the conditions of competition to the detriment of imported motor vehicles and in favour of like domestic motor vehicles".313

5.76. For these reasons, we agree with the Panel that, "under the INOVAR-AUTO programme, the conditions for accreditation in order to receive presumed tax credits ... accord less favourable treatment to imported products than that accorded to like domestic products" within the meaning of Article III:4 of the GATT 1994.³¹⁴

5.1.5 Whether the Panel erred in finding that the INOVAR-AUTO programme is inconsistent with Article 2.1 of the TRIMs Agreement

5.77. Before the Panel, the European Union and Japan argued that the INOVAR-AUTO programme is inconsistent with Article 2.1 of the TRIMs Agreement, both independently and in conjunction with Article 2.2 and paragraph 1(a) of that Agreement's Illustrative List, because it is inconsistent with

(Panel Reports, para. 7.659)

- ³⁰⁹ Brazil's appellant's submission, para. 287. (emphasis original)
- ³¹⁰ Brazil's appellant's submission, para. 288. (emphasis omitted)

³⁰⁴ Panel Reports, fn 1055 to para. 7.658.

³⁰⁵ We observe that the Panel also considered that one of the other requirements involving "investing in R&D in Brazil" is also "linked to the development of their manufacturing activity in the country".

³⁰⁶ Panel Reports, para. 7.659.

³⁰⁷ Panel Reports, para. 7.659.

³⁰⁸ Appellate Body Report, *Thailand – Cigarettes (Philippines)*, para. 126. (fn omitted)

³¹¹ Panel Reports, para. 7.660.

³¹² Panel Reports, para. 7.731.

³¹³ Panel Reports, para. 7.732.

³¹⁴ Panel Reports, para. 7.772.

- 41 -

Article III of the GATT 1994.³¹⁵ Brazil agreed that the INOVAR-AUTO programme is an investment measure, submitting, however, that it does not relate to trade in goods.³¹⁶

5.78. The Panel found that the INOVAR-AUTO programme constitutes a trade-related investment measure within the meaning of the TRIMs Agreement.³¹⁷ The Panel further recalled that it had found that "the INOVAR-AUTO programme is inconsistent with Article III:2 and III:4 of the GATT 1994."³¹⁸ The Panel therefore considered that "those aspects of the INOVAR-AUTO programme found to be inconsistent with Article **III:2** and **III:4** ... are also inconsistent with Article 2.1 of the TRIMs Agreement."³¹⁹

5.79. On appeal, Brazil does not make any specific arguments in connection with the Panel's finding under Article 2.1 of the TRIMs Agreement. Rather, as we understand it, Brazil's request for reversal of the Panel's finding under Article 2.1 of the TRIMs Agreement is premised on us reversing the Panel's findings under Article III:4 of the GATT 1994.³²⁰ We have, however, for the reasons stated above, agreed with the Panel's findings that certain aspects of the INOVAR-AUTO programme are inconsistent with Article III:4 of the GATT 1994. Consequently, we agree with the Panel's finding that those aspects of the INOVAR-AUTO programme found to be inconsistent with Article III:4 of the GATT 1994 are also inconsistent with Article 2.1 of the TRIMs Agreement.³²¹

5.2 Article III:8(b) of the GATT 1994³²²

5.2.1 Introduction

5.80. We now address Brazil's appeal of the Panel's interpretation and application of Article III:8(b) of the GATT 1994. Before examining the complainants' claims of inconsistencies under Article III of the GATT 1994, the Panel addressed the two "general defences" put forth by Brazil.³²³ Rejecting Brazil's first defence that the product-based disciplines of Article III are not applicable to measures directed at producers, the Panel found that "Article III of the GATT 1994 is not per se inapplicable to certain measures, in particular 'pre-market' measures directed at producers."³²⁴ With respect to Brazil's second defence that the measures at issue fall under Article III:8(b), the Panel found that "subsidies that are provided exclusively to domestic producers pursuant to Article III:8(b) of the GATT 1994 are not *per se* exempted from the disciplines of Article III of the GATT 1994."³²⁵ Instead. in the Panel's view, "aspects of a subsidy resulting in product discrimination (including requirements to use domestic goods, as prohibited by Article 3.1 of the SCM Agreement) are not exempted from the disciplines of Article III pursuant to Article III:8(b)."326 The Panel thus turned to assess the consistency of the challenged measures with Article III of the GATT 1994 and Article 2.1 of the TRIMs Agreement. The Panel found, inter alia, that certain aspects of the ICT programmes are inconsistent with Article III:2, first sentence, and Article III:4³²⁷, and that the accreditation requirements under the INOVAR-AUTO programme are inconsistent with Article III: 4.328 Flowing from its findings of inconsistency under Articles III:2 and III:4, the Panel also found the same

(Brazil's appellant's submission, para. 294 (emphasis original))

³²¹ Panel Reports, para. 7.804.

³¹⁵ Panel Reports, paras. 7.793 (referring to European Union's first written submission to the Panel, paras. 435-451; Japan's first written submission to the Panel, paras. 253-266).

³¹⁶ Panel Reports, para. 7.795.

³¹⁷ Panel Reports, para. 7.802.

³¹⁸ Panel Reports, para. 7.804 (referring to Panel Reports, paras. 7.645, 7.688, and 7.772-7.773).

³¹⁹ Panel Reports, para. 7.804.

³²⁰ Having sought reversal of the Panel's findings under Article III: 4 of the GATT 1994, specifically with respect to paras. 7.773, 8.6.b, and 8.17.c of the Panel Reports, Brazil submits that:

As a result, Brazil further requests that the Appellate Body *reverse* the Panel's consequential finding in paragraphs 7.806, 8.6(d) and 8.17(e) of the Panel Report that these aspects of [the] INOVAR-AUTO [programme] are inconsistent with Article 2.1 of the TRIMs Agreement.

³²² One Member of the Division expressed a separate opinion regarding the scope of the term "payment of subsidies" in Article III:8(b) of the GATT 1994. This separate opinion can be found in subsection 5.2.5 of these Reports.

³²³ Panel Reports, para. 7.53.

³²⁴ Panel Reports, para. 7.70. See also Panel Reports, para. 7.67.

³²⁵ Panel Reports, para. 7.87.

³²⁶ Panel Reports, para. 7.88.

³²⁷ Panel Reports, paras. 7.173-7.174, 7.230, 7.243, and 7.255.

³²⁸ Panel Reports, para. 7.772.

aspects of the ICT and INOVAR-AUTO programmes to be inconsistent with Article 2.1 of the TRIMs Agreement.329

5.81. On appeal, Brazil takes issue with the Panel's analysis of its second defence and claims that the Panel erred in its interpretation of Article III:8(b) in finding that "subsidies that are provided exclusively to domestic producers pursuant to Article III:8(b) of the GATT 1994 are not *per se* exempted from the disciplines of Article III of the GATT 1994."³³⁰ Brazil further contends that the Panel, premised on its erroneous interpretation³³¹, also erred in its application of Article III:8(b) in finding that: (i) the ICT programmes are inconsistent with Article III:2, first sentence, and Article III:4 of the GATT 1994 and Article 2.1 of the TRIMs Agreement; and (ii) the accreditation requirements under the INOVAR-AUTO programme are inconsistent with Article III:4 of the GATT 1994 and Article 2.1 of the TRIMs Agreement.³³² We begin our analysis by discussing, briefly, the legal standard under Article III:8(b) of the GATT 1994 before addressing Brazil's claims of error on appeal.

5.2.2 The legal standard under Article III:8(b) of the GATT 1994

5.82. Article III:8 of the GATT 1994 provides as follows:

(a) The provisions of this Article shall not apply to laws, regulations or requirements governing the procurement by governmental agencies of products purchased for governmental purposes and not with a view to commercial resale or with a view to use in the production of goods for commercial sale.

(b) The provisions of this Article shall not prevent the payment of subsidies exclusively to domestic producers, including payments to domestic producers derived from the proceeds of internal taxes or charges applied consistently with the provisions of this Article and subsidies effected through governmental purchases of domestic products.

5.83. Article III:8(b) states that the provisions of Article III "shall not prevent"³³³ the payment of subsidies exclusively to domestic producers. This language is comparable to the chapeau of Article XX of the GATT 1994, which states, in relevant part, that "nothing in this Agreement shall be construed to prevent the adoption or enforcement by any Member of measures" enumerated in paragraphs (a)-(j) of Article XX.³³⁴ It is well established that Article XX is an *affirmative defence* and sets out the "general exceptions" to the obligations contained in other provisions of the GATT 1994.³³⁵ In other words, recourse to Article XX serves as a justification for WTO Members to adopt and enforce measures that are found to be inconsistent with other provisions of the GATT 1994.

5.84. In contrast to the opening clause of Article III:8(b), Article III:8(a) begins with the words "[t]he provisions of this Article shall not apply to" the measures enumerated thereunder.³³⁶ The Appellate Body in Canada - Renewable Energy / Canada - Feed-in Tariff Program explained that:

The opening clause of Article III:8(a) uses the term "apply" in the negative, thus precluding the application of the other provisions of Article III to measures that meet the requirements of that paragraph. Article III:8(a) therefore establishes a derogation from the national treatment obligation of Article III for government procurement activities falling within its scope. Measures satisfying the requirements of Article III:8(a) are not subject to the national treatment obligations set out in other paragraphs of Article III. Article III:8(a) is a derogation limiting the scope of the national treatment

³²⁹ Panel Reports, paras. 7.363 and 7.804.

³³⁰ Brazil's appellant's submission, paras. 5 and 16 (quoting Panel Reports, para. 7.87). See also ibid., paras. 31 and 34.

³³¹ Brazil's appellant's submission, para. 35.

³³² Brazil's appellant's submission, paras. 35-36 (referring to Panel Reports, paras. 7.174, 7.318-7.319,

^{7.365, 7.688, 7.772-7.773, 7.806, 8.5.}a-b, 8.5.d, 8.6.a-b, and 8.6.d).

³³³ Emphasis added.

³³⁴ Emphasis added.

³³⁵ Appellate Body Report, Indonesia – Import Licensing Regimes, para. 5.42 (referring to

Appellate Body Report, US - Wool Shirts and Blouses, p. 16, DSR 1997:1, p. 337). See also Appellate Body Report, *Thailand – Cigarettes*, para. 177. ³³⁶ Emphasis added.

obligation and it is not a justification for measures that would otherwise be inconsistent with that obligation. $^{\rm 337}$

The opening clause of Article III:8(a) thus makes clear that the provision is a *derogation* limiting the scope of the national treatment obligation by making it inapplicable to certain government procurement activities. By contrast, the differently worded opening clause of Article III:8(b), which is similar to the text of the *chapeau* of Article XX, suggests to us that the provision is akin to an *exception* to the national treatment obligation and serves as a justification or affirmative defence for measures that would otherwise be inconsistent with that obligation.³³⁸ Thus, while Article III:8(a) *precludes* the application of the national treatment obligation in Article III to government procurement activities falling within its scope, Article III:8(b) provides a *justification* for measures that would otherwise be inconsistent with the national treatment obligation in Article III.³³⁹

5.85. Turning to the term "payment of subsidies" used in Article III:8(b), we note that neither "payment" nor "subsidies" is defined in the GATT 1994. Although the term "subsidies" is used in several other provisions of the GATT 1994, and Article 1.1 of the SCM Agreement defines what constitutes a "subsidy", Article III:8(b) uses the term "*payment* of subsidies", and not "subsidies" alone. The interpretative issue at hand therefore relates not to the definition of "subsidies" under the GATT 1994 generally or under the SCM Agreement, but focuses instead on the precise scope of the term "payment of subsidies" as used in Article III:8(b), in particular.

5.86. The dictionary meanings of "payment" include "[a] sum of money (or equivalent) paid or payable" and "the remuneration of a person with money or its equivalent".³⁴⁰ We note that whereas the conduct made permissible by Article III:8(b) is the "*payment* of subsidies", other provisions of the GATT 1994, including Article XVI, and the SCM Agreement, dealing with subsidies, use the terms "grant" or "maintain" when referring to subsidies.³⁴¹ This difference in language suggests to us that the term "payment of subsidies" in Article III:8(b) encompasses a narrower range of conduct than that covered by the terms "subsidy" or the "granting" or "maintaining" of a subsidy, as used elsewhere in the GATT 1994 and in the SCM Agreement.

5.87. In *Canada – Periodicals*, the Appellate Body was tasked with determining whether the reduced postal rates for certain eligible Canadian publishers constituted a "payment of subsidies" within the meaning of Article III:8(b).³⁴² Answering in the negative, the Appellate Body concluded that Article III:8(b) "was intended to exempt from the obligations of Article III only the payment of subsidies which involves the expenditure of revenue by a government".³⁴³ In this regard, the Appellate Body quoted and expressly agreed with the GATT panel in *US – Malt Beverages* that:

Article III:8(b) limits, therefore, the permissible producer subsidies to "payments" after taxes have been collected or payments otherwise consistent with Article III. This separation of tax rules, e.g. on tax exemptions or reductions, and subsidy rules makes sense economically and politically. Even if the proceeds from non-discriminatory product

³³⁷ Appellate Body Reports, *Canada – Renewable Energy / Canada – Feed-in Tariff Program*, para. 5.56 (referring to Appellate Body Reports, *China – Raw Materials*, para. 334).

³³⁸ See Appellate Body Reports, *Canada – Renewable Energy / Canada – Feed-in Tariff Program*, para. 5.56 (referring to Appellate Body Reports, *China – Raw Materials*, para. 334).

³³⁹ It is in this vein that the Appellate Body in *US* – *Tax Incentives* observed that Article III:8(b) "exempts from the national treatment obligation in Article III 'the payment of subsidies exclusively to domestic producers'". (Appellate Body Report, *US* – *Tax Incentives*, para. 5.16)

³⁴⁰ Oxford English Dictionary online, definition of "payment", http://www.oed.com/view/Entry/139189?rskey=9c29D2&result=1#eid.

³⁴¹ Article XVI:1 of the GATT 1994, for example, provides, in relevant part:

If any Member grants or maintains any subsidy, including any form of income or price support, which operates directly or indirectly to increase exports of any product from, or to reduce imports of any product into, its territory, it shall notify the WTO in writing of the extent and nature of the subsidization, of the estimated effect of the subsidization on the quantity of the affected product or products imported into or exported from its territory and of the circumstances making the subsidization necessary. In any case in which it is determined that serious prejudice to the interests of any other Member is caused or threatened by any such subsidization, the Member granting the subsidy shall, upon request, discuss with the other Member or Members concerned, or with the WTO, the possibility of limiting the subsidization. (emphasis added)

³⁴² Appellate Body Report, *Canada – Periodicals*, pp. 33-34.

³⁴³ Appellate Body Report, *Canada – Periodicals*, p. 34.

- 44 -

taxes may be used for subsequent subsidies, the domestic producer, like his foreign competitors, must pay the product taxes due. The separation of tax and subsidy rules contributes to greater transparency. It also may render abuses of tax policies for protectionist purposes more difficult, as in the case where producer aids require additional legislative or governmental decisions in which the different interests involved can be balanced.344

5.88. Turning to the immediate context of the term "payment of subsidies", Article III:8(b) defines this term as "including payments to domestic producers derived from the proceeds of internal taxes or charges applied consistently with the provisions of [Article III]". It also includes "subsidies effected through governmental purchases of domestic products". Although the use of the word "including" makes clear that these two examples in Article III:8(b) are not meant to provide an exhaustive list of what constitutes the "payment of subsidies", the Appellate Body in Canada - Periodicals considered that they nonetheless "exemplify the kinds of programmes which are exempted from the obligations of Articles III: 2 and III: 4 of the GATT 1994". ³⁴⁵ These examples must therefore be given due importance in the interpretation of the term "payment of subsidies".

5.89. The text of the first example, namely, "payments to domestic producers derived from the proceeds of internal taxes or charges applied consistently with the provisions of [Article III]", makes it clear that it is not the payment of subsidies that must be consistent with the obligations under Article III of the GATT. Instead, it is the internal taxes applied to products, the proceeds of which are used for the payment of subsidies, which must be consistent with the obligations under Article III.³⁴⁶ When these internal taxes are applied in a manner consistent with Article III, the proceeds derived from such taxes may be used for payments of subsidies exclusively to domestic producers, and such payments of subsidies, as well as any resulting discrimination against like imported products, will be justified under Article III:8(b).³⁴⁷ However, when the internal taxes are higher on imported products than on like domestic products, or otherwise accord less favourable treatment to imported products, and are thus inconsistent with Article III, the payment of subsidies derived from the proceeds of such GATT-inconsistent taxes would not be justified under Article III:8(b). In other words, the text of the first example suggests that subsidies that are paid through the proceeds of discriminatory internal taxes applied, directly or indirectly, on products continue to be subject to the obligations in Article III.³⁴⁸ We note in this regard that the Appellate Body in Canada - Periodicals agreed with the GATT panel in US - Malt Beverages that, "[e]ven if the proceeds from non-discriminatory product taxes may be used for subsequent subsidies, the domestic producer, like his foreign competitors, must pay the product taxes due."349

5.90. In addition to the immediate context, we consider that Article III:2, which sets out the national treatment obligation with respect to internal tax measures, also provides relevant context for the interpretation of the term "payment of subsidies". As we see it, the prohibition against discriminatory internal taxes in Article III:2 may be rendered ineffective if discriminatory internal taxes on imported products could be justified as subsidies for competing domestic producers in terms

³⁴⁴ Appellate Body Report, Canada - Periodicals, p. 34 (quoting GATT Panel Report, US - Malt Beverages, para. 5.10). ³⁴⁵ Appellate Body Report, Canada – Periodicals, pp. 33-34.

³⁴⁶ We note that subsidies to domestic producers can be paid in product markets other than those from which the tax proceeds are derived.

³⁴⁷ Insofar as the payments of subsidies are derived from the general budget, and cannot be traced to a particular tax, they would, of course, be justified under III:8(b).

³⁴⁸ Giving due importance to the examples means that one cannot accept that the opposite of the situation reflected in the first example, namely the payment of subsidies derived from proceeds of taxes applied inconsistently with Article III, would also be covered by Article III:8(b). One cannot but understand the inclusion in the scope of Article III:8(b) of payments of subsidies derived from proceeds of taxes consistent with Article III as the exclusion from the scope of Article III:8(b) of payments of subsidies derived from proceeds of taxes inconsistent with Article III.

³⁴⁹ Appellate Body Report, Canada - Periodicals, p. 34 (quoting GATT Panel Report, US - Malt Beverages, para. 5.10, where the GATT panel further explained that "[t]he separation of tax and subsidy rules contributes to greater transparency. It also may render abuses of tax policies for protectionist purposes more difficult, as in the case where producer aids require additional legislative or governmental decisions in which the different interests involved can be balanced.").

- 45 -

of Article III:8(b).³⁵⁰ For example, instead of applying differential tax rates on imported and like domestic products, a WTO Member could apply the same tax rate to imported and like domestic products and subsequently provide for a reduction of the tax rate for products produced by domestic producers, but not those produced by foreign producers. Indeed, if the scope of "payment of subsidies" is seen as encompassing an exemption or reduction of internal product taxes that are "otherwise due", it would allow WTO Members to circumvent Article III:2 and adopt discriminatory tax measures by disguising them in the form of a scheme of exemption or reduction of internal product taxes for domestic producers alone.

5.91. Our above discussion is supported by the negotiating history of Article III:8(b), which was also relied on by the Appellate Body in Canada - Periodicals.³⁵¹ Specifically, we consider relevant the following discussion from the Reports of the Committees and Principal Sub-Committees of the Interim Commission for the International Trade Organization concerning Article 18 of the Havana Charter, which corresponds to Article III of the GATT 1994:

This sub-paragraph [i.e. what is now Article III:8(b) of the GATT 1994] was redrafted in order to make it clear that nothing in Article [III] could be construed to sanction [i.e. to allow for] the exemption of domestic products from internal taxes imposed on like imported products or the remission of such taxes. At the same time the Sub-Committee recorded its view that nothing in this sub-paragraph or elsewhere in Article [III] would override the provisions [of Article XVI].³⁵²

5.92. An examination of the text and context of Article III:8(b), as supported by its negotiating history, therefore suggests that the term "payment of subsidies" in Article III:8(b) does not include within its scope the exemption or reduction of internal taxes applied, directly or indirectly, on domestic products. Instead, as the Appellate Body has observed, Article III:8(b) "was intended to exempt from the obligations of Article III only the payment of subsidies which involves the expenditure of revenue by a government". 353

5.93. Regarding the phrase "exclusively to domestic producers", we note that the dictionary meaning of "exclusively" is "[s]o as to exclude all except some particular object, subject, etc.; solely."³⁵⁴ Placed in its context, the use of the term "exclusively" therefore indicates that Article III:8(b) exempts from the disciplines of Article III those "payments of subsidies" that are made solely to domestic producers, to the exclusion of foreign producers. Subsidies provided exclusively to domestic producers will often, if not always, have an impact on the conditions of competition between the product produced by the subsidized domestic producers (for example, by lowering the cost of production of such products) and the like imported product produced by foreign producers that are not paid the subsidy.³⁵⁵ In the absence of Article III:8(b), such subsidies paid exclusively to domestic producers could therefore be seen as being inconsistent with the broadly worded national treatment obligation in Article III insofar as they alter the conditions of competition in favour of the product produced by the domestic producer to whom the subsidy is paid.³⁵⁶

http://www.oed.com/view/Entry/65834?redirectedFrom=exclusively#eid.

³⁵⁵ In US – Tax Incentives, the Appellate Body observed that "[s]ubsidies that relate to domestic production ... can ordinarily be expected to increase the supply of the subsidized domestic goods in the relevant market". (Appellate Body Report, US - Tax Incentives, para. 5.15)

³⁵⁰ See e.g. GATT Panel Report, US - Malt Beverages, para. 5.9. The GATT panel further noted that an "expansive interpretation of Article III:8(b) ..., if carried to its logical conclusion, ... would virtually eliminate the prohibition in Article III: 2 of discriminatory internal taxation by enabling contracting parties to exempt all domestic products from indirect taxes." Ibid., para. 5.12.

³⁵¹ Appellate Body Report, Canada - Periodicals, p. 34.

³⁵² Interim Commission for the International Trade Organization, Reports of the Committees and Principal Sub-Committees: ICITO I/8, Geneva, September 1948, p. 66. Article 18 and Section C of Chapter IV of the Havana Charter for an International Trade Organization correspond, respectively, to Article III and Article XVI of the GATT 1947. See also Appellate Body Report, Canada - Periodicals, p. 34.

 ³⁵³ Appellate Body Report, *Canada – Periodicals*, p. 34.
 ³⁵⁴ Oxford English Dictionary online, definition of "exclusively",

³⁵⁶ In this regard, we note that the chair of the working party of Sub-Committee A, which redrafted Article 18 of the Havana Charter (which corresponds to Article III of the GATT 1994), acknowledged that the text of Article III:8(b) was added "because it was felt that if subsidies were paid on domestic and not on imported products, it might be construed that Members were not applying the 'national treatment' rule." (E/CONF.2/c.3/A/W.49, p. 2)

- 46 -

5.94. In US - Tax Incentives, the Appellate Body observed that Article III:8(b) "makes clear that the provision of subsidies to domestic producers only, and not to foreign ones, does not in itself constitute a breach of Article III."³⁵⁷ Insofar as the payment of a subsidy only to domestic producers, to the exclusion of foreign producers, affects the conditions of competition in the relevant product market(s), Article III:8(b) carves out an exception for the payment of such subsidies from the national treatment obligation under Article III. In other words, to the extent that the payment of subsidies exclusively to domestic producers of a given product affects the conditions of competition between such a product and the like imported product, resulting in an inconsistency with the national treatment obligation in Article III, such a payment would be justified under the exception contained in Article III:8(b), provided that the conditions thereunder are met.³⁵⁸

5.95. Moreover, besides the effect of the payment of subsidies exclusively to domestic producers on the conditions of competition in the relevant product market(s), there will often be conditions for eligibility that attach to such payments. For instance, insofar as Article III:8(b) justifies the payment by WTO Members of subsidies exclusively to domestic producers, conditions for eligibility that define the class of eligible "domestic producers" by reference to their activities in the subsidized products' markets would be justified under Article III:8(b). By contrast, a requirement to use domestic over imported goods in order to have access to the subsidy may, however, not be covered by the exception in Article III:8(b) and would therefore continue to be subject to the national treatment obligation in Article III. This is because, while the payment of subsidies and certain eligibility criteria may affect the conditions of competition between the product produced by the producer receiving the subsidy and the like imported products, a requirement to use domestic products in order to have access to the subsidy would impact the conditions of competition between a different set of domestic and like imported products, namely, the domestic product whose use is mandated and the like imported product. 359

5.96. Turning to the term "domestic producers", as used in Article III:8(b), we note that the dictionary meaning of "producer" is "[a] person who ... produces (in various senses)". 360 The scope of Article III:8(b) suggests that the focus of inquiry under that provision ought to be on whether the domestic entity at issue is a *producer* of the product with respect to which a violation of the national treatment obligation arising from the "payment of subsidies" is alleged. This is because Article III:8(b) serves as a justification only for discrimination resulting from the effects of the payment of a subsidy on the conditions of competition in the relevant product market(s). Therefore, whether a domestic entity is a "domestic producer" within the meaning of Article III:8(b) is a question that must be answered in light of the specific facts and circumstances of a given case, including the nature of discrimination that is alleged.

5.2.3 Whether the Panel erred in its interpretation and application of Article III:8(b) of the GATT 1994

5.97. Having set out the legal standard under Article III:8(b), we now turn to address Brazil's claim that the Panel erred in its interpretation and application of Article III:8(b) of the GATT 1994. Brazil claims that the Panel erred in its interpretation of Article III:8(b) in finding that "subsidies that are provided exclusively to domestic producers pursuant to Article III:8(b) of the GATT 1994 are not per se exempted from the disciplines of Article III of the GATT 1994."³⁶¹ Brazil further contends that the Panel, premised on its erroneous interpretation³⁶², also erred in its application of Article III:8(b) in finding that: (i) the ICT programmes are inconsistent with Article III:2, first sentence, and Article III:4 of the GATT 1994 and Article 2.1 of the TRIMs Agreement; and (ii) the accreditation

³⁵⁷ Appellate Body Report, US - Tax Incentives, para. 5.16. (emphasis added)

³⁵⁸ That said, the payment of such subsidies and the effects thereof would continue to be regulated by other relevant provisions of the GATT 1994 and the SCM Agreement.

³⁵⁹ The Appellate Body in US - Tax Incentives observed that, "[t]o the extent that 'domestic producers' may generally be expected to manufacture a certain amount of 'domestic goods' in a Member's territory, Article III:8(b) comports with our reading of Article 3.1(b) under which something more than mere subsidization of domestic production is required for finding an import substitution subsidy." (Appellate Body Report, US - Tax Incentives, para. 5.16)

³⁶⁰ Oxford English Dictionary online, definition of "producer",

http://www.oed.com/view/Entry/151981?redirectedFrom=producer#eid.

³⁶¹ Brazil's appellant's submission, paras. 5 and 16 (quoting Panel Reports, para. 7.87). See also ibid., paras. 31 and 34. ³⁶² Brazil's appellant's submission, para. 35.

requirements under the INOVAR-AUTO programme are inconsistent with Article III:4 of the GATT 1994 and Article 2.1 of the TRIMs Agreement.³⁶³

5.98. According to Brazil, the Panel's interpretation and application of Article III:8(b) erroneously "conflate[] the distinction between the product-related disciplines of Article III and the provisions of the covered agreements concerning subsidies provided to domestic producers, which should be assessed under the SCM Agreement".³⁶⁴ Brazil takes the overarching position that Article III:8(b) "delineat[es]", in a mutually exclusive manner, the respective scopes of application of the "product-related" disciplines of Article III, on the one hand, and the disciplines of Article XVI and the SCM Agreement concerning "producer subsidy measures", on the other hand.³⁶⁵ In Brazil's view, a "harmonious interpretation"³⁶⁶ of these different provisions requires any panel evaluating claims under Article III to determine, as a "threshold analysis",³⁶⁷ whether the measure at issue is a "product-related" measure subject to the disciplines of Article III or whether the measure provides a subsidy to domestic producers and is therefore subject to the disciplines of the SCM Agreement.³⁶⁸ Brazil faults the Panel for not undertaking such a "threshold inquiry".³⁶⁹ In support of its claim, Brazil advances several arguments focusing on different interpretative elements of Article III:8(b), including the introductory clause, the scope of the term "payment of subsidies", and the examples set out in that provision.

5.99. The European Union responds that Brazil's "formalistic interpretation", drawing a "strict line" between the discipline under Article III and the provisions relating to subsidies, goes against the principle of "coherent interpretation".³⁷⁰ Given that the disciplines under Article III of the GATT 1994 and the SCM Agreement are not mutually exclusive, the European Union maintains that a measure may be found to be inconsistent with both sets of disciplines, as in the present case.³⁷¹ Moreover, in the European Union's view, Article III:8(b) cannot be interpreted as excluding the manner in which WTO Members condition the granting of subsidies to domestic producers only because the subsidies are given only to domestic producers. Thus, although the European Union agrees that the product-related effects of payments granted exclusively to domestic producers cannot result in discrimination contrary to Article III, it disagrees that the provision of subsidies containing a requirement to use domestic over imported inputs as a condition to benefit from the subsidy can fall under the carve-out in Article III:8(b).³⁷² According to the European Union, there was no intention to "compartmentalise" the obligations in case of measures consisting of subsidies exclusively within Article XVI of the GATT 1947, nor was that the intention of negotiators during the Uruguay Round when agreeing to Article 3.1(b) of the SCM Agreement.³⁷³

5.100. Japan asserts that the ordinary meaning of the text of Article III:8(b) renders it clear that the provision does not constitute a blanket exemption for subsidies paid exclusively to domestic producers. Instead, for Japan, measures that take the form of subsidies paid exclusively to domestic producers may still be subject to the disciplines under Article III if they, or certain elements thereof, result in discrimination between products.³⁷⁴ For Japan, therefore, if a subsidy is not only contingent on the fact that the recipient is a domestic producer, but also on the use of local content, then this "element" discriminating between foreign and domestic products would be a violation of the national treatment obligation.³⁷⁵ Moreover, Japan also considers that Article III:8(b) applies to not all, but only "some types" of, subsidies to domestic producers.³⁷⁶

5.101. We begin by recalling our discussion above that the national treatment obligation under Article III of the GATT 1994, by the terms of that provision, has a broad scope of application. In

- ³⁶⁵ Brazil's appellant's submission, para. 25.
- ³⁶⁶ Brazil's appellant's submission, para. 26.
- ³⁶⁷ Brazil's appellant's submission, para. 33.
- ³⁶⁸ Brazil's appellant's submission, para. 26.
- ³⁶⁹ Brazil's appellant's submission, para. 30.
- ³⁷⁰ European Union's appellee's submission, para. 42.
- ³⁷¹ European Union's appellee's submission, para. 44.
- ³⁷² European Union's appellee's submission, para. 57.
- ³⁷³ European Union's appellee's submission, para. 45.
- ³⁷⁴ Japan's appellee's submission, paras. 17-19 (referring to Panel Report, *Indonesia Autos*,
- para. 14.43)
 - ³⁷⁵ Japan's appellee's submission, para. 20.
 - ³⁷⁶ Japan's appellee's submission, para. 42.

³⁶³ Brazil's appellant's submission, paras. 35-36 (referring to Panel Reports, paras. 7.174, 7.318-7.319, 7.365, 7.688, 7.772-7.773, 7.806, 8.5.a-b, 8.5.d, 8.6.a-b, and 8.6.d).

³⁶⁴ Brazil's appellant's submission, para. 26.

- 48 -

particular, a discussion of the relevant WTO jurisprudence reveals that, although the national treatment obligation in Articles III:2 and III:4 is made effective in the context of "products", this does not *ipso facto* suggest that measures that are primarily directed at "producers" are excluded from that obligation.³⁷⁷ We thus agree with the Panel's view that, based on the plain text of Article III and relevant WTO jurisprudence, "Article III of the GATT 1994 is not *per se* inapplicable to certain measures, in particular 'pre-market' measures directed at producers."³⁷⁸ The Panel, in our view, **correctly observed that, "if the formalistic approach ... were correct, it would be simple to entirely** avoid the bedrock national treatment requirement of the multilateral trading system."³⁷⁹

5.102. Having agreed with the Panel that the text of Articles III:2 and III:4 does not, in and of itself, exclude from the scope of the national treatment obligation measures that are directed at "producers", we turn to Brazil's argument that Article III:8(b) "delineat[es]", in a *mutually exclusive* manner, the respective scopes of application of the "product-related" disciplines of Article III, on the one hand, and the disciplines of Article XVI and the SCM Agreement concerning "producer subsidy measures", on the other hand.³⁸⁰ On this basis, Brazil claims that the Panel erred in its interpretation of Article III:8(b) in finding that "subsidies that are provided exclusively to domestic producers pursuant to Article III:8(b) of the GATT 1994 are not *per se* exempted from the disciplines of Article III of the GATT 1994."³⁸¹

5.103. The Panel began its analysis under Article III:8(b) of the GATT 1994 by noting, first, that "at a minimum" the text of the provision makes it clear that "subsidies given exclusively to domestic producers do not *per se* and for that reason alone violate Article III of the GATT 1994."³⁸² The Panel explained that, in fact, "subsidies as such are not regulated by Article III, but rather by the provisions of Article XVI of the GATT 1994 and by the SCM Agreement."³⁸³ The Panel further noted that, "without Article III:8(b), Article III as a whole, and Article III:4 in particular, might be seen as prohibiting all subsidization that was provided only to domestic, and not to foreign, producers. This is because both Article III:1 ... and Article III:4 speak of, and discipline, *inter alia* regulatory measures 'affecting' the sale, offering for sale, purchase, transportation, distribution, or use, of products."³⁸⁴ The Panel considered it "clear that a subsidy is a form of regulatory measure that can and often does 'affect' the sale, purchase, etc. of products in ways that create advantages in the domestic market *vis-à-vis* products that have not benefited from the subsidy, including for example imported products".³⁸⁵ In the Panel's view, providing a "competitive advantage" in relation to market conditions is, in fact, the "typical intention" behind subsidization.³⁸⁶

5.104. For the Panel, "the 'adverse effects' provisions of the subsidy disciplines in Article XVI of the GATT 1994 and the SCM Agreement are 'precisely aimed' at the adverse trade effects that can be caused by the competitive advantages provided by subsidization."³⁸⁷ Thus, the Panel explained that, "in the absence of a provision such as Article III:8(b), Article III:4 might be read to require governments of importing Members to provide subsidies to foreign competitor firms whenever they subsidize their own domestic firms; or alternatively to prohibit all subsidies provided only to domestic and not also to foreign producers."³⁸⁸ For the Panel, "[s]uch an approach would be inconsistent with the very existence of the SCM Agreement, which in principle permits subsidies, except for two precise types of prohibited subsidies (namely export contingent subsidies and import substitution subsidies), and contains no requirement that subsidies, to be permitted, must be provided to foreign as well as domestic recipients."³⁸⁹ On this basis, the Panel found that the exclusive provision of subsidies to domestic producers (or any eventual effects thereof in the domestic market) does not

³⁸¹ Brazil's appellant's submission, paras. 16, 31, and 34 (quoting Panel Reports, para. 7.87).

³⁸⁹ Panel Reports, para. 7.78.

³⁷⁷ Appellate Body Report, *China – Publications and Audiovisual Products*, para. 227; GATT Panel Report, *US – Section 337 Tariff Act*, para. 5.10; Panel Report, *Argentina – Hides and Leathers*, para. 11.211; Appellate Body Report, *Thailand – Cigarettes (Philippines)*, para. 114. See also paragraphs 5.14-5.16 above.

Appellate Body Report, *Thailand – Cigarettes (Philippines)*, para. 114. See also paragraphs 5.14-5.16 above. ³⁷⁸ Panel Reports, para. 7.70.

³⁷⁹ Panel Reports, para. 7.63.

³⁸⁰ Brazil's appellant's submission, para. 25.

³⁸² Panel Reports, para. 7.77.

³⁸³ Panel Reports, para. 7.77.

³⁸⁴ Panel Reports, para. 7.78.

³⁸⁵ Panel Reports, para. 7.78.

³⁸⁶ Panel Reports, para. 7.78.

³⁸⁷ Panel Reports, para. 7.78.

³⁸⁸ Panel Reports, para. 7.78.

- 49 -

by itself constitute discriminatory treatment with respect to imported products, as prohibited by Article III.³⁹⁰

5.105. Next, the Panel addressed Brazil's argument that "Article III:8(b) effectively *exempts* all tax and regulatory discrimination between imported and domestic products from the disciplines of Article III to the extent that the measures in question are subsidies to domestic producers."³⁹¹ The Panel noted, first, that "multilateral rules on subsidies have coexisted with those on national treatment – including the proviso in Article III (Article III:8(b)) that clarifies the nature of that coexistence – since the entry into force of the GATT 1947."³⁹² According to the Panel, the adoption of the SCM Agreement did not alter this "basic harmonious coexistence".³⁹³ The Panel considered that "the negotiating history of the Uruguay Round suggest[ed] that the intention of Article III:4 of the GATT 1994, of subsidies contingent upon the use of domestic over imported goods."³⁹⁴ The Panel also noted that "the provisions on discrimination in Article III of the GATT 1994 and the provisions of the SCM Agreement can apply to the same measure simultaneously."³⁹⁵

5.106. Concerning the element of discrimination that could be introduced by a subsidy, the Panel viewed the text of Article III:8(b) as confirming that, "even if a measure is a subsidy that is provided exclusively to domestic producers, this fact is not sufficient to remove the measure from the application of Article III."³⁹⁶ Specifically, with respect to tax-based measures, the Panel noted that Article III:8(b) allows WTO Members to provide subsidies exclusively to domestic products using the proceeds of internal taxes or charges, as long as those taxes or charges are applied consistently with Article III.³⁹⁷ For the Panel, therefore, if "Article III:8(b) exempts tax discrimination from the scope of Article III, the reference in Article III:8(b) itself to 'taxes and charges applied consistently with the provisions of [Article III]' would be meaningless."³⁹⁸ On this basis, the Panel concluded that "Article III:8(b) does not change the applicability of Article III to discriminatory application of a product tax, even where such a discriminatory application constituted a subsidy exclusively to domestic producers."³⁹⁹ The Panel found support for its conclusion in the *travaux préparatoires* of the Havana Charter.⁴⁰⁰ Furthermore, the Panel was of the view that discriminatory non-tax regulatory measures that involve the provision of a subsidy exclusively to domestic producers are also, for that reason alone, not outside the disciplines of Article III: 4, given that the text of Article III:8(b) refers to Article III in its entirety and not just Article III:2.401

5.107. Having agreed with the panel in *Indonesia – Autos* that "a subsidy to domestic producers that introduces discrimination between imported and domestic like products is covered by – and inconsistent with – the provisions of Article III⁴⁰², the Panel thus concluded that "subsidies that are provided exclusively to domestic producers pursuant to Article III:8(b) of the GATT 1994 are not *per se* exempted from the disciplines of Article III of the GATT 1994."⁴⁰³ Specifically, the Panel considered that the "aspects of a subsidy resulting in product discrimination (including requirements)

- ³⁹⁵ Panel Reports, paras. 7.39-7.49 and 7.82.
- ³⁹⁶ Panel Reports, para. 7.85.
- ³⁹⁷ Panel Reports, para. 7.85.
- ³⁹⁸ Panel Reports, para, 7.85.
- ³⁹⁹ Panel Reports, para. 7.85.

⁴⁰⁰ Panel Reports, para. 7.85 and fn 479 thereto (referring to E/CONF.2/C.3/6, p. 17;

E/CONF.2/C.3/A/W.32, p. 2). The Panel observed that:

The Reports of the Committees and Principal Sub-Committees of the Interim Commission for the International Trade Organization concerning the provision of the Havana Charter for an International Trade Organization that corresponds to Article III:8(b) of the GATT 1994 state that: "nothing in Article 18 could be construed to sanction the exemption of domestic products from internal taxes imposed on like imported products or the remission of such taxes".

(Ibid. (quoting Interim Commission for the International Trade Organization, Reports of the Committees and Principal Sub-Committees: ICITO I/8, Geneva, September 1948, p. 66))

⁴⁰¹ Panel Reports, para. 7.86.

⁴⁰² Panel Reports, para. 7.83 (referring to Panel Report, *Indonesia – Autos*, paras. 14.43-14.46).

⁴⁰³ Panel Reports, para. 7.87.

³⁹⁰ Panel Reports, para. 7.79.

³⁹¹ Panel Reports, para. 7.80.

³⁹² Panel Reports, para. 7.82.

³⁹³ Panel Reports, para. 7.82.

³⁹⁴ Panel Reports, para. 7.82

- 50 -

to use domestic goods, as prohibited by Article 3.1 of the SCM Agreement) are not exempted from the disciplines of Article III pursuant to Article III:8(b)."⁴⁰⁴

5.108. We recall that, in US - Tax Incentives, the Appellate Body stated that "Article III:8(b) makes clear that the provision of subsidies to domestic producers only, and not to foreign ones, does not *in itself* constitute a breach of Article III."⁴⁰⁵ As explained in paragraph 5.94 above, insofar as the payment of subsidies exclusively to domestic producers, to the exclusion of foreign producers, affects the conditions of competition in the relevant product market(s), the resulting inconsistency with the national treatment obligation in Article III will be justified under the exception contained in Article III:8(b), provided that the conditions thereunder are met.⁴⁰⁶ We therefore agree with the Panel's view that the "exclusive provision of subsidies (or any eventual effects therefrom in the domestic market) does not by itself constitute discriminatory treatment in respect of imported products of the type prohibited by Article III."⁴⁰⁷ That said, as discussed above, we note that other aspects of a measure directed at producers that go beyond the mere payment of subsidies exclusively to domestic producers to have access to the subsidy, may not be covered by the exception in Article III:8(b) and would therefore continue to be subject to the national treatment obligation in Article III:8(b) and would therefore continue to be subject to the national treatment obligation in Article III.

5.109. Although we agree with the Panel's preliminary observations that discrimination resulting from the payment of subsidies exclusively to domestic producers and the market effects thereof may be justified under Article III:8(b), we have several concerns about the Panel's subsequent analysis leading to its conclusion that "aspects of a subsidy resulting in product discrimination (including requirements to use domestic goods, as prohibited by Article 3.1 of the SCM Agreement) are not exempted from the disciplines of Article III pursuant to Article III:8(b)."⁴⁰⁹ We note, in particular, the Panel's unqualified reference to "aspects of a subsidy", as well as its use of the term "including", when referring to domestic content requirements in the statement quoted above. The Panel adopts a similar view when describing the findings of the panel in *Indonesia – Autos*, noting that the "panel found that Article III:8(b) confirms that subsidies to domestic producers do not violate Article III so long as they do not have *any component* that introduces discrimination between imported and domestic products".⁴¹⁰

5.110. The Panel's interpretation appears to contradict its earlier finding that the "exclusive provision of subsidies (or any eventual effects thereof in the domestic market) does not by itself constitute discriminatory treatment in respect of imported products of the type prohibited by Article III."⁴¹¹ As part of our analysis above, we have agreed with this statement by the Panel, as well as with its observation that "providing a competitive advantage in relation to market conditions is a typical intention behind subsidization".⁴¹² Accepting the Panel's conclusion would suggest that virtually all of the subsidies paid exclusively to domestic producers would be subject to the national treatment obligation set out in the other paragraphs of Article III and would not be justifiable under Article III:8(b). This is because, as the Panel itself acknowledged, subsidies provided to domestic producers will almost always have an impact on the conditions of competition between the product produced by the subsidized domestic producers and the like imported product produced by foreign

⁴⁰⁴ Panel Reports, para. 7.88.

⁴⁰⁵ Appellate Body Report, US - Tax Incentives, para. 5.16. (emphasis added)

⁴⁰⁶ We recall that the chair of the working party of Sub-Committee A, which redrafted Article 18 of the Havana Charter (which corresponds to Article III of the GATT 1994), acknowledged that the text of Article III:8(b) was added "because it was felt that if subsidies were paid on domestic and not on imported products, it might be construed that Members were not applying the 'national treatment' rule". (E/CONF.2/c.3/A/W.49, p. 2)

⁴⁰⁷ Panel Reports, para. 7.79. We also agree with the Panel that "providing a competitive advantage in relation to market conditions is a typical intention behind subsidization". (Ibid., para. 7.78)

⁴⁰⁸ As discussed above, insofar as Article III:8(b) justifies the payment by WTO Members of subsidies exclusively to domestic producers, conditions for eligibility that define the class of eligible "domestic producers" by reference to their activities in the subsidized products' markets may also be justified under Article III:8(b).

⁴⁰⁹ Panel Reports, para. 7.88.

⁴¹⁰ Panel Reports, para. 7.83 (referring to Panel Report, *Indonesia – Autos*, paras. 14.43-14.46). (emphasis added)

⁴¹¹ Panel Reports, para. 7.79.

⁴¹² Panel Reports, para. 7.78.

- 51 -

producers that are not in receipt of the subsidy.⁴¹³ In other words, the Panel's interpretation, taken to its logical conclusion, denies effect to the exception contained in Article III:8(b), because, following the Panel's logic, in order to justify discrimination inconsistent with the national treatment obligations in Article III pursuant to Article III:8(b), the "payment of subsidies exclusively to domestic producers" must not be discriminatory in the first place. Seen in this light, the Panel's conclusion embodies a circular logic inasmuch as it delimits the scope of Article III:8(b) – an exception to the national treatment obligation for certain specific types of subsidies – on the basis of the discriminatory effects of the subsidies themselves.

5.111. To be clear, while we agree with the Panel that certain elements of a subsidy, such as a requirement conditioning access to the subsidy based on the use of domestic over imported goods, may violate the national treatment obligation under Article III in ways other than subsidization, this does not mean that the mere payment of a subsidy, in and of itself, is not justifiable under Article III:8(b).⁴¹⁴ In our view, although the Panel correctly noted that discrimination resulting from requirements to use domestic over imported goods, as prohibited under Article 3.1(b) of the SCM Agreement, is not justified under Article III:8(b), the Panel's unqualified reference to "aspects of a subsidy resulting in product discrimination" not being exempted under Article III:8(b) is overly broad and deprives that provision of any effect because, as acknowledged by the Panel, the very act of subsidization will, in and of itself, often result in product discrimination. Similarly, we find problematic the Panel's endorsement of the panel's conclusion in *Indonesia – Autos* that "Article III:8(b) confirms that subsidies to domestic producers do not violate Article III so long as they do not have *any component* that introduces discrimination between imported and domestic products."⁴¹⁵

5.112. Because of the aforementioned shortcomings in the Panel's reasoning, we <u>reverse</u> the Panel's overly broad and unqualified findings that "subsidies that are provided exclusively to domestic producers pursuant to Article III: 8(b) of the GATT 1994 are not *per se* exempted from the disciplines of Article III of the GATT 1994"⁴¹⁶ and that "aspects of a subsidy resulting in product discrimination (including requirements to use domestic goods, as prohibited by Article 3.1 of the SCM Agreement) are not exempted from the disciplines of Article III pursuant to Article III:8(b)."417 Instead, based on the proper interpretation of Article III:8(b) set out above, insofar as the payment of subsidies exclusively to domestic producers of a given product affects the conditions of competition between such a product and the like imported product, resulting in an inconsistency with the national treatment obligation in Article III, such a payment would be justified under Article III:8(b), provided that the conditions thereunder are met. Moreover, conditions for eligibility for the payment of subsidies that define the class of eligible "domestic producers" by reference to their activities in the subsidized products' markets would be justified under Article III:8(b). By contrast, a requirement to use domestic over imported goods in order to have access to the subsidy would not be covered by the exception in Article III:8(b) and would therefore continue to be subject to the national treatment obligation in Article III.

5.113. The consequences of the Panel's erroneous conclusion that "subsidies that are provided exclusively to domestic producers pursuant to Article III:8(b) are not *per se* exempted from the

 $^{^{413}}$ We recall that in US – Tax Incentives the Appellate Body observed that "[s]ubsidies that relate to domestic production ... can ordinarily be expected to increase the supply of the subsidized domestic goods in the relevant market..." (Appellate Body Report, US – Tax Incentives, para. 5.15)

⁴¹⁴ As discussed above, we consider that conditions for eligibility for a subsidy that relate to the producer's activities in the subsidized products markets are justified under Article III:8(b). In this regard, we note Brazil's argument that:

A logical consequence of the right to provide subsidies under WTO law is that Members must be free to define what constitutes a domestic producer for the purposes of participating in subsidies programmes. It is typical for regulations authorizing a subsidy to define the basic requirements and qualifying conditions for identifying domestic producers to ensure that only domestic producers – and not foreign producers – benefit from the subsidy. It would therefore be

inappropriate to consider that those requirements *per se* violate Article III:4 of the GATT 1994. (Brazil's appellant's submission, para. 136)

⁴¹⁵ Panel Reports, para. 7.83 (referring to Panel Report, *Indonesia – Autos*, paras. 14.43-14.46). (emphasis added)

⁴¹⁶ Panel Reports, para. 7.87.

⁴¹⁷ Panel Reports, para. 7.88.

- 52 -

disciplines of Article III" manifest themselves in its analysis under Article III: 4.418 Having reached the interpretative conclusion above, the Panel turned to determine "whether the product-related aspects of any subsidies that it may find to exist under the challenged measures are discriminatory in a manner inconsistent with Article III: 2, III: 4, and III: 5 of the GATT 1994".⁴¹⁹ With respect to the consistency of the conditions of accreditation under the ICT programmes with Article III: 4, the Panel found "that the conditions of accreditation, which when fulfilled create a lower internal tax burden on domestic products than on like imported products, modify the conditions of competition to the detriment of the imported products".⁴²⁰ Thus, the aspect of the ICT programmes found to be inconsistent is the accreditation requirements that result in less favourable treatment in the form of the differential tax treatment for imported ICT products. Crucially, this aspect of the subsidy that the Panel found to be inconsistent with Article III: 4 was the very aspect of the alleged subsidization sought to be justified by Brazil under Article III:8(b), but never examined by the Panel. In this manner, the Panel's interpretation of Article III:8(b) obviated the need for it to consider whether the "differential tax treatment" arising out of the ICT programmes constitutes the "payment of subsidies" and whether the conditions of accreditation are legitimate conditions for eligibility that are limited to ensuring that such payments are made "exclusively to domestic producers" within the meaning of Article III:8(b).

5.114. Having reversed the Panel's findings under Article III:8(b), we turn to Brazil's argument that the term "subsidies" in Article III:8(b) encompasses all types of subsidies listed in Article 1.1 of the SCM Agreement.⁴²¹ Because the exemption or remission of "indirect taxes" is a "subsidy" within the meaning of the SCM Agreement, for Brazil, it is also a subsidy for the purpose of Article III:8(b) of the GATT 1994.⁴²² Brazil alleges that the Panel's interpretation, therefore, fails to give effect to the definition of a "subsidy" contained in Article 1.1 of the SCM Agreement.⁴²³ Brazil argues that the Appellate Body's interpretation of the term "payment" in the context of Article III:8(b) in Canada -Periodicals is in "direct contradiction" with the Appellate Body's interpretation of the term "payment", in the context of Article 9.1(c) of the Agreement on Agriculture, in Canada - Dairy.⁴²⁴ Moreover, recalling that the Appellate Body in Canada - Periodicals stated that Article III:8(b) was "intended to exempt from the obligations of Article III only the payment of subsidies which involves the expenditure of revenue by a government"⁴²⁵, Brazil asserts that "when a government foregoes the collection of a known amount of tax revenue that it would otherwise collect, it expends revenue in that amount", and "these expenditures are commonly referred to as 'tax expenditures'".⁴²⁶ Finally, Brazil considers that "the benefit to a domestic producer of a subsidy in the form of a 'grant' is equivalent to the benefit of an exemption of indirect taxes in the corresponding ratio", such that a narrow interpretation of Article III:8(b) that limits its scope to certain types of subsidies only would be "void of any economic logic" and would "elevate form over substance". 427

5.115. The European Union submits that the Panel did not need to discuss the overlap between the meaning of the term "subsidy" in Article III:8(b) of the GATT 1994 and the definition provided in Article 1.1 of the SCM Agreement because "[t]he Panel excluded the application of Article III:8(b) on the basis that that provision does not carve out subsidies provided to domestic producers per se."428 Thus, the European Union submits that the Appellate Body need not address Brazil's arguments concerning the interpretation of the term "subsidy" in Article III:8(b) as this did not form a part of the Panel's reasoning.⁴²⁹ Nonetheless, referring to the Appellate Body Report in Canada - Periodicals, the European Union submits that a reduction or exemption of tax does not

⁴¹⁸ Panel Reports, para. 7.87. Our discussion of the Panel's analysis under Articles III: 2 and III: 4 of the GATT 1994 does not indicate an endorsement of the Panel's order of analysis with respect to these provisions and Article III:8(b).

⁴¹⁹ Panel Reports, para. 7.88.

⁴²⁰ Panel Reports, para. 7.225.

⁴²¹ Brazil's appellant's submission, heading B.1.

⁴²² Brazil's appellant's submission, para. 52.

⁴²³ Brazil's appellant's submission, para. 55.

⁴²⁴ Brazil's appellant's submission, para. 60 (referring to Appellate Body Report, Canada - Dairy, paras. 110 and 112).

⁴²⁵ Brazil's appellant's submission, para. 61 (quoting Appellate Body Report, *Canada – Periodicals*,

p. 34). ⁴²⁶ Brazil's appellant's submission, para. 61 (referring to OECD, *Tax Expenditures in OECD Countries* 21 // Alwaseft (and complaint), para, 813) (2010), p. 12; Appellate Body Report, *US – Large Civil Aircraft (2nd complaint)*, para. 813). ⁴²⁷ Brazil's appellant's submission, paras. 62-63.

⁴²⁸ European Union's appellee's submission, para. 66.

⁴²⁹ European Union's appellee's submission, para. 66.

- 53 -

amount to "the payment of subsidies" under Article III:8(b).⁴³⁰ The European Union also relies on the term "payment" used in Article III:8(b), as opposed to "granting" or "subsidies" used in Article XVI of the GATT 1994, to highlight the limited scope of Article III:8(b).⁴³¹ For these reasons, the European Union submits that Article III:8(b) "only carves out 'the payment of subsidies...' (and not any type of subsidies such as revenue foregone), to the extent that such payments do not contain aspects resulting in product discrimination, such as the requirement to use domestic over imported goods."⁴³²

5.116. Japan considers Brazil's argument concerning the term "payment of subsidies" in Article III:8(b) to be "entirely circular" inasmuch as it states that because revenue foregone is included in the definition of a subsidy under the SCM Agreement, it must therefore also fall within the scope of Article III:8(b).⁴³³ As to Brazil's reliance on the Appellate Body Report in *Canada – Dairy*, Japan points out that the Appellate Body's interpretation of the term "payment" in that dispute related to "very specific language" of a provision in a different covered agreement⁴³⁴ and ignores the Appellate Body's "very specific language" in *Canada – Periodicals*.⁴³⁵

5.117. We note the European Union's argument that the Panel did not need to discuss the overlap between the meaning of the term "subsidy" in Article III:8(b) and the definition provided in Article 1.1 of the SCM Agreement because "[t]he Panel excluded the application of Article III:8(b) on the basis that such provision does not carve out subsidies provided to domestic producers *per se.*"⁴³⁶ The European Union submits that the Appellate Body need not address Brazil's arguments concerning the interpretation of the term "payment of subsidies" in Article III:8(b) as it did not form part of the Panel's reasoning.⁴³⁷ We agree with the European Union that one reason why the Panel may not have felt compelled to interpret the terms "payment of subsidies exclusively to domestic producers" could have been its finding that such payments are, in any event, not *per se* exempt from the national treatment obligation under Article III. As discussed, however, this line of reasoning embodies a circular logic and denies effect to the terms of Article III:8(b). Having reversed that finding by the Panel, we consider it appropriate to address Brazil's claim on appeal relating to the interpretation of the term "payment of subsidies" in Article III:8(b).

5.118. Brazil argues that, "[b]ecause the exemption or remission of indirect taxes is a 'subsidy' within the meaning of the SCM Agreement, it is also a 'subsidy' for purposes of Article III:8(b) of the GATT 1994."⁴³⁸ As the European Union rightly points out, however, the issue at hand relates not to the definition of a "subsidy", in the abstract, but, instead, to the scope of the term "payment of subsidies" in Article III:8(b).⁴³⁹ Even if the exemption of indirect product taxes is a "subsidy" within the meaning of Article 1.1 of the SCM Agreement, this does not answer the question whether such exemption constitutes the "*payment* of subsidies" under Article III:8(b).⁴⁴⁰

5.119. As discussed above, while the conduct made permissible by Article III:8(b) is the "*payment* of subsidies", other provisions of the GATT 1994, including Article XVI, and the SCM Agreement,

para. 110).

⁴³⁰ European Union's appellee's submission, paras. 67-68 (referring to Appellate Body Report, *Canada – Periodicals*, p. 34).

⁴³¹ European Union's appellee's submission, para. 70.

⁴³² European Union's appellee's submission, para. 74.

⁴³³ Japan's appellee's submission, para. 43.

⁴³⁴ Japan's appellee's submission, para. 45 (referring to Appellate Body Report, Canada - Dairy,

⁴³⁵ Japan's appellee's submission, para. 46 (quoting Appellate Body Report, *Canada – Periodicals*, p. 34).

⁴³⁶ European Union's appellee's submission, para. 66.

⁴³⁷ European Union's appellee's submission, para. 66.

⁴³⁸ Brazil's appellant's submission, para. 52.

⁴³⁹ European Union's appellee's submission, para. 70.

⁴⁴⁰ Moreover, we recall our discussion above that the definition of a subsidy provided in Article 1.1 of the SCM Agreement is "for the purposes of this Agreement", i.e. the SCM Agreement, and does not apply to the GATT 1994.

- 54 -

dealing with subsidies, use the terms "grant" or "maintain" when referring to subsidies.⁴⁴¹ This difference in terminology suggests that the term "payment of subsidies" in Article III:8(b) encompasses a narrower range of conduct than that covered by the terms "subsidy" or the "granting" or "maintaining" of a subsidy elsewhere in the GATT 1994 and the other covered agreements.

5.120. We recall our discussion above of the first example of what constitutes the "payment of subsidy" under Article III:8(b), namely, "payments to domestic producers derived from proceeds of internal taxes or charges applied consistently with the provisions of [Article III]". If subsidies provided in the form of an exemption or reduction of taxes that are "otherwise due" are seen as being within the scope of Article III:8(b), then such discriminatory taxation effectuated through a scheme of foregoing revenue that reduces the tax burden on like domestic products or otherwise affects the conditions of competition to the detriment of the imported products would be justified under Article III:8(b). However, subsidies that are paid through the proceeds of discriminatory taxation would fall outside the scope of the exception in Article III:8(b) by virtue of the first example and would therefore continue to be subject to the national treatment obligation. We further note in this regard that the Appellate Body in Canada - Periodicals expressly agreed with the GATT panel in US - Malt Beverages that "[e]ven if the proceeds from non-discriminatory product taxes may be used for subsequent subsidies, the domestic producer, like his foreign competitors, must pay the product taxes due."442 Moreover, the prohibition of discriminatory internal taxes in Article III: 2 would also be rendered ineffective if discriminatory internal taxes on imported products could be justified as subsidies for competing domestic producers in terms of Article III: 8(b).⁴⁴³ Indeed, if the scope of "payment of subsidies" is seen as encompassing a reduction or exemption in internal product taxes that are "otherwise due", it would allow WTO Members to adopt discriminatory tax measures by disguising them in the form of a scheme of exemption or reduction of taxes for domestic producers alone.

5.121. We note Brazil's argument that, in *Canada – Dairy*, the Appellate Body interpreted the term "payment", in relation to subsidies, as including "revenue foregone". 444 That dispute addressed the issues of whether the provision of discounted milk to processors or exporters under a Canadian scheme constitutes an export subsidy within the meaning of Article 9.1(c) of the Agreement on Agriculture. The Appellate Body's interpretation of the term "payment" in Article 9.1(c) in that dispute was guided by the language of that provision and was specific to the context in which it appears. For instance, the Appellate Body in Canada - Dairy agreed with the panel that "none of the export subsidies listed in Article 9.1 is restricted to grants made solely in money form and several expressly involve subsidies granted in a form other than money."445 The Appellate Body also considered Article 1(c) of the Agreement on Agriculture as providing relevant context and explained that "[i]n terms of that provision [i.e. Article 1(c)], 'revenue foregone' is to be taken into account in determining whether 'budgetary outlay' commitments, made with respect to export subsidies as listed in Article 9.1, have been exceeded."446 For the Appellate Body in Canada - Dairy, "if a

⁴⁴¹ Article XVI:1 of the GATT 1994, for example, provides, in relevant part:

If any Member grants or maintains any subsidy, including any form of income or price support, which operates directly or indirectly to increase exports of any product from, or to reduce imports of any product into, its territory, it shall notify the WTO in writing of the extent and nature of the subsidization, of the estimated effect of the subsidization on the quantity of the affected product or products imported into or exported from its territory and of the circumstances making the subsidization necessary. In any case in which it is determined that serious prejudice to the interests of any other Member is caused or threatened by any such subsidization, the Member granting the subsidy shall, upon request, discuss with the other Member or Members, or with the WTO, the possibility of limiting the subsidization.

⁽emphasis added)

⁴⁴² Appellate Body Report, Canada - Periodicals, p. 34 (quoting GATT Panel Report, US - Malt Beverages, para. 5.10). The GATT panel further explained that "[t]he separation of tax and subsidy rules contributes to greater transparency. It also may render abuses of tax policies for protectionist purposes more difficult, as in the case where producer aids require additional legislative or governmental decisions in which the different interests involved can be balanced." (Ibid.)

⁴⁴³ See e.g. GATT Panel Report, US - Malt Beverages, para. 5.9. The GATT panel further noted that an "expansive interpretation of Article III:8(b) ..., if carried to its logical conclusion, ... would virtually eliminate the prohibition in Article III: 2 of discriminatory internal taxation by enabling contracting parties to exempt all domestic products from indirect taxes." (Ibid., para. 5.12)

⁴⁴⁴ Brazil's appellant's submission, para. 60 (quoting Appellate Body Report, *Canada – Dairy*, paras. 110

and 112). ⁴⁴⁵ Appellate Body Report, *Canada – Dairy*, para. 109 (referring to Panel Report, *Canada – Dairy*, para. 7.95). (emphasis original)

⁴⁴⁶ Appellate Body Report, Canada - Dairy, para. 110.

- 55 -

restrictive reading of the words 'payments' were adopted, such that 'payments' under Article 9.1(c) had to be monetary, no account could be taken, under Article 9.1(c), of 'revenue foregone'" and this would "prevent a proper assessment of the commitments made by WTO Members under Article 9.2, as envisaged by Article 1(c)".⁴⁴⁷ Given the Appellate Body's context-specific interpretation of the term "payment" in Article 9.1(c) of the Agreement on Agriculture, we consider its observations to be of limited relevance to the interpretation of the term "payment of subsidies" in Article III:8(b).

5.122. An examination of the text and context of Article III:8(b), in light of its object and purpose and as confirmed by the negotiating history, therefore suggests that the term "payment of subsidies" in Article III:8(b) does not include within its scope the exemption or reduction of internal taxes affecting the conditions of competition between like products. Instead, as noted by the Appellate Body in *Canada – Periodicals*, Article III:8(b) "was intended to exempt from the obligations of Article III only the payment of subsidies which involves the expenditure of revenue by a government".⁴⁴⁸

5.2.4 Conclusion with respect to Article III:8(b)

5.123. Although the Panel did, at an early stage of its analysis, acknowledge that the payment of subsidies exclusively to domestic producers (and the resulting market effects) does not by itself constitute discriminatory treatment with respect to imported products of the type prohibited by Article III⁴⁴⁹, the Panel's interpretation of Article III:8(b) and its application to the measures at issue obfuscate the distinction between the effects of the payment of a subsidy to a domestic producer on the conditions of competition in the relevant product market(s) and the conditions for eligibility attaching thereto, on the one hand, and any other effects arising from requirements to use domestic over imported inputs in the production process, on the other hand. Moreover, at no stage did the Panel undertake an assessment of whether the measures at issue constitute the "payment of subsidies exclusively to domestic producers" within the meaning of Article III:8(b). Because of these shortcomings in the Panel's reasoning, we reverse the Panel's overly broad and unqualified findings that "subsidies that are provided exclusively to domestic producers pursuant to Article III:8(b) of the GATT 1994 are not per se exempted from the disciplines of Article III of the GATT 1994"⁴⁵⁰ and that "aspects of a subsidy resulting in product discrimination (including requirements to use domestic goods, as prohibited by Article 3.1 of the SCM Agreement) are not exempted from the disciplines of Article III pursuant to Article III: 8(b)."451

5.124. Article III:8(b) carves out from the national treatment obligation in Article III the payment of subsidies exclusively to domestic producers. Insofar as the payment of subsidies exclusively to domestic producers of a given product affects the conditions of competition between such a product and the like imported product, resulting in an inconsistency with the national treatment obligation in Article III, such a payment would be justified under Article III:8(b), provided that the conditions thereunder are met. Moreover, conditions for eligibility for the payment of subsidies that define the class of eligible "domestic producers" by reference to their activities in the subsidized products' markets would also be justified under Article III:8(b). By contrast, a requirement to use domestic over imported goods in order to have access to the subsidy would not be covered by the exception in Article III:8(b) and would therefore continue to be subject to the national treatment obligation in Article III. Furthermore, an examination of the text and context of Article III:8(b), in light of its object and purpose and as confirmed by the negotiating history, suggests that the term "payment of subsidies" in Article III:8(b) does not include within its scope the exemption or reduction of internal taxes affecting the conditions of competition between like products. Instead, as noted by the Appellate Body in Canada - Periodicals, Article III:8(b) "was intended to exempt from the obligations of Article III only the payment of subsidies which involves the expenditure of revenue by a government".⁴⁵² Under a proper interpretation of Article III:8(b), none of the measures at issue in this dispute is capable of being justified under that provision because they all involve the exemption or reduction of internal taxes affecting the conditions of competition between like products and therefore cannot constitute the "payment of subsidies" within the meaning of Article III:8(b).

⁴⁴⁷ Appellate Body Report, *Canada – Dairy*, para. 110.

⁴⁴⁸ Appellate Body Report, *Canada - Periodicals*, p. 34.

⁴⁴⁹ Panel Reports, para. 7.79.

⁴⁵⁰ Panel Reports, para. 7.87.

⁴⁵¹ Panel Reports, para. 7.88.

⁴⁵² Appellate Body Report, *Canada – Periodicals*, p. 34.

5.2.5 Separate opinion of one Appellate Body Member on Article III:8(b) of the GATT 1994

5.125. While I agree with the majority's reversal of the Panel's "overly broad and unqualified" findings in paragraphs 7.87 and 7.88 of its Reports⁴⁵³, despite our best efforts to arrive at a consensus, I part company with my distinguished colleagues holding the majority view on the interpretation of the term "payment of subsidies" in Article III:8(b) of the GATT 1994 for the reasons set out below.

5.126. While neither Article III:8(b) nor any other provision of the GATT 1994 dealing with subsidies, including Article XVI, defines a "subsidy", a detailed definition of what constitutes a "subsidy" is found in Article 1.1 of the SCM Agreement. Although the *chapeau* of Article 1.1 states that the definition of a subsidy under that provision is "[f]or the purpose of this Agreement", i.e. the SCM Agreement, I note that Article 1.1 contains several textual references to the provision of the GATT 1994 dealing expressly with subsidies, namely Article XVI. In indicating that "a subsidy shall **be deemed to exist if: ... (a)(1)(ii) government revenue that is otherwise due is foregone or not** collected (e.g. fiscal incentives such as tax credits)", footnote 1 to Article 1.1(a)(1)(ii) clarifies that:

In accordance with the provisions of Article XVI of GATT 1994 (Note to Article XVI) and the provisions of Annexes I through III of this Agreement, the exemption of an exported product from duties or taxes borne by the like product when destined for domestic consumption, or the remission of such duties or taxes in amounts not in excess of those which have accrued, shall not be deemed to be a subsidy.⁴⁵⁴

Article 1.1(a)(2) further states that a "subsidy shall be deemed to exist if: ... (a)(2) there is any form of income or price support *in the sense of Article XVI of the GATT 1994*".⁴⁵⁵ There are thus explicit textual linkages between the definition of a subsidy set out in Article 1 of the SCM Agreement and Article XVI of the GATT 1994 dealing with subsidies.

5.127. Moreover, the main object of the SCM Agreement, taken as a whole, is "to increase and improve GATT disciplines relating to the use of both subsidies and countervailing measures".⁴⁵⁶ Indeed, I do not consider that, for the regulation of subsidies, the provisions of the GATT 1994 and the SCM Agreement operate in isolation; instead, the relevant provisions of the GATT 1994 and the SCM Agreement together define and reflect the *whole package of rights and obligations* of WTO Members with respect to subsidies.⁴⁵⁷ Besides the various textual linkages between the GATT 1994 and the SCM Agreement, I consider that the General Interpretative Note to Annex 1A to the WTO Agreement in Annex 1A, ... the other agreement shall prevail to the extent of the conflict", further supports the existence of a package of rights and obligations of WTO Members as reflected in the covered agreements.

5.128. For purposes of this package of rights and obligations, "[t]he SCM Agreement defines the *concept* of 'subsidy'."⁴⁵⁸ Given the textual linkages between the definition of a subsidy set out in Article 1.1 of the SCM Agreement and the provisions of the GATT 1994 dealing with subsidies, as well as the object and purpose of the SCM Agreement and its relationship with the GATT 1994, I see no reason why the term "subsidies" under the GATT 1994, including in Article III:8(b), should have a different meaning than the definition of a subsidy set out in Article 1.1 of the SCM Agreement.⁴⁵⁹

5.129. I now turn to consider the implication of the use of the term "payment" in Article III:8(b), as opposed to "granting" or "maintaining" as used in other provisions dealing with subsidies. As noted by the majority, the dictionary meanings of "payment" include "[a] sum of money

⁴⁵³ See paragraph 5.123 above.

⁴⁵⁴ Emphasis added.

⁴⁵⁵ Emphasis added.

⁴⁵⁶ Appellate Body Report, US - Carbon Steel, para. 73.

⁴⁵⁷ See Appellate Body Report, *Brazil – Desiccated Coconut*, pp. 12-13 and 17.

⁴⁵⁸ Appellate Body Report, US - Carbon Steel, para. 73. (emphasis added)

⁴⁵⁹ As discussed further below, ascribing different definitions to the term "subsidies" under the

GATT 1994 and the SCM Agreement may also alter the carefully constructed balance between the rights and obligations with respect to the provision of subsidies under the covered agreements. See paras. 5.136-5.137 above.

- 57 -

(or equivalent) paid or payable" or "the remuneration of a person with money or its equivalent".⁴⁶⁰ Importantly, the reference to "money" or its "equivalent" in these dictionary definitions suggests that the scope of "payment" is not limited to direct monetary transfers, but may also include other transfers having an "equivalent" value or effect. I recall in this regard that, having considered the dictionary meaning of "payment"⁴⁶¹, which is independent of the context within which the term is used, the Appellate Body in Canada - Dairy observed that "according to these meanings, a 'payment' could be made in a form, other than money, that confers value, such as by way of goods or services."462 I also recall that the Appellate Body in Canada – Periodicals stated that Article III:8(b) "was intended to exempt from the obligations of Article III only the payment of subsidies which involves the expenditure of revenue by a government". 463 While I agree that the "payment of subsidies" must involve the "expenditure of revenue by a government" in order for it to be justified under Article III:8(b), I do not consider that the Appellate Body's statement in Canada - Periodicals quoted above establishes that the only way through which a government can expend revenue is by means of a direct monetary transfer. Instead, I consider that this was clarified by the Appellate Body in its subsequent report in Canada - Dairy, where it explained that a "payment made in the form of goods or services is also 'financed' in the same way as a money payment, and, likewise, 'a charge on the public account' may arise as a result of a payment, or a legally binding commitment to make payment by way of goods or services, or as a result of revenue foregone."464 Indeed, inasmuch as a charge on the public account is also an expenditure of revenue by the government, I consider that "payments" involving the expenditure of revenue by a government may take various forms and are not necessarily limited to direct monetary transfers. As the Appellate Body explained in Canada - Dairy, the foregoing of revenue, such as through a reduction, exemption, or suspension of taxes "otherwise due", incurs a charge on the public account and therefore involves the expenditure of revenue by a government.

5.130. I now address the contextual relevance of the first example of what constitutes the "payment of subsidies" set out under Article III:8(b), namely "payments to domestic producers derived from the proceeds of internal taxes or charges applied consistently with the provisions of [Article III] ...". I note that the text of the first example serves as the lynchpin of both the Panel's analysis as well as my fellow Division Members' interpretation of Article III:8(b). Essentially, they reason that "if ... Article III:8(b) exempts tax discrimination from the scope of Article III, the reference in Article III:8(b) itself to 'taxes and charges applied consistently with the provisions of [Article III]' would be meaningless."⁴⁶⁵ I note, however, that the reference in Article III: 8(b) to "taxes or charges applied consistently with the provisions of [Article III]" is not an independent or express requirement for all measures falling within the scope of Article III:8(b). Instead, this reference is made in the context of the first of the two examples of the "payment of subsidies" contained in that provision, which reads, in relevant part: "the payment of subsidies exclusively to domestic producers, including payments to domestic producers derived from the proceeds of internal taxes or charges applied consistently with the provisions of [Article III] and subsidies effected through governmental purchases of domestic products".⁴⁶⁶ As my distinguished colleagues forming the majority note, the use of the word "including" makes clear that these examples are "not an exhaustive list" of the kind of programmes that would qualify as "payment of subsidies exclusively to domestic producers". 467 The treaty text relied on by the Panel, namely, "taxes or charges applied consistently with the provisions of [Article III]", is, by the terms of that provision, a requirement only in the case of "payments to domestic producers *derived from the proceeds of internal taxes or charges*".⁴⁶⁸ Given

http://www.oed.com/view/Entry/139189?rskey=9c29D2&result=1#eid. (emphasis added)

⁴⁶² Appellate Body Report, *Canada - Dairy*, para. 107.

⁴⁶⁴ Appellate Body Report, *Canada – Dairy*, para. 108. (emphasis added)

⁴⁶⁰ Oxford English Dictionary online, definition of "payment",

⁴⁶¹ We note that the Appellate Body in *Canada – Dairy* referred to definitions of "payment" that are similar to the ones used by us: "[T]he *Oxford English Dictionary* ... **defines** 'payment' as 'the remuneration of a person with money *or its equivalent*'. ... Similarly, the *Shorter Oxford English Dictionary* describes a 'payment' as a 'sum of money (*or other thing*) paid'." (Appellate Body Report, *Canada – Dairy*, para. 107 (quoting Panel Report, *Canada – Dairy*, para. 7.92, in turn quoting *The Shorter Oxford English Dictionary*, C.T. Onions (ed.) (Guild Publishing, 1983), Vol. II, p. 1532) (emphasis added by the Appellate Body; fns omitted))

⁴⁶³ Appellate Body Report, *Canada – Periodicals*, p. 34. Although the Appellate Body in *Canada – Periodicals* relied on the GATT panel report in *US – Malt Beverages*, I consider that report to be of limited relevance as it was issued before the coming into force of the SCM Agreement, in general, and the definition of a subsidy set out in Article 1.1 of the SCM Agreement, in particular.

⁴⁶⁵ Panel Reports, para. 7.85.

⁴⁶⁶ Emphasis added.

⁴⁶⁷ Appellate Body Report, *Canada – Periodicals*, pp. 33-34.

⁴⁶⁸ Emphasis added.

- 58 -

the non-exhaustive nature of the list, I do not consider that this text can be divorced from the context of one specific example in Article III:8(b) and be regarded as an independent and stand-alone requirement that definitively delimits the scope of Article III:8(b) and applies to all instances of "payment of subsidies" that could possibly fall within the scope of that provision. In particular, subsidies provided through the foregoing of government revenue that is otherwise due, such as tax exemptions, are, by definition, not "derived from the proceeds of internal taxes or charges", thereby rendering the first example of limited relevance in such cases.

5.131. Instead of regarding the textual reference in the first example set out in Article III:8(b) as controlling under all circumstances, a proper understanding of the scope of Article III:8(b) ought to be grounded in the interpretation of the term "payment of subsidies" in accordance with the customary international rules of treaty interpretation.

5.132. As for the contextual relevance of Article III: 2 of the GATT 1994, I am not entirely convinced that interpreting "payment of subsidies" as including "revenue foregone" would necessarily render the prohibition against tax discrimination in Article III:2 meaningless. In this regard, I note that Article 1.1(a)(1)(ii) of the SCM Agreement provides that there is a "financial contribution" by a government, sufficient to fulfil that element in the definition of a subsidy, where "government revenue that is otherwise due is foregone or not collected (e.g. fiscal incentives such as tax credits)."469 The Appellate Body has explained that:

[T]he "foregoing" of revenue "otherwise due" implies that less revenue has been raised by the government than would have been raised in a different situation, or, that is, "otherwise". Moreover, the word "foregone" suggests that the government has given up an entitlement to raise revenue that it could "otherwise" have raised. This cannot, however, be an entitlement in the abstract, because governments, in theory, could tax all revenues.470

According to the Appellate Body, there must therefore be "some defined, normative benchmark against which a comparison can be made between the revenue actually raised and the revenue that would have been raised 'otherwise'". 471 In US - FSC, the Appellate Body agreed with the panel that "the basis of comparison must be the tax rules applied by the Member in question."⁴⁷²

5.133. The Appellate Body's interpretation of Article 1.1(a)(1)(ii) of the SCM Agreement therefore suggests that the notion of financial contribution in the form of "revenue foregone" that is "otherwise due" is narrower than just any form of discriminatory taxation which may be prohibited under Article III: 2, in the abstract. In particular, revenue foregone must be "otherwise due" under the tax rules applied by the Member in question. I consider that a scheme applying discriminatory taxes to imported products and like domestic products would not constitute "revenue foregone" in as much as the lesser tax burden on domestic products would not, in and of itself, suggest that the government would have given up an entitlement to raise revenue that it could "otherwise" have raised as there would be no such entitlement under the tax rules in the first place. Seen in this light, "revenue foregone" is a narrower concept than tax discrimination in general, such that not all tax discrimination would ipso facto amount to "revenue foregone" and would thereby not be justified under Article III:8(b).

5.134. Therefore, far from rendering the prohibition against discriminatory taxation in Article III:2 redundant, an interpretation of the term "payment of subsidies" as including revenue foregone gives meaning and effect to the terms of both Articles III:2 and III:8(b), while at the same time respecting the carefully negotiated balance of rights and obligations under the SCM Agreement. As the majority correctly notes, the fact that a measure qualifies as the "payment of subsidies exclusively to domestic producers" and the resulting inconsistency with the national treatment obligation under Article III is therefore justified under Article III:8(b) does not suggest that such a measure would not be subject to the other relevant provisions of the GATT 1994 and the SCM Agreement. This was confirmed by the Appellate Body in US - Tax Incentives, where it explained that "even if the granting of a subsidy is exempt from the GATT national treatment obligation by virtue of it being paid exclusively to domestic producers within the meaning of Article III:8(b) of the GATT 1994, it may

⁴⁶⁹ Emphasis added; fn omitted.

⁴⁷⁰ Appellate Body Report, US - FSC, para. 90. (emphasis original) ⁴⁷¹ Appellate Body Report, US - FSC, para. 90.

⁴⁷² Appellate Body Report, US - FSC, para. 90.

- 59 -

still be found to be contingent upon the use by those producers of domestic over imported goods under Article 3.1(b) of the SCM Agreement."⁴⁷³ Thus, in my view, the payment of subsidies exclusively to domestic producers through a reduction or exemption of internal taxes, though covered by the exception under Article III:8(b), would continue to be subject to the other relevant provisions of the GATT 1994 and the SCM Agreement.

5.135. The GATT 1994 and the SCM Agreement are both a part of a single treaty, namely, the WTO Agreement. The Appellate Body in *Korea – Dairy* explained that:

In light of the interpretive principle of effectiveness, it is the duty of any treaty interpreter to "read all applicable provisions of a treaty in a way that gives meaning to all of them, harmoniously". An important corollary of this principle is that a treaty should be interpreted as a whole, and, in particular, its sections and parts should be read as a whole. Article II: 2 of the WTO Agreement expressly manifests the intention of the Uruguay Round negotiators that the provisions of the WTO Agreement and the Multilateral Trade Agreements included in its Annexes 1, 2 and 3 must be read as a whole.⁴⁷⁴

To me, the restrictive interpretation of "payment of subsidies" as excluding "revenue foregone" arrived at by the majority denies effect to the key legal terms of the SCM Agreement.

5.136. As discussed, with the establishment of the WTO, the relevant provisions of the GATT 1994 and the SCM Agreement together define and reflect the whole package of rights and obligations of WTO Members with respect to subsidies.⁴⁷⁵ The object and purpose of the SCM Agreement includes strengthening and improving the GATT disciplines relating to the use of both subsidies and countervailing measures, while, recognizing at the same time, the right of Members to impose such measures under certain conditions.⁴⁷⁶ The SCM Agreement does not prohibit all subsidies, but instead contains detailed and intricate rules disciplining their use.⁴⁷⁷ While some subsidies are "prohibited" under Part II of the Agreement, Part III makes it clear that others are only "actionable" and must be withdrawn (or their "adverse effects" removed), if they are "specific", within the meaning of Article 2 of the SCM Agreement, and when they cause "adverse effects"⁴⁷⁸ to the interests of other Members.⁴⁷⁹ In contrast to the detailed provisions of the SCM Agreement that, in disciplining the use of "actionable" subsidies, including revenue foregone, take into account the trade effects of the alleged subsidization, Article III: 2 of the GATT 1994 sets out the national treatment obligation with respect to internal taxes and charges. Under Article III:2 any tax, however small, that is borne by products imported into the territory of a WTO Member, but not by the domestic like products would, without more, lead to a breach of that obligation.⁴⁸⁰ Importantly, it is well established that the actual effects of such differential tax burden on imports are irrelevant to the analysis under Article III: 2 of the GATT 1994.⁴⁸¹ The focus on the trade effects of subsidization, including through revenue foregone, which pervades the SCM Agreement disciplines concerning the use of "actionable" subsidies is therefore absent in the context of Article III: 2 of the GATT 1994.

5.137. An interpretation of "payment of subsidies" in Article III:8(b) as excluding revenue foregone would undermine, inconsistently with Article 3.2 of the DSU as well as the fundamental principle of

⁴⁷³ Appellate Body Report, US - Tax Incentives, para. 5.16. (emphasis added)

⁴⁷⁴ Appellate Body Report, Korea - Dairy, para. 81. See also Appellate Body Report,

Argentina - Footwear (EC), para. 81.

⁴⁷⁵ See Appellate Body Report, *Brazil – Desiccated Coconut*, p. 17.

⁴⁷⁶ Appellate Body Report, *US* – *Softwood Lumber IV*, para. 64. The Appellate Body went on to note that "[i]t is in furtherance of this object and purpose that Article 1.1(a)(1)(iii) recognizes that subsidies may be conferred, not only through monetary transfers, but also by the provision of non-monetary inputs." (Ibid.)

⁴⁷⁷ In *Canada – Aircraft (Article 21.5 – Brazil)*, the Appellate Body found it "worth recalling that the granting of a subsidy is not, in and of itself, prohibited under the SCM Agreement. Nor does granting a 'subsidy', without more, constitute an inconsistency with that Agreement. The universe of subsidies is vast. Not all subsidies are inconsistent with the SCM Agreement. The only 'prohibited' subsidies are those identified in Article 3 of the SCM Agreement." (Appellate Body Report, *Canada – Aircraft (Article 21.5 – Brazil)*, para. 47)

⁴⁷⁸ Article 5 of the SCM Agreement defines "adverse effects" as including "injury" to the domestic industry of another WTO Member, "nullification or impairment" of benefits accruing to other Members, and "serious prejudice" to the interests of another Member.

⁴⁷⁹ See e.g. Panel Report, US - Offset Act (Byrd Amendment), para. 7.106.

⁴⁸⁰ Appellate Body Report, *Japan – Alcoholic Beverages II*, p. 23, DSR 1996:1, p. 97; see also Appellate Body Report, *Thailand – Cigarettes (Philippines)*, para. 112.

⁴⁸¹ Appellate Body Report, Japan – Alcoholic Beverages II, pp. 16-17, DSR 1996: I, pp. 109-110.

- 60 -

effectiveness in treaty interpretation, the careful balance of rights and obligations under the SCM Agreement with respect to an entire category of measures that are expressly included within the definition of a subsidy in Article 1.1, namely, the foregoing of government revenue that is otherwise due. In other words, the majority's interpretation of the term "payment of subsidies" in Article III:8(b) would fundamentally alter the carefully constructed balance of rights and obligations under the SCM Agreement and the GATT 1994 with respect to subsidies and would risk rendering redundant the actionable subsidies disciplines of the SCM Agreement insofar as subsidies in the form of the foregoing of revenue are concerned.

5.138. For the above reasons, I am of the view that the term "payment of subsidies" in Article III:8(b) refers to the provision by a WTO Member, whether through monetary or non-monetary transfers having an equivalent effect, of a subsidy, as defined in Article 1.1 of the SCM Agreement. In my view, this is the only interpretation that, consistently with the customary rules of treaty interpretation, gives meaning and effect to the precise terms of Article III:8(b), while at the same time respecting the carefully negotiated balance of rights and obligations under the SCM Agreement, which forms part of the single package under the WTO Agreement. Insofar as they constitute the "payment of subsidies exclusively to domestic producers"⁴⁸², the measures at issue in this dispute, as well as any conditions for eligibility for the payment of subsidies that define the class of eligible "domestic producers" by reference to their activities in the subsidized products' markets, would, in my view, be justified under Article III:8(b).

5.3 Article 3.1(a) of the SCM Agreement

5.3.1 Article 1.1(a)(1)(ii) of the SCM Agreement: PEC and RECAP programmes

5.3.1.1 Introduction

5.139. We now turn to consider Brazil's claim on appeal that the Panel erred in finding that the tax suspensions granted under the PEC programme and the RECAP programme are subsidies within the meaning of Article 1.1(a)(1)(ii) of the SCM Agreement and are contingent upon export performance under Article 3.1(a) of the SCM Agreement.⁴⁸³

5.140. Brazil contends that the Panel erred in the application of Article 1.1(a)(1)(ii) of the SCM Agreement by failing to use the tax treatment of companies that are structural tax accumulators as the benchmark treatment.⁴⁸⁴ Brazil argues, in particular, that, "[i]nstead of examining the principles and structure of Brazil's taxation regime and identifying what constitutes 'comparably situated taxpayers', the Panel erroneously focused on identifying a 'general rule of taxation' and its potential exemptions." 485 In the event that we were to conclude that the Panel did not err in its identification of the relevant benchmark treatment, Brazil submits three alternative claims of error. First, Brazil claims that the Panel erred in its comparison of the selected benchmark treatment with the challenged treatment under the PEC and RECAP programmes by "arbitrarily distinguish[ing] between taxpayers situated within the benchmark".⁴⁸⁶ Second, Brazil submits that the Panel erred in concluding that the possible cash availability and implicit interest income that the Brazilian Government could generate, had the tax been paid in advance and not offset, constitute "government revenue that is otherwise due" under Article 1.1(a)(1)(ii) of the SCM Agreement.487 Finally, Brazil contends that the Panel failed to make an objective assessment of the matter before it pursuant to Article 11 of the DSU by "misreading" and disregarding data in a table submitted by Brazil.488

⁴⁸² As correctly noted by the majority, in addition to the scope of "payment of subsidies", the focus of inquiry under Article III:8(b) is also on whether the domestic entity at issue is a *producer* of the product with respect to which a violation of the national treatment obligation arising from the "payment of subsidies" is alleged.

⁴⁸³ Brazil's Notice of Appeal, para. 26; Brazil's appellant's submission, para. 307.

⁴⁸⁴ Brazil's appellant's submission, paras. 309 and 324.

⁴⁸⁵ Brazil's appellant's submission, para. 316.

⁴⁸⁶ Brazil's appellant's submission, para. 327. (emphasis omitted)

⁴⁸⁷ Brazil's appellant's submission, para. 333

⁴⁸⁸ Brazil's appellant's submission, paras. 345, 351, and 360. Specifically, Brazil argues that, "[i]n direct contradiction to Brazil's explanation that the study was elaborated with data 'from all companies', the Panel found that the study 'appears to reflect the situation of 22 companies'." (Ibid., para. 351)

- 61 -

5.141. The European Union and Japan request that we reject Brazil's claims.⁴⁸⁹ With respect to the Panel's identification of a benchmark for comparison, the European Union and Japan argue that the Panel did not try to identify a "general rule"⁴⁹⁰, as suggested by Brazil, but instead was examining "the structure of [Brazil's] domestic tax regime and its organising principles".⁴⁹¹ With respect to Brazil's alternative claims, the European Union and Japan first submit that the Panel did not disregard the treatment of a particular group of taxpayers (i.e. those that normally offset tax credits during the same taxation period) within the benchmark when making a comparison.⁴⁹² Instead, the Panel weighed "all the scenarios to conclude that there was revenue foregone or not collected in the 'rather frequent scenario' where taxpayers cannot offset their tax credits within the same taxation period".⁴⁹³ Second, the European Union and Japan contend that the Panel did not make an error in considering that the cash availability and implicit interest income that the Brazilian Government foregoes or does not collect by virtue of tax suspensions under the PEC and RECAP programmes amount to a financial contribution under Article 1.1(a)(1)(ii) of the SCM Agreement.⁴⁹⁴ Finally, the European Union and Japan assert that Brazil has failed to show that the Panel made an error under Article 11 of the DSU in assessing the table submitted by Brazil.⁴⁹⁵

5.142. We begin with an overview of the relevant aspects of the PEC and RECAP programmes. We also include a summary of the relevant Panel findings and our understanding of Article 1.1(a)(1)(ii) of the SCM Agreement.

5.3.1.2 PEC and RECAP programmes

5.143. With respect to the programmes at issue, the Panel explained that, under the PEC programme, the IPI tax and the PIS/PASEP, COFINS, PIS/PASEP-Importation, and COFINS-Importation contributions are suspended with respect to raw materials, intermediate goods, and packaging materials purchased by predominantly exporting companies.⁴⁹⁶ The tax suspension applies either at the time when the product leaves the industrial establishment (for purchases in the domestic market) or at the time of the customs clearance (for imports).⁴⁹⁷ The suspension expires and the tax becomes definitively non-due upon exportation or sale in the domestic market of the final good incorporating the raw materials, intermediate goods, and packaging materials for which the taxes were suspended.⁴⁹⁸

5.144. For purposes of the IPI tax suspension under the PEC programme, a predominantly exporting company is a legal person whose gross revenue derived from exports to other countries during the calendar year immediately prior to the year of purchase of the covered products, exceeded 50% of its total gross revenue from the sales of goods and services over the same period, after taxes and other contributions levied on sales.⁴⁹⁹

5.145. For purposes of the suspensions of the PIS/PASEP, COFINS, PIS/PASEP-Importation, and COFINS-Importation contributions under the PEC programme, a predominantly exporting company

⁴⁸⁹ European Union's appellee's submission, para. 369; Japan's appellee's submission, para. 126.

⁴⁹⁰ European Union's appellee's submission, para. 398; Japan's appellee's submission, para. 133.

⁴⁹¹ European Union's appellee's submission, para. 398 (quoting Appellate Body Report, *US – Large Civil Aircraft (2nd complaint)*, para. 815); Japan's appellee's submission, para. 133.

⁴⁹² European Union's appellee's submission, para. 409; Japan's appellee's submission, para. 135.

⁴⁹³ European Union's appellee's submission, para. 412.

⁴⁹⁴ European Union's appellee's submission, para. 428; Japan's appellee's submission, para. 137.

⁴⁹⁵ European Union's appellee's submission, para. 439; Japan's appellee's submission, para. 138. ⁴⁹⁶ Panel Reports, paras. 2.148 and 2.150 (referring to Law 10,637/2002 (Panel Exhibits JE-94 and

BRA-100), Article 29, section 1.11 (for IPI tax); Law 10,865/2004 (Panel Exhibit JE-181), Article 40 (for PIS/PASEP and COFINS contributions)).

⁴⁹⁷ Panel Reports, para. 2.151 (referring to Normative Instruction (*Instrução Normativa*) RFB 948 of 15 June 2009 (Normative Instruction RFB 948/2009) (Panel Exhibit JE-191), Articles 12 and 13; Normative Instruction SRF 595 of 27 December 2005 (Normative Instruction SRF 595/2005) (Panel Exhibit JE-193), Article 8; and Decree 6,759 of 5 February 2009 governing the administration of customs activities and the inspection, control and taxation of foreign trade operations (Decree 6,759/2009) (Panel Exhibit JE-8), Article 247).

⁴⁹⁸ Panel Reports, para. 2.152 (referring to Law 10,637/2002 (Panel Exhibits JE-94 and BRA-100), Article 29, section I.II; Law 10,865/2004 (Panel Exhibit JE-181), Article 40; Normative Instruction RFB 948/2009 (Panel Exhibit JE-191), Article 20; Normative Instruction SRF 595/2005 (Panel Exhibit JE-193), Article 9).

Article 9). ⁴⁹⁹ Panel Reports, paras. 2.156 and 7.1136 (referring to Law 10,637/2002 (Panel Exhibits JE-94 and BRA-100), Article 29, section 3).

- 62 -

is a legal person whose gross revenue from exports, in the calendar year immediately prior to the year of purchase of the covered products, was equal to or greater than 50% of its total gross revenue from the sale of goods and services in the same period, after taxes and other contributions levied on sales.⁵⁰⁰

5.146. Under the RECAP programme, the PIS/PASEP, COFINS, PIS/PASEP-Importation, and COFINS-Importation contributions are suspended with respect to purchases of new machinery, tools, apparatus, instruments, and equipment by predominantly exporting companies.⁵⁰¹ As the Panel explained, "[t]he suspension becomes a zero rate once the export commitments have been achieved."⁵⁰²

5.147. For purposes of the RECAP programme, a predominantly exporting company is a legal person whose gross revenue from exports, in the calendar year immediately before the year in which it became a member of the programme, was equal to or greater than 50% of its total gross revenue from the sale of goods and services over the same period, and who commits to maintaining that percentage of exports for a period of two calendar years.⁵⁰³

5.3.1.3 Whether the Panel erred in the application of Article 1.1(a)(1)(ii) of the SCM Agreement in determining the benchmark treatment

5.148. Brazil claims that the Panel erred in finding that the tax suspensions granted under the PEC and RECAP programmes are subsidies within the meaning of Article 1.1(a)(1)(ii) of the SCM Agreement. As a first ground of its appeal, Brazil considers that the Panel erred in its determination of the benchmark for comparison. In the event that we uphold the Panel's benchmark determination, Brazil submits three alternative claims. As noted above, Brazil argues that the Panel: (i) erred in its comparison of the selected benchmark treatment with the challenged treatment under the PEC and RECAP programmes; (ii) erred in finding that the possible cash availability and implicit interest income constitute "government revenue that is otherwise due"; and (iii) failed to make an objective assessment of the matter before it pursuant to Article 11 of the DSU by disregarding certain evidence submitted by Brazil.⁵⁰⁴

5.149. We recall that, in its analysis, the Panel first examined whether the tax suspensions granted under the PEC and RECAP programmes constituted financial contributions in the form of government revenue that is otherwise due, which is foregone or not collected.

5.150. The Panel noted that the challenged tax treatment comprised the following categories: (i) the IPI tax suspensions on purchases of raw materials, intermediate goods, and packaging materials (for purposes of the PEC programme); (ii) the suspensions of the PIS/PASEP and COFINS contributions, or PIS/PASEP-Importation and COFINS-Importation contributions, on purchases of raw materials, intermediate goods, and packaging materials (for purposes of the PEC programme); and (iii) the suspensions of the PIS/PASEP and COFINS contributions, or PIS/PASEP-Importation and COFINS-Importations, or PIS/PASEP-Importation and COFINS-Importations, or PIS/PASEP-Importation and COFINS-Importation contributions, on purchases of new machinery, apparatus, instruments, and equipment incorporated into the fixed assets of the accredited company (for purposes of the RECAP programme).⁵⁰⁵ The Panel found that there was "evidence demonstrating that the objective reasons

⁵⁰⁰ Panel Reports, paras. 2.157 and 7.1137 (referring to Law 10,865/2004 (Panel Exhibit JE-181), Article 40, section 1). In addition, companies that are just starting activities in Brazil, or that have not achieved the export percentage required in the previous year, may be entitled to the PIS/PASEP and COFINS contribution suspensions if they commit to reach and maintain the required export level for a period of three calendar years. (Ibid., paras. 2.158 and 7.1137 (referring to Normative Instruction SRF 595/2005 (Panel Exhibit JE-193), Article 3, section 1))

⁵⁰¹ Panel Reports, paras. 2.163 and 7.1138.

⁵⁰² Panel Reports, para. 7.1138.

⁵⁰³ Panel Reports, paras. 2.170 and 7.1139. Companies that are starting activities in Brazil, or that have not reached the export percentage required in the previous year, may be entitled to the suspensions if they commit to reach and maintain the required export level for a period of three calendar years. (Ibid., para. 2.171 (referring to Law 11,196 of 21 November 2005 establishing the special scheme for the acquisition of capital goods for exporting companies (RECAP) (Law 11,196/2005) (Panel Exhibit JE-182), Article 13, section 2)).

⁵⁰⁴ Brazil's appellant's submission, paras. 326-327, 333, 345, 351, and 361.

⁵⁰⁵ Panel Reports, para. 7.1156.

behind the tax treatment [are] to tackle the problem of credit-accumulation, and in so doing to increase the competitiveness of Brazilian companies".⁵⁰⁶

5.151. In its identification of the benchmark treatment for each of the three categories of the identified tax treatments, the Panel rejected Brazil's argument that tax suspensions are the benchmark treatment for structurally credit-accumulating companies, including the predominantly exporting companies.⁵⁰⁷ The Panel agreed with Brazil that there are other companies, in addition to predominantly exporting companies, that are entitled to the suspension of relevant taxes.⁵⁰⁸ The Panel considered, however, that nothing in the evidence on the record seemed to suggest that "other companies to which the tax suspensions apply are structural credit accumulators, or that the tax suspensions ... were created to tackle the problem of structural credit accumulation for these companies."⁵⁰⁹

5.152. The Panel further noted that there are producers of low-taxed products, which are more likely to accumulate tax credits, that are not entitled to tax suspensions, on the one hand, and producers of higher taxed products, which are less likely to accumulate tax credits, that *are* entitled to such tax suspensions, on the other hand.⁵¹⁰ The Panel found, with respect to the three categories of tax treatment, that "the benchmarks to be applied are the economy-wide tax treatments from which the suspensions are taken or, in other words, the tax treatment applicable to non-accredited companies."⁵¹¹

5.153. The Panel then compared the challenged treatment with the "rather frequent" scenario, in which companies are not able to offset their credits and have to request compensation or reimbursement.⁵¹² The Panel considered that, "under the challenged treatment, the Government will never be able to benefit from the cash availability and implicit interest income, which could have lasted from the moment of importation (in the case of imported products) or from one taxation period (in the case of domestic products) to 360 days after the request for compensation or reimbursement, under the benchmark treatment."⁵¹³ The Panel thus concluded, with respect to the three categories of tax treatment at issue, that, "under this rather frequent scenario, the Government is foregoing revenue in the form of the implicit interest on the tax revenue collected where the offsetting credits have not (yet) been used."⁵¹⁴ The Panel also concluded that the subsidies at issue confer a benefit and that they thus constitute subsidies within the meaning of Article 1 of the SCM Agreement.⁵¹⁵ Subsequently, the Panel found that the subsidies granted under the PEC and RECAP programmes are contingent upon export performance within the meaning of Article 3.1(a) of the SCM Agreement, and thus prohibited under Articles 3.1(a) and 3.2 of the SCM Agreement.⁵¹⁶

5.154. We start by addressing Brazil's claim that the Panel erred in determining the benchmark for comparison. In Brazil's view, the Panel erroneously rejected the tax treatment of structural credit accumulators as a benchmark for comparison and instead concluded that the proper benchmark was the treatment applicable to purchases by non-accredited companies, i.e. the obligation to pay the full amount of the applicable tax.⁵¹⁷ Brazil submits that "[t]he Panel erroneously sought to ascertain whether a 'general rule' of taxation existed under Brazilian law pursuant to which all suspensions

⁵¹² Panel Reports, paras. 7.1174, 7.1189, and 7.1202. The Panel explained that "[i]t is fairly common that companies are not able to offset the full amount of their credits and have to request compensation or reimbursement, so this scenario seems rather frequent." (Ibid. (referring to Brazil's first written submission, para. 758 (DS472) and para. 691 (DS497))). In this case, "the Brazilian Government will retain the amount of tax paid by the seller until the non-accredited company buying the products is able to offset it (during subsequent taxation periods); or until the Brazilian Government compensates or reimburses the face amount of the credit, without interest, to the non-accredited company buying the products (within 360 days of the date of the request for compensation or reimbursement)." (Panel Reports, paras. 7.1175, 7.1190, and 7.1203) ⁵¹³ Panel Reports, paras. 7.1179, 7.1194, and 7.1207. (fns omitted)

⁵¹⁴ Panel Reports, paras. 7.1179, 7.1194, and 7.1207. (Insom

⁵¹⁵ Panel Reports, paras. 7.1212 and 7.1223.

⁵¹⁶ Panel Reports, para. 7.1237.

⁵⁰⁶ Panel Reports, para. 7.1162.

⁵⁰⁷ Panel Reports, paras. 7.1171, 7.1186, and 7.1199.

⁵⁰⁸ Panel Reports, para. 7.1167 (referring to Law 10,637/2002 (Panel Exhibit JE-94) and to Brazil's response to Panel question No. 21). See also Panel Reports, para. 7.1183 (referring to Brazil's response to Panel question No. 21).

⁵⁰⁹ Panel Reports, para. 7.1167.

⁵¹⁰ Panel Reports, paras. 7.1168, 7.1184, and 7.1199.

⁵¹¹ Panel Reports, para. 7.1164. See also paras. 7.1172, 7.1187, and 7.1200.

⁵¹⁷ Brazil's appellant's submission, paras. 307 and 309.

- 64 -

and exemptions of taxes on inputs and capital goods relate to the problem of structural credit accumulation"⁵¹⁸, despite the Appellate Body's reservations, expressed in *US* – *FSC* (*Article 21.5* – *EC*) and *US* – *Large Civil Aircraft* (2nd complaint), about panels trying to identify such a "general rule of taxation".⁵¹⁹ Brazil recalls that, "in identifying the relevant benchmark treatment, a panel is required to 'develop an understanding of the tax structure and principles that best explains that Member's tax regime, and to provide a reasoned basis for identifying what constitutes comparable income of comparably situated taxpayers."⁵²⁰

5.155. Brazil also maintains that, in order to avoid structural accumulation of indirect taxes, it created "an alternative taxation method for the income of structural credit accumulators", which applies "to companies of many different sectors of the economy which have no or low taxation on their sales".⁵²¹ In particular, Brazil points out that Article 29 of the Law 10,637/2002 "exempts from the payment of taxes on inputs a wide variety of sectors of the Brazilian economy whose products are subject to low or no taxation, including producers of animal and vegetable products and oils, foodstuffs, chemicals, footwear, auto and aircraft parts, informatics products, and predominantly exporting companies".⁵²² In Brazil's view, the appropriate benchmark treatment for comparison is the tax treatment afforded to other structural credit accumulators.⁵²³

5.156. The European Union and Japan contend that the Panel properly applied the test developed by the Appellate Body in US – Large Civil Aircraft (2nd complaint) to determine the benchmark treatment.⁵²⁴ They also submit that, contrary to Brazil's allegations, the Panel did not attempt to identify a "general rule" of taxation pursuant to which all suspensions and exemptions of taxes on inputs and capital goods would relate to a problem of structural credit accumulation, but rather was addressing Brazil's own argument about the existence of such a rule.⁵²⁵

5.157. We recall that, before the Panel, as well as on appeal, Brazil has argued that the appropriate benchmark for comparison with the tax suspensions and exemptions under the PEC and RECAP programmes is the treatment granted to companies that tend to structurally accumulate credits.⁵²⁶ In its analysis of the benchmark treatment with respect to the IPI tax suspensions⁵²⁷, "[t]he Panel agree[d] with Brazil that there are other companies, in addition to the companies accredited or registered as predominantly exporting companies, which are entitled to the IPI tax suspension on the purchase of raw materials, intermediate goods and packaging materials."528 Specifically, the Panel noted that, pursuant to Article 29 of Law 10,637/2002, the tax suspension also applies to: (i) establishments dedicated primarily to the manufacture of products classified in chapters 2, 3, 4, 7, 8, 9, 10, 11, 12, 15, 16, 17, 18, 19, 20, 23, 28, 29, 30, 31, and 64 (21 chapters in total) of the Industrial Goods Tax Classification Table, as well as under codes 2209.00.00 and 2501.00.00, and at positions 21.01 to 21.05.00 of the Industrial Goods Tax Classification Table; (ii) industrial establishments that primarily manufacture components, chassis, bodies, parts, and pieces of automotive products; (iii) industrial establishments that primarily manufacture parts and pieces intended for the aerospace industry; and (iv) industrial establishments that primarily manufacture the goods benefiting from the Informatics programme.⁵²⁹ Moreover, the Panel observed that "[t]he suspension also applies to companies qualified under the Special Regime for the Brazilian Aerospace Industry [(RETAERO)] and the Special Regime to Incentive Computers for Educational Use (REIMCOMP)."530

⁵¹⁸ Brazil's appellant's submission, para. 309 (quoting Panel Reports, paras. 7.1167, 7.1169, and 7.1183). (fn omitted; emphasis omitted)

⁵¹⁹ Brazil's appellant's submission, para. 310 (referring to Appellate Body Reports, *US – FSC (Article 21.5 – EC)*, para. 91; *US – Large Civil Aircraft (2nd complaint)*, para. 815).

⁵²⁰ Brazil's appellant's submission, para. 310 (quoting Appellate Body Report, *US – Large Civil Aircraft* (2nd complaint), para. 813). (emphasis omitted)

⁵²¹ Brazil's appellant's submission, para. 312.

⁵²² Brazil's appellant's submission, para. 313.

⁵²³ Brazil's appellant's submission, para. 315.

⁵²⁴ European Union's appellee's submission, para. 395; Japan's appellee's submission, para. 129.

⁵²⁵ European Union's appellee's submission, para. 398; Japan's appellee's submission, para. 133.

⁵²⁶ Panel Reports, para. 7.1165; Brazil's second written submission to the Panel, para. 216; Brazil's appellant's submission, para. 309.

⁵²⁷ We recall that the Panel subsequently applied this analysis to other categories of the tax treatment. ⁵²⁸ Panel Reports, para. 7.1167.

⁵²⁹ Panel Reports, para. 7.1167.

⁵³⁰ Panel Reports, para. 7.1167 (referring to Brazil's response to Panel question No. 21).

- 65 -

5.158. The Panel, however, considered that the selection of companies entitled to the tax suspensions "d[id] not seem to be directly linked to the problem of credit-accumulation, so as to create a general rule for structurally or predominantly credit accumulating companies".⁵³¹ In the Panel's view, the evidence on the record did not demonstrate that companies to which the suspension applied were structural credit accumulators.⁵³² The Panel observed that, on the one hand, there are producers of low-taxed products, which are more likely to accumulate tax credits, that are not entitled to tax suspensions and, on the other hand, producers of higher taxed products, which are less likely to do so, that were entitled to the suspensions.⁵³³ The Panel further referred to Japan's observation that "companies that export products but do not reach the 50% threshold of gross revenue derived from exports can be credit accumulators and yet cannot benefit from the tax suspensions", which, in the Panel's view, "puts further into question the existence of a general rule for credit accumulators".⁵³⁴ The Panel was "not convinced" that the availability of suspensions for other companies, in addition to the predominantly exporting companies, was "sufficient to prove the existence of a general rule for structurally or predominantly credit accumulating companies".⁵³⁵ The Panel thus found that Brazil has not demonstrated that the tax suspensions were the benchmark treatment for structurally credit-accumulating companies and, instead, defined the benchmark as "the treatment applicable to purchases by non-accredited companies of raw materials, intermediate goods and packaging materials used in the manufacture of their products". 536

5.159. The Panel subsequently relied on these findings in determining the benchmark for comparison with the suspensions of the PIS/PASEP and COFINS contributions and PIS/PASEP-Importation and COFINS-Importation contributions on purchases of raw materials, intermediate goods, and packaging materials (for purposes of PEC)⁵³⁷, as well as the suspensions of the PIS/PASEP and COFINS contributions and PIS/PASEP-Importation and COFINS-Importation contributions on purchases of new machinery, apparatus instruments, and equipment (for purposes of RECAP).⁵³⁸ For these two categories of treatment, the Panel also determined the benchmark to be the treatment applicable to purchases by non-accredited companies, i.e. the obligation to pay the full amount of the applicable tax.⁵³⁹

5.160. On appeal, the participants disagree as to whether, in determining the benchmark treatment, the Panel erroneously sought to ascertain the existence of a "general rule of taxation" instead of examining the organizing principles and structure of Brazil's taxation regime and identifying what constitutes comparable income of comparably situated taxpayers.⁵⁴⁰

- ⁵³⁵ Panel Reports, para. 7.1170.
- ⁵³⁶ Panel Reports, para. 7.1172.
- ⁵³⁷ Panel Reports, paras. 7.1181-7.1187.
- ⁵³⁸ Panel Reports, paras. 7.1197-7.1200.

⁵³⁹ Panel Reports, paras. 7.1186-7.1187 and 7.1200. With respect to the suspension of PIS/PASEP and COFINS contributions and their importation equivalents on purchases of new machinery, apparatus, instruments, and equipment (for purposes of PEC), the Panel observed that "the selection by the Brazilian Government of companies entitled to the suspensions of PIS/PASEP and COFINS contributions, or PIS/PASEP-Importation and COFINS-Importation contributions, does not seem to be directly linked to the problem of credit-accumulation, so as to create a general rule for structurally or predominantly credit accumulating companies." (Ibid., para. 7.1183) With respect to the suspension of PIS/PASEP and COFINS contributions and their importation equivalents on purchases of new machinery, apparatus, instruments, and equipment (for purposes of RECAP), the Panel observed that "the accredited and non-accredited companies purchasing the new machinery, apparatus, instruments and equipment used to manufacture products are identically situated, except for the fact of accreditation." (Ibid., para. 7.1200)

⁵⁴⁰ Brazil's appellant's submission, paras. 309-310, 317, and 323; European Union's appellee's submission, para. 398; Japan's appellee's submission, para. 128.

⁵³¹ Panel Reports, para. 7.1167.

⁵³² Panel Reports, para. 7.1167.

⁵³³ Panel Reports, para. 7.1168. The Panel observed that "an examination of the products for which the IPI tax is suspended or exempted pursuant to Article 29 of Law 10,637/2002 indicates that there is no pattern of suspending the IPI tax only for products with no or low taxation." (Ibid.) In particular, the Panel noted that "many of the products that are subject to special regimes are subject to a tax rate of 5%, 10%, 20% and even 30% and, for instance, Chapters 50 to 54 of the [Table of Application of the Tax on Industrialised Products (TIPI)] are subject to 0% rate and are not covered by Article 29 of Law 10,637/2002." (Ibid. (referring to European Union's response to Panel question No. 41, para. 156))

⁵³⁴ Panel Reports, para. 7.1169.

5.161. Article 1.1(a)(1)(ii) of the SCM Agreement provides:

...

- For the purpose of this Agreement, a subsidy shall be deemed to exist if: 1.1
 - (a)(1)there is a financial contribution by a government or any public body within the territory of a Member (referred to in this Agreement as "government"), i.e. where:
 - (ii) government revenue that is otherwise due is foregone or not collected (e.g. fiscal incentives such as tax credits)[.]⁵⁴¹

5.162. The Appellate Body has previously explained that a panel examining a claim under Article 1.1(a)(1)(ii) of the SCM Agreement must: (i) identify the tax treatment that applies to the income of the alleged subsidy recipients⁵⁴²; (ii) identify a benchmark for comparison⁵⁴³; and (iii) compare the challenged tax treatment and the reasons for it with the benchmark tax treatment.⁵⁴⁴ The Appellate Body has also expressed reservations about a panel seeking to identify a general rule and exception relationship in determining the existence of a revenue otherwise due that is foregone or not collected under Article 1.1(a)(1)(ii). The Appellate Body explained that, "[g]iven the variety and complexity of domestic tax systems, it will usually be very difficult to isolate a 'general' rule of taxation and 'exceptions' to that 'general' rule"⁵⁴⁵, and that an examination under Article 1.1(a)(1)(ii) "must be sufficiently flexible to adjust to the complexities of a Member's domestic rules of taxation".⁵⁴⁶ In fact, "a domestic tax system may be so replete with exceptions" that the rate applicable to the general category of income would no longer represent a general rule but, rather, an exception.⁵⁴⁷ In seeking to identify a general rule and an exception, "a panel might artificially create a rule and an exception where no such distinction exists."548

5.163. The determination of a benchmark for comparison entails, instead, identifying "the tax treatment of comparable income of comparably situated taxpayers". 549 This exercise "involves an examination of the structure of the domestic tax regime and its organizing principles" and requires the panel "to develop an understanding of the tax structure and principles that best explains that Member's tax regime".550 Such an examination must be conducted on the basis of the "rules of taxation that each Member, by its own choice, establishes for itself".⁵⁵¹ In its identification of a benchmark, a panel must be aware of the limitations inherent in identifying and comparing a general rule of taxation and an exception from that rule since such an approach "could result in a finding that government revenue otherwise due has been foregone anytime the tax rate applicable to a recipient is lowered".552

5.164. In addressing Brazil's contention that the suspension of taxes for credit-accumulating companies, including the predominantly exporting companies, is the benchmark treatment, the Panel agreed with Brazil that there were other companies, in addition to those accredited or registered as predominantly exporting companies, for which the IPI tax was suspended.⁵⁵³ We recall the Panel's finding that those were companies referred to in Article 29 of the Law 10,637/2002 and

⁵⁴¹ Fn omitted.

⁵⁴² Appellate Body Report, US – Large Civil Aircraft (2nd complaint), para. 812. The Appellate Body has observed that "[i]dentifying such tax treatment will entail consideration of the objective reasons behind that treatment and, where it involves a change in a Member's tax rules, an assessment of the reasons underlying that change." (Ibid.)

⁵⁴³ Appellate Body Report, US - Large Civil Aircraft (2nd complaint), para. 813.

⁵⁴⁴ Appellate Body Report, *US – Large Civil Aircraft (2nd complaint)*, para. 814. The Appellate Body has explained that "[s]uch a comparison will enable a panel to determine whether, in the light of the treatment of the comparable income of comparably situated taxpayers, the government is foregoing revenue that is otherwise due in relation to the income of the alleged recipients." (Ibid.)

⁵⁴⁵ Appellate Body Report, US - FSC (Article 21.5 - EC), para. 91.

⁵⁴⁶ Appellate Body Report, US - FSC (Article 21.5 - EC), fn 66 to para. 91.

 ⁵⁴⁷ Appellate Body Report, US - Large Civil Aircraft (2nd complaint), para. 815.
 ⁵⁴⁸ Appellate Body Report, US - Large Civil Aircraft (2nd complaint), para. 815.
 ⁵⁴⁹ Appellate Body Report, US - Large Civil Aircraft (2nd complaint), para. 813.

⁵⁵⁰ Appellate Body Report, US - Large Civil Aircraft (2nd complaint), para. 813.

⁵⁵¹ Appellate Body Report, US – Large Civil Aircraft (2nd complaint), para. 813 (quoting Appellate Body Report, US - FSC, para. 90).

⁵⁵² Appellate Body Report, US - Large Civil Aircraft (2nd complaint), para. 815.

⁵⁵³ Panel Reports, para. 7.1167.

- 67 -

under the RETAERO⁵⁵⁴ and REIMCOMP regimes.⁵⁵⁵ We note that, pursuant to Article 29 of the Law 10,637/2002, tax suspensions apply to companies that are dedicated primarily to the manufacture of products classified under 21 chapters and codes 2209.00.00 and 2501.00.00, and positions 21.01 to 21.05.00 of the Industrial Classification Table, as well as companies that primarily manufacture: (i) components, chassis, bodies, parts, and pieces of automotive products; (ii) parts and pieces intended for the aerospace industry; and (iii) goods benefiting from the Informatics programme.⁵⁵⁶

5.165. Notwithstanding this observation, the Panel rejected the treatment of companies that are subject to tax suspensions as a benchmark because, in its view, "the IPI tax suspensions d[id] not seem to be directly linked to the problem of credit-accumulation, so as to create *a general rule* for structurally or predominantly credit accumulating companies."⁵⁵⁷ In reaching this conclusion, the Panel found that there were: (i) producers of low-taxed products, which are more likely to accumulate tax credits, that were not entitled to the suspensions; and (ii) producers of higher taxed products, which are less likely to accumulate tax credits, that were entitled to the suspensions.⁵⁵⁸ Thus, the fact that, on the one hand, not all actual or potential credit accumulators were entitled to the suspensions and, on the other hand, some of the companies entitled to the suspensions were unlikely to accumulate credits, confirmed the Panel's view that there was no "general rule"⁵⁵⁹ for structurally or predominantly credit-accumulating companies. The Panel further mentioned Brazil's argument that "it is not possible to predict as *a general rule* and for all situations when a company that accumulates credits becomes a structurally credit accumulating company."⁵⁶⁰ Having said that, the Panel concluded again that it was "not convinced" that the availability of tax suspensions for other companies, in addition to the predominantly exporting companies, was "sufficient to prove the existence of *a general rule* for structurally or predominantly credit accumulating companies".⁵⁶¹

5.166. In its analysis, the Panel sought to determine the existence of a "general rule" for companies that structurally accumulate credits.⁵⁶² The Panel considered that such a general rule would exist if the tax suspensions were to apply to all the actual or potential credit accumulators and were not to apply to companies that do not or are unlikely to accumulate credits. The Panel rejected the tax suspensions as the benchmark treatment because it considered that some companies to which the tax suspensions applied were not actual or potential credit accumulators, and because some companies to which the suspensions did not apply were accumulating credits. As a result, the Panel could not establish the existence of a "general rule" of taxation.

5.167. However, as noted, in determining a benchmark for comparison, a panel must be cognizant of the limitations inherent in seeking to identify a general rule of taxation and an exception from that rule, because such an approach may lead to an overly narrow conception of which rules are relevant in identifying a benchmark.⁵⁶³ It is not sufficient, once a general rule of taxation has been identified, to conduct an analysis limited to the determination that, but for the challenged measure, a higher tax liability would have attached by virtue of a general rule.⁵⁶⁴ Rather, even if scrutiny of a Member's tax regime indicates the presence of a general rule and an exception relationship, a panel would be expected "to further examine the structure of the domestic tax regime and its organising principles"⁵⁶⁵ in order to determine what is "the tax treatment of comparable income of comparably

⁵⁵⁴ Before the Panel, Brazil explained that RETAERO is aimed at addressing the issue of credit accumulation within suppliers of aerospace inputs by providing a zero rate or suspension of indirect taxes for such inputs. (Brazil's first written submission to the Panel, para. 768 (DS472))

⁵⁵⁵ Panel Reports, para. 7.1167. Brazil explained before the Panel that REIMCOMP was aimed at avoiding tax credit accumulation by computer-equipment manufacturers that win the public bidding process required to sell computers to public schools. (Brazil's first written submission to the Panel, para. 770 (DS472)).

⁵⁵⁶ Panel Reports, para. 7.1167 (referring to Law 10,637/2002 (Panel Exhibit JE-94)).

⁵⁵⁷ Panel Reports, para. 7.1167. (emphasis added)

⁵⁵⁸ Panel Reports, paras. 7.1168-7.1169.

⁵⁵⁹ Panel Reports, paras. 7.1167 and 7.1169-7.1170.

⁵⁶⁰ Panel Reports, para. 7.1170 (quoting Brazil's response to Panel question No. 21). (emphasis added)

⁵⁶¹ Panel Reports, para. 7.1170. (emphasis added) In the similar vein, in its analysis of the benchmark treatment for the suspensions of the PIS/PASEP and COFINS contributions and their importation equivalents, the Panel observed that the selection of companies entitled to the tax suspensions did not seem to be directly linked to the problem of structural credit accumulation, so as to create a general rule for structurally or predominantly credit-accumulating companies. (Panel Reports, para. 7.1183)

⁵⁶² Panel Reports, paras. 7.1167 and 7.1169-7.1170.

⁵⁶³ Appellate Body Report, US – Large Civil Aircraft (2nd complaint), paras. 815 and 818.

⁵⁶⁴ Appellate Body Reports, US – FSC, para. 91; US – Large Civil Aircraft (2nd complaint), paras. 815 and 818.

⁵⁶⁵ Appellate Body Report, US - Large Civil Aircraft (2nd complaint), para. 815.

- 68 -

situated taxpayers".⁵⁶⁶ The Panel commenced its analysis with an examination of Brazil's domestic tax regime by identifying categories of companies subject to tax suspensions. The Panel determined, in this respect, that there were other companies, in addition to those qualified as predominantly exporting companies, that were entitled to the relevant tax suspensions. Having done so, however, the Panel ultimately limited its analysis to seeking to identify the existence of a general rule of taxation to which the challenged treatment would be an exception. By doing so, the Panel effectively predetermined a finding of the existence of the revenue otherwise due that is foregone or not collected under Article 1.1(a)(1)(ii) of the SCM Agreement.

5.168. In our view, instead of seeking to determine the existence of a general rule whereby the tax suspensions would only apply to companies structurally accumulating credits, the Panel should have determined the tax treatment of comparably situated taxpayers.

5.169. We recall, with respect to the IPI tax suspensions, that Article 29 of the Law 10,637/2002, which grants tax suspensions to predominantly exporting companies, also provides for tax suspensions for several other categories of companies.⁵⁶⁷ The range of companies covered by Article 29 is wide, given that the tax suspensions apply, in particular, to companies that primarily manufacture: (i) products classified under 21 chapters of the Industrial Classification Table; (ii) components, chassis, bodies, parts, and pieces of automotive products; (iii) parts and pieces intended for the aerospace industry; and (iv) goods benefiting from the Informatics programme.⁵⁶⁸ In addition, as the Panel found, the IPI tax suspensions apply to companies accredited under the RETAERO and REIMCOMP regimes.⁵⁶⁹ Likewise, with respect to the suspensions of PIS/PASEP and COFINS contributions and their importation equivalents for the purposes of the PEC programme, the Panel found that, in addition to the predominantly exporting companies, such suspensions also apply to other companies. These are the companies that: (i) manufacture certain products when such products are destined for direct public administration bodies; (ii) are accredited under PADIS and PATVD; and (iii) are qualified under the REPES, RETAERO, RETID, and REIMCOMP regimes.⁵⁷⁰

5.170. Thus, the companies that are entitled, or potentially entitled, to the suspension of taxes fall into many categories, each covering a broad number of entities. In our view, the Panel should have considered in detail the treatment of these categories of companies to determine whether they are comparably situated to the predominantly exporting companies rather than seeking to identify the existence of a general rule of taxation for structurally credit-accumulating companies.

5.171. The European Union and Japan argue that the Panel did not seek to establish the existence of a general rule for credit-accumulating companies, but rather was responding to Brazil's own argument that there was a "rule" for credit-accumulating companies providing for the tax suspensions on their supplies.⁵⁷¹ We note, in this respect, that the Panel is not bound by the arguments raised by a party and can use those arguments freely to develop its own legal reasoning to supports its findings and conclusions in the matter under consideration. Moreover, if the panel's analysis was shaped solely by the parties' arguments, the panel may not be able to conduct an objective assessment of the matter, as required under Article 11 of the DSU.⁵⁷² It was therefore incumbent upon the Panel to identify the benchmark for comparison in accordance with the legal

⁵⁶⁶ Appellate Body Report, US - Large Civil Aircraft (2nd complaint), para. 813.

⁵⁶⁷ Panel Reports, para. 7.1167 (referring to Law 10,637/2002 (Panel Exhibit JE-94)).

⁵⁶⁸ Panel Reports, para. 7.1167 (referring to Law 10,637/2002 (Panel Exhibit JE-94)). Similarly, in its determination of the benchmark treatment for the suspensions of the PIS/PASEP and COFINS contributions, or PIS/PASEP-Importation and COFINS-Importation contributions on purchases of raw materials, intermediate goods, and packaging materials for purposes of the PEC programme, the Panel observed that, in addition to the predominantly exporting companies, the relevant suspensions or exemptions apply to: (i) manufacturers of certain products when such products are destined for direct public administration bodies; (ii) companies accredited under PADIS and PATVD; (iii) companies qualified under the Special Tax Regime for the Exportation Platform of Information Technology Services (REPES); (iv) companies qualified under the RETAERO; (v) companies qualified under the Special Regime for the Defense Industry (RETID); and (vi) companies qualified under the REIMCOMP. (Panel Reports, para. 7.1183 (referring to Brazil's response to Panel question No. 21)) 569 Panel Reports, para. 7.1167.

⁵⁷⁰ Panel Reports, para. 7.1183 (referring to Brazil's response to Panel question No. 21).

⁵⁷¹ European Union's appellee's submission, para. 398; Japan's appellee's submission, para. 133.

⁵⁷² Appellate Body Reports, EC - Hormones, para. 156; Korea - Dairy, para. 139; US - Certain EC Products, para. 123.

- 69 -

standard under Article 1.1(a)(1)(ii) of the SCM Agreement, regardless of how Brazil had presented its arguments on this issue.⁵⁷³

5.172. In light of the above, we reverse the Panel's conclusions, in paragraphs 7.1171 and 7.1172 of the Panel Reports, that "Brazil has not demonstrated that the tax suspensions are the benchmark treatment for structurally credit-accumulating companies" ⁵⁷⁴ and that "the treatment applicable to purchases by non-accredited companies of raw materials, intermediate goods and packaging materials ... can be considered as the benchmark treatment or normal rule of general application."575 We recall that the Panel relied on its reasoning and findings concerning the IPI tax suspensions in identifying the benchmark treatment for the remaining two categories of the challenged tax treatment.⁵⁷⁶ We therefore also reverse the Panel's findings, in paragraphs 7.1186-7.1187 and 7.1199-7.1200 of the Panel Reports, that Brazil has not demonstrated that the tax suspensions are the benchmark for comparison and that the appropriate benchmark is, instead, the treatment applicable to purchases by non-accredited companies of the relevant products. As a result, we also reverse the Panel's findings, in paragraphs 7.1211, 7.1223, and 7.1238, as well as in paragraphs 8.7 and 8.18, of the Panel Reports, that the tax suspensions granted to registered or accredited companies under the PEC and RECAP programmes constitute financial contributions in the form of government revenue otherwise due that is foregone or not collected and are hence subsidies within the meaning of Article 1.1 of the SCM Agreement that are contingent upon export performance under Article 3.1(a) of the SCM Agreement.

5.173. Having reversed the Panel's findings concerning the identification of the benchmark for comparison, we do not need to further address Brazil's alternative claims concerning the application of Article 1.1(a)(1)(ii) of the SCM Agreement and the Panel's assessment of evidence under Article 11 of the DSU.

5.174. We now turn to consider whether we can complete the analysis in finding whether the tax suspensions granted to registered or accredited companies under the PEC and RECAP programmes constitute a financial contribution in the form of government revenue otherwise due that is foregone or not collected under Article 1.1(a)(1)(ii) of the SCM Agreement. We recall that, in previous disputes, the Appellate Body has completed the analysis with a view to facilitating the prompt settlement and effective resolution of the dispute⁵⁷⁷ and has completed the analysis where the factual findings in the panel report and undisputed facts on the panel record provided it with a sufficient basis for conducting its own analysis.⁵⁷⁸ The Appellate Body has declined to complete the analysis in light of the complexity of issues, the absence of full exploration of the issues before the panel, and considerations pertaining to the parties' due process rights.⁵⁷⁹

5.175. We recall that the Panel's assessment under Article 1.1(a)(1)(ii) of the SCM Agreement was based on the benchmark that the Panel considered to be correct, i.e. the treatment applicable to purchases by non-accredited companies. Thus, we first have to determine whether we can complete

 $^{^{573}}$ We also note that the main point of Brazil's arguments before the Panel was that "an appropriate comparison under Article 1.1(a)(1)(ii) should be made between the fiscal treatment of predominantly exporting companies and other companies that tend to structurally accumulate credits." (Brazil's second written submission to the Panel, para. 216)

⁵⁷⁴ Panel Reports, para. 7.1171.

⁵⁷⁵ Panel Reports, para. 7.1172.

⁵⁷⁶ We recall that these are: (i) the suspensions of the PIS/PASEP and COFINS contributions, or PIS/PASEP-Importation and COFINS-Importation contributions, on purchases of raw materials, intermediate goods, and packaging materials (for purposes of PEC); and (ii) the suspensions of the PIS/PASEP and COFINS contributions, or PIS/PASEP-Importation and COFINS-Importation contributions, on purchases of new machinery, apparatus, instruments, and equipment incorporated into the fixed assets of the accredited company (for purposes of RECAP). (Panel Reports, paras. 7.1181-7.1187 and 7.1197-7.1200) ⁵⁷⁷ See Appellate Body Reports, *EC and certain member States – Large Civil Aircraft*, para. 1178;

⁵⁷⁷ See Appellate Body Reports, *EC and certain member States – Large Civil Aircraft*, para. 1178; *US – Large Civil Aircraft (2nd complaint)*, para. 1351; *EC – Asbestos*, para. 78.

⁵⁷⁸ See Appellate Body Reports, *Australia – Salmon*, paras. 241 and 255; *Korea – Dairy*, para. 102; *US – Large Civil Aircraft (2nd complaint)*, para. 653; *US – Section 211 Appropriations Act*, para. 343; *EC – Asbestos*, para. 78; *China – HP-SSST (Japan) / China – HP-SSST (EU)*, para. 5.135; *US – Anti-Dumping Methodologies (China)*, paras. 5.163-5.164.

⁵⁷⁹ See Appellate Body Reports, *Korea – Dairy*, para. 102; *Canada – Renewable Energy / Canada – Feed-in Tariff Program*, para. 5.224; *EC – Seal Products*, paras. 5.63 and 5.69; *US – Anti-Dumping Methodologies (China)*, para. 5.178.

- 70 -

the analysis and determine for ourselves the benchmark treatment for the three categories of the tax treatment under the PEC and RECAP programmes identified by the Panel.

5.176. We note that the Panel acknowledged that there are other companies, in addition to the predominantly exporting companies, that are entitled to the tax suspensions.⁵⁸⁰ The Panel, however, did not examine in detail and did not make specific findings concerning the conditions of eligibility for the tax exemptions of these other companies and their operation. Moreover, in their arguments on appeal, the participants focused on whether the Panel erroneously determined the benchmark by focusing on identifying the general rule of taxation, rather than on the identification of the correct benchmark. In these circumstances, we cannot complete the analysis to find whether there exists a financial contribution in the form of government revenue otherwise due that is foregone or not collected within the meaning of Article 1.1(a)(1)(ii) of the SCM Agreement. Accordingly, we are unable to complete the analysis to find whether the PEC and RECAP programmes constitute subsidies under Article 1.1 of the SCM Agreement.

5.4 Article 3.1(b) the SCM Agreement

5.4.1 Article 1.1(a)(1)(ii) of the SCM Agreement: ICT programmes

5.4.1.1 Introduction

5.177. We now turn to consider Brazil's claim on appeal that the Panel erred in finding that the tax treatment of intermediate products and inputs under the ICT programmes constitutes a subsidy under Article 1.1 of the SCM Agreement.⁵⁸¹ Brazil contends that the Panel erred in finding that the "implicit interest income" with respect to intermediate products and inputs constitutes government revenue otherwise due that is foregone under Article 1.1(a)(1)(ii) of the SCM Agreement for two reasons. First, Brazil claims that "the Panel failed to compare the treatment accorded to the *group* of taxpayers in the benchmark treatment with the *group* of taxpayers under the challenged treatment."⁵⁸² Second, Brazil contends that "the Panel erroneously found that the 'implicit interest income' that Brazil purportedly 'foregoes' in respect of intermediate products and inputs under the ICT programmes constitutes 'government revenue' that is otherwise 'due' under Article 1.1(a)(1)(ii) of the SCM Agreement."⁵⁸³ At the oral hearing, Brazil stated that it also takes issue with the Panel's determination of the benchmark treatment for the comparison with the challenged treatment of intermediate products and inputs.⁵⁸⁴

5.178. The European Union and Japan submit that the Panel did not err in finding the existence of a financial contribution under Article 1.1(a)(1)(ii) of the SCM Agreement for certain categories of the tax treatment under the ICT programmes.⁵⁸⁵ In particular, the European Union and Japan note that the Panel did not disregard within the benchmark the treatment of a group of taxpayers that normally offset tax credits within the same taxation period.⁵⁸⁶ Rather, the Panel took, according to the European Union, a "balanced approach" and weighed "all the scenarios that may arise under the normal operation of Brazil's tax system and notably the credit-debit mechanism to conclude that there was revenue foregone or not collected in the 'rather frequent scenario' where taxpayers cannot offset their tax credits within the same taxation period".⁵⁸⁷ The European Union and Japan further disagree with Brazil's claim that the Panel erred in the interpretation and application of

⁵⁸⁵ Panel Reports, para. 7.406.d-h.

⁵⁸⁰ Panel Reports, paras. 7.1167 and 7.1183 (referring to Brazil's response to Panel question No. 21).

⁵⁸¹ Brazil's appellant's submission, para. 177.

⁵⁸² Brazil's appellant's submission, para. 180. (emphasis original)

⁵⁸³ Brazil's appellant's submission, para. 180. Should we agree with Brazil and reverse the relevant findings, Brazil requests that we also reverse the Panel's consequential finding that "the exemption, reduction and suspension of taxes on sales of intermediate goods and on purchases of raw materials, intermediate goods, packaging materials, inputs, capital goods and computational tools under the ICT programmes confer a 'benefit' on the recipient under Article 1.1(b) of the SCM Agreement." (Ibid., para. 242) Brazil further requests that we reverse the Panel's ultimate finding that the relevant tax reductions and exemptions constitute subsidies under Article 1 of the SCM Agreement and declare moot and of no legal effect the Panel's recommendation under Article 4.7 of the SCM Agreement that Brazil withdraw the ICT programmes in 90 days. (Ibid., paras. 242-244 (referring to Panel Reports, paras. 7.489-7.490, 7.493-7.495, 7.500, 8.5.e, 8.11, 8.16.f, and 8.22))

⁵⁸⁴ Brazil's response to questioning at the oral hearing.

⁵⁸⁶ European Union's appellee's submission, para. 279; Japan's appellee's submission, para. 90.

⁵⁸⁷ European Union's appellee's submission, para. 282. See also Japan's appellee's submission, as 92-94

- 71 -

Article 1.1(a)(1)(ii) of the SCM Agreement in finding that the possible cash availability and implicit interest income that the Brazilian Government could earn (had the tax been paid in advance and not offset) constitute revenue otherwise "due" by the taxpayer.⁵⁸⁸

5.4.1.2 Panel's findings

5.179. At the outset of its analysis, the Panel recalled the Appellate Body's statement in US – Large Civil Aircraft (2nd complaint) that, in order to determine whether there is a financial contribution in the form of government revenue otherwise due that is foregone or not collected under Article 1.1(a)(1)(ii) of the SCM Agreement, a panel must: (i) identify the treatment applicable to the alleged recipients of the subsidy and the objective reasons behind it; (ii) identify a benchmark for comparison; and (iii) compare the challenged treatment, and its reasons, to the benchmark treatment.⁵⁸⁹

5.180. First, the Panel identified the challenged tax treatment applicable to the accredited companies and the objective reasons behind that treatment.⁵⁹⁰ The Panel noted that "each of the examined programmes provides a different package of exemptions, reductions, and/or suspensions from one or more particular types of tax."⁵⁹¹ The Panel then classified the different tax treatments at issue on the basis of: (i) the tax at issue; (ii) whether the tax treatment is on the purchase of products by accredited companies or on their sale of incentivized products; and (iii) the degree of transformation of the incentivized products. Based on these criteria, the Panel grouped the tax treatments into ten categories.⁵⁹²

5.181. The Panel then identified a benchmark for each category of tax treatment. In every instance, the Panel found that "the benchmarks to be applied are the economy-wide tax treatments from which the exemptions, reductions and suspensions are taken."⁵⁹³ Having compared the challenged tax treatments to the benchmark treatments, the Panel concluded that each of the challenged tax reductions, exemptions, and suspensions granted to accredited companies on the sales of: (i) finished goods that they produce (under the Informatics, PATVD, and Digital Inclusion programmes); (ii) intermediate goods that they produce (under the Informatics and PADIS programmes); and (iii) raw materials, intermediate goods, and packaging materials (under the Informatics programme) and inputs, capital goods, and computational tools (under the PADIS and PATVD programmes) constitutes financial contributions where "government revenue that is otherwise due is foregone or not collected."⁵⁹⁴

5.182. In its analysis with respect to the various tax exemptions and reductions on the sales of the intermediate goods and raw materials, intermediate goods and packaging materials and inputs, and capital goods and computational goods, the Panel followed the same analytical approach for all the challenged categories of tax treatment.⁵⁹⁵

5.183. In particular, regarding the IPI tax exemptions and reductions on sale of incentivized intermediate goods (for purposes of the Informatics and PADIS programmes), the Panel observed that, under the benchmark treatment, the non-accredited company selling the non-incentivized product would charge the relevant tax to the buyer and would transmit it to the Federal Revenue Service, while the buyer would accrue a credit that it would be allowed to use to offset its IPI debits during the same (i.e. when it pays its monthly liabilities on the twenty-fifth day of the month following the transaction) or subsequent taxation periods. If the buyer is not able to offset the credit after three taxation periods, it could ask the Brazilian Government for compensation with other

⁵⁸⁸ European Union's appellee's submission, paras. 266 and 289; Japan's appellee's submission, para. 99.

⁵⁸⁹ Panel Reports, paras. 7.394-7.395 (referring to Appellate Body Report, *US – Large Civil Aircraft (2nd complaint)*, paras. 812-815).

⁵⁹⁰ Panel Reports, para. 7.401.

⁵⁹¹ Panel Reports, para. 7.406.

⁵⁹² Panel Reports, para. 7.406.a-j. On appeal, Brazil takes issue with the Panel's analysis under

para. 7.406.d-h. (Brazil's appellant's submission, para. 185 (referring to Panel Reports, paras. 7.430-7.475)) ⁵⁹³ Panel Reports, para. 7.414.

⁵⁹⁴ Panel Reports, paras. 7.488-7.490. We thus set out the summary of the Panel's findings with respect to the IPI tax exemptions and reductions on the sale of incentivized intermediate goods (for purposes of the Informatics and PADIS programmes) as an example.

⁵⁹⁵ Panel Reports, paras. 7.430-7.475.

- 72 -

federal debits or reimbursement.⁵⁹⁶ The Panel noted that "[i]t is fairly common that companies are not able to offset the full amount of their credits and have to request compensation or reimbursement, so this scenario seems rather frequent."⁵⁹⁷

5.184. The Panel observed that, "under the best case scenario for the buyer of the non-incentivized intermediate products, the Government will receive the full amount of IPI tax due from the non-accredited company selling the non-incentivized intermediate products and the buyer ... will be able to offset the amount of IPI tax paid during the same taxation period."598 The Panel further observed that, when the buyer of the non-incentivized products is unable to offset the credit during the same taxation period, the Brazilian Government will retain the amount of the tax paid by the non-accredited company selling the non-incentivized product until the buyer is able to offset it or until the Brazilian Government compensates or reimburses the face amount of the credit, without interest, to the buyer. The Panel considered that, under the second scenario (i.e. the scenario under which the buyer is unable to offset the tax credit during same taxation period), the Brazilian Government would "hold the advantage of cash availability or cash flow, along with the associated implicit interest income (revenue) that could be generated on the full amount of that tax that it has collected from the seller".⁵⁹⁹ The cash availability and implicit interest could last from one taxation period (i.e. one month) to 360 days. If the reimbursement is made to the buyer within 360 days after the request, the Brazilian Government does not have to pay any interest on the use of the buyer's money. 600

5.185. By contrast, the Panel noted that, under the challenged treatment, the accredited seller would not charge the tax to the buyer and would not remit it to the Brazilian Government, while the buyer would not accrue any tax credits.⁶⁰¹ The Panel considered that, in the scenario where the tax is paid to the Brazilian Government and offset during the same taxation period, there would be no revenue foregone for the Brazilian Government. However, in the Panel's view, in the "rather frequent scenario" when the buyer is not able to offset the tax credit during the same taxation period, the Brazilian Government would forego revenue in the form of the implicit interest on the tax revenue collected where the offsetting credits have not been used.⁶⁰² The Panel noted that, even though ultimately the Brazilian Government would collect the same nominal amount of tax, this would not diminish or eliminate the advantage that the buyer of the incentivized intermediate products, rather than the Brazilian Government, would enjoy in the form of cash availability and the implicit interest income due to not having to pay the tax.⁶⁰³

5.186. The Panel further recalled that several panels had previously concluded that, whenever there is revenue foregone by the government, a benefit is conferred.⁶⁰⁴ Thus, having concluded that the tax treatments at issue constituted financial contributions where "government revenue that is otherwise due is foregone or not collected", the Panel also concluded that the tax measures at issue conferred a benefit because the buyers of the incentivized products under the relevant programmes are better off with the exemptions and reductions than in the benchmark scenario.⁶⁰⁵ The Panel thus concluded that the measures at issue constitute subsidies within the meaning of Article 1 of the SCM Agreement.⁶⁰⁶

paras. 7.169-7.171; Canada - Renewable Energy / Canada - Feed-in Tariff Program, fn 509 to para. 7.271;

US – *FSC*, para. 7.103; *US* – *FSC* (*Article 21.5* – *EC*), paras. 8.44-8.48; Appellate Body Reports, *US* – *FSC*, para. 140; *US* – *FSC* (*Article 21.5* – *EC*), para. 198).

⁵⁹⁶ Panel Reports, para. 7.432. See also Panel Reports, paras. 7.440, 7.449, 7.458, and 7.468. ⁵⁹⁷ Panel Reports, para. 7.432. (fn omitted) See also Panel Reports, paras. 7.440, 7.449, 7.458,

and 7.468.

⁵⁹⁸ Panel Reports, para. 7.433. See also Panel Reports, paras. 7.441, 7.450, 7.459, and 7.469.

⁵⁹⁹ Panel Reports, para. 7.433. (fn omitted) See also Panel Reports, paras. 7.441, 7.450, 7.459,

and 7.469. The Panel remarked that, "[a]ccording to the European Union, the benchmark rate of the ... Central Bank [of Brazil] ... was, at the time of writing of its first written submission, 13.25%." (Panel Reports, fn 790 to para. 7.433 (referring to European Union's first written submission to the Panel, fn 883 to para. 1201))

⁶⁰⁰ Panel Reports, para. 7.433. See also Panel Reports, paras. 7.441, 7.450, 7.459, and 7.469.

⁶⁰¹ Panel Reports, para. 7.434. See also Panel Reports, paras. 7.442, 7.451, 7.461, and 7.471.

⁶⁰² Panel Reports, para. 7.436. See also Panel Reports, paras. 7.444, 7.454, 7.463, and 7.473.

⁶⁰³ Panel Reports, para. 7.437. See also Panel Reports, paras. 7.445, 7.455, 7.464, and 7.474.

⁶⁰⁴ Panel Reports, para. 7.491 (referring to Panel Reports, US - Large Civil Aircraft (2nd complaint),

⁶⁰⁵ Panel Reports, paras. 7.493-7.494.

⁶⁰⁶ Panel Reports, para. 7.495.

- 73 -

5.4.1.3 The Panel's benchmark determination

5.187. Before turning to address Brazil's claims regarding the Panel's comparison of the benchmark treatment with the challenged treatment, we address Brazil's statement, made at the oral hearing, that it takes issue with the Panel's determination of the benchmark in the context of its analysis under the ICT programmes. We recall that, in its Notice of Appeal, Brazil did not expressly raise a claim concerning the Panel's determination of the benchmark treatment as relevant to its findings under Article 3.1(b) of the SCM Agreement.⁶⁰⁷ Furthermore, in its appellant's submission, Brazil has **argued that "[t]he Panel ... correctly held that the appropriate benchmark treatment is 'the treatment** applicable **to sales ... by non**-accredited companies'."⁶⁰⁸ In response to questioning at the oral hearing, however, Brazil stated that it takes issue with the Panel's identification of the benchmark for comparison with the challenged treatment of intermediate goods and inputs under the ICT programmes.⁶⁰⁹ The European Union responded that, since this claim was not raised by Brazil in its Notice of Appeal and its appellant's submission, we should dismiss it.⁶¹⁰

5.188. We recall, in this respect, that Rule 20(2)(d) and Rule 21(2)(b) of the Working Procedures set out the requirements for the Notice of Appeal and the appellant's submission, respectively. In particular, in accordance with Rule 20(2)(d), the Notice of Appeal "shall include" a brief statement of the nature of the appeal, comprising: (i) "identification of the alleged errors in the issues of law covered in the panel report and legal interpretations developed by the panel"; (ii) "a list of the legal provision(s) of the covered agreements that the panel is alleged to have erred in interpreting or applying"; and (iii) "an indicative list of the paragraphs of the panel report containing the alleged errors".⁶¹¹ Pursuant to the Rule 21(2)(b)(i) of the Working Procedures, the appellant's submission "shall" set out, *inter alia*, "a precise statement of the grounds for the appeal, including the specific allegations of errors in the issues of law covered in the panel report and legal interpretations developed by the panel, and the legal arguments in support thereof".⁶¹²

5.189. Rule 20(2)(d) of the Working Procedures serves an important due process function as it helps to ensure the balance "that must be maintained between the right of Members to exercise the right of appeal meaningfully and effectively, and the right of appellees to receive notice through the Notice of Appeal of the findings under appeal, so that they may exercise their right" to respond.⁶¹³ The Appellate Body has also cautioned that, if a particular claim of error is not raised by the appellant in the Notice of Appeal, then that claim is "not properly within the scope of the appeal, and the Appellate Body will not make findings thereon".⁶¹⁴ With regard to the appellant's submission, the Appellate Body has stated that, compared to the Notice of Appeal, this submission "must be more specific", in that it "must be precise as to the grounds of appeal, the legal arguments which support it, and the provisions of the covered agreements and other legal sources upon which the appellant relies".⁶¹⁵

⁶⁰⁷ Brazil's Notice of Appeal, para. 16. We recall that, by contrast, Brazil raised a claim concerning the Panel's determination of benchmark treatment under the PEC and RECAP programmes. See section 5.3 of these Reports.

⁶⁰⁸ Brazil's appellant's submission, para. 186.

⁶⁰⁹ Brazil's response to questioning at the oral hearing.

⁶¹⁰ European Union's response to questioning at the oral hearing.

⁶¹¹ Working Procedures, Rule 20(2)(d).

⁶¹² Working Procedures, Rule 21(2)(b)(i).

⁶¹³ Appellate Body Report, US – Countervailing Measures on Certain EC Products, para. 62.

⁶¹⁴ Appellate Body Report, EC – Fasteners (China), para. 582 (referring to Appellate Body Reports,

US – Countervailing Measures on Certain EC Products, para. 72; EC – Bananas III (Article 21.5 – Ecuador II) / EC – Bananas III (Article 21.5 – US), para. 285; Japan – Apples, paras. 124-128).

⁶¹⁵ Appellate Body Report, US – Countervailing Measures on Certain EC Products, para. 59.

5.190. We recall that Brazil's Notice of Appeal provides, in relevant part:

The Panel erred in finding that the tax exemptions, suspensions and reductions granted under the ICT programmes both on the sales of intermediate goods and on the purchases of raw materials, intermediate goods, packaging materials, inputs, capital goods and computational tools constitute financial contributions, in the form of government revenue otherwise due that is foregone or not collected, within the meaning of Article 1.1(a)(1)(ii) of the SCM Agreement; [*]⁶¹⁶

[*fn original]¹¹ See Panel Report[s], paras. 7.432-7.433; 7.444-7.445; 7.454-7.455; 7.463-7.464; 7.473-7.474; and 7.495.

5.191. In its Notice of Appeal, Brazil thus framed its claim under Article 1.1(a)(1)(ii) of the SCM Agreement in broad terms by stating that "[t]he Panel erred in finding that [the tax treatments at issue] constitute financial contributions, in the form of government revenue otherwise due that is foregone or not collected." We note, however, that the indicative list of paragraphs in the Panel Reports containing the alleged errors provided by Brazil in footnote 11 of its Notice of Appeal does not refer to paragraphs in which the Panel addressed the benchmark determination. Moreover, we note that, in its appellant's submission, Brazil stated, in several instances, that it agreed with the Panel's determination of the benchmark treatment. In particular, Brazil affirmed that "the Panel began by correctly stating, in broad terms, the appropriate benchmark for comparison."⁶¹⁷ Brazil also stated, to the same effect, that the Panel "correctly held that the appropriate benchmark **treatment is 'the treatment applicable to sales ... by non**-accredited companies, i.e. the obligation to pay the full amount of the applicable tax'".⁶¹⁸

5.192. Thus, in its Notice of Appeal, Brazil formulated its claim under Article 1.1(a)(1)(ii) of the SCM Agreement in broad terms, which, in principle, could include the Panel's identification of the benchmark treatment. However, in its appellant's submission, which is supposed to be "more specific" and "precise as to the grounds of appeal"⁶¹⁹, Brazil clearly indicated that it agreed with the Panel's identification of the benchmark treatment. Accordingly, Brazil's Notice of Appeal, read together with Brazil's appellant's submission, demonstrates that, until the oral hearing, Brazil did not raise a claim regarding the Panel's benchmark determination as it related to its findings under Article 3.1(b) of the SCM Agreement. Moreover, at the oral hearing, Brazil did not develop argumentation in support of that claim.

5.193. In these circumstances, we consider that Brazil's claim that the Panel erred in its determination of the benchmark in the context of its analysis under the ICT programmes was not properly raised before us and hence is not within the scope of this appeal. We thus proceed to address the claims that Brazil has properly raised on appeal.

5.4.1.4 Whether the Panel erred in its comparison of the benchmark treatment with the challenged treatment

5.194. Brazil claims that the Panel erred in its comparison of the treatment of intermediate goods and inputs under the ICT programmes with the benchmark treatment by "arbitrarily distinguish[ing] between taxpayers situated within the benchmark".⁶²⁰ Specifically, Brazil points out that "the Panel failed to compare the treatment accorded to the *group* of taxpayers in the benchmark treatment with the *group* of taxpayers under the challenged treatment."⁶²¹ In Brazil's view, the Panel "dismissed as a mere 'best case scenario' the treatment"⁶²² applicable to the companies offsetting the entire amount of taxes paid during the same taxation period and opted to compare the challenged treatment to a small group of taxpayers within the benchmark, i.e. those that are unable to offset the full amount of tax credits during the same taxation period.⁶²³ Brazil considers that the comparison that the Panel should have undertaken is analogous to the assessment of less favourable

⁶¹⁶ Brazil's Notice of Appeal, para. 16 and fn 11 thereto.

⁶¹⁷ Brazil's appellant's submission, para. 185.

⁶¹⁸ Brazil's appellant's submission, para. 186 (quoting Panel Reports, paras. 7.430, 7.438, 7.447, 7.456, and 7.466).

⁶¹⁹ Appellate Body Report, US – Countervailing Measures on Certain EC Products, para. 59.

⁶²⁰ Brazil's appellant's submission, para. 189. (emphasis omitted)

⁶²¹ Brazil's appellant's submission, para. 180. (emphasis original)

⁶²² Brazil's appellant's submission, para. 188.

⁶²³ Brazil's appellant's submission, para. 189.

- 75 -

treatment under Article III: 4 of the GATT 1994.⁶²⁴ Specifically, Brazil contends that, "[i]n order to establish that the ICT programmes involved the foregoing of revenue otherwise due, the Panel was required to compare the treatment of the *group* of taxpayers under those programmes with the *group* of taxpayers in the benchmark treatment."⁶²⁵

5.195. The European Union and Japan submit that the Panel did not commit an error in comparing the treatment provided under the ICT programmes with the tax treatment granted to the non-accredited companies.⁶²⁶ Rather, as the European Union explains, the Panel conducted an "overall assessment of the tax treatment provided to similarly situated taxpayers, i.e. including those more frequently accumulating tax credits".⁶²⁷ Japan adds that the Panel "made a carefully considered determination that, *at least in some factual situations*, accredited companies ... are better off than those within the benchmark, and the Brazilian Government, by contrast, is foregoing revenue that would be otherwise due".⁶²⁸

5.196. The foregoing (or non-collection) of revenue otherwise due under Article 1.1(a)(1)(ii) of the SCM Agreement implies that less (or no) revenue has been raised by the government than would have been raised in a different situation.⁶²⁹ As noted above, the Appellate Body has previously explained that a panel should follow a three-step test in order to determine whether the revenue that is otherwise due is foregone or not collected under Article 1.1(a)(1)(ii) of the SCM Agreement.⁶³⁰ Specifically, a panel examining a claim under Article 1.1(a)(1)(ii) of the SCM Agreement must: (i) identify the tax treatment that applies to the income of the alleged subsidy recipients⁶³¹; (ii) identify a benchmark for comparison⁶³²; and (iii) compare the challenged treatment and reasons for it with the benchmark tax treatment.⁶³³ Brazil takes issue with the Panel's analysis under the third step of the test, i.e. the comparison of the challenged treatment and its reasons with the benchmark tax treatment.⁶³⁴ We recall that the Appellate Body observed, in this respect, that "[s]uch a comparison will enable a panel to determine whether, in the light of the treatment of the comparable income of comparably situated taxpayers, the government is foregoing revenue that is otherwise due in relation to the income of the alleged recipients."⁶³⁵

5.197. We note that the structure and operation of the tax exemptions and suspensions under the various categories of the tax treatments identified by the Panel are similar. Moreover, the Panel's comparison of the benchmark treatment follows the same logic and structure for all the relevant

⁶²⁴ Brazil's appellant's submission, para. 190 (referring to Appellate Body Report, *EC – Asbestos*, para. 100).

⁶²⁵ Brazil's appellant's submission, para. 191. (emphasis original)

⁶²⁶ European Union's appellee's submission, para. 284; Japan's appellee's submission, para. 95.

⁶²⁷ European Union's appellee's submission, para. 284.

⁶²⁸ Japan's appellee's submission, para. 94.

⁶²⁹ Appellate Body Report, US – Large Civil Aircraft (2nd complaint), para. 806 (referring to Appellate Body Report, US – FSC, para. 90).

⁶³⁰ See paragraph 5.162 above.

⁶³¹ Appellate Body Report, *US* – *Large Civil Aircraft (2nd complaint)*, para. 812. The Appellate Body has observed that "[i]dentifying such tax treatment will entail consideration of the objective reasons behind that treatment and, where it involves a change in a Member's tax rules, an assessment of the reasons underlying that change." (Ibid.)

⁶³² Appellate Body Report, US – Large Civil Aircraft (2nd complaint), para. 813.

⁶³³ Appellate Body Report, US - Large Civil Aircraft (2nd complaint), para. 814.

⁶³⁴ Brazil's appellant's submission, para. 191.

⁶³⁵ Appellate Body Report, US - Large Civil Aircraft (2nd complaint), para. 814.

- 76 -

categories of tax treatment of intermediate products and inputs.⁶³⁶ We thus review the Panel's analysis with respect to the IPI tax exemptions and reductions on the sales of the incentivized intermediate goods under the Informatics and PADIS programmes as an example.⁶³⁷

5.198. We recall that, having identified the benchmark treatment as the treatment of non-accredited companies, the Panel proceeded to examine how the relevant general rules of taxation operate with respect to the non-accredited companies. The Panel explained that, under the benchmark treatment, the buyer of the non-incentivized product would pay the tax to the non-accredited seller, who would transmit it to the Brazilian Government. The buyer, in turn, would accrue a tax credit, which it would be able to offset against its future tax liabilities, i.e. tax debits. It would be able to do so within the same taxation period or within three months since the date of the transaction. If the company is unable to do so, it could request the Brazilian Government to compensate or reimburse the taxes that could not be offset. The Brazilian Government has 360 days to respond to such a request, and if it does not do so within that time, the buyer of the non-incentivized products will be entitled to compensation or reimbursement and the interest generated.⁶³⁸ The Panel then noted that "[i]t is fairly common that companies are not able to offset the full amount of their credits and have to request compensation or reimbursement, so this scenario seems rather frequent."639

5.199. In comparing "the challenged treatment with the best case scenario for the buyer of the non-incentivized products at issue under the benchmark treatment" (i.e. the scenario where the tax is paid to the Brazilian Government and offset during the same taxation period), the Panel found that "there would be no revenue forgone by the Brazilian Government, as the government would not earn the implicit interest on unused credits."640

5.200. By contrast, the Panel noted that, under the "rather frequent scenario", the buyer of the non-incentivized products would not be able to offset the tax credit it has accrued during the same taxation period in which the tax is paid.⁶⁴¹ According to the Panel, in this scenario, the Brazilian Government would retain the amount of tax paid by the non-accredited company until the buyer is able to offset it during the subsequent taxation periods, or until the Brazilian Government compensates or reimburses the amount of the credit to the buyer within 360 days from the request for compensation or reimbursement.⁶⁴² Having compared this scenario to the challenged treatment, the Panel concluded that, "under this rather frequent scenario, the Brazilian Government is foregoing revenue in the form of the implicit interest on the tax revenue collected where the offsetting credits have not (yet) been used."643

5.201. In Brazil's view, the Panel "dismissed as a mere 'best case scenario' the treatment generally applicable to taxpayers in the benchmark", i.e. offsetting the entire amount of taxes paid during the same taxation period, and "opted to compare" the tax treatment at issue with what it considered to

⁶³⁶ Panel Reports, paras. 7.430-7.475. We recall that Brazil takes issue with the Panel's analysis with respect to the following categories of the tax treatment: (i) the IPI tax exemptions (including through zero rates) and reductions granted to accredited companies on their sales of the incentivized intermediate goods that they produce (for purposes of the Informatics and PADIS programmes); (ii) the exemptions (through zero rates) of the PIS/PASEP and COFINS contributions granted to accredited companies on their sales of the incentivized intermediate goods that they produce (for purposes of the PADIS programme); (iii) the IPI tax suspensions on purchases of raw materials, intermediate goods, and packaging materials (for purposes of the Informatics programme) and IPI tax exemptions (through zero rates) on purchases of inputs (for purposes of the PADIS and PATVD programmes) used to manufacture the incentivized products: (iv) the exemptions (though zero rates) of the PIS/PASEP and COFINS contributions or PIS/PASEP-Importation and COFINS-Importation contributions on purchases of inputs (for purposes of the PADIS and PATVD programmes) used to manufacture the incentivized products; and (v) the exemptions (through zero rates) of the PIS/PASEP and COFINS contributions or PIS/PASEP-Importation and COFINS-Importation contributions on purchases of certain capital goods and computational tools (for purposes of the PADIS and PATVD programmes) used to manufacture the incentivized products that will be incorporated into the fixed assets of the accredited company. (Brazil's appellant's submission, para. 185) ⁶³⁷ Panel Reports, paras. 7.431-7.437.

⁶³⁸ Panel Reports, para. 7.432.

⁶³⁹ Panel Reports, para. 7.432 (referring to Brazil's first written submission to the Panel, para. 758 (DS472) and para. 691 (DS497); Exporters 2014 study on Brazilian exports - delays in refunds (Panel Exhibit JE-186), p. 55; European Union's first written submission to the Panel, fn 849 to para. 1172).

^b Panel Reports, para. 7.436.

⁶⁴¹ Panel Reports, paras. 7.432-7433.

⁶⁴² Panel Reports, para. 7.433.

⁶⁴³ Panel Reports, para. 7.436.

be the "'rather frequent' scenario".⁶⁴⁴ By doing so, in Brazil's view, the Panel "arbitrarily distinguish[ed] between taxpayers situated *within* the benchmark".⁶⁴⁵

5.202. We disagree with Brazil's characterization of the Panel's approach to comparing the challenged treatment to the selected benchmark treatment. As we see it, the Panel did not treat only one subset of taxpayers (i.e. those that are unable to offset the amount of the tax paid during the same taxation period) as the benchmark for the purposes of comparison. Rather, having explained in detail how the mechanism of credits and debits under the principle of non-accumulation works, the Panel concluded that, under the normal rule of general application of Brazil's tax system, there are two possible factual scenarios: one in which the buyer of non-incentivized products will be able to offset the amount of the tax paid during the same taxation period, and the other one when this would not be possible. The Panel then reached a conclusion as to whether the Brazilian Government overall foregoes revenue otherwise due based on both scenarios.

5.203. In our view, the Panel did not "dismiss", as Brazil argues, the scenario in which the buyer of non-incentivized products is able to offset the amount of the tax paid during the same taxation period. Rather, the Panel made an express finding that, under this scenario, "there would be no revenue forgone by the Brazilian Government."⁶⁴⁶ We consider that, in comparing the challenged tax treatment to the benchmark treatment, the Panel correctly examined both possible factual scenarios that result from the application of the normal rules of Brazil's tax system.

5.204. Brazil also takes issue with the Panel's characterization of the scenario in which the buyer is not able to offset the amount of the tax paid during the same taxation period as a "rather frequent scenario".⁶⁴⁷ Brazil points out that "normally" taxpayers offset the taxes paid during the same taxation period and that the group of taxpayers that does not is "exceedingly small".⁶⁴⁸ According to Brazil "between 18.5 and 22.5% of *exporting companies* tend to accumulate credits."⁶⁴⁹ We understand credit accumulation to occur when the tax liability is very low or subject to a zero rate, so that companies that have acquired tax credits by purchasing inputs and components would not have sufficient tax debits to offset them.⁶⁵⁰

5.205. As we see it, the Panel's finding that "[i]t is fairly common that companies are not able to offset the full amount of their credits and have to request compensation or reimbursement"⁶⁵¹ was based on the evidence on the record, including that submitted by Brazil. In particular, the Panel explained that:

Brazil acknowledges that "many sectors of the economy have their products subject to low or no taxation, reflecting the selective nature and the extra-fiscal character of indirect taxes in Brazil. These sectors tend to accumulate tax credits, as the tax debits due are lower than the credit acquired in the previous step of the production chain."⁶⁵²

5.206. Indeed, before the Panel, Brazil explained that, while normally companies offset their tax credits within the same taxation period, "many sectors" of the economy "tend to accumulate" them.⁶⁵³ Brazil made this statement in the context of the PEC and RECAP programmes, which were challenged under Article 3.1(a) of the SCM Agreement. In that context, Brazil argued that certain sectors of the Brazilian economy tend to accumulate credits. However, we consider that this

⁶⁵⁰ For example, as Brazil explained before the Panel, in the context of its claims under Article 3.1(a) of the SCM Agreement, the national defence sector accumulates credits because industrial defence products and services are subject to zero PIS/COFINS rates when destined for use by the army. (Brazil's first written submission to the Panel, para. 769 (DS472))

⁶⁴⁴ Brazil's appellant's submission, para. 188.

⁶⁴⁵ Brazil's appellant's submission, para. 189. (emphasis original)

⁶⁴⁶ Panel Reports, para. 7.436.

⁶⁴⁷ Panel Reports, para. 7.432.

⁶⁴⁸ Brazil's appellant's submission, para. 192 and fn 128 thereto.

⁶⁴⁹ Brazil's appellant's submission, fn 128 to para. 192. (emphasis original)

⁶⁵¹ Panel Reports, para. 7.432.

⁶⁵² Panel Reports, fn 789 to para. 7.432. Footnote 789 continues: "Also, for instance, according to a 2014 Study by the Brazilian Industry Confederation, 'a little more than one third of the surveyed exporting companies (34.3%) possess tax credits under PIS/COFINS, IPI and/or ICMS accumulated and not refunded for a period over three months.' (22.5% of the companies possess tax credits under the PIS/COFINS contributions and 18.5% possess tax credits under the IPI tax)." See Exporters 2014 study on Brazilian exports – delays in refunds (Panel Exhibit JE-186), p. 55; European Union's first written submission, fn 849.

⁶⁵³ Brazil's first written submission to the Panel, para. 758 (DS472) and para. 691 (DS497).

- 78 -

statement is also relevant to the ICT programmes. This is because structural accumulation may occur whenever taxes on added value are lower than those on inputs or intermediate products, with the effect that the buyer would not owe enough taxes to make use of its accumulated tax credits.

5.207. We further note Brazil's argument that the Panel should have undertaken a comparison akin to the assessment of less favourable treatment under Article III: 4 of the GATT 1994. Specifically, Brazil refers to the Appellate Body Report in *EC* – *Asbestos*, in which the Appellate Body stated that, in assessing the existence of less favourable treatment, imported products as a group must be compared to the group of like domestic products.⁶⁵⁴

5.208. We do not see how this legal test, used for determining less favourable treatment under Article III:4 of the GATT 1994, could be transplanted into the analysis under Article 1.1(a)(1)(ii) of the SCM Agreement. We recall that, under Article III:4 of the GATT 1994, the less favourable treatment must affect the group of imported products, as compared to the group of like domestic products. There is an inconsistency under Article III:4 only if imported products from the complaining Member, as a group, are treated less favourably than the group of like domestic products.⁶⁵⁵

5.209. By contrast, a subsidy is always conferred upon certain recipients. The Appellate Body has observed that, in determining the existence of the revenue otherwise due that is foregone under Article 1.1(a)(1)(ii), "like will be compared with like", and that it is important to ensure that the examination under Article 1.1(a)(1)(ii) "involves a comparison of the fiscal treatment of the relevant income for taxpayers in comparable situations".⁶⁵⁶ Thus, to determine whether the revenue that is otherwise due is foregone, the challenged treatment must be compared to an objectively identifiable benchmark. This does not presuppose, however, that such a comparison should necessarily be made between the group of the entities that allegedly benefits from a subsidy, on the one hand, and the group of *all* the other entities, on the other hand. Accordingly, even if not all taxpayers in the benchmark group are paying the full amount of the relevant tax, this would not necessarily mean that there is no revenue foregone with respect to the taxpayers benefiting from a subsidy.

5.210. Thus, while both the analysis of less favourable treatment under Article III:4 and the examination under Article 1.1(a)(1)(ii) involve comparisons, this does not mean that the same analytical framework that applies to the examination of less favourable treatment under Article III:4 of the GATT 1994 is also applicable *mutatis mutandis* to the comparison of the benchmark tax treatment and the challenged tax treatment for purposes of determining whether revenue foregone under Article 1.1(a)(1)(ii) of the SCM Agreement exists.

5.211. In any event, comparing the fiscal treatment of taxpayers in comparable situations is exactly what the Panel did by comparing the treatment applicable to sales by non-accredited companies of non-incentivized products to that of the accredited companies. The fact that some non-accredited companies are able to offset their tax credits while others are not stems from the application of the normal rules of the Brazilian tax system, including the functioning of the credit-debit mechanism. We see nothing improper with the manner in which the Panel addressed both possible scenarios in its analysis.

5.4.1.5 Whether the Panel erred in finding that cash availability and implicit interest constitute revenue "otherwise due"

5.212. Brazil claims that "the Panel failed to give adequate meaning to the terms 'government revenue' and 'otherwise due' in Article 1.1(a)(1)(ii) of the SCM Agreement" by considering that the cash availability and the implicit interest on unused credits that the Brazilian Government earns in the situations when tax credits were not offset within the same taxation period fall under the scope of these terms.⁶⁵⁷ Brazil contends that "the alleged 'implicit interest' is not a form of 'government

⁶⁵⁴ Brazil's appellant's submission, para. 190 (referring to Appellate Body Report, *EC – Asbestos*, para. 100).

⁶⁵⁵ Appellate Body Report, *EC – Asbestos*, para. 100.

⁶⁵⁶ Appellate Body Report, US - FSC (Article 21.5 - EC), paras. 90 and 92.

⁶⁵⁷ Brazil's appellant's submission, para. 195.

- 79 -

revenue' that is 'due' to the Brazilian government by the taxpayer."⁶⁵⁸ In Brazil's view, due to the functioning of the credit-debit mechanism, the Brazilian Government is not due to benefit from any cash availability or any implicit interest income in relation to the time when the tax is collected.⁶⁵⁹

5.213. In response, the European Union and Japan submit that Brazil's narrow interpretation of the **"government revenue ... otherwise due [that] is foregone or not collected" should be rejected. The** European Union considers that the foregoing of the cash availability and implicit interest by the Brazilian Government is akin to tax deferrals, which constitute the revenue foregone.⁶⁶⁰ The European Union points out that "paragraph (e) in Annex I to the SCM Agreement explicitly contemplates the 'deferral' of direct taxes or social welfare charges as an illustration of what can amount to an export *subsidy*."⁶⁶¹ Japan submits that it sees "no basis" for Brazil's argument that the interest foregone by the Brazilian Government, even if implicit, "is not ... a form of current or future revenue".⁶⁶²

5.214. The United States, as a third participant, recalls that the Appellate Body stated in *US* – *Large Civil Aircraft (2nd complaint)* that "'the foregoing of revenue otherwise due implies that less revenue has been raised by the government than would have been raised in a different situation,' and that 'the word "foregone" suggests that the government has given up an entitlement to raise revenue that it could "otherwise" have raised."⁶⁶³ To the United States, "if a measure *exempts* taxes that would otherwise have to be paid but for the measure, a financial contribution has been provided: government revenue, otherwise due, is clearly foregone."⁶⁶⁴ Likewise, "if a measure *suspends* taxes that are later paid further down the production chain, a financial contribution has still been provided: at the time in which government revenue would otherwise be due, it is foregone (albeit temporarily)."⁶⁶⁵

5.215. Turning to Brazil's claim, we recall that, in conducting its comparison of the challenged treatment to the benchmark treatment, the Panel considered that, under a "rather frequent" scenario⁶⁶⁶, non-accredited companies will not be able to offset the full amount of their tax credits during the same taxation period and would have to: (i) offset them during the subsequent taxation periods; or (ii) request compensation or reimbursement from the government.⁶⁶⁷ As the Panel observed, in these circumstances, the government "will retain the amount of tax paid by the non-accredited company selling the non-incentivized intermediate product until the buyer of the non-incentivized products is able to offset it (during subsequent taxation periods); or until the Brazilian Government compensates or reimburses the face amount of the credit, without interest, to the buyer of the non-incentivized products".⁶⁶⁸

5.216. By contrast, when the challenged tax exemptions and reductions apply, the buyer of the intermediate products or raw and packaging materials, inputs, capital goods, or computational tools does not pay the IPI tax and does not obtain a tax credit, while the seller does not transmit any money to the Brazilian Government. Only after the industrialized finished or intermediate product is sold to the wholesaler will the industrial establishment transfer the amount of tax to the Brazilian Government.⁶⁶⁹

5.217. In the scenario when a non-accredited company is not able to offset its tax credits during the same taxation period, the key difference between the situation when the tax exemptions and reductions apply and the benchmark treatment is the *time* when the Brazilian Government obtains

⁶⁵⁸ Brazil's appellant's submission, para. 199. Brazil also points out that "[g]overnments are not akin to financial institutions and do not recognize 'float' interest that accrues between the time of collection and payment of taxes as a revenue against which public expenses are met." (Ibid., para. 198)

⁶⁵⁹ Brazil's appellant's submission, para. 200.

⁶⁶⁰ European Union's appellee's submission, paras. 289-291 (referring to Panel Report, US - FSC,

para. 7.98; Appellate Body Report, US - FSC, para. 90).

⁶⁶¹ European Union's appellee's submission, para. 290. (emphasis original)

⁶⁶² Japan's appellee's submission, para. 99.

⁶⁶³ United States' third participant's submission, para. 25 (quoting Appellate Body Report, *US – Large Civil Aircraft (2nd complaint)*, para. 806).

⁶⁶⁴ United States' third participant's submission, para. 25. (emphasis original)

⁶⁶⁵ United States' third participant's submission, para. 25. (emphasis original)

⁶⁶⁶ Panel Reports, para. 7.432. (fn omitted)

⁶⁶⁷ Panel Reports, paras. 7.432-7.433.

⁶⁶⁸ Panel Reports, para. 7.433.

⁶⁶⁹ Panel Reports, paras. 7.434, 7.442, 7.452, 7.461, and 7.471.

- 80 -

the tax from the seller of the product. Under the benchmark treatment, this would be at the beginning of the industrialization process, i.e. when the non-accredited seller transmits the tax to the Brazilian Government. By contrast, when the taxes are exempted or reduced, the Brazilian Government would receive the applicable amount of the tax when the industrialized product is sold to the wholesaler.

5.218. Under the credit-debit mechanism pursuant to the principle of non-accumulation, the amount of the tax paid by the non-accredited company ultimately will be offset either against its tax debits (within three taxation periods) or compensated or reimbursed by the Brazilian Government within 360 days (or later, in which case interest will apply).⁶⁷⁰ In the meantime, however, the Brazilian Government will retain the amount of the tax transmitted to it. The Panel explained, with respect to the position of the Government in these circumstances, the following:

[T]he Brazilian Government thus will hold the advantage of cash availability or cash flow, along with the associated implicit interest income (revenue) that could be generated on the full amount of that tax that it has collected from the seller. This cash availability and associated implicit interest can last from one taxation period (if the buyer is able to offset the credit during the second taxation period) to 360 days after the request for compensation or reimbursement (if the buyer has to request it). During this period, the government is able to make use of the taxes received (i.e., "earning" implicit interest). So long as the compensation or reimbursement is made to [the] buyer of the non-incentivized products within 360 days after the request, the Brazilian Government will not have to pay the buyer of the non-incentivized products any interest on the use of the buyer's money during that period. Only if 360 days are surpassed is the government obligated to compensate or repay to the buyer of the non-incentivized products not just the face amount of the credit but also the associated interest. Thus, it is only after 360 days that the government would no longer enjoy the implicit interest revenue from the free use of the buyer's money that is blocked in the form of its credit.671

5.219. Indeed, as the Panel explained, under the benchmark treatment, the Brazilian Government will collect the tax and retain the amount of the tax paid for a period that can last from one month to 360 days. During this period of time, the Brazilian Government will be able to use it and generate interest on it until the non-accredited company is able to offset its tax credits or until the Brazilian Government compensates or reimburses the amount of the tax paid. In this respect, the amount of the tax and the potential implicit interest that could be generated constitutes what is "due" to the Brazilian Government under the benchmark treatment. We note that, in its analysis, the Panel referred to the European Union's observation that the benchmark interest rate of the Central Bank of Brazil was, at the time of writing of its first written submission to the Panel, 13.25%, a rate that appears to be significant.⁶⁷² As noted, while the amount of the tax that is collected by the Brazilian Government under the challenged treatment and under the benchmark scenario may nominally be the same, the timing of the tax collection will differ. When the relevant taxes are exempted or reduced, the payment of the tax will be postponed until the final stage of the production chain, when the industrialized product is sold to the wholesaler.

5.220. A government foregoes or does not collect the "revenue that is otherwise due" in a situation where it gives up or relinquishes its entitlement to collect revenue that is owed or payable in other circumstances.⁶⁷³ In the present dispute, when the tax exemptions and reductions apply, the Brazilian Government does not collect in full the tax revenue when it normally would, or collects it in part. The fact that, ultimately, the amount of the tax collected under the benchmark treatment and the challenged treatment may nominally be the same does not detract from the fact that, under the benchmark treatment, in the scenario when non-accredited companies are unable to offset their credits immediately, the Brazilian Government would collect and retain, for a certain period, the amount of tax payable to it. During this period of time, the Brazilian Government can enjoy the cash

⁶⁷⁰ Panel Reports, para. 7.433.

⁶⁷¹ Panel Reports, para. 7.433. (fns omitted)

⁶⁷² Panel Reports, fn 790 to para. 7.433 (referring to the European Union's first written submission to the Panel, fn 883 to para. 1201).

⁶⁷³ Appellate Body Report, US – Large Civil Aircraft (2nd complaint), para. 806 (referring to Appellate Body Report, US – FSC, para. 90).

- 81 -

available to it and earn interest on it.⁶⁷⁴ By contrast, when tax exemptions and reductions are applied, the Brazilian Government collects the tax later in time and does not enjoy the availability of cash as it otherwise would under the benchmark treatment. Thus, under the challenged treatment, the Brazilian Government would not collect the tax at the time it normally would under the benchmark treatment. By doing so, in our view, the Brazilian Government would not collect the revenue that would be otherwise due to it.⁶⁷⁵

5.221. Accordingly, in the scenario where the buyer is unable to offset the credit during the same taxation period, the non-collection of the tax revenue by the Brazilian Government at the time when it normally would do so amounts to "government revenue that is otherwise due" being "foregone or not collected" within the meaning of Article 1.1(a)(1)(ii) of the SCM Agreement.

5.222. On the basis of the above, we do not consider that the Panel erred in finding that the tax treatment of intermediate products and inputs under the ICT programmes constitutes a subsidy under Article 1.1 of the SCM Agreement. We thus <u>uphold</u> the Panel's findings, in paragraphs 7.436, 7.444, 7.454, 7.463, 7.473, 7.489-7.490, and 7.495 of the Panel Reports, that each of the challenged tax exemptions, reductions, and suspensions granted to accredited companies on (i) the sales of intermediate goods that they produce, and (ii) the purchases of raw materials, intermediate goods, and packaging materials (under the Informatics programme) and inputs, capital goods, and computational goods (under the PADIS and PATVD programmes) constitutes financial contributions where "government revenue that is otherwise due is foregone or not collected" under Article 1.1(a)(1)(ii) of the SCM Agreement.

5.4.2 Article 3.1(b) the SCM Agreement - import substitution

5.4.2.1 Introduction

5.223. We now turn to consider Brazil's claim on appeal that the Panel erred in finding that the PPBs and other production-step requirements under the four ICT programmes (Informatics, PADIS, PATVD, and Digital Inclusion programmes) are contingent upon the use of domestic goods, inconsistently with Article III: 4 of the GATT 1994, and that they also constitute a contingency on the use of domestic over imported goods within the meaning of Article 3.1(b) of the SCM Agreement.⁶⁷⁶

5.224. Brazil submits that the Panel's findings under Article III:4 of the GATT 1994 imply that "whenever the requirement to perform certain manufacturing steps in Brazil as a condition to receive the subsidy involves the production of a specific input, part or component that could have been sourced from foreign producers, there would be *ipso facto*, and without further examination, discrimination within the meaning of Article III:4."⁶⁷⁷ Moreover, Brazil considers that the Panel found that the ICT programmes are inconsistent with Article 3.1(b) simply because the subsidy at issue is contingent upon production in Brazil. In this respect, Brazil contends that the Panel's reasoning is inconsistent with the Appellate Body's findings under Article 3.1(b) of the SCM Agreement in US - Tax Incentives.⁶⁷⁸ In Brazil's view, the Panel "simply presumed that because PPBs can result in the production of inputs and components and that some of the production steps can be outsourced, the subsidy granted under the ICT programmes will always be *de jure* conditioned upon the use of domestic over imported goods, despite the fact that the PPBs do not contain any language requiring, **either in explicit terms or by necessary implication ... the use of products, whether domestic or** imported".⁶⁷⁹ Brazil underscores that the PPBs and similar production-step requirements relate to

⁶⁷⁴ We recall, in this respect, the Panel's observation that, according to the European Union, the benchmark interest rate of the Central Bank of Brazil was, at the time of the first written submission to the Panel, 13.25%. (Panel Reports, fn 790 to para. 7.433)

⁶⁷⁵ Taxes are deferred when they are collected later in time than they would normally be. We note that paragraph (e) of Annex I to the SCM Agreement refers to "[t]he full or partial exemption remission, or deferral specifically related to exports, of direct taxes or social welfare charges paid or payable by industrial or commercial enterprises" as an example of what could constitute an export subsidy.

⁶⁷⁶ Brazil's appellant's submission, paras. 157 and 240.

⁶⁷⁷ Brazil's appellant's submission, para. 159.

⁶⁷⁸ In Brazil's view, the statements of the Appellate Body in that dispute "apply with equal force in the context of Article III: 4 of the GATT 1994". (Brazil's appellant's submission, para. 163)

⁶⁷⁹ Brazil's appellant's submission, para. 220.

- 82 -

the location of certain manufacturing steps in Brazil and do not prevent the possibility of using imported products.⁶⁸⁰

5.225. The European Union and Japan respond that the Panel's analysis under Article III: 4 focused on the "conditions for companies to obtain accreditation and thereby be eligible for the tax exemptions, reductions or suspensions under the programmes".⁶⁸¹ With respect to Brazil's claim that the Panel erred in finding that the tax exemptions and reductions granted under the ICT programmes constitute prohibited subsidies under Article 3.1(b) of the SCM Agreement, the European Union and Japan submit that the Panel did not base its findings on the effects of the subsidies, but on the conditions of eligibility for them, which impose the use of domestic components under the guise of production-step requirements.⁶⁸² In the European Union's view, by doing so, the Panel correctly applied the legal standard under Article 3.1(b) as elaborated by the Appellate Body in US - Tax Incentives. The European Union notes that, in US - Tax Incentives, the Appellate Body observed that "the term 'use' may also refer to incorporating a component into a separate good and that it may refer to any type of good that may be used by the subsidy recipient, including parts or components that are incorporated into another good."683 In the European Union's view, this is exactly the situation in the present case. For its part, Japan considers that what the Panel found to be problematic was not the outsourcing of the production steps, but the structure of accreditation under which PPBs and other production-step requirements condition the receipt of the subsidies on the use of domestic over imported goods under Article 3.1(b) of the SCM Agreement.⁶⁸⁴

5.226. Before addressing the issues raised on appeal, we provide a summary of the relevant Panel findings and set out the legal standard under Article 3.1(b) of the SCM Agreement.

5.4.2.2 Panel's findings

5.227. Before the Panel, the European Union and Japan raised claims under Article III: 4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement in relation to the four ICT programmes. In particular, the complainants claimed that the ICT programmes: (i) introduce regulatory discrimination against imported inputs, in the form of incentives to use domestically produced components and subassemblies in the production of finished or intermediate products, inconsistently with Article III: 4 of the GATT 1994; and (ii) constitute prohibited subsidies contingent upon the use of domestic over imported products, inconsistent with Article 3.1(b) of the SCM Agreement.⁶⁸⁵

5.228. The European Union and Japan explained before the Panel that, to receive the relevant tax benefits under the ICT programmes, manufacturers must comply with certain production-step requirements. According to the complainants, the process of complying with each required production step specified in PPBs or otherwise results in the creation of an "input" good that then must be "used" in an ensuing step or steps in the production of the incentivized finished or intermediate product.⁶⁸⁶ The complainants made this argument regardless of whether a single company itself performs all of the production steps and thus itself creates the inputs in question (the "in-house" scenario), or instead outsources some production steps to third parties, by acquiring from

⁶⁸⁰ Brazil's appellant's submission, para. 222.

⁶⁸¹ European Union's appellee's submission, para. 259 (quoting Panel Reports, para. 7.193) (emphasis omitted). The European Union notes that the Panel found that the PPBs are a set of product-specific production steps that must be performed in Brazil in order for a company to benefit from the tax incentives with respect to that product. (Ibid.) See also Japan's appellee's submission, para. 79 (referring to Panel Reports, para. 7.313).

⁶⁸² European Union's appellee's submission, para. 320; Japan's appellee's submission, paras. 103 and 108. The European Union recalls, in particular, the Panel's finding that any accredited company has to comply with production steps "in order to obtain accreditation and then [to] obtain the relevant tax benefits". (European Union's appellee's submission, para. 319 (quoting Panel Reports, para. 7.277) (emphasis omitted)) The European Union also notes that the Panel focused on the element of contingency when examining the nested PPBs. (Ibid.)

⁶⁸³ European Union's appellee's submission, para. 321 (referring to Appellate Body Report, *US – Tax Incentives*, paras. 5.8-5.9). (fns omitted)

⁶⁸⁴ Japan's appellee's submission, para. 102.

⁶⁸⁵ Panel Reports, paras. 7.2, 7.271, and 7.366.

⁶⁸⁶ Panel Reports, para. 7.271. The complainants considered that "'the requirement to perform certain manufacturing steps in Brazil [in all cases] is tantamount to requiring the incorporation of domestic content into the finished product', whenever the performance of those manufacturing steps results in the creation of a product." (Ibid., para. 7.272 (quoting Japan's first written submission to the Panel, para. 296; referring to European Union's first written submission to the Panel, para. 600))

them the outputs of those production steps, which it then incorporates as inputs into its production of the incentivized product (the "outsourcing" scenario).⁶⁸⁷

5.229. The Panel observed that a finding that the alleged requirement to use domestic goods exists would "lead *ipso facto* to the further finding of inconsistency with Article III:4 of the GATT 1994"⁶⁸⁸, and would also constitute a finding of contingency in the sense of Article 3.1(b) of the SCM Agreement.⁶⁸⁹

5.230. The Panel noted that all four ICT programmes refer to production-step requirements contained in PPBs or in other legal instruments.⁶⁹⁰ Depending on the PPB, certain production steps must be performed by the accredited company producing the incentivized finished or intermediate product at issue, while others may be performed by "third parties" based in Brazil.⁶⁹¹ The Panel examined how PPBs operate in practice in the ICT programmes, including through two illustrative examples⁶⁹², namely, the PPB for Optical Splice Closures and the one for Speed Alarms, Tracking and Control, which it considered "broadly representative of how the PPBs as a whole operate".⁶⁹³

5.231. With respect to the PPB for Optical Splice Closures, the Panel observed that Article 1 of that PPB requires that all of the relevant steps "must be undertaken in Brazil".⁶⁹⁴ The Panel observed that the envisaged production processes "consist of fundamental manufacturing from basic inputs", as opposed to simple assembly of components and subassemblies, which would not satisfy the PPB.⁶⁹⁵ Similarly, the Panel noted that all of the steps outlined in Article 1 of the PPB for Speed Alarms, Tracking and Control must be undertaken in Brazil. The Panel underscored that "these are not mere assembly operations but fundamental manufacturing processes from basic inputs."⁶⁹⁶ The Panel further observed that the PPB for Speed Alarms, Tracking and Control contains an additional element, "namely a PPB within a PPB (a so-called 'nested PPB')", according to which "at least 90% of the GSM modules used to produce any of the products for speed alarms, tracking and control listed in the Annex to the PPB must be produced in compliance with their own PPB."⁶⁹⁷ According to the Panel, a nested PPB is analytically similar to that of the production-step requirements, with the key difference from main PPBs being that there is "an implicit presumption that the products that are subject to the nested PPBs [would] be outsourced".⁶⁹⁸

5.232. The Panel noted that "in every case the PPBs allow the accredited company to outsource at least some of the required production steps to third parties, so long as those third parties themselves comply with the requirements of the PPB in respect of the steps they perform."⁶⁹⁹ Thus, a producer of a given incentivized product does not necessarily have to perform all the steps by itself in order to retain its accreditation, although it "takes responsibility for all of the production steps required in

para. 220; GATT Panel Reports, *Canada – FIRA*, paras. 5.7-5.11; *Italy – Agricultural Machinery*, para. 13). ⁶⁸⁹ Panel Reports, para. 7.259. The Panel also noted, in footnote 646 of its Reports, that, although its discussion of the alleged imposition under PPBs or other similar requirements of an obligation to use domestic inputs in the production of the incentivized products is limited to issues related to Article III:4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement, this did not mean that such discussion would not also be relevant to the claims under Article 2.1 of the TRIMs Agreement. (Ibid., fn 646 to para. 7.256)

⁶⁹⁰ Panel Reports, para. 7.279. A PPB is defined as "the minimum set of operations, in a manufacturing establishment, which characterizes the effective industrialization of [a] given product". (Ibid., para. 7.280 (referring to Decree 5,906/2006 (Panel Exhibit JE-7), Article 16))

⁶⁸⁷ Panel Reports, para. 7.271.

⁶⁸⁸ Panel Reports, para. 7.258 (referring to Appellate Body Report, US - FSC (Article 21.5 - EC),

⁶⁹¹ Panel Reports, para. 7.281.

⁶⁹² Panel Reports, paras. 7.286-7.287 and 7.289. We reproduce the text of these PPBs in paras. 5.264-5.265.

⁶⁹³ Panel Reports, para. 7.286 (referring to Interministerial Implementing Order (*Portaria*) 93 of 1 April 2013 establishing a Basic Production Process for optical splice closures produced in Brazil (Interministerial Implementing Order 93/2013) (Panel Exhibit JE-31); Interministerial Implementing Order (*Portaria*) 103 of 2 April 2013 on the Basic Production Process for products for speed alarms, tracking and control produced in Brazil (Interministerial Implementing Order 103/2013) (Panel Exhibit JE-32)).

⁶⁹⁴ Panel Reports, para. 7.288.

⁶⁹⁵ Panel Reports, para. 7.288.

⁶⁹⁶ Panel Reports, para. 7.290.

⁶⁹⁷ Panel Reports, para. 7.291. (fn omitted)

⁶⁹⁸ Panel Reports, para. 7.295. (fn omitted)

⁶⁹⁹ Panel Reports, para. 7.292. (fn omitted)

- 84 -

the respective PPB, and correspondingly ... receives the tax advantages in respect of that incentivized product".⁷⁰⁰

5.233. In examining the Informatics programme, the Panel noted that there seemed to be "no theoretical disagreement among the parties that at least in the case in which a company is required, when it acquires a product from an outside source, to only acquire a domestic product, there would be a requirement to use domestic goods in the sense covered by Article III: 4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement."⁷⁰¹ The Panel considered that, given these views of the parties, it would be "useful to separately analyse the two possible scenarios for compliance with the PPBs": the in-house scenario and the outsourcing scenario.⁷⁰²

5.234. The Panel first examined the outsourcing scenario with respect to main PPBs that contain nested PPBs, and began by recalling its earlier finding that the products that are the subject of the challenged measures are "domestic products".⁷⁰³ On this basis, the Panel found that the incentivized products produced in accordance with nested PPBs, which are used as components and subassemblies in the production of the products covered by the main PPBs, constituted Brazilian domestic products in their own right.⁷⁰⁴ The Panel further noted that, because components and subassemblies that are the subject of nested PPBs are in most cases outsourced rather than produced "in-house", the requirements that at least a certain proportion of products subject to nested PPBs comply with their respective PPBs constituted "explicit requirements to use domestic goods", in the sense of Article III: 4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement.⁷⁰⁵ Therefore, according to the Panel, whenever a producer of a product covered by a main PPB obtained components or subassemblies covered by their own nested PPB, it was obtaining "domestic goods". 706 Furthermore, according to the Panel, because nested PPBs imposed a mandatory minimum amount of such "domestic goods" to be used in producing the product subject to the main PPB, the only way to satisfy the requirement, when outsourcing, is to acquire and use domestic goods.⁷⁰⁷ Thus, the Panel concluded that "every PPB with a nested PPB inside contains an explicit requirement to use domestic goods in the cases where the goods covered by the nested PPB are outsourced."708

5.235. The Panel further observed that "[t]his analysis also holds true for the basic production step requirements of all PPBs under the Informatics programme, which typically are set forth in the PPBs' Article 1."⁷⁰⁹ The Panel elaborated that all PPBs "contain outsourcing provisions that require that outsourced production steps comply with the respective requirements of the PPBs, and in every case at least some of those outsourcing provisions apply to production steps for the creation of manufactured components or subassemblies from basic components and raw materials which, as the Panel has found, are for that reason domestic products".⁷¹⁰ The Panel considered that, similarly to nested PPBs, the outsourcing provisions under the main PPBs give rise to a requirement to use domestic goods under Article III:4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement.⁷¹¹

5.236. The Panel added that, even if certain PPBs contain "alternative options to compliance with certain production-steps in the PPBs", the "mere existence of options for compliance that are potentially WTO-consistent could not preclude a finding of inconsistency in respect of the PPBs as a whole", so that the existence of alternative, potentially WTO-consistent options, would not "alter[]

⁷⁰⁴ Panel Reports, para. 7.299.
 ⁷⁰⁵ Panel Reports, para. 7.299.
 ⁷⁰⁶ Panel Reports, para. 7.300.
 ⁷⁰⁷ Panel Reports, para. 7.300.
 ⁷⁰⁸ Panel Reports, para. 7.300.
 ⁷⁰⁹ Panel Reports, para. 7.301.
 ⁷¹⁰ Panel Reports, para. 7.301.
 ⁷¹¹ Panel Reports, para. 7.301.

⁷⁰⁰ Panel Reports, para. 7.292.

⁷⁰¹ Panel Reports, para. 7.297.

⁷⁰² Panel Reports, para. 7.298.

⁷⁰³ Panel Reports, para. 7.299 (referring to Panel Reports, para. 7.117). The Panel considered that the "manufacturing process, starting with basic materials and resulting in an intermediate or finished good to be sold on the market", which must take place in Brazil, "leaves no doubt that these products are Brazilian domestic products". (Ibid., para. 7.117) The Panel thus found that "all products produced in accordance with a PPB are, as such, Brazilian domestic products, by virtue of having been produced in Brazil from basic raw materials and other inputs through a specified mandatory manufacturing process." (Ibid., para. 7.299)

- 85 -

the inconsistency ... of an option that requires the use of domestic products over imported ${\rm products}^{\rm ".712}$

5.237. The Panel applied this reasoning to the other three ICT programmes. With respect to the PATVD programme, the Panel concluded, for the same reasons, that "the PPBs in the PATVD programme require the use of domestic goods in the sense covered by Article III: 4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement."⁷¹³ In the case of the PADIS programme, the Panel found that a company seeking accreditation had to comply with different production-step requirements depending on the type of products, and that for certain products, no PPBs have been adopted yet.⁷¹⁴ In this respect, the Panel noted that "to the extent that any future PPBs adopted under the PADIS programme contain outsourcing provisions or nested PPBs in respect of manufactured components and subassemblies that operate in the same manner as those in the Informatics and PATVD programmes, such PPBs would require the use of domestic goods."⁷¹⁵ With respect to the existing production-step requirements, the Panel concluded, for the same reasons as under the Informatics programme, that they require the use of domestic goods under Article III: 4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement.⁷¹⁶

5.238. With respect to the Digital Inclusion programme, the Panel noted that it is different from other ICT programmes in that, under this programme, "the tax benefits are in respect of sales of certain products by retailers, rather than in respect of production of certain products by producers."⁷¹⁷ Under the Digital Inclusion programme, retailers that sell in Brazil certain digital consumer goods produced in accordance with their respective PPBs qualify for certain tax benefits.⁷¹⁸ Having recalled its earlier finding that all products produced in accordance with PPBs are Brazilian domestic products, the Panel concluded that the only goods eligible for the tax benefits under the Digital Inclusion programme would be domestic goods.⁷¹⁹ Thus, for the same reasons as under other ICT programmes, the Panel found that the Digital Inclusion programme is inconsistent with Article III:4 of the GATT 1994 and that it involves a contingency on the use of domestic over imported goods within the meaning of Article 3.1(b) of the SCM Agreement.⁷²⁰

5.239. The Panel thus concluded that the PPBs and other production-step requirements of the ICT programmes "accord to imported products treatment less favourable than that accorded to like domestic products, inconsistently with Article III:4 of the GATT 1994" and that "the same aspects of these programmes give rise to contingency on the use of domestic over imported goods in the sense of Article 3.1(b) of the SCM Agreement."⁷²¹

5.4.2.3 Article 3.1(b) of the SCM Agreement

5.240. Article 3.1(b) of the SCM Agreement reads:

Except as provided in the Agreement on Agriculture, the following subsidies, within the meaning of Article 1, shall be prohibited:

...

(b) subsidies contingent, whether solely or as one of several other conditions, upon the use of domestic over imported goods.

Article 3.2 adds that "[a] Member shall neither grant nor maintain subsidies referred to in paragraph 1."

⁷¹² Panel Reports, para. 7.303.

⁷¹³ Panel Reports, para. 7.308.

⁷¹⁴ Panel Reports, paras. 7.309 and 7.312.

⁷¹⁵ Panel Reports, para. 7.312.

⁷¹⁶ Panel Reports, para. 7.311.

⁷¹⁷ Panel Reports, para. 7.316.

⁷¹⁸ Panel Reports, para. 7.315.

⁷¹⁹ Panel Reports, para. 7.316 (referring to Panel Reports, para. 7.299).

⁷²⁰ Panel Reports, para. 7.317.

⁷²¹ Panel Reports, para. 7.319.

- 86 -

5.241. Article 3.1(b) of the SCM Agreement prohibits subsidies the granting of which is "contingent ... upon the use of domestic over imported goods". The legal standard for establishing the existence of "contingency" under Article 3.1(b) is the same as under Article 3.1(a) of the SCM Agreement.⁷²² Since the ordinary meaning of "'contingent' is 'conditional' or 'dependent for its existence on something else'", a subsidy would be prohibited under Article 3.1(b) if it is "conditional" or "dependent for its existence" on the use of domestic over imported goods.⁷²³ Therefore, a subsidy would be "contingent" upon the use of domestic over imported goods where the use of those goods is a condition, in the sense of a requirement⁷²⁴, for receiving the subsidy.⁷²⁵

5.242. The Appellate Body in US - Tax Incentives noted that the term "use" in Article 3.1(b) refers to the action of using or employing something⁷²⁶ and "may, depending on the particular circumstances, refer to consuming a good in the process of manufacturing, but may also refer to, for instance, incorporating a component into a separate good, or serving as a tool in the production of a good".⁷²⁷ The Appellate Body also noted that the term "goods" in Article 3.1(b) is qualified by the adjectives "domestic" and "imported", which implies that the "goods" concerned should be at least potentially tradable.⁷²⁸ Finally, the term "over" expresses a preference between two things and, in the context of the phrase "contingent ... upon the use of domestic over imported goods", refers to the use of domestic goods in preference to, or instead of, imported goods.⁷²⁵

5.243. The term "contingency" under Article 3.1(b) covers contingency both in law and in fact. The legal standard expressed by the term "contingent" is the same for both.⁷³⁰ The Appellate Body has said that a subsidy will be *de jure* contingent upon the use of domestic over imported goods "when the existence of that condition can be demonstrated on the basis of the very words of the relevant legislation, regulation or other legal instrument constituting the measure", or can "be derived by necessary implication from the words actually used in the measure".⁷³¹ The existence of *de facto* contingency "must be *inferred* from the total configuration of the facts constituting and surrounding the granting of the subsidy, none of which on its own is likely to be decisive in any given case". 732 The Appellate Body has observed that proving *de facto* contingency "is a much more difficult task".⁷³³

5.244. Where an analysis of contingency does not yield a finding of inconsistency under Article 3.1(b) on the basis of the words actually used in the measure, or any necessary implication therefrom, the existence of a requirement to use domestic over imported goods may still be found de facto on the basis of the factual circumstances that form part of the total configuration of the

⁷²⁶ Appellate Body Report, US - Tax Incentives, para. 5.8 (referring to Appellate Body Report, US - Carbon Steel (India), para. 4.374 and fn 1009 thereto, in turn referring to Shorter Oxford English Dictionary, 6th edn, A. Stevenson (ed.) (Oxford University Press, 2007), Vol. 2, p. 3484).

⁷²² Appellate Body Report, Canada - Autos, para. 123.

⁷²³ Appellate Body Report, Canada - Autos, para. 123 (referring to Appellate Body Report,

Canada - Aircraft, para. 166).

⁷²⁴ For instance, the Appellate Body observed in *Canada – Autos* that the measure at issue in that case would be inconsistent with Article 3.1(b) if "the use of domestic goods [was] a necessity and thus ... required as a condition for eligibility" under the measure. (Appellate Body Report, Canada - Autos, para. 130 (emphasis original))

⁷²⁵ Appellate Body Report, *Canada – Autos*, para. 126. The link between "contingency" and "conditionality" is also borne out by the text of Article 3.1(b), which states that import substitution contingency can be either the sole or "one of several other conditions". (Appellate Body Report, Canada - Aircraft, para. 166 (emphasis added by the Appellate Body)) As with Article 3.1(a), this "relationship of conditionality or dependence" lies at the "very heart" of the legal standard in Article 3.1(b) of the SCM Agreement. (Appellate Body Reports, Canada - Aircraft, para. 171; Canada - Aircraft (Article 21.5 - Brazil), para. 47)

 ⁷²⁷ Appellate Body Report, US - Tax Incentives, para. 5.8.
 ⁷²⁸ Appellate Body Report, US - Tax Incentives, para. 5.9. The Appellate Body explained that this term may refer to any type of good that may be used by the subsidy recipient, including parts or components that are incorporated into another good, material, or substance that are consumed in the production process of another good, or tools or instruments that are used in the production process. (Ibid.)

Appellate Body Report, US - Tax Incentives, para. 5.11.

⁷³⁰ Appellate Body Reports, *Canada – Aircraft*, para. 167; *Canada – Autos*, para. 143.

 ⁷³¹ Appellate Body Report, *Canada – Autos*, paras. 100 and 123. (fn omitted)
 ⁷³² Appellate Body Report, *Canada – Aircraft*, para. 167. (emphasis original) Factors that may be relevant in this regard include the design and structure of the measure granting the subsidy, the modalities of operation set out in such a measure, and the relevant factual circumstances surrounding the granting of the subsidy that provide the context for understanding the measure's design, structure, and modalities of operation. (Appellate Body Report, US - Tax Incentives, para. 5.12 (referring to Appellate Body Report, EC and certain member States - Large Civil Aircraft, para. 1046))

⁷³³ Appellate Body Report, Canada - Aircraft, para. 167.

- 87 -

facts constituting and surrounding the granting of the subsidy.⁷³⁴ The Appellate Body in US - Tax*Incentives* noted that the analysis of *de jure* and *de facto* contingency under Article 3.1(b), in light of the above-mentioned factors and circumstances, should be understood as a continuum, and a panel should conduct a holistic assessment of all relevant elements and evidence on the record.⁷³⁵

5.245. Accordingly, Article 3.1(b) prohibits those subsidies that are *de jure* or *de facto* contingent such that they require the use of domestic goods in preference to, or instead of, imported goods as a condition for receiving the subsidy. While the distinction between *de jure* and *de facto* contingency lies in the "evidence [that] may be employed to prove" that a subsidy is contingent upon the use of domestic over imported goods⁷³⁶, in both its *de jure* and *de facto* analyses, a panel assesses the consistency of the granting of a subsidy under Article 3.1(b) with the same obligation and against a single legal standard of contingency.

5.246. The Appellate Body in *US* – *Tax Incentives* further observed that, insofar as, by its terms, Article 3.1(b) does not prohibit the subsidization of domestic "production" *per se*, but rather the granting of subsidies contingent upon the use of domestic over imported goods, subsidies that relate to domestic production are not, for that reason alone, prohibited under Article 3 of the SCM Agreement.⁷³⁷ In particular, such subsidies can ordinarily be expected to increase the supply of the subsidized domestic goods in the relevant market, thereby increasing the use of these goods downstream and adversely affecting imports, without necessarily requiring the use of domestic over imported goods as a condition for granting the subsidy.⁷³⁸

5.247. With regard to the relevance of Article III:8(b) of the GATT 1994, we note that this provision excepts from the national treatment obligation in Article III "the payment of subsidies exclusively to domestic producers", and thus makes clear that the provision of subsidies only to domestic producers, and not to foreign producers, does not in itself constitute a breach of Article III.⁷³⁹ The Appellate Body in *US* – *Tax Incentives* observed that, while Article III:8(b) of the GATT 1994 comports with a reading of Article 3.1(b) of the SCM Agreement under which something more than mere subsidization of domestic production is required for finding an import substitution subsidy, a subsidy excepted from the Article III national treatment obligation by virtue of it being paid exclusively to domestic producers within the meaning of Article III:8(b) of the GATT 1994 may still be found to be contingent upon the use by those producers of domestic over imported goods under Article 3.1(b) of the SCM Agreement.⁷⁴⁰

5.248. As the Appellate Body observed in US - Tax Incentives, the relevant question in determining the existence of contingency under Article 3.1(b) is not whether the eligibility requirements under a subsidy may *result* in the use of more domestic and fewer imported goods. Rather, the question is whether *a condition requiring* the use of domestic over imported goods can be discerned from the terms of the measure itself or inferred from its design, structure, modalities of operation, and the relevant factual circumstances constituting and surrounding the granting of the subsidy that provide context for understanding the operation of these factors.⁷⁴¹

⁷³⁴ Appellate Body Report, US - Tax Incentives, para. 5.13. In that report, the Appellate Body mentioned that, for instance, factual circumstances potentially relevant to an assessment of whether a subsidy is *de facto* contingent may include the existence of a multi-stage production process, the level of specialization of the subsidized inputs, or the level of integration of the production chain in the relevant industry. (Ibid., fn 49 to para_5.13)

⁷³⁵ Appellate Body Report, US - Tax Incentives, para. 5.13.

⁷³⁶ Appellate Body Report, *Canada – Aircraft (Article 21.5 – Brazil)*, fn 46 to para. 47 (quoting Appellate Body Report, *Canada – Aircraft*, para. 167).

⁷³⁷ Appellate Body Report, *US* – *Tax Incentives*, para. 5.15 (referring to Appellate Body Report, *Canada* – *Aircraft (Article 21.5* – *Brazil)*, para. 47).

⁷³⁸ Appellate Body Report, US - Tax Incentives, para. 5.15.

⁷³⁹ See Appellate Body Report, US - Tax Incentives, para. 5.16.

⁷⁴⁰ Appellate Body Report, US - Tax Incentives, para. 5.16.

⁷⁴¹ Appellate Body Report, *US* – *Tax Incentives*, para. 5.18. The Appellate Body further explained that, to the extent that no condition requiring the use of domestic over imported goods exists, but the effect of the subsidy is to displace or impede, or otherwise cause adverse effects to imports, those effects are disciplined under Part III of the SCM Agreement. (Ibid.)

- 88 -

5.4.2.4 Whether the Panel erred in finding that the PPBs and other production-step requirements under the ICT programmes are inconsistent with Article 3.1(b) of the SCM Agreement and Article III:4 of the GATT 1994

5.249. On appeal, Brazil takes issue with the Panel's finding that the PPBs or analogous production-step requirements in the ICT programmes are explicitly contingent upon the use of domestic goods, within the meaning of Article III:4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement.⁷⁴² Brazil argues that the Panel erroneously considered that "whenever the requirement to perform certain manufacturing steps in Brazil as a condition to receive the subsidy involves the production of a specific input, part or component that could have been sourced from foreign producers, there would be *ipso facto*, and without further examination, discrimination within the meaning of Article III:4."⁷⁴³ In a similar vein, with respect to the Panel's finding under Article 3.1(b), Brazil submits that the PPBs and other production-step requirements "establish the production process of specific products and leave no doubt that the tax incentives under the ICT programmes are contingent on *domestic production*"⁷⁴⁴, which, according to the Appellate Body in *US – Tax Incentives*, is "*insufficient* to establish a prohibited import substitution subsidy under Article 3.1(b) of the SCM Agreement".⁷⁴⁵ Thus, in Brazil's view, the Panel erroneously equated a condition that certain production activities take place domestically with a contingency on the use of domestic over imported goods.⁷⁴⁶

5.250. In addition, Brazil contends that the Panel made "a sweeping finding" that all ICT programmes are contingent upon the use of domestic over imported goods despite the fact that not all of these measures require compliance with production steps as a necessary condition for payment of subsidies.⁷⁴⁷ Brazil submits that this is particularly the case with respect to the PADIS and PATVD programmes, both of which allow accreditation for companies regardless of whether they comply with PPBs.⁷⁴⁸

5.251. The European Union and Japan contend that the Panel did not err in its findings under Article III:4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement. The European Union disagrees with Brazil's contention that "the Panel based its findings on the effects of the subsidies, or on the fact that accredited companies would likely use more domestic products in their downstream production activities."⁷⁴⁹ Rather, in the European Union's view, "[t]he Panel focused on the conditions of eligibility for the subsidies, and found that those conditions impose the use of domestic components and subassemblies, under the guise of production step[] requirements, in order to obtain the subsidy."⁷⁵⁰

5.252. Japan submits that, "because of the very structure of the measure involved, the requirements that covered ICT products be produced in accordance with PPBs can only be satisfied if the final products are domestic, and certain inputs are domestic as well", which "means that the **use of ... imported inputs** can disqualify a final product for eligibility under the ICT programmes".⁷⁵¹ Japan agrees with the proposition that "granting subsidies to firms so long as they engage in domestic production activities, without more, should not be equated to making those subsidies contingent on the use of domestic over imported goods and hence prohibited."⁷⁵² However, in Japan's view, in the present case, the ICT programmes are not mere production subsidies because they condition the receipt of the tax advantages upon use of domestic over imported goods through production-step requirements.⁷⁵³

⁷⁴² Panel Reports, paras. 7.313 and 7.317.

⁷⁴³ Brazil's appellant's submission, para. 159.

⁷⁴⁴ Brazil's appellant's submission, para. 221. (emphasis original)

⁷⁴⁵ Brazil's appellant's submission, para. 221 (referring to Appellate Body Report, *US – Tax Incentives*, para. 5.16). (emphasis original)

⁷⁴⁶ Brazil's opening statement at the oral hearing.

⁷⁴⁷ Brazil's opening statement at the oral hearing.

⁷⁴⁸ Brazil's opening statement at the oral hearing.

⁷⁴⁹ European Union's appellee's submission, para. 295.

⁷⁵⁰ European Union's appellee's submission, para. 295.

⁷⁵¹ Japan's appellee's submission, para. 78.

⁷⁵² Japan's appellee's submission, para. 79 (quoting Panel Report, EC and certain member

States - Large Civil Aircraft (Article 21.5 - US), para. 6.785).

⁷⁵³ Japan's appellee's submission, para. 79.

- 89 -

5.253. We recall that the Panel addressed the question whether the ICT programmes contain a requirement to use domestic goods in the section of the Panel Reports addressing the complainants' claims under Article III:4 and subsequently referred to this analysis in its examination under Article 3.1(b).⁷⁵⁴ The Panel considered that a finding that the production-step requirements listed in PPBs and other legal instruments required the use of domestic components and subassemblies in the manufacturing of incentivized products as a condition to obtain a subsidy would entail an inconsistency under Article III:4 of the GATT 1994 and would satisfy the contingency element under Article 3.1(b) of the SCM Agreement.⁷⁵⁵

5.254. We note that the legal standard under Article 3.1(b) of the SCM Agreement is not the same as that under Article III: 4 of the GATT 1994. In order to establish an inconsistency with Article 3.1(b) of the SCM Agreement, a measure must be "contingent ... upon the use of the domestic over imported goods". By contrast, to find an inconsistency with Article III: 4 of the GATT, it is sufficient that the measure at issue alters the conditions of competition to the detriment of the imported products by providing an incentive to use domestic goods. Establishing the existence of a contingency requirement to use domestic over imported products under Article 3.1(b) of the SCM Agreement is thus a more demanding standard than demonstrating that an incentive to use domestic goods exists under Article III: 4 of the GATT 1994. Accordingly, while establishing that a measure provides an incentive to producers to use domestic goods would be sufficient to find an inconsistency with Article III: 4 of the GATT 1994, it would not suffice to also find that the same measure is contingent upon the use of domestic over imported goods under Article 3.1(b) of the SCM Agreement.

5.255. Nevertheless, we consider that the Panel's reliance on its findings of inconsistency of the ICT programmes with Article III: 4 of the GATT 1994 to find the existence of the requirement to use domestic over imported goods within the meaning of Article 3.1(b) of the SCM Agreement is not necessarily, for that reason alone, incorrect. This is so if the findings under Article III: 4, on which the Panel relied in its analysis under Article 3.1(b), establish the existence of a contingency requirement to use domestic over imported goods. In other words, as long as the Panel made findings of inconsistency with Article III: 4 due to the existence of a contingency *requirement*, as opposed to a mere incentive, to use domestic goods, it could rely on these findings as a basis for its findings of inconsistency with Article 3.1(b) of the SCM Agreement. However, if we were to find that the Panel relied in its analysis under Article 3.1(b) on findings it made under Article III: 4 that merely establish the existence of an *incentive* to favour domestic over imported goods, such reliance would be incorrect.⁷⁵⁶

5.256. Brazil has set out its claims of error regarding the Panel's findings that the ICT programmes require the use of domestic components and subassemblies in the sense of Article III:4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement in separate subsections of its appellant's submission. The essence of Brazil's arguments with respect to both claims of error is, however, the same and concerns whether the Panel correctly found that the PPBs and similar production-step requirements require accredited producers to use domestic goods. Moreover, Brazil's arguments under both provisions rely preponderantly on the Appellate Body's findings in *US – Tax Incentives*. In our assessment, we will focus on whether the Panel was correct in finding that the PPBs and other production-step requirements of the four ICT programmes establish a requirement to use domestic over imported goods in the sense of Article 3.1(b) of the SCM Agreement. To the extent that they do, this would also be inconsistent with Article III:4 of the GATT 1994. If, however, we establish that the Panel made a finding of inconsistency under Article 3.1(b) of the SCM Agreement.

5.257. As noted, the European Union and Japan contend that the Panel correctly found that the PPBs and other production-step requirements explicitly require the use of domestic over imported goods.⁷⁵⁷ By contrast, Brazil considers that the Panel equated a production requirement with the requirement to use domestic over imported goods.⁷⁵⁸ All the participants refer, in their

⁷⁵⁴ Panel Reports, paras. 7.313, 7.319, and 7.386.

⁷⁵⁵ Panel Reports, paras. 7.258-7.259.

⁷⁵⁶ We recall, in this respect, that the Panel found that imported products alleged to be discriminated against are similar to the domestic products at issue. (Panel Reports, para. 7.269) This finding is not challenged on appeal.

⁷⁵⁷ European Union's appellee's submission, para. 295; Japan's appellee's submission, paras. 78-79.

⁷⁵⁸ Brazil's appellant's submission, paras. 159 and 221.

- 90 -

argumentation, to the Appellate Body Report in *US* – *Tax Incentives.* We recall that, in that report, the Appellate Body explained that both import substitution subsidies and other subsidies that relate to domestic production may have detrimental effects with respect to imported goods. Subsidies contingent upon import substitution, by their nature, adversely affect the competitive conditions of imported goods. Yet also subsidies that relate to the production of certain goods in a Member's domestic territory can ordinarily be expected to increase the supply of the subsidized goods in the relevant market, which would have the consequence of increasing the use of subsidies may foster the use of subsidized domestic goods and affect imported goods, such effects do not, in and of themselves, demonstrate the existence of a requirement to use domestic over imported goods.⁷⁵⁹ As observed, the relevant question in determining the existence of contingency under Article 3.1 (b) is not whether conditions for eligibility and access to the subsidy may *result* in the use of more domestic and fewer imported goods.⁷⁶⁰

5.258. We note that the Panel did not expressly indicate whether it conducted a *de jure* or a *de facto* analysis of inconsistency with Article 3.1(b) of the SCM Agreement.⁷⁶¹ We understand the Panel, however, to have made a *de jure* finding of inconsistency. First, the Panel relied on the text of the relevant legal instruments and its own understanding of the operation of the PPBs and other production-step requirements. The Panel did not examine the factual circumstances surrounding the granting of the subsidy. Second, the Panel stated that it was analysing "the specific provisions of the PPBs" and that the PPBs contain "an explicit requirement to use domestic goods".⁷⁶² Finally, in response to questions at the oral hearing, the participants agreed that the Panel made a finding of *de jure* inconsistency with Article 3.1(b) of the SCM Agreement with respect to the ICT programmes.⁷⁶³

5.259. In setting out its understanding of Article 3.1(b) of the SCM Agreement, the Panel noted that "a subsidy is 'contingent upon the use of domestic over imported goods', and thus, prohibited under Article 3.1(b) of the SCM Agreement, if the use of domestic goods is *required* or *necessary* in order to receive the subsidy."⁷⁶⁴ We thus consider that the Panel understood the legal standard under Article 3.1(b) correctly and sought to determine, in its analysis, whether the PPBs and other production-step requirements require the use of domestic goods. Indeed, the Panel concluded that certain aspects of the ICT programmes are inconsistent with Article 3.1(b) of the SCM Agreement precisely because, in its view, they constitute requirements to use domestic goods, and not because it considered that they condition the receipt of the subsidy upon production in Brazil.

5.4.2.4.1 Informatics programme

5.260. We start our examination of Brazil's claims of error concerning the Panel's findings under the Informatics programme by recalling the relevant aspects of the Panel's analysis that led it to reach the finding that the PPBs and other production-step requirements in the ICT programmes require the use of domestic over imported goods. We note that the Panel's description of the operation of PPBs and other production-step requirements is not challenged on appeal.

5.261. The Panel started by setting out the relevant factual aspects of "the PPBs and other production step requirements themselves".⁷⁶⁵ As noted above, the Panel explained that "a PPB is essentially a set of product-specific production steps that must be performed in Brazil, in order for a company to benefit from the tax incentives in respect of that product under the relevant

⁷⁵⁹ Appellate Body Reports, US – Tax Incentives, para. 5.49; EC and certain member States – Large Civil Aircraft (Article 21.5 – US), para. 5.65.

⁷⁶⁰ Appellate Body Reports, US – Tax Incentives, para. 5.18; EC and certain member States – Large Civil Aircraft (Article 21.5 – US), para. 5.65.

⁷⁶¹ Panel Reports, paras. 7.285 and 7.300.

⁷⁶² Panel Reports, paras. 7.285 and 7.300.

⁷⁶³ Brazil's, the European Union's, and Japan's responses to questioning at the oral hearing.

⁷⁶⁴ Panel Reports, para. 7.261. (emphasis original)

⁷⁶⁵ Panel Reports, para. 7.274.

programme(s)."⁷⁶⁶ The Panel also found that all products produced in accordance with PPBs are Brazilian domestic products.⁷⁶⁷ Brazil does not challenge this finding on appeal.⁷⁶⁸

5.262. The Panel further noted that not all production-step requirements are contained in PPBs. In particular, there are no PPBs for some of the products covered by the PADIS programme; instead there are "production-step details".⁷⁶⁹ Under the PATVD programme, the tax benefits can be obtained either by complying with the production steps in the PPBs, or by meeting the criteria for the product to be considered "developed in Brazil".⁷⁷⁰ Finally, the Panel specified that "not all PPBs or other production step requirements are relevant under all of the ICT programmes."⁷⁷¹

5.263. The Panel did not analyse in detail each of the numerous PPBs and other relevant production-step requirements under the ICT programmes in addressing the complainants' claims. Instead, the Panel summarized the details of each PPB and other production-step requirements in the Appendix to the Panel Reports⁷⁷² and, "by way of example", examined the PPB for Optical Splice Closures and the PPB for Speed Alarms, Tracking and Control, which, in the Panel's view, "are broadly representative of how the PPBs as a whole operate".⁷⁷³

5.264. The Panel reproduced the wording of the two PPBs. The PPB for Optical Splice Closures provides, in relevant part:

Article 1. Establishes the following Basic Production Process for OPTICAL SPLICE CLOSURES:

- I manufacture of the moulds for the injection of the plastic parts;
- II injection of the plastic parts;
- III stamping of the metal parts;
- IV assembly of the air valve and closure kit sub-assemblies and base items;
- ${\sf V}$ final integration of the product; and
- VI product impermeability test.

Sole Paragraph. Provided that the Basic Production Process is complied with, the activities or operations required in the production stages may be carried out by third parties in Brazil, except with regard to stages V and VI which may not be conducted by third parties.

Article 2. Whenever duly corroborated technical or economic factors so determine, any stage of the Basic Production Process may be temporarily suspended or changed through a joint Implementing Order issued by the Ministers of State for Development, Industry and Foreign Trade and Science, Technology and Innovation.⁷⁷⁴

5.265. In turn, the PPB for Speed Alarms, Tracking and Control provides, in relevant part:

Article 1. The Basic Production Process for PRODUCTS FOR SPEED ALARMS, TRACKING AND CONTROL industrialized in Brazil, as set out in the Annex to

⁷⁶⁶ Panel Reports, para. 7.280.

⁷⁶⁷ Panel Reports, paras. 7.115 and 7.299. The Panel explained that, "to qualify as an incentivized product, the production process, all of which must be carried out in Brazil, starts with basic raw materials and components unassembled and in many cases unmanufactured." (Ibid., para. 7.116) The Panel explained that, because the "manufacturing process, starting with basic materials and resulting in an intermediate or finished good to be sold on the market, must take place in Brazil", there is "no doubt that these products are Brazilian domestic products". (Ibid., para. 7.117)

⁷⁶⁸ Brazil's response to questioning at the oral hearing.

⁷⁶⁹ Panel Reports, para. 7.276.

⁷⁷⁰ Panel Reports, para. 7.282.

⁷⁷¹ Panel Reports, para. 7.283. The Panel noted that "a large number of PPBs are relevant under the Informatics programme, and a subset of these is relevant under the Digital Inclusion programme. Other PPBs are relevant, respectively, under the PADIS and PATVD programmes. In addition, only PADIS has production step requirements that are not contained in separate PPBs." (Ibid.)

⁷⁷² Panel Reports, paras. 9.1-9.323.

⁷⁷³ Panel Reports, para. 7.286.

⁷⁷⁴ Panel Reports, para. 7.287. (emphasis original)

Interministerial Implementing Order MDIC/MCT No 14 of 22 January 2007, shall now read as follows:

I - stamping, cutting, folding and surface treatment of metal parts, when applicable;

II - injection of the housing's plastic parts, when applicable;

III - manufacture of the printed circuits from laminate;

IV - assembly and soldering, or equivalent process, of all components on the printed circuit boards;

V - assembly of the electrical and mechanical parts, totally separated, at a basic component level; and

VI - integration of the printed circuit boards and the electrical and mechanical parts in the formation of the final product, in accordance with indents I to V above.

§ 1 Provided that the Basic Production Process is complied with, the activities or operations required in the production stages may be carried out by third parties, except with regard to the stage set out in indent VI which may not be conducted by third parties.

§ 2 Liquid crystal, plasma and other display technologies are temporarily exempted from assembly.

§ 3 The following assembled modules and sub-assemblies are temporarily exempted from compliance with the stages set out in indents III and IV, of the header paragraph to this Article:

I - Frequency Modulation (FM) communication modules;

- II Pager communication modules;
- III Global Positioning System (GPS) communication modules;
- IV satellite communication modules;
- V thermal printer mechanisms; and
- VI Code Division Multiple Access (CDMA) communication modules.

Article 2. 90% of the total number of Global System for Mobile Communications (GSM) communication modules used in the production of PRODUCTS FOR SPEED ALARMS, TRACKING AND CONTROL as set out in the Annex to this Implementing Order shall be produced in accordance with their respective Basic Production Process in the calendar year.⁷⁷⁵

5.266. With respect to both PPBs, the Panel noted that all of the production steps outlined in Article 1 must be undertaken in Brazil. According to the Panel, these steps are not mere assembly operations, but "fundamental manufacturing" processes from basic inputs, and a product produced through simple assembly of components and subassemblies would not satisfy the conditions of the PPBs.⁷⁷⁶ We have reservations with respect to the Panel's unqualified statement that all the steps outlined in Article 1 of these PPBs are not assembly operations, but fundamental manufacturing processes. It is not clear whether certain of the steps listed in these PPBs constitute manufacturing or assembly operations (for example, the injection of the plastic parts in the process of manufacturing the optical splice closures, and the injection of the housing's plastic parts and assembly and soldering, or equivalent process, of all components on the printed circuit boards in the production of speed alarms, tracking and control products). More generally, we have reservations as to whether a clear line can be drawn between manufacturing and assembly, and as to its relevance to the question whether a measure requires the use of domestic over imported goods.

5.267. Notwithstanding, we recall that, in order to find a *de jure* inconsistency with Article 3.1(b) "*a condition requiring* the use of domestic over imported goods [must] be discerned from the terms of the measure itself", or by necessary implication therefrom.⁷⁷⁷ Thus, for purposes of analysis, the distinction between manufacturing and assembly, while instructive, is not determinative. What

⁷⁷⁵ Panel Reports, para. 7.289. (emphasis original)

⁷⁷⁶ Panel Reports, paras. 7.288 and 7.290.

⁷⁷⁷ Appellate Body Report, US – Tax Incentives, para. 5.18. (emphasis original)

- 93 -

matters, instead, is whether the measure reflects a condition requiring the use of domestic over imported goods.

5.268. The Panel also noted that both PPBs allow for outsourcing of some production steps to the third parties.⁷⁷⁸ However, unlike the PPB for Optical Splice Closures, the PPB for Speed Alarms, Tracking and Control contains an additional feature, "a PPB within a PPB", which the Panel termed a "nested PPB".⁷⁷⁹ Specifically, Article 2 of that PPB provides that at least 90% of the GSM modules used to produce any of the products for speed alarms, tracking and control listed in the Annex to the PPB must be produced in compliance with their own PPB. The Panel explained that this means that a separate PPB exists for GSM modules, and that this separate PPB must be complied with for at least 90% of the GSM modules used in production of speed alarms, tracking and control products listed in the Annex.⁷⁸⁰

5.269. In its analysis in the context of the Informatics programme, the Panel separately addressed the case of the "main" PPBs that contain nested PPBs and the case of basic production-step requirements for all PPBs. With respect to the nested PPBs, the Panel first noted that all products produced in accordance with PPBs, including those that are produced in accordance with nested PPBs, are domestic products. Moreover, the Panel recalled that at least some proportion of the components and subassemblies covered by the nested PPBs must be produced in accordance with those nested PPBs. Finally, the Panel recalled that, in most cases, the components and subassemblies subject to the nested PPBs will be outsourced.⁷⁸¹ On this basis, the Panel arrived at the conclusion that, "where products subject to nested PPBs are outsourced, the requirements that (at least a certain proportion of) such products comply with their respective PPBs thus constitute explicit requirements to use domestic goods – the components and subassemblies covered by the nested PPBs – in the sense covered by Article III:4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement."⁷⁸² The Panel then elaborated that:

[W]henever a producer of a product covered by a main PPB obtains components or subassemblies covered by their own nested PPB it is obtaining domestic goods. Because the nested PPB imposes a mandatory minimum amount of such domestic goods to be used in producing the product subject to the main PPB, the only way to satisfy this requirement, when outsourcing, is by acquiring and using domestic goods. Thus, every PPB with a nested PPB inside contains an explicit requirement to use domestic goods in the cases where the goods covered by the nested PPB are outsourced.⁷⁸³

According to the Panel, the same analysis holds true for the basic production-step requirements of all PPBs, all of which contain outsourcing provisions. The Panel concluded that "the outsourcing of production steps for the manufacture in Brazil of components and subassemblies from basic components and raw materials gives rise to a requirement to use domestic goods in the sense covered by Article III: 4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement."⁷⁸⁴

5.270. We recall that PPBs are defined as a "minimum set of operations performed at a manufacturing facility that characterizes the *actual industrialization of a given product*".⁷⁸⁵ The Panel characterized the PPBs as "a set of product-specific production steps that must be performed in Brazil, in order for a company to benefit from the tax incentives".⁷⁸⁶ In other words, to obtain tax incentives under the relevant ICT programmes, a company has to manufacture a given product in accordance with the requirements of the PPBs or other production-step requirements.

- ⁷⁸³ Panel Reports, para. 7.300.
- ⁷⁸⁴ Panel Reports, para. 7.301.
- ⁷⁸⁵ Panel Reports, para. 2.62. (emphasis added)
- ⁷⁸⁶ Panel Reports, para. 7.280.

⁷⁷⁸ The PPB for Optical Splice Closures allows an accredited producer to outsource steps I-IV, i.e. manufacture the injection moulds, inject the plastic parts, stamp the metal parts, and assemble the air valve and closure kit subassemblies and base items. Under PPB for Speed Alarms, Tracking and Control, the accredited producer can outsource all of the production steps other than the final step of integrating the printed circuit boards, and all of the electrical and mechanical parts to form the final product. (Panel Reports, para. 7.293)

⁷⁷⁹ Panel Reports, para. 7.291. (fn omitted)

⁷⁸⁰ Panel Reports, para. 7.291.

⁷⁸¹ Panel Reports, para. 7.299.

⁷⁸² Panel Reports, para. 7.299.

- 94 -

5.271. The Panel started its analysis by examining the "nested" PPBs, i.e. those PPBs that contain an additional PPB within a main PPB. As the Panel observed, "the main PPBs that contain nested PPBs require that at least some minimum proportion of the components and subassemblies of the type covered by the nested PPBs must have been produced in accordance with those nested PPBs."⁷⁸⁷ We understand the Panel to have seen "nested" PPBs as an explicit manifestation of the requirement to use domestic goods over imported products. For example, Article 2 of the PPB for Speed Alarms, Tracking and Control requires that "90% of the total number of Global System for Mobile Communications (GSM) communication modules used in the production of PRODUCTS FOR SPEED ALARMS, TRACKING AND CONTROL ... shall be produced in accordance with their respective [PPB]."⁷⁸⁸ The PPB for GSM communication modules is thus "nested" into the main PPB for Speed Alarms, Tracking and Control.

5.272. The main PPB mandates that 90% of the GSM communication modules used in the production of products for speed alarms, tracking and control be produced in accordance with their own (nested) PPB. Pursuant to the Panel's findings that are not contested on appeal, goods produced in accordance with PPBs are Brazilian domestic goods.⁷⁸⁹ The GSM communication modules manufactured in line with their respective PPBs will thus be Brazilian domestic products. Accordingly, it follows that 90% of the GSM communication modules used in the production of products for speed alarms, tracking and control "shall be"⁷⁹⁰ Brazilian domestic goods. Thus, by requiring that 90% of the GSM communication modules used in the production of products for speed alarms, tracking and control be produced in accordance with their respective PPBs, the PPB at issue effectively requires that 90% of GSM communication modules used in the production of speed alarms, tracking and control be of domestic origin.

5.273. We recall that a subsidy is *de jure* contingent upon the use of domestic over imported goods when the existence of that condition can be demonstrated on the basis of the very words of the measure, or can be derived by necessary implication from them.⁷⁹¹ The above-mentioned example demonstrates that, the nested PPBs within the main PPBs, by their very words, or at least by necessary implication therefrom, require that a certain percentage of inputs used in the production steps performed in accordance with the main PPB be sourced domestically. Thus, the use of domestic components and subassemblies, for which there is a nested PPB, in the production of the products covered by the main PPB will not be merely incidental.⁷⁹² Rather, it is a condition that must be fulfilled in order for the relevant product to benefit from the tax incentives under the ICT programmes.

5.274. The main PPBs that incorporate a nested PPB thus contain a condition requiring the use of domestic over imported goods in the sense of Article 3.1(b) of the SCM Agreement that stems from the very text of the nested PPBs, or at least by necessary implication therefrom. As a consequence of that condition, the nested PPBs also provide less favourable treatment to imported goods than to like domestic products within the meaning of Article III: 4 of the GATT 1994.

5.275. We thus agree with the Panel's conclusion that the main PPBs that incorporate nested PPBs contain a requirement to use domestic goods, i.e. the components and subassemblies covered by the nested PPBs, in the sense of Article 3.1(b) of the SCM Agreement and Article III:4 of the GATT 1994.⁷⁹³ Accordingly, we uphold the Panel's findings, in paragraphs 7.299-7.300, 7.302, 7.313, 7.319, 8.5.b, 8.5.e, 8.16.c, and 8.16.f of the Panel Reports, that the main PPBs that incorporate nested PPBs under the Informatics programme are inconsistent with Article 3.1(b) of the SCM Agreement and Article III: 4 of the GATT 1994.

⁷⁸⁸ Panel Reports, para. 7.289 (quoting PPB for Products for Speed Alarms, Tracking and Control).

- ⁷⁹⁰ Panel Reports, para. 7.289 (quoting PPB for Products for Speed Alarms, Tracking and Control).

⁷⁸⁷ Panel Reports, para. 7.299.

⁷⁸⁹ Panel Reports, paras. 7.117 and 7.299.

 ⁷⁹¹ Appellate Body Report, *Canada – Autos*, paras. 100 and 123.
 ⁷⁹² As the Appellate Body observed in *US – Tax Incentives*, production subsidies "can ordinarily be expected to increase the supply of the subsidized domestic goods in the relevant market, thereby increasing the use of these goods downstream and adversely affecting imports". This, however, does not necessarily indicate the existence of a requirement to use domestic over imported goods as a condition for granting the subsidy. (Appellate Body Report, *US – Tax Incentives*, para. 5.15) ⁷⁹³ Panel Reports, para. 7.299.

- 95 -

5.276. We now turn to examine the main PPBs that do not contain nested PPBs.⁷⁹⁴ We recall that PPBs are defined as the "minimum set of operations, in a manufacturing establishment, which characterizes the effective industrialization of [a] given product⁷⁹⁵ The PPBs prescribe "a set of product-specific production steps that must be performed in Brazil, in order for a company to benefit from the tax incentives in respect of that product under the relevant programme(s)".⁷⁹⁶ In its analysis, the Panel observed that all PPBs "contain outsourcing provisions that require that outsourced production steps comply with the respective requirements of the PPBs, and in every case at least some of those outsourcing provisions apply to production steps for the creation of manufactured components or subassemblies from basic components and raw materials which, as the Panel has found, are for that reason domestic products".⁷⁹⁷ The Panel considered that "the outsourcing of production steps for the manufacture in Brazil of components and subassemblies from basic components and raw materials gives rise to a requirement to use domestic goods."⁷⁹⁸ Thus, to the Panel, the possibility to outsource the production of components and subassemblies in Brazil under the PPBs amounted to a requirement to use domestic over imported goods under Article 3.1(b) of the SCM Agreement.

5.277. We recall that a subsidy is said to be *de jure* contingent upon the use of domestic over imported goods "when the existence of that condition can be demonstrated on the basis of the very words of the relevant legislation, regulation or other legal instrument constituting the measure".⁷⁹⁹ A requirement to use domestic over imported products that is not set out expressly in the relevant legislation may nevertheless be derived by necessary implication if it results inevitably from the words actually used in the legislation, or if any other interpretation would be unreasonable.⁸⁰⁰ In addition, while production subsidies "can ordinarily be expected to increase the supply of the subsidized domestic goods in the relevant market, thereby increasing the use of these goods downstream and adversely affecting imports", this does not necessarily indicate the existence of a requirement to use of domestic over imported goods as a condition for granting the subsidy.⁸⁰¹

5.278. The Panel considered that compliance with the PPBs is possible by either outsourcing the performance of production steps to a third party in Brazil or by performing such production steps in-house in Brazil.⁸⁰² In the Panel's view, the possibility to outsource the production of components and subassemblies in Brazil under the PPBs "g[ave] rise to a requirement to use domestic goods"⁸⁰³ under Article 3.1(b). We disagree with the Panel that the mere possibility of outsourcing under PPBs of production steps to be performed by a third party in Brazil, in and of itself, gives rise to a requirement to use domestic over imported goods under Article 3.1(b) of the SCM Agreement. Instead, the Panel should have explored, when establishing whether there was *de jure* inconsistency with Article 3.1(b), how the text of the PPBs gives rise to a requirement to use domestic over imported goods or how such a requirement can be derived from the text of the PPBs by necessary implication.

5.279. Moreover, in reaching its conclusion, the Panel observed that, "when an accredited company outsources components and subassemblies it can only comply with the PPB in question if the production step requirements for those components and subassemblies are respected, and respecting those requirements in turn means that those components and subassemblies must be

 $^{^{794}}$ We note that the Panel refers to them also as "basic production step requirements". (Panel Reports, para. 7.301)

⁷⁹⁵ Panel Reports, para. 7.280 (referring to Decree 5,906/2006 (Panel Exhibit JE-7), Article 16).

⁷⁹⁶ Panel Reports, para. 7.280.

⁷⁹⁷ Panel Reports, para. 7.301.

⁷⁹⁸ Panel Reports, para. 7.301.

⁷⁹⁹ Appellate Body Report, *Canada – Autos*, para. 100.

⁸⁰⁰ The panel in US - Tax Incentives explained that, for the purpose of the *de jure* determination, including by necessary implication, the relevant facts are therefore the text of the legislation at issue and any additional facts that can assist the Panel in understanding the meaning of the terms as used in that legislation. For example, such facts could include relevant context within the legislation itself, or other clarifications of legal meaning within the domestic legal system in question (such as the interpretation of pertinent terms by domestic courts, or administrative regulations directly implementing the legislation). However, any such facts would serve to illuminate the meaning of words used in the legislation and would not extend to other factual circumstances surrounding the granting of the subsidy that provide the context for understanding the measure's design, structure, and modalities of operation. (Panel Report, *US – Tax Incentives*, para. 7.273 and fn 556 thereto)

⁸⁰¹ Appellate Body Report, US – Tax Incentives, para. 5.15.

⁸⁰² Panel Reports, para. 7.298.

⁸⁰³ Panel Reports, para. 7.301.

- 96 -

Brazilian domestic goods."⁸⁰⁴ The focus of this statement is on the domestic origin of the goods produced in accordance with production-step requirements under the PPBs. Article 3.1(b) of the SCM Agreement indeed prohibits subsidies contingent "upon the use of *domestic* over imported goods".⁸⁰⁵ Brazil has not challenged the Panel's finding that components and subassemblies produced in accordance with PPBs will be Brazilian domestic goods.⁸⁰⁶ However, the mere fact that goods produced in accordance with the production-step requirements under the PPBs will be domestic does not indicate the existence of a condition requiring the use of domestic over imported goods. We recall, in this respect, that under the Siting Provisions, which were part of the measure at issue in *US – Tax Incentives*, it was uncontested, but not determinative, that the wings and fuselages that had to be produced in Washington State, and the planes that had to be assembled in Washington State, were of US origin because of the location where these production and assembly steps took place. We further emphasize, as the Appellate Body explained in *US – Tax Incentives*, that the relevant question in determining the existence of a contingency under Article 3.1(b) is "whether a condition requiring the use of domestic over imported goods can be discerned" from the measure.⁸⁰⁷

5.280. As noted, the PPBs are defined as "a minimum set of operations, in a manufacturing establishment, which characterizes the effective industrialization of a given product".⁸⁰⁸ It follows that, to characterize the actual industrialization of a given product and to qualify for the tax incentives, the steps outlined in the PPBs, such as manufacture of components and subassemblies, product testing, and final integration, must be performed in Brazil. For instance, the PPB for Optical Splice Closures provides that the following production steps be performed in Brazil: (i) manufacture of the moulds for the injection of the plastic parts; (ii) injection of the plastic parts; (iii) stamping of the metal parts; (iv) assembly of the air valve and closure kit subassemblies and base items; (v) final integration of the product; and (vi) product impermeability test.⁸⁰⁹ Similarly, steps I-V of the PPB for Speed Alarms, Tracking and Control PPB require performance in Brazil of certain production steps, in particular: (i) stamping, cutting, folding and surface treatment of metal parts; (ii) injection of the housing's plastic parts; (iii) manufacture of the printed circuits from laminate; (iv) assembly and soldering, or equivalent process, of all components on the printed circuit boards; and (v) assembly of the electrical and mechanical parts, totally separated, at a basic component level. Production step VI of this PPB further provides for "integration of the printed circuit boards and the electrical and mechanical parts in the formation of the final product" in accordance with steps I-V of that PPB.810

5.281. The PPBs thus set out a number of sequential production steps, starting from manufacturing of components and subassemblies and ending with the final assembly and testing of the product. As the Panel put it, "[i]n terms of their content, all of the PPBs consist of a collection of production steps, most of which involve the conversion through a manufacturing process of basic raw materials in completely disaggregated form into a component or set of components or subassemblies, with the final step or steps involving the integration of all of these components or subassemblies into the final product along with, in some cases, testing and other final steps."811 The structure of the PPBs suggests that the subsidy recipients will likely "use" in a subsequent production step the domestic components and subassemblies that were manufactured in a previous production step. For example, production step I under the PPB for Optical Splice Closures provides for the "manufacture of the moulds for the injection of the plastic parts", while production step II envisages the "injection of the plastic parts".⁸¹² Thus, the moulds produced for the injection of the plastic parts in accordance with the first production step are likely to be used in the injection of the plastic parts under the second production step. Subsequently, the moulds and the components and subassemblies resulting from the ensuing production steps are likely to be incorporated into the incentivized product as part of final integration under production step V. However, while such use of domestic goods may be a likely consequence of the eligibility requirements for the tax incentives under the Informatics programme,

⁸⁰⁷ Appellate Body Report, US - Tax Incentives, para. 5.18. (emphasis original)

⁸⁰⁹ Panel Reports, para. 7.287 (quoting Interministerial Implementing Order 93/2013 (Panel Exhibit JE-31)).

⁸¹⁰ Panel Reports, para. 7.289 (quoting PPB for Products for Speed Alarms, Tracking and Control).

⁸¹¹ Panel Reports, para. 7.285.

⁸¹² Panel Reports, para. 7.287 (quoting Interministerial Implementing Order 93/2013 (Panel Exhibit JE-31), Article 1).

⁸⁰⁴ Panel Reports, para. 7.301.

⁸⁰⁵ Emphasis added.

⁸⁰⁶ Brazil's response to questioning at the oral hearing.

⁸⁰⁸ Panel Reports, para. 7.280 (quoting Decree 5,906/2006 (Panel Exhibit JE-7), Article 16).

- 97 -

this does not, in and of itself, indicate the existence of a condition requiring the use of domestic over imported products.

5.282. Article 3.1(b) does not prohibit *per se* conditioning eligibility for tax incentives on conducting certain production, processing, or assembly steps domestically.⁸¹³ Inherent effects of production subsidies are not sufficient for a finding of contingency upon import substitution.⁸¹⁴ Instead, as the Appellate Body explained in *US* – *Tax Incentives*, the key question for a measure to be found *de jure* inconsistent with Article 3.1(b) of the SCM Agreement is whether "*a condition requiring* the use of domestic over imported goods" can be discerned from its very words or by necessary implication therefrom.⁸¹⁵

5.283. In this respect, we do not consider that the main PPBs that do not contain nested PPBs provide for more than "a collection of production steps", which must be carried out in Brazil, in order for a company to benefit from the tax incentives with respect to the product subject to the PPB under the relevant programme.⁸¹⁶ Although compliance with the production steps set out in the PPBs is likely to result in the use of domestic components and subassemblies, this is not sufficient for a *de jure* finding of inconsistency to be made under Article 3.1(b) of the SCM Agreement.⁸¹⁷ Such use of domestic products will be a consequence of the requirement to perform the production steps in Brazil, but this does not mean that a contingency requirement not to use imported products can be discerned from the wording of the PPBs or by necessary implication therefrom.

5.284. The Panel also found that the main PPBs that do not contain nested PPBs are inconsistent with Article III: 4 of the GATT 1994 because it considered that they give rise to a requirement to use domestic over imported goods. We recall that, in an inquiry under Article III: 4 of the GATT 1994, **"what is relevant is whether ... regulatory differences distort the conditions of competition to the** detriment of imported products".⁸¹⁸ Thus, an examination of whether a measure involves treatment less favourable "must be founded on a careful analysis of the contested measure and of its implications in the marketplace".⁸¹⁹ As noted, while the existence of an incentive to use domestic over imported goods will not be sufficient to establish an inconsistency with Article 3.1(b) of the SCM Agreement, it will suffice to find less favourable treatment under Article III: 4 of the GATT 1994. We recall that compliance with the PPBs is mandatory in order for a company to receive tax benefits under the ICT programmes.⁸²⁰ We also recall that the Panel found that "the products alleged by the complaining parties to be disadvantaged are *like* the products that are allegedly favoured."⁸²¹ As we

⁸¹³ As the Appellate Body noted, in agreeing with the panel in *US* – *Tax Incentives*, "while the terms of the First Siting Provision could result in the use by Boeing of some wings and fuselages produced in Washington, this did not necessarily mean that the provision, by its terms, requires Boeing to use domestic over imported wings and fuselages." (Appellate Body Report, *US* – *Tax Incentives*, para. 5.27 (fn omitted)) The panel in *US* – *Tax Incentives* also concluded that "[t]he contingency on siting certain production activities within the state of Washington [under the First Siting Provision] does not entail any explicit, or any necessarily implied, requirement to use domestic goods", and that "[n]o express or obvious contingency results from the terms used in the [Second Siting Provision], nor can one be derived inevitably from its terms." (Panel Report, *US* – *Tax Incentives*, paras. 7.296 and 7.310)

⁸¹⁴ In this vein, the Appellate Body concurred in *US* – *Tax Incentives* with "the panel's conclusions that the First and Second Siting Provisions are not *de jure* contingent under Article 3.1(b) were based on [the Panel's] findings that: (i) the contingencies set out in the terms of these provisions relate to the location of certain assembly operations within Washington; (ii) the provisions are silent as to the use of domestic or imported goods; and (iii) the terms of the provisions do not, by necessary implication, prevent the possibility of using imported goods." (Appellate Body Report, *US* – *Tax Incentives*, para. 5.30)

⁸¹⁵ Appellate Body Report, *US – Tax Incentives*, para. 5.18. (emphasis original) Indeed, in *US – Tax Incentives*, in the context of its *de jure* analysis of the First Siting Provision, the Appellate Body found that, even if Boeing was "likely [to] use some amount of domestically produced wings and fuselages, this observation [was] not in itself sufficient to establish the existence of a condition, reflected in the measure's terms or arising by necessary implication therefrom, requiring the use of domestic over imported goods". (Ibid., para. 5.40)

⁸¹⁶ Panel Reports, para. 7.285.

⁸¹⁷ As discussed above, we note that, in response to questioning at the oral hearing, the participants agreed that the Panel's findings under the ICT programmes were of a *de jure* nature. (Brazil's, the European Union's, and Japan's responses to questioning at the oral hearing)

⁸¹⁸ Appellate Body Report, *Thailand – Cigarettes (Philippines)*, para. 128.

⁸¹⁹ Appellate Body Report, *US* – *FSC (Article 21.5 – EC)*, para. 215. The Appellate Body, however, cautioned that "the examination need not be based on the *actual effects* of the contested measure in the marketplace". (Ibid. (emphasis original))

⁸²⁰ Panel Reports, para. 7.280.

⁸²¹ Panel Reports, para. 7.269. (emphasis original)

- 98 -

observed above, given the structure of the PPBs, which comprises a number of sequential production steps, it is likely that components and subassemblies produced in compliance with PPBs will be used as inputs in the subsequent production steps. Accordingly, given that compliance with the PPBs is mandatory in order for a company to qualify for the tax incentives and that, in complying with the PPBs, the producers of an incentivized product will be likely to use domestic components and subassemblies, we consider that the main PPBs without nested PPBs provide an incentive to use domestic over imported goods. By doing so, the main PPBs in the Informatics programme accord treatment less favourable to imported goods than that accorded to like domestic goods inconsistently with Article III: 4 of the GATT 1994. We thus agree with the Panel, albeit for different reasons, that the main PPBs under the Informatics programme are inconsistent with Article III: 4 of the GATT 1994.

5.285. In light of the above, with respect to the main PPBs that do not incorporate nested PPBs, we consider that the Panel erred in finding that, "under the Informatics programme, regarding the outsourcing requirements in respect of the production step requirements for components and subassemblies used in the production of an incentivized product, the PPBs require the use of domestic goods."⁸²² Accordingly, we <u>reverse</u> the Panel's findings, in paragraphs 7.301-7.302, 7.313, 7.319, 8.5.e, and 8.16.f of the Panel Reports, that the main PPBs without nested PPBs under the Informatics programme are contingent upon the use of domestic over imported goods within the meaning of Article 3.1(b) of the SCM Agreement. However, we <u>uphold</u>, albeit for different reasons, the Panel's findings, in paragraphs 7.301-7.302, 7.313, 7.319, 8.5.b, and 8.16.c of the Panel Reports, that the main PPBs that do not incorporate nested PPBs under the Informatics programme are inconsistent with Article III: 4 of the GATT 1994.

5.4.2.4.2 PADIS, PATVD, and Digital Inclusion programmes

5.286. Brazil further argues that the Panel made "a sweeping finding" that all ICT programmes are contingent upon the use of domestic over imported goods despite the fact that the PADIS and PATVD programmes allow accreditation regardless of whether companies comply with PPBs.⁸²³ We recall that, in order to obtain accreditation and the relevant tax benefits under the PATVD programme, a company must either comply with the relevant PPBs or meet the criteria for the product to be "developed in Brazil".⁸²⁴ We understand Brazil to take issue with the fact that the Panel made a finding of inconsistency with respect to the PPBs operating under the PATVD programme despite the fact that complying with PPBs is only one of the options for obtaining accreditation under the PATVD programme by virtue of meeting the criteria for a product to be "developed in Brazil" as "a potentially WTO-consistent alternative option for compliance", which, in its view, did not alter its conclusion that the PPBs in the PATVD programme required the use of domestic goods under Article 111:4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement.⁸²⁵

5.287. Indeed, the Panel found that "*the PPBs in the PATVD programme* require the use of domestic goods in the sense covered by Article III:4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement."⁸²⁶ Accordingly, we understand the Panel to have made findings of inconsistency only with respect to the PPBs, in the scenario when a company obtains accreditation by virtue of complying with the relevant PPB, and not with respect to the criteria for a product to be "developed in Brazil". The Panel found that the PPBs in the PATVD programme require the use of domestic goods under Article III:4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement "[f]or the same reasons [as it gave] in respect of the Informatics programme".⁸²⁷

5.288. We understand that the PPBs under the PATVD programme follow the same structure and logic as those under the Informatics programme.⁸²⁸ Above, we have agreed with the Panel's

⁸²² Panel Reports, para. 7.302.

⁸²³ Brazil's opening statement at the oral hearing.

⁸²⁴ Panel Reports, para. 7.307. (fn omitted)

⁸²⁵ Panel Reports, para. 7.308.

⁸²⁶ Panel Reports, para. 7.308. (emphasis added)

⁸²⁷ Panel Reports, para. 7.308.

⁸²⁸ As the Panel observed, "all of the PPBs consist of a collection of production steps, most of which involve the conversion through a manufacturing process of basic raw materials in completely disaggregated form into a component or set of components or subassemblies, with the final step or steps involving the integration of all of these components or subassemblies into the final product along with, in some cases, testing and other final steps." (Panel Reports, para. 7.285)

- 99 -

conclusion that the main PPBs that incorporate nested PPBs contain "explicit requirements to use domestic goods", i.e. the components and subassemblies covered by the nested PPBs, under Article 3.1(b) of the SCM Agreement and Article III: 4 of the GATT 1994. For the same reasons, to the extent the PPBs under the PATVD programme incorporate nested PPBs, we consider them to require *de jure* the use of domestic over imported goods inconsistently with Article 3.1(b) of the SCM Agreement and Article III: 4 of the GATT 1994. With respect to the main PPBs that do not incorporate nested PPBs, however, we have reversed the Panel finding that they reflect a condition requiring the use of domestic over imported goods under Article 3.1(b). This is so because, even if compliance with the production steps set out in the PPBs is likely to result in the use of more domestic components and subassemblies, this is not sufficient for a finding of inconsistency under Article 3.1(b) of the SCM Agreement.

5.289. We thus <u>uphold</u> the Panel's findings of inconsistency under Article 3.1(b) of the SCM Agreement concerning the PATVD programme, in paragraphs 7.308, 7.313, 7.317, 7.319, 8.5.e, and 8.16.f of the Panel Reports, to the extent they relate to the main PPBs that contain nested PPBs, and we <u>reverse</u> the Panel's findings, contained in the same paragraphs of the Panel Reports, to the extent they relate to the main PPBs.

5.290. Under the PADIS programme accredited companies are exempted from paying certain taxes with respect to semiconductors and information displays, as well as to inputs, tools, equipment, machinery, and software for the production of semiconductors and displays.⁸²⁹ In order to become accredited under the PADIS programme, a company must, in particular, engage in certain manufacturing activities in Brazil that depend on the products for which a company is seeking tax benefits.⁸³⁰ Specifically, with respect to semiconductor electronic devices, a company must perform: (i) concept, development, and design; (ii) diffusion or physical-chemical processing; or (iii) cutting, encapsulation, and testing.⁸³¹ With respect to information displays, a company must perform: (i) concept, development, and design; (ii) manufacture of photosensitive, photo, or electroluminescent elements and light-emitting diodes; or (iii) final assembly of displays and electrical and optical testing.⁸³² Finally, with respect to inputs and equipment intended for the manufacture of electronic semiconductor devices and information displays, the company must manufacture those products in accordance with the relevant PPBs.⁸³³

5.291. We note that, in order to be accredited with respect to semiconductor electronic devices or information displays, it is sufficient for a company to comply with one of the listed production-step requirements.⁸³⁴ For example, in order to receive tax benefits for manufacturing semiconductor electronic devices, a company must perform either concept, development, and design; diffusion or physical-chemical processing; or cutting, encapsulation, and testing in Brazil. While the company could, in principle, engage in all of these activities and be accredited under the PADIS programme,

⁸³² Panel Reports, para. 2.76. (referring to Law 11,484/2007 (Panel Exhibit JE-71), Article 2(II); Decree 6,233/2007 (Panel Exhibit JE-73), Article 6(II), as amended by Decree 8,247/2014 (Panel Exhibit JE-75), Article 1: Normative Instruction RFB 852/2008 (Panel Exhibit JE-76), Article 4(II); Decree 6,759/2009 (Panel Exhibit JE-8), Article 283(II); Decree 7,212/2010 (Panel Exhibit JE-9) Article 150, section 1(II)).

⁸³³ Panel Reports, para. 2.76 (referring to Law 11,484/2007 (Panel Exhibit JE-71), Article 2, section 3, as amended by Law 12,715/2012 (Panel Exhibit JE-95), Article 57; Decree 6,233/2007 (Panel Exhibit JE-73), Article 6, section 3, as amended by Decree 8,247/2014 (Panel Exhibit JE-75), Article 1).

⁸³⁴ Panel Reports, para. 7.310.

⁸²⁹ Panel Reports, para. 2.71.

⁸³⁰ Panel Reports, para. 7.309.

⁸³¹ Panel Reports, para. 2.76. (referring to Law 11,484 of 31 May 2007 providing for the incentives to the Digital TV equipment and electronic semiconductor components industries and on the intellectual property protection of integrated circuit topographies, establishing the Semiconductor Technological Development Support Program (PADIS) and Support Program for the Technological Development of the Digital TV Equipment Industry (PATVD) (Law 11,484/2007) (Panel Exhibit JE-71), Article 2(I), as amended by Law 12,715 of 17 September 2012 (Law 12,715/2012) (Panel Exhibit JE-95), Article 57; Decree 6,233 of 11 October 2007 establishing the criteria for the purpose of accreditation to the Semiconductor Technological Development Support Program (PADIS), which grants income tax exemption and reduces PIS/PASEP, COFINS and IPI contributions to zero (Decree 6,233/2007) (Panel Exhibit JE-73), Article 6(I), as amended by Decree 8,247 of 23 May 2014, amending Decree 6,233/2007 (Decree 8,247/2014) (Panel Exhibit JE-75), Article 1; Regulatory Instruction RFB 852 of 13 June 2008 laying down procedures for qualification for the Programme for the Technological Development of the Semiconductor Industry (PADIS) by Secretary of the Federal Revenue Department of Brazil (Regulatory Instruction RFB 852/2008) (Panel Exhibit JE-76), Article 4(I); Decree 6,759/2009 (Panel Exhibit JE-8), Article 283(I); Decree 7,212 of 15 June 2010 regulating the charging, inspection, collection and administration of the Industrial Goods Tax (IPI) (Decree 7,212/2010) (Panel Exhibit JE-9), Article 150, section 1(I)).

- 100 -

performing just one of them would suffice to get accreditation. Thus, in contrast to some PPBs that require incorporation of components and subassemblies produced in accordance with the PPBs in the ensuing stage of production, the PADIS programme does not reflect such a requirement. It is possible that, in the scenario when a company seeking accreditation complies with all or several requirements under the PADIS programme, it would use the inputs produced in accordance with the PADIS programme in the ensuing stage of production. In our view, however, this would be a *result* of the eligibility requirements under the PADIS programme rather than a *condition* requiring the use of domestic over imported goods. As the Appellate Body noted in *US – Tax Incentives*, production subsidies "can ordinarily be expected to increase the supply of the subsidized domestic goods in the relevant market, thereby increasing the use of domestic over imported goods as a condition for granting the subsidy."⁸³⁵ Such effects of eligibility criteria for tax incentives are not sufficient for a finding of contingency upon the use of domestic over imported goods.

5.292. Accordingly, we agree with the Panel that the eligibility requirements under the PADIS programme are inconsistent with Article III:4 of the GATT 1994 because these eligibility requirements provide an *incentive* to use domestic ICT products and accord treatment less favourable to like imported ICT products covered under the PADIS programme. We do not consider, however, that these eligibility requirements under the PADIS programme also constitute a contingency *requirement* to use domestic over imported goods under Article 3.1(b) of the SCM Agreement.

5.293. We further note Brazil's argument that no PPBs have been developed under the PADIS programme for inputs and equipment intended for the manufacture of electronic semiconductor devices and information displays.⁸³⁶ We note that the Panel expressly recognized this fact in its analysis.⁸³⁷ The Panel further stated that, although the PADIS programme contemplates the adoption of PPBs in the future, the Panel could not "make findings in respect of production-step requirements that do not (yet) exist".⁸³⁸ The Panel also remarked that it sufficed for it to note that, "to the extent that any future PPBs adopted under the PADIS programme contain outsourcing provisions or nested PPBs in respect of manufactured components and subassemblies that operate in the same manner as those in the Informatics and PATVD programmes, such PPBs would require the use of domestic goods."⁸³⁹ We consider the Panel's statement unfortunate and unnecessary given that it concerned a potential WTO-inconsistency of a measure that Brazil had not taken. While the Panel made this statement, the Panel was clear in that it was not making a finding of inconsistency with respect to the PPBs that do not exist.

5.294. In light of the above, we <u>reverse</u> the Panel's findings, in paragraphs 7.313, 7.319, 8.5.e, and 8.16.f of the Panel Reports, that the PADIS programme requires the use of domestic over imported goods inconsistently with Article 3.1(b) of the SCM Agreement.

5.295. Finally, we recall that, "[t]he Digital Inclusion programme provides for zero rates with respect to PIS/PASEP and COFINS contributions for companies that sell in Brazil at retail level certain digital consumer goods produced in accordance with the relevant PPBs."⁸⁴⁰ As the Panel observed, "the role of PPBs under the Digital Inclusion programme departs from that under the other ICT **programmes** ... because under this programme the tax benefits are in respect of sales of certain products by retailers, rather than in respect of production of certain products by producers."⁸⁴¹ Having recalled its finding that all products produced in accordance with PPBs are Brazilian domestic products, the Panel found that only domestic goods are eligible for the tax benefits under the Digital Inclusion programme.⁸⁴² The Panel concluded that "this is a straightforward situation of incentives that are provided in respect of a preference (in this case by retailers) for domestic over imported goods."⁸⁴³ In the Panel's view, for the same reasons as with respect to the Informatics programme,

⁸³⁵ Appellate Body Report, US - Tax Incentives, para. 5.15.

⁸³⁶ Brazil's response to questioning at the oral hearing.

⁸³⁷ Panel Reports, para. 7.312. The Panel noted that, in reviewing the evidence before it, it found that "no PPBs have as yet been adopted in respect of these particular products." (Ibid.)

⁸³⁸ Panel Reports, para. 7.312.

⁸³⁹ Panel Reports, para. 7.312.

⁸⁴⁰ Panel Reports, para. 2.95.

⁸⁴¹ Panel Reports, para. 7.316.

⁸⁴² Panel Reports, para. 7.316.

⁸⁴³ Panel Reports, para. 7.317.

- 101 -

the Digital Inclusion programme is inconsistent with Article III: 4 of the GATT 1994 and "involves a contingency on the use of domestic over imported goods in the sense covered by Article 3.1(b) of the SCM Agreement".⁸⁴⁴

5.296. As the Panel noted, the Digital Inclusion programme provides a tax incentive to the retailers **"in respect of a preference** ... **for domestic over imported goods".**⁸⁴⁵ In our view, such a preference results in the less favourable treatment under Article III: 4 of the GATT 1994 for like imported digital consumer products due to the differential tax burden that imported products are subjected to by virtue of the fact that foreign producers cannot be accredited under that programme. We thus agree with the Panel that the accreditation requirements under the Digital Inclusion programme are inconsistent with Article III: 4 of the GATT 1994 since they provide an incentive to use domestic over imported goods.

5.297. We note, however, that no inconsistency with Article 3.1(b) of the SCM Agreement arises unless the granting of a subsidy is found to be contingent "upon the use of domestic over imported goods". We recall, in this respect, that the Panel found that the Digital Inclusion programme presents "a straightforward situation of incentives that are provided in respect of a preference (in this case by retailers) for domestic over imported goods".⁸⁴⁶ On that basis, the Panel found that the Digital Inclusion programme is inconsistent both with Article III: 4 of the GATT 1994 and with Article 3.1(b) of the SCM Agreement. While the existence of an incentive to buy and resell like domestic goods may be sufficient to meet the legal standard under Article III: 4 of the GATT 1994, it is not sufficient to establish an inconsistency with Article 3.1(b). Instead, in order to find an inconsistency with Article 3.1(b) of the SCM Agreement, a measure must be found to contain "*a condition requiring* the use of domestic over imported goods".⁸⁴⁷

5.298. We observe, in addition, that the Panel stated that it was reaching its findings "for the same reasons as outlined in [section 7.3.2.2.4.1 of its Reports]".⁸⁴⁸ That section of the Panel Reports, however, merely sets out the factual background of the operation of the PPBs and other production-step requirements. We thus fail to see how the discussion in that section could provide the reasons for the Panel's finding that the Digital Inclusion programme contains a contingency requirement that is inconsistent with Article 3.1(b) of the SCM Agreement.

5.299. In light of the above, we consider that the Panel did not have a proper basis to conclude that the Digital Inclusion programme contains a requirement to use domestic over imported goods under Article 3.1(b) of the SCM Agreement. We thus <u>reverse</u> the Panel's findings, in paragraphs 7.317, 7.319, 8.5.e, and 8.16.f of the Panel Reports, as they relate to the Digital Inclusion programme involves a contingency upon the use of domestic over imported goods that is inconsistent with Article 3.1(b) of the SCM Agreement.

5.300. We, however, agree with the Panel that the PPBs and other production-step requirements under the PATVD, PADIS, and Digital Inclusion programmes provide an incentive to use domestic ICT products. We therefore <u>uphold</u> the Panel's findings, in paragraphs 7.308, 7.311, 7.313, 7.317, 7.319, 8.5.b, and 8.16.c of the Panel Reports, that the Informatics, PATVD, PADIS, and Digital Inclusion programmes accord less favourable treatment to imported ICT products than that accorded to like domestic products, inconsistently with Article III:4 of the GATT 1994.

5.4.2.4.3 INOVAR-AUTO programme

5.301. Brazil further requests, "[t]o the extent that the Panel's findings are based on the erroneous assumption that production step requirements are sufficient to establish a contingency upon the use of domestic over imported goods within the meaning of Article 3.1(b) of the SCM Agreement"⁸⁴⁹, that we reverse the Panel's findings that certain aspects of the INOVAR-AUTO programme constitute a prohibited import substitution subsidy under Articles 3.1(b) and 3.2 of the SCM Agreement.⁸⁵⁰

⁸⁴⁴ Panel Reports, para. 7.317.

⁸⁴⁵ Panel Reports, para. 7.317.

⁸⁴⁶ Panel Reports, para. 7.317.

⁸⁴⁷ Appellate Body Report, US – Tax Incentives, para. 5.18. (emphasis original)

⁸⁴⁸ Panel Reports, para. 7.317.

⁸⁴⁹ Brazil's Notice of Appeal, para. 25.

⁸⁵⁰ Brazil's Notice of Appeal, para. 25 (referring to Brazil's appellant's submission, para. 295; Panel Reports, paras. 7.751, 7.823, 7.847, 8.6.e, and 8.17.f).

- 102 -

Brazil, however, has not developed any argumentation in support of this claim. In response to questioning at the oral hearing, Brazil explained that it requests the Appellate Body to reverse those findings to the extent the Appellate Body would reverse the corresponding findings under the ICT programmes.

5.302. The Panel made a finding of inconsistency with Article 3.1(b) of the SCM Agreement with respect to: (i) the accreditation requirement to perform a minimum number of manufacturing steps in Brazil; (ii) the rules on accrual of presumed tax credits, with respect to purchases of strategic inputs and tools; and (iii) the accreditation requirements with respect to expenditure and investment in R&D in Brazil, pertaining to the purchase of Brazilian laboratory equipment.⁸⁵¹ We understand that Brazil's claim concerns the Panel's findings relating to the accreditation requirements to perform a minimum number of manufacturing steps in Brazil.

5.303. In its reasoning concerning the accreditation requirement to perform a minimum number of manufacturing steps in Brazil, the Panel relied on its previous analysis with respect to the PPBs and other production-step requirements under the ICT programmes.⁸⁵² The Panel observed that the "production step requirements under the INOVAR-AUTO programme operate in an analogous manner" to those under the ICT programmes.⁸⁵³ As we see it, the requirement to perform a minimum number of manufacturing steps in Brazil operates in a way similar to the main PPBs that do not incorporate nested PPBs under the ICT programmes. The Panel's findings under the INOVAR-AUTO programme, similarly to those under the ICT programmes, also concern manufacturing steps that must be performed in Brazil in order to qualify for certain tax benefits.

5.304. Above, we have reversed the Panel's findings of inconsistency with Article 3.1(b) of the SCM Agreement with respect to the main PPBs that do not incorporate nested PPBs under the ICT programmes. As noted, the production-step requirements under the INOVAR-AUTO programme operate in a similar manner to the main PPBs that do not incorporate nested PPBs and, in reaching its findings of inconsistency with respect to the production-step requirements under the ICT programmes. Accordingly, having reversed the Panel's findings with respect to the main PPBs that do not incorporate nested PPBs under the ICT programme, the Panel relied on its analysis under the ICT programmes. Accordingly, having reversed the Panel's findings with respect to the main PPBs that do not incorporate nested PPBs under the ICT programmes, we also reverse the Panel's findings of inconsistency with Article 3.1(b) with respect to the requirement to perform a minimum number of manufacturing steps under the INOVAR-AUTO programme contained in paragraphs 7.751, 7.823, 7.847, 8.6.e, and 8.17.f of the Panel Reports.

5.4.3 The European Union's and Japan's appeal concerning the in-house scenario

5.4.3.1 Introduction

5.305. We now turn to consider the European Union's and Japan's claims on appeal. The European Union and Japan submit that, by not making specific findings on the in-house scenario, the Panel exercised false judicial economy and failed to make an objective assessment of the matter before it under Article 11 of the DSU.⁸⁵⁴ The European Union and Japan contend that they have challenged the production-step requirements of the ICT and INOVAR-AUTO programmes "as a whole", without distinguishing between the in-house and outsourcing scenarios⁸⁵⁵, such that the matter referred to the Panel "comprised both" scenarios.⁸⁵⁶ The European Union and Japan request that we reverse the relevant Panel findings and complete the legal analysis to find that the production-step requirements in the ICT and INOVAR-AUTO programmes are inconsistent with

⁸⁵¹ Panel Reports, para. 7.773.

⁸⁵² Panel Reports, paras. 7.747-7.750 and 7.770 (referring to Panel Reports, paras. 7.290-7.304 and 7.307-7.313).

⁸⁵³ Panel Reports, para. 7.747.

⁸⁵⁴ European Union's other appellant's submission, para. 36: Japan's other appellant's submission, paras. 20, 31, and 50.

⁸⁵⁵ European Union's other appellant's submission, para. 36: Japan's other appellant's submission, paras. 35-36 and 51.

⁸⁵⁶ European Union's other appellant's submission, paras. 56 and 63. See also Japan's other appellant's submission, paras. 18-19.

- 103 -

Article III:4 of the GATT 1994, Article 2.1 of the TRIMs Agreement, and Article 3.1(b) of the SCM Agreement under both the in-house and outsourcing scenarios.⁸⁵⁷

5.306. In the alternative, if we were to consider that the Panel correctly exercised judicial economy by not making specific findings in the in-house scenario, the European Union requests that we "*review*, pursuant to Article 17.6 of DSU, the legal interpretations developed by the Panel and modify, pursuant to Article **17.13 of DSU, the findings** ... so as *to make it clearer* that the Panel indeed did not need to rule twice on the production-step requirements (in the in-house and outsourcing scenarios) because [it] had already found that those steps are *per se* inconsistent with the covered agreements".⁸⁵⁸ Similarly, if we were to find that the Panel actually made a finding with respect to the ICT and INOVAR-AUTO programmes "as a whole", Japan requests that we review, pursuant to Article 17.6 of the DSU, the legal interpretations developed by the Panel, and modify, pursuant to Article 17.13 of the DSU, the relevant findings, to the extent they may be understood as referring solely to the outsourcing scenario.⁸⁵⁹ Should we decide that the Panel did not exercise judicial economy at all with respect to the in-house scenario, Japan also claims that the Panel acted inconsistently with Article 11 of the DSU because it failed to provide "coherent reasoning" under that provision.⁸⁶⁰

5.307. Finally, the European Union raises "a subordinate claim of error, subject to the Appellate Body rejecting" both of the above-mentioned claims.⁸⁶¹ The European Union requests that we "find that the Panel, by failing to consider the European Union's claims under Article III: 4 of [the] GATT 1994, Article 2.1 of [the] TRIMs Agreement and Article 3.1(b) of [the] SCM Agreement in light of all the relevant facts of the case (which included both the in-house and outsourcing scenarios), erred in the application of those provisions."⁸⁶²

5.308. In response, Brazil claims that the other appellants' claims of error under Article 11 of the DSU should be rejected.⁸⁶³ Brazil argues that the issue of judicial economy is "only relevant to the manner in which a panel deals with a party's *claims*"⁸⁶⁴, and that the in-house scenario was not a claim but an argument.⁸⁶⁵ Brazil considers, however, that the Panel duly examined and made findings with respect to all of the legal claims raised by the European Union and Japan under the ICT and the INOVAR-AUTO programmes under Article III:4 of the GATT, Article 2.1 of the TRIMs Agreement, and Article 3.1(b) of the SCM Agreement.⁸⁶⁶ Accordingly, Brazil considers that the Panel "did not act inconsistently with Article 11 of the DSU by not addressing the European Union and Japan's *argument* that in the in-house scenario the ICT programmes and INOVAR-AUTO gave rise to a requirement to use domestic over imported goods within the meaning of Article III:4 of the GATT 1994, Article 2.1 of the TRIMs Agreement, and Article 3.1(b) of the SCM Agreement, and Article 3.1(b) of the SCM Agreement.⁸⁶⁷

5.4.3.2 Panel's findings

5.309. We note that the other appeals in this dispute relate to a specific aspect of the ICT and INOVAR-AUTO programmes, namely, the in-house scenario in the context of compliance with the production-step requirements contained in PPBs or other instruments under the relevant programmes. In section 5.4.2.2 above, we have summarized the Panel's findings concerning the inconsistency of the PPBs and other production-step requirements under the ICT programmes with

⁸⁵⁷ The European Union requests we reverse the Panel findings in paragraphs 6.10-6.11, 6.13, 7.314, 7.749, and 8.5 of the Panel Reports. (European Union's other appellant's submission, para. 125) Japan requests that we reverse the Panel findings in paragraphs 7.302, 7.308, 7.314, and 7.317 of the Panel Reports. (Japan's other appellant's submission, paras. 52-53)

⁸⁵⁸ European Union's other appellant's submission, para. 125. (emphasis original)

⁸⁵⁹ Japan's other appellant's submission, paras. 30 and 49.

⁸⁶⁰ Japan's other appellant's submission, para. 21.

⁸⁶¹ European Union's other appellant's submission, para. 125.

⁸⁶² European Union's other appellant's submission, para. 125. (fn omitted) The European Union thus requests that we reverse the Panel's findings, complete the legal analysis with regard to the in-house scenario and find that the production-step requirements contained in the ICT and INOVAR-AUTO programmes are inconsistent with Article III: 4 of the GATT 1994, Article 2.1 of the TRIMs Agreement, and Article 3.1(b) of the SCM Agreement also under the in-house scenario. (Ibid.)

⁸⁶³ Brazil's appellee's submission, para. 29.

⁸⁶⁴ Brazil's appellee's submission, para. 6 (quoting Appellate Body Report, *EC – Fasteners (China)*, para. 511 (emphasis original)).

⁸⁶⁵ Brazil's appellee's submission, para. 8.

⁸⁶⁶ Brazil's appellee's submission, para. 7.

⁸⁶⁷ Brazil's appellee's submission, para. 12.

- 104 -

Article III:4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement.⁸⁶⁸ In the summary below, we recall the Panel's findings relevant for the in-house scenario under the ICT and the INOVAR-AUTO programmes.

5.310. Before the Panel, the complainants argued, *inter alia*, that the production-step requirements with which manufacturers must comply to receive the tax incentives under the four ICT programmes involve requirements to use domestic inputs in the production of the incentivized products⁸⁶⁹, and that "the requirement to perform certain manufacturing steps in Brazil [in all cases] is tantamount to requiring the incorporation of domestic content into the finished product", whenever the performance of those manufacturing steps results in the creation of a product.⁸⁷⁰ The Panel observed that the complainants "ma[de] this argument without regard for whether a single company itself performs all of the production steps and thus itself creates the 'inputs' in question (so-called 'in-house' production), or instead outsources some production steps to third parties, by acquiring from them the outputs of those product."⁸⁷¹

5.311. The Panel observed that, depending on the PPB, "[c]ertain production-step requirements must be performed by the company accredited as the producer of the incentivized finished or intermediate product that is subject of the PPB, while other production-step requirements may be performed by 'third parties' based in Brazil."⁸⁷² The Panel further noted that, in the complainants' view, the production-step requirements "on their own" constituted requirements to use domestic goods in the production of incentivized products, without regard to whether those requirements are met "in-house" or through outsourcing some of those steps.⁸⁷³ The Panel further observed that Brazil had argued that production-step requirements had "exclusively to do with production and in no case require the use of domestic goods".⁸⁷⁴ Nevertheless, the Panel stated that there seemed to be "no theoretical disagreement among the parties that at least in the case in which a company is required, when it acquires a product from an outside source, to only acquire a domestic product, there would be a requirement to use domestic goods in the sense covered by Article III: 4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement."⁸⁷⁵ The Panel considered that "[g]iven these views of the parties", it would be "useful to separately analyse the two possible scenarios for compliance with the PPBs ... : the in-house scenario and the outsourcing scenario".⁸⁷⁶

5.312. With respect to the outsourcing scenario, the Panel found that, because nested PPBs imposed a mandatory minimum amount of such "domestic goods" to be used in producing the product subject to the main PPB, the only way to satisfy the requirement, when outsourcing, was to acquire and use domestic goods.⁸⁷⁷ The Panel further considered that "[t]his analysis also holds true for the basic production step requirements of all PPBs under the Informatics programme."⁸⁷⁸

5.313. The Panel added that, even if certain PPBs contained alternative options to comply with certain production steps in those PPBs, the "mere existence of options for compliance that are potentially WTO-consistent could not preclude a finding of inconsistency in respect of the PPBs as a whole", so that the existence of alternative, potentially WTO-consistent options would not "alte[r] the inconsistency ... of an option that requires the use of domestic products over imported

⁸⁶⁸ See section 5.4.2.2 above.

⁸⁶⁹ Panel Reports, para. 7.256.

⁸⁷⁰ Panel Reports, para. 7.272 (quoting Japan's first written submission to the Panel, para. 296;

referring to European Union's first written submission to the Panel, para. 600)

⁸⁷¹ Panel Reports, para. 7.271.

⁸⁷² Panel Reports, para. 7.281. (emphasis omitted; fn omitted)

⁸⁷³ Panel Reports, para. 7.297. See also ibid., para. 7.271.

⁸⁷⁴ Panel Reports, para. 7.297.

⁸⁷⁵ Panel Reports, para. 7.297.

⁸⁷⁶ Panel Reports, para. 7.298.

⁸⁷⁷ Panel Reports, para. 7.300.

⁸⁷⁸ Panel Reports, para. 7.301. The Panel applied this reasoning to each of the four ICT programmes and concluded for each of them that, "regarding the outsourcing requirements" with respect to the production-step requirements for components and subassemblies used in the production of an incentivized product, the PPBs or analogous production-step requirements in the ICT programmes require the use of domestic goods, in the sense covered by Article III:4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement. (Panel Reports, paras. 7.302, 7.308, 7.311, and 7.316)

- 105 -

products".⁸⁷⁹ In light of this, and given the parties' "divergent views in respect of the in-house scenario for complying with the PPBs", the Panel found it "unnecessary" to address the in-house scenario.⁸⁸⁰ In the Panel's view, even if it found that the in-house scenario did not involve a requirement to use domestic goods, this would not alter its finding with respect to the outsourcing scenario that such measures are inconsistent with Article III: 4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement.881

5.314. With respect to the INOVAR-AUTO programme, the Panel noted that, to be eligible for the tax benefits under this programme, a company must comply with a minimum number of defined manufacturing and engineering infrastructure activities that "must be performed in Brazil either by the accredited company or through third parties".⁸⁸² Further, the performance of these steps must cover at least 80% of manufactured vehicles, and is subject to a schedule of the minimum number of activities to be performed, which varies by calendar year and type of vehicle manufactured.⁸⁸³ The Panel then recalled its analysis regarding the outsourcing scenario in the context of the ICT programmes, and considered that the production-step requirements in the INOVAR-AUTO programme operated in an "analogous manner".⁸⁸⁴ Thus, the Panel concluded that, in the outsourcing scenario, the production-step requirements under the INOVAR-AUTO programme require the use of domestic goods under Article III: 4 and Article 3.1(b).885

5.315. Furthermore, the Panel considered that for the same reasons as under the ICT programmes, it did not need to examine "whether under the 'in-house' scenario the production step requirements also would constitute a requirement to use domestic goods".⁸⁸⁶ The Panel added that it did not consider that the existence of an alternative, WTO-consistent option for compliance would alter the fact that the option that requires the use of domestic over imported goods is inconsistent with Article III: 4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement.887

5.316. In its comments on the Interim Report, the European Union, raising concerns about paragraphs 7.314 and 7.747 of the Report, argued that "the Panel's completion of its legal analysis in respect of the 'in-house scenario' is 'essential to secure a positive solution' to the dispute", because otherwise, "the 'complainants would be compelled to start a new panel procedure and repeat their legal claims, even though those legal claims [were] properly before [the] Panel.""888 The European Union thus requested the Panel to complete its legal analysis with regard to the in-house scenario, or to include elements in the Panel Reports that would allow the Appellate Body to do so.889

5.317. In responding to the European Union's request, the Panel first recalled the Appellate Body's statement in EC - Fasteners (China), that:

[A] panel has the discretion "to address only those arguments it deems necessary to resolve a particular claim" and "the fact that a particular argument relating to that claim is not specifically addressed in the 'Findings' section of a panel report will not, in and of itself, lead to the conclusion that that panel has failed to make the 'objective assessment of the matter before it' required by Article 11 of the DSU."890

5.318. For the Panel, its findings that the "relevant aspects of the programmes concerning production-step requirements, as challenged by the complaining parties in this dispute", are

⁸⁸⁴ Panel Reports, para. 7.747.

⁸⁸⁶ Panel Reports, para. 7.749.

⁸⁷⁹ Panel Reports, para. 7.303. See also ibid., paras. 7.304, 7.308, 7.311, and 7.314. In the Appendix to its Report, the Panel set out its analysis regarding individual production-step requirements that must be complied with to benefit from tax treatments under the four ICT programmes, in the outsourcing scenario. The Panel did not set out the potentially WTO-consistent alternative options referred to above. (Ibid., para. 9.4) ⁸⁸⁰ Panel Reports, para. 7.314.

⁸⁸¹ Panel Reports, para. 7.314.

⁸⁸² Panel Reports, para. 7.743.

⁸⁸³ Panel Reports, para. 7.743.

⁸⁸⁵ Panel Reports, para. 7.748.

⁸⁸⁷ Panel Reports, para. 7.750.

⁸⁸⁸ Panel Reports, para. 6.9 (quoting European Union's comments on the Interim Report, para. 26).

⁸⁸⁹ Panel Reports, para. 6.9 (quoting European Union's comments on the Interim Report, para. 26).

⁸⁹⁰ Panel Reports, para. 6.10 (quoting Appellate Body Report, EC - Fasteners (China), para. 511, in turn referring to Appellate Body Report, EC - Poultry, para. 135).

- 106 -

inconsistent with certain provisions of the covered agreements makes it unnecessary to address the complainants' argument that the relevant production steps are inconsistent with the exact same provisions, for reasons other than those identified by the Panel.⁸⁹¹ The Panel further noted that "its factual findings are sufficient should the Appellate Body decide to rule on this issue", and that "should the Appellate Body want to review the Panel's analysis, it will be able to benefit from the descriptive part of the Report, the exhibits contained in the Panel record (and identified in the descriptive part), and the Appendix attached to the Report."⁸⁹² Finally, the Panel considered it "inappropriate at this stage to prejudge the manner in which Brazil may come into compliance with [its] obligations", and found that issues pertaining to the manner of Brazil's compliance can be addressed in Article 21.5 proceedings.⁸⁹³ Thus, the Panel refused to make additional findings on the in-house scenario.

5.4.3.3 Analysis

5.319. The European Union and Japan maintain that they have challenged the production-step requirements of the ICT and the INOVAR-AUTO programmes "as a whole", without distinguishing between the in-house and outsourcing scenarios⁸⁹⁴, such that the matter referred to the Panel "comprised both" scenarios.⁸⁹⁵ They also raise before us a series of alternative claims concerning the Panel's findings regarding the in-house scenario under the ICT and INOVAR-AUTO programmes.

5.320. As noted, the European Union and Japan submit that the Panel, by deciding not to make specific findings on the in-house scenario, exercised false judicial economy and failed to make an objective assessment of the matter before it contrary to the requirements of Article 11 of the DSU.⁸⁹⁶ In the European Union's and Japan's view, by distinguishing between the in-house scenario and outsourcing scenarios on its own volition and by making specific findings only with regard to the latter scenario, the Panel failed to secure "a positive solution" to the dispute, in the sense of Article 3.7 of the DSU.⁸⁹⁷

5.321. As an alternative to this claim, the European Union and Japan each raise a similar claim predicated on different conditions. If we were to consider that the Panel correctly exercised judicial economy by not making findings concerning the in-house scenario, the European Union requests that we "*review*, pursuant to Article 17.6 of [the] DSU, the legal interpretations developed by the Panel and modify, pursuant to Article 17.13 of [the] **DSU, the findings** ... so as to make it clearer that the Panel indeed did not need to rule twice on the production step requirements (in the in-house and in the outsourcing scenarios) because [it] had already found that those steps are *per se* inconsistent with the covered agreements".⁸⁹⁸ Japan makes a similar request conditioned upon us concluding that the Panel actually made a finding with respect to the ICT and INOVAR-AUTO programmes "as a whole".⁸⁹⁹ In the alternative, if we were to decide that the Panel did not exercise judicial economy at all with respect to the in-house scenario, Japan claims that the Panel acted inconsistently with Article 11 of the DSU because it failed to provide "coherent reasoning" under that provision.⁹⁰⁰

5.322. Finally, the European Union also raises "a subordinate claim of error", subject to us rejecting both of the above-mentioned claims.⁹⁰¹ The European Union thus requests that we reverse the

- ⁸⁹⁸ European Union's other appellant's submission, para. 125. (emphasis original)
- ⁸⁹⁹ Japan's other appellant's submission, paras. 30 and 49.
- ⁹⁰⁰ Japan's other appellant's submission, para. 21.

⁸⁹¹ Panel Reports, para. 6.11 (referring to sections 7.3.2.2.4, 7.3.4, 7.4.2.4.2, and 7.4.5 of the Panel Reports).

⁸⁹² Panel Reports, para. 6.12.

⁸⁹³ Panel Reports, para. 6.13.

⁸⁹⁴ European Union's other appellant's submission, para. 36: Japan's other appellant's submission, paras. 35-36 and 51.

⁸⁹⁵ European Union's other appellant's submission, paras. 56 and 63. See also Japan's other appellant's submission, paras. 18-19.

 ⁸⁹⁶ European Union's other appellant's submission, para. 36; Japan's other appellant's submission, paras. 20, 31, and 50.
 ⁸⁹⁷ Japan's other appellant's submission, para. 20; European Union's other appellant's submission,

⁸⁹⁷ Japan's other appellant's submission, para. 20; European Union's other appellant's submission, para. 48.

⁹⁰¹ European Union's other appellant's submission, para. 125. In particular, the European Union submits that "the Panel made an error in the application of Article III:4 of [the] GATT 1994, Article 2.1 of [the] TRIMs Agreement and Article 3.1(b) of [the] SCM Agreement by failing to consider those provisions in light of all the relevant facts of the case, i.e. the in-house scenario." (Ibid., para. 122. (emphasis omitted))

Panel's findings and complete the legal analysis with regard to the in-house scenario and find that the production-step requirements contained in the ICT and INOVAR-AUTO programmes are inconsistent with Article III: 4 of the GATT 1994, Article 2.1 of the TRIMs Agreement, and Article 3.1(b) of the SCM Agreement under both the in-house and outsourcing scenarios.⁹⁰²

5.323. The European Union and Japan submit that the lack of clarity concerning the WTO-consistency of the in-house scenario could prevent effective implementation of the Panel's recommendations and rulings in this dispute.⁹⁰³ The European Union considers that the Panel Reports could be understood as "not requiring Brazil to amend in any way the challenged measures, in so far as applicable to the in-house scenario", and leaving the participants and other WTO Members "in a situation of legal uncertainty" as regards the compatibility of the in-house scenario with the relevant WTO provisions.⁹⁰⁴ Similarly, Japan argues that the lack of findings regarding the in-house scenario may lead to future disagreement as to whether Brazil is still permitted to maintain the measures at issue with respect to the in-house scenario. Furthermore, because a compliance panel's terms of reference are limited to measures taken to comply with the recommendations and rulings of the DSB, and it is unclear whether the Panel's findings, recommendations, and rulings (and thus Brazil's future compliance obligations) extend to the accreditation requirement as a whole, a positive solution to the dispute is not secured.⁹⁰⁵

5.324. For its part, Brazil does not contest that the European Union and Japan, in making their claims before the Panel, did not distinguish between in-house and outsourcing scenarios.⁹⁰⁶ Brazil also agrees with the European Union and Japan that the scope of the Panel's findings at issue is unclear.⁹⁰⁷ Brazil, however, claims that the issue of judicial economy is "only relevant to the manner in which a panel deals with a party's *claims*"⁹⁰⁸, and that the in-house scenario was not a claim, but an argument.⁹⁰⁹ Hence, in Brazil's view, the European Union's and Japan's claims of error under Article 11 of the DSU should be rejected because judicial economy cannot be exercised improperly with respect to arguments.⁹¹⁰

5.325. Brazil submits that the Panel was not required to "address every conceivable 'factual situation'" in which the European Union and Japan considered that the measures at issue are inconsistent with the covered agreements, and neither was the Panel's decision not to address the in-house scenario "internally contradictory with any interpretative finding made by the Panel, as Japan incorrectly posits".⁹¹¹ Rather, the Panel simply "considered all of the arguments put forward ... **but effectively decided t**o attribute to the in-house scenario the weight and significance it considered appropriate", within the bounds of its discretion under Article 11.⁹¹²

5.326. As we see it, at the heart of the European Union's and Japan's appeal is the concern that, due to an alleged lack of clarity in the Panel's findings concerning the in-house scenario, the implementation of the Panel's recommendations and rulings in this dispute may be compromised and certain issues may be left unresolved.⁹¹³ We thus start our analysis by examining whether the Panel's findings cover the in-house scenario.

5.327. We recall that, prior to its analysis of the ICT programmes, the Panel noted that there seemed to be "no theoretical disagreement among the parties" that, in case a company is required, when it acquires a product from an outside source, to only acquire a domestic product, there would be a requirement to use domestic goods.⁹¹⁴ The Panel, however, also noted that, while the

⁹⁰⁷ Brazil's responses to questioning at the oral hearing.

⁹⁰² European Union's other appellant's submission, para. 125.

⁹⁰³ European Union's and Japan's responses to questioning at the oral hearing.

⁹⁰⁴ European Union's other appellant's submission, para. 85.

⁹⁰⁵ Japan's other appellant's submission, paras. 38-39. See also European Union's other appellant's submission, para. 86.

⁹⁰⁶ Brazil's responses to questioning at the oral hearing.

⁹⁰⁸ Brazil's appellee's submission, para. 6 (quoting Appellate Body Report, EC - Fasteners (China),

para. 511 (emphasis original)).

⁹⁰⁹ Brazil's appellee's submission, para. 8.

⁹¹⁰ Brazil's appellee's submission, para. 29.

⁹¹¹ Brazil's appellee's submission, para. 11 (referring to European Union's other appellant's submission,

para. 77; Japan's other appellant's submission, para. 21, in turn referring to Appellate Body Report,

Colombia - Textiles, para. 5.27).

⁹¹² Brazil's appellee's submission, para. 12.

⁹¹³ European Union's and Japan's responses to questioning at the oral hearing.

⁹¹⁴ Panel Reports, para. 7.297.

- 108 -

complainants considered that "the production-step requirements on their own constitute requirements to use domestic goods", Brazil argued that production-step requirements have exclusively to do with production and do not require the use of domestic goods.⁹¹⁵ The Panel therefore considered that it would be "useful" to separately analyse the two possible scenarios, starting with the outsourcing one.⁹¹⁶ Accordingly, the Panel's subsequent analysis explored how the production-step requirements may result in a "requirement to use domestic goods" in the context of the "outsourcing scenario".⁹¹⁷ The Panel observed, in particular, that, whenever a producer of a product covered by a main PPB obtained components or subassemblies covered by its own nested PPB, it was obtaining "domestic goods". Furthermore, in the Panel's view, because nested PPBs imposed a mandatory minimum amount of such "domestic goods" to be used in producing the product subject to the main PPB, the only way to satisfy the requirement, when outsourcing, was to acquire and use domestic goods.918

5.328. In addition, with regard to three out of four of the ICT programmes (i.e. the Informatics, PATVD, and PADIS programmes), the Panel found that, "regarding the outsourcing requirements in respect of the production step requirements for components and subassemblies"919 used in the production of an incentivized product, the PPBs required the use of domestic goods in the sense of Article III: 4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement.⁹²⁰ The explicit references to the outsourcing scenario in those findings may be viewed as suggesting that the Panel's findings of inconsistency, in paragraphs 7.302, 7.308, and 7.311-7.314 of its Report, concern the measures only as they apply in the outsourcing scenario, without prejudice to how they may apply in the in-house scenario.⁹²¹ Similarly, with regard to the INOVAR-AUTO programme, the Panel found that, in the outsourcing scenario, the production-step requirements under the INOVAR-AUTO programme require the use of domestic goods under Article III: 4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement.922

5.329. Other elements of the Panel's analysis, however, can be understood as covering the in-house scenario. In particular, in its analysis under the ICT programmes, the Panel repeatedly observed that, even if certain PPBs contain "alternative options to compliance with certain production-steps in the PPBs", the "mere existence of options for compliance that are potentially WTO-consistent could not preclude a finding of inconsistency in respect of the PPBs as a whole", so that the existence of alternative, potentially WTO-consistent options, would not "alte[r] the inconsistency ... of an option that requires the use of domestic products over imported products".⁹²³ The Panel subsequently referred to these observations in its analysis of the consistency of the INOVAR-AUTO programme with Article III: 4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement.⁹²⁴ These statements by the Panel may be seen as suggesting that the in-house scenario could be a "potentially WTO-consistent" option, had the Panel examined it.925 We also note that the Panel is unclear in stating, on the one hand, that "potentially WTO-consistent [options] could not preclude a finding of inconsistency in respect of the PPBs as a whole", and, on the other hand, that potentially

⁹²¹ We note that, for the Digital Inclusion programme, however, the Panel did not make such explicit references to the outsourcing scenario, because it applies to sales of certain products by retailers. The Panel simply found that "this programme involves a contingency on the use of domestic over imported goods in the sense covered by Article 3.1(b) of the SCM Agreement." (Panel Reports, para. 7.317)

⁹²⁴ Panel Reports, paras. 7.749-7.750 (referring to Panel Reports, paras. 7.303-7.304) and 7.770 (referring to Panel Reports, paras. 7.258, 7.303, and fn 648 to para. 7.258). ⁹²⁵ Panel Reports, para. 7.314.

⁹¹⁵ Panel Reports, para. 7.297.

⁹¹⁶ Panel Reports, para. 7.298.

⁹¹⁷ Panel Reports, paras. 7.298-7.314 and 7.746-7.751.

⁹¹⁸ Panel Reports, para. 7.300.

⁹¹⁹ Panel Reports, para. 7.302. (emphasis added)

⁹²⁰ Panel Reports, paras. 7.302, 7.308, 7.311, and 7.316. In the case of the PADIS programme, the Panel found that a company seeking accreditation had to comply with different production-step requirements depending on the type of products, and that for certain products, no PPBs have been adopted yet. In this respect, the Panel noted that "to the extent that any future PPBs adopted under the PADIS programme contain outsourcing provisions or nested PPBs in respect of manufactured components and subassemblies that operate in the same manner as those in the Informatics and PATVD programmes, such PPBs would require the use of domestic goods." (Ibid., para. 7.312 (emphasis added))

⁹²² Panel Reports, para. 7.748.

⁹²³ Panel Reports, para. 7.303. See also ibid., para. 7.304.

WTO-consistent options would not "alte[r] the inconsistency ... of an option that requires the use of domestic products over imported products".⁹²⁶

5.330. We also note that the Panel's overall conclusions regarding the ICT and INOVAR-AUTO programmes under Article III: 4 of the GATT 1994 appear to suggest that the Panel's findings of inconsistency cover the measures as a whole. In particular, the Panel found, with respect to the ICT programmes, that:

[T]he production-step requirements and the requirement for products to obtain the status of "developed" in Brazil under the Informatics, PADIS, and PATVD programmes, **and certain eligibility requirements under the Digital Inclusion programme** ... accord to imported products treatment less favourable than that accorded to like domestic products, inconsistently with Article III: 4 of the GATT 1994.⁹²⁷

5.331. Similarly, in the conclusions regarding the INOVAR-AUTO programme under Article III: 4 of the GATT 1994, the Panel found that, "under the INOVAR-AUTO programme, the conditions for **accreditation in order to receive presumed tax credits** ... **accord less favourable treatment to imported** products than that accorded to like domestic products, inconsistently with Article III: 4 of the GATT 1994."⁹²⁸ Thus, in its overall conclusions on those two programmes, the Panel did not specify the factual scenarios in which the relevant production-step requirements and analogous requirements are WTO-inconsistent. Neither did the Panel refer to such distinction in the conclusions and recommendations of its Reports.⁹²⁹

5.332. In its overall conclusions and recommendations, the Panel may be seen as having made a finding of inconsistency with the relevant provisions for the measures as a whole, without distinguishing between the two factual scenarios used by the Panel in its analysis.

5.333. Our examination of the relevant Panel's analysis and findings thus reveals that some aspects of the Panel's reasoning and conclusions appear to refer to the outsourcing scenario, while excluding the in-house scenario. Other aspects of the Panel's analysis, including the overall conclusions and recommendations can be understood as covering measures as a whole and thus extending to the in-house scenario. In this respect, we share the concern of the European Union and Japan that the lack of clarity with respect to the scope of the Panel's findings may compromise the effective implementation of the recommendations and rulings in this dispute. This would not contribute to achieving a positive solution to this dispute, as required under Article 3.7 of the DSU.

5.334. We recall, that the Panel reached its findings of inconsistency under Article III: 4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement with respect to the ICT programmes because it considered that the PPBs and other production-step requirements require the use of domestic goods.⁹³⁰ Subsequently, the Panel referred to its analysis under the ICT programmes in reaching findings of inconsistency under Article III: 4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement with respect to the accreditation requirement to perform certain manufacturing steps in Brazil under the INOVAR-AUTO programme.⁹³¹ The Panel also relied on these findings to reach its findings of inconsistency under Article 2.1 of the TRIMs Agreement with respect to the ICT and INOVAR-AUTO programmes.⁹³² Accordingly, the Panel used its findings of the existence of the requirement to use domestic goods under Article III: 4 of the GATT and Article 3.1(b) of the SCM Agreement under the ICT programmes.⁹³² Accordingly, the Panel used its findings of the existence of the requirement to use domestic goods under Article III: 4 of the GATT and Article 3.1(b) of the SCM Agreement under the ICT programmes as a basis for its subsequent findings.

5.335. We also recall that the participants agree that the Panel's finding of inconsistency with respect to the PPBs and other production-step requirements under the ICT programmes was of a *de jure* nature.⁹³³ In its analysis, under the ICT programmes, the Panel relied on the text of the relevant legal instruments and its own understanding of the operation of the PPBs and other production-step requirements and did not examine the factual circumstances surrounding the

⁹²⁶ Panel Reports, para. 7.303. (emphasis added)

⁹²⁷ Panel Reports, para. 7.318.

⁹²⁸ Panel Reports, para. 7.772.

⁹²⁹ Panel Reports, paras. 8.5.b, 8.5.d-e, 8.6b, 8.6.d-e. 8.16.c, 8.16.e-f, 8.17.c, and 8.17.e-f.

⁹³⁰ Panel Reports, paras. 7.313 and 7.319.

⁹³¹ Panel Reports, paras. 7.747-7.751.

⁹³² Panel Reports, paras. 7.364 and 7.805.

⁹³³ Participants' responses to questioning at the oral hearing.

- 110 -

granting of the subsidy.⁹³⁴ Moreover, in examining the claims concerning the ICT programmes, the Panel indicated that it was analysing "the specific provisions of the PPBs"⁹³⁵ and that the PPBs contain "an explicit requirement to use domestic goods". 936 Similarly, in the context of the INOVAR-AUTO programme, the Panel considered "that the accreditation requirements to invest in R&D in Brazil and make expenditure[s] in engineering, basic industrial technology and capacity-building of suppliers in Brazil, with respect to laboratory equipment used in performing R&D in Brazil, by the necessary implication of the wording of the measure, require the use of domestic over imported goods in order to be accredited and obtain the tax benefits."937 Likewise, in our analysis above, we have focused on whether the PPBs and other production-step requirements under the ICT programmes contain in their very terms, or by necessary implication therefrom, a requirement to use domestic over imported goods under Article 3.1(b) and provide less favourable treatment under Article III: 4 of the GATT 1994.938

5.336. We further recall that the legal standard under Article 3.1(b) of the SCM Agreement and Article III: 4 of the GATT is not the same. While an inquiry under Article 3.1(b) of the SCM Agreement focuses on whether there is a condition requiring the use of domestic over imported goods, an incentive to use domestic goods is sufficient to find an inconsistency with Article III:4 of the GATT 1994.939

5.337. We recall that the legal standard under Article 3.1(b) of the SCM Agreement requires that a condition requiring the use of domestic over imported goods be discerned from the terms of the measure itself, or inferred from its design, structure, modalities of operation, and the relevant factual circumstances constituting and surrounding the granting of the subsidy that provide context for understanding the operation of these factors.⁹⁴⁰ Accordingly, for purposes of establishing an inconsistency with Article 3.1(b) of the SCM Agreement, whether a company produces goods in-house or whether it outsources its production is not decisive. What matters, instead, is whether such a measure reflects a condition requiring the use of domestic over imported goods.⁹⁴¹

5.338. By contrast, local content requirements that alter the conditions of competition to the detriment of the imported products by providing an incentive to use domestic goods will be found to be inconsistent with Article III: 4 of the GATT 1994.942 For purposes of analysis under Article III: 4 of the GATT, as well as under Article 2.1 of the TRIMs Agreement, whether a company produces goods in-house or whether it outsources its production would not be determinative. Instead, the relevant inquiry is whether the measure accords to imported products treatment less favourable than that accorded to the domestic products.943

5.339. In light of the above, we consider that it did not matter, for purposes of the Panel's analysis, what factual scenarios were available for compliance with the requirements under the ICT and INOVAR-AUTO programmes. The Panel's bifurcation of its analysis into the two possible factual scenarios was thus unnecessary. Moreover, as noted, for purposes of establishing an inconsistency

⁹⁴⁰ Appellate Body Report, US - Tax Incentives, para. 5.18.

⁹⁴¹ We recall, however, that, in US - Tax Incentives, the Appellate Body explained that "the existence of de facto contingency 'must be inferred from the total configuration of the facts constituting and surrounding the granting of the subsidy, none of which on its own is likely to be decisive in any given case'. (Appellate Body Report, US - Tax Incentives, para. 5.12 (quoting Appellate Body Report, Canada - Aircraft, para. 167. (emphasis original))) Such "factual circumstances potentially relevant to an assessment of whether a subsidy is de facto contingent may include the existence of a multi-stage production process, the level of specialization of the subsidized inputs, or the level of integration of the production chain in the relevant industry." (Ibid., fn 49 to para. 5.13) 942 Appellate Body Report, US - FSC (Article 21.5 - EC), para. 220.

⁹⁴³ We also recall that the Panel considered that the in-house scenario is one of "the two possible [factual] scenarios for compliance with the PPBs" under the ICT programmes. (Panel Reports, para. 7.298) We note, in this respect, that, for purposes of a de jure analysis, the factual circumstances surrounding the measure are of limited importance. Rather, a subsidy will be de jure contingent upon the use of domestic over imported goods under Article 3.1(b) of the SCM Agreement when the existence of that condition can be demonstrated on the basis of the very words of the measure, or can be derived from them by necessary implication. (Appellate Body Report, Canada - Autos, paras. 100 and 123)

⁹³⁴ Panel Reports, para. 7.285.

⁹³⁵ Panel Reports, para. 7.285.

⁹³⁶ Panel Reports, para. 7.300.

⁹³⁷ Panel Reports, para. 7.770. (emphasis added)

⁹³⁸ See section 5.4.2.4.1 above.

⁹³⁹ See paragraph 5.254 above.

- 111 -

with Article 3.1(b) of the SCM Agreement, Article III:4 of the GATT 1994, and Article 2.1 of the TRIMs Agreement, the possible factual scenarios existing under the measure are not decisive. What matters, instead, is whether the respective legal standard has been met.

5.340. In light of the above, we <u>reverse</u> the Panel's findings, in paragraphs 7.303-7.304 and 7.314 of the Panel Reports, made in the context of its analysis under ICT programmes, to the extent that they can be understood as suggesting that the in-house scenario was not covered by the Panel's findings. We also <u>reverse</u> the Panel's findings, in paragraphs 7.749-7.750 and 7.770 of the Panel Reports, made in the context of INOVAR-AUTO programme and referring to the mentioned Panel's findings under the ICT programmes, to the extent that they can also be understood as suggesting that the in-house scenario was not covered by the Panel's findings. We thus consider that the Panel's findings, in paragraphs 7.319, 7.772-7.773, 8.5.e, 8.6.b, 8.6.e, 8.16.c, 8.16.f, 8.17.c, and 8.17.f of the Panel Reports, apply also in the in-house scenario. For these reasons, we need not further address the European Union's and Japan's claims raised on appeal.

5.5 Article I:1 of the GATT 1994 and the Enabling Clause

5.341. We now turn to consider Brazil's claim on appeal that the Panel erred in finding that the claims raised by the European Union and Japan under Article I: 1 of the GATT 1994 were within its terms of reference and that the differential and more favourable treatment in the form of internal tax reductions accorded to imports from Argentina, Mexico, and Uruguay under the INOVAR-AUTO programme was not justified under paragraphs 2(b) and 2(c) of the Enabling Clause.⁹⁴⁴ We will begin by examining whether the Panel erred in its interpretation and application of paragraph 4(a) of the Enabling Clause. In so doing, we will first address the notification requirement in paragraph 4(a). We then review whether the differential tax treatment under the INOVAR-AUTO programme was notified pursuant to paragraph 4(a) as having been adopted under paragraphs 2(b) and 2(c) of the Enabling Clause, such that the complaining parties could be considered to have been on notice and, consequently, could have been expected to raise the Enabling Clause and identify the relevant provision(s) thereof in their panel requests. We then examine whether the Panel erred in its interpretation and application of paragraph 2(b) of the Enabling Clause, where we first examine the scope of that provision. Next, we review whether, as claimed by Brazil, the differential tax treatment under the INOVAR-AUTO programme falls within the scope of paragraph 2(b) and is therefore substantively justified under that provision. Finally, we examine Brazil's claim on appeal that the Panel erred in finding that Brazil has not identified any arrangement adopted under paragraph 2(c) that has a genuine link to the internal tax reductions under the INOVAR-AUTO programme such that the differential and more favourable treatment at issue can be substantively justified under that provision.

5.5.1 Whether the Panel erred in finding that the claims raised by the European Union and Japan under Article I:1 of the GATT 1994 were within its terms of reference

5.342. Before the Panel, the European Union and Japan raised claims under Article I:1 of the GATT 1994 with respect to certain aspects of the INOVAR-AUTO programme. The European Union and Japan argued that Brazil accords an advantage in the form of internal tax reductions implemented through Articles 21 and 22(I) of Decree 7,819/2012 to motor vehicles imported into Brazil from a member-country of MERCOSUR and Mexico that is not accorded to like motor vehicles imported into Brazil from other WTO Members, including the European Union and Japan.⁹⁴⁵

5.343. The Panel found that the tax reductions accorded to motor vehicles imported from MERCOSUR members and Mexico under the INOVAR-AUTO programme are advantages granted by Brazil to products originating in those countries that are not accorded immediately and unconditionally to like products originating in other WTO Members, inconsistently with Article I:1 of the GATT 1994.⁹⁴⁶

5.344. The Panel then turned to consider Brazil's defences under the Enabling Clause. Brazil argued that the differential tax treatment accorded to Argentina, Mexico, and Uruguay under the INOVAR-AUTO programme was justified under both paragraphs 2(b) and 2(c) of the Enabling Clause

⁹⁴⁴ GATT Document L/4903, 28 November 1979, BISD 26S/203.

⁹⁴⁵ Panel Reports, para. 7.1012 (referring to European Union's first written submission to the Panel,

paras. 346-365; Japan's first written submission to the Panel, paras. 274-281).

⁹⁴⁶ Panel Reports, para. 7.1048.

- 112 -

and was notified to the WTO, as required under paragraph 4(a) of the Enabling Clause.947 Brazil further argued that the European Union and Japan had the burden of invoking the Enabling Clause in their panel requests and "since they did not do so, they cannot challenge the right of Brazil to invoke paragraph[s] 2(b) and 2(c) of the Enabling Clause to justify the inconsistency of the INOVAR-AUTO programme with Article I:1 of the GATT 1994."948

5.345. The Panel began by examining who had the burden of invoking the Enabling Clause. The Panel stated that the issue of whether the Enabling Clause has been properly invoked pertains to the jurisdiction of the Panel, and specifically "whether the complaining parties' claim under Article I: 1 of the GATT 1994 is within the Panel's terms of reference". 949

5.346. The Panel noted Brazil's argument that the Appellate Body has established that "a complaining party has the burden of indicating in its panel request that a particular challenged measure is not consistent with the relevant provisions of the Enabling Clause."950 The Panel considered that the Appellate Body's statements in EC - Tariff Preferences were made in the specific context of that dispute.⁹⁵¹ According to the Panel, the Appellate Body's findings indicated that "the burden of invoking the Enabling Clause is placed on the complaining party in situations where the complaining party is on notice that the challenged measure was adopted (and in the view of the adopting member, justified) under the Enabling Clause."952 The Panel, therefore, considered that in situations where a WTO Member has notified a particular arrangement imposing discriminatory treatment as adopted or modified under the Enabling Clause, other WTO Members are presumed to be aware that the specific discriminatory treatment was adopted pursuant to the Enabling Clause.⁹⁵³ The Panel concluded that a complaining party does not have the burden to invoke the Enabling Clause in its panel request, unless that complaining party is informed that the responding party considers the challenged measure to have been adopted pursuant to the Enabling Clause.954

5.347. The Panel next considered "whether the challenged measures and their related justifications were notified to the WTO"955 such that the complaining parties were on notice and therefore had the burden to invoke the relevant provisions of the Enabling Clause in their panel requests.

5.348. The Panel recalled that Brazil submitted that the differential tax treatment under the INOVAR-AUTO programme was notified to the WTO, as required under paragraph 4(a) of the Enabling Clause.⁹⁵⁶ Brazil argued that the notification requirement was satisfied because "the 1980 Treaty of Montevideo establishing the Latin American Integration Association (LAIA) was notified to the WTO under [paragraph] 2(c) of the Enabling Clause"957 and that the differential tax treatment granted to Argentina, Mexico, and Uruguay is based on economic complementation agreements (ECAs) negotiated under the auspices of the 1980 Treaty of Montevideo.⁹⁵⁸ Brazil also argued that the "ECAs 'are implementation measures of the {1980} Treaty of Montevideo and [we]re notified to

⁹⁴⁷ Panel Reports, para. 7.1054. The Panel noted that Brazil did not invoke the Enabling Clause with respect to the differential and more favourable treatment accorded to motor vehicles imported from Paraguay and Venezuela. (Ibid.) In response to questioning at the oral hearing in these appellate proceedings, Brazil confirmed that it does not challenge the Panel's findings contained in paragraphs 7.1048, 8.6.g-h, and 8.17.h-i of the Panel Reports that the internal tax reductions under the INOVAR-AUTO programme are inconsistent with Article I:1 of the GATT 1994 to the extent they concern Paraguay and Venezuela. (Brazil's response to questioning at the oral hearing)

⁹⁴⁸ Panel Reports, para. 7.1054 (referring to Brazil's first written submissions to the Panel. paras. 694-742 (DS472) and paras. 627-674 (DS497); second written submission to the Panel,

paras. 166-181; response to Panel questions Nos. 4 and 53-55; comments on other parties' responses to Panel questions Nos. 53-55).

⁹⁴⁹ Panel Reports, para. 7.1062.

⁹⁵⁰ Panel Reports, para. 7.1057 (referring to Brazil's response to Panel question No. 4, in turn referring to Appellate Body Report, *EC – Tariff Preferences*, para. 110). ⁹⁵¹ Panel Reports, para. 7.1063.

⁹⁵² Panel Reports, para. 7.1064.

⁹⁵³ Panel Reports, para. 7.1066.

⁹⁵⁴ Panel Reports, para. 7.1068. We note that, although one panelist appended a separate opinion, this panelist, however, does not disagree with this standard articulated by the Panel. (Panel Reports, para. 7.1130) ⁹⁵⁵ Panel Reports, para. 7.1068.

⁹⁵⁶ Panel Reports, para. 7.1069.

⁹⁵⁷ Panel Reports, para. 7.1072.

⁹⁵⁸ Panel Reports, para. 7.1072 (referring to Brazil's first written submissions to the Panel, paras. 706-707 (DS472) and paras. 638-639 (DS497)).

- 113 -

the WTO'" and that the differential tax treatment under the INOVAR-AUTO programme is "the corollary of these ECAs and does not require further notification".959

5.349. The Panel noted that the 1980 Treaty of Montevideo was notified to the WTO by Uruguay on behalf of the LAIA on 1 July 1982, "as adopted pursuant to paragraph 2(c) of the Enabling Clause".⁹⁶⁰ The Panel, however, stated that "no explicit notification in respect of paragraph 2(b) ha[d] been identified by any party, or submitted to the Panel as evidence."961 Therefore, in order to determine whether the complaining parties were on notice that the challenged measure was adopted under paragraph 2(b) and, accordingly, whether or not they were required to invoke this provision of the Enabling Clause in their panel requests, the Panel turned to the question whether a notification of an arrangement adopted pursuant to paragraph 2(c) could also serve as a notification of an arrangement adopted pursuant to paragraph 2(b) of the Enabling Clause.⁹⁶² The Panel found that a notification of a regional trade agreement (RTA) adopted under paragraph 2(c) of the Enabling Clause, even if valid, is not sufficient to serve as a notification of a preferential trade arrangement (PTA) adopted under paragraph 2(b) of the Enabling Clause and that "there was no notification made under [paragraph] 4(a) to support a justification under paragraph 2(b)."963 The Panel concluded that "Brazil has not demonstrated that any arrangement providing for the differential and more favourable treatment at issue was notified to the WTO as adopted pursuant to paragraph 2(b)."964

5.350. The Panel next considered whether the notification of the 1980 Treaty of Montevideo and the ECAs could substantively serve as a notification of the adoption under paragraph 2(c) of the Enabling Clause of the arrangement introducing the differential and more favourable treatment found to be inconsistent with Article I:1 of the GATT 1994, i.e. the differential tax treatment under the INOVAR-AUTO programme.965 The Panel stated that "the differential and more favourable treatment sought to be justified under paragraph 2(c) must have a close and genuine link to the arrangement notified to the WTO such as to put other WTO Members on notice as to the adoption of the differential and more favourable treatment pursuant to the Enabling Clause."966 The Panel found that "none of the provisions cited to in the [1980] Treaty of Montevideo bear the slightest relation[,] in and of themselves, to the internal tax reductions found to be inconsistent with Article I:1 of the GATT 1994."967 The Panel also found that Brazil had failed to point to "a single provision of any ECA that would attest to the fundamental premise of Brazil's argument, namely that the INOVAR-AUTO programme is implementing the objectives of the ECAs". ⁹⁶⁸ Accordingly, the Panel found that "Brazil has not demonstrated how the relevant tax reductions found to be inconsistent under Article I:1 of the GATT 1994" are related to the 1980 Treaty of Montevideo or the ECAs that were notified as having been adopted pursuant to paragraph 2(c).969

⁹⁵⁹ Panel Reports, para. 7.1072 (quoting Brazil's second written submission, para. 181; referring to Brazil's response to Panel question No. 53; Communication from LAIA to the Chair of the Trade and Development Committee of the WTO, 19 May 2016 (Panel Exhibits BRA-114 and BRA-115)). (curly brackets added) The Panel noted:

[[]T]he MERCOSUR agreement was allegedly notified on 5 March 1992 in document L/6985 ECA No. 55 [between MERCOSUR and Mexico] was allegedly notified on 21 November 2012 in document WT/COMTD/77; the 38th Additional Protocol to ECA No. 14 [between Argentina and Brazil] and the 68th Additional Protocol to ECA No. 2 [between Brazil and Uruguay] were allegedly notified on 8 November 2010, in document WT/COMTD/72; and the 69th Additional Protocol to ECA No. 2 [between Brazil and Uruguay] was allegedly notified on 25 October 2013 in document WT/COMTD/82.

⁽Panel Reports, fn 1443 to para. 7.1080 (referring to Communication from LAIA to the Chair of the Trade and Development Committee of the WTO, 19 May 2016 (Panel Exhibits BRA-114 and BRA-115)))

⁹⁶⁰ Panel Reports, para. 7.1074. (fn omitted)

⁹⁶¹ Panel Reports, para. 7.1074.

⁹⁶² Panel Reports, para. 7.1075.

⁹⁶³ Panel Reports, para. 7.1081. The Panel considered that paragraph 4(a) of the Enabling Clause "does not ... permit[] notification of a measure adopted under one provision of the Enabling Clause to serve equally as a notification of that measure being adopted under another provision of the Enabling Clause, unless indicated in the notification itself". (Panel Reports, para. 7.1079) 964 Panel Reports, para. 7.1082. (fn omitted)

⁹⁶⁵ Panel Reports, para. 7.1105.

⁹⁶⁶ Panel Reports, para. 7.1108. (emphasis original)

⁹⁶⁷ Panel Reports, para. 7.1112.

⁹⁶⁸ Panel Reports, para. 7.1112.

⁹⁶⁹ Panel Reports, para. 7.1115.

- 114 -

5.351. Thus, the Panel concluded that the challenged measure and its related justifications under the Enabling Clause (i.e. under paragraphs 2(b) and 2(c) thereof) were not notified to the WTO pursuant to paragraph 4(a) of the Enabling Clause, such that the complaining parties could be considered to have been on notice. Consequently, the Panel found that there was no burden on the complaining parties to invoke paragraphs 2(b) and 2(c) of the Enabling Clause in their panel requests, and therefore their claims under Article I: 1 of the GATT 1994 were within the Panel's terms of reference.⁹⁷⁰

5.5.1.1 Whether the Panel erred in its interpretation of paragraph 4(a) of the Enabling Clause

5.352. On appeal, Brazil takes issue with the Panel's finding that "the obligation to invoke the Enabling Clause would only apply if the complaining party had been 'appropriately informed that the responding party considers the challenged measure to have been adopted pursuant to (and justified under) the Enabling Clause."⁹⁷¹ According to Brazil, the Panel considered that "a disagreement regarding whether the notification was deemed appropriate suffices to waive the complainants' burden of invoking the Enabling Clause in their panel requests."⁹⁷² In Brazil's view, "[t]his logic, if it were to stand, would raise important systemic questions", because "a disagreement on the method of notification could justify *ex post* the lack of inclusion of the Enabling Clause" in a complainant's panel request.⁹⁷³ Therefore, Brazil submits that, by conflating the substantive obligation to notify properly the measures and the threshold presumption of being informed that a Member considers a measure to be justified under the Enabling Clause, the Panel's understanding deprives of any meaningful effect the Appellate Body's finding that it is incumbent on the complaining party to raise the Enabling Clause in its panel request.⁹⁷⁴

5.353. In response, the European Union submits that the Appellate Body's findings in *EC* – *Tariff Preferences* have to be understood in the context of that particular case and of the specific provision of the Enabling Clause at issue.⁹⁷⁵ The European Union explains that measures taken pursuant to paragraph 2(a) of the Enabling Clause "are different from measures taken pursuant to other paragraphs of the Enabling Clause, and in particular those pertaining to regional trade agreements".⁹⁷⁶

5.354. Japan contends that the present dispute can be distinguished from that in *EC* – *Tariff Preferences*, which involved tariff preferences for developing countries adopted pursuant to paragraph 2(a) of the Enabling Clause, as opposed to issues relating to RTAs.⁹⁷⁷ According to Japan, "[i]n the case of a preference accorded pursuant to a Generalized System of Preferences ('GSP') it can be expected that the legal dispute would revolve around the (in)application of the Enabling Clause" and thus the complaining party "has to 'present the problem clearly' as required

⁹⁷⁰ Panel Reports, paras. 7.1083 and 7.1120. One Panelist appended a separate opinion on Brazil's defence under paragraph 2(c) of the Enabling Clause. According to this Panelist, references in the relevant ECAs to "tariff preferences [were] sufficient to put Members on notice that, at a minimum, motor vehicles imported into Brazil from Mexico, Argentina and Uruguay will be treated differently to motor vehicles imported into Brazil from other Members." (Ibid., para. 7.1127) In the view of this Panelist, it was therefore "reasonable for Members to assume that such preferences would not be limited to fiscal measures applied at the border, but could potentially include internal fiscal measures" and Members could be considered to have been informed that "motor vehicles imported into Brazil from Mexico, Argentina and Uruguay may be subject to a differential tax burden". (Ibid.) Accordingly, this Panelist considered that "the ECAs are ... substantively sufficient to satisfy the notification obligation in paragraph 4(a) of the Enabling Clause." (Ibid., para. 7.1128) Moreover, this Panelist was of the view that "Brazil has at a minimum demonstrated that the European Union and Japan were informed that the Treaty of Montevideo, the MERCOSUR Agreement, and the relevant ECAs were adopted pursuant to paragraph 2(c) of the Enabling Clause." (Ibid., para. 7.1129) This Panelist therefore considered that "the European Union and Japan were sufficiently on notice that the Enabling Clause might be invoked by Brazil, a signatory to those ECAs, as a defence to a claim under Article I:1 of the GATT 1994, in respect of preferences granted to the signatories to those ECAs." (Ibid.) For these reasons, this Panelist concluded that the European Union and Japan were under an obligation to have included the relevant provisions of the Enabling Clause within their respective panel requests and the failure to do so "mean[t] that their claims under Article I:1 of the GATT 1994 [were] outside the Panel's terms of reference". (Ibid., para. 7.1131)

⁹⁷¹ Brazil's appellant's submission, para. 374 (quoting Panel Reports, para. 7.1068).

⁹⁷² Brazil's appellant's submission, para. 376.

⁹⁷³ Brazil's appellant's submission, para. 377.

⁹⁷⁴ Brazil's appellant's submission, para. 378.

⁹⁷⁵ European Union's appellee's submission, para. 451.

⁹⁷⁶ European Union's appellee's submission, para. 453.

⁹⁷⁷ Japan's appellee's submission, paras. 153 and 155.

- 115 -

under Article 6.2 of the DSU by referring to the Enabling Clause explicitly in its panel request".⁹⁷⁸ However, Japan submits that, in a dispute such as the present one, a complaining party cannot be expected to assume that the "measures were adopted pursuant to the Enabling Clause (particularly when the responding Member has never said so and never made any notification of the sort to the WTO)."⁹⁷⁹

5.355. We begin our analysis with the text of paragraph 4(a) of the Enabling Clause, which provides, in relevant part:

Any Member taking action to introduce an arrangement pursuant to paragraphs 1, 2 and 3 above or subsequently taking action to introduce modification or withdrawal of the differential and more favourable treatment so provided shall: [*]

(a) notify the WTO and furnish [Members] with all the information they may deem appropriate relating to such action[.]

[*fn original]⁴ Nothing in these provisions shall affect the rights of Members under the General Agreement.

5.356. Paragraph 4(a) of the Enabling Clause thus deals with the requirement to notify any action by a Member seeking to introduce, modify, or withdraw differential and more favourable arrangements adopted pursuant to paragraphs 1 through 3 of the Enabling Clause. The Appellate Body has described paragraph 4 as setting forth "procedural conditions for the introduction, modification, or withdrawal of a preferential measure for developing countries".⁹⁸⁰

5.357. The use of the word "shall" indicates that paragraph 4(a) imposes an obligation on a Member according differential and more favourable treatment to notify the WTO of the arrangement it has adopted. In addition to the obligation to notify the introduction of an arrangement, paragraph 4(a) also imposes an obligation on Members to notify any modification or the withdrawal of the arrangement according differential and more favourable treatment. Paragraph 4(a) thus envisages that, at all times, Members are kept informed of any changes to, including the withdrawal of, an arrangement according differential and more favourable treatment.

5.358. Moreover, we observe that paragraph 4(a) provides that a Member adopting an arrangement according differential and more favourable treatment "furnish" Members "with all the information they may deem appropriate" relating to the introduction, modification, or withdrawal of the arrangements adopted. This requirement to furnish "all"⁹⁸¹ the information suggests that a notification pursuant to paragraph 4(a) should be sufficiently detailed so as to put the Members on notice regarding any "action" taken pursuant to paragraphs 1 through 3 of the Enabling Clause. The need for notifications under paragraph 4(a) to be sufficiently detailed is also borne out by the requirement to notify not only the introduction or withdrawal of an arrangement according differential and more favourable treatment but also of any modifications thereof.

5.359. Turning to the immediate context provided by paragraph 4(b) of the Enabling Clause, we observe that paragraph 4(b) stipulates that a Member introducing, modifying, or withdrawing an arrangement according differential and more favourable treatment shall "afford adequate opportunity for prompt consultations at the request of any interested Member with respect to any difficulty or matter that may arise" in connection with the adopted arrangement. Paragraph 4(b) thus builds upon the notification issued pursuant to paragraph 4(a) of the Enabling Clause by calling upon the WTO Member adopting an arrangement according differential and more favourable treatment to "afford adequate opportunity" for "prompt consultations" as may be requested by any other WTO Member in connection with the adopted arrangement.

5.360. A notification pursuant to paragraph 4(a) of the Enabling Clause thus speaks to and has a direct bearing on a complaining party's knowledge and, consequently, on the question whether it is

⁹⁷⁸ Japan's appellee's submission, para. 155. (fn omitted)

⁹⁷⁹ Japan's appellee's submission, para. 155.

⁹⁸⁰ Appellate Body Report, *EC – Tariff Preferences*, para. 112.

⁹⁸¹ Emphasis added.

- 116 -

required to raise the Enabling Clause and identify the relevant provision(s) thereof in its panel request.

5.361. We recall that although the dispute in *EC* – *Tariff Preferences* did not involve the notification requirement in paragraph 4(a) specifically, the Appellate Body set out relevant considerations concerning the interpretation of the Enabling Clause. The Appellate Body considered that the Enabling Clause is not a "typical 'exception', or 'defence'". 982 The Appellate Body stated that, when a complaining party considers that a preference scheme of another Member does not meet one or more of the requirements set forth in the Enabling Clause, "the specific provisions of the Enabling Clause with which the scheme allegedly falls afoul[] form critical components of the 'legal basis of the complaint' and, therefore, of the 'matter' in dispute."983 However, at the same time, the Appellate Body cautioned that "[t]he responsibility of the complaining party in such an instance ... should not be overstated."984 Although the Appellate Body found that "it is insufficient in WTO dispute settlement for a complainant to allege inconsistency with Article I:1 of the GATT 1994 if the complainant seeks also to argue that the measure is not justified under the Enabling Clause"985, the Appellate Body also explained that "[t]his is especially so if the challenged measure ... is plainly taken pursuant to the Enabling Clause."986

5.362. However, the Appellate Body's statements in EC - Tariff Preferences concerning the burden on the complaining party to raise the Enabling Clause and identify the relevant provision(s) thereof in its panel request should be read in the context of the challenged measure at issue in that dispute, i.e. the tariff preference scheme, which as the Appellate Body itself indicated, was "plainly taken pursuant to the Enabling Clause".⁹⁸⁷ The Appellate Body further noted that the challenged measure in that dispute was "unmistakably a preferential tariff scheme, granted by a developed-country Member in favour of developing countries, and proclaiming to be in accordance with the GSP". 988 Thus, the Appellate Body found it "clear, on the face of the Regulation and from official, publicly-available explanatory documentation", that the "Drug Arrangements" at issue in that dispute were "part of a preferential tariff scheme implemented by the European Communities pursuant to the authorization in paragraph 2(a) of the Enabling Clause". 989 Accordingly, in that dispute, the Appellate Body noted that India would have been "well aware" that the Drug Arrangements must comply with the requirements of the Enabling Clause, and that "the European Communities was likely to invoke the Enabling Clause in response to a challenge of inconsistency with Article I: 1."990

5.363. Paragraph 4(a) of the Enabling Clause envisages a degree of specificity in the notification adopted thereunder. At a minimum, a notification pursuant to paragraph 4(a) should state under which provision of the Enabling Clause the differential and more favourable treatment has been adopted. Paragraph 4(a) indicates that arrangements or measures adopted under different subparagraphs of paragraph 2 would have to be notified to the WTO so as to put other Members on notice regarding the relevant differential and more favourable treatment sought to be accorded and justified under the Enabling Clause. In such circumstances, the mere procedural propriety of the notification itself, for example, in terms of "whether such notification was ... sent by the right actor or body, under the right procedure, at the right time, etc."⁹⁹¹, is, however, not sufficient to dislodge the presumption that the complaining party is on notice that the responding party has adopted an arrangement or a measure that may be inconsistent with its obligations under Article I:1 of the GATT 1994, but that may nonetheless be justified under the Enabling Clause.

5.364. Moreover, paragraph 4(a) does not exclude the possibility that a single notification can state that the notifying Member considers an arrangement or a measure to have been adopted pursuant to one or more subparagraphs of paragraph 2 of the Enabling Clause. We do not detect anything in the text of paragraph 4(a) that indicates otherwise. To the contrary, paragraph 4 is broadly worded

⁹⁸² Appellate Body Report, EC - Tariff Preferences, para. 106.

⁹⁸³ Appellate Body Report, EC - Tariff Preferences, para. 113 (quoting Article 6.2 of the DSU; Appellate Body Report, US - Carbon Steel, para. 125, in turn referring to Appellate Body Report, Guatemala - Cement I, paras. 69-76).

⁹⁸⁴ Appellate Body Report, *EC – Tariff Preferences*, para. 115.
⁹⁸⁵ Appellate Body Report, *EC – Tariff Preferences*, para. 110.
⁹⁸⁶ Appellate Body Report, *EC – Tariff Preferences*, para. 110.

⁹⁸⁷ Appellate Body Report, *EC – Tariff Preferences*, para. 110. (emphasis added)

⁹⁸⁸ Appellate Body Report, EC - Tariff Preferences, para. 116. (emphasis added)

 ⁹⁸⁹ Appellate Body Report, *EC – Tariff Preferences*, para. 117. (emphasis added)
 ⁹⁹⁰ Appellate Body Report, *EC – Tariff Preferences*, para. 117.

⁹⁹¹ Panel Reports, para. 7.1105.

- 117 -

in providing that any "action to introduce an arrangement pursuant to *paragraphs 1, 2 and 3*" of the Enabling Clause be notified to the WTO.⁹⁹² However, in the absence of any such indication in the notification issued under paragraph 4(a), it cannot be taken for granted that a complaining party is on notice of those subparagraphs of paragraph 2 that the notifying Member considers applicable.

5.365. A complaining party is therefore required to raise the Enabling Clause and identify the relevant provisions thereof in its panel request when a measure according differential and more favourable treatment is: (i) plainly taken pursuant to the Enabling Clause, or when it is clear from the face of the measure itself that it has been adopted pursuant to the Enabling Clause; and/or (ii) notified pursuant to paragraph 4(a) of the Enabling Clause. However, the complaining party "is merely to *identify* those provisions of the Enabling Clause with which the [measure] is allegedly inconsistent, without bearing the burden of *establishing* the facts necessary to support such inconsistency".⁹⁹³ Thus, while it is for the complaining party to identify the relevant provision(s) of the Enabling Clause in its panel request, the burden to "*prove*" that the measure "satisf[ies] the conditions set out in the Enabling Clause"⁹⁹⁴ still "remains on the responding party" relying on "the Enabling Clause as a defence".⁹⁹⁵

5.366. In light of the foregoing considerations, we agree with the Panel that, in situations where a Member has notified a particular arrangement imposing discriminatory treatment as adopted or modified under the Enabling Clause, other Members would be considered to be aware that the specific discriminatory treatment was adopted pursuant to the Enabling Clause.⁹⁹⁶ Therefore, to the extent that the Panel conditioned a complaining party's knowledge of an arrangement or a measure according differential and more favourable treatment as having been adopted under the Enabling Clause to the notification by the responding party of that arrangement or measure pursuant to paragraph 4(a), we see no reason to disagree with this standard articulated by the Panel. As explained, a notification pursuant to paragraph 4(a) of the Enabling Clause speaks to and has a direct bearing on a complaining party's knowledge and, consequently, on its burden to raise the Enabling Clause and identify the relevant provision(s) thereof in its panel request. We, therefore, <u>uphold</u> the Panel's finding that a complaining party has to raise the Enabling Clause and identify the relevant provision(s) thereof in its panel request "in situations where the complaining party is on notice that the challenged measure was adopted (and in the view of the adopting member, justified) under the Enabling Clause".⁹⁹⁷

5.5.1.2 Whether the Panel erred in finding that the differential tax treatment under the INOVAR-AUTO programme was not notified pursuant to paragraph 4(a) as having been adopted under paragraph 2(b) of the Enabling Clause

5.367. We now turn to review whether the measure at issue (the differential tax treatment under the INOVAR-AUTO programme in the form of internal tax reductions accorded to some but not other Members) was notified pursuant to paragraph 4(a) of the Enabling Clause as having been adopted under paragraph 2(b) thereof, such that the complaining parties could be considered to have been on notice and, consequently, had the burden to raise and identify paragraph 2(b) in their panel requests.

5.368. We recall that Decree No. 7,819/2012 is one of the instruments that administers the INOVAR-AUTO programme⁹⁹⁸ and provides for the differential and more favourable treatment at issue in the form of internal tax reductions on imports of motor vehicles from certain countries.

5.369. Brazil's case before the Panel rested on its contention that, since the 1980 Treaty of Montevideo and the relevant ECAs were notified to the WTO as having been adopted under paragraph 2(c), the notification requirement in paragraph 4(a) with respect to the differential tax

⁹⁹⁵ Appellate Body Report, EC - Tariff Preferences, para. 115.

⁹⁹⁸ Panel Reports, para. 2.100 (referring to Decree 7,819 of 3 October 2012 regulating Articles 40 to 44 of Law 12,715 of 17 September 2012, which contain provisions on the Incentive Scheme for Technological Innovation and Consolidation of the Automotive Supply Chain (INOVAR-AUTO) and Articles 5 and 6 of Law 12,546 of 14 December 2011, which contain provisions on a reduction of the Industrial Goods Tax, as the case dictates (Decree 7,819/2012) (Panel Exhibit JE-132), Articles 11-19).

⁹⁹² Emphasis added.

⁹⁹³ Appellate Body Report, EC - Tariff Preferences, para. 115. (emphasis original)

⁹⁹⁴ Appellate Body Report, *EC - Tariff Preferences*, para. 105. (emphasis original)

⁹⁹⁶ Panel Reports, para. 7.1066.

⁹⁹⁷ Panel Reports, para. 7.1064.

- 118 -

treatment under the INOVAR-AUTO programme stood satisfied.⁹⁹⁹ However, in its defence, Brazil contended that the differential tax treatment under the INOVAR-AUTO programme was justified not only under paragraph 2(c)¹⁰⁰⁰, but also under paragraph 2(b).¹⁰⁰¹ We therefore understand, as did the Panel, Brazil to have contended that the differential tax treatment under the INOVAR-AUTO programme was adopted pursuant to both paragraphs 2(b) and 2(c) of the Enabling Clause and did not require additional notification, since the 1980 Treaty of Montevideo and the relevant ECAs were notified as adopted pursuant to paragraph 2(c).

5.370. It is undisputed that the differential tax treatment under the INOVAR-AUTO programme, which is the measure at issue in this case, was not specifically notified to the WTO pursuant to paragraph 4(a) as having been adopted under paragraph 2(b). The Panel, as we recall, noted that "no explicit notification in respect of paragraph 2(b) ha[d] been identified by any party, or submitted to the Panel as evidence."¹⁰⁰² The Panel therefore considered that its task was to determine whether a notification of an arrangement adopted pursuant to paragraph 2(c) could also serve as a notification of an arrangement adopted pursuant to paragraph 2(b).¹⁰⁰³ As a second step, the Panel considered that if the answer to the first question was in the positive, it would then determine whether the notification under paragraph 2(c) could substantively serve as a notification of the specific differential and more favourable treatment sought to be justified under paragraph 2(b).¹⁰⁰⁴

5.371. Unlike what Brazil seems to contend¹⁰⁰⁵, the Panel did not consider the guestion before it to concern the procedural propriety of the notification. Rather, the Panel's enquiry focused on whether or not the complaining parties could be considered to have been on notice that the differential tax treatment under the INOVAR-AUTO programme was notified as having been adopted pursuant to paragraph 2(b) by virtue of the notification under paragraph 2(c). The analysis of whether or not an arrangement or a measure alleged to be adopted under paragraph 2(b) was notified pursuant to paragraph 4(a) was necessary in determining the complaining parties' burden to raise the Enabling Clause and identify the relevant provisions(s) thereof in their panel requests and allege the inconsistency of the differential tax treatment under the INOVAR-AUTO programme with these relevant provisions of the Enabling Clause, including paragraph 2(b). However, in so doing, neither do we nor did the Panel prejudge the question whether the differential tax treatment under the INOVAR-AUTO programme was notified pursuant to paragraph 4(a) as having been adopted under paragraph 2(c).¹⁰⁰⁶ Indeed, that is the question we address in the next section of these Reports when we review Brazil's claim on appeal that the Panel erred in finding that the differential tax treatment under the INOVAR-AUTO programme was not notified pursuant to paragraph 4(a) as having been adopted under paragraph 2(c).

5.372. Brazil also challenges the Panel's finding that "a notification of an RTA adopted under paragraph 2(c) of the Enabling Clause, even if valid, is not sufficient to serve as a notification of a PTA adopted under paragraph 2(b) of the Enabling Clause."¹⁰⁰⁷ Brazil asserts that the Panel's reasoning was based on the provisions of the Transparency Mechanism for Preferential Trade Arrangements¹⁰⁰⁸ and the Transparency Mechanism for Regional Trade Agreements¹⁰⁰⁹, which the Panel found to constitute subsequent agreements within the meaning of Article 31(3)(a) of the Vienna Convention on the Law of Treaties (Vienna Convention) regarding the interpretation of the

- ¹⁰⁰³ Panel Reports, para. 7.1075.
- ¹⁰⁰⁴ Panel Reports, para. 7.1075.

¹⁰⁰⁵ We note that Brazil argues that "the Panel concluded that because the challenged tax treatment was not notified in a specific format, Brazil had not met the notification requirements under the Enabling Clause, and the complainants could not have been presumed to be informed that the challenged measures were adopted under the Enabling Clause." (Brazil's appellant's submission, para. 375)

¹⁰⁰⁶ The Panel, as we recall, eventually concluded that "a notification of an RTA adopted under paragraph 2(c) of the Enabling Clause, even if valid, is not sufficient to serve as a notification of a PTA adopted under paragraph 2(b) of the Enabling Clause." (Panel Reports, para. 7.1081 (italics original; underlining added)) ¹⁰⁰⁷ Brazil's appellant's submission, para. 398 (quoting Panel Reports, para. 7.1081). (emphasis original)

¹⁰⁰⁸ Transparency Mechanism for Preferential Trade Arrangements, General Council Decision of 14 December 2010, WT/L/806.

¹⁰⁰⁹ Transparency Mechanism for Regional Trade Agreements, General Council Decision of 14 December 2006, WT/L/671.

⁹⁹⁹ Panel Reports, para. 7.1072. See also Brazil's appellant's submission, para. 395.

¹⁰⁰⁰ Panel Reports, para. 7.1099 (referring to Brazil's second written submission to the Panel,

para. 181).

¹⁰⁰¹ Panel Reports, para. 7.1069.

¹⁰⁰² Panel Reports, para. 7.1074.

- 119 -

notification obligation under the Enabling Clause.¹⁰¹⁰ In particular, Brazil asserts that the Panel found that the Transparency Mechanisms indicate that "Members notifying an RTA or PTA must specify under which precise provision of the Enabling Clause the RTA or PTA is being notified."¹⁰¹¹ According to Brazil, "the Transparency Mechanisms both explicitly state that they do not affect the 'substance and timing **of the notifications required under** *...* **the Enabling** Clause' in the case of the Transparency Mechanism for RTAs"¹⁰¹² nor "the 'substance of the relevant provisions of the Enabling Clause' in the case of the Transparency Mechanism for PTAs".¹⁰¹³

5.373. The European Union contends that Brazil "misrepresents the Panel's findings and attempts to create the impression that the Panel based its decision on a flawed interpretation of the Transparency Mechanisms".¹⁰¹⁴ The European Union submits that the Panel made it clear that it had already reached its conclusions by analysing the provisions of paragraph 4(a) and that "the Transparency Mechanisms were assessed only in order *to confirm* its conclusions."¹⁰¹⁵ The European Union adds that even documents that were not found by previous panels to constitute "subsequent agreements regarding the interpretation of the treaty or the application of its provisions" under Article 31(3)(a) of the Vienna Convention were nevertheless considered as providing useful guidance.¹⁰¹⁶ Accordingly, the European Union submits that, "even if Brazil would be right (*quod non*), the Transparency Mechanisms would still be able to play a useful interpretative role on the notification issue."¹⁰¹⁷

5.374. Similarly, Japan submits that "Brazil appears to make much of the fact" that the Panel referred to the Transparency Mechanisms as "subsequent agreements regarding the interpretation of the treaty or the application of its provisions" under Article 31(3)(a) of the Vienna Convention.¹⁰¹⁸

5.375. In interpreting paragraph 4(a), we have considered that this provision envisages a degree of specificity in the notification adopted thereunder and, at a minimum, a notification pursuant to paragraph 4(a) should state under which provision of the Enabling Clause the differential and more favourable treatment has been adopted. At the same time, we have also considered that paragraph 4(a) does not exclude the possibility that a single notification can state that the notifying Member considers an arrangement or a measure to have been adopted pursuant to one or more subparagraphs of paragraph 2. However, we have also explained that in the absence of any such indication, it cannot be taken for granted that a complaining party is on notice that the notifying Member considers the notified arrangement or measure to have been adopted pursuant to one or more subparagraphs of paragraph 2. Therefore, to the extent that the Panel observed that paragraph 4(a) "does not explicitly indicate what precisely is required to be notified" 1019 , we disagree.

5.376. That said, we note that the Panel ultimately found that paragraph **4(a) "does not ... permit[]** notification of a measure adopted under one provision of the Enabling Clause to serve equally as a notification of that measure being adopted under another provision of the Enabling Clause, unless indicated in the notification itself".¹⁰²⁰ We agree with this conclusion of the Panel.

5.377. Contrary to what Brazil contends, in reaching this conclusion, the Panel did not base its reasoning exclusively on the provisions of the Transparency Mechanisms. Rather, the Panel found that "the Transparency Mechanisms *serve to further confirm*" that paragraph 4(a) does not permit a notification of a measure adopted under one provision of the Enabling Clause to function as a

¹⁰¹⁰ Brazil's appellant's submission, para. 397 (referring to Panel Reports, fn 1443 to para. 7.1080).

¹⁰¹¹ Brazil's appellant's submission, para. 397.

¹⁰¹² Brazil's appellant's submission, para. 399 (quoting Transparency Mechanism for Regional

Trade Agreements, General Council Decision of 14 December 2006, WT/L/671, para. 1).

 ¹⁰¹³ Brazil's appellant's submission, para. 399 (quoting Transparency Mechanism for Preferential Trade Arrangements, General Council Decision of 14 December 2010, WT/L/806, Article A.2).
 ¹⁰¹⁴ European Union's appellee's submission, para. 486 (referring to Brazil's appellant's submission,

¹⁰¹⁴ European Union's appellee's submission, para. 486 (referring to Brazil's appellant's submission, para. 397).

¹⁰¹⁵ European Union's appellee's submission, para. 485 (referring to Panel Reports, fn 1443 to para. 7.1080). (emphasis original)

¹⁰¹⁶ European Union's appellee's submission, para. 488 and fns 465-466 thereto (referring to Appellate Body Report, *US – Tuna II (Mexico)*, para. 372; Panel Report, *India – Agricultural Products*, fn 1197 to para. 7.679).

¹⁰¹⁷ European Union's appellee's submission, para. 490.

¹⁰¹⁸ Japan's appellee's submission, para. 163.

¹⁰¹⁹ Panel Reports, para. 7.1106.

¹⁰²⁰ Panel Reports, para. 7.1079.

- 120 -

notification of adoption of that same measure under another provision of the Enabling Clause.¹⁰²¹ The Panel further recalled that it had "*concluded ... without reference* to the Transparency Mechanisms" that paragraph 4(a) does not permit notification of a measure adopted pursuant to one provision of the Enabling Clause to suffice as notification of that same measure being adopted pursuant to a different provision of the Enabling Clause.¹⁰²² Thus, having already reached a conclusion on what is permitted under paragraph 4(a), the Panel found added support for its conclusion from the provisions of the Transparency Mechanisms.

5.378. Brazil further contends that "in the case of developing country Members, paragraphs 2(a) and 2(b) of the Enabling Clause are both contained in paragraph 2(c)."¹⁰²³ According to Brazil, under paragraph 2(c) "[d]eveloping country Members may adopt tariff preferences *vis-à-vis* other developing country Members without having to create a Generalized System of Preferences."¹⁰²⁴ Similarly, under that provision "a developing country Member may adopt preferential treatment with regard to non-tariff measures."¹⁰²⁵ Therefore, Brazil submits that the fact that a given agreement has been notified under paragraph 2(c), as was the case in the present dispute, suffices for the complaining parties to have been on notice that the measures at issue fell under the Enabling Clause.¹⁰²⁶

5.379. Paragraph 2 of the Enabling Clause provides, in relevant part:

- 2. The provisions of paragraph 1 apply to the following:
 - (a) Preferential tariff treatment accorded by developed country Members to products originating in developing countries in accordance with the Generalized System of Preferences;
 - (b) Differential and more favourable treatment with respect to the provisions of the General Agreement concerning non-tariff measures governed by the provisions of instruments multilaterally negotiated under the auspices of the GATT;
 - (c) Regional or global arrangements entered into amongst developing country Members for the mutual reduction or elimination of tariffs and, in accordance with criteria or conditions which may be prescribed by the WTO, for the mutual reduction or elimination of non-tariff measures, on products imported from one another[.]¹⁰²⁷

5.380. Subparagraphs (a) to (c) of paragraph 2 of the Enabling Clause provide for differential and more favourable treatment with respect to which the authorization of paragraph 1 of the Enabling Clause applies. Paragraph 2(a) provides for differential and more favourable treatment in the form of preferential tariff treatment accorded by developed-country Members to products originating from developing countries. Paragraph 2(b) provides for differential and more favourable treatment concerning non-tariff measures. Unlike paragraph 2(a), which specifically speaks of "[p]referential tariff treatment accorded by developed country Members to ... developing countries", paragraph 2(b) does not define either the grantor or the beneficiary of the differential and more favourable treatment. Paragraph 2(c), unlike both paragraphs 2(a) and 2(b), provides for differential and more favourable treatment concerning tariff and non-tariff measures between developing country Members pursuant to "[r]egional or global arrangements".

5.381. As noted above, Brazil asserts that "in the case of developing country Members, paragraphs **2(a) and 2(b)** ... are both contained in paragraph 2(c)."¹⁰²⁸ Even assuming it to be so, it does not necessarily follow that the notification of a measure as having been adopted under

¹⁰²⁶ Brazil's appellant's submission, para. 400.

¹⁰²¹ Panel Reports, para. 7.1080. (emphasis added) The Panel noted this to be the case "at least in respect of those alleged notifications that took place *subsequent* to the adoption of one or both transparency mechanisms". (Ibid. (emphasis original))

¹⁰²² Panel Reports, fn 1443 to para. 7.1080. (emphasis added) The Panel noted that "with respect to those alleged notifications that occurred *prior* to the adoption of the transparency mechanisms, it [was] sufficient for the Panel's purposes to note that its interpretation of paragraph 4(a) of the Enabling Clause is not contradicted by the Transparency Mechanisms." (Ibid. (emphasis original))

¹⁰²³ Brazil's appellant's submission, para. 400.

¹⁰²⁴ Brazil's appellant's submission, para. 400.

¹⁰²⁵ Brazil's appellant's submission, para. 400.

¹⁰²⁷ Fns omitted.

¹⁰²⁸ Brazil's appellant's submission, para. 400.

- 121 -

paragraph 2(c) would suffice for the purposes of paragraph 2(b) insofar as the complaining party's burden to raise and identify paragraph 2(b) of the Enabling Clause in its panel request is concerned. This would be the case in circumstances where, as we have explained and as the Panel also noted, it is "indicated in the notification itself".¹⁰²⁹ As we have explained, a notification pursuant to paragraph 4(a) speaks to and has a direct bearing on a complaining party's knowledge and, consequently, on its burden to raise the Enabling Clause and identify the relevant provision(s) thereof in its panel request.

5.382. In light of the foregoing considerations, we <u>uphold</u> the Panel's findings, in paragraphs 7.1082-7.1083 of the Panel Reports, that Brazil has not demonstrated that the differential tax treatment under the INOVAR-AUTO programme was notified to the WTO as adopted pursuant to paragraph 2(b), and therefore, in the circumstances of this case, the complaining parties were not required to raise and identify paragraph 2(b) of the Enabling Clause in their panel requests.

5.5.1.3 Whether the Panel erred in finding that the differential tax treatment under the INOVAR-AUTO programme was not notified pursuant to paragraph 4(a) as having been adopted under paragraph 2(c) of the Enabling Clause

5.383. We now turn to review whether the differential tax treatment under the INOVAR-AUTO programme was notified pursuant to paragraph 4(a) of the Enabling Clause as having been adopted under paragraph 2(c) of the Enabling Clause, such that the complaining parties could be considered to have been on notice and, consequently, had the burden to raise and identify paragraph 2(c) in their panel requests.

5.384. Before the Panel, Brazil contended that "the complainants were sufficiently 'on notice' that the challenged measure was adopted (and justified) under the Enabling Clause because the 1980 **Treaty of Montevideo** ... was notified to the WTO under paragraph 2(c) of the Enabling Clause."¹⁰³⁰ Brazil further submitted that the differential and more favourable treatment granted to Argentina, Mexico, and Uruguay (in the form of internal tax reductions) is based on ECAs that were negotiated under the auspices of the 1980 Treaty of Montevideo and that in turn "were also notified to the WTO".¹⁰³¹ Brazil contended that the differential tax treatment at issue under the INOVAR-AUTO programme is "a corollary of these ECAs, and thus did not require additional notification".¹⁰³²

5.385. The Panel stated that it should first decide "whether the notification of the [1980] Treaty of Montevideo and the ECAs could substantively serve as a notification of the adoption under paragraph **2(c)** ... of the arrangement introducing the differential and more favourable treatment (in the form of tax treatment) found to be inconsistent with Article I:1 of the GATT 1994".¹⁰³³ The Panel considered that "the differential and more favourable treatment sought to be justified under paragraph 2(c) must have a close and genuine link to the arrangement notified to the WTO such as to put other WTO Members on notice as to the adoption of the differential and more favourable treatment pursuant to the Enabling Clause."¹⁰³⁴

5.386. The differential tax treatment under the INOVAR-AUTO programme was not specifically notified to the WTO pursuant to paragraph 4(a) of the Enabling Clause as having been adopted under paragraph 2(c). In order to determine that the complaining parties could be considered to have been on notice that the differential tax treatment under the INOVAR-AUTO programme was taken pursuant to arrangements adopted under paragraph 2(c), the Panel needed to determine whether the notification of the 1980 Treaty of Montevideo and the ECAs could substantively serve as a

¹⁰²⁹ Panel Reports, para. 7.1079.

¹⁰³⁰ Brazil's appellant's submission, para. 395. (fn omitted)

¹⁰³¹ Brazil's appellant's submission, para. 395 (referring to Brazil's first written submission to the Panel, para. 742 (DS472)). The Panel, as we recall, noted that the ECAs were allegedly notified at various times between 1992 and 2013. (Panel Reports, fn 1443 to para. 7.1080)

¹⁰³² Brazil's appellant's submission, para. 395 (referring to Brazil's second written submission to the Panel, para. 181; first written submission to the Panel, paras. 638-639 (DS 472)).

¹⁰³³ Panel Reports, para. 7.1105.

¹⁰³⁴ Panel Reports, para. 7.1108. (emphasis omitted) As explained below in section 5.5.3 of these Reports, it suffices that the differential and more favourable treatment sought to be justified under paragraph 2(c) has a "genuine" link or a rational connection with the arrangement notified to the WTO. Therefore, we disagree with the Panel to the extent it considered that in order for any differential and more favourable treatment to be justified under paragraph 2(c), there must exist *both* a "close" and "genuine" link to the arrangement notified to the WTO.

- 122 -

notification of the adoption under paragraph 2(c) of the differential tax treatment under the INOVAR-AUTO programme found to be inconsistent with Article I:1 of the GATT 1994.¹⁰³⁵ In so doing, the Panel rightly considered the question to be whether Brazil had demonstrated that "the RTA notified to the WTO put the rest of the Membership on notice as to the adoption of the particular differential and more favourable treatment sought to be justified under paragraph 2(c)."¹⁰³⁶

5.387. We recall that Decree No. 7,819/2012 is the instrument that provides for the differential tax treatment under the INOVAR-AUTO programme, the measure at issue in this case.¹⁰³⁷ Articles 21 and 22(I) of Decree 7,819/2012, respectively, provide for a 30-percentage-point reduction of the IPI tax rates on certain categories of motor vehicles if:

a. Under Article 21, those motor vehicles are imported into Brazil by companies accredited as "domestic manufacturers" or "investors" under the INOVAR-AUTO programme, and the motor vehicles are imported from "countries that are signatories to the agreements established by Legislative Decree 350 of 21 November 1991, Decree 4,458 of 5 November 2002 and Decree 6[,]500 of 2 July 2008"; or

b. Under Article 22(I), those motor vehicles are imported into Brazil by any company (whether accredited or unaccredited) under the INOVAR-AUTO programme, and the motor vehicles are imported "under the agreement established by Decree 6,518 of 30 July 2008 and Decree 7,658 of 23 December 2011".¹⁰³⁸

5.388. Decree 350 of 21 November 1991, referred to in Article 21 of Decree 7,819/2012, provides for the implementation of the MERCOSUR Treaty for the formation of a common market between Argentina, Brazil, Paraguay, and Uruguay.¹⁰³⁹ Decree 4,458 of 5 November 2002, referred to in Article 21 of Decree 7,819/2012, provides for the implementation of the Economic Complementation Agreement No. 55 between MERCOSUR and Mexico (ECA No. 55).¹⁰⁴⁰ Decree 6,500 of 2 July 2008, referred to in Article 21 of Decree 7,819/2012, provides for the implementation Agreement No. 14 between Argentina and Brazil (ECA No. 14).¹⁰⁴¹ Decree 6,518 of 30 July 2008, referred to in Article 22(I) of Decree 7,819/2012, provides for the implementation of the 68th Additional Protocol to Economic Complementation Agreement No. 2 between Brazil and Uruguay (ECA No. 2).¹⁰⁴² Decree 7,658 of 23 December 2011, referred to in Article 22(I) of Decree 7,819/2012, lays down provisions concerning the implementation of the 69th Additional Protocol to Economic Complementation Agreement No. 2 between Brazil and Uruguay.¹⁰⁴³

5.389. Brazil submits that the 1980 Treaty of Montevideo and the ECAs "have an ample scope comprising internal tax reduction measures".¹⁰⁴⁴ Brazil refers to Articles 3(c), 3(e), and, in particular, Article 9(g) of the 1980 Treaty of Montevideo.¹⁰⁴⁵ Brazil asserts that Article 21 of

¹⁰⁴¹ Panel Reports, fn 246 to para. 2.102 (referring to Decree 6,500 of 2 July 2008 on the Economic Complementation Agreement No. 14 between the Republic of Argentina and the Federative Republic of Brazil (Decree 6,500/2008) (Panel Exhibit JE-165 (rev))).

¹⁰⁴² Panel Reports, fn 255 to para. 2.102 (referring to Decree 6,518 of 30 July 2008 providing for the implementation of the 68th Additional Protocol to Economic Complementation Agreement No. 2, signed between the Governments of the Federative Republic of Brazil and the Oriental Republic of Uruguay (Decree 6,518/2008) (Panel Exhibit JE-203 (rev))).

¹⁰⁴³ Panel Reports, fn 255 to para. 2.102 (referring to Decree 7,658 of 23 December 2011 laying down provisions concerning the implementation of the 69th Additional Protocol to Economic

Complementation Agreement No. 2 signed by the Governments of the Federal Republic of Brazil and the Oriental Republic of Uruguay (Decree 7,658/2011) (Panel Exhibit JE-204 (rev))).

¹⁰³⁵ Panel Reports, para. 7.1105.

¹⁰³⁶ Panel Reports, para. 7.1108.

¹⁰³⁷ Panel Reports, para. 2.100 (referring to Decree 7,819/2012 (Panel Exhibit JE-132)).

¹⁰³⁸ Panel Reports, para. 7.1014. (emphasis omitted; fns omitted)

¹⁰³⁹ Panel Reports, fn 246 to para. 2.102 (referring to Decree 350 of 21 November 1991 on the Treaty for the Formation of a Common Market between the Republic of Argentina, the Federative Republic of Brazil, the Republic of Paraguay and the Oriental Republic of Uruguay (MERCOSUR) (Degree 350/1991) (Panel Exhibit JE-163 (rev))).

¹⁰⁴⁰ Panel Reports, fn 246 to para. 2.102 (referring to Decree 4,458 of 5 November 2002 on the Economic Complementation Agreement No. 55 between MERCOSUR and the United Mexican States (Decree 4,458/2002) (Panel Exhibit JE-164 (rev))).

¹⁰⁴⁴ Brazil's appellant's submission, para. 422.

¹⁰⁴⁵ Brazil's appellant's submission, para. 422.

Decree 7,819/2012 "makes an explicit reference to these legal instruments"¹⁰⁴⁶, and therefore "the tax treatment at issue under [the] INOVAR-AUTO [programme] was a corollary of these ECAs, and thus did not require additional notification."¹⁰⁴⁷

5.390. In response, Japan submits that the 1980 Treaty of Montevideo "does not contemplate the specific preferential tax treatment conferred by Articles 21 and 22 of Decree 7,819".¹⁰⁴⁸ The European Union adds that the references to the three decrees in Article 21 of Decree 7,819/2012 "nowhere mention the provisions in those decrees referring to preferential tax treatment".¹⁰⁴⁹ Both the European Union and Japan further contend that Brazil makes reference to those decrees in Article 21 of Decree 7,819/2012 simply because it wants to avoid referring to the beneficiary countries directly by name.¹⁰⁵⁰

5.391. We recall that the Panel examined the provisions of the 1980 Treaty of Montevideo to determine if the internal tax reductions under the INOVAR-AUTO programme, i.e. the differential tax treatment at issue, had a genuine link to the 1980 Treaty of Montevideo that had been notified as adopted under paragraph 2(c) of the Enabling Clause. The Panel, in particular, reviewed Articles 3(c) and $3(e)^{1051}$, Article 9^{1052} , and Article 11^{1053} of the 1980 Treaty of Montevideo and, based on its

(c) Flexibility, characterized by the capacity to allow the conclusion of partial agreements, regulated in a manner compatible with the gradual achievement of their convergence and with the strengthening of ties of integration;

(e) Multiplicity, to make possible various types of agreement between member countries in harmony with the objectives and functions of the integration process, using all instruments capable of activating and expanding regional markets.

(Panel Reports, para. 7.1110 and fn 1472 thereto (quoting the 1980 Treaty of Montevideo (Panel Exhibit BRA-93), Article 3; referring to Brazil's response to Panel question No. 52))

¹⁰⁵² Article 9 of the 1980 Treaty of Montevideo provides:

Partial agreements shall be subject to the following general rules:

(a) They shall be open for accession, after prior negotiations, to the other member countries;
 (b) They shall contain clauses advocating convergence so that their benefits extend to all member countries;

(c) They may contain clauses advocating convergence with other Latin American countries, in conformity with the mechanisms established in this Treaty;

(d) They shall recommend different treatment for the three categories of countries recognized by this Treaty; each agreement shall specify the kind of treatment to be applied and negotiation procedures for its periodic revision at the request of any member country which considers itself at a disadvantage;

(e) Tariff reductions may be applied to the same products or tariff sub-items and on the basis of a percentage reduction in the tariffs on imports originating from non-participating countries; (f) They shall be applied for a minimum period of one year; and

(g) They may include, among others, specific rules regarding origin, safeguard clauses, non-tariff restrictions, withdrawal of concessions, renegotiation of concessions, denunciation, co-ordination and harmonization of policies. In the event that such specific rules have not been adopted, account will be taken of the provisions adopted by member countries on the matters in question.

(Panel Reports, para. 7.1110 and fn 1473 thereto (quoting the 1980 Treaty of Montevideo (Panel Exhibit BRA-93), Article 9; referring to Brazil's second written submission to the Panel, para. 179))

¹⁰⁵³ Article 11 of the 1980 Treaty of Montevideo provides:

The economic complementarity agreements shall be designed *inter alia* to promote the maximum utilization of the factors of production, to stimulate economic complementarity, to ensure equitable conditions of competition, to facilitate the competitiveness of products on the international market and to encourage the balanced and harmonious development of member countries.

These agreements shall be subject to the specific rules to be established for this purpose.

¹⁰⁴⁶ Brazil's appellant's submission, para. 382.

¹⁰⁴⁷ Brazil's appellant's submission, para. 395. (fn omitted) See also Panel Reports, para. 7.1072 (referring to Brazil's second written submission to the Panel, para. 181). We observe that the Panel noted that "Brazil maintains that the tax reductions under Article 21 of Decree 7,819/2012 'are not accorded because of the MERCOSUR Treaty'". (Panel Reports, para. 7.1031 (quoting Brazil's comments to other parties' responses to Panel question No. 52)) On appeal, Brazil does not take issue with this statement by the Panel.

¹⁰⁴⁸ Japan's appellee's submission, para. 160.

¹⁰⁴⁹ European Union's appellee's submission, para. 515.

 ¹⁰⁵⁰ European Union's appellee's submission, para. 515; Japan's appellee's submission, para. 175.
 ¹⁰⁵¹ Article 3 of the 1980 Treaty of Montevideo provides, in relevant part:

In the implementation of this Treaty and in the advance towards its final objective, the member countries shall bear in mind the following principles:

- 124 -

review, found that "none of the provisions cited to in the [1980] Treaty of Montevideo" had any relation "in and of themselves" to the differential tax treatment under the INOVAR-AUTO programme (in the form of internal tax reductions accorded to some but not other Members) found to be inconsistent with Article I:1 of the GATT 1994.¹⁰⁵⁴

5.392. Articles 21 and 22(I) of Decree 7,819/2012, which provide for the differential tax treatment under the INOVAR-AUTO programme at issue, do not refer to the 1980 Treaty of Montevideo. We observe that Article 4 of the 1980 Treaty of Montevideo provides for the conclusion of partial agreements, which in turn, pursuant to Article 9(g), "may include ... specific rules regarding ... non-tariff restrictions". Article 3 of the 1980 Treaty of Montevideo enumerates the "principles" that signatories shall bear in mind and includes in subparagraph (e) the principle of "[m]ultiplicity", pursuant to which signatories are "to make possible various types of agreement between member countries in harmony with the objectives and functions of the integration process, using all instruments capable of activating and expanding regional markets". The 1980 Treaty of Montevideo thus does not itself specify any rules regarding internal tax reductions. We therefore see no reason to disagree with the Panel's finding that "none of the provisions cited to in the [1980] Treaty of Montevideo" have any relation, "in and of themselves", to the differential tax treatment under the INOVAR-AUTO programme (in the form of internal tax reductions accorded to some but not other Members) found to be inconsistent with Article I:1 of the GATT 1994.¹⁰⁵⁵

5.393. That said, we note that Articles 21 and 22(I) of Decree 7,819/2012 fall under Chapter VII thereof, entitled "Tax Rates and Suspension of IPI". As noted, Articles 21 and 22(I) of Decree 7,819/2012 provide for a reduction of the IPI tax rates on certain categories of motor vehicles when they are imported into Brazil from the countries that are signatories to the above-mentioned ECAs. The reference to the ECAs in Articles 21 and 22(I) of Decree 7,819/2012, in our view, is limited to identifying the countries, imports from which benefit from the IPI tax reduction. These ECAs, in turn, do not, however, in and of themselves, refer to internal taxation. In this regard, we recall that the Panel reviewed the provisions of the relevant ECAs to determine if the internal tax reductions (in the form of IPI tax reductions) under the INOVAR-AUTO programme had a genuine link to these ECAs that had been notified as adopted under paragraph 2(c) of the Enabling Clause.

5.394. The Panel noted that "Brazil has not pointed to a single provision of any ECA" that would attest to "the fundamental premise of Brazil's argument, namely that the INOVAR-AUTO programme is implementing the objectives of the ECAs".¹⁰⁵⁶ Based on its review, the Panel found that it "could not discern any such relationship". 1057

5.395. We observe that, in reaching this conclusion, the Panel noted that, while "ECA No. 55, between MERCOSUR and Mexico, indicates that its objective is to 'lay the foundations for the establishment of free trade in the motor vehicle sector and to promote the productive integration and complementation of their respective motor vehicle sectors'", the definition of "free trade" is limited to "tariff reductions", and the term "tariff" is defined as concerning "taxes or charges 'in connection with importation of goods'".¹⁰⁵⁸ With respect to the 38th Additional Protocol to ECA No. 14 between Argentina and Brazil, the Panel noted that this Additional Protocol indicates that "automotive goods shall be placed on the market between the parties with a 100% preferential tariff (0% of ad valorem tariff within the area) provided that the origin requirements and the conditions laid down in the Agreements have been met."¹⁰⁵⁹ Similarly, we observe that with respect to the 68th Additional Protocol to ECA No. 2 between Brazil and Uruguay, the Panel noted that this Additional Protocol indicates that "motor vehicle products shall be sold between the Parties with 100% (one hundred) preference (0% (zero) 'ad valorem' intra-zone tariff), whenever they meet the

⁽Panel Reports, para. 7.1110 and fn 1474 thereto (quoting the 1980 Treaty of Montevideo (Panel Exhibit BRA-93), Article 11; referring to Brazil's first written submissions to the Panel, para. 706 (DS472) and para. 638 (DS497)))

¹⁰⁵⁴ Panel Reports, para. 7.1112.

¹⁰⁵⁵ Panel Reports, para. 7.1112.

¹⁰⁵⁶ Panel Reports, para. 7.1112.

¹⁰⁵⁷ Panel Reports, para. 7.1112.

¹⁰⁵⁸ Panel Reports, para. 7.1113 (quoting Decree 4,458/2002 (Panel Exhibit JE-164 (rev)), Articles 1 and 2). ¹⁰⁵⁹ Panel Reports, para. 7.1113 (quoting Decree 6,500/2008 (Panel Exhibit JE-165 (rev)), Article 9).

- 125 -

requirements of origin and the conditions set out in this Agreement."¹⁰⁶⁰ The Panel thus found that **"these provisions are indicative of tariff preferences ...** [and] do not refer to internal taxation."¹⁰⁶¹

5.396. On appeal, Brazil has not identified any provisions of the ECAs that would support its **contention that "the [differential] tax treatment ... under [the] INOVAR**-AUTO [programme is] a corollary of these ECAs."¹⁰⁶² Neither do we find Brazil to have demonstrated that there is a rational connection between the IPI tax reduction provided under the INOVAR-AUTO programme and the above-mentioned ECAs, save to the extent of identifying the countries that are signatories to these ECAs. To the contrary, the above-mentioned ECAs, in our view (and as also noted by the Panel), provide for the adoption of tariff preferences in the automotive sector and do not refer to internal taxation and therefore have no genuine link to the differential tax treatment in the form of IPI tax reductions under the INOVAR-AUTO programme according the differential and more favourable treatment in the form of IPI tax reductions could be considered to have been notified as having been adopted under paragraph 2(c), pursuant to paragraph 4(a), of the Enabling Clause. Accordingly, we **agree with the Panel that "Brazil has not demonstrated how the ... tax reductions found to be** inconsistent under Article I: 1 of the GATT 1994 are related to the RTA that Brazil has notified to the WTO (the [1980] Treaty of Montevideo) or the ECAs allegedly implementing that RTA."¹⁰⁶³

5.397. In light of the foregoing considerations, we <u>uphold</u> the Panel's finding, in paragraph 7.1119 of the Panel Reports, that the differential and favourable treatment (i.e. the differential tax treatment in the form of internal tax reductions) under the INOVAR-AUTO programme was not notified as adopted under paragraph 2(c) of the Enabling Clause, as required pursuant to paragraph 4(a). Consequently, we also <u>uphold</u> the Panel's finding, in paragraph 7.1120 of the Panel Reports, that, in the circumstances of this case, there was no burden on the complaining parties to raise and identify paragraph 2(c) of the Enabling Clause in their panel requests.

5.5.2 Whether the Panel erred in its interpretation of paragraph 2(b) of the Enabling Clause and in finding that the differential tax treatment under the INOVAR-AUTO programme was not justified under that provision

5.398. Having upheld the Panel's finding that the complaining parties' claims were within its terms of reference, we now turn to address Brazil's claim on appeal that the Panel erred in its interpretation and application of the Enabling Clause in the present dispute, starting with paragraph 2(b) of the Enabling Clause.¹⁰⁶⁴ In so doing, we first examine the scope of paragraph 2(b), particularly the phrase "non-tariff measures governed by the provisions of instruments multilaterally negotiated under the auspices of the GATT", following which we review the Panel's finding of whether the differential tax treatment under the INOVAR-AUTO programme falls within the scope of that provision.

5.399. Before the Panel, Brazil argued that "the tax reductions challenged by the complaining parties fall within the scope of paragraph 2(b)."¹⁰⁶⁵ Brazil submitted that the internal tax reductions at issue "are non-tariff measures (NTMs) because they constitute internal taxes subject to Article III of the GATT 1994 and, consequently, are subject to the [most-favoured nation] MFN obligation under Article I:1 of the GATT 1994".¹⁰⁶⁶ Brazil further submitted that "internal taxes are NTMs 'governed by the provisions of instruments multilaterally negotiated under the auspices of the GATT' because the GATT 1947 (and subsequently the GATT 1994) are relevant multilaterally-negotiated instruments covering internal taxation, and there is no specific agreement covering internal taxes."¹⁰⁶⁷

¹⁰⁶⁰ Panel Reports, para. 7.1113 (quoting Decree 6,518/2008 (Panel Exhibit JE-203 (rev)), Article 3).

¹⁰⁶¹ Panel Reports, para. 7.1114.

¹⁰⁶² Brazil's appellant's submission, para. 395.

¹⁰⁶³ Panel Reports, para. 7.1115.

¹⁰⁶⁴ Brazil's appellant's submission, paras. 387-388.

¹⁰⁶⁵ Panel Reports, para. 7.1086.

¹⁰⁶⁶ Panel Reports, para. 7.1086 (referring to Brazil's first written submissions to the Panel,

paras. 708-709 and 729 (DS472) and paras. 640-641 and 661 (DS497); second written submission to the Panel, paras. 167-171).

¹⁰⁶⁷ Panel Reports, para. 7.1086 (quoting Brazil's second written submission to the Panel, paras. 167-175).

- 126 -

5.400. The Panel noted that the issue before it included "whether internal taxes are 'non-tariff measures governed by the provisions of instruments multilaterally negotiated under the auspices of the GATT'".¹⁰⁶⁸ The Panel, however, did not consider it necessary to define the term "non-tariff measures" in isolation from the rest of paragraph 2(b)¹⁰⁶⁹, and considered instead the question of "[w]hether the alleged non-**tariff measures at issue ... have been demonstrated to be w**ithin the scope of [paragraph] 2(b)".¹⁰⁷⁰ Accordingly, the Panel proceeded to assess whether paragraph 2(b) applies to non-tariff measures governed exclusively by those provisions of the GATT 1994 that were incorporated from the GATT 1947.¹⁰⁷¹

5.401. The Panel considered that paragraph 2(b), in referring to "[d]ifferential and more favourable treatment with respect to the provisions of the General Agreement concerning *non-tariff measures governed by the provisions of instruments multilaterally negotiated under the auspices of the GATT*{}" at the time the Enabling Clause was adopted, "meant non-tariff measures *other than* those non-tariff measures governed exclusively by the provisions of the GATT 1947".¹⁰⁷² Recalling that the Enabling Clause was adopted by the GATT Contracting Parties during the Tokyo Round when a number of plurilateral agreements covering certain non-tariff measures were concluded¹⁰⁷³, the Panel considered that "the intended application of paragraph 2(b) must have been limited to the discrimination explicitly provided for in specific [special and differential (S&D)] provisions of the Tokyo Round Codes."¹⁰⁷⁴ The Panel noted that the provisions of the GATT 1994 that Brazil relied on, namely Articles III: 2 and III: 4, "do not introduce any special and differential treatment for taxes in the form of non-tariff measures" and "are substantively identical to provisions in the GATT 1947".¹⁰⁷⁵ Accordingly, the Panel concluded that "a non-tariff measure within the scope of paragraph 2(b) must be governed by specific provisions on special and differential treatment[] that are distinct from the provisions of the GATT 1994 incorporating the GATT 1947.^{"1076}

5.402. For these reasons, the Panel found that "the tax reductions accorded to imported products from Argentina, Mexico and Uruguay and found to be inconsistent under Article I: 1 of the GATT 1994 are not justified under paragraph 2(b) of the Enabling Clause."¹⁰⁷⁷

5.403. On appeal, Brazil disagrees that "instruments multilaterally negotiated under the auspices of the GATT" must be "'distinct from the provisions of the GATT 1994 incorporating the GATT 1947' because the GATT 1994 itself is an instrument which was multilaterally negotiated under the auspices of the GATT (institution)".¹⁰⁷⁸ Brazil argues that "[t]he GATT 1994 is the covered agreement that governs internal taxation, in Article III" and the Enabling Clause itself "as it was incorporated to the WTO as part of the GATT 1994, is, therefore, an 'instrument multilaterally negotiated under the auspices of the GATT.'"¹⁰⁷⁹ Brazil contends that the Panel interpreted paragraph 2(b) to apply "only to specific Special and Differential (S&D) provisions present in the covered agreements other than the GATT itself".¹⁰⁸⁰ Brazil submits that, according to the Panel's reasoning, if the Enabling Clause is incorporated in the GATT 1994 but only applies to provisions outside of the GATT, "it also follows that the S&D provisions prevail over the GATT and therefore the Enabling Clause

¹⁰⁶⁸ Panel Reports, para. 7.1088.

¹⁰⁶⁹ Panel Reports, para. 7.1088. The Panel further considered that paragraph 2(b) when read as a whole does not refer to "non-tariff measures" generally, but rather refers to a limited category of non-tariff measures, namely "non-tariff measures governed by the provisions of instruments multilaterally negotiated under the auspices of the GATT". Therefore, the Panel considered that "[i]t would be unnecessary and **inappropriate ... to interpret only one part of** paragraph 2(b) without reference to the rest of the provision." (Ibid., fn 1449 thereto) We note that, to the extent that the Panel chose not to define the term "non-tariff measures" and determine whether internal taxes are "non-tariff measures governed by the provisions of instruments multilaterally negotiated under the auspices of the GATT", on appeal, none of the participants take issue with the Panel's approach.

¹⁰⁷⁰ Panel Reports, fn 1449 to para. 7.1088.

¹⁰⁷¹ Panel Reports, para. 7.1088.

¹⁰⁷² Panel Reports, para. 7.1089. (emphasis original; curly brackets added)

¹⁰⁷³ Panel Reports, para. 7.1092.

¹⁰⁷⁴ Panel Reports, para. 7.1093.

¹⁰⁷⁵ Panel Reports, para. 7.1096. The Panel further noted that Brazil conceded that other than the GATT 1994 there is no specific covered agreement dealing with internal taxation. (Ibid.)

¹⁰⁷⁶ Panel Reports, para. 7.1096. (fn omitted)

¹⁰⁷⁷ Panel Reports, para. 7.1097.

¹⁰⁷⁸ Brazil's appellant's submission, para. 410 (quoting Panel Reports, para. 7.1096).

¹⁰⁷⁹ Brazil's appellant's submission, para. 410.

¹⁰⁸⁰ Brazil's appellant's submission, para. 408 (referring to Panel Reports, para. 7.1096).

- 127 -

itself, according to [the general interpretative note to] Annex 1A of the Marrakesh Agreement."1081 Brazil therefore asserts that "[t]he Panel's interpretation renders the text of paragraph 2(b) of the Enabling Clause *inutile*."¹⁰⁸² Brazil therefore seeks reversal of the Panel's conclusion that a non-tariff measure within the scope of paragraph 2(b) must be governed by specific provisions on S&D treatment that are distinct from the provisions of the GATT 1994 incorporating the GATT 1947.¹⁰⁸³ Brazil also seeks reversal of the Panel's "consequential finding" that the internal tax reductions accorded to imported products from Argentina, Mexico, and Uruguay and found to be inconsistent under Article I:1 of the GATT 1994 are not justified under paragraph 2(b) of the Enabling Clause¹⁰⁸⁴ and requests the Appellate Body to complete the analysis and find that "the differential tax treatment is justified under paragraph 2(b) and complies with the requirements of paragraph 3 of the Enabling Clause." 1085

5.404. In response, the European Union asserts that "Brazil's over-creative reading", according to which paragraph 2(b) refers to "all the provisions of the GATT relating to non-tariff measures as being negotiated under the auspices of the GATT (or of the WTO), does not find any support in the text, context or the object and purpose of the Enabling Clause". ¹⁰⁸⁶ The European Union points to the Panel's explanation that the Enabling Clause was adopted by the GATT CONTRACTING PARTIES during the Tokyo Round in the context of which a number of plurilateral agreements governing certain non-tariff measures were concluded.¹⁰⁸⁷ The European Union submits that it is in this context that the reference in paragraph 2(b) to "instruments multilaterally negotiated under the auspices of the GATT" should be understood. 1088

5.405. Japan submits that paragraph 2(b) "does not endorse exceptions to the MFN principle with respect to 'non-tariff measures' themselves" but instead "pertains to differential and more favourable treatment with respect to 'the provisions of the General Agreement concerning non-tariff measures governed by the provisions of instruments multilaterally negotiated under the auspices of the GATT".¹⁰⁸⁹ According to Japan, the Panel carefully interpreted those terms and confirmed that they mean "since 'Article III: 2 and III: 4 of the GATT 1994 [] do not introduce any special and differential treatment for taxes in the form of non-tariff measures'" and "there is no specific WTO Agreement dealing with internal taxation, Brazil's measures cannot be substantially covered by paragraph 2(b)."1090

5.406. We begin by analysing the scope of the phrase "non-tariff measures governed by the provisions of instruments multilaterally negotiated under the auspices of the GATT" as it appears in paragraph 2(b) of the Enabling Clause. This paragraph provides, in relevant part:

2. The provisions of paragraph 1 apply to the following:

...

(b) Differential and more favourable treatment with respect to the provisions of the General Agreement concerning non-tariff measures governed by the provisions of instruments multilaterally negotiated under the auspices of the GATT[.]¹⁰⁹¹

5.407. Paragraph 2(b) thus identifies a certain form of differential and more favourable treatment to which the authorization of paragraph 1 of the Enabling Clause applies. In other words, a measure that a Member claims to be excepted from a finding of inconsistency with Article I of the

and 8.17.j).

¹⁰⁸¹ Brazil's appellant's submission, para. 412. (fn omitted)

¹⁰⁸² Brazil's appellant's submission, para. 412. (emphasis original)

¹⁰⁸³ Brazil's appellant's submission, para. 431 (referring to Panel Reports, para. 7.1096).

¹⁰⁸⁴ Brazil's appellant's submission, para. 431 (referring to Panel Reports, paras. 7.1097, 8.6.i,

¹⁰⁸⁵ Brazil's appellant's submission, para. 432.

¹⁰⁸⁶ European Union's appellee's submission, para. 496.

¹⁰⁸⁷ European Union's appellee's submission, para. 494.

¹⁰⁸⁸ European Union's appellee's submission, para. 494.

¹⁰⁸⁹ Japan's appellee's submission, para. 168 (quoting paragraph 2(b) of the Enabling Clause). (emphasis added by Japan) 1090 Japan's appellee's submission, para. 168 (quoting Panel Reports, para. 7.1096; referring to

para. 7.1097). ¹⁰⁹¹ Fn omitted.

- 128 -

"General Agreement" "must fit within"¹⁰⁹² the meaning of paragraph 2(b). To this effect, paragraph 2(b) provides for the adoption of a limited category of differential and more favourable treatment, namely treatment that concerns "non-tariff measures governed by the provisions of instruments multilaterally negotiated under the auspices of the GATT [as an institution]". The text of paragraph 2(b) does not, however, support a reading of that provision as extending to the adoption of differential and more favourable treatment concerning non-tariff measures governed by "provisions of the General Agreement" itself. Indeed, had it been so, the latter part of paragraph 2(b) in referring to "provisions of instruments multilaterally negotiated under the auspices of the GATT" would be deprived of any meaning.

5.408. We find support for this reading from the contextual history surrounding the adoption of the Enabling Clause. We recall that the Enabling Clause was adopted in 1979 during the Tokyo Round of multilateral trade negotiations, which also witnessed the conclusion of a number of plurilateral agreements governing various non-tariff measures, i.e. the Tokyo Round Codes.¹⁰⁹³ The Tokyo Round Codes were "negotiated under the auspices of the GATT [as an institution]". We observe that a number of these plurilateral agreements sought to further the objectives of and/or build upon existing provisions of the GATT 1947, and contained provisions on S&D treatment for developing countries. The reference in paragraph 2(b) to differential and more favourable treatment "with respect to the provisions of the General Agreement concerning non-tariff measures governed by the provisions of instruments multilaterally negotiated under the auspices of the GATT" was in relation to these plurilateral agreements that were negotiated under the auspices of the GATT, as an institution and furthered the objectives of and/or built upon existing provisions of the GATT 1947. Moreover, in using the phrase "provisions of instruments multilaterally negotiated under the auspices of the GATT"¹⁰⁹⁴, as opposed to "instruments multilaterally negotiated under the auspices of the GATT", paragraph 2(b) referred to specific provisions of these plurilateral agreements, in particular, the S&D treatment provisions, and not the entire agreements themselves.

5.409. We find additional support from contemporaneous decisions adopted during the Tokyo Round of multilateral trade negotiations. In particular, we recall the decision entitled "Action by the CONTRACTING PARTIES on the Multilateral Trade Negotiations"¹⁰⁹⁵, which recognized in paragraph 2 thereof that "as a result of the Multilateral Trade Negotiations, a number of Agreements covering certain non-**tariff measures** ... have been drawn up."¹⁰⁹⁶ We observe that paragraph 1 of that decision provided that the CONTRACTING PARTIES "reaffirm their intention to ensure the unity and consistency of the GATT system, and to this end they shall oversee the operation of the system as a whole and take action as appropriate".¹⁰⁹⁷ Paragraph 3, in particular, stated that "[t]he CONTRACTING PARTIES also note that existing rights and benefits under the GATT of contracting parties not being parties to these Agreements, including those derived from Article I, are not affected by these Agreements."¹⁰⁹⁸

5.410. In other words, the GATT CONTRACTING PARTIES addressed the issue of MFN treatment arising out of Article I of the GATT 1947 by reaffirming "their intention to ensure the unity and consistency of the GATT system" and expressly confirming that the benefits of the Tokyo Round plurilateral agreements were to accrue to all the contracting parties to the GATT, even those that were not parties to the plurilateral agreements, insofar as the subject matter of those agreements were covered by Article I of the GATT 1947. Therefore, at the time of the conclusion of the Tokyo

¹⁰⁹⁴ Emphasis added.

¹⁰⁹² Appellate Body Report, EC - Tariff Preferences, para. 112.

¹⁰⁹³ The Tokyo Round witnessed the conclusion of (i) the Agreement on Technical Barriers to Trade (TBT Agreement); (ii) the Agreement on Government Procurement; (iii) the Agreement on Interpretation and Application of Articles VI, XVI, and XXIII of the General Agreement on Tariffs and Trade (Subsidies and Countervailing Duties); (iv) the Agreement on Implementation of Article VII of the General Agreement on Tariffs and Trade (Customs Valuation Agreement); (v) the Agreement on Import Licensing Procedures; (vi) the Agreement on Implementation of Article VI of the General Agreement on Tariffs and Trade (Anti-Dumping Agreement); (vii) the Agreement on Trade in Civil Aircraft; (viii) the International Dairy Agreement; and (ix) the International Bovine Meat Agreement.

¹⁰⁹⁵ Action by the Contracting Parties on the Multilateral Trade Negotiations, Decision by the Contracting Parties of 28 November 1979, GATT Document L/4905, BISD 26S/201.

¹⁰⁹⁶ Action by the Contracting Parties on the Multilateral Trade Negotiations, Decision by the Contracting Parties of 28 November 1979, GATT Document L/4905, BISD 26S/201, para. 2.

¹⁰⁹⁷ Action by the Contracting Parties on the Multilateral Trade Negotiations, Decision by the Contracting Parties of 28 November 1979, GATT Document L/4905, BISD 26S/201, para. 1.

¹⁰⁹⁸ Action by the Contracting Parties on the Multilateral Trade Negotiations, Decision by the Contracting Parties of 28 November 1979, GATT Document L/4905, BISD 26S/201, para. 3.

- 129 -

Round Codes, absent the Enabling Clause, a Contracting Party who was not a party to a Tokyo Round plurilateral agreement could have challenged a measure taken by a party to that plurilateral agreement pursuant to a S&D treatment provision thereof in favour of a developing country as being inconsistent with Article I of the GATT 1947.

5.411. The adoption of the Enabling Clause, particularly paragraph 2(b), addressed this situation. Paragraph 2(b) provided an umbrella by excepting differential and more favourable treatment concerning non-tariff measures governed by the provisions of instruments multilaterally negotiated under the auspices of the GATT, i.e. differential and more favourable treatment accorded pursuant to the S&D treatment provisions of the Tokyo Round Codes, from the purview of a challenge under Article I of the GATT 1947.

5.412. The foregoing considerations therefore suggest that the phrase "non-tariff measures governed by the provisions of instruments multilaterally negotiated under the auspices of the GATT" in paragraph 2(b), at the time of the adoption of the Enabling Clause, concerned non-tariff measures taken pursuant to the S&D treatment provisions of the Tokyo Round Codes and not the provisions of the GATT 1947.

5.413. We note that with the entry into effect of the WTO Agreement, the Tokyo Round Codes are no longer in force. The Enabling Clause, however, stands incorporated as an "integral part" of the GATT 1994.¹⁰⁹⁹ The Appellate Body considered in *EC* – *Tariff Preferences* that "Members reaffirmed the significance of the Enabling Clause ... with [its] incorporation ... into the GATT 1994."1100 The Uruguay Round of multilateral trade negotiations culminated in the establishment of the WTO, following which GATT as an institution was replaced by the WTO. Article II: 1 of the WTO Agreement expressly recognizes that "[t]he WTO shall provide the common institutional framework for the conduct of trade relations among its Members in matters related to the agreements and associated legal instruments included in the Annexes to [the WTO] Agreement".¹¹⁰¹ The Enabling Clause as an "integral part" of the GATT 1994 falls within the scope of Article II:1 of the WTO Agreement. Therefore, while at the time of its adoption, paragraph 2(b) of the Enabling Clause speaks of "instruments multilaterally negotiated under the auspices of the GATT" as an institution, following the entry into force of the WTO Agreement, paragraph 2(b) refers to "instruments multilaterally negotiated under the auspices of the [WTO]" as an institution.¹¹⁰² Paragraph 2(b) of the Enabling Clause, following the entry into force of the WTO Agreement, thus provides for the adoption of a limited category of differential and more favourable treatment, namely treatment that concerns non-tariff measures governed by provisions of instruments multilaterally negotiated under the auspices of the WTO. The GATT 1994, while an integral part of the WTO Agreement, was not negotiated under the auspices of the WTO as an institution.

5.414. These considerations, read in light of the text, context, and circumstances surrounding the adoption of the Enabling Clause and thereafter the establishment of the WTO, indicate that paragraph 2(b) does not concern non-tariff measures governed by the provisions of the GATT 1994. Instead, paragraph 2(b) speaks to non-tariff measures taken pursuant to S&D treatment provisions of "instruments multilaterally negotiated under the auspices of the [WTO]".¹¹⁰³ Brazil's contention that paragraph 2(b) applies to non-tariff measures taken pursuant to the provisions of the GATT 1994 incorporating the GATT 1947¹¹⁰⁴, in our view, calls for paragraph 2(b) to be given a meaning that was not ascribed to it either at the time of its adoption or thereafter with the establishment of the WTO. We therefore uphold the Panel's finding, in paragraph 7.1096 of the

 ¹⁰⁹⁹ Appellate Body Report, *EC – Tariff Preferences*, para. 90.
 ¹¹⁰⁰ Appellate Body Report, *EC – Tariff Preferences*, para. 108 (referring to paragraph 1(b)(iv) of the language of Annex 1A to the WTO Agreement incorporating the GATT 1994 into the WTO Agreement).

¹⁰¹ Article II:1 of the WTO Agreement. We also note that Article III:2 of the WTO Agreement states in relevant part that "[t]he WTO shall provide the forum for negotiations among its Members concerning their multilateral trade relations in matters dealt with under the agreements in the Annexes to [the WTO] Agreement." (Article III: 2 of the WTO Agreement)

¹¹⁰² We note Brazil's contention that the terms "under the auspices of the GATT" in paragraph 2(b) of the Enabling Clause, "at present, should be read 'under the auspices of the WTO'". (Brazil's appellant's submission, para. 410 (quoting Brazil's second written submission to the Panel, para. 174))

¹¹⁰³ This reading of paragraph 2(b) of the Enabling Clause does not, however, affect the S&D treatment provisions contained in the other multilateral agreements on trade in goods in Annex 1A to the WTO Agreement. The S&D treatment provisions of these multilateral agreements on trade in goods apply in their own right in the context of their respective agreements as among all WTO Members.

¹¹⁰⁴ Brazil's appellant's submission, para. 410.

Panel Reports, that "a non-tariff measure within the scope of paragraph 2(b) must be governed by specific provisions on special and differential treatment, that are distinct from the provisions of the GATT 1994 incorporating the GATT 1947."¹¹⁰⁵

5.415. Turning to the Panel's application of paragraph 2(b) of the Enabling Clause, we recall that the Panel found that the provisions of the GATT 1994 relied on by Brazil, namely Articles III:2 and III:4, "do not introduce any special and differential treatment for taxes in the form of non-tariff measures".¹¹⁰⁶ We have considered above that paragraph 2(b) of the Enabling Clause did not apply, at the time of its adoption, with respect to Articles III:2 and III:4 of the GATT 1947, and following the entry into force of the WTO Agreement, does not apply with respect to Articles III:2 and III:4 of the GATT 1947. Thus, subjecting like products of different WTO Members to different internal taxes inconsistently with Article I:1 of the GATT 1994 cannot be justified under paragraph 2(b) of the Enabling Clause. We therefore <u>uphold</u> the Panel's finding, in paragraph 7.1097 of the Panel Reports, that the internal tax reductions under the INOVAR-AUTO programme accorded to imported products from Argentina, Mexico, and Uruguay and found to be inconsistent under Article I:1 of the GATT 1994 are not justified under paragraph 2(b) of the Enabling Clause.¹¹⁰⁷

5.5.3 Whether the Panel erred in its interpretation of paragraph 2(c) of the Enabling Clause and in finding that the differential tax treatment under the INOVAR-AUTO programme was not justified under that provision

5.416. We now turn to paragraph 2(c) of the Enabling Clause. We recall that we have agreed with the Panel to the extent the Panel found that there is no genuine link or a rational connection between differential tax treatment under the INOVAR-AUTO programme and the notified arrangements adopted under paragraph 2(c). Consequently, we have upheld the Panel's finding that the differential and favourable treatment (i.e. the differential tax treatment in the form of internal tax reductions accorded to some but not other Members) under the INOVAR-AUTO programme was not notified as adopted under paragraph 2(c), as required pursuant to paragraph 4(a) of the Enabling Clause. In this section, we review whether the Panel erred in finding that the differential tax treatment under the INOVAR-AUTO programme was not substantively justified under paragraph 2(c).

5.417. The Panel recalled that, to satisfy the notification requirement in paragraph 4(a), any differential and more favourable treatment adopted under paragraph 2(c) must have "a close and genuine link to an RTA sufficient to alert other WTO Members to the adoption of such differential and more favourable treatment pursuant to the Enabling Clause".¹¹⁰⁸ The Panel considered that a "similar standard" applies with respect to the substantive justification under paragraph 2(c) itself.¹¹⁰⁹ Thus, the Panel found that, "[i]n order for any differential and more favourable treatment to be justified under paragraph 2(c) of the Enabling Clause, there must exist a close and genuine link to a 'regional arrangement entered into amongst less-developed contracting parties'."¹¹¹⁰ In this case, the Panel found that Brazil had not identified an RTA with a close and genuine link to the internal tax reductions at issue.¹¹¹¹ The Panel stated that, while "Brazil has made assertions regarding the [1980] Treaty of Montevideo and the ECAs", Brazil has not pointed to a provision providing for tax preferences in those RTAs nor "demonstrated how the tax reductions at issue are related to those RTAs", and therefore how the relevant differential and more favourable treatment could be justified under paragraph 2(c).¹¹¹² The Panel therefore found that "Brazil has not met its burden of proof in respect of the substantive requirements of paragraph 2(c)."¹¹¹³

¹¹⁰⁵ Panel Reports, para. 7.1096. (fn omitted)

¹¹⁰⁶ Panel Reports, para. 7.1096. In particular, the Panel noted that "Brazil concedes that other than the GATT 1994 there is no specific WTO covered agreement dealing with internal taxation." (Ibid. (referring to Brazil's second written submission to the Panel, para. 174))

¹¹⁰⁷ Panel Reports, para. 7.1097.

¹¹⁰⁸ Panel Reports, para. 7.1117 (referring to ibid., para. 7.1108).

¹¹⁰⁹ Panel Reports, para. 7.1117.

¹¹¹⁰ Panel Reports, para. 7.1117.

¹¹¹¹ Panel Reports, para. 7.1118.

¹¹¹² Panel Reports, para. 7.1118. (fn omitted)

¹¹¹³ Panel Reports, para. 7.1118.

- 131 -

5.418. For these reasons, the Panel concluded that "the tax reductions accorded to imported products from Argentina, Mexico and Uruguay and found to be inconsistent under Article I:1 of the GATT 1994 are not justified under paragraph 2(c) of the Enabling Clause."1114

5.419. On appeal, Brazil contends that the Panel's substantive evaluation of paragraph 2(c) "is essentially indistinguishable from its evaluation of whether the notification was sufficient under paragraph 4(a)".¹¹¹⁵ Brazil submits that the Panel rested its finding on its "flawed conclusion that because the [1980] Treaty of Montevideo and the provisions of the relevant ECAs do not expressly make reference to internal taxation, they did not have a genuine link with paragraph 2(c)" and therefore "WTO Members could not have been expected to be informed that Brazil intended to accord internal tax reductions to motor vehicles from Argentina, Mexico and Uruguay".¹¹¹⁶ Brazil further submits that the Panel's finding that "Brazil has not demonstrated 'how the tax reductions at issue are related to [the] RTAs'"¹¹¹⁷ is "directly contradicted by the facts on the record".¹¹¹⁸ Brazil asserts that the 1980 Treaty of Montevideo, the relevant ECAs, and LAIA have "an ample scope comprising internal tax reduction measures".¹¹¹⁹ Accordingly, Brazil requests the Appellate Body to reverse the Panel's conclusion that Brazil did not meet its burden of proof with respect to the substantive requirements of paragraph 2(c)¹¹²⁰ and to complete the analysis and find that "the differential tax treatment is justified under paragraph 2(c) and complies with the requirements of paragraph 3 of the Enabling Clause."1121

5.420. In response, the European Union asserts that what Brazil "in essence" claims in its appeal is that "the Panel failed to make an objective assessment of the matter before it, while Brazil has not raised a claim under Article 11 of DSU."¹¹²² Thus, the European Union submits that "the Appellate Body can and should ... dismiss this part of Brazil's appeal."¹¹²³ That said, the European Union disagrees with Brazil's contention that Article 21 of Decree 7,819/2012 contains the necessary references and argues instead that the article does not mention the provisions in the relevant decrees referring to preferential tax treatment.1124

5.421. Japan submits that "Brazil's claim appears to be that the Panel did not properly analyse and weigh the facts."¹¹²⁵ Japan contends that "[s]uch a claim ... cannot be raised by Brazil at this stage (at least not under any other basis than potentially a DSU Article 11 claim, which Brazil did not raise)."1126 That said, Japan submits that the Panel reached the correct conclusion concerning paragraph 2(c) of the Enabling Clause.¹¹²⁷ Japan submits that Article 21 of Decree 7,819/2012 refers to "vehicles [...] originating from countries that are signatories to the agreements" established by the relevant decrees.¹¹²⁸ According to Japan, "[t]he references to the relevant decrees do not identify the provisions in those decrees referring to preferential tax treatment."¹¹²⁹ Japan contends that "Brazil makes reference to those decrees ... because it wants to avoid referring to the beneficiary countries directly by name."¹¹³⁰

¹¹¹⁴ Panel Reports, para. 7.1121.

¹¹¹⁵ Brazil's appellant's submission, para. 414.

¹¹¹⁶ Brazil's appellant's submission, para. 419 (referring to Panel Reports, para. 7.1115).

¹¹¹⁷ Brazil's appellant's submission, para. 421 (quoting Panel Reports, para. 7.1118).

¹¹¹⁸ Brazil's appellant's submission, para. 422

¹¹¹⁹ Brazil's appellant's submission, para. 422.

¹¹²⁰ Brazil's appellant's submission, para. 433 (referring to Panel Reports, para. 7.1118).

¹¹²¹ Brazil's appellant's submission, para. 434.

¹¹²² European Union's appellee's submission, para. 508. (fn omitted)

¹¹²³ European Union's appellee's submission, para. 508.

¹¹²⁴ European Union's appellee's submission, para. 515.

¹¹²⁵ Japan's appellee's submission, para. 171.

¹¹²⁶ Japan's appellee's submission, para. 171.

¹¹²⁷ Japan's appellee's submission, paras. 172-174.

¹¹²⁸ Japan's appellee's submission, para. 175 (quoting Decree 7,819/2012 (Panel Exhibit JE-132),

Article 21). ¹¹²⁹ Japan's appellee's submission, para. 175.

¹¹³⁰ Japan's appellee's submission, para. 175.

5.422. Paragraph 2 of the Enabling Clause provides, in relevant part:

The provisions of paragraph 1 apply to the following:

...

(c) Regional or global arrangements entered into amongst developing country Members for the mutual reduction or elimination of tariffs and, in accordance with criteria or conditions which may be prescribed by the [WTO Members], for the mutual reduction or elimination of non-tariff measures, on products imported from one another[.]¹¹³¹

5.423. Paragraph 2(c) excepts differential and more favourable treatment accorded pursuant to "[r]egional or global arrangements entered into amongst" developing country Members from a finding of inconsistency with Article I of the GATT 1994. Paragraph 2(c) limits the kind of differential and more favourable treatment to the: (i) mutual reduction or elimination of tariffs; and (ii) mutual reduction or elimination of non-tariff measures. In case of the latter, paragraph 2(c) adds that the "mutual reduction or elimination of non-tariff measures" have to be "in accordance with criteria or conditions which may be prescribed" by the WTO Members. Paragraph 2(c) does not exclude the possibility that developing country Members that are parties to regional or global arrangements may adopt such instruments that they may deem appropriate for the mutual reduction or elimination of tariffs and non-tariff measures. However, it suffices that the instrument adopted that way, to be justified under paragraph 2(c) for the differential and more favourable treatment it accords, has a 'genuine" link or a rational connection with the regional or global arrangement adopted and notified to the WTO. Therefore, we disagree with the Panel to the extent it considered that, in order for any differential and more favourable treatment to be justified under paragraph 2(c), there must exist both a "close" and "genuine" link to a "regional arrangement entered into amongst" developing country Members. 1132

5.424. As noted, Brazil submits that the Panel rested its finding on its "flawed conclusion that because the [1980] Treaty of Montevideo and the provisions of the relevant ECAs do not expressly make reference to internal taxation, they did not have a genuine link with paragraph 2(c)". ¹¹³³ We consider that Brazil mischaracterizes the Panel's finding. The Panel did not find, as Brazil contends, that the 1980 Treaty of Montevideo and the relevant ECAs do not bear a genuine link with the requirements of paragraph 2(c).¹¹³⁴ Instead, the Panel found that "Brazil has not demonstrated how the relevant tax reductions [under the INOVAR-AUTO programme] found to be inconsistent under Article I: 1 of the GATT 1994 are *related* to the RTA that Brazil has notified to the WTO (the Treaty of Montevideo) or the ECAs allegedly implementing that RTA."¹¹³⁵ Consequently, the Panel was not satisfied "how the relevant differential and more favourable treatment could be justified under paragraph 2(c)".¹¹³⁶ In reaching this conclusion, the Panel examined the provisions of the 1980 Treaty of Montevideo and found that "none of the provisions cited to in the [1980] Treaty of Montevideo" had any relation "in and of themselves" to the differential tax treatment under the INOVAR-AUTO programme (in the form of internal tax reductions accorded to some but not other Members) found to be inconsistent with Article I:1 of the GATT 1994.¹¹³⁷ Turning to the relevant ECAs referred to in Articles 21 and 22(I) of Decree 7,819/2012, the Panel noted that it "could not discern any ... relationship" that would attest to "the fundamental premise of Brazil's argument, namely that the INOVAR-AUTO programme is implementing the objectives of the ECAs". ¹¹³⁸

5.425. The Panel was required to undertake this analytical exercise given that, as Brazil explains on appeal, it contended before the Panel that "the tax treatment accorded within the framework of [the] INOVAR-AUTO [programme] ... is inscribed in the context of the implementation process of the ... [ECAs] negotiated ... under the umbrella of the 1980 Treaty of Montevideo."¹¹³⁹ Consequently,

¹¹³¹ Fn omitted.

¹¹³² Panel Reports, para. 7.1117.

¹¹³³ Brazil's appellant's submission, para. 419 (referring to Panel Reports, para. 7.1115).

¹¹³⁴ Brazil's appellant's submission, para. 418.

¹¹³⁵ Panel Reports, para. 7.1115. (emphasis added)

¹¹³⁶ Panel Reports, para. 7.1118. (fn omitted)

¹¹³⁷ Panel Reports, para. 7.1112.

¹¹³⁸ Panel Reports, para. 7.1112.

¹¹³⁹ Brazil's appellant's submission, para. 367.

as Brazil further explains, the differential tax treatment was not only justified under paragraph 2(c)¹¹⁴⁰ but also "did not require additional notification".¹¹⁴¹

5.426. We have considered above the question whether or not the differential tax treatment under the INOVAR-AUTO programme was notified pursuant to paragraph 4(a) of the Enabling Clause as having been adopted under paragraph 2(c) by virtue of the notification of the 1980 Treaty of Montevideo and the relevant ECAs under paragraph 2(c). In so doing, we have noted that Articles 21 and 22(I) of Decree 7,819/2012 do not refer to the 1980 Treaty of Montevideo and that the 1980 Treaty of Montevideo does not itself specify any rules regarding internal tax reductions. We have also considered that, while Articles 21 and 22(I) of Decree 7,819/2012 refer to the relevant ECAs, those ECAs provide for the adoption of tariff preferences in the automotive sector and do not refer to internal taxation. Accordingly, in the absence of a genuine link between the differential tax treatment under the INOVAR-AUTO programme, on the one hand, and the 1980 Treaty of Montevideo and the relevant ECAs, on the other hand, we have agreed with the Panel's finding that "Brazil has not demonstrated how the ... tax reductions found to be inconsistent under Article 1:1 of the GATT 1994 are related to the RTA that Brazil has notified to the WTO (the [1980] Treaty of Montevideo) or the ECAs allegedly implementing that RTA."¹¹⁴² Therefore, to the extent that the Panel relied on its earlier analysis concerning whether or not the INOVAR-AUTO programme, which accords the differential and more favourable treatment (i.e. the differential tax treatment in the form of internal tax reductions accorded to some but not other Members), had a genuine link to "the arrangement notified to the WTO"¹¹⁴³ in determining if the differential and more favourable tax treatment was substantively justified under paragraph 2(c), we find no error in the Panel's approach. The considerations outlined by the Panel in that part of its analysis were bound to have a substantial bearing on whether or not the differential and more favourable treatment under the INOVAR-AUTO programme was substantively justified under paragraph 2(c), given the manner in which Brazil framed its arguments before the Panel. Indeed, if there is no genuine link between the measure at issue according the differential and more favourable treatment and the arrangements notified to the WTO, it is difficult to see how the measure at issue could be substantively justified under paragraph 2(c).

5.427. In light of the foregoing considerations, we <u>uphold</u> the Panel's finding, in paragraph 7.1118 of the Panel Reports, to the extent that the Panel found that Brazil has not identified any arrangement with a genuine link to the differential tax treatment envisaged under the INOVAR-AUTO programme. Consequently, we also <u>uphold</u> the Panel's finding, in paragraph 7.1121 of the Panel Reports, that the internal tax reductions accorded under the INOVAR-AUTO programme to imported products from Argentina, Mexico, and Uruguay and found to be inconsistent under Article I: 1 of the GATT 1994 are not justified under paragraph 2(c) of the Enabling Clause.

5.5.4 Conclusion with respect to the Enabling Clause

5.428. Paragraph 4(a) of the Enabling Clause envisages a degree of specificity in the notification adopted thereunder. At a minimum, a notification pursuant to paragraph 4(a) should state under which particular provision of the Enabling Clause the differential and more favourable treatment has been adopted so as to put other Members on notice. Paragraph 4(a) does not exclude the possibility that a single notification can state that the notifying Member considers an arrangement or a measure to have been adopted pursuant to one or more subparagraphs of paragraph 2 of the Enabling Clause. However, in the absence of any such indication, it cannot be taken for granted that a complaining party is on notice of those subparagraphs of paragraph 2 that the notifying Member considers to be applicable. A notification pursuant to paragraph 4(a) therefore speaks to and has a direct bearing on a complaining party's knowledge. Consequently, it speaks to whether the complaining party is required to raise the Enabling Clause and identify the relevant provision(s) thereof in its panel request and assert that an arrangement or a measure adopted by the responding party is inconsistent not only with Article I:1 of the GATT 1994, but also with the relevant provisions of the Enabling Clause and identify the relevant provisions of the Enabling Clause and identify the relevant provisions of the Enabling Clause and identify the relevant provisions of the Enabling Clause.

¹¹⁴⁰ Brazil's appellant's submission, para. 367.

¹¹⁴¹ Brazil's appellant's submission, para. 395. (fn omitted)

¹¹⁴² Panel Reports, para. 7.1115.

¹¹⁴³ Panel Reports, para. 7.1108.

- 134 -

conditions set out in the Enabling Clause remains on the responding party relying on the Enabling Clause as a defence.

5.429. With regard to whether the measure at issue (i.e. differential tax treatment under the INOVAR-AUTO programme in the form of internal tax reductions accorded to some but not other Members) was notified pursuant to paragraph 4(a) as having been adopted under paragraph 2(b) of the Enabling Clause, our review of the Panel's findings indicates that Brazil has not demonstrated that the differential tax treatment under the INOVAR-AUTO programme was notified to the WTO as adopted under paragraph 2(b), as required pursuant to paragraph 4(a) of the Enabling Clause. We therefore <u>uphold</u> the Panel's finding, in paragraphs 7.1082-7.1083 of the Panel Reports, that Brazil has not demonstrated that the differential tax treatment under the INOVAR-AUTO programme was notified to the WTO as adopted pursuant to paragraph 2(b), and therefore, in the circumstances of this case, there was no burden on the complaining parties to raise and identify paragraph 2(b) of the Enabling Clause in their panel requests.

5.430. With regard to whether the measure at issue (i.e. the differential tax treatment under the INOVAR-AUTO programme in the form of internal tax reductions accorded to some but not other Members) was notified pursuant to paragraph 4(a) as having been adopted under paragraph 2(c) of the Enabling Clause, our review of the Panel's findings indicates that Brazil has not demonstrated that the 1980 Treaty of Montevideo and the ECAs notified to the WTO have a genuine link to the INOVAR-AUTO programme providing for the differential and more favourable treatment at issue, and, consequently, the differential and more favourable treatment at issue was not notified to the WTO as adopted under paragraph 2(c), as required pursuant to paragraph 4(a) of the Enabling Clause. We therefore <u>uphold</u> the Panel's finding, in paragraph 7.1119 of the Panel Reports, that the differential tax treatment under the INOVAR-AUTO programme was not notified as adopted under paragraph 2(c), as required pursuant to paragraph 2(c), we also <u>uphold</u> the Panel's finding, in paragraph 2(c), we also <u>uphold</u> the Panel's finding, in paragraph 2(c), of the Panel's finding, in paragraph 2(c), of the Panel's finding, in paragraph 2(c), as required pursuant to paragraph 4(a). Consequently, we also <u>uphold</u> the Panel's finding, in paragraph 2(c) of the Panel's finding, in paragraph 2(c) of the Panel's finding, in paragraph 2(c), of the Panel's finding, in paragraph 2(c) of the Panel's finding, in paragraph 2(c), as required pursuant to paragraph 4(a). Consequently, we also <u>uphold</u> the Panel's finding, in paragraph 7.1120 of the Panel Reports, that, in the circumstances of this case, there was no burden on the complaining parties to raise and identify paragraph 2(c) of the Enabling Clause in their panel requests.

5.431. In light of the foregoing considerations, we <u>uphold</u> the Panel's findings, in paragraphs 8.6.h and 8.17.i of the Panel Reports, that there was no burden on the complaining parties to raise and identify the relevant provisions of the Enabling Clause in their panel requests, and their claims under Article I: 1 of the GATT 1994 were therefore within the Panel's terms of reference.

5.432. With respect to paragraph 2(b) of the Enabling Clause, we note that paragraph 2(b) provides for the granting of "[d]ifferential and more favourable treatment with respect to the provisions of the General Agreement concerning non-tariff measures governed by the provisions of instruments multilaterally negotiated under the auspices of the GATT". Paragraph 2(b) provides for the adoption of a limited category of differential and more favourable treatment, namely treatment that concerns "non-tariff measures governed by the provisions of instruments multilaterally negotiated under the auspices of the GATT" as an institution. The phrase "non-tariff measures governed by the provisions of instruments multilaterally negotiated under the auspices of the GATT", at the time of the adoption of the Enabling Clause, concerned non-tariff measures taken pursuant to the S&D treatment provisions of the Tokyo Round Codes, and not the provisions of the GATT 1947. Following the entry into force of the WTO Agreement, paragraph 2(b) of the Enabling Clause provides for the adoption of a limited category of differential and more favourable treatment, namely treatment that concerns non-tariff measures governed by the provisions of instruments multilaterally negotiated under the auspices of the WTO. The GATT 1994, while an integral part of the WTO Agreement, was not negotiated under the auspices of the WTO. These considerations read in light of the text, context, and circumstances surrounding the adoption of the Enabling Clause and thereafter the establishment of the WTO indicates that paragraph 2(b) does not concern non-tariff measures governed by the provisions of the GATT 1994. Instead, paragraph 2(b) speaks to non-tariff measures taken pursuant to S&D treatment provisions of "instruments multilaterally negotiated under the auspices of the [WTO]".

5.433. We therefore <u>uphold</u> the Panel's finding, in paragraph 7.1096 of the Panel Reports, that "a non-tariff measure within the scope of paragraph 2(b) must be governed by specific provisions on special and differential treatment, that are distinct from the provisions of the GATT 1994 incorporating the GATT 1947."¹¹⁴⁴ We also <u>uphold</u> the Panel's findings, in paragraphs 7.1097, 8.6.i,

¹¹⁴⁴ Panel Reports, para. 7.1096. (fn omitted)

and 8.17.j of the Panel Reports, that the tax reductions accorded under the INOVAR-AUTO programme to imported products from Argentina, Mexico, and Uruguay and found to be inconsistent under Article I:1 of the GATT 1994 are not justified under paragraph 2(b) of the Enabling Clause.

5.434. With respect to paragraph 2(c) of the Enabling Clause, we note that paragraph 2(c) excepts differential and more favourable treatment accorded pursuant to "[r]egional or global arrangements entered into amongst" developing country Members from a finding of inconsistency with Article I of the GATT 1994. Paragraph 2(c) does not exclude the possibility that developing country Members that are parties to regional or global arrangements may adopt such instruments that they may deem appropriate for the mutual reduction or elimination of tariffs and non-tariff measures. However, it suffices that the instrument adopted that way, to be justified under paragraph 2(c) for the differential and more favourable treatment it accords, has a "genuine" link or a rational connection with the regional or global arrangement adopted and notified to the WTO.

5.435. Our review of the Panel's findings indicates that the Panel did not find, as Brazil contends, that the 1980 Treaty of Montevideo and the relevant ECAs do not bear a genuine link with the requirements of paragraph 2(c). Instead, the Panel found that Brazil has not demonstrated how the internal tax reductions under the INOVAR-AUTO programme found to be inconsistent under Article I: 1 of the GATT 1994 are related to the RTA (the 1980 Treaty of Montevideo) that Brazil has notified to the WTO or the ECAs allegedly implementing that RTA. Consequently, the Panel was not satisfied how the relevant differential and more favourable treatment under the INOVAR-AUTO programme could be justified under paragraph 2(c). Therefore, to the extent that the Panel relied on its earlier analysis concerning whether or not the INOVAR-AUTO programme, according the differential and more favourable treatment (i.e. the differential tax treatment in the form of internal tax reductions accorded to some but not other Members), had a genuine link to "the arrangement notified to the WTO" in determining if the differential and more favourable treatment was substantively justified under paragraph 2(c), we find no error in the Panel's approach. Indeed, if there is no genuine link between the measure at issue according the differential and more favourable treatment and the arrangements notified to the WTO, we find it difficult to see how the measure at issue could be substantively justified under paragraph 2(c).

5.436. We therefore <u>uphold</u> the Panel's finding, in paragraph 7.1118 of the Panel Reports, to the extent that the Panel found that Brazil has not identified any arrangement with a genuine link to the differential tax treatment envisaged under the INOVAR-AUTO programme. Consequently, we also <u>uphold</u> the Panel's findings, in paragraphs 7.1121, 8.6.i, and 8.17.j of the Panel Reports, that the tax reductions accorded under the INOVAR-AUTO programme to imported products from Argentina, Mexico, and Uruguay and found to be inconsistent under Article I:1 of the GATT 1994 are not justified under paragraph 2(c) of the Enabling Clause.

5.6 Articles 11 and 12.7 of the DSU and Article 4.7 of the SCM Agreement

5.6.1 Introduction

5.437. In this section, we address Brazil's claim that the Panel acted inconsistently with Articles 11 and 12.7 of the DSU in recommending, pursuant to Article 4.7 of the SCM Agreement, that Brazil withdraw the prohibited subsidies identified by the Panel within 90 days.¹¹⁴⁵ To recall briefly, during the interim review stage of the Panel proceedings, the European Union and Japan requested that the Panel specify in its recommendations the time period within which the measures must be withdrawn, pursuant to Article 4.7 of the SCM Agreement.¹¹⁴⁶ On 6 December 2016, the Panel thus submitted an additional question to the parties regarding the appropriate time period that the Panel should specify in its recommendations. The Panel received the parties' responses to this question

¹¹⁴⁵ Brazil's appellant's submission, para. 436. Brazil's claim is conditional upon us upholding the Panel's findings that the measures at issue are prohibited subsidies under Articles 3.1(a) and 3.1(b) of the SCM Agreement. (Ibid., para. 436) Having partially upheld the Panel's findings under Article 3.1(b) of the SCM Agreement in section 5.4 above, we are therefore called upon to address Brazil's claim under Articles 11 and 12.7 of the DSU and Article 4.7 of the SCM Agreement.

¹¹⁴⁶ Panel Reports, para. 6.15. Article 4.7 of the SCM Agreement specifies that "the panel *shall* recommend that the subsidizing Member withdraw the subsidy without delay". (emphasis added)

- 136 -

and the parties' comments on each other's responses on 9 December 2016 and 12 December 2016, respectively.¹¹⁴⁷

5.438. In its Reports, the Panel noted that previous panels addressing the specific time period contemplated under Article 4.7 of the SCM Agreement have "tended" to specify 90 days as the time period in which the subsidy must be withdrawn.¹¹⁴⁸ The Panel considered it "difficult" to "reconcile" the "reasonable period of time' (ostensibly up to 15 months, subject to the arbitrator's discretion)" contemplated in Article 21.3(c) of the DSU and the requirement to withdraw prohibited subsidies "without delay' (i.e. typically 90 days)" under Article 4.7 of the SCM Agreement.¹¹⁴⁹ Recalling the Appellate Body's statements in *Canada – Renewable Energy / Canada – Feed-in Tariff Program* that "the expedited remedy provided under Article **4.7** ... **is an 'important consideration''', the Panel** decided to "respect the specific remedy contemplated in Article 4".¹¹⁵⁰ The Panel thus added paragraphs 8.11 and 8.22 to its Report, wherein, "[t]aking into account the procedures that may be required to implement [its] recommendation on the one hand, and the requirement that Brazil withdraw its subsidies 'without delay' on the other", the Panel recommended that the prohibited subsidies identified by it be withdrawn within 90 days.¹¹⁵¹

5.439. On appeal, Brazil claims that the Panel acted inconsistently with Articles 11 and 12.7 of the DSU because it failed to engage with the arguments put forward by the parties and to provide a "reasoned and adequate explanation" or a "basic rationale" for its conclusion that the prohibited subsidies at issue be withdrawn within 90 days.¹¹⁵² Noting that the Panel's reasoning was limited to a single paragraph of its Reports, Brazil asserts, in particular, that the Panel was required – but failed – to "explain why 90 days was appropriate in light of the 'nature of the measures' and the 'procedures that may be required to implement [the panel's] recommendation".¹¹⁵³

5.440. We begin our analysis by discussing, briefly, the relevant legal standards under Articles 11 and 12.7 of the DSU as well as Article 4.7 of the SCM Agreement. Subsequently, we consider whether the Panel acted inconsistently with Articles 11 and 12.7, as alleged by Brazil.

5.6.2 Relevant legal standards under Articles 11 and 12.7 of the DSU and Article 4.7 of the SCM Agreement

5.441. Article 11 of the DSU provides, in relevant part, that:

[A] panel should make an objective assessment of the matter before it, including an objective assessment of the facts of the case and the applicability of and conformity with the relevant covered agreements, and make such other findings as will assist the DSB in making the recommendations or in giving the rulings provided for in the covered agreements.

The provision requires a panel to make an objective assessment of the matter before it, which includes a requirement that the panel provide a "'reasoned and adequate' explanation" and coherent reasoning for its findings as the initial trier of facts.¹¹⁵⁴ A panel is required to "consider all the evidence presented to it, assess its credibility, determine its weight, and ensure that its factual

¹¹⁴⁷ Panel Reports, para. 6.16.

¹¹⁴⁸ Panel Reports, para. 6.17 (referring to Panel Reports, *Australia – Automotive Leather II*, para. 10.7; Brazil – Aircraft, para. 8.5; Canada – Aircraft Credits and Guarantees, para. 8.4; Canada – Autos, para. 11.7; Korea – Commercial Vessels, para. 8.5; US – Tax Incentives, para. 8.6).

¹¹⁴⁹ Panel Reports, para. 6.17.

¹¹⁵⁰ Panel Reports, para. 6.17 (referring to Appellate Body Reports, *Canada – Renewable Energy / Canada – Feed-in Tariff Program*, para. 5.7, in turn referring to Appellate Body Report, *EC – Export Subsidies on Sugar*, paras. 332-335).

¹¹⁵¹ Panel Reports, paras. 8.11 and 8.22.

¹¹⁵² Brazil's appellant's submission, paras. 437, 443, and 445.

¹¹⁵³ Brazil's appellant's submission, para. 453 (referring to Panel Reports, *Australia – Automotive* Leather II, para. 10.7; *Brazil – Aircraft*, para. 8.5; *Canada – Aircraft*, para. 10.4; *Canada – Autos*, para. 11.6; EC and certain member States – Large Civil Aircraft, para. 8.6; *Korea – Commercial Vessels*, para. 8.5; US – Tax Incentives, para. 8.6).

¹¹⁵⁴ Appellate Body Report, *EC and certain member States – Large Civil Aircraft*, para. 881 (quoting and referring to Appellate Body Reports, *US – Softwood Lumber VI (Article 21.5 – Canada)*, para. 97; *US – Upland Cotton (Article 21.5 – Brazil)*, para. 293 and fn 618 thereto and para. 294).

- 137 -

findings have a proper basis in that evidence".¹¹⁵⁵ Thus, when called on to review a panel's analysis, the Appellate Body "cannot base a finding of inconsistency under Article 11 simply on the conclusion that [it] might have reached a different factual finding from the one the panel reached".¹¹⁵⁶ Neither does a panel err simply because it declines to accord to the evidence the weight that one of the parties believes should be accorded to it.1157

5.442. Article 12.7 of the DSU provides that "the report of a panel shall set out the findings of fact, the applicability of relevant provisions and the basic rationale behind any findings and recommendations that it makes." What constitutes a "basic rationale" is not defined in the covered agreements. In Mexico - Corn Syrup (Article 21.5 - US), the Appellate Body noted that Article 12.7 "establishes a minimum standard for the reasoning that panels must provide in support of their findings and recommendations".¹¹⁵⁸ However, the Appellate Body cautioned that it was neither possible nor desirable "to determine, in the abstract, the minimum standard of reasoning that will constitute a 'basic rationale' for the findings and recommendations made by a panel."¹¹⁵⁹ Instead, whether a basic rationale has been set out should be decided on a case-by-case basis, taking into account the "facts of the case, the specific legal provisions at issue, and the particular findings and recommendations made by a panel".¹¹⁶⁰ In all circumstances, panels must set forth explanations and reasons sufficient to disclose the "essential, or fundamental, justification" for their findings and recommendations.¹¹⁶¹ In particular, panels must identify the relevant facts and the applicable legal norms. In applying those legal norms to the relevant facts, the reasoning of the panel must reveal how and why the law applies to the facts. In this way, panels will, in their reports, disclose the essential or fundamental justification for their findings and recommendations.¹¹⁶² This does not, however, necessarily imply that Article 12.7 requires panels to expound at length on the reasons for their findings and recommendations. There may, for example, be instances where a panel's "basic rationale" could be found in reasoning that is set out in other documents, such as in previous panel or Appellate Body reports - provided that such reasoning is quoted or, at a minimum, incorporated by reference.¹¹⁶³

5.443. We now consider the legal standard under Article 4.7 of the SCM Agreement, which is the specific legal provision applied by the Panel in the present dispute. Article 4.7 provides:

If the measure in question is found to be a prohibited subsidy, the panel shall recommend that the subsidizing Member withdraw the subsidy without delay. In this regard, the panel shall specify in its recommendation the time-period within which the measure must be withdrawn.

5.444. Through the use of the word "shall", Article 4.7 requires a panel to recommend that prohibited subsidies found to exist be withdrawn "without delay".¹¹⁶⁴ Further, Article 4.7 mandates that a panel must specify the time period within which such withdrawal must occur.¹¹⁶⁵

¹¹⁵⁵ Appellate Body Reports, China - Rare Earths, para. 5.178 (quoting Appellate Body Report,

- Brazil Retreaded Tyres, para. 185, in turn referring to Appellate Body Report, EC Hormones,
- paras. 132-133). See also Appellate Body Reports, Australia Salmon, para. 266; EC Asbestos, para. 161; EC - Bed Linen (Article 21.5 - India), paras. 170, 177, and 181; EC - Sardines, para. 299; EC - Tube or Pipe Fittings, para. 125; Japan – Apples, para. 221; Japan – Agricultural Products II, paras. 141-142; Korea – Alcoholic Beverages, paras. 161-162; Korea – Dairy, para. 138; US – Carbon Steel, para. 142;
- US Gambling, para. 363; US Oil Country Tubular Goods Sunset Reviews, para. 313; EC Selected Customs Matters, para. 258.

¹¹⁵⁷ Appellate Body Reports, Australia - Salmon, para. 267; Japan - Apples, para. 221;

Appellate Body Report, Korea - Alcoholic Beverages, para. 168).

¹¹⁵⁶ Appellate Body Report, EC and certain member States - Large Civil Aircraft, para. 1317 (quoting Appellate Body Report, US - Wheat Gluten, para. 151).

Korea - Alcoholic Beverages, para. 164.

 ¹¹⁵⁸ Appellate Body Report, *Mexico – Corn Syrup (Article 21.5 – US)*, para. 106.
 ¹¹⁵⁹ Appellate Body Report, *Mexico – Corn Syrup (Article 21.5 – US)*, para. 108 (referring to

¹¹⁶⁰ Appellate Body Report, Mexico - Corn Syrup (Article 21.5 - US), para. 108. See also Appellate Body Report, US - Carbon Steel (India), para. 4.194.

¹¹⁶¹ Appellate Body Report, *Mexico - Corn Syrup (Article 21.5 - US)*, para. 106.

¹¹⁶² Appellate Body Report, Mexico - Corn Syrup (Article 21.5 - US), para. 108 (referring to Appellate Body Report, Chile - Alcoholic Beverages, para. 78).

¹¹⁶³ Appellate Body Report, Mexico - Corn Syrup (Article 21.5 - US), paras. 108-109.

¹¹⁶⁴ Appellate Body Report, Brazil - Aircraft, para. 192.

¹¹⁶⁵ Appellate Body Report, *Brazil – Aircraft*, paras. 191-192.

- 138 -

5.445. The remedy under Article 4.7 of the SCM Agreement to withdraw prohibited subsidies "without delay" is distinct from the remedy contemplated under Article 21 of the DSU. Paragraph 1 of Article 21 of the DSU requires "prompt compliance with recommendations or rulings of the DSB", and paragraph 3 thereof allows the implementing Member "a reasonable period of time" to implement the recommendations or rulings of the DSB, where it is impracticable to comply "immediately".¹¹⁶⁶ We note in this regard that Article 4.7 of the SCM Agreement is listed in Appendix 2 to the DSU as a "special or additional rule or procedure" on dispute settlement, and the provisions of Article 21.3 of the DSU therefore "are not relevant in determining the period of time for implementation of a finding of inconsistency with the prohibited subsidies provisions of Part II of the SCM Agreement".¹¹⁶⁷

5.446. As regards the meaning of "without delay" in Article 4.7, the dictionary meanings of "delay" include, as a noun, "procrastination ... ; waiting, lingering"¹¹⁶⁸, and as a verb, "[t]o put off to a later time; to defer, postpone".¹¹⁶⁹ As to the context, we note that, in requiring that prohibited subsidies be withdrawn "without delay", Article 4.7 states that panels have to specify the time period within which the measures must be withdrawn. Furthermore, Article 4.10 requires the DSB to grant authorization to the complaining Member to take appropriate countermeasures in the event the recommendations of the DSB are not followed within the time period specified by a panel pursuant to Article 4.7. The text and the context of Article 4.7 therefore suggest to us that the term "without delay" in Article 4.7 is not used in the sense of requiring *immediate* compliance. Nor does the term "without delay", combined with the requirement that the panel specify a time period, impose a single standard or time period applicable in all cases. Instead, Article 4.7 requires a panel to specify a time period that constitutes "without delay" within the realm of possibilities in a given case and considering the domestic legal system of the implementing Member. In determining the time period under Article 4.7 that constitutes "without delay", a panel should typically take into account the nature of the measure(s) to be revoked or modified and the domestic procedures available for such revocation or modification. These domestic procedures include any extraordinary procedures that may be available within the legal system of a WTO Member.¹¹⁷⁰

5.447. Finally, we consider it useful to contrast the text of Article 4.7 of the SCM Agreement with that of Article 21.3(c) of the DSU. Article 21.3 of the DSU specifies that, "[i]f it is impracticable to comply immediately with the recommendations and rulings, the Member concerned shall have a reasonable period of time in which to do so." Article 21.3(c) of the DSU in turn provides that an arbitrator may be appointed where a reasonable time period cannot be agreed on, and that "a guideline for the arbitrator should be that the reasonable period of time ... should not exceed 15 months from the date of adoption", although that time period "may be shorter or longer, depending upon the particular circumstances".¹¹⁷¹ By contrast, Article 4.7 of the SCM Agreement contains no reference to flexibilities depending on "circumstances". Article 4.7 simply mandates that "the panel shall recommend that the subsidizing Member withdraw the subsidy without delay" and that "the panel shall specify in its recommendation the time period within which the measure must be withdrawn." Therefore, in contrast to Article 21.3(c) of the DSU, the use of the term "without delay" in Article 4.7 constrains the latitude available to a panel in specifying the time period under that provision.

http://www.oed.com/view/Entry/49278?rskey=23TzIE&result=2&isAdvanced=false#eid.
 ¹¹⁷⁰ By contrast, we note that the existence of, and recourse to, extraordinary procedures within the domestic legal system of a WTO Member State is a factor that is generally not taken into account in determining the "reasonable period of time" under Article 21.3(c) of the DSU. See Award of the Arbitrator, *EC – Chicken Cuts (Article 21.3(c))*, para. 49 (referring to Award of the Arbitrator, *Korea – Alcoholic Beverages (Article 21.3(c))*, para. 42; Award of the Arbitrator, *Chile – Price Band System (Article 21.3(c))*, para. 51; Award of the Arbitrator, *US – Offset Act (Byrd Amendment) (Article 21.3(c))*, para. 74).

¹¹⁶⁶ Appellate Body Report, *Brazil – Aircraft*, para. 191.

¹¹⁶⁷ Appellate Body Report, *Brazil – Aircraft*, para. 192.

¹¹⁶⁸ Oxford English Dictionary online, definition of "delay" (noun),

http://www.oed.com/view/Entry/49277?rskey=23TzIE&result=1&isAdvanced=false#eid.

¹¹⁶⁹ Oxford English Dictionary online, definition of "delay" (verb),

¹¹⁷¹ The three governing principles applicable to an arbitrator's determination of the reasonable period of time under Article 21.3(c) of the DSU are as follows: (i) the reasonable period of time should be the shortest period of time possible within the legal system of the implementing Member; (ii) the implementing Member must utilize all the flexibility and discretion available within its legal and administrative system in order to implement within the shortest period of time possible; and (iii) the "particular circumstances" of the case must be taken into account in determining the reasonable period of time. (Award of the Arbitrator, *EC – Export Subsidies on Sugar (Article 21.3(c))*, para. 61)

- 139 -

5.6.3 Whether the Panel acted inconsistently with Articles 11 and 12.7 of the DSU in recommending that Brazil withdraw the prohibited subsidies within 90 days

5.448. On appeal, Brazil claims that the Panel acted inconsistently with Articles 11 and 12.7 of the DSU in recommending that Brazil withdraw the prohibited subsidies found to exist in the present dispute within 90 days.¹¹⁷² In Brazil's view, the Panel failed to engage with the arguments put forward by the parties and to provide a "reasoned and adequate explanation" or a "basic rationale" for its conclusion that the recommended timeframe of 90 days was adequate, as required under those provisions.¹¹⁷³ Brazil alleges that the Panel did not explain why "respect[ing] the specific remedy contemplated in Article 4 of the SCM Agreement" necessitated the granting of only 90 days, when other panels have afforded considerably more than 90 days when making specific recommendations under Article 4.7.1174 In Brazil's view, in order to provide a "reasoned and adequate explanation" or "basic rationale" for its conclusion, the Panel was required to explain why 90 days was appropriate in light of the "nature of the measures"¹¹⁷⁵ and the "procedures that may be required to implement [the panel's] recommendation".¹¹⁷⁶ By failing to provide such an explanation, Brazil alleges that the Panel acted inconsistently with its obligations under Articles 11 and 12.7 of the DSU.¹¹⁷⁷ Consequently, Brazil requests us to reverse the Panel's findings, in paragraphs 8.11 and 8.22 of its Reports, and to complete the legal analysis and recommend that Brazil withdraw the prohibited subsidies at issue within 18 months from the adoption of the recommendations and rulings of the DSB.1178

5.449. Requesting the Appellate Body to reject Brazil's claim¹¹⁷⁹, the European Union and Japan observe that panels do not need to address in their reports every argument or all evidence raised by the parties¹¹⁸⁰, and the fact that panels are required to set forth explanations and reasons "sufficient to disclose the essential, or fundamental, justification" for their findings and recommendations does not necessarily imply that Article 12.7 requires panels "to expound at length on the reasons for their findings and recommendations".¹¹⁸¹ In the appellees' view, the Panel came to its conclusion after considering the parties' views and had "taken into account the procedures that may be required to implement the recommendation", on the one hand, and the requirement under Article 4.7 of the SCM Agreement that prohibited subsidies be withdrawn "without delay", on the other hand.¹¹⁸² The European Union further notes that the Panel supported its conclusion by observing a trend to specify 90 days in previous cases.¹¹⁸³ Moreover, the European Union considers that Brazil "appears to disagree with the Panel's conclusion" and that this does not mean that the Panel has failed to provide a basic rationale for its recommendation.¹¹⁸⁴ Japan makes a similar argument that Brazil's claim on appeal "appears to be simply an effort to relitigate the question that the Panel already raised and on which it carefully considered all of the parties' views, as reflected in Panel question 83 to the Parties". ¹¹⁸⁵ The European Union observes that panels in previous cases "have motivated the 90-days period on the same basis [as the Panel], i.e. without the need to go

¹¹⁸² European Union's appellee's submission, para. 529; Japan's appellee's submission, para. 182.

¹¹⁷² Brazil's appellant's submission, para. 436.

¹¹⁷³ Brazil's appellant's submission, paras. 437, 443, and 445.

¹¹⁷⁴ Brazil's appellant's submission, para. 447 (referring to Panel Reports, *US – Upland Cotton*, para. 8.3(b); *US – FSC*, para. 8.8).

¹¹⁷⁵ Brazil's appellant's submission, para. 453 (quoting Panel Reports, *Australia – Automotive Leather II*, para. 10.7; *Brazil – Aircraft*, para. 8.5; *EC and certain member States – Large Civil Aircraft*, para. 8.6;

US - Tax Incentives, para. 8.6).

¹¹⁷⁶ Brazil's appellant's submission, para. 453 (quoting Panel Reports, Brazil - Aircraft, para. 8.5;

Canada - Aircraft, para. 10.4; Canada - Autos, para. 11.6; Korea - Commercial Vessels, para. 8.5).

¹¹⁷⁷ Brazil's appellant's submission, para. 453.

¹¹⁷⁸ Brazil's appellant's submission, paras. 454 and 461.

¹¹⁷⁹ European Union's appellee's submission, para. 532; Japan's appellee's submission, para. 184.

¹¹⁸⁰ European Union's appellee's submission, para. 528; Japan's appellee's submission, para. 181.

¹¹⁸¹ European Union's appellee's submission, para. 528 (referring to Appellate Body Report,

Mexico – Corn Syrup (Article 21.5 – US), paras. 108-109); Japan's appellee's submission, para. 181 (referring to Appellate Body Report, *Mexico – Corn Syrup (Article 21.5 – US)*, paras. 108-109).

¹¹⁸³ European Union's appellee's submission, para. 529.

¹¹⁸⁴ European Union's appellee's submission, para. 530.

¹¹⁸⁵ Japan's appellee's submission, para. 184. Panel question No. 83 to the parties stated: "in light of the reference in Article 4.7 of the SCM Agreement to withdrawal of a subsidy 'without delay', could the parties please indicate what they consider is the time period within which the relevant subsidies in this dispute should be withdrawn, as well as their reasons for indicating such a time period?".

- 140 -

into a detailed reasoning to support their conclusions" 1186, and considers that, in any event, the Panel did not err in recommending a timeframe of 90 days, because Brazil "can withdraw the prohibited subsidies immediately through provisional measures (*Medida Provisória*) which are adopted by the executive branch".¹¹⁸⁷

5.450. The main issue before us in the present case is whether the Panel provided a "reasoned and adequate explanation" or "basic rationale", in accordance with Articles 11 and 12.7 of the DSU, for its recommendation, in paragraphs 8.11 and 8.22 of its Reports, that Brazil withdraw the prohibited subsidies found to exist within 90 days. We do not share the appellees' view that Brazil simply appears to take issue with the Panel's conclusion and is thus merely recasting its arguments before the Panel in the guise of a claim under Article 11 of the DSU.¹¹⁸⁸ Although Brazil also disagrees with the Panel's final conclusion regarding the time period of 90 days, we note that its appeal is directed at the Panel's analysis in its Reports and whether that analysis constituted a "reasoned and adequate explanation" or "basic rationale" for its conclusion that 90 days is the appropriate time. We thus do not consider that Brazil is simply recasting factual arguments made before the Panel in the guise of an Article 11 claim.¹¹⁸⁹ Rather, the issue that Brazil raises on appeal concerns the Panel's treatment of information on the record and, in particular, whether its failure to engage with the parties' arguments and to "explain why 90 days was appropriate in light of the 'nature of the measures' and the 'procedures that may be required to implement [the panel's] recommendation'" constitute errors under Article 11 and Article 12.7 of the DSU.1190

5.451. Turning to the Panel's analysis, we note that the Panel referred to the practice of previous panels and noted that they have "tended" to specify 90 days as the time period under Article 4.7 of the SCM Agreement.¹¹⁹¹ The Panel further considered the differences between the timeframe contemplated under Article 21.3 of the DSU and Article 4.7 of the SCM Agreement and observed that:

[A] local content requirement in a subsidy would need to be remedied "without delay" (i.e. typically 90 days) when challenged as a prohibited subsidy under the SCM Agreement, whereas the same local content requirement in the same instruments when challenged under the GATT 1994 would need to be remedied within a "reasonable period of time" (ostensibly up to 15 months, subject to the arbitrator's discretion). It is difficult for the Panel to reconcile these provisions.¹¹⁹²

Noting the Appellate Body's statements in Canada - Renewable Energy / Canada - Feed-in Tariff Program "that the expedited remedy provided under Article 4.7 ... is an 'important consideration"¹¹⁹³, the Panel decided to "respect the specific remedy contemplated in Article 4"¹¹⁹⁴ and added paragraphs 8.10, 8.11, 8.21, and 8.22 to its conclusions. In those paragraphs, the Panel recommended that Brazil withdraw the subsidies identified by the Panel "without delay".¹¹⁹⁵ Furthermore, "[t]aking into account the procedures that may be required to implement [the Panel's] recommendation on the one hand, and the requirement that Brazil withdraw its subsidies 'without

¹¹⁸⁶ European Union's appellee's submission, para. 530 (referring to Panel Reports,

Australia - Automotive Leather II, para. 10.7; Brazil - Aircraft, para. 8.5; Canada - Aircraft Credits and Guarantees, para. 8.4; Canada - Autos, para. 11.7; Korea - Commercial Vessels, para. 8.5; US - Tax *Incentives*, para. 8.6). ¹¹⁸⁷ European Union's appellee's submission, para. 531. The European Union incorporated by reference

its arguments in response to Panel question No. 83 and its comments on Brazil's response to Panel question No. 83. (Ibid., fn 510 thereto)

¹¹⁸⁸ European Union's appellee's submission, para. 530; Japan's appellee's submission, para. 184.

¹¹⁸⁹ Appellate Body Report, EC - Fasteners (China), para. 442.

¹¹⁹⁰ Brazil's appellant's submission, para. 453 (referring to Panel Reports, Australia - Automotive Leather II, para. 10.7; Brazil - Aircraft, para. 8.5; Canada - Aircraft, para. 10.4; Canada - Autos, para. 11.6; EC and certain member States - Large Civil Aircraft, para. 8.6; Korea - Commercial Vessels, para. 8.5; US - Tax Incentives, para. 8.6).

¹¹⁹¹ Panel Reports, para. 6.17 (referring to Panel Reports, Australia – Automotive Leather II, para. 10.7; Brazil - Aircraft, para. 8.5; Canada - Aircraft Credits and Guarantees, para. 8.4; Canada - Autos, para. 11.7; Korea - Commercial Vessels, para. 8.5; US - Tax Incentives, para. 8.6).

¹¹⁹² Panel Reports, para. 6.17.

¹¹⁹³ Panel Reports, para. 6.17 (quoting Appellate Body Reports, Canada - Renewable Energy / Canada - Feed-in Tariff Program, para. 5.7, in turn referring to Appellate Body Report, EC - Export Subsidies on Sugar, paras. 332-335). ¹¹⁹⁴ Panel Reports, para. 6.17.

¹¹⁹⁵ Panel Reports, paras. 8.10 and 8.21.

- 141 -

delay' on the other", the Panel specified a time period of 90 days for the withdrawal of the subsidies at issue.¹¹⁹⁶

5.452. The Panel's analysis under Article 4.7 proceeded in three steps. First, referring to the practice of previous panels, the Panel considered that the time period for a recommendation under Article 4.7 of the SCM Agreement was "typically" 90 days. As the second step, referring to the Appellate Body's statements in *Canada – Renewable Energy / Canada – Feed-in Tariff Program*, the Panel considered that, because the specific remedy under Article 4.7 of the SCM Agreement is an "important consideration"¹¹⁹⁷, the stricter time period of "typically 90 days" under Article 4.7 of the SCM Agreement should prevail over the possibly longer time frame of up to 15 months contemplated under Article 21.3(c) of the DSU. Finally, "[t]aking into account the procedures that may be required to implement [the Panel's] recommendation on the one hand, and the requirement that Brazil withdraw its subsidies 'without delay' on the other", the Panel specified a time period of 90 days for the withdrawal of the subsidies at issue.¹¹⁹⁸

5.453. The Appellate Body has found that a panel provides a "basic rationale" where it incorporates, as part of its findings, a reference to past jurisprudence supporting its reasoning.¹¹⁹⁹ In the present case, the Panel referred to prior panel reports specifying 90 days as the time period under Article 4.7 as well as the Appellate Body Reports in *Canada – Renewable Energy / Canada – Feed-in Tariff Program*. Beginning with the latter, we do not consider that the Panel's reference to the Appellate Body's observations in *Canada – Renewable Energy / Canada – Feed-in Tariff Program* is germane to the issue at hand in the present dispute. In those earlier disputes, the Appellate Body's statement that the "specific remedy provided under Article 4.7 of the SCM Agreement is an important consideration" related to the issue of whether the panel had erred in the sequence of its analysis by choosing not to start with the claims under the SCM Agreement.¹²⁰⁰ We fail to see how the Appellate Body's statement in *Canada – Renewable Energy / Canada – Feed-in Tariff Program*, although not incorrect, serves as reasoning underpinning the Panel's indication of 90 days as the time period under Article 4.7 in light of the specific facts and circumstances of this case.

5.454. Next, we turn to the Panel's observation that "previous panels addressing the specific time period under Article 4.7 of the SCM Agreement have tended to specify 90 days as the time period in which the subsidy must be withdrawn."¹²⁰¹ In considering whether the Panel's reference to the practice of prior panels specifying 90 days as the time period under Article 4.7 constitutes sufficient reasoning for its conclusions in the present case, we note that the Panel itself acknowledged that the time period specified by prior panels was "typically 90 days".¹²⁰² Indeed, while many panels have tended to specify 90 days as the time period for withdrawal of prohibited subsidies "without delay", other panels have specified longer periods under Article 4.7.1203 This is in line with our discussion above that the use of the term "without delay" in Article 4.7 does not establish a bright-line standard of 90 days to be specified in all cases under that provision. Instead, a panel's duty under Article 4.7 is to specify the time period taking into account the facts and circumstances of each case, such that it applies the "without delay" standard in Article 4.7 on a case-by-case basis. Although there may well be a trend of 90 days as the time period for the withdrawal of prohibited subsidies, this does not ipso facto support the Panel's conclusion that 90 days was the appropriate time period in the present dispute. Moreover, while we agree that the time period under Article 21.3(c) of the DSU is not relevant to the determination of the appropriate time period under Article 4.7 of the SCM Agreement¹²⁰⁴, we also do not consider that the language "without delay" in Article 4.7 should

¹¹⁹⁶ Panel Reports, paras. 8.11 and 8.22.

¹¹⁹⁷ Panel Reports, para. 6.17 (quoting Appellate Body Reports, *Canada – Renewable Energy / Canada – Feed-in Tariff Program*, para. 5.7, in turn referring to Appellate Body Report, *EC – Export Subsidies on Sugar*, paras. 332-335).

¹¹⁹⁸ Panel Reports, paras. 8.11 and 8.22.

¹¹⁹⁹ Appellate Body Report, US - Carbon Steel (India), para. 4.195.

¹²⁰⁰ Appellate Body Reports, Canada - Renewable Energy / Canada - Feed-in Tariff Program, para. 5.7.

¹²⁰¹ Panel Reports, para. 6.17 (referring to Panel Reports, *Korea – Commercial Vessels*, para. 8.5; *Canada – Aircraft Credits and Guarantees*, para. 8.4; *Canada – Autos*, para. 11.7; *Australia – Automotive Leather II*, para. 10.7; *Brazil – Aircraft*, para. 8.5; *US – Tax Incentives*, para. 8.6).

¹²⁰² Panel Reports, para. 6.17. (emphasis added)

¹²⁰³ Panel Reports, US – FSC, para. 8.8; US – Upland Cotton, para. 8.3.

¹²⁰⁴ The Appellate Body has noted in past jurisprudence that Article 4.7 of the SCM Agreement is listed in Appendix 2 to the DSU as a "special or additional rule[] and procedure[]" on dispute settlement, and that the provisions of Article 21.3 of the DSU "are not relevant in determining the period of time for implementation of a finding of inconsistency with the prohibited subsidies provisions of Part II of the SCM Agreement". (Appellate Body Report, *Brazil – Aircraft*, para. 192)

- 142 -

automatically lead to a recommendation of 90 days for the withdrawal of prohibited subsidies in all circumstances.

5.455. As the third step in its analysis, in specifying a period of 90 days as part of its conclusions under Article 4.7 in this case, the Panel stated that it took into account "procedures that may be required to implement [its] recommendation on the one hand, and the requirement that Brazil withdraw its subsidies 'without delay' on the other".¹²⁰⁵ We recall that a panel is not required under Article 11 of the DSU to "consider each and every argument put forward by the parties in support of their respective cases".¹²⁰⁶ Furthermore, it is generally "within the competence of a panel 'freely to use arguments submitted by any of the parties – or to develop its own legal reasoning – to support its own findings and conclusions on the matter under its consideration'."¹²⁰⁷ Thus, the fact that the Panel did not engage with each of the parties' arguments would not in and of itself constitute a failure to make an objective assessment of the matter in the sense of Article 11 of the DSU. A panel is, however, required to "consider all the evidence presented to it, assess its credibility, determine its weight, and ensure that its factual findings have a proper basis in that evidence".¹²⁰⁸

5.456. As Brazil rightly points out, during the interim review stage of the panel proceedings, the parties presented extensive arguments as part of their responses and comments on each other's responses to the Panel's additional question on this issue. The European Union and Japan sought a time period of 90 days¹²⁰⁹, whereas Brazil advocated for a period of 18 months.¹²¹⁰ In particular, Japan argued, recalling previous disputes where panels established a time period of 90 days, that, in determining the time period for a withdrawal "without delay", a panel may take into account "the nature of the measures and the procedures which may be required to implement [the] recommendation"¹²¹¹, but need not consider the time required to "design replacement measures". 1212 The European Union emphasized that it was possible for Brazil to withdraw the measures found to be prohibited subsidies within 90 days, through the use of provisional measures in the Brazilian system (Medida Provisória), and alleged that such provisional measures had been used many times to amend the programmes at issue in this dispute.¹²¹³ The European Union had also noted that many of the legal instruments ensuring the operation of the challenged programmes, and notably those containing the discriminatory local content requirements, were administrative acts, such as the Portarias containing PPBs, which the executive branch can amend by simple administrative action.¹²¹⁴ The European Union further argued that it would be "incongruous" to interpret "without delay", under Article 4.7, as being equal or longer than the maximum period by which an actionable subsidy shall be withdrawn or its adverse effects removed, namely, six months.1215

5.457. Brazil, for its part, pointed out that, "in circumstances where the withdrawal of the subsidies would entail significant economic consequences, or where legislative action was required, panels

¹²⁰⁹ European Union's response to Panel question No. 83; Japan's response to Panel question No. 83.

¹²⁰⁵ Panel Reports, paras. 8.11 and 8.22.

¹²⁰⁶ Appellate Body Report, Dominican Republic - Import and Sale of Cigarettes, para. 125.

¹²⁰⁷ Appellate Body Reports, *China – HP-SSST (Japan) / China – HP-SSST (EU)*, para. 5.116 (quoting Appellate Body Reports, *Canada – Renewable Energy / Canada – Feed-in Tariff Program*, para. 5.215, in turn quoting Appellate Body Reports, *EC – Hormones*, para. 156; referring to Appellate Body Report, *US – Certain EC Products*, para. 123).

¹²⁰⁸ Appellate Body Reports, *China – Rare Earths*, para. 5.178 (quoting Appellate Body Report, *Brazil – Retreaded Tyres*, para. 185, in turn referring to Appellate Body Report, *EC – Hormones*, paras. 132-133; referring to Appellate Body Reports, *Australia – Salmon*, para. 266; *EC – Asbestos*, para. 161; *EC – Bed Linen (Article 21.5 – India)*, paras. 170, 177, and 181; *EC – Sardines*, para. 299; *EC – Tube or Pipe Fittings*, para. 125; *Japan – Apples*, para. 221; *Japan – Agricultural Products II*, paras. 141-142; *Korea – Alcoholic Beverages*, paras. 161-162; *Korea – Dairy*, para. 138; *US – Carbon Steel*, para. 142; *US – Gambling*, para. 363; *US – Oil Country Tubular Goods Sunset Reviews*, para. 313; *EC – Selected Customs Matters*, para. 258).

¹²¹⁰ Brazil's response to Panel question No. 83.

¹²¹¹ Panel Reports, Australia - Automotive Leather II, para. 10.4; Brazil - Aircraft, para. 8.5.

¹²¹² Japan's response to Panel question No. 83, paras. 4 and 6.

¹²¹³ See European Union's response to Panel question No. 83, para. 4. According to the European Union, provisional measures "allow the executive branch to swiftly enact infra-constitutional norms [that] ... are as a rule immediately effective but ... must be submitted to the Parliament for conversion into law (possibly with amendments) within a certain period (60 to 120 days), failing which they expire." (Ibid., para. 3 (fn omitted))

¹²¹⁴ European Union's response to Panel question No. 83, para. 6.

¹²¹⁵ European Union's response to Panel question No. 83, para. 8.

- 143 -

ha[d] afforded up to one year for implementation under Article 4.7."¹²¹⁶ Brazil also elaborated extensively on: (i) the nature of the measures; and (ii) the procedures that would be required to implement the Panel's recommendation. According to Brazil, the nature of those measures meant that they "constitute a series of inter-connected tax exemptions, tax reductions, and tax suspensions established under distinct federal laws and regulations that apply to a wide range of production processes in key economic sectors of the Brazilian economy".¹²¹⁷ Further, Brazil considered that the procedures for withdrawal would involve legislative action, entailing "a complex and lengthy process that usually takes more than 2 years, involving both the executive and the legislative branch".¹²¹⁸ Brazil estimated that, following the normal legislative process, "even under the most expeditious scenario the withdrawal of the measures at issue could not be accomplished in less than 18 months from adoption of the DSB recommendations and rulings in this dispute."¹²¹⁹

5.458. The parties further submitted extensive comments rebutting each other's arguments, with the European Union arguing, *inter alia*, that the difficulties described by Brazil were "contradicted by **the frequency with which those programmes have been amended ... during their life".**¹²²⁰ Likewise, Japan commented that, while Brazil was free to consider each step of the entire production chain in the affected sectors or the risk of judicial challenges, this should have no bearing on the Panel's interpretation and application of the phrase "without delay", and that these were factors to be considered within Brazil in the design of compliance measures, not factors for the Panel to consider when determining the time period for withdrawal.¹²²¹ By contrast, Brazil denied that provisional measures could be used to implement the Panel's recommendation, as the European Union argued, because they "lose effect retroactively if not converted into law within [120 days], through a legislative process similar to the one described by Brazil in its response to the Panel, and cannot be re-enacted".¹²²² Accordingly, in Brazil's view, the enactment of a provisional measure would "at most affect a *temporary* withdrawal of the measures at issue, and [would] not obviate the need to go through Brazil's ordinary legislative process".¹²²³

5.459. Other than the Panel's indication, in paragraphs 8.11 and 8.22 of its Reports, that it had taken into account "the procedures that may be required to implement [its] recommendation on the one hand, and the requirement that Brazil withdraw its subsidies 'without delay' on the other", there is no way to determine the extent to which the Panel took into account the practical aspects of the procedures necessary for withdrawal in the present case. In particular, we consider that the parties' arguments regarding the two potential avenues for implementation - namely, through the normal legislative process or through provisional measures - are relevant to the Panel's assessment of whether or not 90 days was appropriate in view of the facts of the case. While we disagree with Brazil's suggestion that the Panel was required to engage in an inquiry similar to that under Article 21.3(c) of the DSU, we nonetheless consider that the Panel's failure to engage with these arguments is particularly glaring in light of Brazil's argument that certain panels have afforded more than 90 days "in circumstances ... where legislative action was required"1224 and the European Union's rebuttal thereof.¹²²⁵ The single statement by the Panel in its conclusions on Article 4.7 does not allow us to ascertain whether the panel's findings and recommendations were based on a sufficient evidentiary basis on the record and an adequate engagement with the arguments presented by the parties, and whether its recommendation was based on a reasoned and

¹²¹⁶ Brazil's response to Panel question No. 83 (referring to Panel Reports, *US – Upland Cotton*, para. 8.3(b); *US – FSC*, para. 8.8).

¹²¹⁷ Brazil's response to Panel question No. 83.

¹²¹⁸ Brazil's response to Panel question No. 83.

¹²¹⁹ Brazil's response to Panel question No. 83.

¹²²⁰ European Union's comments on Brazil's response to Panel question No. 83, para. 14. See also Japan's comments on Brazil's response to Panel question No. 83, para. 5.

¹²²¹ Japan's comments on Brazil's response to Panel question No. 83, para. 3.

 ¹²²² Brazil's comments on the European Union's and Japan's responses to Panel question No. 83, para. 3.
 ¹²²³ Brazil's comments on the European Union's and Japan's responses to Panel question No. 83, para. 3.
 (emphasis original)

¹²²⁴ Brazil's response to Panel question No. 83 (referring to Panel Reports, *US – Upland Cotton*, para. 8.3(b); *US – FSC*, para. 8.8). See Brazil's appellant's submission, para. 448.

¹²²⁵ The European Union argued, *inter alia*, that the panel in *US* – *FSC* did not set the time period for the United States to withdraw the export contingent subsidies only because legislative action was necessary. Instead, the panel looked into "what was absolutely necessary for the United States to withdraw 'without delay'" the subsidies, which required legislative action, as acknowledged by the European Communities. (European Union's comments on Brazil's response to Panel question No. 83, paras. 8-9)

- 144 -

adequate explanation, in accordance with Article 11 of the DSU.¹²²⁶ Moreover, in our view, as a result of the Panel's failure to engage with the participants' arguments on the relevant procedures necessary to implement its recommendation, the Panel's analysis does not "reveal how and why the law applies" to the facts at hand, in accordance with Article 12.7 of the DSU.¹²²⁷

5.460. In light of the above, we do not consider that the Panel established a sufficient link between the time period of 90 days specified by it for the withdrawal of the subsidies at issue and the domestic procedure within Brazil for such withdrawal. Instead, the Panel appears to have treated the practice of specifying 90 days by some prior panels as the *de facto* standard to be applied in all cases. As discussed, Article 4.7 requires a case-by-case analysis of the time period to be prescribed for the withdrawal of prohibited subsidies "without delay". We therefore find that by failing to provide a "reasoned and adequate explanation" or a "basic rationale" in recommending a time period of 90 days under Article 4.7 of the SCM Agreement in the present case, the Panel acted inconsistently with Articles 11 and 12.7 of the DSU. Consequently, we reverse the Panel's conclusions, in paragraphs 8.11 and 8.22 of its Reports, that Brazil withdraw the prohibited subsidies identified at paragraphs 8.5.e, 8.6.e, 8.7, 8.16.f, 8.17.f, and 8.18 of the Panel Reports within 90 days.

5.461. Having reversed the Panel's findings, we turn to Brazil's request to complete the legal analysis.¹²²⁸ Brazil alleges that it is "undisputed that legislation is required in order to withdraw the relevant measures" and that the "unresolved disagreement between the parties concerns whether the measures must be withdrawn using Brazil's 'ordinary legislative process', or whether the withdrawal of the relevant measures could be made effective through provisional measures".¹²²⁹ Further, Brazil asserts that, because "the enactment of a provisional measure will at most effect a temporary withdrawal of the measures", the "question left unresolved by the Panel is whether the temporary withdrawal of the subsidies is consistent with the requirement under Article 4.7".¹²³⁰ Accordingly, in Brazil's view, the Appellate Body "should conclude that the only avenue for Brazil to withdraw the subsidies in a manner consistent with Article 4.7 [was] pursuant to the ordinary legislative process in Brazil", namely, within 18 months.¹²³¹

5.462. In previous disputes, the Appellate Body has emphasized that it can complete the analysis "only if the factual findings of the panel, or the undisputed facts in the panel record" provide a sufficient basis for it to do so.¹²³² We have already noted that the Panel made no factual findings relating to the domestic procedure for the withdrawal of the subsidies at issue under Brazil's legal system. We further note that, before the Panel, besides agreeing that the measures at issue are contained in multiple legal instruments having the nature of laws, regulations, and administrative acts¹²³³, the parties disagreed over all factual aspects related to the procedures that may be employed to implement the Panel's recommendation. 1234

5.463. We do not, for example, agree with Brazil's assertion that it is *undisputed* that "legislation is required in order to withdraw the relevant measures" and that the "unresolved disagreement between the parties concerns whether the measures must be withdrawn using Brazil's 'ordinary legislative process', or whether the withdrawal of the relevant measures could be made effective through provisional measures".¹²³⁵ In this regard, we recall that the European Union, in refuting

¹²²⁶ Appellate Body Reports, China - Rare Earths, para. 5.178 (quoting Appellate Body Report, Brazil - Retreaded Tyres, para. 185, in turn referring to Appellate Body Report, EC - Hormones,

paras. 132-133; referring to Appellate Body Reports, *Australia – Salmon*, para. 266; *EC – Asbestos*, para. 161; *EC – Bed Linen (Article 21.5 – India)*, paras. 170, 177, and 181; *EC – Sardines*, para. 299; *EC – Tube or Pipe* Fittings, para. 125; Japan - Apples, para. 221; Japan - Agricultural Products II, paras. 141-142;

Korea - Alcoholic Beverages, paras. 161-162; Korea - Dairy, para. 138; US - Carbon Steel, para. 142; US - Gambling, para. 363; US - Oil Country Tubular Goods Sunset Reviews, para. 313; EC - Selected Customs Matters, para. 258).

Appellate Body Report, Mexico - Corn Syrup (Article 21.5 - US), para. 108.

¹²²⁸ Brazil's appellant's submission, para. 456.

¹²²⁹ Brazil's appellant's submission, para. 457.

¹²³⁰ Brazil's appellant's submission, para. 457.

¹²³¹ Brazil's appellant's submission, paras. 458 and 461.

¹²³² Appellate Body Report, US - Hot-Rolled Steel, para. 235.

¹²³³ European Union's response to Panel question No. 83, para. 2; Brazil's comments on the

European Union's and Japan's responses to Panel question No. 83, para. 2.

¹²³⁴ We note that because this issue was raised during the interim review stage, the information on the record concerning this issue is limited to the Panel's analysis, the parties' responses to Panel question No. 83, and their comments on each other's responses.

¹²³⁵ Brazil's appellant's submission, para. 457.

- 145 -

before the Panel that legislative action is absolutely necessary, alleged that Brazil is not only equipped with provisional measures, but also with "other administrative acts at Brazil's disposal that would eliminate the discriminatory elements of the subsidy programmes found to be inconsistent in this case".¹²³⁶ As the Panel's description of the challenged programmes states, each of the programmes that were found to grant prohibited subsidies operated through various legal instruments including laws, decrees, implementing orders, interministerial implementing orders, and accreditations granted pursuant to the programmes.¹²³⁷ Determining whether it is possible to withdraw the relevant measures at issue through means other than Brazil's normal legislative process would therefore require us to study each of the legal instruments at issue and make our own factual findings about the means of implementation available to Brazil, which would be outside the scope of appellate review. Similarly, we find no undisputed facts on the record concerning the time periods for the legislative process in Brazil¹²³⁸, as well as on the issue of whether provisional measures by Brazil would result in a "permanent" or a "temporary" withdrawal of the measures at issue.1239

5.464. In such circumstances, we are unable to complete the legal analysis, as requested by Brazil, and cannot specify the time period within which the prohibited subsidies identified by the Panel must be withdrawn by Brazil. We note, however, that the Panel's recommendations, in paragraphs 8.10 and 8.21, that Brazil withdraw the subsidies identified by the Panel "without delay", stand undisturbed.

6 FINDINGS AND CONCLUSIONS

6.1. For the reasons set out in these Reports, the Appellate Body makes the following findings and conclusions.1240

6.1 Articles III:2 and III:4 of the GATT 1994 and Article 2.1 of the TRIMs Agreement

6.1.1 Whether the Panel erred in finding that imported finished ICT products were taxed in excess of like domestic finished ICT products inconsistently with Article III:2, first sentence, of the GATT 1994

6.2. Imported finished ICT products are not eligible for either tax reductions or exemptions because foreign producers cannot be accredited under the ICT programmes and, consequently, bear the full tax burden, as opposed to like domestic finished ICT products. In the case of an imported finished ICT product, when an importer sells the imported finished ICT product to a wholesaler, retailer, or distributor, the importer will charge the IPI tax to the wholesaler, retailer, or distributor and remit the tax to the Brazilian Government.¹²⁴¹ In contrast, in the case of a like domestic finished ICT product that is subject to IPI tax exemption or reduction under the ICT programmes, the seller does not charge any tax or charges a reduced tax, as the case may be, to the wholesaler, retailer,

¹²³⁶ European Union's comments on Brazil's response to Panel question No. 83, para. 9.

¹²³⁷ Panel Reports, para. 2.38.

¹²³⁸ Brazil estimated before the Panel that, following the normal legislative process, "even under the most expeditious scenario the withdrawal of the measures at issue could not be accomplished in less than 18 months from adoption of the DSB recommendations and rulings in this dispute". (Brazil's response to Panel question No. 83) Japan contested the fact that Brazil's normal legislative process must necessarily take longer than 90 days, because "Brazil fails to explain why these processes cannot be accelerated in order to bring itself into compliance with its international legal obligations." (Japan's comments on Brazil's response to Panel question No. 83, para. 4)

¹²³⁹ According to Brazil, legislative action is required to withdraw the measures at issue because it must be the case that the term "withdraw the subsidy" in Article 4.7 requires the permanent withdrawal of the subsidies, and provisional measures would at most affect only "a *temporary* withdrawal of the measures at issue, and [would] not obviate the need to go through Brazil's ordinary legislative process". (Brazil's comments on the European Union's and Japan's responses to Panel question No. 83, para. 3 (emphasis original)) The European Union and Japan both contested Brazil's position, asserting that Brazil had used provisional measures to introduce or modify several of the prohibited subsidies at issue in this dispute and citing multiple factual examples of such modifications. (See European Union's response to Panel question No. 83, para. 4; Japan's comments on Brazil's response to Panel question No. 83, para. 5)

¹²⁴⁰ One Member of the Division expressed a separate opinion regarding the scope of the term "payment of subsidies" in Article III:8(b) of the GATT 1994. This separate opinion can be found in section 5.2.5 of these Reports. ¹²⁴¹ Panel Reports, para. 2.14.

- 146 -

or distributor.¹²⁴² At this last stage, the tax rate is thus higher for imported finished ICT products than for like domestic finished ICT products, and the tax burden on the former is necessarily in excess of that on the latter.

6.3. We therefore <u>uphold</u> the Panel's findings, in paragraphs 7.154, 8.5.a, and 8.16.a of the Panel Reports, that imported finished ICT products are subject to a higher tax burden than like domestic ICT products and are consequently taxed in excess of like domestic finished ICT products, contrary to Article III: 2, first sentence, of the GATT 1994.

6.1.2 Whether the Panel erred in finding that imported intermediate ICT products were taxed in excess of like domestic intermediate ICT products inconsistently with Article III:2, first sentence, of the GATT 1994

6.4. Under the credit-debit system, purchases of non-incentivized imported intermediate ICT products involve the payment of a tax upfront that is not faced by companies that purchase incentivized like domestic intermediate ICT products, which are exempted from the relevant taxes.¹²⁴³ Even in the case of tax reductions, companies purchasing incentivized like domestic intermediate ICT products have to pay a lower tax compared to companies purchasing non-incentivized imported intermediate ICT products. We fail to see how these situations do not have the effect of limiting the availability of cash flow for companies purchasing non-incentivized imported intermediate ICT products. The fact that purchasers of imported intermediate ICT products have to pay the relevant taxes under the ICT programmes, irrespective of the point in time, in comparison to purchasers of incentivized like domestic intermediate ICT products, who do not have to pay the relevant tax or pay a reduced amount, "limit[s] the availability of cash flow"¹²⁴⁴, resulting in a higher effective tax burden on imported intermediate ICT products. Moreover, the value of the tax credit that is generated upon the payment of the relevant tax on the sale of a non-incentivized imported intermediate ICT product will depreciate over time until it is used or adjusted. To that extent, in as much as there is a time lag between the accrual of the tax credit and the adjustment or use thereof, it necessarily results in the value of money (in the form of accrued tax credits) depreciating over time. Therefore, imported intermediate ICT products, the purchase of which is subject to a payment of tax upfront, bear a higher tax burden than that faced by the incentivized like domestic intermediate ICT products, which benefit from tax exemption or reduction.

6.5. We therefore <u>uphold</u> the Panel's findings, in paragraphs 7.172, 8.5.a, and 8.16.a of the Panel Reports, that imported intermediate ICT products are taxed in excess of like domestic incentivized intermediate ICT products contrary to Article III: 2, first sentence, of the GATT 1994.

6.1.3 Whether the Panel erred in finding that the accreditation requirements under the ICT programmes accord treatment less favourable to imported products than that accorded to like domestic products inconsistently with Article III:4 of the GATT 1994

6.6. The aspect of the ICT programmes challenged by the complaining parties as being inconsistent with Article III: 4 of the GATT 1994 concerned the accreditation requirements, the fulfilment of which enabled the obtaining of the relevant tax exemption, reduction, or suspension on the sales or purchases of ICT products. It is undisputed that in order to be eligible for the tax exemption, reduction, or suspension under the ICT programmes, companies must fulfil the accreditation requirements. The accreditation requirements under the ICT programmes therefore result in less favourable treatment for imported ICT products in the form of the differential tax burden that imported ICT products are subjected to by virtue of the fact that foreign producers cannot be accredited under the ICT programmes. The consequence being, as the Panel also noted, that foreign producers "can never qualify for the tax exemptions, reductions or suspensions".¹²⁴⁵ We note that the aspects of the ICT programmes found to be inconsistent with Article III: 2, first sentence, and Article III: 4 are distinct. In the case of Article III: 2, first sentence, the aspect of the ICT products. Whereas, for the purposes of Article III: 4, the aspect of the ICT programmes found to be

¹²⁴² Panel Reports, para. 2.15.

¹²⁴³ Panel Reports, para. 7.170.

¹²⁴⁴ Panel Reports, para. 7.170.

¹²⁴⁵ Panel Reports, para. 7.223.

inconsistent is the accreditation requirements that result in less favourable treatment in the form of the differential tax treatment for imported ICT products.

6.7. We therefore uphold the Panel's findings, in paragraphs 7.225, 8.5.b, and 8.16.c of the Panel Reports, that the accreditation requirements of the ICT programmes, by restricting access to the tax incentives only to domestic products, modify the conditions of competition to the detriment of imported products and result in less favourable treatment being accorded to imported ICT products than to like domestic ICT products inconsistently with Article III: 4 of the GATT 1994.

6.1.4 Whether the Panel erred in finding that the ICT programmes are inconsistent with Article III:4 of the GATT 1994 by virtue of the lower administrative burden on companies purchasing incentivized domestic intermediate products

6.8. Under the credit-debit system, purchasers of imported intermediate ICT products that are not incentivized under the ICT programmes will have to anticipate and pay the full amount of tax due on such imported intermediate ICT products. Although any such tax paid on the purchase of imported intermediate ICT products will generate a corresponding tax credit in favour of the purchaser, nonetheless, offsetting this tax credit entails an administrative burden that is not faced, or faced to a lesser extent, by a purchaser of domestic intermediate ICT products that are incentivized.¹²⁴⁶ This is the case because under the credit-debit system, "if the tax credit cannot be offset by debits after three taxation periods", the process of compensating the tax credit with other federal taxes, or reimbursement thereof can "be burdensome for companies, and can take years."¹²⁴⁷

6.9. We therefore uphold the Panel's findings, in paragraphs 7.255, 8.5.b, and 8.16.c of the Panel Reports, that the ICT programmes are inconsistent with Article III: 4 of the GATT 1994, because they accord to imported intermediate ICT products treatment less favourable than that accorded to like domestic intermediate ICT products, due to the lower administrative burden imposed on firms purchasing incentivized domestic intermediate ICT products.

6.1.5 Whether the Panel erred in finding that the PPBs and other production-step requirements under the ICT programmes are contingent upon the use of domestic goods, inconsistently with Article III:4 of the GATT 1994

6.10. We agree with the Panel that the PPBs and other production-step requirements under the Informatics, PATVD, PADIS, and Digital Inclusion programmes provide an incentive to use domestic ICT products. We therefore uphold the Panel's findings, in paragraphs 7.299, 7.301-7.302, 7.308, 7.311, 7.313, 7.317, 7.319, 8.5.b, and 8.16.c of the Panel Reports, that the Informatics, PATVD, PADIS, and Digital Inclusion programmes accord less favourable treatment to imported intermediate ICT products than that accorded to like domestic products.

6.1.6 Whether the Panel erred in finding that the ICT programmes are inconsistent with Article 2.1 of the TRIMs Agreement

6.11. On appeal, Brazil does not make any specific arguments in connection with the Panel's finding under Article 2.1 of the TRIMs Agreement. Rather, Brazil's request for reversal of the Panel's finding under that provision is premised on us reversing the Panel's findings under Article III:4 of the GATT 1994.¹²⁴⁸ We have, however, for the reasons stated above, upheld the Panel's findings that certain aspects of the ICT programmes are inconsistent with Article III: 4 of the GATT 1994.

¹²⁴⁶ We note that purchasers of incentivized intermediate domestic ICT products, in most cases, will not need to anticipate the tax due on the purchase of such intermediate ICT products since the Informatics, PATVD, PADIS, and Digital Inclusion programmes provide for tax exemptions, through zero rates, to accredited companies selling domestic intermediate ICT products. It is only in the context of the IPI tax reduction provided under the Informatics programme that purchasers of domestic intermediate ICT products may have

to anticipate and pay the reduced amount of IPI tax due. ¹²⁴⁷ Panel Reports, para. 7.251 (referring to Brazil's first written submission to the Panel, para. 702 (DS497)). ¹²⁴⁸ Brazil's appellant's submission, para. 172.

- 148 -

6.12. Consequently, we <u>uphold</u> the Panel's findings, in paragraphs 7.363, 8.5.d, and 8.16.e of the Panel Reports, that those aspects of the ICT programmes found to be inconsistent with Article III:4 of the GATT 1994 are also inconsistent with Article 2.1 of the TRIMs Agreement.

6.1.7 Whether the Panel erred in finding that the accreditation requirements under the INOVAR-AUTO programme are inconsistent with Article III:4 of the GATT 1994 because they are more burdensome for companies seeking accreditation as importers/distributors as opposed to domestic manufacturers

6.13. It is undisputed that, in order for companies to obtain any sort of accreditation under the INOVAR-AUTO programme, which entitles them to accruing and using presumed IPI tax credits, they must either be located and operate in Brazil, in the case of domestic manufacturers and importers/distributors, or be in the process of establishing in the country as domestic manufacturers, in the case of investors.¹²⁴⁹ The only viable way for foreign manufacturers to be able to enjoy the benefit of the presumed IPI tax credits in reducing their IPI tax liability under the INOVAR-AUTO programme is to become accredited as importers/distributors. However, in order to do so, foreign manufacturers must, first and foremost, be located and operate in Brazil.¹²⁵⁰ This indicates that foreign manufacturers seeking accreditation as importers/distributors face a corresponding burden that necessarily comes with having to operate in, or establish themselves in, Brazil, unlike domestic manufacturers, who already operate or are established in Brazil. Moreover, we note that in order to become accredited as importers/distributors, a company shall comply with the following three specific requirements: (i) investments in R&D in Brazil; (ii) expenditure on engineering, basic industrial technology, and capacity-building of suppliers in Brazil; and (iii) participation in the vehicle-labelling programme by INMETRO.¹²⁵¹ A fourth requirement also exists, which calls for the performance in Brazil of certain manufacturing steps.¹²⁵² These activities cannot be considered to be typical for foreign manufacturers seeking to import motor vehicles into Brazil. The fact that foreign manufacturers have to undertake these activities to get accredited as importers/distributors implies that foreign manufacturers face a burden that domestic manufacturers do not face. Almost all of these requirements can be considered to be typical of the nature of activity carried out by a domestic manufacturer. Indeed, any domestic manufacturer will carry out and perform a minimum number of manufacturing activities in Brazil, and, in that process, it is likely to make investments in R&D in Brazil and make expenditures in the categories indicated in the INOVAR-AUTO programme. The INOVAR-AUTO programme is thus designed in such a manner that the accreditation requirements thereunder adversely modify the competitive conditions for imported products in comparison to like domestic products.

6.14. We therefore <u>uphold</u> the Panel's findings, in paragraphs 7.772, 8.6.b, and 8.17.c of the Panel Reports, that, under the INOVAR-AUTO programme, the conditions for accreditation in order to receive presumed tax credits accord less favourable treatment to imported products than that accorded to like domestic products inconsistently with Article III:4 of the GATT 1994.

6.1.8 Whether the Panel erred in finding that the INOVAR-AUTO programme is inconsistent with Article 2.1 of the TRIMs Agreement

6.15. On appeal, Brazil does not make any specific arguments in connection with the Panel's finding under Article 2.1 of the TRIMs Agreement. Rather Brazil's request for reversal of the Panel's finding under Article 2.1 of the TRIMs Agreement is premised on us reversing the Panel's findings under Article III:4 of the GATT 1994.¹²⁵³ We have, however, for the reasons stated above, upheld the Panel's findings that certain aspects of the INOVAR-AUTO programme are inconsistent with Article III:4 of the GATT 1994.

6.16. Consequently, we <u>uphold</u> the Panel's findings, in paragraphs 7.804, 8.6.d, and 8.17.e of the Panel Reports, that those aspects of the INOVAR-AUTO programme found to be inconsistent with Article III:4 of the GATT 1994 are also inconsistent with Article 2.1 of the TRIMs Agreement.

¹²⁴⁹ Panel Reports, para. 7.657 (referring to Brazil's responses to Panel questions Nos. 28 and 57).

¹²⁵⁰ Panel Reports, para. 7.660 (referring to Brazil's response to Panel question No. 57).

¹²⁵¹ Panel Reports, para. 7.658.

¹²⁵² Panel Reports, fn 1056 to para. 7.658.

¹²⁵³ Brazil's appellant's submission, para. 294.

- 149 -

6.2 Article III:8(b) of the GATT 1994¹²⁵⁴

6.17. Insofar as the payment of subsidies exclusively to domestic producers of a given product affects the conditions of competition between such a product and the like imported product, the resulting inconsistency with the national treatment obligation under Article III is justified under Article III:8(b), provided that the conditions thereunder are met. Moreover, conditions for eligibility for the payment of subsidies that define the class of eligible "domestic producers" by reference to their activities in the subsidized products' markets are also justified under Article III:8(b). By contrast, a requirement to use domestic over imported goods in order to have access to the subsidy is not covered by the exception in Article III:8(b) and would therefore continue to be subject to the national treatment obligation in Article III:8(b) and would therefore continue to be subject to the national treatment obligation in Article III:8(b) and sconfirmed by the negotiating history, suggests that the term "payment of subsidies" in Article III:8(b) does not include within its scope the exemption or reduction of internal taxes affecting the conditions of competition between like products. Instead, as noted by the Appellate Body in *Canada – Periodicals*, Article III:8(b) "was intended to exempt from the obligations of Article III only the payment of subsidies which involves the expenditure of revenue by a government".¹²⁵⁵

6.18. The Panel's interpretation and application of Article III:8(b) to the measures at issue obfuscate the distinction between the effects of the payment of a subsidy to a domestic producer on the conditions of competition in the relevant product markets and the conditions for eligibility attaching thereto, on the one hand, and any other effects arising from requirements to use domestic over imported inputs in the production process, on the other hand. Moreover, at no stage did the Panel undertake an assessment of whether the measures at issue constitutes the "payment of subsidies exclusively to domestic producers" within the meaning of Article III:8(b).

6.19. Because of these shortcomings in the Panel's reasoning, we <u>reverse</u> the Panel's overly broad and unqualified findings, in paragraphs 7.87-7.88 of the Panel Reports, that "subsidies that are provided exclusively to domestic producers pursuant to Article **III:8(b)** ... **are not** *per se* exempted from the disciplines of Article III" and that "aspects of a subsidy resulting in product discrimination (including requirements to use domestic goods, as prohibited by Article 3.1 of the SCM Agreement) are not exempted from the disciplines of Article III pursuant to Article III:8(b)." Under a proper interpretation of Article III:8(b), none of the measures at issue in this dispute are capable of being justified under that provision because they all involve the exemption or reduction of internal taxes affecting the conditions of competition between like products and therefore cannot constitute the "payment of subsidies" within the meaning of Article III:8(b).

6.3 Article 3.1(a) of the SCM Agreement

6.20. In its identification of the benchmark treatment for the three categories of the treatments under the PEC and RECAP programmes, the Panel sought to determine the existence of a general rule for companies that structurally accumulate credits. The Panel limited its analysis to seeking to identify the existence of a general rule of taxation to which the challenged treatment would be an exception. Instead of doing so, however, the Panel should have determined the tax treatment of comparably situated taxpayers, as the legal standard under Article 1.1(a)(1)(ii) of the SCM Agreement requires.

6.21. We therefore <u>reverse</u> the Panel's conclusions, in paragraphs 7.1171-7.1172, 7.1186-7.1187, and 7.1199-7.1200 of the Panel Reports, that Brazil has not demonstrated that the tax suspensions are the benchmark for comparison and that the appropriate benchmark is, instead, the treatment applicable to purchases by non-accredited companies of the relevant products. As a result, we also reverse the Panel's findings, in paragraphs 7.1211, 7.1223, and 7.1238, as well as in paragraphs 8.7 and 8.18 of the Panel Reports, that the tax suspensions granted to registered or accredited companies under the PEC and RECAP programmes constitute financial contributions in the form of government revenue otherwise due that is foregone or not collected and are hence subsidies within the meaning of Article 1.1 of the SCM Agreement and contingent upon export performance within the meaning of Article 3.1(a) of the SCM Agreement.

¹²⁵⁴ For the separate opinion on this issue, see section 5.2.5 of these Reports.

¹²⁵⁵ Appellate Body Report, *Canada – Periodicals*, p. 34.

- 150 -

6.4 Article 3.1(b) of the SCM Agreement

6.22. With respect to Brazil's claim that the Panel erred in its comparison of the treatment of intermediate goods and inputs under the ICT programmes with the benchmark treatment by arbitrarily distinguishing between taxpayers within the benchmark, we disagree with Brazil's approach to comparing the challenged treatment to the selected benchmark treatment. As we see it, the Panel did not treat only one subset of taxpayers (i.e. those that are unable to offset the amount of the tax paid during the same taxation period) as the benchmark for the purposes of comparison. Rather, having explained in detail how the mechanism of credits and debits under the principle of non-accumulation works, the Panel concluded that, under the normal rule of general application of Brazil's tax system, there are two possible factual scenarios: one in which the buyer of non-incentivized products will be able to offset the amount of the tax paid during the same taxation period, and the other one when this would not be possible. We consider that, in comparing the challenged tax treatment to the benchmark treatment, the Panel correctly examined both possible factual scenarios that result from the application of the normal rules of Brazil's tax system.

6.23. With respect to Brazil's claim that the Panel erred in finding that cash availability and implicit interest constitute revenue "otherwise due" under Article 1.1(a)(1)(ii) of the SCM Agreement, we consider that, in the present dispute, when the tax exemptions and reductions apply, the Brazilian Government does not collect in full the tax revenue when it normally would, or collects it in part. The fact that, ultimately, the amount of the tax collected under the benchmark treatment and the challenged treatment may nominally be the same does not detract from the fact that, under the benchmark treatment, in the scenario when non-accredited companies are unable to offset their credits immediately, the Brazilian Government would collect and retain, for a certain period, the amount of tax payable to it. During this period of time, the Brazilian Government can enjoy the cash available to it and earn interest on it. By contrast, when tax exemptions and reductions are applied, the Brazilian Government collects the tax later in time and does not enjoy the availability of cash as it otherwise would under the benchmark treatment. Thus, under the challenged treatment, the Brazilian Government would not collect the tax at the time it normally would under the benchmark treatment. By doing so, in our view, the Brazilian Government would not collect the revenue that would be otherwise due to it.

6.24. In light of the above, we do not consider that the Panel erred in finding that the tax treatment of intermediate products and inputs under the ICT programmes constitutes a subsidy under Article 1.1 of the SCM Agreement. We thus <u>uphold</u> the Panel's findings, in paragraphs 7.436, 7.444, 7.454, 7.463, 7.473, 7.489-7.490, and 7.495 of the Panel Reports, that each of the challenged tax exemptions, reductions, and suspensions granted to accredited companies on (i) the sales of intermediate goods that they produce, and (ii) the purchases of raw materials, intermediate goods, and packaging materials (under the Informatics programme) and inputs, capital goods, and computational goods (under the PADIS and PATVD programmes) constitutes financial contributions where "government revenue that is otherwise due is foregone or not collected" under Article 1.1(a)(1)(ii) of the SCM Agreement.

6.25. With respect to the main PPBs that incorporate nested PPBs under the Informatics programme, we consider that they contain a condition requiring the use of domestic components and subassemblies, which must be fulfilled in order for the relevant products to benefit from the tax incentives. We therefore agree with the Panel's conclusion that the main PPBs that incorporate nested PPBs contain a requirement to use domestic goods, i.e. the components and subassemblies covered by the nested PPBs, under Article 3.1(b) of the SCM Agreement. Accordingly, we <u>uphold</u> the Panel's findings, in paragraphs 7.299-7.300, 7.302, 7.313, 7.319, 8.5.e, and 8.16.f of the Panel Reports, that the main PPBs that incorporate nested PPBs under the Informatics programme are inconsistent with Article 3.1(b) of the SCM Agreement.

6.26. With respect to the main PPBs that do not contain nested PPBs under the Informatics programme, we consider that they do not provide for more than a collection of production steps, which must be carried out in Brazil, in order for a company to benefit from the tax incentives with respect to the product subject to the PPB under the relevant programme. Although compliance with the production steps set out in the PPBs is likely to result in the use of domestic components and subassemblies, such use of domestic products will be a consequence of the requirement to perform the production steps in Brazil. Accordingly, we reverse the Panel's findings, in paragraphs 7.301-7.302, 7.313, 7.319, 8.5.e, and 8.16.f of the Panel Reports, that the main PPBs without

- 151 -

nested PPBs under the Informatics programme are contingent upon the use of domestic over imported goods within the meaning of Article 3.1(b) of the SCM Agreement.

6.27. With respect to the PATDV programme, we understand that the PPBs under that programme follow the same structure and logic as those under the Informatics programme. Accordingly, for the same reasons as under the Informatics programme, to the extent the main PPBs under the PATVD programme incorporate nested PPBs, we consider them to require the use of domestic over imported goods inconsistently with Article 3.1(b) of the SCM Agreement. However, we do not consider that the main PPBs that do not contain nested PPBs contain a condition requiring the use of domestic over imported goods. We thus <u>uphold</u> the Panel's findings concerning the PATVD programme, in paragraphs 7.308, 7.313, 7.317, 7.319, 8.5.e, and 8.16.f of the Panel Reports, to the extent they relate to the main PPBs that contain nested PPBs, and we <u>reverse</u> the Panel's findings, contained in the same paragraphs of the Panel Reports, to the extent they relate to the main PPBs.

6.28. With respect to the PADIS programme, we do not consider that the eligibility requirements under that programme constitute a contingency requirement to use domestic over imported goods under Article 3.1(b) of the SCM Agreement.

6.29. With respect to the Digital Inclusion programme, our review of the Panel's findings demonstrates that the Panel did not have a proper basis to conclude that the Digital Inclusion programme contains a requirement to use domestic over imported goods inconsistent with Article 3.1(b) of the SCM Agreement.

6.30. In light of the above, we <u>reverse</u> the Panel's findings, in paragraphs 7.313, 7.317, 7.319, 8.5.e-f of the Panel Reports, that the PADIS and Digital Inclusion programmes require the use of domestic over imported goods inconsistently with Article 3.1(b) of the SCM Agreement.

6.31. With respect to the INOVAR-AUTO programme, we note that the requirement to perform a minimum number of manufacturing steps in Brazil under that programme operates in a way similar to the main PPBs that do not incorporate nested PPBs under the ICT programmes. Thus, having reversed the Panel's findings with respect to the main PPBs that do not incorporate nested PPBs under the ICT programmes, we also <u>reverse</u> the Panel's findings of inconsistency with Article 3.1(b) with respect to the manufacturing steps under the INOVAR-AUTO programme, contained in paragraphs 7.751, 7.823, 7.847, 8.6.e, and 8.17.f of the Panel Reports.

6.32. With respect to the European Union's and Japan's appeals of the Panel's findings concerning the in-house scenario, we note that, for purposes of establishing an inconsistency with Article 3.1(b) of the SCM Agreement, whether a company produces goods in-house or whether it outsources their production is not decisive. What matters, instead, is whether such measure reflects a condition requiring the use of domestic over imported goods. We thus consider that it did not matter, for purposes of the Panel's analysis, what factual scenarios were available for compliance with the requirements under the ICT and INOVAR-AUTO programmes. We thus <u>reverse</u> the Panel's findings, in paragraphs 7.303-7.304 and 7.314 of the Panel Reports, made in the context of its analysis under ICT programmes, to the extent that they can be understood as suggesting that the in-house scenario was not covered by the Panel's findings. We also <u>reverse</u> the Panel's findings, in paragraphs 7.749-7.750, and 7.770 of the Panel Reports, made in the context of INOVAR-AUTO programme and referring to the mentioned Panel's findings under the ICT programmes, to the extent that they can also be understood as suggesting that the in-house scenario was not covered by the Panel's findings under the ICT programmes, to the extent that the Panel's findings under the ICT programmes, to the extent that they can also be understood as suggesting that the in-house scenario was not covered by the Panel's findings, in paragraphs 7.319, 7.772-7.773, 8.5.e, 8.6.b, 8.6.e, 8.16.c, 8.16.f, 8.17.c, and 8.17.f of the Panel Reports, also apply the in-house scenario.

6.5 Article I:1 of the GATT 1994 and the Enabling Clause

6.5.1 Whether the Panel erred in finding that the claims raised by the European Union and Japan under Article I:1 of the GATT 1994 were within its terms of reference

6.33. Paragraph 4(a) of the Enabling Clause envisages a degree of specificity in the notification adopted thereunder. At a minimum, a notification pursuant to paragraph 4(a) should state under which particular provision of the Enabling Clause the differential and more favourable treatment has been adopted so as to put other Members on notice. A notification pursuant to paragraph 4(a) speaks

- 152 -

to and has a direct bearing on a complaining party's knowledge. Consequently, it speaks to whether the complaining party is required to raise the Enabling Clause and identify the relevant provision(s) thereof in its panel request and assert that an arrangement or a measure adopted by the responding party is inconsistent not only with Article I: 1 of the GATT 1994, but also with the relevant provisions of the Enabling Clause. While it is for the complaining party to raise the Enabling Clause and identify the relevant provision(s) thereof in its panel request, the burden to prove that the measure satisfies the conditions set out in the Enabling Clause remains on the responding party relying on the Enabling Clause as a defence.

6.34. With respect to whether the measure at issue (i.e. differential tax treatment under the INOVAR-AUTO programme in the form of internal tax reductions accorded to some but not other Members) was notified pursuant to paragraph 4(a) as having been adopted under paragraph 2(b) of the Enabling Clause, our review of the Panel's findings indicates that Brazil has not demonstrated that the differential tax treatment under the INOVAR-AUTO was notified to the WTO as adopted under paragraph 2(b), as required pursuant to paragraph 4(a) of the Enabling Clause.

6.35. We therefore <u>uphold</u> the Panel's finding, in paragraphs 7.1082-7.1083 of the Panel Reports, that Brazil has not demonstrated that the differential tax treatment under the INOVAR-AUTO programme was notified to the WTO as adopted pursuant to paragraph 2(b), and therefore, in the circumstances of this case, there was no burden on the complaining parties to raise and identify paragraph 2(b) of the Enabling Clause in their panel requests.

6.36. With respect to whether the measure at issue (i.e. the differential tax treatment under the INOVAR-AUTO programme in the form of internal tax reductions accorded to some but not other Members) was notified pursuant to paragraph 4(a) as having been adopted under paragraph 2(c) of the Enabling Clause, our review of the Panel's findings indicates that Brazil has not demonstrated that the 1980 Treaty of Montevideo and the ECAs notified to the WTO have a genuine link to the INOVAR-AUTO programme providing for the differential and more favourable treatment at issue, and consequently, the differential and more favourable treatment at issue was not notified to the WTO as adopted under paragraph 2(c), as required pursuant to paragraph 4(a) of the Enabling Clause.

6.37. We therefore <u>uphold</u> the Panel's finding, in paragraph 7.1119 of the Panel Reports, that the differential tax treatment under the INOVAR-AUTO programme was not notified as adopted under paragraph 2(c), as required pursuant to paragraph 4(a). Consequently, we also <u>uphold</u> the Panel's finding, in paragraph 7.1120 of the Panel Reports, that, in the circumstances of this case, there was no burden on the complaining parties to raise and identify paragraph 2(c) of the Enabling Clause in their panel requests.

6.38. For these reasons, we <u>uphold</u> the Panel's findings, in paragraphs 8.6.h and 8.17.i of the Panel Reports, that there was no burden on the complaining parties to raise and identify the relevant provisions of the Enabling Clause in their panel requests, and their claims under Article I:1 of the GATT 1994 were therefore within the Panel's terms of reference.

6.5.2 Whether the Panel erred in its interpretation of paragraph 2(b) of the Enabling Clause and in finding that the differential tax treatment under the INOVAR-AUTO programme was not justified under that provision

6.39. Paragraph 2(b) provides for the granting of "[d]ifferential and more favourable treatment with respect to the provisions of the General Agreement concerning non-tariff measures governed by the provisions of instruments multilaterally negotiated under the auspices of the GATT". Paragraph 2(b) provides for the adoption of a limited category of differential and more favourable treatment, namely treatment that concerns "non-tariff measures governed by the provisions of instruments multilaterally negotiated under the auspices" of the GATT, as an institution. The phrase "non-tariff measures governed by the provisions of the adoption of the Enabling Clause, concerned non-tariff measures taken pursuant to the S&D treatment provisions of the Tokyo Round Codes and not the provisions of the GATT 1947. Following the entry into force of the WTO Agreement, paragraph 2(b) of the Enabling Clause provides for the adoption of a limited category of differential and more favourable treatment, namely treatment that concerns non-tariff measures governed by the provisions of the matter and the provisions of the GATT 1947. Following the entry into force of the WTO Agreement, paragraph 2(b) of the Enabling Clause provides for the adoption of a limited category of differential and more favourable treatment, namely treatment that concerns non-tariff measures governed by the provisions of instruments multilaterally negotiated under the auspices of the WTO. The GATT 1994, while an integral part of

- 153 -

the WTO Agreement, was not negotiated under the auspices of the WTO. These considerations read in light of the text, context, and circumstances surrounding the adoption of the Enabling Clause, and thereafter the establishment of the WTO, indicate that paragraph 2(b) does not concern non-tariff measures governed by the provisions of the GATT 1994. Instead, paragraph 2(b) speaks to non-tariff measures taken pursuant to S&D treatment provisions of "instruments multilaterally negotiated under the auspices of the [WTO]".

6.40. We therefore <u>uphold</u> the Panel's finding, in paragraph 7.1096 of the Panel Reports, that "a non-tariff measure within the scope of paragraph 2(b) must be governed by specific provisions on special and differential treatment, that are distinct from the provisions of the GATT 1994 incorporating the GATT 1947."¹²⁵⁶ We also <u>uphold</u> the Panel's findings, in paragraphs 7.1097, 8.6.i, and 8.17.j of the Panel Reports, that the tax reductions accorded under the INOVAR-AUTO programme to imported products from Argentina, Mexico, and Uruguay and found to be inconsistent under Article I:1 of the GATT 1994 are not justified under paragraph 2(b) of the Enabling Clause.

6.5.3 Whether the Panel erred in its interpretation of paragraph 2(c) of the Enabling Clause and in finding that the differential tax treatment under the INOVAR-AUTO programme was not justified under that provision

6.41. Paragraph 2(c) excepts differential and more favourable treatment accorded pursuant to "[r]egional or global arrangements entered into amongst" developing country Members from a finding of inconsistency with Article I:1 of the GATT 1994. To the extent that the Panel relied on its earlier analysis concerning whether or not the INOVAR-AUTO programme, according the differential and more favourable treatment (i.e. the differential tax treatment in the form of internal tax reductions accorded to some but not other Members), had a genuine link to "the arrangement notified to the WTO" in determining if the differential and more favourable treatment was substantively justified under paragraph 2(c), we find no error in the Panel's approach. Indeed, if there is no genuine link between the measure at issue according the differential and more favourable treatment sourable treatment and the arrangements notified to the WTO, we find it difficult to see how the measure at issue could be substantively justified under paragraph 2(c).

6.42. We therefore <u>uphold</u> the Panel's finding, in paragraph 7.1118 of the Panel Reports, to the extent that the Panel found that Brazil has not identified any arrangement with a genuine link to the differential tax treatment envisaged under the INOVAR-AUTO programme. Consequently, we also <u>uphold</u> the Panel's findings, in paragraphs 7.1121, 8.6.i, and 8.17.j of the Panel Reports, that the tax reductions accorded under the INOVAR-AUTO programme to imported products from Argentina, Mexico, and Uruguay and found to be inconsistent under Article I:1 of the GATT 1994 are not justified under paragraph 2(c) of the Enabling Clause.

6.6 Article 4.7 of the SCM Agreement

6.43. The term "without delay" in Article 4.7 is not used in the sense of requiring *immediate* compliance. Nor does the term "without delay", combined with the requirement that the panel specify a time period, impose a single standard or time period applicable in all cases. Instead, Article 4.7 requires a panel to specify a time period that constitutes "without delay" within the realm of possibilities in a given case and considering the domestic legal system of the implementing Member. In determining the time period under Article 4.7 that constitutes "without delay", a panel should typically take into account the nature of the measure(s) to be revoked or modified and the domestic procedures available for such revocation or modification. These domestic procedures include any extraordinary procedures that may be available within the legal system of a WTO Member. The Panel did not establish a sufficient link between the time period of 90 days specified by it for the withdrawal of the subsidies at issue and the domestic procedure within Brazil for such withdrawal. Instead, the Panel treated the practice of specifying 90 days by some prior panels as the *de facto* standard to be applied in all cases.

6.44. We, therefore, find that by failing to provide a "reasoned and adequate explanation" or a "basic rationale" in recommending a time period of 90 days under Article 4.7 of the SCM Agreement in the present case, the Panel acted inconsistently with Articles 11 and 12.7 of the DSU. Consequently, we <u>reverse</u> the Panel's conclusions, in paragraphs 8.11 and 8.22 of the Panel Reports,

¹²⁵⁶ Panel Reports, para. 7.1096. (fn omitted)

that Brazil withdraw the prohibited subsidies identified at paragraphs 8.5.e, 8.6.e, 8.7, 8.16.f, 8.17.f, and 8.18 of the Panel Reports within 90 days.

6.45. Furthermore, we are unable to complete the legal analysis, as requested by Brazil, and cannot specify the time period within which the prohibited subsidies identified by the Panel must be withdrawn by Brazil. We note, however, that the Panel's recommendations, in paragraphs 8.10 and 8.21 of the Panel Reports, that Brazil withdraw the subsidies identified by the Panel "without delay", stand undisturbed.

6.7 Recommendation

6.46. The Appellate Body <u>recommends</u> that the DSB request Brazil to bring its measures, found in these Reports, and in the Panel Reports as modified by these Reports, to be inconsistent with Articles I:1, III:2, and III:4 of the GATT 1994, Article 2.1 of the TRIMs Agreement, and Article 3.1(b) of the SCM Agreement into conformity with its obligations under those Agreements.

Signed in the original in Geneva this 15th day of October 2018 by:

Thomas R. Graham

Presiding Member

Peter Van den Bossche Member

Hong Zhao Member



13 December 2018

Page: 1/42

Original: English

BRAZIL - CERTAIN MEASURES CONCERNING TAXATION AND CHARGES

AB-2017-7 AB-2017-8

Reports of the Appellate Body

Addendum

This *Addendum* contains Annexes A to D to the Report of the Appellate Body circulated as documents WT/DS472/AB/R, WT/DS497/AB/R.

The Notices of Appeal and Other Appeal and the executive summaries of written submissions contained in this Addendum are attached as they were received from the participants and third participants. The content has not been revised or edited by the Appellate Body, except that paragraph and footnote numbers that did not start at one in the original may have been re-numbered to do so, and the text may have been formatted in order to adhere to WTO style. The executive summaries do not serve as substitutes for the submissions of the participants and third participants in the Appellate Body's examination of the appeal.

(18-7794)

- 2 -

LIST OF ANNEXES

ANNEX A

NOTICES OF APPEAL AND OTHER APPEALS

	Contents	Page
Annex A-1	Brazil's Notice of Appeal	4
Annex A-2	The European Union's Notice of Other Appeal	11
Annex A-3	Japan's Notice of Other Appeal	12

ANNEX B

ARGUMENTS OF THE PARTICIPANTS

	Contents	Page
Annex B-1	Executive summary of Brazil's appellant's submission	14
Annex B-2	Executive summary of the European Union's other appellant's submission	23
Annex B-3	Executive summary of Japan's other appellant's submission	24
Annex B-4	Executive summary of Japan's appellee's submission	26
Annex B-5	Executive summary of the European Union's appellee's submission	30
Annex B-6	Executive summary of Brazil's appellee's submission	36

ANNEX C

ARGUMENTS OF THE THIRD PARTICIPANTS

	Contents	Page
Annex C-1	Executive summary of Argentina's third participant's submission	38
Annex C-2	Executive summary of Australia's third participant's submission	39
Annex C-3	Executive summary of the United States' third participant's submission	40

ANNEX D

PROCEDURAL RULING

	Contents	Page
Annex D-1	Procedural ruling of 12 October 2017 regarding the deadline for filing of third participants' submissions	42

- 3 -

ANNEX A

NOTICES OF APPEAL AND OTHER APPEALS

	Contents	Page
Annex A-1	Brazil's Notice of Appeal	4
Annex A-2	The European Union's Notice of Other Appeal	11
Annex A-3	Japan's Notice of Other Appeal	12

- 4 -

ANNEX A-1

BRAZIL'S NOTICE OF APPEAL*

1. Pursuant to Articles 16.4 and 17 of the Understanding on Rules and Procedures Governing the Settlement of Disputes ("DSU") and Rule 20 of the Working Procedures for Appellate Review (WT/AB/WP/6) ("Working Procedures"), Brazil hereby notifies the Dispute Settlement Body of its decision to appeal certain issues of law and legal interpretation in the reports of the Panel in Brazil – Certain Measures Concerning Taxation and Charges (WT/DS472, DS497) ("Panel Report").

2. The seven programmes at issue are: (a) the Informatics programme; (b) the programme of Incentives for the Semiconductors Sector ("PADIS programme"); (c) the programme of Support for the Technological Development of the Industry of Digital TV Equipment ("PATVD programme"); (d) the programme for Digital Inclusion ("Digital Inclusion programme"); (e) the programme of Incentive to the Technological Innovation and Densification of the Automotive Supply Chain ("INOVAR-AUTO"); (f) the regime for predominantly exporting companies ("PEC programme"); and the Special Regime for the Purchase of Capital Goods for Exporting Enterprises ("RECAP programme").¹

3. The issues that Brazil raises in this appeal relate to the Panel's findings and conclusions with respect to the consistency of the challenged programmes with various provisions of the General Agreement on Tariffs and Trade 1994 (GATT 1994), the Agreement on Subsidies and Countervailing Measures (SCM Agreement), and the Agreement on Trade-Related Investment Measures (TRIMS Agreement).

4. Pursuant to Rules 20(1) and 21(1) of the Working Procedures, Brazil files this Notice of Appeal together with its Appellant's Submission with the Appellate Body Secretariat.

5. Pursuant to Rule 20(2)(d)(iii) of the Working Procedures, this Notice of Appeal provides an indicative list of the paragraphs of the Panel Report containing the alleged errors of law and legal interpretation by the Panel, without prejudice to Brazil's ability to rely on other paragraphs of the Panel Report in its appeal.

I. REVIEW OF THE PANEL'S INTERPRETATION AND APPLICATION OF ARTICLE III:8(b) OF THE GATT 1994 TO THE INFORMATICS, PADIS, PATVD, DIGITAL INCLUSION PROGRAMMES ("ICT PROGRAMMES") AND THE INOVAR-AUTO PROGRAMME

6. Brazil seeks review by the Appellate Body of the Panel's finding that domestic production subsidies within the meaning of Article III:8(b) of the GATT 1994 are not exempt from the disciplines of Article III thereof. The Panel's errors of law and legal interpretation include:

- The Panel erred in finding that the scope of Article III:8(b) of the GATT 1994 is limited to the provision of subsidies to domestic producers where those subsidies do not have any component that introduces discrimination between imported and domestic products;²
- The Panel erred in finding that Article III:8(b) of the GATT 1994 only serves to clarify that Members are not required to subsidize foreign producers in tandem with domestic producers;³
- The Panel erred by failing to determine, in the first instance, whether the measures at issue are product-related measures subject to the disciplines of Article III of the

 $^{^{\}ast}$ This document, dated 28 September 2017, was circulated to Members as document WT/DS472/8 - WT/DS497/6.

¹ The complainants explained in their panel requests that these programmes are "set up and implemented", "established and administered" through the measures identified in paragraph 2.38 of the Panel Report.

² See Panel Report, paras. 7.83-7.87.

³ See Panel Report, para. 7.84.

GATT 1994 or, conversely, whether the measures provide subsidies to domestic producers and are therefore subject to the disciplines of the SCM Agreement.

7. For these reasons, Brazil respectfully requests that the Appellate Body *reverse* the Panel's interpretative finding, in paragraphs 7.87, 8.3, and 8.14 of the Panel Report, that subsidies that are provided exclusively to domestic producers pursuant to Article III:8(b) of the GATT 1994 are not *per se* exempted from the disciplines of Article III of the GATT. As a result, Brazil also requests that the Appellate Body *reverse* all of the Panel's consequential findings to this effect in paragraphs 7.93, 7.184, 7.631, and 7.696 of the Panel Report.

8. Brazil respectfully requests that the Appellate Body *find* that the Panel erred by failing to determine, in the first instance, whether the measures at issue are product-related measures subject to the disciplines of Article III of the GATT 1994 or, conversely, whether the measures provide subsidies to domestic producers and are therefore subject to the disciplines of the SCM Agreement.

9. Consequently, Brazil also requests that the Appellate Body *reverse* the Panel's findings in paragraphs 7.174, 7.318, 7.319, 7.688, 7.772, 7.773, 8.5(a), 8.5(b), 8.6(a), 8.6(b), 8.16(a), 8.16(c), 8.17(a), and 8.17(c) of the Panel Report that the ICT programmes and INOVAR-AUTO are inconsistent with Articles III:2 and III:4 of the GATT 1994. The Appellate Body should also *reverse* the Panel's findings in paragraphs 7.365, 7.806, 8.5(d), 8.6(d), 8.16(e), and 8.17(e) of the Panel Report that the ICT programmes and INOVAR-AUTO are inconsistent with Article 2.1 of the TRIMs Agreement.

II. REVIEW OF THE PANEL'S INTERPRETATION AND APPLICATION OF ARTICLE III:2 OF THE GATT 1994 TO THE ICT PROGRAMMES

10. Brazil seeks review by the Appellate Body of the Panel's finding that production-step requirements under the ICT programmes, and the requirement for products to obtain the status of "developed" in Brazil, result in imported products being subject to internal taxes in excess of those applied to like domestic products, inconsistently with Article III: 2, first sentence, of the GATT 1994. The Panel's errors of law and legal interpretation include:

- The Panel erred in failing to determine that the ICT programmes constitute "payments of subsidies exclusively to domestic producers" within the meaning of Article III:8(b) of the GATT 1994;⁴
- The Panel erred in finding that the complainants had demonstrated that imported products were taxed in excess of domestic like products under the ICT programmes within the meaning of Article III: 2; ⁵
- The Panel erred in finding that the "cash flow" and the "time-value of money" effects under the ICT programmes result in a higher tax burden on imported intermediate products.⁶

11. For these reasons, Brazil requests further that the Appellate Body *reverse* the Panel's findings in paragraphs 7.174, 8.5(a), and 8.16(a) of the Panel Report that the ICT programmes result in imported products being subject to internal taxes in excess of those applied to like domestic products, inconsistently with Article III:2, first sentence, of the GATT 1994.

12. As a result, Brazil also requests that the Appellate Body *reverse* the Panel's consequential finding in paragraphs 7.365, 8.5(d), and 8.16(e) of the Panel Report that the ICT programmes are inconsistent with Article 2.1 of the TRIMS Agreement.

III. REVIEW OF THE PANEL'S INTERPRETATION AND APPLICATION OF ARTICLE III:4 OF THE GATT 1994 TO THE ICT PROGRAMMES

13. Brazil seeks review by the Appellate Body of the Panel's finding that the ICT programmes accord to imported products treatment less favourable than that accorded to like domestic products,

⁴ See Panel Report, paras. 7.77-7.87, 7.93.

⁵ See Panel Report, para. 7.151.

⁶ See Panel Report, paras. 7.170-7.172.

- 6 -

inconsistently with Article III: 4 of the GATT 1994. The Panel's errors of law and legal interpretation include:

- The Panel erred in failing to determine that the ICT programmes constitute "payments of subsidies exclusively to domestic producers" within the meaning of Article III:8(b) of the GATT 1994;⁷
- The Panel erred in finding that the accreditation requirements under the ICT programmes are inconsistent with Article III:4 of the GATT 1994;⁸
- The Panel erred in finding that the ICT programmes are inconsistent with Article III: 4 of the GATT 1994 by virtue of purported administrative burdens on intermediate products;⁹
- The Panel erred in finding that the PPBs and other production step requirements under the ICT programmes are inconsistent with Article III: 4 of the GATT 1994 because they require the use domestic products.¹⁰

14. For these reasons, Brazil respectfully requests that the Appellate Body *reverse* the Panel's finding in paragraphs 7.318, 7.319, 8.5(b), and 8.16(c) of the Panel Report that the ICT programmes accord to imported products treatment less favourable than that accorded to like domestic products, inconsistently with Article III: 4 of the GATT 1994.

15. As a result, Brazil also requests that the Appellate Body reverse the Panel's consequential finding in paragraphs 7.365, 8.5(d), and 8.16(e) of the Panel Report that the ICT programmes are inconsistent with Article 2.1 of the TRIMS Agreement.

IV. REVIEW OF THE PANEL'S INTERPRETATION AND APPLICATION OF ARTICLE 3.1(b) OF THE SCM AGREEMENT TO THE ICT PROGRAMMES

16. Brazil seeks review by the Appellate Body of the Panel's finding that the ICT programmes provide subsidies that are contingent upon the use of domestic over imported goods under

Article 3.1(b) of the SCM Agreement. The Panel's errors of law and legal interpretation include:

- The Panel erred in finding that the tax exemptions, suspensions and reductions granted under the ICT programmes both on the sales of intermediate goods and on the purchases of raw materials, intermediate goods, packaging materials, inputs, capital goods and computational tools constitute financial contributions, in the form of government revenue otherwise due that is foregone or not collected, within the meaning of Article 1.1(a)(1)(ii) of the SCM Agreement;¹¹
- The Panel erred in finding that the tax exemptions, reductions and suspensions granted under the ICT programmes are prohibited subsidies within the meaning of Article 3.1(b) of the SCM Agreement;¹²

17. For these reasons, Brazil respectfully requests that the Appellate Body *reverse* the Panel's findings in paragraphs 7.489 and 7.490 that the exemptions, suspensions and reductions of taxes on the sales of intermediate products and on the purchases of inputs by accredited companies under the ICT programmes constitute financial contributions in the form of "government revenue that is otherwise due is foregone or not collected" within the meaning of Article 1.1(a)(1)(iii) of the SCM Agreement.

⁷ See Panel Report, paras. 7.77-7.87, 7.184.

⁸ See Panel Report, para. 7.220-7.225.

⁹ See Panel Report, paras. 7.252-7.255.

¹⁰ See Panel Report, paras. 7.288, 7.299-7.301; and 7.311-7-313.

¹¹ See Panel Report, paras. 7.432-7.433, 7.444-7.445; 7.454-7.455; 7.463-7.464; 7.473-7.474; and 7.495.

¹² See Panel Report, paras. 7.274-7.302, 7.319.

- 7 -

18. Consequently, the Appellate Body should also *reverse* the Panel's finding in paragraphs 7.493 and 7.494 that the exemption, reduction and suspension of taxes on sales of intermediate products and on purchases of inputs by accredited companies under the ICT programmes confer a benefit on the recipient under Article 1.1(b) of the SCM Agreement. As a result, the Appellate Body should also *reverse* the Panel's ultimate conclusion in paragraphs 7.495, 8.5(e), and 8.16(f) that the that the exemption, reduction and suspension of taxes on sales of intermediate products and on purchases of inputs by accredited companies under the ICT programmes constitutes subsidies within the meaning of Article 1 of the SCM Agreement.

19. Brazil further requests that the Appellate Body *reverse* the Panel's finding in paragraphs 7.500, 8.5(e), and 8.16(f) that the exemptions, reductions, and suspensions granted under the ICT programmes are subsidies contingent upon the use of domestic over imported goods within the meaning of Article 3.1(b) of the SCM Agreement, and thus are prohibited subsidies, inconsistent with Article 3.1(b) and 3.2 of the SCM Agreement.

20. Consequently, Brazil also respectfully requests that the Appellate Body *declare moot and of no legal effect* the Panel's recommendation under Article 4.7 of the SCM Agreement that Brazil withdraw the ICT programs in 90 days, as reflected in paragraphs 8.11 and 8.22 of the Panel Report.

V. REVIEW OF THE PANEL'S INTERPRETATION AND APPLICATION OF ARTICLE III:4 OF THE GATT 1994 TO THE INOVAR-AUTO PROGRAMME

21. Brazil seeks review by the Appellate Body of the Panel's finding that certain aspects of the accreditation requirements under the INOVAR-AUTO Programme related to investments in R&D expenditures on engineering, basic industrial technology and to the performance of certain manufacturing steps in Brazil accord less favourable treatment to imported products than that accorded to like domestic products, inconsistently with Article III:4 of the GATT 1994. The Panel's errors of law and legal interpretation include:

- The Panel erred in failing to determine that the INOVAR-AUTO Programme constitutes "payments of subsidies exclusively to domestic producers" within the meaning of Article III:8(b);¹³
- The Panel erred in failing to conclude that because the accreditation requirements under the INOVAR-AUTO Programme do not relate to products, they cannot be *per se* inconsistent with Article III: 4 of the GATT 1994 or Article 2.1 of the TRIMs Agreement; ¹⁴
- The Panel erred in finding that accreditation requirements to perform a minimum number of manufacturing steps in Brazil are *per se* inconsistent with Article III:4 of the GATT 1994;¹⁵
- The Panel erred in finding that the different accreditation processes for companies seeking accreditation as "domestic manufacturers" and companies seeking accreditation as "investors" or "importers/distributors" would *per se* result in less favourable treatment being accorded to imported finished motor vehicles, under Article III: 4.¹⁶

22. For these reasons, Brazil respectfully requests that the Appellate Body *reverse* the Panel's finding, in paragraph 7.696 of the Panel Report, that Article III:8(b) of the GATT 1994 "does not exempt from the disciplines of Article III components of [domestic] production subsidies that introduce product discrimination in the form of less favourable treatment on imported like products."

23. Brazil also requests that the Appellate Body *reverse* the Panel's finding in paragraphs 7.773, 8.6(b) and 8.17(c) of the Panel Report that the accreditation requirements under INOVAR-AUTO to: (a) perform a minimum number of manufacturing steps in Brazil; and (b) invest in R&D in Brazil and make expenditures on engineering, basic industrial technology and capacity-building of suppliers in

¹³ See Panel Report, para. 7.696.

¹⁴ See Panel Report, para. 7.70.

¹⁵ See Panel Report, paras. 7.737-7.751.

¹⁶ See Panel Report, paras. 7.658-7.661, 7.732.

Brazil accord less favourable treatment to imported products than that accorded to like domestic products, inconsistently with Article III:4 of the GATT 1994.

24. As a result, Brazil requests that the Appellate Body also reverse the Panel's consequential finding in paragraphs 7.806, 8.6(d), and 8.17(e) of the Panel Report that these aspects of the accreditation requirements are inconsistent with Article 2.1 of the TRIMS Agreement.

25. To the extent that the Panel's findings are based on the erroneous assumption that production step requirements are sufficient to establish a contingency upon the use of domestic over imported goods within the meaning of Article 3.1(b) of the SCM Agreement, Brazil requests that the Appellate Body *reverse* the Panel's findings in paragraphs 7.751, 7.823, 7.847, 8.6(e), and 8.17(f) of the Panel Report that INOVAR-AUTO constitutes a prohibited import substitution subsidy under Article 3.1(b) and 3.2 of the SCM Agreement.

VI. REVIEW OF THE PANEL'S FINDINGS INTERPRETATION AND APPLICATION OF ARTICLES 1.1 AND 3.1(a) OF THE SCM AGREEMENT TO THE TAX SUSPENSIONS GRANTED UNDER THE PEC AND RECAP PROGRAMMES

26. Brazil seeks review by the Appellate Body of the Panel's finding that the tax suspensions granted under the PEC and RECAP programmes are subsidies within the meaning of Article 1.1 of the SCM Agreement and contingent upon export performance within the meaning of Article 3.1(a) of the SCM Agreement. The Panel's errors of law and legal interpretation include:

- The Panel erred in failing to use the tax treatment of structural tax accumulators as the benchmark treatment;¹⁷
- The Panel erred in its comparison of its preferred benchmark treatment with the challenged treatment under PEC and RECAP;¹⁸
- The Panel erred in finding that "cash availability" and "implicit interest income" are government revenue otherwise due;¹⁹
- The Panel acted inconsistently with Article 11 of the DSU in its assessment of the evidence of export contingency;²⁰

27. For these reasons, Brazil respectfully requests that the Appellate Body *reverse* the Panel's conclusions in paragraph 7.1211 of the Panel Report that the PEC and RECAP programmes constitute financial contributions in the form of government revenue that is otherwise due and is foregone or not collected within the meaning of Article 1.1(a)(1)(ii) of the SCM Agreement.

28. As a result, the Appellate Body should also *reverse* the Panel's finding in paragraph 7.1212 of the Panel Report that the PEC and RECAP programmes confer a benefit under Article 1.1(b) of the SCM Agreement, and its ultimate conclusion in paragraphs 7.1223, 8.7, and 8.18 that the tax suspensions under the PEC and RECAP programmes constitute subsidies within the meaning of Article 1.1 of the SCM Agreement.

29. Brazil respectfully requests that the Appellate Body *find* that the Panel acted inconsistently with its duty to conduct an objective assessment of the matter under Article 11 of the DSU in examining the evidence on export contingency.

30. Brazil also respectfully requests that the Appellate Body *reverse* the Panel's finding in paragraphs 7.1238, 8.7, and 8.18 that the tax suspensions granted under the PEC and RECAP programmes are subsidies that are contingent upon export performance within the meaning of Article 3.1(a) of the SCM Agreement.

¹⁷ See Panel Report, paras. 7.1167, 7.1169, and 7.183.

¹⁸ See Panel Report, paras. 7.1175, 7.1179, 7.1203, 7.1207.

¹⁹ See Panel Report, para. 7.1179, 7.1190, 7.1194.

²⁰ See Panel Report, para. 7.1234.

- 9 -

31. As a consequence, Brazil respectfully requests that the Appellate Body declare *moot and of no legal effect* the Panel's recommendation under Article 4.7 of the SCM Agreement in paragraphs 8.11 and 8.22 that Brazil withdraws these alleged subsidies within 90 days.

VII. REVIEW OF THE PANEL'S INTERPRETATION AND APPLICATION OF THE ENABLING CLAUSE TO THE TAX TREATMENT OF MERCOSUR COUNTRIES UNDER INOVAR-AUTO

32. Brazil seeks review by the Appellate Body of the Panel's finding that the complainants' claims under Article I:1 of the GATT 1994 were within the Panel's terms of reference, that the differential tax treatment accorded to Mexico, Argentina, and Uruguay under INOVAR-AUTO was not properly notified to the WTO, and that this differential tax treatment is not justified under paragraphs 2(b) and 2(c) of the Enabling Clause. The Panel's errors of law and legal interpretation include:

- The Panel erred in finding that the complainants were not on notice that the differential tax treatment at issue was adopted (and, in Brazil's view, justified) under the Enabling Clause, such that the complainants were required to invoke the Enabling Clause in their panel requests in order for the complainants claims under Article I:1 of the GATT 1994 to be within the Panel's terms of reference;²¹
- The Panel erred in finding that Brazil had not properly notified the differential tax treatment at issue under paragraph 4(a) of the Enabling Clause; ²²
- The Panel erred in finding that for a non-tariff measure to be within the scope of paragraph 2(b) of the Enabling Clause, it must be governed by specific provisions on special and differential treatment that are distinct from the provisions of the GATT 1994 incorporating the GATT 1947;²³
- The Panel erred by conflating the substantive analysis under paragraph 2(c) with the obligation to notify under paragraph 4(a), concluding that the differential tax treatment at issue is not justified under paragraph 2(c) of the Enabling Clause for the same reason that it concluded that the differential tax treatment was not properly notified.²⁴

33. For these reasons, Brazil respectfully requests that the Appellate Body *reverse* the Panel's conclusions in paragraphs 7.1083, 7.1120, 8.6(h), and 8.17(i) that the complainants did not have an obligation to invoke the Enabling Clause in their panel requests, and reverse the Panel's consequential conclusion that the complaints' claims under Article I:1 of the GATT 1994 were within the Panel's terms of reference.

34. Consequently, Brazil requests that the Appellate Body *declare moot and of no legal effect* the Panel's findings under paragraphs 8.6(g) and 8.17(h) that the tax reductions accorded to imported products from Argentina, Mexico and Uruguay under INOVAR-AUTO are advantages granted by Brazil to products originating in those countries, which are not accorded immediately and unconditionally to like products originating in other WTO Members, inconsistently with Article I: 1 of the GATT 1994.

35. In case the Appellate Body were to find that the complainants' claim under Article I:1 was properly within the Panel's terms of reference, Brazil respectfully requests that the Appellate Body *reverse* the Panel's conclusions in paragraphs 7.1079, 7.1081, 7.1083, 7.115 and 7.119 that the differential tax treatment at issue was not properly notified under paragraph 4(a) of the Enabling Clause.

36. Brazil respectfully requests that the Appellate Body *reverse* the Panel's conclusion in paragraphs 7.1096 that a non-tariff measure within the scope of paragraph 2(b) of the Enabling Clause must be governed by specific provisions on special and differential treatment that are distinct from the provisions of the GATT 1994 incorporating the GATT 1947. Brazil also requests that the Appellate Body *reverse* the Panel's consequential finding in paragraphs 7.1097, 8.6(i), and 8.17(j) that the tax reductions accorded to imported products from Argentina, Mexico and Uruguay and

²² See Panel Report, paras. 7.1076-7.1083, 7.1108-7.1115, 7.1119.

²¹ See Panel Report, paras. 7.1076-7.1083, 7.1108-7.1115, 7.1120.

²³ See Panel Report, paras. 7.1088-7.1097.

²⁴ See Panel Report, paras. 7.1108-7.1118.

found to be inconsistent under Article I:1 of the GATT 1994 are not justified under paragraph 2(b) of the Enabling Clause.

37. Brazil further requests that the Appellate Body *complete the analysis* and find that the differential tax treatment is justified under paragraph 2(b) and complies with the requirements of paragraph 3 of the Enabling Clause.

38. Brazil respectfully requests that the Appellate Body *reverse* the Panel's conclusion in paragraph 7.1118 that Brazil did not meet its burden of proof in respect of the substantive requirements of paragraph 2(c) of the Enabling Clause, and the Panel's ultimate conclusion in paragraphs 8.6(i) and 8.17(j) that the tax reductions accorded to imported products from Argentina, Mexico and Uruguay and found to be inconsistent under Article I:1 of the GATT 1994 are not justified under paragraph 2(c) of the Enabling Clause.

39. Brazil further requests that the Appellate Body *complete the analysis* and find that the differential tax treatment is justified under paragraph 2(c) and complies with the requirements of paragraph 3 of the Enabling Clause.

VIII. REVIEW OF THE PANEL'S RECOMMENDATION UNDER ARTICLE 4.7 OF THE SCM AGREEMENT

40. If the Appellate Body were to uphold the Panel's findings that the measures at issue are prohibited subsidies under Articles 3.1(a) and 3.1(b) of the SCM Agreement, Brazil then seeks review by the Appellate Body of the Panel's recommendation that Brazil withdraw the subsidies identified in paragraphs 8.5(e), 8.6(e), 8.7, 8.16(f), 8.17(f), and 8.18 of the Panel's Report within 90 days.²⁵ The Panel's errors of law and legal interpretation include:

- The Panel erred by failing to provide "reasoned and adequate explanations and coherent reasoning" for its 90-day recommendation, as required by Article 11 of the DSU;²⁶
- The Panel erred by failing to provide the "basic rationale" for its 90-day recommendation, as required by Article 12.7 of the DSU;²⁷

41. For these reasons, Brazil respectfully requests that the Appellate Body *find* that the Panel acted inconsistently with Articles 11 and 12.7 of the DSU in making its recommendations under Article 4.7 of the SCM Agreement and *reverse* the Panel's recommendation in paragraphs 8.11 and 8.22 of the Panel Report that Brazil withdraw the subsidies identified in paragraphs 8.5(e), 8.6(e), 8.7, 8.16(f), 8.17(f), and 8.18 within 90 days.

42. Brazil further requests that the Appellate Body *complete the analysis*, and specify a time period of 18 months for implementation pursuant to Article 4.7 of the SCM Agreement.

 $^{^{25}}$ Brazil notes that the Panel refers to the prohibited subsidies in paragraphs 8.5(f) and paragraphs 8.6(f) in paragraphs 8.11 and 8.22 of the Panel Report, but the relevant paragraphs are in fact 8.5(e) and 8.6(e).

²⁶ See Panel Report, para. 6.17.

²⁷ See Panel Report, para. 6.17.

- 11 -

ANNEX A-2

THE EUROPEAN UNION'S NOTICE OF OTHER APPEAL*

Pursuant to Article 16.4 and Article 17.1 of DSU, the European Union hereby notifies to the Dispute Settlement Body its decision to appeal to the Appellate Body certain issues of law covered in the Panel Report and certain legal interpretations developed by the Panel in the dispute *Brazil – Certain Measures Concerning Taxation and Charges* (WT/DS472). Pursuant to Rule 23(1) of the Working Procedures for Appellate Review, the European Union simultaneously files this Notice of Other Appeal with the Appellate Body Secretariat.

For the reasons to be further elaborated in its submissions to the Appellate Body, the European Union appeals, and requests the Appellate Body to reverse or modify the findings and conclusions of the Panel with respect to the errors of law and legal interpretations contained in the Panel Report described below, and where indicated to complete the analysis on the basis of the Panel's findings and uncontested facts on the record:¹

- 1. The Panel failed to make an objective assessment of the matter before it in accordance with Article 11 of DSU by exercising false judicial economy in relation to the European Union's request to make findings also on the in-house scenario in the context of the ICT and INOVAR-AUTO programmes.² Thus, the European Union requests the Appellate Body *to reverse* the Panel's findings, *to complete the legal analysis* with regard to the in-house scenario and *to find* that the production-step requirements contained in the ICT and INOVAR-AUTO programmes are inconsistent with Articles III: 4 of GATT 1994, Article 2.1 of TRIMS Agreement and Article 3.1(b) of SCM Agreement also under the in-house scenario.
- 2. In the alternative, were the Appellate Body to consider that the Panel correctly exercised judicial economy when failing to make specific findings on the in-house scenario, the European Union requests the Appellate Body *to review*, pursuant to Article 17.6 of DSU, the legal interpretations developed by the Panel and modify, pursuant to Article 17.13 of DSU, the findings contained in paragraphs 7.298-7.314 and 7.746-7.751 of its Report, so as *to make it clearer* that the Panel indeed did not need to rule twice on the production step requirements (in the in-house and in the outsourcing scenarios) because the Panel had already found that those steps are *per se* inconsistent with the covered agreements, since they make an advantage/subsidy contingent upon the use of domestic over imported goods, regardless of whether those domestic goods are manufactured by the accredited company or a third party.³
- 3. Finally, the European Union makes a subordinate claim of error, subject to the Appellate Body rejecting both the first claim of error formulated by the European Union (that the Panel exercised false judicial economy) and the alternative claim of appeal. In that case, the European Union requests the Appellate Body to find that the Panel, by failing to consider the European Union's claims under Article III:4 of GATT 1994, Article 2.1 of TRIMS Agreement and Article 3.1(b) of SCM Agreement in light of all the relevant facts of the case (which included both the in-house and outsourcing scenarios), erred in the application of those provisions.⁴ Accordingly, the European Union requests the Appellate Body *to reverse* the Panel's findings, *to complete the legal analysis* with regard to the in-house scenario and *to find* that the production-step requirements contained in the ICT and INOVAR-AUTO programmes are inconsistent with Articles III:4 of GATT 1994, Article 2.1 of TRIMS Agreement and Article 3.1(b) of SCM Agreement also under the in-house scenario.

^{*} This document, dated 3 October 2017, was circulated to Members as document WT/DS472/9.

¹ Pursuant to Rule 23(2)(c)(ii)(C) of the Working Procedures for Appellate Review this Notice of Other Appeal includes an indicative list of the paragraphs of the Panel Report containing the alleged errors, without prejudice to the ability of the European Union to refer to other paragraphs of the Panel Report in the context of its appeal.

² Panel Report, paras. 6.10, 6.11 and 6.13, 7.314, 7.749, and 8.5.

³ See Panel Report, paras. 6.11, final sentence, 7.292, footnote 683, 7.319, penultimate sentence, 7.500 and 8.5.

⁴ Panel Report, paras. 6.10, 6.11 and 6.13, 7.314, 7.749, and 8.5.

- 12 -

ANNEX A-3

JAPAN'S NOTICE OF OTHER APPEAL*

Pursuant to Article 16.4 and Article 17 of the Understanding on Rules and Procedures Governing the Settlement of Disputes ("DSU") and Rule 23 of the Working Procedures for Appellate Review, Japan hereby notifies its decision to appeal to the Appellate Body certain issues of law covered in the Panel Report in Brazil – Certain Measures Concerning Taxation and Charges (WT/DS497/R) and certain legal interpretations developed by the Panel in this dispute.

In particular, Japan seeks review by the Appellate Body of the Panel's findings that it did not need to review the WTO-consistency of the challenged INOVAR-AUTO and ICT programmes under an "inhouse" scenario. The Panel's findings to this effect are in error and are based on an erroneous interpretation and application of Article 11 of the DSU, by erroneously exercising judicial economy not to extend findings to the so-called "in-house" scenario. Alternatively, the Panel failed to provide "coherent reasoning as required under Article 11 of the DSU. The paragraphs relating to these errors include paragraphs 6.11, 7.302, 7.308, 7.314 and 7.749 of the Panel Report.

Japan respectfully requests the Appellate Body to reverse and modify the related findings, conclusions and recommendations of the Panel. Moreover, to the extent that it reverses the Panel's findings in this respect, Japan also requests that the Appellate Body complete the legal analysis by extending the Panel's findings regarding the "outsourcing" scenario to the "in-house" scenario, and finding that the accreditation and production step requirements under the INOVAR-AUTO programme and ICT programmes "as a whole" – i.e., under both the "outsourcing" and "in-house" scenarios – are inconsistent with Article III:4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement.

In the alternative, should the Appellate Body consider that the Panel did not exercise judicial economy, Japan respectfully requests the Appellate Body to review, pursuant to Article 17.6 of DSU, the legal interpretations developed by the Panel and modify, pursuant to Article 17.13 of the DSU, the findings contained in paragraphs 7.298-7.314 and 7.746-7.751 of the Report, to the extent that they may be understood as referring solely to the "outsourcing" scenario, and clarify that the Panel's findings of the WTO-inconsistency of the challenged INOVAR-AUTO programme and ICT programmes are indeed not limited to the "outsourcing" scenario, but that the Panel made a finding with respect to the accreditation requirement under INOVAR-AUTO programme as a whole, as well as the production-step requirements of the ICT programmes as a whole. To this end, Japan also requests that the Appellate Body complete the analysis, where appropriate.

^{*} This document, dated 3 October 2017, was circulated to Members as document WT/DS497/7.

- 13 -

ANNEX B

ARGUMENTS OF THE PARTICIPANTS

	Contents	Page
Annex B-1	Executive summary of Brazil's appellant's submission	14
Annex B-2	Executive summary of the European Union's other appellant's submission	23
Annex B-3	Executive summary of Japan's other appellant's submission	24
Annex B-4	Executive summary of Japan's appellee's submission	26
Annex B-5	Executive summary of the European Union's appellee's submission	30
Annex B-6	Executive summary of Brazil's appellee's submission	36

- 14 -

ANNEX B-1

EXECUTIVE SUMMARY OF BRAZIL'S APPELLANT'S SUBMISSION*

I. Introduction

1. With the present appeal, Brazil requests that the Appellate Body review and reverse a number of issues of law and legal interpretation contained in Panel Report. Beyond the specific legal provisions under discussion, this appeal concerns more broadly the ability of WTO Members to provide subsidies to domestic producers.

2. Central to the resolution of this dispute is a proper understanding of the role that Article III:8(b) of the GATT 1994 plays in delineating the product-related disciplines of Article III from the subsidy-related disciplines of the SCM Agreement. The Appellate Body recently observed in US - Tax Incentives that "Article III:8(b) of the GATT 1994 exempts from the national treatment obligation in Article III 'the payment of subsidies exclusively to domestic producers'."¹ In diametric opposition to this conclusion, the Panel in the present dispute found that "subsidies that are provided exclusively to domestic producers pursuant to Article III:8(b) of the GATT 1994."²

3. In effect, the Panel interpreted Article III:8(b) of the GATT 1994 as serving only one purpose: to "make[] explicit" that the non-discrimination provisions of Article III do "not require subsidization of foreign producers in tandem with domestic producers".³ While Article III:8(b) does serve that rather axiomatic purpose, Article III:8(b) also makes clear that subsidies provided to domestic producers, including the effects that those subsidies have on the conditions of competition for domestic versus imported products, are not a basis for finding a violation of the national treatment obligations in Article III.

4. Moreover, and contrary to at least one implication of the Panel's analysis, it is immaterial for purposes of Article III:8(b) what type of subsidy a Member chooses to provide exclusively to domestic producers. Article III:8(b) of the GATT 1994 does not distinguish among different types of producer subsidies. The exemption or remission of indirect taxes—the type of subsidies at issue in the present dispute—is undoubtedly a "subsidy" within the meaning of Article 1 of the SCM Agreement. In addition, the Appellate Body has previously found that the ordinary meaning of the term "payment" encompasses revenue foregone, i.e. the exemption or remission of a tax or other revenue owed to the government. The "payment of subsidies exclusively to domestic producers" under Article III:8(b) therefore includes the exemption or remission of indirect taxes.

5. A proper application of Article III:8(b) of the GATT 1994 would have led the Panel to the conclusion that the measures at issue are producer subsidy measures, not measures that discriminate against imported products on the basis of their origin. The challenged measures provide for the exemption or remission of taxes based on whether a producer carries out certain production steps in Brazil. A producer that manufactures a covered product using 100% imported components will obtain the subsidy if it carries out the required production steps in Brazil, while a producer that manufactures the same product using 100% domestic components will *not* obtain the subsidy if it does not carry out the required production steps in Brazil.

6. The Panel committed the same basic errors of interpretation and application in its assessment of the complainants' claims under Article 3.1(b) of the SCM Agreement pertaining to the ICT programmes. Unable to identify any respect in which the ICT programmes condition the provision of a subsidy upon the use of domestic over imported goods, either in the text of the measures at issue or by necessary implication thereof, the Panel based its finding of prohibited import substitution entirely on the effect that the subsidies at issue might have on producers' sourcing decisions. The Appellate Body expressly rejected this line of reasoning in *US - Tax Incentives*, ruling that the fact

^{*} Total word count: 5542

¹ Appellate Body Report, US - Tax Incentives, para. 5.16.

² Panel Report, para. 7.87.

³ Ibid. para. 7.79.

that a domestic production subsidy might affect input sourcing decisions is not a sufficient basis to conclude that it is contingent upon the use of domestic over imported goods.

7. A proper delineation of the national treatment obligations of Article III of the GATT 1994 and the subsidy disciplines of the SCM Agreement is essential to preserving the rights that the covered agreements reserve for Members to subsidize domestic producers. The Panel Report in the present dispute, if adopted, would impermissibly constrain the right of Members to subsidize domestic production. It would, moreover, result in arbitrary distinctions among the *means* by which Members subsidize domestic production more commonly practiced by developed Members (such as outright grants) over methods of subsidization that are more readily available to developing Members (such as, in this case, the exemption or remission of indirect taxes based on whether certain production steps take place within the country).

8. These are not the only important systemic issues presented by this appeal. Another example relates to the Panel's finding that the ICT programmes and the PEC and RECAP programmes result in "revenue foregone" within the meaning of the SCM Agreement, based entirely on the flawed notion that "implicit interest income" is tax revenue that is "due" to a government when a government decides, for reasons of tax administration, either to collect a tax at a later point in time or to forgo the collection of a tax that the government would later need to refund. This interpretation, if accepted, would draw the covered agreements into the realm of tax administration in ways that would have far-reaching consequences for WTO Members.

II. The Panel Erred in Its Interpretation and Application of Article III:8(b) of the GATT 1994 to the ICT programmes and to INOVAR-AUTO and Its Relationship with the Other Provisions of Article III of the GATT 1994

9. The Panel erred in its interpretation and application of Article III:8(b) of the GATT 1994 to the ICT programmes and to INOVAR-AUTO by failing to determine, at the threshold, that these measures are producer subsidy measures subject to the disciplines of the SCM Agreement, rather than to the product-related disciplines of Article III of the GATT 1994.

10. Article III:8(b) makes clear that subsidies provided to domestic producers, *including* the effects that those subsidies might have on the conditions of competition for domestic versus imported *products*, are not a basis for finding a violation of the product-related disciplines of Article III. Production subsidies and their effects are to be analysed under the SCM Agreement, not under Article III of the GATT 1994.

11. In contrast, the concern with origin-based discrimination is reflected in Articles III: 2 and III: 4 of the GATT 1994. In both cases, the discipline applies to *products*, and guards against discriminatory treatment of imported products ("**products of the territory of any contracting party imported** ...") relative to products of national origin ("domestic products", "products of national origin"). Articles III: 2 and III: 4 do not apply, by force of Article III: 8(b), to differences in treatment that may result from payment of subsidies exclusively to domestic <u>producers</u>.

12. The Panel conflated the distinction between the product-related disciplines of Article III and the provisions of the covered agreements concerning subsidies provided to domestic producers, which should be assessed under the SCM Agreement. A harmonious interpretation of these different provisions requires any panel evaluating claims under Article III to determine, in the first instance, whether the measure at issue is a product-related measure subject to the disciplines of Article III or, conversely, whether that measure provides a subsidy to domestic producers under Article III:8(b), and is therefore subject to the disciplines of the SCM Agreement.

13. The Panel did not undertake this threshold inquiry. It simply assumed that, because the challenged programmes resulted in imported products not having access to the subsidy provided therein to domestic producers, the subsidy would not be exempted from the national treatment obligations of Article III and, therefore, would automatically constitute a violation of Articles III:2 and III:4 of the GATT 1994 and of Article 2.1 of TRIMS.

14. Accordingly, the Appellate Body should *reverse* the Panel's interpretative finding that measures providing subsidies exclusively to domestic producers "are not *per se* exempted from the

- 16 -

disciplines of Article III of the GATT 1994"⁴, and to *reverse* its ultimate conclusion that, because the ICT programmes and INOVAR-AUTO are not available to foreign *producers*, they are necessarily inconsistent with Articles III: 2 and III: 4 of the GATT 1994, and Article 2.1 of the TRIMs Agreement.⁵

III. The Panel Erred in Finding that the ICT programmes Are Inconsistent with Article III:2, First Sentence, of the GATT 1994

15. The Panel erred in finding that the ICT programmes result in imported finished and intermediate products being subject to taxation in excess of domestic like products, within the meaning of Article III: 2, first sentence, of the GATT 1994, in three distinct respects.

16. First, the Panel erred in failing to establish, at the threshold, that the ICT programmes constitute the "payment of subsidies to domestic producers" under Article III:8(b) of the GATT 1994 and therefore are not subject to the disciplines of Article III:2 of the GATT 1994. In ruling that Article III:8(b) authorized the provision of domestic production subsidies "so long as they do not have any component that introduces discrimination between imported and domestic products"⁶, the Panel effectively held that such subsidies can *never* take the form of reductions or exemptions of indirect taxes. However, the terms "subsidies" in Article III:8(b) encompasses all types of subsidies listed in Article 1.1 of the SCM Agreement, including foregoing of government revenue otherwise due under Article 1.1(a)(1)(ii) of the SCM Agreement, in the form of the exemption or remission of indirect taxes unquestionably constitute a "payment" of subsidies under Article III:8(b), and the negotiating history referred to the Panel is of little interpretative value considering that it precedes the definition of subsidies contained in the SCM Agreement.

17. In addition, the Panel failed to recognize that the list of measures in Article III:8(b) is merely illustrative, and does not exhaustively define the type of permissible domestic production subsidies. The Panel also erred in finding that Articles III:8(b) and Article III:2 have overlapping scopes of application. Contrary to the Panel's finding to this effect, the Appellate Body expressly held in *US* – *Tax Incentives* that Article III:8(b) *exempts* domestic production subsidies from the product disciplines of Article III.

18. Second, the Panel erred in finding that the complainants had established that the ICT programmes result in imported *finished* products being subject to taxes in excess of domestic like products. Setting aside the Panel's failure to determine the threshold applicability of Article III:8(b) to the ICT programmes, it was still incumbent upon the Panel to hold the complainants to their burden of proof of establishing that the ICT programmes result in *finished* imported products being subject to taxation in excess of domestic like products. The Panel simply assumed that, because foreign producers are ineligible for the subsidies, the ICT programmes *de jure* discriminated against imported products. However, because importers are also subject to the application of the debit/credit rule under Brazil's non-cumulative tax system, the payment of taxes on purchases of non-incentivized products generates a corresponding credit that may be offset against the importer's liability. Accordingly, the fact that foreign producers are ineligible for the tax incentives does not establish, without more, that imported finished products are taxed in excess of domestic like products.

19. Third and finally, the Panel erred in finding that the ICT programmes result in imported *intermediate* ICT products being taxed in excess of domestic like products by virtue of "cash flow" and "time value of money" effects allegedly generated on the purchase of incentivized products. At most, such unsubstantiated effects would benefit *producers*, and the complainants did not produce, and the Panel did not examine, any evidence tending to demonstrate that these effects resulted in higher taxation of imported intermediate *products*. Under Brazil's non-cumulative tax system, the relevant taxes are not paid upfront but are rather levied each month pursuant to the credits and debits recorded in the taxpayer's accounting books. As a general rule all credits are offset against debits in the same taxation period, and are neither tax nor product-specific. Thus the purchase of

⁴ Panel Report, para. 7.87.

⁵ Panel Report, paras. paragraphs 7.174, 7.318, 7.319, 7.365, 7.688, 7.772, 7.773, 7.806, 8.5(a), 8.5(b), 8.5(d), 8.6(a), 8.6(b) and 8.6(d).

⁶ Panel Report, para. 7.83.

- 17 -

non-incentivized products neither reduces the availability of cash flow by the purchaser, nor does the value of the tax credit diminishes over time.

20. For these reasons, the Appellate Body should *reverse* the Panel's finding that the ICT programmes result in taxation of imported like products in excess of domestic like products, inconsistently with Article III:2, first sentence, of the GATT 1994.⁷

IV. The Panel Erred in Finding that the ICT Programmes Accord to Imported Products Treatment Less Favourable than that Accorded to Like Domestic Products, Inconsistent with Article III:4 of the GATT 1994 and Article 2.1 of the TRIMs Agreement

21. Since the Panel erred in failing to determine the threshold applicability of Article III:8(b) of the GATT 1994 to the ICT programmes, it misconstrued the relationship of that provision with Article III:4 of the GATT 1994. Consistent with the analytical framework adopted by the Appellate Body in US - Tax Incentives and by previous panels, domestic production subsidies are *exempt* from the product disciplines of Article III:4 of the GATT 1994. Under the Panel's erroneous interpretation, it is unclear under what circumstances a WTO Member would *ever* be able to provide subsidies that do not distort the conditions of competition in the marketplace to the detriment of imported products. Quite the contrary, as the Panel itself recognized, the very purpose of providing a domestic production subsidy is to provide a competitive advantage exclusively to domestic producers.

22. For this reason, the Panel erred in finding that accreditation requirements under the ICT programmes are inconsistent with Article III: 4 because they result in foreign producers not being eligible to the subsidies. A corollary of Article III:8(b) is that only producers that produce domestically will benefit from the subsidies. The accreditation requirements merely define the producers that are eligible to the subsidies (i.e. those that undertake certain production steps in Brazil), and in and of itself do not reflect any regulatory discrimination between imported and domestic products. In this respect, the Panel impermissibly conflated tax discrimination under Article III:2 with regulatory discrimination under Article III:4, insofar as purported discrimination really stems from the reduction or exemption of indirect taxes for eligible producers. Moreover, because the accreditation requirements apply at the *producer* level pursuant to the authorization contained in Article III:8(b), by definition they cannot be implicated by the product disciplines of Article III:4

23. The Panel's finding that the ICT programmes are inconsistent with Article III:4 because imported intermediate products are purportedly subject to more onerous administrative burdens is also in error, and reflects a misapprehension of Brazil's non-cumulative tax system. Regardless of whether they purchase incentivized or non-incentivized products, all companies in Brazil are required to keep and operate a ledger in which all purchases of inputs and sales of final goods are recorded, and debits and credits are offset. The suspension, exemption, or reduction of indirect taxes under the ICT programmes do not alter in any way the administrative burden incurred by companies to assess their monthly tax liabilities, irrespective of their sourcing decisions.

24. Finally, the Panel erred in finding that the PPBs and other production-step requirements are tantamount to a requirement to use domestic goods within the meaning of Article III: 4 of the GATT 1994. The PPBs and other production step requirements are entirely silent as to the origin of the inputs used in the production process. A producer using 100% of imported components will be eligible for the subsidies, provided that it undertakes the relevant production steps in Brazil. The Panel's rationale that production step requirements are tantamount to a requirement to use domestic goods in respect of a producer's component or sub-assemblies sourcing decisions has been expressly addressed – and rejected – by the Appellate Body in *US – Tax Incentives*. In that dispute, the Appellate Body expressly held that domestic goods in the relevant market, thereby increasing the use of these goods downstream and adversely affecting imports, without necessarily requiring the use of domestic over imported goods as a condition for granting the subsidy."⁸ For this reason, the "siting criterion" imposed by the production step requirements under the ICT programmes, even when it involves the outsourcing of certain steps, does not constitute a requirement to use domestic

⁷ Panel Report, paras. 7.174, 8.5(a), and 8.16(a).

⁸ Appellate Body Report, US – Tax Incentives, para. 5.15.

- 18 -

products to the detriment of imported products in a manner inconsistent with Article III:4 of the GATT 1994.

25. Accordingly, the Appellate Body should *reverse* the Panel's finding that in respect of (i) accreditation requirements; (ii) purported lower administrative burdens on purchases of domestic intermediate products; and (iii) PPBs and other production step requirements, the ICT programmes accord to imported products treatment less favourable than that accorded to domestic like products, inconsistently with Article III: 4 of the GATT 1994 and Article 2.1 of the TRIMs Agreement.⁹

V. The Panel Erred in Finding that the ICT programmes Provide Subsidies that Are Contingent upon the Use of Domestic over Imported Goods under Article 3.1(b) of the SCM Agreement

26. The Panel erred in finding that the ICT programmes provide prohibited import substitution subsidies under Article 3.1(b) of the SCM Agreement in two respects.

27. First, the Panel erred in finding that the exemptions, suspensions and reductions of taxes provided under the ICT programmes on the sale of intermediate goods and on the purchase of raw materials, intermediate goods, packaging materials, inputs, capital goods and computational tools (collectively, "inputs") provide a subsidy under Article 1.1 of the SCM Agreement. More specifically, the Panel erred in finding that the ICT programmes provide a financial contribution in the form of government revenue that is otherwise due and is foregone and not collected under Article 1.1(a)(1)(ii) of the SCM Agreement. By focusing on the exceptional scenario in which the taxpayer is unable to fully offset the total amount of taxes paid in the same taxation period, the Panel impermissibly failed to compare the treatment of the group of taxpayers in the benchmark treatment with the treatment of taxpayers under the ICT programmes. Moreover, the "implicit interest income" that the Brazilian government allegedly foregoes under the ICT programmes as compared to the situation of that subset of taxpayers is not a "government revenue" that is "due" within the meaning of Article 1.1(a)(1)(ii) of the SCM Agreement.

28. Second, the Panel erred in finding that the ICT programmes are contingent upon the use of domestic over imported goods under Article 3.1(b) of the SCM Agreement because the PPBs and other production step requirements imply a contingency upon production in Brazil. As the Appellate Body held in *US* – *Tax Incentives*, "*something more than mere subsidization of domestic production* is required for finding an import substitution subsidy."¹⁰ The contingencies set out in the PPBs and similar production step requirements relate to the location of certain manufacturing operations in Brazil, and do not prevent the possibility of using imported inputs in the manufacturing process.

29. The Panel reasoned that because PPBs provide that outsourced components must be produced domestically, such condition imposes, in respect of the accredited producer of the incentivized finished product, a requirement to use domestic over imported goods. This line of argument was expressly addressed and rejected by the Appellate Body in US - Tax Incentives, where the Appellate Body categorically held that, in situations where both an input and the finished product are subsidized, the consequences of a requirement to produce an input domestically in respect of the producer's input-sourcing decisions, and the fact that the producer is likely to use a domestic product in its downstream production activities is insufficient, without more, to establish a condition requiring the use of domestic over imported goods under Article 3.1(b) of the SCM Agreement.¹¹

30. For these reasons, the Appellate Body should *reverse* the Panel's finding that the exemptions, reductions, and suspensions of taxes granted under the ICT programmes are subsidies contingent upon the use of domestic over imported goods within the meaning of Articles 3.1(b) and 3.2 of the SCM Agreement.¹²

⁹ Panel Report, paras. 7.318, 7.319, 7.365, 8.5(b), 8.5(d), 8.16(c) and 8.16(e).

¹⁰ Appellate Body Report, US – Tax Incentives, para. 5.16. (emphasis added)

¹¹ See Appellate Body Report, US - Tax Incentives, para. 5.49; see also ibid. para. 5.76.

¹² Panel Report, paras. 7.500, 8.5(e) and 8.16(f).

- 19 -

VI. The Panel Erred in Finding that Certain Aspects of the Accreditation Requirements under INOVAR-AUTO Related to Investments in R&D, Expenditures on Engineering, Basic Industrial Technology and to the Performance of Certain Manufacturing Steps in Brazil Accord Less Favourable Treatment to Imported Products than that Accorded to Like Domestic Products, Inconsistently with Article III:4 of the GATT 1994 and Article 2.1 of the TRIMs Agreement

31. Similar to the Panel's finding in respect of the ICT programmes, the Panel's finding that certain accreditation requirements under INOVAR-AUTO are inconsistent with Article III: 4 of the GATT 1994 is based on the incorrect assumption that Article III:8(b) does not exempt domestic production subsidies from the product disciplines of Article III. In so finding, the Panel failed to recognize that Articles III:8(b) and III:4 have distinct scopes of application. While the former concerns producer subsidy measures, the latter concern regulatory discrimination as between imported and domestic *products*. Consequently, any adverse effects that domestic production subsidies may have on the competitive opportunities for imported products should fall within the purview of Part III of the SCM Agreement.

32. The Panel's failure to determine, at the threshold, that INOVAR-AUTO constituted a domestic production subsidy under Article III:8(b) effectively converted permissible domestic production subsidies into *per se* violations of Article III:4 of the GATT 1994. Had the Panel properly interpreted and applied Articles III:8(b) and III:4 of the GATT 1944 to INOVAR-AUTO, it would have concluded that the conditioning of domestic production subsidies to the performance of certain production steps in Brazil is insufficient, without more, to establish regulatory discrimination under Article III:4 of the GATT 1994.

33. Moreover, the requirement to perform certain manufacturing steps in Brazil and to invest in R&D and/or engineering in Brazil in order to benefit from the tax incentives established in the INOVAR-AUTO programme are imposed on *producers*, and not on the products themselves. As such, these requirements cannot be deemed in violation of the product disciplines of Article III: 4 of the GATT 1994.

34. The Panel also erred in finding that different accreditation requirements for domestic manufacturers vis-à-vis the categories of "investors" or "importer/distributors" entail a more burdensome requirement in respect of imported products. The requirement that domestic producers undertake certain manufacturing or investment activities in Brazil is a natural corollary of Brazil's right under Article III:8(b) of the GATT to pay subsidies exclusively to domestic producers. The Panel impermissibly undertook a quantitative assessment of the INOVAR-AUTO requirements, rather than a qualitative assessment. Had the Panel adopted a qualitative approach instead, it would have recognized that the differences set forth in the INOVAR-AUTO programme with respect to "companies that manufacture vehicles in Brazil" and "companies that do not manufacture but market vehicles in Brazil" are not discriminatory because the differences are inherent to the level of contribution made by both manufacturing and distributing companies to the programme's objectives.

35. For these reasons, the Appellate Body should *reverse* the Panel's finding that the accreditation requirements under INOVAR-AUTO to: (a) perform a minimum number of manufacturing steps in Brazil; and (b) invest in R&D in Brazil and make expenditures on engineering, basic industrial technology and capacity-building of suppliers in Brazil accord less favourable treatment to imported products than that accorded to like domestic products, inconsistently with Article III: 4 of the GATT 1994 and Article 2.1 of the TRIMs Agreement.¹³

36. Moreover, to the extent that the Panel concluded that domestic production requirements alone are sufficient to establish that INOVAR-AUTO is "contingent upon the use of domestic over imported goods" within the meaning of Article 3.1(b) of the SCM Agreement, Brazil further requests that the Appellate Body *reverse* the Panel's finding that INOVAR-AUTO is a prohibited import substitution subsidies under Article 3.1(b) and 3.2 of the SCM Agreement.¹⁴

¹³ Panel Report, paras. 7.773, 7.806, 8.6(b), 8.6(d), 8.17(c) and 8.17(e).

¹⁴ Panel Report, paras. 7.751, 7.823, 7.847, 8.6(e), and 8.17(f).

- 20 -

VII. The Panel Erred in Finding that the Tax Suspensions under the PEC and RECAP Programmes Are Subsidies Contingent Upon Export Performance Under Article 3.1(a) of the SCM Agreement

37. The Panel erred in finding that PEC and RECAP provide subsidies in the form of government revenue otherwise due that is foregone or not collected under Article 1.1(a)(1)(ii) of the SCM Agreement, in three different respects.

38. First, the Panel erred in failing to use the treatment of structural credit accumulators as the benchmark treatment for comparison with the tax treatment under the PEC and RECAP programmes. The Panel erroneously sought to ascertain whether a "general rule" of taxation existed under Brazilian law pursuant to which *all* suspensions and exemptions of taxes on inputs and capital goods is directly linked to the problem of structural credit accumulation, rather than ascertain the "comparably situated taxpayers" that make up the relevant normative benchmark for comparison. Had the Panel applied a proper analytical framework, it would have concluded that all structural credit accumulators are comparably situated, as it is evident from the relevant statute. The fact that sectors whose products were formerly subject to high taxation may also be eligible for tax suspensions confirms rather than disproves the existence of this normative benchmark, insofar as companies in those sectors became structural tax credit accumulators by virtue of exemptions or suspensions granted in other programmes.

39. Second, the Panel erred in failing to compare the tax treatment applicable to the *group* of taxpayers subject to the benchmark treatment with the tax treatment of the *group* of taxpayers under PEC and RECAP. The Panel arbitrarily distinguished between taxpayers subject to the benchmark treatment, disregarding the overwhelming majority of taxpayers that are able to fully offset their tax credits in the same taxation period, focusing instead on the small subset of taxpayers who are unable to do so. In doing so, the Panel found the foregoing of revenue where there is none.

40. Third, the Panel erred in finding that alleged "implicit interest income" that the Brazilian government purportedly "foregoes" under PEC and RECAP constitutes "government revenue" that is otherwise "due" and foregone or not collected under Article 1.1(a)(1)(ii) of the SCM Agreement. The terms "government revenue" that is "due" refer to government income that is owed or payable to the government as an obligation or debt by the taxpayer (such as taxes or charges), and does not encompass any "implicit interest" that the Brazilian Government supposedly "foregoes" as a result of tax administration measures which do not alter the amount of taxes due.

41. Should the Appellate Body consider nonetheless that PEC and RECAP provide a subsidy within the meaning of Article 1.1 of the SCM Agreement, Brazil then claims that the Panel acted inconsistently with Article 11 of the DSU in its assessment of the evidence on export contingency, and therefore erred in finding that the PEC and RECAP programmes are contingent upon export performance within the meaning of Article 3.1(a) of the SCM Agreement.

42. On the basis of the foregoing, the Appellate Body should *reverse* the Panel's finding that PEC and RECAP are subsidies contingent upon export performance under Articles 3.1(a) and 3.2 of the SCM Agreement.¹⁵

VIII. The Panel Erroneously Found that the Tax Reduction Accorded to Imports from Argentina, Mexico and Uruguay under INOVAR-AUTO Is Not Justified under Paragraphs 2(b) or 2(c) of the Enabling Clause

43. The Panel erred in finding that the tax reduction accorded to imports from MERCOSUR countries under INOVAR-AUTO are inconsistent with Article I:1 of the GATT 1994 and not justified under paragraphs 2(b) or 2(c) of the Enabling Clause.

44. The Panel preliminarily erred in finding that the complainants' claim under Article I:1 of the GATT 1994 was properly within its terms of reference. The measures had unquestionably been notified under paragraph 4(a) of the Enabling Clause, and contrary to what the Panel found, any

¹⁵ Panel Report, paras. 7.1211, 7.1212, 7.1238, 8.7 and 8.18.

substantive disagreement as to the proper characterization of the measures did not alleviate the complainants' burden of invoking the Enabling Clause in their panel requests.

45. The Panel also erred in failing to examine whether the notification of the Treaty of Montevideo and the ECAs under paragraph 2(c) could substantively serve as a notification of the specific differential and more favourable treatment sought to be justified under paragraph 2(b) of the Enabling Clause. The Panel adopted an impermissibly restrictive interpretation of the Enabling Clause, failing to recognize that paragraph 4(a) does not impose on Members any requirement to precisely identify the basis pursuant to which it notifies its measures. This suggests that the drafters of the Enabling Clause provided some flexibility in such notification provisions.

46. The Panel also erred in limiting the scope of paragraph 2(b) of the Enabling Clause to the S&D provisions in the covered agreements other than the GATT itself. The GATT 1994 is unquestionably an "instrument multilaterally negotiated under the auspices of the GATT" within the meaning of paragraph 2(b). The Panel's restrictive interpretation to the contrary impermissibly renders the provision effectively *inutile*, insofar as the S&D provisions provided for in the agreements in Annex 1A of the Marrakesh Agreement are mostly hortatory and impose no substantive obligations.

47. Finally, the Panel impermissibly conflated the notification requirements with the substantive requirements imposed under paragraph 2(c) of the Enabling Clause. The challenged treatment had been duly notified under paragraph 2(c). The Treaty of Montevideo, the ECAs, and the LAIA clearly comprise internal taxation measures, and as the dissenting Panelist correctly observed, refer to both "regional integration" and to tariff preferences in the automotive sector. Internal taxation measures are unquestionably "non-tariff measures" within the meaning of paragraph 2(c) of the Enabling Clause, and the Panel erred in failing to undertake a substantive analysis of whether the challenged treatment satisfied the requirements of that provision.

48. Accordingly, the Appellate Body should *reverse* the Panel's conclusion that the complainants did not have an obligation to invoke the Enabling Clause in their panel requests, and *reverse* the Panel's consequential finding that the complainants' claims under Article I: 1 of the GATT 1994 were within the Panel's terms of reference.¹⁶

49. The Appellate Body should also *reverse* the Panel's finding that the differential tax treatment at issue was not properly notified under paragraph 4(a) of the Enabling Clause; *reverse* the Panel's finding that the tax treatment at issue is not justified under paragraphs 2(b) or 2(c) of the Enabling Clause; *complete the legal analysis* and find that the differential tax treatment is justified under paragraphs 2(b) or 2(c) and complete with paragraph 3 of the Enabling Clause.¹⁷

IX. The Panel Acted Inconsistently with Articles 11 and 12.7 of the DSU in Recommending that Brazil Withdraw the Alleged Prohibited Subsidies in 90 Days

50. Should the Appellate Body uphold the Panel's finding that the measures at issue are prohibited subsidies under Article 3.1(a) and 3.1(b) of the SCM Agreement, Brazil then claims, in the alternative, that the Panel acted inconsistently with its duties under Articles 11 and 12.7 of the DSU in recommending pursuant to Article 4.7 of the SCM Agreement that Brazil withdraws these measures in 90 days.

51. The fact that the Appellate Body has said that "the expedited remedy provided under Article 4.7 of the SCM Agreement is an 'important consideration'"¹⁸ does not constitute a "reasoned and adequate explanation" for the Panel's 90-day recommendation, because it does not explain why the Panel believed that a 90-day timeframe was appropriate. In order for the Panel to provide a "reasoned and adequate explanation" or "basic rationale" for its conclusion, the Panel needed to explain why 90 days was appropriate in light of the nature of the measures and the procedures that may be required to implement [the panel's] recommendation. The Panel's failure to provide such an explanation cannot be reconciled with the Panel's obligations under Articles 11 and 12.7 of the DSU.

¹⁶ Panel Report, paras. 7.1083, 7.1120, 8.6(h), and 8.17(i).

¹⁷ Panel Report, paras. 7.1079, 7.1081, 7.1083, 7.1096, 7.115, 7.119, 8.6(i) and 8.17(j).

¹⁸ Panel Report, para. 6.17 (quoting Appellate Body Report, *Canada – Renewable Energy / Canada – Feed-in Tariff Program*, para. 5.7) (referring to Appellate Body Report, *EC – Export Subsidies on Sugar*, paras. 332-335).

52. Brazil therefore requests that the Appellate Body *reverse* the Panel's recommendation that Brazil withdraw the measures within 90 days¹⁹, *complete the legal analysis*, and recommend, in light of the nature of the measures at issue and the length of the legislative process that will be required, withdrawal within 18 months.

¹⁹ Panel Report, paras. 8.11 and 8.22.

- 23 -

ANNEX B-2

EXECUTIVE SUMMARY OF THE EUROPEAN UNION'S OTHER APPELLANT'S SUBMISSION¹

1. The European Union brings this other appeal in the interest of clarifying the scope of Brazil's implementing obligations in this dispute. The Panel's findings lack clarity as to Brazil's obligations under Articles III:4 of GATT 1994, Article 2.1 of TRIMS Agreement and Article 3.1(b) of SCM Agreement with regard to the in-house scenario.

2. First, the European Union submits that the Panel failed to make an objective assessment of the matter before it in accordance with Article 11 of DSU by exercising false judicial economy in relation to the European Union's request to make findings also on the in-house scenario in the context of the ICT and INOVAR-AUTO programmes. Thus, the European Union requests the Appellate Body *to reverse* the Panel's findings, *to complete the legal analysis* with regard to the in-house scenario and *to find* that the production-step requirements contained in the ICT and INOVAR-AUTO programmes are inconsistent with Articles III: 4 of GATT 1994, Article 2.1 of TRIMS Agreement and Article 3.1(b) of SCM Agreement also under the in-house scenario.

3. Second, in the alternative, were the Appellate Body to consider that the Panel correctly exercised judicial economy when failing to make specific findings on the in-house scenario, the European Union requests the Appellate Body *to review*, pursuant to Article 17.6 of DSU, the legal interpretations developed by the Panel and modify, pursuant to Article 17.13 of DSU, the findings contained in paragraphs 7.298-7.314 and 7.746-7.751 of its Report, so as *to make it clearer* that the Panel indeed did not need to rule twice on the production step requirements (in the in-house and in the outsourcing scenarios) because the Panel had already found that those steps are *per se* inconsistent with the covered agreements, since they make an advantage/subsidy contingent upon the use of domestic over imported goods, regardless of whether those domestic goods are manufactured by the accredited company or a third party.

4. Finally, the European Union makes a subordinate claim of error, subject to the Appellate Body rejecting both the first claim of error formulated by the European Union (that the Panel exercised false judicial economy) and the alternative claim of appeal. In that case, the European Union requests the Appellate Body to find that the Panel, by failing to consider the European Union's claims under Article III:4 of GATT 1994, Article 2.1 of TRIMS Agreement and Article 3.1(b) of SCM Agreement in light of all the relevant facts of the case (which included both the in-house and outsourcing scenarios), erred in the application of those provisions. Accordingly, the European Union requests the Appellate Body *to reverse* the Panel's findings, *to complete the legal analysis* with regard to the in-house scenario and *to find* that the production-step requirements contained in the ICT and INOVAR-AUTO programmes are inconsistent with Articles III:4 of GATT 1994, Article 2.1 of TRIMS Agreement and Article 3.1(b) of SCM Agreement and INOVAR-AUTO programmes are inconsistent with Articles III:4 of GATT 1994, Article 2.1 of TRIMS Agreement and Article 3.1(b) of SCM Agreement also under the in-house scenario.

¹ Total number of words (including footnotes but excluding executive summary) = 12 851; total number of words of the executive summary = 511.

- 24 -

ANNEX B-3

EXECUTIVE SUMMARY OF JAPAN'S OTHER APPELLANT'S SUBMISSION¹

1. Japan's Other Appellant Submission is focused on one single issue only where, in Japan's view, the Panel could and should have done more. As the Panel appears to have applied judicial economy with respect to the INOVAR-AUTO programme and ICT programmes, Japan submits that the Panel erred by (1) distinguishing the scenario in which a company undertakes minimum production processes by itself ("in-house"), from that in which the company delegates these activities to a third party ("outsourcing") and (2) refraining from addressing the former, which is inconsistent with Article III: 4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement.

2. First, Panel explicitly finds that the production-step requirements and the accreditation requirements formulated under the relevant legal instruments implementing the challenged programmes do not themselves distinguish between an "in-house" and an "outsourcing" scenario. The Panel draws this distinction merely as an analytical one to facilitate and organize its own analysis.

3. Second, as the Panel itself acknowledged, the complainants challenged the programmes' production-step requirements as a whole WTO-inconsistent, irrespective of which scenario applies. In Japan's view, the Panel's failure to make findings with respect to each accreditation and production step requirements as a whole would lead to a partial resolution of the matter.

4. Third, because the Panel limited its finding to the "outsourcing" scenario, a positive solution to the dispute is not secured. Due to the lack of necessary findings in the Panel report with respect to the "in-house" scenario, the Panel leaves open the possibility of disagreement between the Parties within the meaning of Article 21.5 of the DSU. This will undermine the "effective resolution of disputes to the benefit of all Members" under Article 21.1 of the DSU. Such failure constitutes "false judicial economy" and constitutes a breach of the panel's obligation under Article 11 of the DSU.

5. Therefore, Japan respectfully asks the Appellate Body to reverse the Panel's decision to limit its finding to the "outsourcing" scenario and complete the legal analysis by finding that the accreditation and production step requirements "as a whole," encompassing both the "outsourcing" and "in-house" scenarios, are inconsistent with Article III:4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement.

6. Should the Appellate Body consider the Panel's application of judicial economy to be unwarranted, Japan requests that the Appellate Body complete the legal analysis to find the production step requirements under the challenged programmes, as a whole, to be inconsistent with Article III:4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement. The Panel's findings were sufficient for the Appellate Body to complete the analysis and to make such findings. Moreover, much of the evidence is in fact identical or substantially the same for the "outsourcing" and "inhouse" scenarios. Thus, the Appellate Body can complete the analysis as needed.

7. Japan underscores that the each production step requirement under the INOVAR-AUTO and the ICT programmes extends accreditation to the companies that carry out the production steps in Brazil by themselves, as well as to the companies that outsource to third parties within Brazil. This necessarily implies that the programmes offer subsidies based on the use of domestic products, rather than carrying out a particular production steps. Accordingly, they constitute "a condition requiring" manufacturers to use Brazilian products as a precondition to receiving tax benefits, which are inconsistent with Article 3.1(b) of the SCM Agreement.

8. Alternatively, should the Appellate Body find that the Panel did not exercise judicial economy with respect to the "in-house" scenario, Japan respectfully requests the Appellate Body to find that the Panel's finding of the WTO-inconsistency of the accreditation and the production-step requirements under INOVAR-AUTO and the ICT programme are interpreted to cover the challenged

¹ Other Appellant Submission 7,434 words, Executive Summary 670 words.

- 25 -

two measures as a whole including both the "outsourcing" and "in-house" scenarios and the Panel erred by failing to provide "coherent reasoning" as required under Article 11 of the DSU.

- 26 -

ANNEX B-4

EXECUTIVE SUMMARY OF JAPAN'S APPELLEE'S SUBMISSION¹

I. INTRODUCTION

1. Japan notes that the Panel's findings did *not* put into question Brazil's or any other country's ability to adopt policies to promote development, support domestic research and development (R&D), and/or develop human capital. The Panel found that Brazil's measures fall into the much more narrowly defined categories of measures that are covered by the National Treatment obligations of Article III: 2 and III:4 of the GATT 1994, the prohibitions on export subsidies and import-substitution subsidies in Articles 3.1(a), 3.1(b) and 3.2 of the SCM Agreement, and Article 2.1 of the TRIMS Agreement.

II. THE PANEL DID NOT ERR IN ITS INTERPRETATION AND APPLICATION OF ARTICLE III:8(b) OF THE GATT 1994 TO THE ICT AND INOVAR-AUTO PROGRAMMES

2. Contrary to Brazil's assertion, as the Panel found, the mere fact that a subsidy was provided exclusively to domestic producers is not sufficient for it to be exempt from Articles III:2 and III:4 of the GATT 1994. Brazil's argument is not only legally incorrect but very far-reaching in its potential consequences, which would lead to a wide loophole for countries to circumvent the disciplines under Articles III:2 and III:4 of the GATT 1994.

3. By its ordinary meaning, Article III:8(b) does not constitute a blanket exemption for subsidies paid exclusively to domestic producers. Instead, measures that take the forms of subsidies paid exclusively to domestic producers may still be subject to the disciplines under Article III of the GATT 1994 if they result in discrimination between products. Brazil also appears to take issue with the Panel's interpretation of the term "payment" in Article III:8(b). Its argument, however, is entirely circular.

4. In addition, Brazil claims that while the Panel did not rely on it, Japan and the EU, as co-complainants misinterpreted the Appellate Body's findings in *Canada – Periodicals*. Contrary to Brazil's argument, however, the Appellate Body made clear that a reduction of tax rates is not covered by Article III:8(b). Because INOVAR-AUTO and the ICT programmes both involve a reduction of tax rates, Article III:8(b) does not apply.

III. THE PANEL PROPERLY FOUND THAT THE ICT PROGRAMMES ARE INCONSISTENT WITH ARTICLE III:2, FIRST SENTENCE, OF THE GATT 1994

5. As a threshold argument, Brazil alleges that the tax treatment at issue was not within the scope of Article III:2. Again, Brazil's interpretation of Article III:8(b) of the GATT 1994, and its relationship to the substantive disciplines of Article III, is incorrect.

6. Further, Brazil alleges that the Panel failed to "carefully scrutinize the design, structure and operation of the ICT programmes in applying Article III:2 of the GATT 1994 to the facts of this dispute." The Panel, however, conducted a thorough analysis of the ICT programmes, and correctly found that the tax burden on imported finished products is in excess of that on domestic like products.

7. Additionally, Brazil argument that the Panel erred in finding that the "cash flow" and the "timevalue of money" effects under the ICT programmes result in a higher tax burden on imported intermediate products is flawed. Brazil's domestic manufacturers of intermediate products must pay certain taxes on inputs, which are subsequently offset by revenues from selling downstream products. Thus, during the lag between purchase of inputs and sale of outputs, funds related to the payment of taxes are temporarily unavailable. However, the ICT programmes mitigate the negative financial consequences of this lag, by reducing or eliminating the required taxes otherwise due when

¹ Appellee Submission 27,908 words, Executive Summary 2,202 words.

- 27 -

domestic manufacturers of intermediate goods purchase inputs covered by the ICT programmes, and thus are not consistent with Article III:2.

IV. THE PANEL PROPERLY FOUND THAT THE ICT PROGRAMMES ARE INCONSISTENT WITH ARTICLE III:4 OF THE GATT 1994 AND ARTICLE 2.1 OF THE TRIMS AGREEMENT

8. Brazil's argument once again that Article III:8(b) shields a measure from being found to be WTO-inconsistent is flawed. Japan also disagrees with Brazil's claim that the Panel conflated Articles III:2 and IIII:4 of the GATT 1994. A violation of Article III:4 of the GATT 1994 can be found when an underlying measure is tax.

9. Brazil also disputes the Panel's finding that the requirements to perform a series of production steps under the ICT programmes require the use of domestic products. However, the Panel correctly found such requirements to be inconsistent with Article III: 4 of the GATT 1994 because they "accord to imported products treatment less favourable than that accorded to like domestic products."

V. THE PANEL PROPERLY FOUND THAT CERTAIN ASPECTS OF THE ICT PROGRAMMES ARE INCONSISTENT WITH ARTICLE 3.1(b) OF THE SCM AGREEMENT

10. The Panel applied the correct benchmark in finding that the tax exemptions under the ICT programmes constitute financial contributions within the meaning of Article 1.1(a)(1)(ii) of the SCM Agreement. The Panel determined that, *at least in some factual situations*, accredited companies benefiting from the ICT Programmes are better off than those within the benchmark, and the Brazilian Government, by contrast, is foregoing revenue that would be otherwise due.

11. Brazil also claims that the Panel committed legal error when it found that the "cash availability" and "implicit interest" constituted revenue "due" that could have been forgone within the meaning of Article 1.1(a)(1)(ii) of the SCM Agreement. At the outset, Japan notes that Brazil appears to be asking for a *de novo* review of the facts or to accept certain facts that are simply not supported by the Panel's findings. Further, there is no basis for Brazil's argument that the interest foregone by the Brazilian government is not in fact a form of current or future revenue. Brazil's claim also appears to rest on the premise that it is only in "exceptional" cases but the Panel record, makes clear that such a situation is "rather frequent".

12. Brazil argues that the Panel erred in finding that the production-step requirements under the ICT programmes require the use of domestic over imported goods. However, the production-step requirements under each of the challenged ICT programmes and ultimately concluded that the structure of accreditation requirements demonstrated that the challenged measures "give rise to a contingency on the use of domestic over imported goods in the sense of Article 3.1(b) of the SCM Agreement."

VI. THE PANEL PROPERLY FOUND THAT THE INOVAR-AUTO PROGRAMME ACCORDS TREATMENT LESS FAVOURABLE TO IMPORTS THAN TO LIKE DOMESTIC PRODUCTS, INCONSISTENT WITH ARTICLE III:4 OF THE GATT 1994 AND ARTICLE 2.1 OF THE TRIMS AGREEMENT

13. Brazil's assertions against the Panel's findings on the measure's inconsistency with Article III: 4 of the GATT 1994 are premised on the flawed claim that Article III: 8 shields a measure from being found to be WTO-inconsistent, so long as the challenged measure is in the form of a subsidy provided exclusively to domestic producers.

14. Brazil challenges what it calls the Panel's "quantitative" analysis and appears to request the Appellate Body to engage in a weighing and balancing of the facts before the Panel to determine the burden that INOVAR-AUTO's requirements imposed on foreign manufactures. This appears to be a request for *de novo* review of the facts and thus the Appellate Body should reject Brazil's argument.

15. The Panel consider[ed] that foreign companies seeking accreditation under the INOVAR-AUTO programme must go through a more burdensome accreditation process than domestic manufacturers." Additionally, the level of taxation applied to imported motor vehicles, sold by foreign companies that are not accredited, exceed that applied to domestic vehicles sold by accredited

- 28 -

domestic manufacturers. Japan underscores that the conditions for accreditation accord less favourable treatment to foreign *products* imported into Brazil than that accorded to like *products* of Brazilian origin.

VII. THE PANEL PROPERLY FOUND THAT THE TAX SUSPENSIONS GRANTED UNDER THE PEC AND RECAP PROGRAMME ARE CONTINGENT UPON EXPORT PERFORMANCE,

16. According to Brazil, the Panel incorrectly applied Article 1.1(a)(1)(ii) of the SCM Agreement because it failed to identify the treatment of structural accumulators as the proper benchmark treatment with respect to PEC and RECAP programmes.

17. In conducting its benchmark analysis, contrary to Brazil's assertion, the Panel did not attempt to identify a "general rule" of taxation under Brazilian law pursuant to which *all* suspensions and exemptions of taxes on inputs and capital goods *relate to the problem of structural credit accumulation*, but rather, responded to Brazil's assertion about the "rule". Based on its analysis, the Panel correctly found that there was no basis for defining the benchmark as narrowly as Brazil proposed.

18. As an alternative argument, Brazil asserts that even if the Appellate Body finds that nonaccredited companies are the correct benchmark, the Panel incorrectly "opted to compare the treatment under PEC and RECAP with what it erroneously termed the "rather frequent" scenario in which non-accredited companies are unable to fully offset their tax credits in the same taxation period, and have to formally request the reimbursement of taxes paid." Japan has already explained why Brazil's argument is flawed in the analogous in the context of the ICT programmes is flawed. The same reasoning applies here.

19. Further, Brazil's claim, asserting that the Panel erred in the interpretation and application of Article 1.1(a)(1)(ii) of the SCM Agreement when finding that the possible cash availability and implicit interest income that the Brazilian Government could earn (had the tax been paid in advance and not offset) constitute a kind of revenue otherwise 'due' by the taxpayer, is flawed.

20. Brazil also argues that the Panel failed to properly assess a table it submitted to show that the PEC and RECAP programmes were indeed providing subsidies to credit-accumulating companies and not in contingency with exportation under Article 3.1(a) of the SCM Agreement, and thus acted inconsistently with Article 11 of the DSU. However, it is very clear that the Panel performed an objective assessment consistent with that provision. Brazil's argument should be rejected because it is effectively seeking a de novo review of the facts, but has failed to show any failure by the Panel to meet the standard reflected in Article 11 of the DSU.

VIII. THE PANEL PROPERLY FOUND THAT THE TAX REDUCTION ACCORDED TO IMPORTS FROM ARGENTINA, MEXICO AND URUGUAY UNDER INOVAR-AUTO IS NOT JUSTIFIED UNDER PARAGRAPHS 2(b) OR 2(c) OF THE ENABLING CLAUSE

21. Paragraph 4 of the Enabling Clause in particular, explicitly requires the Member taking an action under the Enabling Clause to notify, *ex ante*, other Members. A complaining party cannot be obliged to invoke the Enabling Clause in its panel request, unless it is appropriately notified. If the Appellate Body were to follow Brazil's overly broad approach, this would effectively mean that Members are required to simply guess whether some past action by another Member was or was not meant as a measure taken pursuant to the Enabling Clause.

22. Brazil also takes issue with the Panel's substantive rejection of its ex post facto attempts at an Enabling Clause-based justification. However, Brazil does not even assert, much less establish, that INOVAR-AUTO provides for an exception with respect to such provisions. In particular, with respect to paragraph 2(c), Japan notes that Brazil's claim appears to be that the Panel did not properly analyse and weigh the facts. In addition, Japan also notes that the INOVAR-AUTO programme did not have a close and genuine link to the Treaty of Montevideo or ECAs, as the Panel correctly concluded.

23. Finally, Japan also notes, that in any event, the preferential treatment for other MERCOSUR countries and Mexico is contrary to paragraph 3 of the Enabling Clause. Pursuant to paragraph 3(a), measures taken under the Enabling Clause "shall be designed to facilitate and promote the trade of

- 29 -

developing countries and not to raise barriers or to create undue difficulties for the trade of any other contracting parties".

IX. THE PANEL'S RECOMMENDATION THAT BRAZIL WITHDRAW THE ALLEGED PROHIBITED SUBSIDIES IN 90 DAYS IS CONSISTENT WITH ARTICLES 11 AND 12.7 OF THE DSU

24. With its final ground for appeal, Brazil submits that the Panel erred in recommending 90 days as the appropriate time period for Brazil to withdraw the prohibited subsidies. At the outset, Japan notes that panels do not need to address in their reports every argument or evidence raised by the parties. Similarly, panels must set forth explanations and reasons sufficient to disclose the essential, or fundamental, justification for those findings and recommendations, in light of Article 12.7 of the DSU. This does not, however, necessarily imply that Article 12.7 requires panels to expound at length on the reasons for their findings and recommendations.

25. In any event, Japan observes that, the Panel explicitly explained that it had taken into account the procedures that may be required to implement the recommendation on the one hand, and the requirement that Brazil withdraw its subsidies "without delay" on the other hand, when recommending that Brazil shall withdraw the subsidies identified in paragraphs 8.16(f), 8.17(f), and 8.18 within 90 days. Brazil's appeal, if anything, appears to be simply an effort to re-litigate the question.

X. CONCLUSION

26. As discussed and for the reasons set out above, Brazil's claims of error are all without merit and Japan respectfully requests the Appellate Body to reject them.

- 30 -

ANNEX B-5

EXECUTIVE SUMMARY OF THE EUROPEAN UNION'S APPELLEE'S SUBMISSION¹

1. The European Union considers that overall the Panel made balanced findings in relation to the challenged programmes as a whole as well as several specific elements thereof, on issues which, to certain extent, are a *déjà vu* for the GATT/WTO system: the use of discriminatory taxation and local content requirements as a tool to boosting industrial policy development, coupled with tax reliefs for companies that achieve an export threshold. Far from disputing each and every legal finding made by the Panel, Brazil appears to concede some of these findings and conclusions by not raising an appeal against them. In contrast, Brazil's appeal is premised on seeking to spot weakness in how the Panel interpreted and applied the law to the relevant facts, whilst in reality, the Panel's assessment is reasoned and reasonable. At times, Brazil also appears to question how the Panel weighted all the evidence on specific matters without raising a claim under Article 11 of the DSU. The European Union considers that, whilst Brazil may disagree with the Panel's appreciation of the facts and ultimate conclusions, Brazil has failed to show the existence of any reversible error.

1.1 Article III:8(b) as a threshold issue

2. The Panel properly interpreted and applied Article III:8(b) of the GATT 1994 as not excluding *per se* from the disciplines of Article III subsidies granted to domestic producers. Article III:8(b) of the GATT 1994 exempts payments to domestic *producers* from the national treatment obligation, but only to the extent that e.g. those payments do not discriminate between domestic and imported *goods*. Brazil's formalistic attempt to isolate the disciplines under the SCM Agreement from the national treatment rules under Article III of the GATT 1994 should be rejected as it goes against the principle that the covered agreements should be interpreted in a coherent and consistent manner, giving meaning to all applicable provisions harmoniously.

1.2 ICT programmes

1.2.1 Article III:2 of the GATT 1994

With the first claim of appeal under Article III: 2 of the GATT Brazil in substance repeats its 3 line of argument that the Panel erred when finding that domestic production subsidies do not fall per se outside the purview of the disciplines of Article III (including paragraph 2 of that Article) and it adds a few more arguments. The European Union notes first that the expression "domestic production subsidies" is not contained in the WTO treaties, which shows that Brazil's claim in predicated on an undefined concept, perhaps in the hope that the Appellate Body will make the case for Brazil. Second, the European Union notes that the Panel found that Article III:8(b) does not carve out from the disciplines of Article III: 2 subsidies provided exclusively to domestic producers per se, since aspects of a subsidy resulting in product discrimination may fall within that provision. Hence the Panel did not need to discuss whether or not Article III:8(b) of the GATT 1994 applies to the provision of subsidies in the form of reductions in indirect taxes. Therefore, the Appellate Body does not need to examine Brazil's allegation that term "subsidies" in Article III:8(b) of the GATT 1994 encompasses all types of subsidies included in Article 1.1 of the SCM Agreement, as that legal issue is not covered by the Panel's findings. In any event, the European Union submits that Brazil's argument is also incorrect as a matter of law.

4. Brazil's second claim of appeal under Article III: 2 of the GATT relates to the findings of the Panel concerning the taxation of finished incentivised products as opposed to like imported finished products. However, Brazil's claim is predicated on a number of false premises, which are contradicted by factual findings of the Panel and what Brazil conceded before the Panel, which Brazil does not challenge. The Panel analysis of the parties arguments concerning incentivised finished products under Article III: 2, first sentence, of the GATT 1994 reflects the rules on the allocation of the burden of proof in WTO law. Hence, Brazil's claim boils down to criticizing the Panel for not having made the case for Brazil. In any case, Brazil's claim is wrong on substance: a lower tax rate imposed on

¹ Total number of words of the executive summary = 3744.

incentivised finished products (i.e. domestic) results in a lower fiscal burden for those products compared to like imported finished products.

5. Brazil's third claim of appeal under Article III: 2 of the GATT is in essence that the Panel erred in finding that the "cash-flow" and the "time-value of money" effects under the ICT programmes result in a higher tax burden on imported intermediate products compared to like incentivised intermediate domestic products. Brazil's claim of appeal is however based on a wrong reading of the Panel Report. First, nowhere the Panel agreed with Brazil that the nominal amount of taxes is decisive, but it clearly indicated that it was concerned with determining the effective tax burden and for that purpose it took into account the functioning of the credit-debit system governing Brazil's direct taxes at issue. The Panel found that the application of the rule of credits and debits for purchases of imported (and, therefore, non-incentivized) intermediate ICT products involves the payment of a tax that is not faced by companies purchasing incentivized intermediate domestic ICT products. This payment of the tax upfront has the effect of limiting the availability of cash flow by companies purchasing imported intermediate ICT products and results in a higher effective tax burden on these products. Moreover, imported intermediate ICT products are subject to a higher tax burden than like incentivized domestic intermediate ICT products, due to depreciation in the value of the tax credits over time. The Panel's analysis reflects elementary concepts of economics or rather simple common sense considerations and is confirmed by WTO case law. Moreover, Brazil does not assert that the Panel failed to make an objective assessment of the facts when describing the functioning of the credit-debit system governing Brazil's direct taxes at issue.

1.2.2 Article III:4 of the GATT 1994

6. The first claim of appeal by Brazil under Article III: 4 of the GATT relate again to Brazil's reading of Article III: 8(b) as exempting domestic production subsidies *per se* from the disciplines of Article III (including obviously Article III: 4). The European Union has already explained that this reading Article III: 8(b) is unwarranted and will not repeat the same reasoning.

7. With its second claim of appeal under Article III: 4 of the GATT, Brazil argues in substance that the Panel conflated tax discrimination under Article III: 2 with regulatory discrimination under Article III: 4. However, a careful reading of the Panel Report is sufficient to reject Brazil's claim. Indeed, the Panel clearly recalled the Appellate Body jurisprudence which admits that the same measure, or certain aspects of the same measures, can fall both within Article III: 4 and Article III: 2. The Panel hence analysed the complainants claims in detail and checked weather the complainants demonstrated that the challenged ICT programmes met all the conditions for finding a violation of Article III: 4, which Brazil does not seriously contest. The fact that one instance of less favourable treatment accorded to imported products results from the different tax burden imposed on those products vis-à-vis like domestic products is simply a consequence of the characteristics of the regulatory measure examined by the Panel. At the same time, it is entirely possible that the same measure or certain aspects of the same measure also violate Article III: 2, without that demonstrating that the Panel made any error in its analysis.

With its third claim of appeal under Article III: 4 of the GATT, Brazil contests the Panels finding 8. that requirements related to production of a product may fall within Article III: 4 of the GATT, because according to Brazil they would fall under Article III:8(b) of the GATT. However, Brazil does not appeal Section 7.2.1 of the Report, where the Panel found that that the scope of Article III:4 extends to measures presented in the form of requirements for producers as opposed to direct requirements on products. Brazil does not even formulate any argument trying to distinguish this case from a long line of cases, which confirms the Panel's findings. Therefore, Brazil's appeal appears pointless as it is limited to affirming some errors in the Panel Report without challenging the legal and factual premises for those findings. For the rest Brazil is making some undemonstrated assertions and ultimately is subsuming its first general defence before the Panel about "pre-market" requirements in its second general defence according to which the challenged ICT programmes are exempted from the disciplines of Article III, by virtue of Article III:8(b). Hence, Brazil's arguments are based again on the wrong premise that, just because Brazil grants subsidies only to domestic producers, it is entitled to introduce any differences in treatment between domestic and foreign producers and products together with those subsidies.

9. With the fourth claim of appeal under Article III:4 of the GATT, Brazil challenges the Panel findings that the challenged ICT programmes result in a heavier administrative burden on imported product inconsistent with Article III:4. However, the Panel reasoning is firmly grounded on the

- 32 -

functioning of the credit-debit system as described in the Panel Report and on what Brazil acknowledged before the Panel. At the same time. Brazil does not even argue (let alone demonstrate) that the Panel violated Article 11 of the DSU by failing to make an objective assessment of the facts of the case when it described the functioning of the Brazilian tax system and the credit-debit mechanism.

10 With its fifth claim of appeal under Article III:4 of the GATT, Brazil challenges the Panel's finding that the PPBs and other production step requirements under the ICT programmes are inconsistent with Article III: 4 of the GATT 1994 because they require the use of domestic products to the detriment of imported products, when accredited companies outsource the production of components and subassemblies to be used in the production of incentivised products. The European Union thinks that the Panel's findings concerning the so called "outsourcing scenario" are correct. The complainants claimed before the Panel that the challenged ICT programmes impose on the companies accredited for the production of a given incentivised product (intermediate or finished) a condition to use components and subassemblies produced in Brazil (hence domestic products) in order to obtain a tax advantage/subsidy in relation with the production of the incentivised product. This condition or requirement results from the production step requirements (contemplated by the ICT programmes), which are specific to each incentivised product, and which must be complied with in order for a company to be accredited and be eligible for the tax advantages in relation to that product. Some of those production steps require the manufacturing in Brazil of components or subassemblies for the incentivised product. Hence, those components or subassemblies are domestic products that must be used in the production of the incentivised product.

11. Brazil confirms that the conditions for companies to obtain accreditation define the condition of eligibility for the tax advantages provided for under the challenged ICT programmes and it does not claim that the Panel made whatever error in the description of the functioning of the PPBs or the production steps requirements more generally. Brazil also confirms that the production steps requirements or subassemblies in Brazil and it does not question that these products should be considered as domestic.

12. Accordingly, the Panel did not err in finding that the certain production step requirements under the challenged ICT programme require that domestic components and subassemblies be incorporated into the incentivised (finished or intermediate product), as a condition for receiving the tax advantages in relation with that product. Given that accredited companies are required to use domestic components if they want to benefit of tax advantages in relation to their incentivised products, this condition or requirement distorts the conditions of competition between domestic components and subassemblies and like imported components and subassemblies and therefore accords less favourable treatment to imported components and subassemblies.

13. Brazil's argument that under the ICT programmes accredited producers are free to source inputs from foreign producers provided that the relevant production steps are undertaken in Brazil is simply missing the point, since the claim adjudicated by the Panel does not encompass raw materials and other basic components, but the domestic components and subassemblies that result from the performance in Brazil of certain production steps.

14. Brazil is also wrong in arguing that the Panel's conclusions imply that the only national production process that could be subsidized in a WTO-consistent manner is the assembly of inputs, not their production.

15. It follows from the above that Brazil's claims of appeal under Article III: 4 of the GATT (to the exception of the claim based on the interpretation of Article III: 8(b)) do not concern the Panel's findings contained in Section 7.3.2.1.4.2 relating to the mechanism for the calculation of the amount of resources required to be invested in R&D under the Informatics and PADIS programmes. Indeed, none of the arguments formulated by Brazil take issue with anything the Panel found in that Section. Hence, it must necessarily be concluded that Brazil does not challenge the Panel's finding in paragraph 7.243 of the Report, according to which the mechanisms under the Informatics and PADIS programmes for the calculation of the amount of resources required to be invested in R&D accord to imported products treatment less favourable than that accorded to like domestic products, and thus are inconsistent with Article III: 4 of the GATT 1994.

- 33 -

16. In light of the foregoing, the European Union submits that Brazil has failed to show the existence of any reversible error when the Panel found that the ICT programmes were inconsistent with Article III: 4 of the GATT 1994.

1.2.3 Article 1.1(a)(1)(ii) of the SCM Agreement

17. Brazil wrongly argues that the Panel made errors when finding that the ICT programmes amounted to prohibited subsidies under Article 3.1(b) of the SCM Agreement. The Panel properly interpreted and applied Article 1.1(a)(1)(ii) of the SCM Agreement when finding that the tax exemptions, suspensions and reductions granted under the ICT programmes amounted to "government revenue that is otherwise foregone or not collected". In fact, the Panel carried out a balanced assessment when comparing the tax treatment of similarly situated taxpayers in accordance with Brazil's general and predominant rules of taxation in relation to non-accredited companies.

1.2.4 Article 3.1(b) of the SCM Agreement

With this claim of appeal Brazil argues that the Panel erred in finding that the tax exemptions, 18. reductions and suspensions granted under the ICT programmes are prohibited import substitution subsidies within the meaning of Article 3.1(b) of the SCM Agreement. However, the Panel defined and applied the correct legal standard under Article 3.1(b) of the SCM agreement. In line with the Appellate Body's findings in US - Tax Incentives, the Panel focussed its analysis from the very beginning on the conditions for companies to obtain accreditation and thereby be eligible for the tax exemptions, reductions or suspensions under the programmes. The Panel stressed that element of conditionality all along its analysis. Therefore, it is not correct to say that the Panel based its findings on the effects of the subsidies, or on the fact that accredited companies would likely use more domestic products in their downstream production activities. The Panel focused on the conditions of eligibility for the subsidies, and found that those conditions impose the use of domestic components and subassemblies, under the guise of production steps requirements, in order to obtain the subsidy in relation to the production of the incentivised products. At the same time there is no dispute that the components and subassemblies at issue in the present case are domestic because they are produced in Brazil (regardless of the origin of their raw materials or inputs).

19. It is also worth recalling that there are some key differences between this case and the situation in *US* – *Tax Incentives* and the Appellate Body itself clarified that whether a subsidy is conditional upon the use by the subsidy recipient of domestic over imported goods, should be assessed on a case-by-case basis. The production steps requirements at issue in the present case are not limited to simply specifying the location of an assembly operation of the incentivised product, rather they concern the production in Brazil of component and subassemblies of the incentivised product. Moreover, those subsidies are not granted because the accredited companies are engaged in the production of the components of the incentivised product, but they are explicitly made contingent upon the use of certain components or subassemblies made in Brazil (by the accredited company or a third company). Hence, this case (unlike the *US-Tax Incentives* case) concerns subsidies granted exclusively for the production of the incentivised products (the final goods) and not for the production of their inputs, too. Finally, the components that are manufactured in Brazil in order to comply with the production step requirements of the ICT programmes must, by definition, be used in the production of the incentivised products.

20. In nutshell all of Brazil's claims of appeal in respect of the Panel's findings with regard to the imposition of local content requirements boil down to say that if a requirement for granting a subsidy is drafted in terms of production activity, it is indifferent, for the purpose of Article 3.1(b) of the SCM Agreement, if such production activity results in the creation of a domestic component that *must be used* in the production of the incentivised product. As Brazil would have it, Article 3.1(b) of the SCM Agreement would prohibit Brazil from requiring accredited companies to "use *domestic* component and subassemblies" when producing incentivised products. However, it would leave it completely free to require the same companies to "use component and subassemblies *produced in Brazil*" when producing incentivised products. It goes without saying that this cannot be the correct interpretation of Article 3.1(b) of the SCM Agreement.

- 34 -

1.3 INOVAR-AUTO PROGRAMME

21. In essence, Brazil claims that the Panel erred by excluding the application of Article III:8(b) of GATT 1994 and when it found that the accreditation requirements are inconsistent with Article III:4. It also maintains that the Panel erred in finding that the different processes for companies seeking accreditation as "domestic manufacturers" and companies seeking accreditation as "investors" or "importers/distributors" would *per se* result in less favourable treatment being accorded to imported finished motor vehicles, under Article III:4 of GATT 1994. The European Union will explain below why all those claims should be rejected by the Appellate Body.

22. Brazil's arguments should be rejected entirely. Far from just being a subsidy programme exclusively for domestic producers, the INOVAR-AUTO programme discriminates between domestic and imported vehicles. Foreign companies seeking accreditation under the INOVAR-AUTO programme should opt for becoming "importers/distributors" by establishing in Brazil in order to carry out certain marketing and sale activities in the Brazilian market for the imported vehicles to benefit from the same presumed IPI credits granted to domestic vehicles produced by domestic manufacturers or investors in Brazil accredited under the programme. Therefore, the Panel properly found that the accreditation requirements under the INOVAR-AUTO modify the conditions of competition to the detriment of imported motor vehicles and in favour of like domestic motor vehicles.

1.4 PEC and RECAP programmes

Brazil raises four claims against the Panel's findings that the PEC and RECAP programmes 23. amounted to prohibited export subsidies under Article 3.1(a) of the SCM Agreement. In particular, Brazil disputes the Panel's conclusions that the tax treatment granted through the PEC and RECAP programmes amounted to "revenue otherwise due that is foregone or not collected", as well as the Panel's conclusions on export contingency. Brazil's four claims of error should be rejected. The Panel properly applied the test developed by the Appellate Body in US - Large Civil Aircraft (2nd Complaint) when, having examined the structure and organising principles of Brazil's tax system, the Panel found that the benchmark for comparison was the tax treatment of non-registered/non-accredited companies under the PEC/RECAP programmes, as similarly situated taxpayers in comparable transactions. The Panel did not err either when examining the tax treatment of those taxpayers including the rather frequent scenario where there is tax credit accumulation. The Panel properly interpreted and applied Article 1.1(a)(1)(ii) of the SCM Agreement when finding that the cash availability and the implicit interest income that the Brazilian Government could earn (had the tax been paid in advance and not offset) fall under the terms "revenue otherwise due is foregone or not collected". Finally, the European Union considers that the Panel properly weighted all the evidence in accordance with its duties under Article 11 of the DSU when finding that the PEC/RECAP programmes were contingent upon export performance, in particular in view of the clear requirement for companies to have more than 50% of their revenues derived from exports.

1.5 Enabling Clause

24. The European Union considers that Brazil's claims of error with respect to the Panel's findings on the Enabling Clause should be rejected by the Appellate Body. First, the Panel correctly found that a complaining party cannot be under an obligation to invoke the Enabling Clause in its panel request, unless that complaining party is appropriately informed that the responding party considers the challenged measure to have been adopted pursuant to (and justified under) the Enabling Clause. Second, the Panel correctly established that the conditions in paragraphs 2(b) and 2(c) of the Enabling Clause are not met by the Brazilian measures in the present case. In addition, whilst Brazil complaints about the Panel's assessment and characterisation of the facts (such as the Panel's finding that the accreditation process under the INOVAR-AUTO programme is more burdensome to foreign manufacturers than to domestic ones), Brazil has no claim under Article 11 of DSU with respect to the Panel's assessment of Brazil's invocation of the Enabling Clause under paragraph 2(c).

1.6 Article 4.7 of the SCM Agreement

25. The European Union maintains that, contrary to Brazil's allegations, the Panel did not make an error under Articles 11 or 12.7 of the DSU when recommending, pursuant to Article 4.7 of the SCM Agreement, that Brazil withdraw the prohibited subsidies found within 90 days. The Panel's - 35 -

rationale and explanations are sufficiently evident from the Panel Report. Brazil's disagreement on substance should not be construed as a reversible error.

- 36 -

ANNEX B-6

EXECUTIVE SUMMARY OF BRAZIL'S APPELLEE'S SUBMISSION¹

1. Brazil requests that the Appellate Body reject all of the claims of error raised by the European Union and Japan in their other appeals, for the following reasons:

2. First, the complainants failed to establish that the Panel acted inconsistently with Article 11 of the DSU in failing to address their *arguments* that in the "in-house" scenario the ICT programmes and INOVAR-AUTO give rise to a contingency on the use of domestic over imported goods under Articles III:4 of the GATT 1994, Article 2.1 of the TRIMs Agreement, and Article 3.1(b) of the SCM Agreement. The Panel's decision not to address the complainants' *arguments* in this respect could not be properly characterized as exercise of judicial economy, which occurs when a panel decides not to address multiple *claims* raised by a WTO Member in respect of the same measure. The Panel clearly examined and made findings in respect of all of the legal claims raised by the complainants in respect of the ICT programmes and INOVAR-AUTO. In any event, the Panel acted within the bounds of its discretion under Article 11 of the DSU in deciding not to address the complainants' *arguments* that in the "in-house" scenario the ICT programmes and INOVAR-AUTO allegedly gave rise to a requirement to use domestic over imported goods.

3. Second, the European Union's "subordinate claim" that the Panel erred in its application of Articles III:4 of the GATT 1994, Article 2.1 of the TRIMs Agreement, and Article 3.1(b) of the SCM Agreement by failing to address arguments concerning the "in-house" scenario is not properly within the scope of appellate review. The European Union's allegation that the Panel failed to consider "all relevant facts of the case" is directed at the Panel's appreciation of the facts, and therefore should have been brought under Article 11 of the DSU. The European Union's failure to bring its "subordinate" appeal claim under Article 11 of the DSU is dispositive, and for this reason alone should be rejected by the Appellate Body.

4. Third, the complainants remain unable to establish that in the "in-house" scenario, both the ICT programmes and INOVAR-AUTO are inconsistent with Article III:4 of the GATT 1994, Article 2.1 of the TRIMs Agreement, and Article 3.1(b) of the SCM Agreement. The complainants' interpretation that these provisions prohibit subsidies contingent upon domestic production in-country renders Article III:8(b) of the GATT 1994 entirely meaningless. Moreover, the complainants' arguments in respect of both the "in-house" and "outsourcing" scenarios are based on the proposition expressly rejected by the Appellate Body in *US* – *Tax Incentives* that subsidies contingent on producing certain goods domestically *per se* constitute prohibited import-substitution subsidies.

¹ Total word count: 463

- 37 -

ANNEX C

ARGUMENTS OF THE THIRD PARTICIPANTS

	Contents	Page
Annex C-1	Executive summary of Argentina's third participant's submission	38
Annex C-2	Executive summary of Australia's third participant's submission	39
Annex C-3	Executive summary of the United States' third participant's submission	40

- 38 -

ANNEX C-1

EXECUTIVE SUMMARY OF ARGENTINA'S THIRD PARTICIPANT'S SUBMISSION

1. Argentina has a strong interest in the proper interpretation of the Enabling Clause and, in particular, in the development of an effective and coherent approach to interpreting its paragraphs.

2. Argentina will refer to the following issues: (1) Whether the Panel erred in finding that the claims under Article I:1 of the GATT 1994 were properly included in the terms of reference; (2) Whether the challenged treatment accorded to imports from Argentina, Mexico and Uruguay under INOVAR-AUTO is justified under paragraph 2(b) of the Enabling Clause; (3) Whether the challenged treatment accorded to imports from Argentina, Mexico and Uruguay under INOVAR-AUTO is justified under paragraph 2(c) of the Enabling Clause.

3. In particular, Argentina considers that the differential treatment accorded through Decree 7,819/2012 is *part of* a notified regional arrangement and therefore does not require additional notification.¹ As *part of* that notified arrangement, Argentina considers that the complainants should have been on notice that Brazil could invoke the Enabling Clause as a justification for the differential treatment granted under INOVAR-AUTO notwithstanding the provisions of Article I:1 of the GATT 1994.

4. Argentina shares the view of Brazil that the Panel's interpretation renders paragraph 2(b) useless, since it would limit its provisions to what is already established in the covered agreements. Hence, Argentina considers that the analysis under this paragraph should be reversed and completed by the Appellate Body.

5. Regarding the "criteria or conditions for the mutual reduction or elimination of non-tariff measures", paragraph 2(c) uses the phrase "may be prescribed". Argentina considers that a correct interpretation of this phrase would result in concluding that it does not constitute a positive obligation. The fact that in the last 38 years WTO Members have not established such criteria and conditions demonstrated that this phrase cannot be interpreted as a prohibition to apply the provisions in paragraph 2(c) related to non-tariff measures.

¹ See Brazil's second written submission, para. 181; see also Brazil's first written submission (DS472), paras. 638-639.

- 39 -

ANNEX C-2

EXECUTIVE SUMMARY OF AUSTRALIA'S THIRD PARTICIPANT'S SUBMISSION

Australia considers that the Panel was correct in finding that, while subsidies provided 1. exclusively to domestic producers under Article III:8(b) of the GATT 1994 do not per se violate Article III: 2,¹ such a subsidy could still violate Article III: 2 if designed and applied in a manner that discriminates between imported and domestic products.²

A panel is therefore not required to undertake a 'threshold inquiry' as to whether a subsidy 2. is product-related or production-related because the disciplines of Article III apply to both types of subsidies.³

3. Australia shares the concerns of Japan and the EU that the Panel's use of judicial economy casts doubt on whether the Panel found that the measures in their entirety, or merely as they operated in the outsourcing scenario, violated Brazil's WTO obligations.

4. In Australia's view, a panel's mandate under the DSU to make an objective assessment of the matter before it - with a view to its positive resolution - limits the degree to which a panel can exercise judicial economy; and requires a panel to ensure its findings can lead to a satisfactory settlement of the dispute.

¹ Panel Report, *Brazil – Taxation*, para. 7.77 ² Panel Report, *Brazil – Taxation*, paras. 7.85 and 7.88.

³ Panel Report, Brazil - Taxation, para. 7.88.

- 40 -

ANNEX C-3

EXECUTIVE SUMMARY OF THE UNITED STATES' THIRD PARTICIPANT'S SUBMISSION

I. GATT 1994 ARTICLE III:8(B)¹

1. A measure conditioning an advantage, such as a subsidy, on the use of a domestic good would breach Article III: 4. In light of Article III: 8(b), Article III: 4 must be interpreted as not prohibiting the provision of subsidies *because of* the exclusion of foreign producers as eligible subsidy recipients. Such a payment should not be understood, on the basis of this exclusion alone, to be in conflict with Article III: 4 (including its requirement of national treatment for measures affecting the "use" of a product).

2. SCM Agreement Article 3.1(b) and GATT 1994 Article III share similar disciplines. Subsidies contingent on the use of domestic over imported products would be inconsistent with *both* provisions. The Panel thus correctly analyzed the disputed programs under both Article III and Article 3.1(b).

3. While Members must be free to define the domestic producers who are to receive subsidies, including the productive activities that make them a domestic producer, we consider that subsidies conditioned not solely on domestic production, but also the use of domestic products, fall outside Article III:8(b)'s derogation.

II. GATT 1994 ARTICLE III:4

4. The Panel found that "nested" PPBs—which require the use of inputs that themselves conform to another PPB—contain an explicit requirement to use domestic over imported goods. Brazil has not appealed these findings. Such a requirement is inconsistent with Article III: 4.

III. SCM AGREEMENT ARTICLE 3.1(B)

5. With respect to intermediate goods, we note that if a measure *exempts* taxes, government revenue, otherwise due, is clearly foregone under SCM Agreement Article 1.1(a)(1)(ii). If a measure *suspends* taxes that are later paid, a financial contribution is still provided: at the time when government revenue would otherwise be due, it is foregone (albeit temporarily). In addition, implicit interest may also be foregone.

6. Brazil has not appealed the Panel's factual findings regarding the disputed programs. Those facts show that the tax advantages under the programs are contingent on the use of domestic over imported goods under Article 3.1(b) with respect to nested PPBs.

¹ This executive summary contains a total of 382 words (including footnotes), and the U.S. third participant submission contains 3831 words (including footnotes).

- 41 -

ANNEX D

PROCEDURAL RULING

	Contents	Page
Annex D-1	Procedural ruling of 12 October 2017 regarding the deadline for filing of third participants' submissions	42

- 42 -

ANNEX D-1

PROCEDURAL RULING OF 12 OCTOBER 2017

1. On 28 September 2017, Brazil filed a Notice of Appeal in the above proceedings. In accordance with Rule 26 of the Working Procedures for Appellate Review (Working Procedures), a Working Schedule for Appeal was drawn up by the Appellate Body Division hearing this appeal and communicated to the participants and third parties on 29 September 2017.

2. On 9 October 2017, the Division received a joint letter from Argentina, Australia, Canada and the United States requesting an extension of the deadline for the filing of their third participants' submissions in these proceedings (the joint request). These third participants explained that some of them are also third parties in another pending appellate proceeding, namely, *Indonesia – Safeguard on Certain Iron or Steel Products* (DS490 / DS496), and that the third participants' submissions in that proceeding are due on the same day as the deadline for the filing of their third participants' submissions in the present proceedings, i.e. 19 October 2017. In their joint request, the four third participants also highlighted the length of the panel report, and of Brazil's appellant's submission in these appellate proceedings. Argentina, Australia, Canada and the United States therefore requested that the deadline for the filing of the third participants' submissions be extended by one week, until 26 October 2017.

3. On 9 October 2017, we invited the participants and third parties, if they so wished, to comment on the joint request by 12 noon on 11 October 2017. We received no objections to the joint request. Brazil indicated that it did not oppose the request and Japan stated it had no comments.

4. We consider the reasons identified by Argentina, Australia, Canada and the United States, in particular the fact that the deadlines for the filing of third participants' submissions in two separate appeal proceedings currently fall on the same day, to be relevant factors in our assessment of the joint request. We further note that Brazil, the European Union and Japan and the other third parties have not raised any objection to the joint request.

5. In these circumstances, the Division has decided, in accordance with Rule 16(2) of the Working Procedures, to extend the deadline for the filing of third participants' submissions by one week to <u>Thursday, 26 October 2017</u>.

Modified Dates for the Submission of Documents

Process	<u>Rule</u>	<u>Date</u>
Notice of Appeal	Rule 20	Thursday, 28 September 2017
Appellant's submission and executive summary	Rule 21(1)	Thursday, 28 September 2017
Notice of Other Appeal	Rule 23(1)	Tuesday, 3 October 2017
Other appellant's submission and executive summary	Rule 23(3)	Tuesday, 3 October 2017
Appellee's submission(s) and executive summary(ies)	Rules 22 and 23(4)	Monday, 16 October 2017
Third participants' submissions and executive summaries	Rule 24(1)	Thursday, 26 October 2017
Third participants' notifications	Rule 24(2)	Thursday, 26 October 2017



13 December 2018

Page: 1/42

Original: English

BRAZIL - CERTAIN MEASURES CONCERNING TAXATION AND CHARGES

AB-2017-7 AB-2017-8

Reports of the Appellate Body

Addendum

This *Addendum* contains Annexes A to D to the Report of the Appellate Body circulated as documents WT/DS472/AB/R, WT/DS497/AB/R.

The Notices of Appeal and Other Appeal and the executive summaries of written submissions contained in this Addendum are attached as they were received from the participants and third participants. The content has not been revised or edited by the Appellate Body, except that paragraph and footnote numbers that did not start at one in the original may have been re-numbered to do so, and the text may have been formatted in order to adhere to WTO style. The executive summaries do not serve as substitutes for the submissions of the participants and third participants in the Appellate Body's examination of the appeal.

(18-7794)

- 2 -

LIST OF ANNEXES

ANNEX A

NOTICES OF APPEAL AND OTHER APPEALS

	Contents	Page
Annex A-1	Brazil's Notice of Appeal	4
Annex A-2	The European Union's Notice of Other Appeal	11
Annex A-3	Japan's Notice of Other Appeal	12

ANNEX B

ARGUMENTS OF THE PARTICIPANTS

	Contents	Page
Annex B-1	Executive summary of Brazil's appellant's submission	14
Annex B-2	Executive summary of the European Union's other appellant's submission	23
Annex B-3	Executive summary of Japan's other appellant's submission	24
Annex B-4	Executive summary of Japan's appellee's submission	26
Annex B-5	Executive summary of the European Union's appellee's submission	30
Annex B-6	Executive summary of Brazil's appellee's submission	36

ANNEX C

ARGUMENTS OF THE THIRD PARTICIPANTS

	Contents	Page
Annex C-1	Executive summary of Argentina's third participant's submission	38
Annex C-2	Executive summary of Australia's third participant's submission	39
Annex C-3	Executive summary of the United States' third participant's submission	40

ANNEX D

PROCEDURAL RULING

	Contents	Page
Annex D-1	Procedural ruling of 12 October 2017 regarding the deadline for filing of third participants' submissions	42

- 3 -

ANNEX A

NOTICES OF APPEAL AND OTHER APPEALS

	Contents	Page
Annex A-1	Brazil's Notice of Appeal	4
Annex A-2	The European Union's Notice of Other Appeal	11
Annex A-3	Japan's Notice of Other Appeal	12

- 4 -

ANNEX A-1

BRAZIL'S NOTICE OF APPEAL*

1. Pursuant to Articles 16.4 and 17 of the Understanding on Rules and Procedures Governing the Settlement of Disputes ("DSU") and Rule 20 of the Working Procedures for Appellate Review (WT/AB/WP/6) ("Working Procedures"), Brazil hereby notifies the Dispute Settlement Body of its decision to appeal certain issues of law and legal interpretation in the reports of the Panel in Brazil – Certain Measures Concerning Taxation and Charges (WT/DS472, DS497) ("Panel Report").

2. The seven programmes at issue are: (a) the Informatics programme; (b) the programme of Incentives for the Semiconductors Sector ("PADIS programme"); (c) the programme of Support for the Technological Development of the Industry of Digital TV Equipment ("PATVD programme"); (d) the programme for Digital Inclusion ("Digital Inclusion programme"); (e) the programme of Incentive to the Technological Innovation and Densification of the Automotive Supply Chain ("INOVAR-AUTO"); (f) the regime for predominantly exporting companies ("PEC programme"); and the Special Regime for the Purchase of Capital Goods for Exporting Enterprises ("RECAP programme").¹

3. The issues that Brazil raises in this appeal relate to the Panel's findings and conclusions with respect to the consistency of the challenged programmes with various provisions of the General Agreement on Tariffs and Trade 1994 (GATT 1994), the Agreement on Subsidies and Countervailing Measures (SCM Agreement), and the Agreement on Trade-Related Investment Measures (TRIMS Agreement).

4. Pursuant to Rules 20(1) and 21(1) of the Working Procedures, Brazil files this Notice of Appeal together with its Appellant's Submission with the Appellate Body Secretariat.

5. Pursuant to Rule 20(2)(d)(iii) of the Working Procedures, this Notice of Appeal provides an indicative list of the paragraphs of the Panel Report containing the alleged errors of law and legal interpretation by the Panel, without prejudice to Brazil's ability to rely on other paragraphs of the Panel Report in its appeal.

I. REVIEW OF THE PANEL'S INTERPRETATION AND APPLICATION OF ARTICLE III:8(b) OF THE GATT 1994 TO THE INFORMATICS, PADIS, PATVD, DIGITAL INCLUSION PROGRAMMES ("ICT PROGRAMMES") AND THE INOVAR-AUTO PROGRAMME

6. Brazil seeks review by the Appellate Body of the Panel's finding that domestic production subsidies within the meaning of Article III:8(b) of the GATT 1994 are not exempt from the disciplines of Article III thereof. The Panel's errors of law and legal interpretation include:

- The Panel erred in finding that the scope of Article III:8(b) of the GATT 1994 is limited to the provision of subsidies to domestic producers where those subsidies do not have any component that introduces discrimination between imported and domestic products;²
- The Panel erred in finding that Article III:8(b) of the GATT 1994 only serves to clarify that Members are not required to subsidize foreign producers in tandem with domestic producers;³
- The Panel erred by failing to determine, in the first instance, whether the measures at issue are product-related measures subject to the disciplines of Article III of the

 $^{^{\}ast}$ This document, dated 28 September 2017, was circulated to Members as document WT/DS472/8 - WT/DS497/6.

¹ The complainants explained in their panel requests that these programmes are "set up and implemented", "established and administered" through the measures identified in paragraph 2.38 of the Panel Report.

² See Panel Report, paras. 7.83-7.87.

³ See Panel Report, para. 7.84.

GATT 1994 or, conversely, whether the measures provide subsidies to domestic producers and are therefore subject to the disciplines of the SCM Agreement.

7. For these reasons, Brazil respectfully requests that the Appellate Body *reverse* the Panel's interpretative finding, in paragraphs 7.87, 8.3, and 8.14 of the Panel Report, that subsidies that are provided exclusively to domestic producers pursuant to Article III:8(b) of the GATT 1994 are not *per se* exempted from the disciplines of Article III of the GATT. As a result, Brazil also requests that the Appellate Body *reverse* all of the Panel's consequential findings to this effect in paragraphs 7.93, 7.184, 7.631, and 7.696 of the Panel Report.

8. Brazil respectfully requests that the Appellate Body *find* that the Panel erred by failing to determine, in the first instance, whether the measures at issue are product-related measures subject to the disciplines of Article III of the GATT 1994 or, conversely, whether the measures provide subsidies to domestic producers and are therefore subject to the disciplines of the SCM Agreement.

9. Consequently, Brazil also requests that the Appellate Body *reverse* the Panel's findings in paragraphs 7.174, 7.318, 7.319, 7.688, 7.772, 7.773, 8.5(a), 8.5(b), 8.6(a), 8.6(b), 8.16(a), 8.16(c), 8.17(a), and 8.17(c) of the Panel Report that the ICT programmes and INOVAR-AUTO are inconsistent with Articles III:2 and III:4 of the GATT 1994. The Appellate Body should also *reverse* the Panel's findings in paragraphs 7.365, 7.806, 8.5(d), 8.6(d), 8.16(e), and 8.17(e) of the Panel Report that the ICT programmes and INOVAR-AUTO are inconsistent with Article 2.1 of the TRIMs Agreement.

II. REVIEW OF THE PANEL'S INTERPRETATION AND APPLICATION OF ARTICLE III:2 OF THE GATT 1994 TO THE ICT PROGRAMMES

10. Brazil seeks review by the Appellate Body of the Panel's finding that production-step requirements under the ICT programmes, and the requirement for products to obtain the status of "developed" in Brazil, result in imported products being subject to internal taxes in excess of those applied to like domestic products, inconsistently with Article III: 2, first sentence, of the GATT 1994. The Panel's errors of law and legal interpretation include:

- The Panel erred in failing to determine that the ICT programmes constitute "payments of subsidies exclusively to domestic producers" within the meaning of Article III:8(b) of the GATT 1994;⁴
- The Panel erred in finding that the complainants had demonstrated that imported products were taxed in excess of domestic like products under the ICT programmes within the meaning of Article III: 2;⁵
- The Panel erred in finding that the "cash flow" and the "time-value of money" effects under the ICT programmes result in a higher tax burden on imported intermediate products.⁶

11. For these reasons, Brazil requests further that the Appellate Body *reverse* the Panel's findings in paragraphs 7.174, 8.5(a), and 8.16(a) of the Panel Report that the ICT programmes result in imported products being subject to internal taxes in excess of those applied to like domestic products, inconsistently with Article III:2, first sentence, of the GATT 1994.

12. As a result, Brazil also requests that the Appellate Body *reverse* the Panel's consequential finding in paragraphs 7.365, 8.5(d), and 8.16(e) of the Panel Report that the ICT programmes are inconsistent with Article 2.1 of the TRIMS Agreement.

III. REVIEW OF THE PANEL'S INTERPRETATION AND APPLICATION OF ARTICLE III:4 OF THE GATT 1994 TO THE ICT PROGRAMMES

13. Brazil seeks review by the Appellate Body of the Panel's finding that the ICT programmes accord to imported products treatment less favourable than that accorded to like domestic products,

⁴ See Panel Report, paras. 7.77-7.87, 7.93.

⁵ See Panel Report, para. 7.151.

⁶ See Panel Report, paras. 7.170-7.172.

- 6 -

inconsistently with Article III: 4 of the GATT 1994. The Panel's errors of law and legal interpretation include:

- The Panel erred in failing to determine that the ICT programmes constitute "payments of subsidies exclusively to domestic producers" within the meaning of Article III:8(b) of the GATT 1994;⁷
- The Panel erred in finding that the accreditation requirements under the ICT programmes are inconsistent with Article III:4 of the GATT 1994;⁸
- The Panel erred in finding that the ICT programmes are inconsistent with Article III: 4 of the GATT 1994 by virtue of purported administrative burdens on intermediate products;⁹
- The Panel erred in finding that the PPBs and other production step requirements under the ICT programmes are inconsistent with Article III: 4 of the GATT 1994 because they require the use domestic products.¹⁰

14. For these reasons, Brazil respectfully requests that the Appellate Body *reverse* the Panel's finding in paragraphs 7.318, 7.319, 8.5(b), and 8.16(c) of the Panel Report that the ICT programmes accord to imported products treatment less favourable than that accorded to like domestic products, inconsistently with Article III: 4 of the GATT 1994.

15. As a result, Brazil also requests that the Appellate Body reverse the Panel's consequential finding in paragraphs 7.365, 8.5(d), and 8.16(e) of the Panel Report that the ICT programmes are inconsistent with Article 2.1 of the TRIMS Agreement.

IV. REVIEW OF THE PANEL'S INTERPRETATION AND APPLICATION OF ARTICLE 3.1(b) OF THE SCM AGREEMENT TO THE ICT PROGRAMMES

16. Brazil seeks review by the Appellate Body of the Panel's finding that the ICT programmes provide subsidies that are contingent upon the use of domestic over imported goods under

Article 3.1(b) of the SCM Agreement. The Panel's errors of law and legal interpretation include:

- The Panel erred in finding that the tax exemptions, suspensions and reductions granted under the ICT programmes both on the sales of intermediate goods and on the purchases of raw materials, intermediate goods, packaging materials, inputs, capital goods and computational tools constitute financial contributions, in the form of government revenue otherwise due that is foregone or not collected, within the meaning of Article 1.1(a)(1)(ii) of the SCM Agreement;¹¹
- The Panel erred in finding that the tax exemptions, reductions and suspensions granted under the ICT programmes are prohibited subsidies within the meaning of Article 3.1(b) of the SCM Agreement;¹²

17. For these reasons, Brazil respectfully requests that the Appellate Body *reverse* the Panel's findings in paragraphs 7.489 and 7.490 that the exemptions, suspensions and reductions of taxes on the sales of intermediate products and on the purchases of inputs by accredited companies under the ICT programmes constitute financial contributions in the form of "government revenue that is otherwise due is foregone or not collected" within the meaning of Article 1.1(a)(1)(iii) of the SCM Agreement.

⁷ See Panel Report, paras. 7.77-7.87, 7.184.

⁸ See Panel Report, para. 7.220-7.225.

⁹ See Panel Report, paras. 7.252-7.255.

¹⁰ See Panel Report, paras. 7.288, 7.299-7.301; and 7.311-7-313.

¹¹ See Panel Report, paras. 7.432-7.433, 7.444-7.445; 7.454-7.455; 7.463-7.464; 7.473-7.474; and 7.495.

¹² See Panel Report, paras. 7.274-7.302, 7.319.

- 7 -

18. Consequently, the Appellate Body should also *reverse* the Panel's finding in paragraphs 7.493 and 7.494 that the exemption, reduction and suspension of taxes on sales of intermediate products and on purchases of inputs by accredited companies under the ICT programmes confer a benefit on the recipient under Article 1.1(b) of the SCM Agreement. As a result, the Appellate Body should also *reverse* the Panel's ultimate conclusion in paragraphs 7.495, 8.5(e), and 8.16(f) that the that the exemption, reduction and suspension of taxes on sales of intermediate products and on purchases of inputs by accredited companies under the ICT programmes constitutes subsidies within the meaning of Article 1 of the SCM Agreement.

19. Brazil further requests that the Appellate Body *reverse* the Panel's finding in paragraphs 7.500, 8.5(e), and 8.16(f) that the exemptions, reductions, and suspensions granted under the ICT programmes are subsidies contingent upon the use of domestic over imported goods within the meaning of Article 3.1(b) of the SCM Agreement, and thus are prohibited subsidies, inconsistent with Article 3.1(b) and 3.2 of the SCM Agreement.

20. Consequently, Brazil also respectfully requests that the Appellate Body *declare moot and of no legal effect* the Panel's recommendation under Article 4.7 of the SCM Agreement that Brazil withdraw the ICT programs in 90 days, as reflected in paragraphs 8.11 and 8.22 of the Panel Report.

V. REVIEW OF THE PANEL'S INTERPRETATION AND APPLICATION OF ARTICLE III:4 OF THE GATT 1994 TO THE INOVAR-AUTO PROGRAMME

21. Brazil seeks review by the Appellate Body of the Panel's finding that certain aspects of the accreditation requirements under the INOVAR-AUTO Programme related to investments in R&D expenditures on engineering, basic industrial technology and to the performance of certain manufacturing steps in Brazil accord less favourable treatment to imported products than that accorded to like domestic products, inconsistently with Article III:4 of the GATT 1994. The Panel's errors of law and legal interpretation include:

- The Panel erred in failing to determine that the INOVAR-AUTO Programme constitutes "payments of subsidies exclusively to domestic producers" within the meaning of Article III:8(b);¹³
- The Panel erred in failing to conclude that because the accreditation requirements under the INOVAR-AUTO Programme do not relate to products, they cannot be *per se* inconsistent with Article III: 4 of the GATT 1994 or Article 2.1 of the TRIMs Agreement; ¹⁴
- The Panel erred in finding that accreditation requirements to perform a minimum number of manufacturing steps in Brazil are *per se* inconsistent with Article III:4 of the GATT 1994;¹⁵
- The Panel erred in finding that the different accreditation processes for companies seeking accreditation as "domestic manufacturers" and companies seeking accreditation as "investors" or "importers/distributors" would *per se* result in less favourable treatment being accorded to imported finished motor vehicles, under Article III: 4.¹⁶

22. For these reasons, Brazil respectfully requests that the Appellate Body *reverse* the Panel's finding, in paragraph 7.696 of the Panel Report, that Article III:8(b) of the GATT 1994 "does not exempt from the disciplines of Article III components of [domestic] production subsidies that introduce product discrimination in the form of less favourable treatment on imported like products."

23. Brazil also requests that the Appellate Body *reverse* the Panel's finding in paragraphs 7.773, 8.6(b) and 8.17(c) of the Panel Report that the accreditation requirements under INOVAR-AUTO to: (a) perform a minimum number of manufacturing steps in Brazil; and (b) invest in R&D in Brazil and make expenditures on engineering, basic industrial technology and capacity-building of suppliers in

¹³ See Panel Report, para. 7.696.

¹⁴ See Panel Report, para. 7.70.

¹⁵ See Panel Report, paras. 7.737-7.751.

¹⁶ See Panel Report, paras. 7.658-7.661, 7.732.

Brazil accord less favourable treatment to imported products than that accorded to like domestic products, inconsistently with Article III:4 of the GATT 1994.

24. As a result, Brazil requests that the Appellate Body also reverse the Panel's consequential finding in paragraphs 7.806, 8.6(d), and 8.17(e) of the Panel Report that these aspects of the accreditation requirements are inconsistent with Article 2.1 of the TRIMS Agreement.

25. To the extent that the Panel's findings are based on the erroneous assumption that production step requirements are sufficient to establish a contingency upon the use of domestic over imported goods within the meaning of Article 3.1(b) of the SCM Agreement, Brazil requests that the Appellate Body *reverse* the Panel's findings in paragraphs 7.751, 7.823, 7.847, 8.6(e), and 8.17(f) of the Panel Report that INOVAR-AUTO constitutes a prohibited import substitution subsidy under Article 3.1(b) and 3.2 of the SCM Agreement.

VI. REVIEW OF THE PANEL'S FINDINGS INTERPRETATION AND APPLICATION OF ARTICLES 1.1 AND 3.1(a) OF THE SCM AGREEMENT TO THE TAX SUSPENSIONS GRANTED UNDER THE PEC AND RECAP PROGRAMMES

26. Brazil seeks review by the Appellate Body of the Panel's finding that the tax suspensions granted under the PEC and RECAP programmes are subsidies within the meaning of Article 1.1 of the SCM Agreement and contingent upon export performance within the meaning of Article 3.1(a) of the SCM Agreement. The Panel's errors of law and legal interpretation include:

- The Panel erred in failing to use the tax treatment of structural tax accumulators as the benchmark treatment;¹⁷
- The Panel erred in its comparison of its preferred benchmark treatment with the challenged treatment under PEC and RECAP;¹⁸
- The Panel erred in finding that "cash availability" and "implicit interest income" are government revenue otherwise due;¹⁹
- The Panel acted inconsistently with Article 11 of the DSU in its assessment of the evidence of export contingency;²⁰

27. For these reasons, Brazil respectfully requests that the Appellate Body *reverse* the Panel's conclusions in paragraph 7.1211 of the Panel Report that the PEC and RECAP programmes constitute financial contributions in the form of government revenue that is otherwise due and is foregone or not collected within the meaning of Article 1.1(a)(1)(ii) of the SCM Agreement.

28. As a result, the Appellate Body should also *reverse* the Panel's finding in paragraph 7.1212 of the Panel Report that the PEC and RECAP programmes confer a benefit under Article 1.1(b) of the SCM Agreement, and its ultimate conclusion in paragraphs 7.1223, 8.7, and 8.18 that the tax suspensions under the PEC and RECAP programmes constitute subsidies within the meaning of Article 1.1 of the SCM Agreement.

29. Brazil respectfully requests that the Appellate Body *find* that the Panel acted inconsistently with its duty to conduct an objective assessment of the matter under Article 11 of the DSU in examining the evidence on export contingency.

30. Brazil also respectfully requests that the Appellate Body *reverse* the Panel's finding in paragraphs 7.1238, 8.7, and 8.18 that the tax suspensions granted under the PEC and RECAP programmes are subsidies that are contingent upon export performance within the meaning of Article 3.1(a) of the SCM Agreement.

¹⁷ See Panel Report, paras. 7.1167, 7.1169, and 7.183.

¹⁸ See Panel Report, paras. 7.1175, 7.1179, 7.1203, 7.1207.

¹⁹ See Panel Report, para. 7.1179, 7.1190, 7.1194.

²⁰ See Panel Report, para. 7.1234.

- 9 -

31. As a consequence, Brazil respectfully requests that the Appellate Body declare *moot and of no legal effect* the Panel's recommendation under Article 4.7 of the SCM Agreement in paragraphs 8.11 and 8.22 that Brazil withdraws these alleged subsidies within 90 days.

VII. REVIEW OF THE PANEL'S INTERPRETATION AND APPLICATION OF THE ENABLING CLAUSE TO THE TAX TREATMENT OF MERCOSUR COUNTRIES UNDER INOVAR-AUTO

32. Brazil seeks review by the Appellate Body of the Panel's finding that the complainants' claims under Article I:1 of the GATT 1994 were within the Panel's terms of reference, that the differential tax treatment accorded to Mexico, Argentina, and Uruguay under INOVAR-AUTO was not properly notified to the WTO, and that this differential tax treatment is not justified under paragraphs 2(b) and 2(c) of the Enabling Clause. The Panel's errors of law and legal interpretation include:

- The Panel erred in finding that the complainants were not on notice that the differential tax treatment at issue was adopted (and, in Brazil's view, justified) under the Enabling Clause, such that the complainants were required to invoke the Enabling Clause in their panel requests in order for the complainants claims under Article I:1 of the GATT 1994 to be within the Panel's terms of reference;²¹
- The Panel erred in finding that Brazil had not properly notified the differential tax treatment at issue under paragraph 4(a) of the Enabling Clause; ²²
- The Panel erred in finding that for a non-tariff measure to be within the scope of paragraph 2(b) of the Enabling Clause, it must be governed by specific provisions on special and differential treatment that are distinct from the provisions of the GATT 1994 incorporating the GATT 1947;²³
- The Panel erred by conflating the substantive analysis under paragraph 2(c) with the obligation to notify under paragraph 4(a), concluding that the differential tax treatment at issue is not justified under paragraph 2(c) of the Enabling Clause for the same reason that it concluded that the differential tax treatment was not properly notified.²⁴

33. For these reasons, Brazil respectfully requests that the Appellate Body *reverse* the Panel's conclusions in paragraphs 7.1083, 7.1120, 8.6(h), and 8.17(i) that the complainants did not have an obligation to invoke the Enabling Clause in their panel requests, and reverse the Panel's consequential conclusion that the complaints' claims under Article I:1 of the GATT 1994 were within the Panel's terms of reference.

34. Consequently, Brazil requests that the Appellate Body *declare moot and of no legal effect* the Panel's findings under paragraphs 8.6(g) and 8.17(h) that the tax reductions accorded to imported products from Argentina, Mexico and Uruguay under INOVAR-AUTO are advantages granted by Brazil to products originating in those countries, which are not accorded immediately and unconditionally to like products originating in other WTO Members, inconsistently with Article I: 1 of the GATT 1994.

35. In case the Appellate Body were to find that the complainants' claim under Article I:1 was properly within the Panel's terms of reference, Brazil respectfully requests that the Appellate Body *reverse* the Panel's conclusions in paragraphs 7.1079, 7.1081, 7.1083, 7.115 and 7.119 that the differential tax treatment at issue was not properly notified under paragraph 4(a) of the Enabling Clause.

36. Brazil respectfully requests that the Appellate Body *reverse* the Panel's conclusion in paragraphs 7.1096 that a non-tariff measure within the scope of paragraph 2(b) of the Enabling Clause must be governed by specific provisions on special and differential treatment that are distinct from the provisions of the GATT 1994 incorporating the GATT 1947. Brazil also requests that the Appellate Body *reverse* the Panel's consequential finding in paragraphs 7.1097, 8.6(i), and 8.17(j) that the tax reductions accorded to imported products from Argentina, Mexico and Uruguay and

²² See Panel Report, paras. 7.1076-7.1083, 7.1108-7.1115, 7.1119.

²¹ See Panel Report, paras. 7.1076-7.1083, 7.1108-7.1115, 7.1120.

²³ See Panel Report, paras. 7.1088-7.1097.

²⁴ See Panel Report, paras. 7.1108-7.1118.

found to be inconsistent under Article I:1 of the GATT 1994 are not justified under paragraph 2(b) of the Enabling Clause.

37. Brazil further requests that the Appellate Body *complete the analysis* and find that the differential tax treatment is justified under paragraph 2(b) and complies with the requirements of paragraph 3 of the Enabling Clause.

38. Brazil respectfully requests that the Appellate Body *reverse* the Panel's conclusion in paragraph 7.1118 that Brazil did not meet its burden of proof in respect of the substantive requirements of paragraph 2(c) of the Enabling Clause, and the Panel's ultimate conclusion in paragraphs 8.6(i) and 8.17(j) that the tax reductions accorded to imported products from Argentina, Mexico and Uruguay and found to be inconsistent under Article I:1 of the GATT 1994 are not justified under paragraph 2(c) of the Enabling Clause.

39. Brazil further requests that the Appellate Body *complete the analysis* and find that the differential tax treatment is justified under paragraph 2(c) and complies with the requirements of paragraph 3 of the Enabling Clause.

VIII. REVIEW OF THE PANEL'S RECOMMENDATION UNDER ARTICLE 4.7 OF THE SCM AGREEMENT

40. If the Appellate Body were to uphold the Panel's findings that the measures at issue are prohibited subsidies under Articles 3.1(a) and 3.1(b) of the SCM Agreement, Brazil then seeks review by the Appellate Body of the Panel's recommendation that Brazil withdraw the subsidies identified in paragraphs 8.5(e), 8.6(e), 8.7, 8.16(f), 8.17(f), and 8.18 of the Panel's Report within 90 days.²⁵ The Panel's errors of law and legal interpretation include:

- The Panel erred by failing to provide "reasoned and adequate explanations and coherent reasoning" for its 90-day recommendation, as required by Article 11 of the DSU;²⁶
- The Panel erred by failing to provide the "basic rationale" for its 90-day recommendation, as required by Article 12.7 of the DSU;²⁷

41. For these reasons, Brazil respectfully requests that the Appellate Body *find* that the Panel acted inconsistently with Articles 11 and 12.7 of the DSU in making its recommendations under Article 4.7 of the SCM Agreement and *reverse* the Panel's recommendation in paragraphs 8.11 and 8.22 of the Panel Report that Brazil withdraw the subsidies identified in paragraphs 8.5(e), 8.6(e), 8.7, 8.16(f), 8.17(f), and 8.18 within 90 days.

42. Brazil further requests that the Appellate Body *complete the analysis*, and specify a time period of 18 months for implementation pursuant to Article 4.7 of the SCM Agreement.

 $^{^{25}}$ Brazil notes that the Panel refers to the prohibited subsidies in paragraphs 8.5(f) and paragraphs 8.6(f) in paragraphs 8.11 and 8.22 of the Panel Report, but the relevant paragraphs are in fact 8.5(e) and 8.6(e).

²⁶ See Panel Report, para. 6.17.

²⁷ See Panel Report, para. 6.17.

- 11 -

ANNEX A-2

THE EUROPEAN UNION'S NOTICE OF OTHER APPEAL*

Pursuant to Article 16.4 and Article 17.1 of DSU, the European Union hereby notifies to the Dispute Settlement Body its decision to appeal to the Appellate Body certain issues of law covered in the Panel Report and certain legal interpretations developed by the Panel in the dispute *Brazil – Certain Measures Concerning Taxation and Charges* (WT/DS472). Pursuant to Rule 23(1) of the Working Procedures for Appellate Review, the European Union simultaneously files this Notice of Other Appeal with the Appellate Body Secretariat.

For the reasons to be further elaborated in its submissions to the Appellate Body, the European Union appeals, and requests the Appellate Body to reverse or modify the findings and conclusions of the Panel with respect to the errors of law and legal interpretations contained in the Panel Report described below, and where indicated to complete the analysis on the basis of the Panel's findings and uncontested facts on the record:¹

- 1. The Panel failed to make an objective assessment of the matter before it in accordance with Article 11 of DSU by exercising false judicial economy in relation to the European Union's request to make findings also on the in-house scenario in the context of the ICT and INOVAR-AUTO programmes.² Thus, the European Union requests the Appellate Body *to reverse* the Panel's findings, *to complete the legal analysis* with regard to the in-house scenario and *to find* that the production-step requirements contained in the ICT and INOVAR-AUTO programmes are inconsistent with Articles III: 4 of GATT 1994, Article 2.1 of TRIMS Agreement and Article 3.1(b) of SCM Agreement also under the in-house scenario.
- 2. In the alternative, were the Appellate Body to consider that the Panel correctly exercised judicial economy when failing to make specific findings on the in-house scenario, the European Union requests the Appellate Body *to review*, pursuant to Article 17.6 of DSU, the legal interpretations developed by the Panel and modify, pursuant to Article 17.13 of DSU, the findings contained in paragraphs 7.298-7.314 and 7.746-7.751 of its Report, so as *to make it clearer* that the Panel indeed did not need to rule twice on the production step requirements (in the in-house and in the outsourcing scenarios) because the Panel had already found that those steps are *per se* inconsistent with the covered agreements, since they make an advantage/subsidy contingent upon the use of domestic over imported goods, regardless of whether those domestic goods are manufactured by the accredited company or a third party.³
- 3. Finally, the European Union makes a subordinate claim of error, subject to the Appellate Body rejecting both the first claim of error formulated by the European Union (that the Panel exercised false judicial economy) and the alternative claim of appeal. In that case, the European Union requests the Appellate Body to find that the Panel, by failing to consider the European Union's claims under Article III:4 of GATT 1994, Article 2.1 of TRIMS Agreement and Article 3.1(b) of SCM Agreement in light of all the relevant facts of the case (which included both the in-house and outsourcing scenarios), erred in the application of those provisions.⁴ Accordingly, the European Union requests the Appellate Body *to reverse* the Panel's findings, *to complete the legal analysis* with regard to the in-house scenario and *to find* that the production-step requirements contained in the ICT and INOVAR-AUTO programmes are inconsistent with Articles III:4 of GATT 1994, Article 2.1 of TRIMS Agreement and Article 3.1(b) of SCM Agreement also under the in-house scenario.

^{*} This document, dated 3 October 2017, was circulated to Members as document WT/DS472/9.

¹ Pursuant to Rule 23(2)(c)(ii)(C) of the Working Procedures for Appellate Review this Notice of Other Appeal includes an indicative list of the paragraphs of the Panel Report containing the alleged errors, without prejudice to the ability of the European Union to refer to other paragraphs of the Panel Report in the context of its appeal.

² Panel Report, paras. 6.10, 6.11 and 6.13, 7.314, 7.749, and 8.5.

³ See Panel Report, paras. 6.11, final sentence, 7.292, footnote 683, 7.319, penultimate sentence, 7.500 and 8.5.

⁴ Panel Report, paras. 6.10, 6.11 and 6.13, 7.314, 7.749, and 8.5.

- 12 -

ANNEX A-3

JAPAN'S NOTICE OF OTHER APPEAL*

Pursuant to Article 16.4 and Article 17 of the Understanding on Rules and Procedures Governing the Settlement of Disputes ("DSU") and Rule 23 of the Working Procedures for Appellate Review, Japan hereby notifies its decision to appeal to the Appellate Body certain issues of law covered in the Panel Report in Brazil – Certain Measures Concerning Taxation and Charges (WT/DS497/R) and certain legal interpretations developed by the Panel in this dispute.

In particular, Japan seeks review by the Appellate Body of the Panel's findings that it did not need to review the WTO-consistency of the challenged INOVAR-AUTO and ICT programmes under an "inhouse" scenario. The Panel's findings to this effect are in error and are based on an erroneous interpretation and application of Article 11 of the DSU, by erroneously exercising judicial economy not to extend findings to the so-called "in-house" scenario. Alternatively, the Panel failed to provide "coherent reasoning as required under Article 11 of the DSU. The paragraphs relating to these errors include paragraphs 6.11, 7.302, 7.308, 7.314 and 7.749 of the Panel Report.

Japan respectfully requests the Appellate Body to reverse and modify the related findings, conclusions and recommendations of the Panel. Moreover, to the extent that it reverses the Panel's findings in this respect, Japan also requests that the Appellate Body complete the legal analysis by extending the Panel's findings regarding the "outsourcing" scenario to the "in-house" scenario, and finding that the accreditation and production step requirements under the INOVAR-AUTO programme and ICT programmes "as a whole" – i.e., under both the "outsourcing" and "in-house" scenarios – are inconsistent with Article III:4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement.

In the alternative, should the Appellate Body consider that the Panel did not exercise judicial economy, Japan respectfully requests the Appellate Body to review, pursuant to Article 17.6 of DSU, the legal interpretations developed by the Panel and modify, pursuant to Article 17.13 of the DSU, the findings contained in paragraphs 7.298-7.314 and 7.746-7.751 of the Report, to the extent that they may be understood as referring solely to the "outsourcing" scenario, and clarify that the Panel's findings of the WTO-inconsistency of the challenged INOVAR-AUTO programme and ICT programmes are indeed not limited to the "outsourcing" scenario, but that the Panel made a finding with respect to the accreditation requirement under INOVAR-AUTO programme as a whole, as well as the production-step requirements of the ICT programmes as a whole. To this end, Japan also requests that the Appellate Body complete the analysis, where appropriate.

^{*} This document, dated 3 October 2017, was circulated to Members as document WT/DS497/7.

- 13 -

ANNEX B

ARGUMENTS OF THE PARTICIPANTS

	Contents	Page
Annex B-1	Executive summary of Brazil's appellant's submission	14
Annex B-2	Executive summary of the European Union's other appellant's submission	23
Annex B-3	Executive summary of Japan's other appellant's submission	24
Annex B-4	Executive summary of Japan's appellee's submission	26
Annex B-5	Executive summary of the European Union's appellee's submission	30
Annex B-6	Executive summary of Brazil's appellee's submission	36

- 14 -

ANNEX B-1

EXECUTIVE SUMMARY OF BRAZIL'S APPELLANT'S SUBMISSION*

I. Introduction

1. With the present appeal, Brazil requests that the Appellate Body review and reverse a number of issues of law and legal interpretation contained in Panel Report. Beyond the specific legal provisions under discussion, this appeal concerns more broadly the ability of WTO Members to provide subsidies to domestic producers.

2. Central to the resolution of this dispute is a proper understanding of the role that Article III:8(b) of the GATT 1994 plays in delineating the product-related disciplines of Article III from the subsidy-related disciplines of the SCM Agreement. The Appellate Body recently observed in US - Tax Incentives that "Article III:8(b) of the GATT 1994 exempts from the national treatment obligation in Article III 'the payment of subsidies exclusively to domestic producers'."¹ In diametric opposition to this conclusion, the Panel in the present dispute found that "subsidies that are provided exclusively to domestic producers pursuant to Article III:8(b) of the GATT 1994."²

3. In effect, the Panel interpreted Article III:8(b) of the GATT 1994 as serving only one purpose: to "make[] explicit" that the non-discrimination provisions of Article III do "not require subsidization of foreign producers in tandem with domestic producers".³ While Article III:8(b) does serve that rather axiomatic purpose, Article III:8(b) also makes clear that subsidies provided to domestic producers, including the effects that those subsidies have on the conditions of competition for domestic versus imported products, are not a basis for finding a violation of the national treatment obligations in Article III.

4. Moreover, and contrary to at least one implication of the Panel's analysis, it is immaterial for purposes of Article III:8(b) what type of subsidy a Member chooses to provide exclusively to domestic producers. Article III:8(b) of the GATT 1994 does not distinguish among different types of producer subsidies. The exemption or remission of indirect taxes—the type of subsidies at issue in the present dispute—is undoubtedly a "subsidy" within the meaning of Article 1 of the SCM Agreement. In addition, the Appellate Body has previously found that the ordinary meaning of the term "payment" encompasses revenue foregone, i.e. the exemption or remission of a tax or other revenue owed to the government. The "payment of subsidies exclusively to domestic producers" under Article III:8(b) therefore includes the exemption or remission of indirect taxes.

5. A proper application of Article III:8(b) of the GATT 1994 would have led the Panel to the conclusion that the measures at issue are producer subsidy measures, not measures that discriminate against imported products on the basis of their origin. The challenged measures provide for the exemption or remission of taxes based on whether a producer carries out certain production steps in Brazil. A producer that manufactures a covered product using 100% imported components will obtain the subsidy if it carries out the required production steps in Brazil, while a producer that manufactures the same product using 100% domestic components will *not* obtain the subsidy if it does not carry out the required production steps in Brazil.

6. The Panel committed the same basic errors of interpretation and application in its assessment of the complainants' claims under Article 3.1(b) of the SCM Agreement pertaining to the ICT programmes. Unable to identify any respect in which the ICT programmes condition the provision of a subsidy upon the use of domestic over imported goods, either in the text of the measures at issue or by necessary implication thereof, the Panel based its finding of prohibited import substitution entirely on the effect that the subsidies at issue might have on producers' sourcing decisions. The Appellate Body expressly rejected this line of reasoning in *US - Tax Incentives*, ruling that the fact

^{*} Total word count: 5542

¹ Appellate Body Report, US - Tax Incentives, para. 5.16.

² Panel Report, para. 7.87.

³ Ibid. para. 7.79.

that a domestic production subsidy might affect input sourcing decisions is not a sufficient basis to conclude that it is contingent upon the use of domestic over imported goods.

7. A proper delineation of the national treatment obligations of Article III of the GATT 1994 and the subsidy disciplines of the SCM Agreement is essential to preserving the rights that the covered agreements reserve for Members to subsidize domestic producers. The Panel Report in the present dispute, if adopted, would impermissibly constrain the right of Members to subsidize domestic production. It would, moreover, result in arbitrary distinctions among the *means* by which Members subsidize domestic production more commonly practiced by developed Members (such as outright grants) over methods of subsidization that are more readily available to developing Members (such as, in this case, the exemption or remission of indirect taxes based on whether certain production steps take place within the country).

8. These are not the only important systemic issues presented by this appeal. Another example relates to the Panel's finding that the ICT programmes and the PEC and RECAP programmes result in "revenue foregone" within the meaning of the SCM Agreement, based entirely on the flawed notion that "implicit interest income" is tax revenue that is "due" to a government when a government decides, for reasons of tax administration, either to collect a tax at a later point in time or to forgo the collection of a tax that the government would later need to refund. This interpretation, if accepted, would draw the covered agreements into the realm of tax administration in ways that would have far-reaching consequences for WTO Members.

II. The Panel Erred in Its Interpretation and Application of Article III:8(b) of the GATT 1994 to the ICT programmes and to INOVAR-AUTO and Its Relationship with the Other Provisions of Article III of the GATT 1994

9. The Panel erred in its interpretation and application of Article III:8(b) of the GATT 1994 to the ICT programmes and to INOVAR-AUTO by failing to determine, at the threshold, that these measures are producer subsidy measures subject to the disciplines of the SCM Agreement, rather than to the product-related disciplines of Article III of the GATT 1994.

10. Article III:8(b) makes clear that subsidies provided to domestic producers, *including* the effects that those subsidies might have on the conditions of competition for domestic versus imported *products*, are not a basis for finding a violation of the product-related disciplines of Article III. Production subsidies and their effects are to be analysed under the SCM Agreement, not under Article III of the GATT 1994.

11. In contrast, the concern with origin-based discrimination is reflected in Articles III: 2 and III: 4 of the GATT 1994. In both cases, the discipline applies to *products*, and guards against discriminatory treatment of imported products ("**products of the territory of any contracting party imported** ...") relative to products of national origin ("domestic products", "products of national origin"). Articles III: 2 and III: 4 do not apply, by force of Article III: 8(b), to differences in treatment that may result from payment of subsidies exclusively to domestic <u>producers</u>.

12. The Panel conflated the distinction between the product-related disciplines of Article III and the provisions of the covered agreements concerning subsidies provided to domestic producers, which should be assessed under the SCM Agreement. A harmonious interpretation of these different provisions requires any panel evaluating claims under Article III to determine, in the first instance, whether the measure at issue is a product-related measure subject to the disciplines of Article III or, conversely, whether that measure provides a subsidy to domestic producers under Article III:8(b), and is therefore subject to the disciplines of the SCM Agreement.

13. The Panel did not undertake this threshold inquiry. It simply assumed that, because the challenged programmes resulted in imported products not having access to the subsidy provided therein to domestic producers, the subsidy would not be exempted from the national treatment obligations of Article III and, therefore, would automatically constitute a violation of Articles III:2 and III:4 of the GATT 1994 and of Article 2.1 of TRIMS.

14. Accordingly, the Appellate Body should *reverse* the Panel's interpretative finding that measures providing subsidies exclusively to domestic producers "are not *per se* exempted from the

- 16 -

disciplines of Article III of the GATT 1994"⁴, and to *reverse* its ultimate conclusion that, because the ICT programmes and INOVAR-AUTO are not available to foreign *producers*, they are necessarily inconsistent with Articles III: 2 and III: 4 of the GATT 1994, and Article 2.1 of the TRIMs Agreement.⁵

III. The Panel Erred in Finding that the ICT programmes Are Inconsistent with Article III:2, First Sentence, of the GATT 1994

15. The Panel erred in finding that the ICT programmes result in imported finished and intermediate products being subject to taxation in excess of domestic like products, within the meaning of Article III: 2, first sentence, of the GATT 1994, in three distinct respects.

16. First, the Panel erred in failing to establish, at the threshold, that the ICT programmes constitute the "payment of subsidies to domestic producers" under Article III:8(b) of the GATT 1994 and therefore are not subject to the disciplines of Article III:2 of the GATT 1994. In ruling that Article III:8(b) authorized the provision of domestic production subsidies "so long as they do not have any component that introduces discrimination between imported and domestic products"⁶, the Panel effectively held that such subsidies can *never* take the form of reductions or exemptions of indirect taxes. However, the terms "subsidies" in Article III:8(b) encompasses all types of subsidies listed in Article 1.1 of the SCM Agreement, including foregoing of government revenue otherwise due under Article 1.1(a)(1)(ii) of the SCM Agreement, in the form of the exemption or remission of indirect taxes unquestionably constitute a "payment" of subsidies under Article III:8(b), and the negotiating history referred to the Panel is of little interpretative value considering that it precedes the definition of subsidies contained in the SCM Agreement.

17. In addition, the Panel failed to recognize that the list of measures in Article III:8(b) is merely illustrative, and does not exhaustively define the type of permissible domestic production subsidies. The Panel also erred in finding that Articles III:8(b) and Article III:2 have overlapping scopes of application. Contrary to the Panel's finding to this effect, the Appellate Body expressly held in *US* – *Tax Incentives* that Article III:8(b) *exempts* domestic production subsidies from the product disciplines of Article III.

18. Second, the Panel erred in finding that the complainants had established that the ICT programmes result in imported *finished* products being subject to taxes in excess of domestic like products. Setting aside the Panel's failure to determine the threshold applicability of Article III:8(b) to the ICT programmes, it was still incumbent upon the Panel to hold the complainants to their burden of proof of establishing that the ICT programmes result in *finished* imported products being subject to taxation in excess of domestic like products. The Panel simply assumed that, because foreign producers are ineligible for the subsidies, the ICT programmes *de jure* discriminated against imported products. However, because importers are also subject to the application of the debit/credit rule under Brazil's non-cumulative tax system, the payment of taxes on purchases of non-incentivized products generates a corresponding credit that may be offset against the importer's liability. Accordingly, the fact that foreign producers are ineligible for the tax incentives does not establish, without more, that imported finished products are taxed in excess of domestic like products.

19. Third and finally, the Panel erred in finding that the ICT programmes result in imported *intermediate* ICT products being taxed in excess of domestic like products by virtue of "cash flow" and "time value of money" effects allegedly generated on the purchase of incentivized products. At most, such unsubstantiated effects would benefit *producers*, and the complainants did not produce, and the Panel did not examine, any evidence tending to demonstrate that these effects resulted in higher taxation of imported intermediate *products*. Under Brazil's non-cumulative tax system, the relevant taxes are not paid upfront but are rather levied each month pursuant to the credits and debits recorded in the taxpayer's accounting books. As a general rule all credits are offset against debits in the same taxation period, and are neither tax nor product-specific. Thus the purchase of

⁴ Panel Report, para. 7.87.

⁵ Panel Report, paras. paragraphs 7.174, 7.318, 7.319, 7.365, 7.688, 7.772, 7.773, 7.806, 8.5(a), 8.5(b), 8.5(d), 8.6(a), 8.6(b) and 8.6(d).

⁶ Panel Report, para. 7.83.

- 17 -

non-incentivized products neither reduces the availability of cash flow by the purchaser, nor does the value of the tax credit diminishes over time.

20. For these reasons, the Appellate Body should *reverse* the Panel's finding that the ICT programmes result in taxation of imported like products in excess of domestic like products, inconsistently with Article III:2, first sentence, of the GATT 1994.⁷

IV. The Panel Erred in Finding that the ICT Programmes Accord to Imported Products Treatment Less Favourable than that Accorded to Like Domestic Products, Inconsistent with Article III:4 of the GATT 1994 and Article 2.1 of the TRIMs Agreement

21. Since the Panel erred in failing to determine the threshold applicability of Article III:8(b) of the GATT 1994 to the ICT programmes, it misconstrued the relationship of that provision with Article III:4 of the GATT 1994. Consistent with the analytical framework adopted by the Appellate Body in US - Tax Incentives and by previous panels, domestic production subsidies are *exempt* from the product disciplines of Article III:4 of the GATT 1994. Under the Panel's erroneous interpretation, it is unclear under what circumstances a WTO Member would *ever* be able to provide subsidies that do not distort the conditions of competition in the marketplace to the detriment of imported products. Quite the contrary, as the Panel itself recognized, the very purpose of providing a domestic production subsidy is to provide a competitive advantage exclusively to domestic producers.

22. For this reason, the Panel erred in finding that accreditation requirements under the ICT programmes are inconsistent with Article III: 4 because they result in foreign producers not being eligible to the subsidies. A corollary of Article III:8(b) is that only producers that produce domestically will benefit from the subsidies. The accreditation requirements merely define the producers that are eligible to the subsidies (i.e. those that undertake certain production steps in Brazil), and in and of itself do not reflect any regulatory discrimination between imported and domestic products. In this respect, the Panel impermissibly conflated tax discrimination under Article III:2 with regulatory discrimination under Article III:4, insofar as purported discrimination really stems from the reduction or exemption of indirect taxes for eligible producers. Moreover, because the accreditation requirements apply at the *producer* level pursuant to the authorization contained in Article III:8(b), by definition they cannot be implicated by the product disciplines of Article III:4

23. The Panel's finding that the ICT programmes are inconsistent with Article III:4 because imported intermediate products are purportedly subject to more onerous administrative burdens is also in error, and reflects a misapprehension of Brazil's non-cumulative tax system. Regardless of whether they purchase incentivized or non-incentivized products, all companies in Brazil are required to keep and operate a ledger in which all purchases of inputs and sales of final goods are recorded, and debits and credits are offset. The suspension, exemption, or reduction of indirect taxes under the ICT programmes do not alter in any way the administrative burden incurred by companies to assess their monthly tax liabilities, irrespective of their sourcing decisions.

24. Finally, the Panel erred in finding that the PPBs and other production-step requirements are tantamount to a requirement to use domestic goods within the meaning of Article III: 4 of the GATT 1994. The PPBs and other production step requirements are entirely silent as to the origin of the inputs used in the production process. A producer using 100% of imported components will be eligible for the subsidies, provided that it undertakes the relevant production steps in Brazil. The Panel's rationale that production step requirements are tantamount to a requirement to use domestic goods in respect of a producer's component or sub-assemblies sourcing decisions has been expressly addressed – and rejected – by the Appellate Body in *US – Tax Incentives*. In that dispute, the Appellate Body expressly held that domestic goods in the relevant market, thereby increasing the use of these goods downstream and adversely affecting imports, without necessarily requiring the use of domestic over imported goods as a condition for granting the subsidy."⁸ For this reason, the "siting criterion" imposed by the production step requirements under the ICT programmes, even when it involves the outsourcing of certain steps, does not constitute a requirement to use domestic

⁷ Panel Report, paras. 7.174, 8.5(a), and 8.16(a).

⁸ Appellate Body Report, US – Tax Incentives, para. 5.15.

- 18 -

products to the detriment of imported products in a manner inconsistent with Article III:4 of the GATT 1994.

25. Accordingly, the Appellate Body should *reverse* the Panel's finding that in respect of (i) accreditation requirements; (ii) purported lower administrative burdens on purchases of domestic intermediate products; and (iii) PPBs and other production step requirements, the ICT programmes accord to imported products treatment less favourable than that accorded to domestic like products, inconsistently with Article III: 4 of the GATT 1994 and Article 2.1 of the TRIMs Agreement.⁹

V. The Panel Erred in Finding that the ICT programmes Provide Subsidies that Are Contingent upon the Use of Domestic over Imported Goods under Article 3.1(b) of the SCM Agreement

26. The Panel erred in finding that the ICT programmes provide prohibited import substitution subsidies under Article 3.1(b) of the SCM Agreement in two respects.

27. First, the Panel erred in finding that the exemptions, suspensions and reductions of taxes provided under the ICT programmes on the sale of intermediate goods and on the purchase of raw materials, intermediate goods, packaging materials, inputs, capital goods and computational tools (collectively, "inputs") provide a subsidy under Article 1.1 of the SCM Agreement. More specifically, the Panel erred in finding that the ICT programmes provide a financial contribution in the form of government revenue that is otherwise due and is foregone and not collected under Article 1.1(a)(1)(ii) of the SCM Agreement. By focusing on the exceptional scenario in which the taxpayer is unable to fully offset the total amount of taxes paid in the same taxation period, the Panel impermissibly failed to compare the treatment of the group of taxpayers in the benchmark treatment with the treatment of taxpayers under the ICT programmes. Moreover, the "implicit interest income" that the Brazilian government allegedly foregoes under the ICT programmes as compared to the situation of that subset of taxpayers is not a "government revenue" that is "due" within the meaning of Article 1.1(a)(1)(ii) of the SCM Agreement.

28. Second, the Panel erred in finding that the ICT programmes are contingent upon the use of domestic over imported goods under Article 3.1(b) of the SCM Agreement because the PPBs and other production step requirements imply a contingency upon production in Brazil. As the Appellate Body held in *US* – *Tax Incentives*, "*something more than mere subsidization of domestic production* is required for finding an import substitution subsidy."¹⁰ The contingencies set out in the PPBs and similar production step requirements relate to the location of certain manufacturing operations in Brazil, and do not prevent the possibility of using imported inputs in the manufacturing process.

29. The Panel reasoned that because PPBs provide that outsourced components must be produced domestically, such condition imposes, in respect of the accredited producer of the incentivized finished product, a requirement to use domestic over imported goods. This line of argument was expressly addressed and rejected by the Appellate Body in US - Tax Incentives, where the Appellate Body categorically held that, in situations where both an input and the finished product are subsidized, the consequences of a requirement to produce an input domestically in respect of the producer's input-sourcing decisions, and the fact that the producer is likely to use a domestic product in its downstream production activities is insufficient, without more, to establish a condition requiring the use of domestic over imported goods under Article 3.1(b) of the SCM Agreement.¹¹

30. For these reasons, the Appellate Body should *reverse* the Panel's finding that the exemptions, reductions, and suspensions of taxes granted under the ICT programmes are subsidies contingent upon the use of domestic over imported goods within the meaning of Articles 3.1(b) and 3.2 of the SCM Agreement.¹²

⁹ Panel Report, paras. 7.318, 7.319, 7.365, 8.5(b), 8.5(d), 8.16(c) and 8.16(e).

¹⁰ Appellate Body Report, US – Tax Incentives, para. 5.16. (emphasis added)

¹¹ See Appellate Body Report, US - Tax Incentives, para. 5.49; see also ibid. para. 5.76.

¹² Panel Report, paras. 7.500, 8.5(e) and 8.16(f).

- 19 -

VI. The Panel Erred in Finding that Certain Aspects of the Accreditation Requirements under INOVAR-AUTO Related to Investments in R&D, Expenditures on Engineering, Basic Industrial Technology and to the Performance of Certain Manufacturing Steps in Brazil Accord Less Favourable Treatment to Imported Products than that Accorded to Like Domestic Products, Inconsistently with Article III:4 of the GATT 1994 and Article 2.1 of the TRIMs Agreement

31. Similar to the Panel's finding in respect of the ICT programmes, the Panel's finding that certain accreditation requirements under INOVAR-AUTO are inconsistent with Article III: 4 of the GATT 1994 is based on the incorrect assumption that Article III:8(b) does not exempt domestic production subsidies from the product disciplines of Article III. In so finding, the Panel failed to recognize that Articles III:8(b) and III:4 have distinct scopes of application. While the former concerns producer subsidy measures, the latter concern regulatory discrimination as between imported and domestic *products*. Consequently, any adverse effects that domestic production subsidies may have on the competitive opportunities for imported products should fall within the purview of Part III of the SCM Agreement.

32. The Panel's failure to determine, at the threshold, that INOVAR-AUTO constituted a domestic production subsidy under Article III:8(b) effectively converted permissible domestic production subsidies into *per se* violations of Article III:4 of the GATT 1994. Had the Panel properly interpreted and applied Articles III:8(b) and III:4 of the GATT 1944 to INOVAR-AUTO, it would have concluded that the conditioning of domestic production subsidies to the performance of certain production steps in Brazil is insufficient, without more, to establish regulatory discrimination under Article III:4 of the GATT 1994.

33. Moreover, the requirement to perform certain manufacturing steps in Brazil and to invest in R&D and/or engineering in Brazil in order to benefit from the tax incentives established in the INOVAR-AUTO programme are imposed on *producers*, and not on the products themselves. As such, these requirements cannot be deemed in violation of the product disciplines of Article III: 4 of the GATT 1994.

34. The Panel also erred in finding that different accreditation requirements for domestic manufacturers vis-à-vis the categories of "investors" or "importer/distributors" entail a more burdensome requirement in respect of imported products. The requirement that domestic producers undertake certain manufacturing or investment activities in Brazil is a natural corollary of Brazil's right under Article III:8(b) of the GATT to pay subsidies exclusively to domestic producers. The Panel impermissibly undertook a quantitative assessment of the INOVAR-AUTO requirements, rather than a qualitative assessment. Had the Panel adopted a qualitative approach instead, it would have recognized that the differences set forth in the INOVAR-AUTO programme with respect to "companies that manufacture vehicles in Brazil" and "companies that do not manufacture but market vehicles in Brazil" are not discriminatory because the differences are inherent to the level of contribution made by both manufacturing and distributing companies to the programme's objectives.

35. For these reasons, the Appellate Body should *reverse* the Panel's finding that the accreditation requirements under INOVAR-AUTO to: (a) perform a minimum number of manufacturing steps in Brazil; and (b) invest in R&D in Brazil and make expenditures on engineering, basic industrial technology and capacity-building of suppliers in Brazil accord less favourable treatment to imported products than that accorded to like domestic products, inconsistently with Article III: 4 of the GATT 1994 and Article 2.1 of the TRIMs Agreement.¹³

36. Moreover, to the extent that the Panel concluded that domestic production requirements alone are sufficient to establish that INOVAR-AUTO is "contingent upon the use of domestic over imported goods" within the meaning of Article 3.1(b) of the SCM Agreement, Brazil further requests that the Appellate Body *reverse* the Panel's finding that INOVAR-AUTO is a prohibited import substitution subsidies under Article 3.1(b) and 3.2 of the SCM Agreement.¹⁴

¹³ Panel Report, paras. 7.773, 7.806, 8.6(b), 8.6(d), 8.17(c) and 8.17(e).

¹⁴ Panel Report, paras. 7.751, 7.823, 7.847, 8.6(e), and 8.17(f).

- 20 -

VII. The Panel Erred in Finding that the Tax Suspensions under the PEC and RECAP Programmes Are Subsidies Contingent Upon Export Performance Under Article 3.1(a) of the SCM Agreement

37. The Panel erred in finding that PEC and RECAP provide subsidies in the form of government revenue otherwise due that is foregone or not collected under Article 1.1(a)(1)(ii) of the SCM Agreement, in three different respects.

38. First, the Panel erred in failing to use the treatment of structural credit accumulators as the benchmark treatment for comparison with the tax treatment under the PEC and RECAP programmes. The Panel erroneously sought to ascertain whether a "general rule" of taxation existed under Brazilian law pursuant to which *all* suspensions and exemptions of taxes on inputs and capital goods is directly linked to the problem of structural credit accumulation, rather than ascertain the "comparably situated taxpayers" that make up the relevant normative benchmark for comparison. Had the Panel applied a proper analytical framework, it would have concluded that all structural credit accumulators are comparably situated, as it is evident from the relevant statute. The fact that sectors whose products were formerly subject to high taxation may also be eligible for tax suspensions confirms rather than disproves the existence of this normative benchmark, insofar as companies in those sectors became structural tax credit accumulators by virtue of exemptions or suspensions granted in other programmes.

39. Second, the Panel erred in failing to compare the tax treatment applicable to the *group* of taxpayers subject to the benchmark treatment with the tax treatment of the *group* of taxpayers under PEC and RECAP. The Panel arbitrarily distinguished between taxpayers subject to the benchmark treatment, disregarding the overwhelming majority of taxpayers that are able to fully offset their tax credits in the same taxation period, focusing instead on the small subset of taxpayers who are unable to do so. In doing so, the Panel found the foregoing of revenue where there is none.

40. Third, the Panel erred in finding that alleged "implicit interest income" that the Brazilian government purportedly "foregoes" under PEC and RECAP constitutes "government revenue" that is otherwise "due" and foregone or not collected under Article 1.1(a)(1)(ii) of the SCM Agreement. The terms "government revenue" that is "due" refer to government income that is owed or payable to the government as an obligation or debt by the taxpayer (such as taxes or charges), and does not encompass any "implicit interest" that the Brazilian Government supposedly "foregoes" as a result of tax administration measures which do not alter the amount of taxes due.

41. Should the Appellate Body consider nonetheless that PEC and RECAP provide a subsidy within the meaning of Article 1.1 of the SCM Agreement, Brazil then claims that the Panel acted inconsistently with Article 11 of the DSU in its assessment of the evidence on export contingency, and therefore erred in finding that the PEC and RECAP programmes are contingent upon export performance within the meaning of Article 3.1(a) of the SCM Agreement.

42. On the basis of the foregoing, the Appellate Body should *reverse* the Panel's finding that PEC and RECAP are subsidies contingent upon export performance under Articles 3.1(a) and 3.2 of the SCM Agreement.¹⁵

VIII. The Panel Erroneously Found that the Tax Reduction Accorded to Imports from Argentina, Mexico and Uruguay under INOVAR-AUTO Is Not Justified under Paragraphs 2(b) or 2(c) of the Enabling Clause

43. The Panel erred in finding that the tax reduction accorded to imports from MERCOSUR countries under INOVAR-AUTO are inconsistent with Article I:1 of the GATT 1994 and not justified under paragraphs 2(b) or 2(c) of the Enabling Clause.

44. The Panel preliminarily erred in finding that the complainants' claim under Article I:1 of the GATT 1994 was properly within its terms of reference. The measures had unquestionably been notified under paragraph 4(a) of the Enabling Clause, and contrary to what the Panel found, any

¹⁵ Panel Report, paras. 7.1211, 7.1212, 7.1238, 8.7 and 8.18.

substantive disagreement as to the proper characterization of the measures did not alleviate the complainants' burden of invoking the Enabling Clause in their panel requests.

45. The Panel also erred in failing to examine whether the notification of the Treaty of Montevideo and the ECAs under paragraph 2(c) could substantively serve as a notification of the specific differential and more favourable treatment sought to be justified under paragraph 2(b) of the Enabling Clause. The Panel adopted an impermissibly restrictive interpretation of the Enabling Clause, failing to recognize that paragraph 4(a) does not impose on Members any requirement to precisely identify the basis pursuant to which it notifies its measures. This suggests that the drafters of the Enabling Clause provided some flexibility in such notification provisions.

46. The Panel also erred in limiting the scope of paragraph 2(b) of the Enabling Clause to the S&D provisions in the covered agreements other than the GATT itself. The GATT 1994 is unquestionably an "instrument multilaterally negotiated under the auspices of the GATT" within the meaning of paragraph 2(b). The Panel's restrictive interpretation to the contrary impermissibly renders the provision effectively *inutile*, insofar as the S&D provisions provided for in the agreements in Annex 1A of the Marrakesh Agreement are mostly hortatory and impose no substantive obligations.

47. Finally, the Panel impermissibly conflated the notification requirements with the substantive requirements imposed under paragraph 2(c) of the Enabling Clause. The challenged treatment had been duly notified under paragraph 2(c). The Treaty of Montevideo, the ECAs, and the LAIA clearly comprise internal taxation measures, and as the dissenting Panelist correctly observed, refer to both "regional integration" and to tariff preferences in the automotive sector. Internal taxation measures are unquestionably "non-tariff measures" within the meaning of paragraph 2(c) of the Enabling Clause, and the Panel erred in failing to undertake a substantive analysis of whether the challenged treatment satisfied the requirements of that provision.

48. Accordingly, the Appellate Body should *reverse* the Panel's conclusion that the complainants did not have an obligation to invoke the Enabling Clause in their panel requests, and *reverse* the Panel's consequential finding that the complainants' claims under Article I: 1 of the GATT 1994 were within the Panel's terms of reference.¹⁶

49. The Appellate Body should also *reverse* the Panel's finding that the differential tax treatment at issue was not properly notified under paragraph 4(a) of the Enabling Clause; *reverse* the Panel's finding that the tax treatment at issue is not justified under paragraphs 2(b) or 2(c) of the Enabling Clause; *complete the legal analysis* and find that the differential tax treatment is justified under paragraphs 2(b) or 2(c) and complete with paragraph 3 of the Enabling Clause.¹⁷

IX. The Panel Acted Inconsistently with Articles 11 and 12.7 of the DSU in Recommending that Brazil Withdraw the Alleged Prohibited Subsidies in 90 Days

50. Should the Appellate Body uphold the Panel's finding that the measures at issue are prohibited subsidies under Article 3.1(a) and 3.1(b) of the SCM Agreement, Brazil then claims, in the alternative, that the Panel acted inconsistently with its duties under Articles 11 and 12.7 of the DSU in recommending pursuant to Article 4.7 of the SCM Agreement that Brazil withdraws these measures in 90 days.

51. The fact that the Appellate Body has said that "the expedited remedy provided under Article 4.7 of the SCM Agreement is an 'important consideration'"¹⁸ does not constitute a "reasoned and adequate explanation" for the Panel's 90-day recommendation, because it does not explain why the Panel believed that a 90-day timeframe was appropriate. In order for the Panel to provide a "reasoned and adequate explanation" or "basic rationale" for its conclusion, the Panel needed to explain why 90 days was appropriate in light of the nature of the measures and the procedures that may be required to implement [the panel's] recommendation. The Panel's failure to provide such an explanation cannot be reconciled with the Panel's obligations under Articles 11 and 12.7 of the DSU.

¹⁶ Panel Report, paras. 7.1083, 7.1120, 8.6(h), and 8.17(i).

¹⁷ Panel Report, paras. 7.1079, 7.1081, 7.1083, 7.1096, 7.115, 7.119, 8.6(i) and 8.17(j).

¹⁸ Panel Report, para. 6.17 (quoting Appellate Body Report, *Canada – Renewable Energy / Canada – Feed-in Tariff Program*, para. 5.7) (referring to Appellate Body Report, *EC – Export Subsidies on Sugar*, paras. 332-335).

52. Brazil therefore requests that the Appellate Body *reverse* the Panel's recommendation that Brazil withdraw the measures within 90 days¹⁹, *complete the legal analysis*, and recommend, in light of the nature of the measures at issue and the length of the legislative process that will be required, withdrawal within 18 months.

¹⁹ Panel Report, paras. 8.11 and 8.22.

- 23 -

ANNEX B-2

EXECUTIVE SUMMARY OF THE EUROPEAN UNION'S OTHER APPELLANT'S SUBMISSION¹

1. The European Union brings this other appeal in the interest of clarifying the scope of Brazil's implementing obligations in this dispute. The Panel's findings lack clarity as to Brazil's obligations under Articles III:4 of GATT 1994, Article 2.1 of TRIMS Agreement and Article 3.1(b) of SCM Agreement with regard to the in-house scenario.

2. First, the European Union submits that the Panel failed to make an objective assessment of the matter before it in accordance with Article 11 of DSU by exercising false judicial economy in relation to the European Union's request to make findings also on the in-house scenario in the context of the ICT and INOVAR-AUTO programmes. Thus, the European Union requests the Appellate Body *to reverse* the Panel's findings, *to complete the legal analysis* with regard to the in-house scenario and *to find* that the production-step requirements contained in the ICT and INOVAR-AUTO programmes are inconsistent with Articles III: 4 of GATT 1994, Article 2.1 of TRIMS Agreement and Article 3.1(b) of SCM Agreement also under the in-house scenario.

3. Second, in the alternative, were the Appellate Body to consider that the Panel correctly exercised judicial economy when failing to make specific findings on the in-house scenario, the European Union requests the Appellate Body *to review*, pursuant to Article 17.6 of DSU, the legal interpretations developed by the Panel and modify, pursuant to Article 17.13 of DSU, the findings contained in paragraphs 7.298-7.314 and 7.746-7.751 of its Report, so as *to make it clearer* that the Panel indeed did not need to rule twice on the production step requirements (in the in-house and in the outsourcing scenarios) because the Panel had already found that those steps are *per se* inconsistent with the covered agreements, since they make an advantage/subsidy contingent upon the use of domestic over imported goods, regardless of whether those domestic goods are manufactured by the accredited company or a third party.

4. Finally, the European Union makes a subordinate claim of error, subject to the Appellate Body rejecting both the first claim of error formulated by the European Union (that the Panel exercised false judicial economy) and the alternative claim of appeal. In that case, the European Union requests the Appellate Body to find that the Panel, by failing to consider the European Union's claims under Article III:4 of GATT 1994, Article 2.1 of TRIMS Agreement and Article 3.1(b) of SCM Agreement in light of all the relevant facts of the case (which included both the in-house and outsourcing scenarios), erred in the application of those provisions. Accordingly, the European Union requests the Appellate Body *to reverse* the Panel's findings, *to complete the legal analysis* with regard to the in-house scenario and *to find* that the production-step requirements contained in the ICT and INOVAR-AUTO programmes are inconsistent with Articles III:4 of GATT 1994, Article 2.1 of TRIMS Agreement and Article 3.1(b) of SCM Agreement and INOVAR-AUTO programmes are inconsistent with Articles III:4 of GATT 1994, Article 2.1 of TRIMS Agreement and Article 3.1(b) of SCM Agreement also under the in-house scenario.

¹ Total number of words (including footnotes but excluding executive summary) = 12 851; total number of words of the executive summary = 511.

- 24 -

ANNEX B-3

EXECUTIVE SUMMARY OF JAPAN'S OTHER APPELLANT'S SUBMISSION¹

1. Japan's Other Appellant Submission is focused on one single issue only where, in Japan's view, the Panel could and should have done more. As the Panel appears to have applied judicial economy with respect to the INOVAR-AUTO programme and ICT programmes, Japan submits that the Panel erred by (1) distinguishing the scenario in which a company undertakes minimum production processes by itself ("in-house"), from that in which the company delegates these activities to a third party ("outsourcing") and (2) refraining from addressing the former, which is inconsistent with Article III: 4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement.

2. First, Panel explicitly finds that the production-step requirements and the accreditation requirements formulated under the relevant legal instruments implementing the challenged programmes do not themselves distinguish between an "in-house" and an "outsourcing" scenario. The Panel draws this distinction merely as an analytical one to facilitate and organize its own analysis.

3. Second, as the Panel itself acknowledged, the complainants challenged the programmes' production-step requirements as a whole WTO-inconsistent, irrespective of which scenario applies. In Japan's view, the Panel's failure to make findings with respect to each accreditation and production step requirements as a whole would lead to a partial resolution of the matter.

4. Third, because the Panel limited its finding to the "outsourcing" scenario, a positive solution to the dispute is not secured. Due to the lack of necessary findings in the Panel report with respect to the "in-house" scenario, the Panel leaves open the possibility of disagreement between the Parties within the meaning of Article 21.5 of the DSU. This will undermine the "effective resolution of disputes to the benefit of all Members" under Article 21.1 of the DSU. Such failure constitutes "false judicial economy" and constitutes a breach of the panel's obligation under Article 11 of the DSU.

5. Therefore, Japan respectfully asks the Appellate Body to reverse the Panel's decision to limit its finding to the "outsourcing" scenario and complete the legal analysis by finding that the accreditation and production step requirements "as a whole," encompassing both the "outsourcing" and "in-house" scenarios, are inconsistent with Article III:4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement.

6. Should the Appellate Body consider the Panel's application of judicial economy to be unwarranted, Japan requests that the Appellate Body complete the legal analysis to find the production step requirements under the challenged programmes, as a whole, to be inconsistent with Article III:4 of the GATT 1994 and Article 3.1(b) of the SCM Agreement. The Panel's findings were sufficient for the Appellate Body to complete the analysis and to make such findings. Moreover, much of the evidence is in fact identical or substantially the same for the "outsourcing" and "inhouse" scenarios. Thus, the Appellate Body can complete the analysis as needed.

7. Japan underscores that the each production step requirement under the INOVAR-AUTO and the ICT programmes extends accreditation to the companies that carry out the production steps in Brazil by themselves, as well as to the companies that outsource to third parties within Brazil. This necessarily implies that the programmes offer subsidies based on the use of domestic products, rather than carrying out a particular production steps. Accordingly, they constitute "a condition requiring" manufacturers to use Brazilian products as a precondition to receiving tax benefits, which are inconsistent with Article 3.1(b) of the SCM Agreement.

8. Alternatively, should the Appellate Body find that the Panel did not exercise judicial economy with respect to the "in-house" scenario, Japan respectfully requests the Appellate Body to find that the Panel's finding of the WTO-inconsistency of the accreditation and the production-step requirements under INOVAR-AUTO and the ICT programme are interpreted to cover the challenged

¹ Other Appellant Submission 7,434 words, Executive Summary 670 words.

- 25 -

two measures as a whole including both the "outsourcing" and "in-house" scenarios and the Panel erred by failing to provide "coherent reasoning" as required under Article 11 of the DSU.

- 26 -

ANNEX B-4

EXECUTIVE SUMMARY OF JAPAN'S APPELLEE'S SUBMISSION¹

I. INTRODUCTION

1. Japan notes that the Panel's findings did *not* put into question Brazil's or any other country's ability to adopt policies to promote development, support domestic research and development (R&D), and/or develop human capital. The Panel found that Brazil's measures fall into the much more narrowly defined categories of measures that are covered by the National Treatment obligations of Article III: 2 and III:4 of the GATT 1994, the prohibitions on export subsidies and import-substitution subsidies in Articles 3.1(a), 3.1(b) and 3.2 of the SCM Agreement, and Article 2.1 of the TRIMS Agreement.

II. THE PANEL DID NOT ERR IN ITS INTERPRETATION AND APPLICATION OF ARTICLE III:8(b) OF THE GATT 1994 TO THE ICT AND INOVAR-AUTO PROGRAMMES

2. Contrary to Brazil's assertion, as the Panel found, the mere fact that a subsidy was provided exclusively to domestic producers is not sufficient for it to be exempt from Articles III:2 and III:4 of the GATT 1994. Brazil's argument is not only legally incorrect but very far-reaching in its potential consequences, which would lead to a wide loophole for countries to circumvent the disciplines under Articles III:2 and III:4 of the GATT 1994.

3. By its ordinary meaning, Article III:8(b) does not constitute a blanket exemption for subsidies paid exclusively to domestic producers. Instead, measures that take the forms of subsidies paid exclusively to domestic producers may still be subject to the disciplines under Article III of the GATT 1994 if they result in discrimination between products. Brazil also appears to take issue with the Panel's interpretation of the term "payment" in Article III:8(b). Its argument, however, is entirely circular.

4. In addition, Brazil claims that while the Panel did not rely on it, Japan and the EU, as co-complainants misinterpreted the Appellate Body's findings in *Canada – Periodicals*. Contrary to Brazil's argument, however, the Appellate Body made clear that a reduction of tax rates is not covered by Article III:8(b). Because INOVAR-AUTO and the ICT programmes both involve a reduction of tax rates, Article III:8(b) does not apply.

III. THE PANEL PROPERLY FOUND THAT THE ICT PROGRAMMES ARE INCONSISTENT WITH ARTICLE III:2, FIRST SENTENCE, OF THE GATT 1994

5. As a threshold argument, Brazil alleges that the tax treatment at issue was not within the scope of Article III:2. Again, Brazil's interpretation of Article III:8(b) of the GATT 1994, and its relationship to the substantive disciplines of Article III, is incorrect.

6. Further, Brazil alleges that the Panel failed to "carefully scrutinize the design, structure and operation of the ICT programmes in applying Article III:2 of the GATT 1994 to the facts of this dispute." The Panel, however, conducted a thorough analysis of the ICT programmes, and correctly found that the tax burden on imported finished products is in excess of that on domestic like products.

7. Additionally, Brazil argument that the Panel erred in finding that the "cash flow" and the "timevalue of money" effects under the ICT programmes result in a higher tax burden on imported intermediate products is flawed. Brazil's domestic manufacturers of intermediate products must pay certain taxes on inputs, which are subsequently offset by revenues from selling downstream products. Thus, during the lag between purchase of inputs and sale of outputs, funds related to the payment of taxes are temporarily unavailable. However, the ICT programmes mitigate the negative financial consequences of this lag, by reducing or eliminating the required taxes otherwise due when

¹ Appellee Submission 27,908 words, Executive Summary 2,202 words.

- 27 -

domestic manufacturers of intermediate goods purchase inputs covered by the ICT programmes, and thus are not consistent with Article III:2.

IV. THE PANEL PROPERLY FOUND THAT THE ICT PROGRAMMES ARE INCONSISTENT WITH ARTICLE III:4 OF THE GATT 1994 AND ARTICLE 2.1 OF THE TRIMS AGREEMENT

8. Brazil's argument once again that Article III:8(b) shields a measure from being found to be WTO-inconsistent is flawed. Japan also disagrees with Brazil's claim that the Panel conflated Articles III:2 and IIII:4 of the GATT 1994. A violation of Article III:4 of the GATT 1994 can be found when an underlying measure is tax.

9. Brazil also disputes the Panel's finding that the requirements to perform a series of production steps under the ICT programmes require the use of domestic products. However, the Panel correctly found such requirements to be inconsistent with Article III: 4 of the GATT 1994 because they "accord to imported products treatment less favourable than that accorded to like domestic products."

V. THE PANEL PROPERLY FOUND THAT CERTAIN ASPECTS OF THE ICT PROGRAMMES ARE INCONSISTENT WITH ARTICLE 3.1(b) OF THE SCM AGREEMENT

10. The Panel applied the correct benchmark in finding that the tax exemptions under the ICT programmes constitute financial contributions within the meaning of Article 1.1(a)(1)(ii) of the SCM Agreement. The Panel determined that, *at least in some factual situations*, accredited companies benefiting from the ICT Programmes are better off than those within the benchmark, and the Brazilian Government, by contrast, is foregoing revenue that would be otherwise due.

11. Brazil also claims that the Panel committed legal error when it found that the "cash availability" and "implicit interest" constituted revenue "due" that could have been forgone within the meaning of Article 1.1(a)(1)(ii) of the SCM Agreement. At the outset, Japan notes that Brazil appears to be asking for a *de novo* review of the facts or to accept certain facts that are simply not supported by the Panel's findings. Further, there is no basis for Brazil's argument that the interest foregone by the Brazilian government is not in fact a form of current or future revenue. Brazil's claim also appears to rest on the premise that it is only in "exceptional" cases but the Panel record, makes clear that such a situation is "rather frequent".

12. Brazil argues that the Panel erred in finding that the production-step requirements under the ICT programmes require the use of domestic over imported goods. However, the production-step requirements under each of the challenged ICT programmes and ultimately concluded that the structure of accreditation requirements demonstrated that the challenged measures "give rise to a contingency on the use of domestic over imported goods in the sense of Article 3.1(b) of the SCM Agreement."

VI. THE PANEL PROPERLY FOUND THAT THE INOVAR-AUTO PROGRAMME ACCORDS TREATMENT LESS FAVOURABLE TO IMPORTS THAN TO LIKE DOMESTIC PRODUCTS, INCONSISTENT WITH ARTICLE III:4 OF THE GATT 1994 AND ARTICLE 2.1 OF THE TRIMS AGREEMENT

13. Brazil's assertions against the Panel's findings on the measure's inconsistency with Article III: 4 of the GATT 1994 are premised on the flawed claim that Article III: 8 shields a measure from being found to be WTO-inconsistent, so long as the challenged measure is in the form of a subsidy provided exclusively to domestic producers.

14. Brazil challenges what it calls the Panel's "quantitative" analysis and appears to request the Appellate Body to engage in a weighing and balancing of the facts before the Panel to determine the burden that INOVAR-AUTO's requirements imposed on foreign manufactures. This appears to be a request for *de novo* review of the facts and thus the Appellate Body should reject Brazil's argument.

15. The Panel consider[ed] that foreign companies seeking accreditation under the INOVAR-AUTO programme must go through a more burdensome accreditation process than domestic manufacturers." Additionally, the level of taxation applied to imported motor vehicles, sold by foreign companies that are not accredited, exceed that applied to domestic vehicles sold by accredited

- 28 -

domestic manufacturers. Japan underscores that the conditions for accreditation accord less favourable treatment to foreign *products* imported into Brazil than that accorded to like *products* of Brazilian origin.

VII. THE PANEL PROPERLY FOUND THAT THE TAX SUSPENSIONS GRANTED UNDER THE PEC AND RECAP PROGRAMME ARE CONTINGENT UPON EXPORT PERFORMANCE,

16. According to Brazil, the Panel incorrectly applied Article 1.1(a)(1)(ii) of the SCM Agreement because it failed to identify the treatment of structural accumulators as the proper benchmark treatment with respect to PEC and RECAP programmes.

17. In conducting its benchmark analysis, contrary to Brazil's assertion, the Panel did not attempt to identify a "general rule" of taxation under Brazilian law pursuant to which *all* suspensions and exemptions of taxes on inputs and capital goods *relate to the problem of structural credit accumulation*, but rather, responded to Brazil's assertion about the "rule". Based on its analysis, the Panel correctly found that there was no basis for defining the benchmark as narrowly as Brazil proposed.

18. As an alternative argument, Brazil asserts that even if the Appellate Body finds that nonaccredited companies are the correct benchmark, the Panel incorrectly "opted to compare the treatment under PEC and RECAP with what it erroneously termed the "rather frequent" scenario in which non-accredited companies are unable to fully offset their tax credits in the same taxation period, and have to formally request the reimbursement of taxes paid." Japan has already explained why Brazil's argument is flawed in the analogous in the context of the ICT programmes is flawed. The same reasoning applies here.

19. Further, Brazil's claim, asserting that the Panel erred in the interpretation and application of Article 1.1(a)(1)(ii) of the SCM Agreement when finding that the possible cash availability and implicit interest income that the Brazilian Government could earn (had the tax been paid in advance and not offset) constitute a kind of revenue otherwise 'due' by the taxpayer, is flawed.

20. Brazil also argues that the Panel failed to properly assess a table it submitted to show that the PEC and RECAP programmes were indeed providing subsidies to credit-accumulating companies and not in contingency with exportation under Article 3.1(a) of the SCM Agreement, and thus acted inconsistently with Article 11 of the DSU. However, it is very clear that the Panel performed an objective assessment consistent with that provision. Brazil's argument should be rejected because it is effectively seeking a de novo review of the facts, but has failed to show any failure by the Panel to meet the standard reflected in Article 11 of the DSU.

VIII. THE PANEL PROPERLY FOUND THAT THE TAX REDUCTION ACCORDED TO IMPORTS FROM ARGENTINA, MEXICO AND URUGUAY UNDER INOVAR-AUTO IS NOT JUSTIFIED UNDER PARAGRAPHS 2(b) OR 2(c) OF THE ENABLING CLAUSE

21. Paragraph 4 of the Enabling Clause in particular, explicitly requires the Member taking an action under the Enabling Clause to notify, *ex ante*, other Members. A complaining party cannot be obliged to invoke the Enabling Clause in its panel request, unless it is appropriately notified. If the Appellate Body were to follow Brazil's overly broad approach, this would effectively mean that Members are required to simply guess whether some past action by another Member was or was not meant as a measure taken pursuant to the Enabling Clause.

22. Brazil also takes issue with the Panel's substantive rejection of its ex post facto attempts at an Enabling Clause-based justification. However, Brazil does not even assert, much less establish, that INOVAR-AUTO provides for an exception with respect to such provisions. In particular, with respect to paragraph 2(c), Japan notes that Brazil's claim appears to be that the Panel did not properly analyse and weigh the facts. In addition, Japan also notes that the INOVAR-AUTO programme did not have a close and genuine link to the Treaty of Montevideo or ECAs, as the Panel correctly concluded.

23. Finally, Japan also notes, that in any event, the preferential treatment for other MERCOSUR countries and Mexico is contrary to paragraph 3 of the Enabling Clause. Pursuant to paragraph 3(a), measures taken under the Enabling Clause "shall be designed to facilitate and promote the trade of

- 29 -

developing countries and not to raise barriers or to create undue difficulties for the trade of any other contracting parties".

IX. THE PANEL'S RECOMMENDATION THAT BRAZIL WITHDRAW THE ALLEGED PROHIBITED SUBSIDIES IN 90 DAYS IS CONSISTENT WITH ARTICLES 11 AND 12.7 OF THE DSU

24. With its final ground for appeal, Brazil submits that the Panel erred in recommending 90 days as the appropriate time period for Brazil to withdraw the prohibited subsidies. At the outset, Japan notes that panels do not need to address in their reports every argument or evidence raised by the parties. Similarly, panels must set forth explanations and reasons sufficient to disclose the essential, or fundamental, justification for those findings and recommendations, in light of Article 12.7 of the DSU. This does not, however, necessarily imply that Article 12.7 requires panels to expound at length on the reasons for their findings and recommendations.

25. In any event, Japan observes that, the Panel explicitly explained that it had taken into account the procedures that may be required to implement the recommendation on the one hand, and the requirement that Brazil withdraw its subsidies "without delay" on the other hand, when recommending that Brazil shall withdraw the subsidies identified in paragraphs 8.16(f), 8.17(f), and 8.18 within 90 days. Brazil's appeal, if anything, appears to be simply an effort to re-litigate the question.

X. CONCLUSION

26. As discussed and for the reasons set out above, Brazil's claims of error are all without merit and Japan respectfully requests the Appellate Body to reject them.

- 30 -

ANNEX B-5

EXECUTIVE SUMMARY OF THE EUROPEAN UNION'S APPELLEE'S SUBMISSION¹

1. The European Union considers that overall the Panel made balanced findings in relation to the challenged programmes as a whole as well as several specific elements thereof, on issues which, to certain extent, are a *déjà vu* for the GATT/WTO system: the use of discriminatory taxation and local content requirements as a tool to boosting industrial policy development, coupled with tax reliefs for companies that achieve an export threshold. Far from disputing each and every legal finding made by the Panel, Brazil appears to concede some of these findings and conclusions by not raising an appeal against them. In contrast, Brazil's appeal is premised on seeking to spot weakness in how the Panel interpreted and applied the law to the relevant facts, whilst in reality, the Panel's assessment is reasoned and reasonable. At times, Brazil also appears to question how the Panel weighted all the evidence on specific matters without raising a claim under Article 11 of the DSU. The European Union considers that, whilst Brazil may disagree with the Panel's appreciation of the facts and ultimate conclusions, Brazil has failed to show the existence of any reversible error.

1.1 Article III:8(b) as a threshold issue

2. The Panel properly interpreted and applied Article III:8(b) of the GATT 1994 as not excluding *per se* from the disciplines of Article III subsidies granted to domestic producers. Article III:8(b) of the GATT 1994 exempts payments to domestic *producers* from the national treatment obligation, but only to the extent that e.g. those payments do not discriminate between domestic and imported *goods*. Brazil's formalistic attempt to isolate the disciplines under the SCM Agreement from the national treatment rules under Article III of the GATT 1994 should be rejected as it goes against the principle that the covered agreements should be interpreted in a coherent and consistent manner, giving meaning to all applicable provisions harmoniously.

1.2 ICT programmes

1.2.1 Article III:2 of the GATT 1994

With the first claim of appeal under Article III: 2 of the GATT Brazil in substance repeats its 3 line of argument that the Panel erred when finding that domestic production subsidies do not fall per se outside the purview of the disciplines of Article III (including paragraph 2 of that Article) and it adds a few more arguments. The European Union notes first that the expression "domestic production subsidies" is not contained in the WTO treaties, which shows that Brazil's claim in predicated on an undefined concept, perhaps in the hope that the Appellate Body will make the case for Brazil. Second, the European Union notes that the Panel found that Article III:8(b) does not carve out from the disciplines of Article III: 2 subsidies provided exclusively to domestic producers per se, since aspects of a subsidy resulting in product discrimination may fall within that provision. Hence the Panel did not need to discuss whether or not Article III:8(b) of the GATT 1994 applies to the provision of subsidies in the form of reductions in indirect taxes. Therefore, the Appellate Body does not need to examine Brazil's allegation that term "subsidies" in Article III:8(b) of the GATT 1994 encompasses all types of subsidies included in Article 1.1 of the SCM Agreement, as that legal issue is not covered by the Panel's findings. In any event, the European Union submits that Brazil's argument is also incorrect as a matter of law.

4. Brazil's second claim of appeal under Article III: 2 of the GATT relates to the findings of the Panel concerning the taxation of finished incentivised products as opposed to like imported finished products. However, Brazil's claim is predicated on a number of false premises, which are contradicted by factual findings of the Panel and what Brazil conceded before the Panel, which Brazil does not challenge. The Panel analysis of the parties arguments concerning incentivised finished products under Article III: 2, first sentence, of the GATT 1994 reflects the rules on the allocation of the burden of proof in WTO law. Hence, Brazil's claim boils down to criticizing the Panel for not having made the case for Brazil. In any case, Brazil's claim is wrong on substance: a lower tax rate imposed on

¹ Total number of words of the executive summary = 3744.

incentivised finished products (i.e. domestic) results in a lower fiscal burden for those products compared to like imported finished products.

5. Brazil's third claim of appeal under Article III: 2 of the GATT is in essence that the Panel erred in finding that the "cash-flow" and the "time-value of money" effects under the ICT programmes result in a higher tax burden on imported intermediate products compared to like incentivised intermediate domestic products. Brazil's claim of appeal is however based on a wrong reading of the Panel Report. First, nowhere the Panel agreed with Brazil that the nominal amount of taxes is decisive, but it clearly indicated that it was concerned with determining the effective tax burden and for that purpose it took into account the functioning of the credit-debit system governing Brazil's direct taxes at issue. The Panel found that the application of the rule of credits and debits for purchases of imported (and, therefore, non-incentivized) intermediate ICT products involves the payment of a tax that is not faced by companies purchasing incentivized intermediate domestic ICT products. This payment of the tax upfront has the effect of limiting the availability of cash flow by companies purchasing imported intermediate ICT products and results in a higher effective tax burden on these products. Moreover, imported intermediate ICT products are subject to a higher tax burden than like incentivized domestic intermediate ICT products, due to depreciation in the value of the tax credits over time. The Panel's analysis reflects elementary concepts of economics or rather simple common sense considerations and is confirmed by WTO case law. Moreover, Brazil does not assert that the Panel failed to make an objective assessment of the facts when describing the functioning of the credit-debit system governing Brazil's direct taxes at issue.

1.2.2 Article III:4 of the GATT 1994

6. The first claim of appeal by Brazil under Article III: 4 of the GATT relate again to Brazil's reading of Article III: 8(b) as exempting domestic production subsidies *per se* from the disciplines of Article III (including obviously Article III: 4). The European Union has already explained that this reading Article III: 8(b) is unwarranted and will not repeat the same reasoning.

7. With its second claim of appeal under Article III: 4 of the GATT, Brazil argues in substance that the Panel conflated tax discrimination under Article III: 2 with regulatory discrimination under Article III: 4. However, a careful reading of the Panel Report is sufficient to reject Brazil's claim. Indeed, the Panel clearly recalled the Appellate Body jurisprudence which admits that the same measure, or certain aspects of the same measures, can fall both within Article III: 4 and Article III: 2. The Panel hence analysed the complainants claims in detail and checked weather the complainants demonstrated that the challenged ICT programmes met all the conditions for finding a violation of Article III: 4, which Brazil does not seriously contest. The fact that one instance of less favourable treatment accorded to imported products results from the different tax burden imposed on those products vis-à-vis like domestic products is simply a consequence of the characteristics of the regulatory measure examined by the Panel. At the same time, it is entirely possible that the same measure or certain aspects of the same measure also violate Article III: 2, without that demonstrating that the Panel made any error in its analysis.

With its third claim of appeal under Article III: 4 of the GATT, Brazil contests the Panels finding 8. that requirements related to production of a product may fall within Article III: 4 of the GATT, because according to Brazil they would fall under Article III:8(b) of the GATT. However, Brazil does not appeal Section 7.2.1 of the Report, where the Panel found that that the scope of Article III:4 extends to measures presented in the form of requirements for producers as opposed to direct requirements on products. Brazil does not even formulate any argument trying to distinguish this case from a long line of cases, which confirms the Panel's findings. Therefore, Brazil's appeal appears pointless as it is limited to affirming some errors in the Panel Report without challenging the legal and factual premises for those findings. For the rest Brazil is making some undemonstrated assertions and ultimately is subsuming its first general defence before the Panel about "pre-market" requirements in its second general defence according to which the challenged ICT programmes are exempted from the disciplines of Article III, by virtue of Article III:8(b). Hence, Brazil's arguments are based again on the wrong premise that, just because Brazil grants subsidies only to domestic producers, it is entitled to introduce any differences in treatment between domestic and foreign producers and products together with those subsidies.

9. With the fourth claim of appeal under Article III:4 of the GATT, Brazil challenges the Panel findings that the challenged ICT programmes result in a heavier administrative burden on imported product inconsistent with Article III:4. However, the Panel reasoning is firmly grounded on the

- 32 -

functioning of the credit-debit system as described in the Panel Report and on what Brazil acknowledged before the Panel. At the same time. Brazil does not even argue (let alone demonstrate) that the Panel violated Article 11 of the DSU by failing to make an objective assessment of the facts of the case when it described the functioning of the Brazilian tax system and the credit-debit mechanism.

10 With its fifth claim of appeal under Article III:4 of the GATT, Brazil challenges the Panel's finding that the PPBs and other production step requirements under the ICT programmes are inconsistent with Article III: 4 of the GATT 1994 because they require the use of domestic products to the detriment of imported products, when accredited companies outsource the production of components and subassemblies to be used in the production of incentivised products. The European Union thinks that the Panel's findings concerning the so called "outsourcing scenario" are correct. The complainants claimed before the Panel that the challenged ICT programmes impose on the companies accredited for the production of a given incentivised product (intermediate or finished) a condition to use components and subassemblies produced in Brazil (hence domestic products) in order to obtain a tax advantage/subsidy in relation with the production of the incentivised product. This condition or requirement results from the production step requirements (contemplated by the ICT programmes), which are specific to each incentivised product, and which must be complied with in order for a company to be accredited and be eligible for the tax advantages in relation to that product. Some of those production steps require the manufacturing in Brazil of components or subassemblies for the incentivised product. Hence, those components or subassemblies are domestic products that must be used in the production of the incentivised product.

11. Brazil confirms that the conditions for companies to obtain accreditation define the condition of eligibility for the tax advantages provided for under the challenged ICT programmes and it does not claim that the Panel made whatever error in the description of the functioning of the PPBs or the production steps requirements more generally. Brazil also confirms that the production steps requirements or subassemblies in Brazil and it does not question that these products should be considered as domestic.

12. Accordingly, the Panel did not err in finding that the certain production step requirements under the challenged ICT programme require that domestic components and subassemblies be incorporated into the incentivised (finished or intermediate product), as a condition for receiving the tax advantages in relation with that product. Given that accredited companies are required to use domestic components if they want to benefit of tax advantages in relation to their incentivised products, this condition or requirement distorts the conditions of competition between domestic components and subassemblies and like imported components and subassemblies and therefore accords less favourable treatment to imported components and subassemblies.

13. Brazil's argument that under the ICT programmes accredited producers are free to source inputs from foreign producers provided that the relevant production steps are undertaken in Brazil is simply missing the point, since the claim adjudicated by the Panel does not encompass raw materials and other basic components, but the domestic components and subassemblies that result from the performance in Brazil of certain production steps.

14. Brazil is also wrong in arguing that the Panel's conclusions imply that the only national production process that could be subsidized in a WTO-consistent manner is the assembly of inputs, not their production.

15. It follows from the above that Brazil's claims of appeal under Article III: 4 of the GATT (to the exception of the claim based on the interpretation of Article III: 8(b)) do not concern the Panel's findings contained in Section 7.3.2.1.4.2 relating to the mechanism for the calculation of the amount of resources required to be invested in R&D under the Informatics and PADIS programmes. Indeed, none of the arguments formulated by Brazil take issue with anything the Panel found in that Section. Hence, it must necessarily be concluded that Brazil does not challenge the Panel's finding in paragraph 7.243 of the Report, according to which the mechanisms under the Informatics and PADIS programmes for the calculation of the amount of resources required to be invested in R&D accord to imported products treatment less favourable than that accorded to like domestic products, and thus are inconsistent with Article III: 4 of the GATT 1994.

- 33 -

16. In light of the foregoing, the European Union submits that Brazil has failed to show the existence of any reversible error when the Panel found that the ICT programmes were inconsistent with Article III: 4 of the GATT 1994.

1.2.3 Article 1.1(a)(1)(ii) of the SCM Agreement

17. Brazil wrongly argues that the Panel made errors when finding that the ICT programmes amounted to prohibited subsidies under Article 3.1(b) of the SCM Agreement. The Panel properly interpreted and applied Article 1.1(a)(1)(ii) of the SCM Agreement when finding that the tax exemptions, suspensions and reductions granted under the ICT programmes amounted to "government revenue that is otherwise foregone or not collected". In fact, the Panel carried out a balanced assessment when comparing the tax treatment of similarly situated taxpayers in accordance with Brazil's general and predominant rules of taxation in relation to non-accredited companies.

1.2.4 Article 3.1(b) of the SCM Agreement

With this claim of appeal Brazil argues that the Panel erred in finding that the tax exemptions, 18. reductions and suspensions granted under the ICT programmes are prohibited import substitution subsidies within the meaning of Article 3.1(b) of the SCM Agreement. However, the Panel defined and applied the correct legal standard under Article 3.1(b) of the SCM agreement. In line with the Appellate Body's findings in US - Tax Incentives, the Panel focussed its analysis from the very beginning on the conditions for companies to obtain accreditation and thereby be eligible for the tax exemptions, reductions or suspensions under the programmes. The Panel stressed that element of conditionality all along its analysis. Therefore, it is not correct to say that the Panel based its findings on the effects of the subsidies, or on the fact that accredited companies would likely use more domestic products in their downstream production activities. The Panel focused on the conditions of eligibility for the subsidies, and found that those conditions impose the use of domestic components and subassemblies, under the guise of production steps requirements, in order to obtain the subsidy in relation to the production of the incentivised products. At the same time there is no dispute that the components and subassemblies at issue in the present case are domestic because they are produced in Brazil (regardless of the origin of their raw materials or inputs).

19. It is also worth recalling that there are some key differences between this case and the situation in *US* – *Tax Incentives* and the Appellate Body itself clarified that whether a subsidy is conditional upon the use by the subsidy recipient of domestic over imported goods, should be assessed on a case-by-case basis. The production steps requirements at issue in the present case are not limited to simply specifying the location of an assembly operation of the incentivised product, rather they concern the production in Brazil of component and subassemblies of the incentivised product. Moreover, those subsidies are not granted because the accredited companies are engaged in the production of the components of the incentivised product, but they are explicitly made contingent upon the use of certain components or subassemblies made in Brazil (by the accredited company or a third company). Hence, this case (unlike the *US-Tax Incentives* case) concerns subsidies granted exclusively for the production of the incentivised products (the final goods) and not for the production of their inputs, too. Finally, the components that are manufactured in Brazil in order to comply with the production step requirements of the ICT programmes must, by definition, be used in the production of the incentivised products.

20. In nutshell all of Brazil's claims of appeal in respect of the Panel's findings with regard to the imposition of local content requirements boil down to say that if a requirement for granting a subsidy is drafted in terms of production activity, it is indifferent, for the purpose of Article 3.1(b) of the SCM Agreement, if such production activity results in the creation of a domestic component that *must be used* in the production of the incentivised product. As Brazil would have it, Article 3.1(b) of the SCM Agreement would prohibit Brazil from requiring accredited companies to "use *domestic* component and subassemblies" when producing incentivised products. However, it would leave it completely free to require the same companies to "use component and subassemblies *produced in Brazil*" when producing incentivised products as a substant this cannot be the correct interpretation of Article 3.1(b) of the SCM Agreement.

- 34 -

1.3 INOVAR-AUTO PROGRAMME

21. In essence, Brazil claims that the Panel erred by excluding the application of Article III:8(b) of GATT 1994 and when it found that the accreditation requirements are inconsistent with Article III:4. It also maintains that the Panel erred in finding that the different processes for companies seeking accreditation as "domestic manufacturers" and companies seeking accreditation as "investors" or "importers/distributors" would *per se* result in less favourable treatment being accorded to imported finished motor vehicles, under Article III:4 of GATT 1994. The European Union will explain below why all those claims should be rejected by the Appellate Body.

22. Brazil's arguments should be rejected entirely. Far from just being a subsidy programme exclusively for domestic producers, the INOVAR-AUTO programme discriminates between domestic and imported vehicles. Foreign companies seeking accreditation under the INOVAR-AUTO programme should opt for becoming "importers/distributors" by establishing in Brazil in order to carry out certain marketing and sale activities in the Brazilian market for the imported vehicles to benefit from the same presumed IPI credits granted to domestic vehicles produced by domestic manufacturers or investors in Brazil accredited under the programme. Therefore, the Panel properly found that the accreditation requirements under the INOVAR-AUTO modify the conditions of competition to the detriment of imported motor vehicles and in favour of like domestic motor vehicles.

1.4 PEC and RECAP programmes

Brazil raises four claims against the Panel's findings that the PEC and RECAP programmes 23. amounted to prohibited export subsidies under Article 3.1(a) of the SCM Agreement. In particular, Brazil disputes the Panel's conclusions that the tax treatment granted through the PEC and RECAP programmes amounted to "revenue otherwise due that is foregone or not collected", as well as the Panel's conclusions on export contingency. Brazil's four claims of error should be rejected. The Panel properly applied the test developed by the Appellate Body in US - Large Civil Aircraft (2nd Complaint) when, having examined the structure and organising principles of Brazil's tax system, the Panel found that the benchmark for comparison was the tax treatment of non-registered/non-accredited companies under the PEC/RECAP programmes, as similarly situated taxpayers in comparable transactions. The Panel did not err either when examining the tax treatment of those taxpayers including the rather frequent scenario where there is tax credit accumulation. The Panel properly interpreted and applied Article 1.1(a)(1)(ii) of the SCM Agreement when finding that the cash availability and the implicit interest income that the Brazilian Government could earn (had the tax been paid in advance and not offset) fall under the terms "revenue otherwise due is foregone or not collected". Finally, the European Union considers that the Panel properly weighted all the evidence in accordance with its duties under Article 11 of the DSU when finding that the PEC/RECAP programmes were contingent upon export performance, in particular in view of the clear requirement for companies to have more than 50% of their revenues derived from exports.

1.5 Enabling Clause

24. The European Union considers that Brazil's claims of error with respect to the Panel's findings on the Enabling Clause should be rejected by the Appellate Body. First, the Panel correctly found that a complaining party cannot be under an obligation to invoke the Enabling Clause in its panel request, unless that complaining party is appropriately informed that the responding party considers the challenged measure to have been adopted pursuant to (and justified under) the Enabling Clause. Second, the Panel correctly established that the conditions in paragraphs 2(b) and 2(c) of the Enabling Clause are not met by the Brazilian measures in the present case. In addition, whilst Brazil complaints about the Panel's assessment and characterisation of the facts (such as the Panel's finding that the accreditation process under the INOVAR-AUTO programme is more burdensome to foreign manufacturers than to domestic ones), Brazil has no claim under Article 11 of DSU with respect to the Panel's assessment of Brazil's invocation of the Enabling Clause under paragraph 2(c).

1.6 Article 4.7 of the SCM Agreement

25. The European Union maintains that, contrary to Brazil's allegations, the Panel did not make an error under Articles 11 or 12.7 of the DSU when recommending, pursuant to Article 4.7 of the SCM Agreement, that Brazil withdraw the prohibited subsidies found within 90 days. The Panel's - 35 -

rationale and explanations are sufficiently evident from the Panel Report. Brazil's disagreement on substance should not be construed as a reversible error.

- 36 -

ANNEX B-6

EXECUTIVE SUMMARY OF BRAZIL'S APPELLEE'S SUBMISSION¹

1. Brazil requests that the Appellate Body reject all of the claims of error raised by the European Union and Japan in their other appeals, for the following reasons:

2. First, the complainants failed to establish that the Panel acted inconsistently with Article 11 of the DSU in failing to address their *arguments* that in the "in-house" scenario the ICT programmes and INOVAR-AUTO give rise to a contingency on the use of domestic over imported goods under Articles III:4 of the GATT 1994, Article 2.1 of the TRIMs Agreement, and Article 3.1(b) of the SCM Agreement. The Panel's decision not to address the complainants' *arguments* in this respect could not be properly characterized as exercise of judicial economy, which occurs when a panel decides not to address multiple *claims* raised by a WTO Member in respect of the same measure. The Panel clearly examined and made findings in respect of all of the legal claims raised by the complainants in respect of the ICT programmes and INOVAR-AUTO. In any event, the Panel acted within the bounds of its discretion under Article 11 of the DSU in deciding not to address the complainants' *arguments* that in the "in-house" scenario the ICT programmes and INOVAR-AUTO allegedly gave rise to a requirement to use domestic over imported goods.

3. Second, the European Union's "subordinate claim" that the Panel erred in its application of Articles III:4 of the GATT 1994, Article 2.1 of the TRIMs Agreement, and Article 3.1(b) of the SCM Agreement by failing to address arguments concerning the "in-house" scenario is not properly within the scope of appellate review. The European Union's allegation that the Panel failed to consider "all relevant facts of the case" is directed at the Panel's appreciation of the facts, and therefore should have been brought under Article 11 of the DSU. The European Union's failure to bring its "subordinate" appeal claim under Article 11 of the DSU is dispositive, and for this reason alone should be rejected by the Appellate Body.

4. Third, the complainants remain unable to establish that in the "in-house" scenario, both the ICT programmes and INOVAR-AUTO are inconsistent with Article III:4 of the GATT 1994, Article 2.1 of the TRIMs Agreement, and Article 3.1(b) of the SCM Agreement. The complainants' interpretation that these provisions prohibit subsidies contingent upon domestic production in-country renders Article III:8(b) of the GATT 1994 entirely meaningless. Moreover, the complainants' arguments in respect of both the "in-house" and "outsourcing" scenarios are based on the proposition expressly rejected by the Appellate Body in *US* – *Tax Incentives* that subsidies contingent on producing certain goods domestically *per se* constitute prohibited import-substitution subsidies.

¹ Total word count: 463

- 37 -

ANNEX C

ARGUMENTS OF THE THIRD PARTICIPANTS

	Contents	Page
Annex C-1	Executive summary of Argentina's third participant's submission	38
Annex C-2	Executive summary of Australia's third participant's submission	39
Annex C-3	Executive summary of the United States' third participant's submission	40

- 38 -

ANNEX C-1

EXECUTIVE SUMMARY OF ARGENTINA'S THIRD PARTICIPANT'S SUBMISSION

1. Argentina has a strong interest in the proper interpretation of the Enabling Clause and, in particular, in the development of an effective and coherent approach to interpreting its paragraphs.

2. Argentina will refer to the following issues: (1) Whether the Panel erred in finding that the claims under Article I:1 of the GATT 1994 were properly included in the terms of reference; (2) Whether the challenged treatment accorded to imports from Argentina, Mexico and Uruguay under INOVAR-AUTO is justified under paragraph 2(b) of the Enabling Clause; (3) Whether the challenged treatment accorded to imports from Argentina, Mexico and Uruguay under INOVAR-AUTO is justified under paragraph 2(c) of the Enabling Clause.

3. In particular, Argentina considers that the differential treatment accorded through Decree 7,819/2012 is *part of* a notified regional arrangement and therefore does not require additional notification.¹ As *part of* that notified arrangement, Argentina considers that the complainants should have been on notice that Brazil could invoke the Enabling Clause as a justification for the differential treatment granted under INOVAR-AUTO notwithstanding the provisions of Article I:1 of the GATT 1994.

4. Argentina shares the view of Brazil that the Panel's interpretation renders paragraph 2(b) useless, since it would limit its provisions to what is already established in the covered agreements. Hence, Argentina considers that the analysis under this paragraph should be reversed and completed by the Appellate Body.

5. Regarding the "criteria or conditions for the mutual reduction or elimination of non-tariff measures", paragraph 2(c) uses the phrase "may be prescribed". Argentina considers that a correct interpretation of this phrase would result in concluding that it does not constitute a positive obligation. The fact that in the last 38 years WTO Members have not established such criteria and conditions demonstrated that this phrase cannot be interpreted as a prohibition to apply the provisions in paragraph 2(c) related to non-tariff measures.

¹ See Brazil's second written submission, para. 181; see also Brazil's first written submission (DS472), paras. 638-639.

- 39 -

ANNEX C-2

EXECUTIVE SUMMARY OF AUSTRALIA'S THIRD PARTICIPANT'S SUBMISSION

Australia considers that the Panel was correct in finding that, while subsidies provided 1. exclusively to domestic producers under Article III:8(b) of the GATT 1994 do not per se violate Article III: 2,¹ such a subsidy could still violate Article III: 2 if designed and applied in a manner that discriminates between imported and domestic products.²

A panel is therefore not required to undertake a 'threshold inquiry' as to whether a subsidy 2. is product-related or production-related because the disciplines of Article III apply to both types of subsidies.³

3. Australia shares the concerns of Japan and the EU that the Panel's use of judicial economy casts doubt on whether the Panel found that the measures in their entirety, or merely as they operated in the outsourcing scenario, violated Brazil's WTO obligations.

4. In Australia's view, a panel's mandate under the DSU to make an objective assessment of the matter before it - with a view to its positive resolution - limits the degree to which a panel can exercise judicial economy; and requires a panel to ensure its findings can lead to a satisfactory settlement of the dispute.

¹ Panel Report, *Brazil – Taxation*, para. 7.77 ² Panel Report, *Brazil – Taxation*, paras. 7.85 and 7.88.

³ Panel Report, Brazil - Taxation, para. 7.88.

- 40 -

ANNEX C-3

EXECUTIVE SUMMARY OF THE UNITED STATES' THIRD PARTICIPANT'S SUBMISSION

I. GATT 1994 ARTICLE III:8(B)¹

1. A measure conditioning an advantage, such as a subsidy, on the use of a domestic good would breach Article III: 4. In light of Article III: 8(b), Article III: 4 must be interpreted as not prohibiting the provision of subsidies *because of* the exclusion of foreign producers as eligible subsidy recipients. Such a payment should not be understood, on the basis of this exclusion alone, to be in conflict with Article III: 4 (including its requirement of national treatment for measures affecting the "use" of a product).

2. SCM Agreement Article 3.1(b) and GATT 1994 Article III share similar disciplines. Subsidies contingent on the use of domestic over imported products would be inconsistent with *both* provisions. The Panel thus correctly analyzed the disputed programs under both Article III and Article 3.1(b).

3. While Members must be free to define the domestic producers who are to receive subsidies, including the productive activities that make them a domestic producer, we consider that subsidies conditioned not solely on domestic production, but also the use of domestic products, fall outside Article III:8(b)'s derogation.

II. GATT 1994 ARTICLE III:4

4. The Panel found that "nested" PPBs—which require the use of inputs that themselves conform to another PPB—contain an explicit requirement to use domestic over imported goods. Brazil has not appealed these findings. Such a requirement is inconsistent with Article III: 4.

III. SCM AGREEMENT ARTICLE 3.1(B)

5. With respect to intermediate goods, we note that if a measure *exempts* taxes, government revenue, otherwise due, is clearly foregone under SCM Agreement Article 1.1(a)(1)(ii). If a measure *suspends* taxes that are later paid, a financial contribution is still provided: at the time when government revenue would otherwise be due, it is foregone (albeit temporarily). In addition, implicit interest may also be foregone.

6. Brazil has not appealed the Panel's factual findings regarding the disputed programs. Those facts show that the tax advantages under the programs are contingent on the use of domestic over imported goods under Article 3.1(b) with respect to nested PPBs.

¹ This executive summary contains a total of 382 words (including footnotes), and the U.S. third participant submission contains 3831 words (including footnotes).

- 41 -

ANNEX D

PROCEDURAL RULING

	Contents	Page
Annex D-1	Procedural ruling of 12 October 2017 regarding the deadline for filing of third participants' submissions	42

- 42 -

ANNEX D-1

PROCEDURAL RULING OF 12 OCTOBER 2017

1. On 28 September 2017, Brazil filed a Notice of Appeal in the above proceedings. In accordance with Rule 26 of the Working Procedures for Appellate Review (Working Procedures), a Working Schedule for Appeal was drawn up by the Appellate Body Division hearing this appeal and communicated to the participants and third parties on 29 September 2017.

2. On 9 October 2017, the Division received a joint letter from Argentina, Australia, Canada and the United States requesting an extension of the deadline for the filing of their third participants' submissions in these proceedings (the joint request). These third participants explained that some of them are also third parties in another pending appellate proceeding, namely, *Indonesia – Safeguard on Certain Iron or Steel Products* (DS490 / DS496), and that the third participants' submissions in that proceeding are due on the same day as the deadline for the filing of their third participants' submissions in the present proceedings, i.e. 19 October 2017. In their joint request, the four third participants also highlighted the length of the panel report, and of Brazil's appellant's submission in these appellate proceedings. Argentina, Australia, Canada and the United States therefore requested that the deadline for the filing of the third participants' submissions be extended by one week, until 26 October 2017.

3. On 9 October 2017, we invited the participants and third parties, if they so wished, to comment on the joint request by 12 noon on 11 October 2017. We received no objections to the joint request. Brazil indicated that it did not oppose the request and Japan stated it had no comments.

4. We consider the reasons identified by Argentina, Australia, Canada and the United States, in particular the fact that the deadlines for the filing of third participants' submissions in two separate appeal proceedings currently fall on the same day, to be relevant factors in our assessment of the joint request. We further note that Brazil, the European Union and Japan and the other third parties have not raised any objection to the joint request.

5. In these circumstances, the Division has decided, in accordance with Rule 16(2) of the Working Procedures, to extend the deadline for the filing of third participants' submissions by one week to <u>Thursday, 26 October 2017</u>.

Modified Dates for the Submission of Documents

Process	Rule	<u>Date</u>
Notice of Appeal	Rule 20	Thursday, 28 September 2017
Appellant's submission and executive summary	Rule 21(1)	Thursday, 28 September 2017
Notice of Other Appeal	Rule 23(1)	Tuesday, 3 October 2017
Other appellant's submission and executive summary	Rule 23(3)	Tuesday, 3 October 2017
Appellee's submission(s) and executive summary(ies)	Rules 22 and 23(4)	Monday, 16 October 2017
Third participants' submissions and executive summaries	Rule 24(1)	Thursday, 26 October 2017
Third participants' notifications	Rule 24(2)	Thursday, 26 October 2017